## UNITED STATES BANKRUPTCY COURT NORTHERN DISTRICT OF OHIO EASTERN DIVISION

In re:	)
CEP HOLDINGS, LLC, et al.,1	) Case No. 06-51848 ) (Jointly Administered)
Debtors.	) Chapter 11
Deptors.	) Honorable Marilyn Shea-Stonum

## **AFFIDAVIT OF SERVICE**

I, James H. Myers, state as follows:

- 1. I am over eighteen years of age and I believe the statements contained herein are true based on my personal knowledge. I am employed by BMC Group, Inc., the Notice, Claims and Balloting Agent for the Debtors and Debtors-in-Possession in the above-captioned cases, whose business address is 1330 East Franklin Avenue, El Segundo, California 90245.
- 2. At the direction of Baker & Hostetler LLP, Counsel for the Debtors and Debtors-in-Possession, copies of the documents attached hereto as Exhibits "1" through "3" and Exhibit "4" were sent to those persons listed on the Service Lists attached hereto as Exhibits "A" through "E" at the addresses shown thereon. All parties were served via first-class mail and deposited with the United States Postal Service with postage thereon fully prepaid. Except as noted in Exhibit "A", service commenced on June 13, 2007 and concluded on June 14, 2007:

///

<sup>&</sup>lt;sup>1</sup> The Debtors include: CEP Holdings, LLC, Creative Engineered Polymer Products, LLC and Thermoplastics Acquisition, LLC.

SOLICITATION INFORMATION BOOK FOR CEP HOLDINGS, LLC Attached hereto as ET AL. (comprised of)

Exhibit "1"

- Notice of (A) Deadline for Casting Votes to Accept or Reject Proposed Joint Plan of Liquidation, (B) Hearing to Consider Confirmation of Proposed Joint Plan of Liquidation and (C) Related Matters;
- Letters Recommending Acceptance of the Plan
- First Amended Disclosure Statement to Accompany First Amended Joint Plan of Liquidation Under Chapter 11 of the Bankruptcy Code Proposed by the Debtors and the Official Committee of Unsecured Creditors Dated May 25, 2007:
  - Exhibit A: First Amended Joint Plan of Liquidation Under Chapter 11 of the Bankruptcy Code Proposed by the Debtors and the Official Committee of Unsecured Creditors Dated May 25, 2007, with Plan Exhibit 1, the Trust Agreement
  - Exhibit B: Order (A) Approving Proposed Disclosure Statement, (B) Establishing Procedures for Solicitation and Tabulation of Votes to Accept or Reject Proposed Joint Plan of Liquidation and (C) Scheduling Certain Dates in Connection Therewith
  - Exhibit C: Distribution Analysis

NOTICE OF (A) DEADLINE FOR CASTING VOTES TO ACCEPT OR Attached hereto as REJECT PROPOSED JOINT PLAN OF LIQUIDATION, (B) HEARING TO CONSIDER CONFIRMATION OF PROPOSED JOINT PLAN OF LIQUIDATION AND (C) RELATED MATTERS

Exhibit "2"

BALLOT[S] FOR ACCEPTING OR REJECTING THE FIRST AMENDED JOINT PLAN OF LIQUIDATION OF CEP HOLDINGS, LLC AND ITS AFFILIATED DEBTORS AND DEBTORS IN POSSESSION CLASS 4: GENERAL UNSECURED CLAIMS

Sample attached hereto as Exhibit "3"

PRE-ADDRESSED RETURN ENVELOPE [no postage attached] [Sample not attached hereto]

Exhibit "4"

## Exhibit "A" Exhibit "1" was served on

- Class 1 Subordinated Participating Customers Secured Claimants
- Class 2 Other Secured Claimants
- Class 3 Priority Non-Tax Claimants
- Class 5 Equity Interest Holders
- Administrative Expense Claimants
- Compensation and Reimbursement Claimants
- Wachovia Secured Claimant
- Priority Tax Claimants
- Governmental Units listed in the Schedules
- Parties to Executory Contracts and Unexpired Leases
- The Securities and Exchange Commission
- The Core Group, parties listed on the General Service List established by the Case Management Order, those parties who have requested special notice and the Office of The United States Trustee
- Exhibit "B" Exhibit "2" was served on parties listed in the Debtors' Schedules as holding contingent, unliquidated or disputed clams that did not file a timely proof of claim, and on general matrix parties that are not otherwise included in Exhibits "A", "C", "D" or "E"
- Exhibit "C" Exhibits "1", "3A" and "4" were served on the Class 4 General Unsecured Claimants

Exhibit "D" Exhibits "1", "3B" and "4" were served on the Class 4 General Unsecured Claimants

Exhibit "E" Exhibits "1", "3C" and "4" were served on the Class 4 General Unsecured Claimants

DATED: June / , 2007 El Segundo, California

James H. Myers

State of California ) ss County of Los Angeles )

Personally appeared before me on June 1, 2007, James H. Myers, an individual, known to me to be the person who executed the foregoing instrument and acknowledged the same.

JAMES F. BARTLETT
Comm. # 1523213
NOTARY PUBLIC-CALIFORNIA
Los Angeles County
My Comm. Expires Oct. 30, 2008

# **EXHIBIT 1**

## **SOLICITATION INFORMATION**

## CEP HOLDINGS, LLC, ET AL.

DIRECT ALL INQUIRIES TO VOTING AGENT: BMC GROUP, INC. TELEPHONE: (888) 909-0100 WEBSITE: www.bmcgroup.com/cep

## UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF OHIO, EASTERN DIVISION

In re:	) Case No. 06-51848
	) (Jointly Administered)
CEP HOLDINGS, LLC, <u>et</u> <u>al</u> ., <sup>1</sup>	) Chapter 11
	)
Debtors.	) Honorable Marilyn Shea-Stonum
	) United States Bankruptcy Judge

## **IMPORTANT DATES**

June 6, 2007 Voting Record Date: July 13, 2007 at 5:00 p.m. Eastern Time Voting Deadline: July 13, 2007 at 4:00 p.m. Eastern Time Confirmation Objection Deadline: July 24, 2007 at 1:30 p.m. Eastern Time Hearing to Confirm the Plan:

<u>CC</u>	ONTENTS		<b>PAGE</b>
•		Deadline for Casting Votes to Accept or Reject Proposed Joint Plan of Liquidation, (B) asider Confirmation of Proposed Joint Plan of Liquidation and (C) Related Matters	1
•	Letters Recomi	mending Acceptance of the Plan.	9
•	Chapter 11 of t	Disclosure Statement to Accompany First Amended Joint Plan of Liquidation Under he Bankruptcy Code Proposed by the Debtors and the Official Committee of Unsecured d May 25, 2007.	15
	o Exhibit A:	First Amended Joint Plan of Liquidation Under Chapter 11 of the Bankruptcy Code Proposed by the Debtors and the Official Committee of Unsecured Creditors Dated May 25, 2007	75
		■ Plan Exhibit 1: Trust Agreement	113
	o Exhibit B:	Order (A) Approving Proposed Disclosure Statement, (B) Establishing Procedures for Solicitation and Tabulation of Votes to Accept or Reject Proposed Joint Plan of Liquidation and (C) Scheduling Certain Dates in Connection Therewith	141
	o Exhibit C:	Distribution Analysis.	153

## BAKER & HOSTETLER LLP

Joseph F. Hutchinson, Jr., Esquire Thomas M. Wearsch, Esquire Eric R. Goodman, Esquire 3200 National City Center 1900 East 9th Street Cleveland, Ohio 44114

Telephone: (216) 621-0200 Facsimile: (216) 696-0740

Counsel to the Debtors and Debtors-in-Possession

#### MCGUIREWOODS LLP

Mark E. Freedlander, Esquire Sally E. Edison, Esquire William C. Price, Esquire 625 Liberty Avenue 23rd Floor Dominion Tower Pittsburgh, PA 15222 Telephone: (412) 667-6000

Facsimile: (412) 667-6050

Counsel to the Official Committee of Unsecured Creditors

<sup>&</sup>lt;sup>1</sup> The Debtors are: CEP Holdings, LLC, Creative Engineered Polymer Products, LLC and Thermoplastics Acquisition, LLC.

Notice of (A) Deadline for Casting Votes to Accept or Reject Proposed Joint Plan of Liquidation, (B) Hearing to Consider Confirmation of Proposed Joint Plan of Liquidation and (C) Related Matters [THIS PAGE INTENTIONALLY LEFT BLANK]

## UNITED STATES BANKRUPTCY COURT NORTHERN DISTRICT OF OHIO EASTERN DIVISION

	X	
In re:	:	Case No. 06-51848
	:	(Jointly Administered)
CEP HOLDINGS, LLC, et al., 1	:	
	:	Chapter 11
Debtors.	:	
	:	Honorable Marilyn Shea-Stonum
	:	
	:	Hearing Date: 7/24/07 at 1:30 p.m.
	:	Objection Deadline: 7/13/07 at 4:00 p.m
	X	

## NOTICE OF (A) DEADLINE FOR CASTING VOTES TO ACCEPT OR REJECT PROPOSED JOINT PLAN OF LIQUIDATION, (B) HEARING TO CONSIDER CONFIRMATION OF PROPOSED JOINT PLAN OF LIQUIDATION AND (C) RELATED MATTERS

### PLEASE TAKE NOTICE OF THE FOLLOWING:

1. On May 25, 2007, the above-captioned debtors and debtors in possession (collectively, the "Debtors") and the Official Committee of Unsecured Creditors (the "Creditors' Committee") filed: (a) the First Amended Joint Plan of Liquidation Under Chapter 11 of the Bankruptcy Code Proposed by the Debtors and the Official Committee of Unsecured Creditors Dated May 25, 2007 (as it may be amended or modified, the "Plan"); and (b) the related First Amended Disclosure Statement to Accompany First Amended Joint Plan of Liquidation Under Chapter 11 of the Bankruptcy Code Proposed by the Debtors and the Official Committee of Unsecured Creditors Dated May 25, 2007 (Docket No. 591) (as it may be

The Debtors are: CEP Holdings, LLC, Creative Engineered Polymer Products, LLC and Thermoplastics Acquisition, LLC.

<sup>&</sup>lt;sup>2</sup> Capitalized terms used but not otherwise defined herein shall have the meanings given to them in the Plan.

amended or modified, the "Disclosure Statement") under section 1125 of title 11 of the United States Code (the "Bankruptcy Code").

- 2. Pursuant to an order of the Court dated June 6, 2007 (the "Disclosure Statement Order"), the Disclosure Statement and certain related materials (collectively, the "Solicitation Materials") have been approved for solicitation of votes to accept or reject the Plan. In accordance with the Disclosure Statement Order, certain Solicitation Materials have been provided with this Notice, including a copy of the Disclosure Statement and the Plan (including all exhibits thereto filed with the Court).
- 3. A hearing to consider the confirmation of the Plan (the "Confirmation Hearing") will be held before the Honorable Marilyn Shea-Stonum, United States Bankruptcy Judge, in the Judge's usual courtroom at United States Bankruptcy Court, 455 U.S. Courthouse, 2 South Main Street, Akron, Ohio 44308, at 1:30 p.m., Eastern Time, on July 24, 2007.
- 4. Pursuant to the Disclosure Statement Order, the Court approved certain procedures for the tabulation of votes to accept or reject the Plan. If you are the holder of a claim against one of the Debtors as of **June 6, 2007** (the record date established in the Disclosure Statement Order) in a class entitled to vote on the Plan, you have received with this Notice a ballot form (a "**Ballot**") and voting instructions appropriate for your claim. The following procedures apply with respect to voting your claim:
  - a. Except as provided in subparagraph (b) below, for your vote to accept or reject the Plan to be counted, you must complete all required information on the Ballot, execute the Ballot and return the completed Ballot to the address indicated on the Ballot so that it is received by 5:00 p.m., Eastern Time, on July 13, 2007 (the "Voting Deadline"). Any failure to follow the voting instructions included

with the Ballot or to return a properly completed Ballot so that it is received by the Voting Deadline may disqualify your Ballot and your vote.

- b. Your claim has been temporarily allowed solely for purposes of voting to accept or reject the Plan in accordance with the following tabulation rules approved by the Court in the Disclosure Statement Order (the "Tabulation Rules"):
  - i. Unless one or more of the Tabulation Rules set forth below applies and provides otherwise, a claim will be deemed temporarily allowed for voting purposes in an amount equal to: (A) the noncontingent, liquidated and undisputed amount of such claim, as set forth in the Schedules; or (B) if a proof of claim has been timely filed in respect of such claim, the noncontingent, liquidated and undisputed amount set forth in such proof of claim.
  - ii. If a claim is deemed allowed in accordance with the Plan, such claim will be temporarily allowed for voting purposes in the deemed allowed amount set forth in the Plan.
  - iii. If a claim for which a proof of claim has been timely filed and has not been disallowed is listed or marked or otherwise referenced on its face as contingent, unliquidated or disputed, either in whole or in part, only the noncontingent, liquidated and undisputed portion, if any, of such claim will be deemed temporarily allowed for voting purposes, subject to the other Tabulation Rules, and the remaining portion of such claim will be disallowed for voting purposes.
  - iv. If a claim for which a proof of claim has been timely filed is marked as a priority claim, either in whole or in part, but is listed in the Schedules as a nonpriority claim or as a priority claim only in part, such claim will be temporarily allowed for voting purposes as a nonpriority claim in an amount equal to the lesser of (A) the entire amount of such claim as set forth in the proof of claim or (B) the nonpriority claim set forth in the Schedules, provided that such claim is not listed in the Schedules or marked on the proof of claim as contingent, unliquidated or disputed.

- v. If a claim has been allowed pursuant to a stipulation approved by the Court, such claim will be deemed temporarily allowed for voting purposes in the amount set forth in the stipulation.
- vi. If a claim has been estimated or otherwise allowed for voting purposes by order of the Court, such claim will be temporarily allowed for voting purposes in the amount so estimated or allowed by the Court.
- vii. With respect to a claim as to which (A) the claim is (I) listed in the Schedules as contingent, unliquidated or disputed or in a zero amount or (II) not listed in the Schedules and (B) a proof of claim was not timely filed and no stipulation allowing the claim has been approved by the Court, such claim will be disallowed for voting purposes.
- viii. If the Debtors have filed and served an objection to a claim on or before the date that is 10 days after the date of service of the Confirmation Hearing Notice, such claim will be disallowed for voting purposes in accordance with the relief sought in the objection, unless the objection only seeks disallowance of part of such claim, in which event only the amount subject to the objection will be disallowed for voting purposes.
- ix. If a claim holder identifies a claim amount on its Ballot that is less than the amount otherwise calculated in accordance with the Tabulation Rules, the claim will be temporarily allowed for voting purposes in the lesser amount identified on such Ballot.
- c. Temporary allowance or disallowance of your Claim for voting purposes does not constitute an allowance or disallowance of your Claim for purposes of receiving distributions under the Plan and is without prejudice to the rights of the Debtors in any other context, including the right of the Debtors to contest the amount or validity of any claim for purposes of allowance and distribution under the Plan. If you wish to challenge the allowance or disallowance of your Claim for voting purposes, you must file a motion, pursuant to Bankruptcy Rule 3018(a), for an order temporarily allowing your Claim in a different amount or classification for purposes of voting to

accept or reject the Plan and serve such motion on the counsel to the Debtors so that it is received no later than 10 days after the later of (i) the date of service of this Notice and (ii) the date of service of a notice of an objection, if any, to the underlying claim. The Debtors shall have 7 days to file and serve any responses to such motions. Unless the Court orders otherwise, your Claim will not be counted for voting purposes in excess of the amount as determined in accordance with the Tabulation Rules.

- 5. Objections, if any, to the confirmation of the Plan must: (a) be in writing; (b) state the name and address of the objecting party and the nature of the claim or interest of such party; (c) state with particularity the basis and nature of any objection to the confirmation of the Plan; and (d) be filed with the Court and served so that they are received no later than **4:00** p.m., Eastern Time, on July 13, 2007 by (i) Joseph F. Hutchinson, Esq., Baker & Hostetler LLP, 3200 National City Center, 1900 East 9th Street, Cleveland, OH 44114-3485; (ii) Mark E. Freedlander, Esq., McGuireWoods LLP, 625 Liberty Avenue, 23rd Floor Dominion Tower, Pittsburgh, PA 15222; and (iii) Maria D. Giannirakis, Esq., Office of the U.S. Trustee, Howard M. Metzenbaum U.S. Courthouse, 201 Superior Ave., East, Suite 441, Cleveland, OH 44114. A pre-hearing conference on any objection to the confirmation of the Plan that is filed by the Confirmation Objection Deadline will be held on July 19, 2007 at 3:30 p.m., Eastern Time.
- 6. Requests for additional copies of the Disclosure Statement and the Plan (including all exhibits thereto filed with the Court) by parties in interest may be made in writing or orally to BMC Group, Inc., CEP HOLDINGS, LLC, et al., c/o BMC Group, PO Box 903, El Segundo, CA 90245-0903 (telephone number (888) 909-0100). Copies of the Disclosure Statement and the Plan (with all exhibits filed with the Court) also are available for review, without charge, at http://www.bmcgroup.com/cep.

7. The Confirmation Hearing may be continued from time to time without further notice, other than the announcement of the adjourned date(s) at the Confirmation Hearing or any continued hearing.

Dated: June 13, 2007 Cleveland, OH

> CEP HOLDINGS, LLC, et al., Debtors and Debtors-in-Possession

By: /s/ Joseph F. Hutchinson, Jr.
One of Their Attorneys

Joseph F. Hutchinson, Jr. (0018210) Thomas M. Wearsch (0078403) Eric R. Goodman (0076035) BAKER & HOSTETLER LLP 3200 National City Center 1900 East 9th Street Cleveland, Ohio 44114-3485

Phone: 216.621.0200 Fax: 216.696.0740

Counsel for the Debtors and Debtors-in-Possession

OFFICIAL COMMITTEE OF UNSECURED CREDITORS

By: /s/ Mark E. Freedlander
One of Their Attorneys

Mark E. Freedlander Sally E. Edison MCGUIREWOODS LLP 625 Liberty Avenue 23rd Floor Dominion Tower Pittsburgh, PA 15222 Phone: 412.667.6000

Fax: 412.667-6050

Counsel for the Creditors' Committee

**Letters Recommending Acceptance of the Plan** 

[THIS PAGE INTENTIONALLY LEFT BLANK]

# Baker Hostetler

Baker&Hostetler LLP

3200 National City Center 1900 East 9th Street Cleveland, OH 44114-3485

T 216.621.0200 F 216.696.0740 www.bakerlaw.com

June 12, 2007

Re: Confirmation of Joint Plan of Liquidation Filed by Debtors and the Committee

To: Unsecured Creditors of CEP Holdings, LLC:

Counsel for CEP Holdings, LLC and its affiliated debtors and debtors in possession (the "<u>Debtors</u>") writes to encourage you to **VOTE IN FAVOR** of the First Amended Joint Plan of Liquidation (as it may be amended, the "Plan").

A hearing to consider confirmation of the Plan is scheduled for 1:30 p.m., Eastern Time, on July 24, 2007 in the Judge's usual courtroom at United States Bankruptcy Court, 455 U.S. Courthouse, 2 South Main Street, Akron, Ohio 44308. This hearing may be continued from time to time without further notice, other than the announcement of the adjourned date(s).

In order for your vote to be counted, you must complete all required information on the ballot and return the completed ballot to the address indicated on the ballot so that it is received by 5:00 p.m., Eastern Time, on July 13, 2007.

The Plan is proposed jointly by the Debtors and the Committee and is the product of extensive negotiations with numerous parties. The Debtors believe that, under the circumstances, the Plan represents a fair and equitable distribution of the Debtors' assets, as more fully described in the Disclosure Statement.

Requests for additional copies of the Plan and the related Disclosure Statement may be made in writing or orally to BMC Group, Inc., CEP HOLDINGS, LLC, et al., c/o BMC Group, PO Box 903, El Segundo, CA 90245-0903 (telephone number (888) 909-0100). Copies of the Plan and Disclosure Statement also are available for review, without charge, at http://www.bmcgroup.com/cep.

oseph F. Hutchinson, Jr. Thomas M. Wearsch

AKER & HOSTETLER LLP

Eric R. Goodman 3200 National City Center

1900 East 9th Street

Cleveland, Ohio 44114-3485

[THIS PAGE INTENTIONALLY LEFT BLANK]

McGuireWoods LLP
Dominion Tower
625 Liberty Avenue, 23rd Floor
Pittsburgh, PA 15222-3142
Phone: 412.667.6000
Fax: 412.667.6050
www.mcguirewoods.com

Mark E. Freedlander Direct: 412.667.7928



mfreedlander@mcguirewoods.com Direct Fax: 412.667.7967

June 5, 2007

Re:

In re CEP Holdings, LLC, et al (the "Debtors") Case No. 06-51848-MSS (Jointly Administered) First Amended Joint Plan of Liquidation

To Creditors and Parties-in-Interest:

McGuireWoods is counsel to the Official Committee of Unsecured Creditors of CEP Holdings, LLC, Creative Engineered Polymer Products, LLC and Thermoplastics Acquisition LLC (the "Committee"). Enclosed please find the First Amended Joint Plan of Liquidation Under Chapter 11 of the Bankruptcy Code Proposed by the Debtors and the Committee Dated May 25, 2007 (the "First Amended Plan"). Please be advised that the Committee actively participated in the preparation and negotiation of the First Amended Plan and is, in fact, a co-proponent of the First Amended Plan. The Committee submits that the First Amended Plan provides the greatest means of return to the Debtors' creditors. As such, the Committee recommends that creditors vote to accept the confirmation of the First Amended Plan.

Please feel free to call me with any questions or comments that you may have.

Very truly yours,

McGuireWoods LL

Mark E. Freedlander

\4603674.2

[THIS PAGE INTENTIONALLY LEFT BLANK]

First Amended Disclosure Statement to Accompany First Amended Joint Plan of Liquidation Under Chapter 11 of the Bankruptcy Code Proposed by the Debtors and the Official Committee of Unsecured Creditors Dated May 25, 2007 [THIS PAGE INTENTIONALLY LEFT BLANK]

## UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF OHIO EASTERN DIVISION

IN RE:	)	Case No. 06-51848
	)	(Jointly Administered)
CEP HOLDINGS, LLC, et.al., <sup>1</sup>	)	Chapter 11
	)	
Debtors.	)	Honorable Marilyn Shea-Stonum
	)	United States Bankruptcy Judge

FIRST AMENDED DISCLOSURE STATEMENT TO ACCOMPANY
FIRST AMENDED JOINT PLAN OF LIQUIDATION UNDER CHAPTER 11 OF THE
BANKRUPTCY CODE PROPOSED BY THE DEBTORS AND THE OFFICIAL COMMITTEE
OF UNSECURED CREDITORS DATED MAY 25, 2007

### **BAKER & HOSTETLER LLP**

Joseph F. Hutchinson, Jr., Esquire Thomas M. Wearsch, Esquire Eric R. Goodman, Esquire 3200 National City Center 1900 East 9th Street Cleveland, OH 44114

Telephone: (216) 621-0200 Facsimile: (216) 696-0740 Counsel to the Debtors

-and-

MCGUIREWOODS LLP

Mark E. Freedlander, Esquire Sally E. Edison, Esquire William C. Price, Esquire Dominion Tower, 23rd Floor 625 Liberty Avenue Pittsburgh, PA 15222

Telephone: (412) 667-6000 Facsimile: (412) 667-6050

Counsel to the Official Committee of Unsecured

Creditors of the Debtors

The Debtors include: CEP Holdings, LLC, Creative Engineered Polymer Products, LLC and Thermoplastics Acquisition, LLC

## **IMPORTANT NOTICE**

This Disclosure Statement<sup>2</sup> and its related documents are the only documents authorized by the Bankruptcy Court to be used in connection with the solicitation of votes to accept the Plan. No representations have been authorized by the Bankruptcy Court concerning the Debtors, their business operations or the value of their assets, except as explicitly set forth in this Disclosure Statement.

Please refer to the Plan (or, where indicated, certain motions filed with the Bankruptcy Court) for definitions of the capitalized terms used in this Disclosure Statement.

The Plan Proponents reserve the right to file an amended Plan and Disclosure Statement from time to time. The Plan Proponents urge you to read this Disclosure Statement carefully for a discussion of voting instructions, recovery information, classification of claims, the history of the Debtors and the Cases and a summary and analysis of the Plan.

The Plan and this Disclosure Statement have not been required to be prepared in accordance with federal or state securities laws or other applicable non-bankruptcy law. This Disclosure Statement has been approved by the Bankruptcy Court as containing "adequate information"; however, such approval does not constitute endorsement of the Plan or Disclosure Statement by the Bankruptcy Court and none of the United States Securities and Exchange Commission, any state securities commission or similar public, governmental or regulatory authority has approved this Disclosure Statement, or the Plan, or has passed on the accuracy or adequacy of the statements in this Disclosure Statement. Persons trading in or otherwise purchasing, selling or transferring securities, if any, of the Debtors should evaluate the Plan in light of the purposes for which it was prepared.

This Disclosure Statement contains only a summary of the Plan. This Disclosure Statement is not intended to replace a careful and detailed review of the Plan, only to aid and supplement such review. This Disclosure Statement is qualified in its entirety by reference to the Plan, and the exhibits attached thereto and the agreements and documents described therein. If there is a conflict between the Plan and this Disclosure Statement, the provisions of the Plan will govern. In the event of a conflict between this Disclosure Statement and the Agreement, the Agreement shall control. Further, in the event of a conflict between the Plan and the Agreement, the Agreement shall control. You are encouraged to review the full text of the Plan and to read carefully the entire Disclosure Statement, including all exhibits, before deciding how to vote with respect to the Plan.

Except as expressly otherwise indicated, the statements in this Disclosure Statement are made as of May 25, 2007, and the delivery of this Disclosure Statement will not, under any circumstances, imply that the information contained in this Disclosure Statement is correct at any time after May 25, 2007. Any estimates of claims or interests in this Disclosure Statement may vary from the final amounts of claims or interests allowed by the Bankruptcy Court. In addition, the treatment of creditors under the Plan described herein is subject to change as such treatment continues to be negotiated.

<sup>&</sup>lt;sup>2</sup> Unless otherwise defined herein, capitalized terms shall have the meanings ascribed to them in the First Amended Joint Plan of Liquidation under Chapter 11 of the Bankruptcy Code Proposed by the Debtors and the Official Committee of Unsecured Creditors Dated May 25, 2007 (the "Plan").

YOU SHOULD NOT CONSTRUE THIS DISCLOSURE STATEMENT AS PROVIDING ANY LEGAL, BUSINESS, FINANCIAL OR TAX ADVICE. YOU SHOULD, THEREFORE, CONSULT WITH YOUR OWN LEGAL, BUSINESS, FINANCIAL AND TAX ADVISORS AS TO ANY SUCH MATTERS IN CONNECTION WITH THE PLAN, THE SOLICITATION OF VOTES ON THE PLAN AND THE TRANSACTIONS CONTEMPLATED BY THE PLAN.

As to contested matters, adversary proceedings and other actions or threatened actions, this Disclosure Statement is not, and is in no event to be construed as, an admission or stipulation as to any fact or allegation.

## **TABLE OF CONTENTS**

			Page
I.	OVE	RVIEW OF THE DISCLOSURE STATEMENT	1
II.		IMARY AND OVERVIEW OF THE PLAN	
III.		ERAL INFORMATION	
111.	A.	Overview of Chapter 11	
	В.	Description of Debtors	
IV.		PETITION SECURED DEBT	
V.		CHAPTER 11 CASES	
•	Α.	Debtors' Chapter 11 Filings	
	В.	Debtor-in-Possession Financing	
	C.	The Official Committee of Unsecured Creditors	
	D.	Compliance with Bankruptcy Code, Bankruptcy Rules, Local Court Rules and U.S. Trustee Deadlines	i
	E.	Sale of Substantially All of the Debtors' Assets	
	F.	Assumption of Pension Plans	
	G.	Termination of Collective Bargaining Agreements	
	H.	Extension of the Debtors' Exclusivity Periods	19
VI.	PLA	N OF LIQUIDATION	20
	A.	General Description of the Plan	20
	B.	Substantive Consolidation	20
	C.	Treatment of CEP's 99.9% Ownership Interest in CEP Mexico	20
	D.	Unclassified Allowed Claims and Their Treatment	21
	E.	Classification and Treatment of Allowed Claims and Equity Interests	22
	F.	Claims Bar Date	24
	G.	Treatment and Status of Certain Claims	25
	Н.	Confirmation Hearing	26
	I.	Cancellation of Existing Securities and Agreements	27
	J.	Means of Executing the Plan	27
	K.	Treatment of Disputed, Contingent and Unliquidated Claims	36
	L.	The Committee	38

M.

## TABLE OF CONTENTS

(continued)

	O.	Releases	38
	P.	Executory Contracts	39
	Q.	Retention of Jurisdiction of the Bankruptcy Court	
	R.	Conditions to Effectiveness of the Plan	40
	S.	Reserved.	41
	T.	Other Miscellaneous Provisions	41
VII.	PROJI	ECTED DISTRIBUTION TO CREDITORS	42
VIII.	UNITI	ED STATES FEDERAL INCOME TAX CONSIDERATIONS	42
	A.	Introduction	42
	B.	Certain Material United States Federal Income Tax Consequences to the Debtors	43
	C.	Certain Material United States Federal Income Tax Consequences to Holders of Claims	43
IX.	ALTE	RNATIVES TO THE PLAN	45
	A.	Other Plans of Reorganization	45
	B.	Liquidation Under Chapter 7 of the Bankruptcy Code	45
X.	CONF	IRMATION REQUIREMENTS	46
	A.	The Confirmation Hearing	46
	B.	Acceptances Necessary to Confirm the Plan	46
	C.	Best Interests of Creditors	47
	D.	Feasibility	48
	E.	Confirmation of the Plan	48
XI.	CERT	AIN RISK FACTORS TO BE CONSIDERED	48
	A.	Parties-In-Interest May Object to the Plan Proponents' Classification of Claims	48
	B.	The Plan Proponents May Not Be Able to Secure Confirmation of the Plan	48
	C.	The Plan Proponents and/or the Trustee May Object to the Amount or Classification of Your Claim	49
	D.	Insurance Claims	49
	E.	Assumptions Forming the Basis for Calculating Estimated Returns to Creditors Could be Incorrect	49
XII.	WHE	RE YOU CAN OBTAIN MORE INFORMATION	50

Page

## **EXHIBITS**

Exhibit A -- Joint Plan of Liquidation Under Chapter 11 of the Bankruptcy Code Proposed by the Debtors and The Official Committee of Unsecured Creditors Dated May 25, 2007

Exhibit B -- Order of the Bankruptcy Court Approving the Disclosure Statement Dated **June 6, 2007** 

Exhibit C -- Distribution Analysis

## UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF OHIO EASTERN DIVISION

IN RE:	)	Case No. 06-51848
	)	(Jointly Administered)
CEP HOLDINGS, LLC, et.al., 1	)	Chapter 11
	)	
Debtors.	)	Honorable Marilyn Shea-Stonum
	)	United States Bankruptcy Judge

### I. OVERVIEW OF THE DISCLOSURE STATEMENT

### PURPOSE OF DISCLOSURE STATEMENT

CEP Holdings, LLC, Creative Engineered Polymer Products, LLC and Thermoplastics Acquisition, LLC, the above-captioned debtors and debtors-in-possession, together with the Official Committee of Unsecured Creditors prepared this First Amended Disclosure Statement to accompany and in connection with their solicitation of acceptances of the First Amended Joint Plan of Liquidation Under Chapter 11 of the Bankruptcy Code Proposed by the Debtors and The Official Committee of Unsecured Creditors Dated May 25, 2007, filed in the Debtors' reorganization proceedings under the Bankruptcy Code, pending in the Bankruptcy Court. After notice and hearing, and upon order of the Bankruptcy Court dated **June 6, 2007**, the Bankruptcy Court approved this Disclosure Statement as containing information of a kind and in sufficient detail that would enable a hypothetical reasonable investor, typical of holders of claims and interests of the classes being solicited, to make an informed judgment whether to vote to accept or reject the Plan.

A copy of the Plan is attached to this Disclosure Statement and incorporated into this Disclosure Statement by reference as Exhibit A. [A copy of the order of the Bankruptcy Court approving this Disclosure Statement is attached to this Disclosure Statement as Exhibit B.] Unless otherwise specifically noted, all capitalized terms utilized herein shall have the meaning ascribed to such terms as set forth in the Plan.

You should read this Disclosure Statement and the Plan in their entirety before voting on the Plan. No statements or information concerning the Debtors, their subsidiaries or affiliates or any other entity described in this Disclosure Statement or the Plan, particularly, but not limited to, the Debtors' profits, financial condition, assets or liabilities are authorized by the Plan Proponents other than as set forth in this Disclosure Statement or Exhibits hereto.

The financial information set forth in this Disclosure Statement has not been audited by independent certified public accountants, nor has it necessarily been prepared in accordance with generally accepted accounting principles, except as specifically set forth herein. For that reason, and as a result of the complexity of the financial affairs of the Debtors (and their subsidiaries and/or affiliates, to the extent applicable), the Plan Proponents are not able to represent and warrant that the information set

The Debtors include: CEP Holdings, LLC, Creative Engineered Polymer Products, LLC and Thermoplastics Acquisition, LLC

forth in this Disclosure Statement is without any inaccuracy. To the extent possible, however, the information has been prepared from the Debtors' financial books and records, and every reasonable effort has been made to ensure that all information in this Disclosure Statement has been fairly presented.

### PROCEDURAL INFORMATION

### Voting

Each holder of a Claim or Equity Interest of a Class that is "Impaired" under the Plan, but is not deemed to have rejected the Plan, will receive this Disclosure Statement, the Plan, the Voting Procedures Order, notice of the Confirmation Hearing and a ballot for accepting or rejecting the Plan. Any holder of a Claim or Equity Interest whose legal, contractual or equitable rights are altered, modified or changed by the proposed treatment under the Plan, or whose treatment under the Plan is not provided for in section 1124 of the Bankruptcy Code, is considered "Impaired." Each holder of a Claim or Equity Interest of a Class that is deemed to accept or reject the Plan will receive the Voting Procedures Order, notice of the Confirmation Hearing and a notice of non-voting status in the form approved by the Bankruptcy Court, but will not receive a ballot and will not be eligible to vote on the Plan. Holders of Claims or Equity Interests of a Class deemed to accept or reject the Plan will not receive copies of the Plan or the Disclosure Statement, but may obtain copies of these documents by mailing a written request for such materials to the Balloting Agent identified below.

#### Which Classes of Claims are Entitled to Vote on the Plan?

Classes of Claims are entitled to vote on the Plan as follows:

- Claims in Class 4 are Impaired and entitled to vote on the Plan (each a "Voting Class" and together the "Voting Classes").
- Claims in Classes 1, 2 and 3 are unimpaired under the Plan, are deemed to have accepted the Plan and will not be entitled to vote on the Plan.
- Equity Interests in Class 5 are not expected to receive a distribution under the Plan, are deemed to have rejected the Plan and will not be entitled to vote on the Plan.

For a description of the Classes of Claims and Equity Interests and their respective treatment under the Plan, see Section II and Section VI D below.

You may only vote on the Plan with respect to a Claim or Equity Interest if that Claim belongs to a Voting Class under the Plan. All holders of Equity Interests are deemed to have rejected the Plan and will not be entitled to vote. The Bankruptcy Court has fixed June 6, 2007, as the voting record date. To be eligible to vote on the Plan, persons with Claims that belong to the Voting Classes must have held them on the voting record date.

Under the Bankruptcy Code, the Plan will be deemed accepted by an Impaired Class of Claims if BMC Group, Inc. (the "Balloting Agent") receives votes accepting the Plan representing at least:

- two-thirds of the total dollar amount of the allowed Claims in the Class that cast a vote;
   and
- more than one-half of the total number of allowed Claims in the Class that cast a vote.

The Voting Procedures Order will set forth which Claims are "allowed" for purposes of voting and designate the form of ballot to be used by each Voting Class. For more information on voting procedures, please consult the Voting Procedures Order.

All properly completed ballots received by the Balloting Agent on **July 13, 2007 at 5:00 p.m.** (EST) (the "Voting Deadline"), will be counted in determining whether each Impaired Class entitled to vote on the Plan has accepted the Plan. Any ballots received after the Voting Deadline will not be counted. All ballots must contain an original signature to be counted. No ballots received by facsimile will be accepted.

### Voting on the Plan

When does the vote need to be received? The deadline for the receipt by the Balloting Agent of properly completed ballots July 13, 2007 at 5:00 p.m. (EST).

*Which Classes may vote?* Persons may vote to accept or reject the Plan only with respect to Allowed Claims that belong to a Class that is Impaired under the Plan and is not deemed to have rejected the Plan. This is Class 4.

Which members of the Impaired Classes may vote? The voting record date for determining which members of Impaired Classes may vote on the Plan is June 6, 2007. Persons may vote on the Plan only with respect to Claims that were held on the voting record date.

How do I vote on the Plan? For a vote to be counted, the Balloting Agent must receive an original signed copy of the ballot form approved by the Bankruptcy Court. Faxed copies and votes sent on other forms will not be accepted.

Who should I contact if I have questions or need a ballot? You may contact the Balloting Agent at the address or phone number listed below.

This Disclosure Statement and the Plan are the only materials that you should use in determining how to vote on the Plan. The Plan Proponents believe that approval of the Plan provides the greatest return to holders of Claims in the Voting Classes.

### **Voting Recommendations**

The Plan Proponents believe that the Plan presents the best opportunity for holders of Claims to maximize their respective recoveries. The Plan Proponents encourage holders of Impaired Claims to vote to <u>accept</u> the Plan.

The ballots have been specifically designed for the purpose of soliciting votes on the Plan from each Class entitled to vote. For this reason, in voting on the Plan, please use only the ballot sent to you with this Disclosure Statement. If you hold Claims in more than one Class, you must use a separate ballot for voting with respect to each Class of Claims that you hold. If you believe you have received the incorrect form of ballot, you need another ballot or you have any questions concerning the form of ballot, please contact the Balloting Agent.

Please complete and sign your ballot and return it in the enclosed pre-addressed envelope to the Balloting Agent. All correspondence in connection with voting on the Plan should be directed to the Balloting Agent at the following address:

### **Balloting Agent**

### By mail:

CEP Ballot Process Center c/o BMC Group, Inc. Attn: CEP Holdings, LLC Balloting Agent P.O. Box 903 El Segundo, CA 90245-0903

## By overnight delivery:

CEP Ballot Process Center c/o BMC Group, Inc. Attn: CEP Holdings, LLC Balloting Agent 1330 E. Franklin Avenue El Segundo, CA 90245 Phone: 888-909-0100

The Balloting Agent will prepare and file with the Bankruptcy Court a certification of the results of the voting on the Plan on a Class-by-Class basis.

Additional copies of the ballots, this Disclosure Statement and the Plan are available upon request made to the Balloting Agent. Please contact the Balloting Agent with any questions relating to voting on the Plan.

### **Your Vote Is Important**

Your vote on the Plan is important because:

- Under the Bankruptcy Code, a chapter 11 plan can only be confirmed if certain majorities in dollar amount and number of claims (as described above) of each Voting Class under the plan vote to accept the plan, unless the "cram down" provisions of the Bankruptcy Code are used.
- Under the Bankruptcy Code, only the votes of those holders of claims or
  interests who actually submit votes on a plan are counted in determining
  whether the specified majorities of votes in favor of the plan have been
  received.
- If you are eligible to vote with respect to a Claim and do not deliver a properly completed ballot relating to that Claim by the Voting Deadline, you will be deemed to have abstained from voting with respect to that Claim and your eligibility to vote with respect to that Claim will *not* be considered in determining the number and dollar amount of ballots needed to make up the specified majority of that Claim's Class for the purpose of approving the Plan.

All pleadings and other documents referred to in this Disclosure Statement as being on file with the Bankruptcy Court are available for inspection and review during normal business hours at the Office of the Clerk of the United States Bankruptcy Court for the Northern District of Ohio (Eastern Division), 455 United States Courthouse, 2 South Main Street, Akron, Ohio, 44308; telephone 330.252.6100, or on-line at the Bankruptcy Court's website: http://www.ohnb.uscourts.gov.

THIS DISCLOSURE STATEMENT CONTAINS SUMMARIES OF CERTAIN PROVISIONS OF THE PLAN AND OTHER DOCUMENTS RELATING TO THE PLAN. WHILE THE PLAN PROPONENTS SUBMIT THAT THOSE SUMMARIES PROVIDE ADEQUATE INFORMATION WITH RESPECT TO THE DOCUMENTS SUMMARIZED, THESE SUMMARIES ARE QUALIFIED BY THE COMPLETE TEXT OF SUCH DOCUMENTS. IF ANY INCONSISTENCIES EXIST BETWEEN THE TERMS AND PROVISIONS OF THIS DISCLOSURE STATEMENT AND THE TERMS AND PROVISIONS OF THE PLAN OR OTHER DOCUMENTS DESCRIBED HEREIN, THE TERMS AND PROVISIONS OF THE PLAN AND OTHER DOCUMENTS ARE CONTROLLING. EACH HOLDER OF AN IMPAIRED CLAIM SHOULD REVIEW THE ENTIRE PLAN AND ALL RELATED DOCUMENTS AND SEEK THE ADVICE OF ITS OWN COUNSEL AND FINANCIAL CONSULTANT BEFORE VOTING TO ACCEPT OR REJECT THE PLAN. ANY CHANGES TO THESE DOCUMENTS WILL BE DESCRIBED AT THE HEARING ON THE CONFIRMATION OF THE PLAN.

THIS DISCLOSURE STATEMENT MAY NOT BE RELIED UPON BY ANY PERSON OR ENTITY FOR ANY PURPOSE OTHER THAN BY HOLDERS OF IMPAIRED CLAIMS OR INTERESTS ENTITLED TO VOTE ON THE PLAN IN DETERMINING WHETHER TO VOTE TO ACCEPT OR REJECT THE PLAN. NOTHING CONTAINED IN THIS DISCLOSURE STATEMENT WILL CONSTITUTE AN ADMISSION OF ANY FACT OR LIABILITY BY ANY PARTY, OR BE ADMISSIBLE IN ANY PROCEEDING INVOLVING THE DEBTORS OR ANY OTHER PARTY, OR BE DEEMED CONCLUSIVE EVIDENCE OF THE TAX OR OTHER LEGAL EFFECTS OF THE PLAN ON HOLDERS OF CLAIMS OR EQUITY INTERESTS.

EXCEPT TO THE EXTENT OTHERWISE SPECIFICALLY NOTED HEREIN, THE INFORMATION CONTAINED IN THIS DISCLOSURE STATEMENT IS GENERALLY INTENDED TO DESCRIBE FACTS AND CIRCUMSTANCES ONLY AS OF **MAY 25, 2007,** AND NEITHER THE DELIVERY OF THIS DISCLOSURE STATEMENT NOR THE CONFIRMATION OF THE PLAN WILL CREATE ANY IMPLICATION, UNDER ANY CIRCUMSTANCES, THAT THE INFORMATION CONTAINED IN THIS DISCLOSURE STATEMENT IS CORRECT AT ANY TIME AFTER **MAY 25, 2007** OR THAT THE PLAN PROPONENTS WILL BE UNDER ANY OBLIGATION TO UPDATE SUCH INFORMATION IN THE FUTURE.

THE PLAN PROPONENTS BELIEVE THAT ACCEPTANCE OF THE PLAN IS IN THE BEST INTEREST OF EVERY CREDITOR AND RECOMMEND THAT YOU VOTE TO ACCEPT THE PLAN.

### II. SUMMARY AND OVERVIEW OF THE PLAN

The following table briefly summarizes the classifications and treatment of Claims and Equity Interests under the Plan.

Voting	°Z	٥ Z	N <sub>O</sub>	No
<u>Impairment</u>	N/A	N/A	N/A	N/A
Estimated Recovery	100%	100%	100%	100%
Treatment	Except to the extent that a holder of an Allowed Administrative Expense Claim agrees to a different treatment, on the Effective Date, or as soon thereafter as is reasonably practicable, the Disbursing Agent shall pay to each holder of an Allowed Administrative Expense Claim, in full satisfaction of such Claim, an amount in Cash equal to the Allowed amount of such Claim.	All Professionals or other Persons seeking an award by the Bankruptcy Court of compensation for services rendered or reimbursement of expenses incurred through and including the Confirmation Date under sections 327, 328, 330, 331, 503(b)(2), 503(b)(3), 503(b)(4), or 503(b)(5) of the Bankruptcy Code (i) shall file their respective final applications for allowance of compensation for services rendered and reimbursement of expenses incurred by the date that is thirty (30) days after the Confirmation Date, unless otherwise ordered by the Bankruptcy Court, and (ii) shall be paid in full by the Disbursing Agent in such amounts as are allowed by the Bankruptcy Court (A) upon the date upon which the order relating to any such Administrative Expense Claim is entered, or (B) upon such other terms as may be mutually agreed upon between the holder of such an Administrative Expense Claim and the Committee.	Paid on a quarterly basis as required by statute until the Cases are closed.	Except to the extent that the Wachovia Secured Claim has been paid and satisfied prior to the Effective Date in accordance with
Type of Claim or Equity Interest	Administrative Expense Claims (other than those set forth separately below)	Compensation and Reimbursement Claims	U.S. Trustee Fees	Wachovia Secured Claim
Class	N/A	N/A	N/A	N/A

Voting		oN N	oN N
<u>Impairment</u>		N/A	Unimpaired
Estimated Recovery		100%	100%
Treatment	the Final DIP Order, on or as soon as reasonably practicable after the Effective Date, Wachovia shall receive in full satisfaction of such Claim, Cash in an amount equal to one hundred percent (100%) of the unpaid amount of the Allowed Wachovia Secured Claim. The liens securing such Allowed Wachovia Secured Claim shall be deemed released at such time as the Wachovia Secured Claim is paid in full.	Except to the extent that a holder of an Allowed Priority Tax Claim agrees to a different treatment, on the Effective Date, or as soon thereafter as is reasonably practicable, the Disbursing Agent shall pay to each holder of an Allowed Priority Tax Claim, in full satisfaction of such Claim, an amount in Cash equal to the Allowed amount of such Claim. To the extent that the holder of an Allowed Priority Tax Claim is not paid Cash equal to the Allowed amount of such Claim on the Effective Date, such holder shall be entitled to interest at a rate determined under applicable non-bankruptcy law.	To the extent that the Subordinated Secured Participation Claims have not been indefeasibly paid and satisfied prior to the Effective Date in accordance with the Final DIP Order on or as soon as reasonably practicable after the Effective Date, the Participating Customers shall receive in full satisfaction of their respective Subordinated Secured Participation Claims Cash in an amount of the Allowed Subordinated Secured Participation Claims, plus interest accrued thereon from and after the Petition Date, less the amount of Agreed AR due from each respective Participating Customer. The Liens securing such Allowed Subordinated Secured Participation Claims shall be deemed
Type of Claim or Equity Interest		Priority Tax Claims	Subordinated Participating Customers Secured Claims
Class		N/A	Class 1

Voting		o N	o <sub>N</sub>
<u>Impairment</u>		Unimpaired	Unimpaired
Estimated Recovery		100%	100%
Treatment	released at such time as the Subordinated Secured Participation Claims are paid in full.	On or as soon as reasonably practicable after the Effective Date, each holder of an Allowed Other Secured Claim shall receive, at the option of the Liquidating Trustee, and in full satisfaction of such Claim, either (i) Cash in an amount equal to one hundred percent (100%) of the Allowed Other Secured Claim (taking into account sections 506(a)(1) and 506(b) of the Bankruptcy Code); (ii) the proceeds of the sale or disposition of the Collateral securing such Allowed Other Secured Claim to the extent of the value of the holder's secured interest in such Collateral, net of the costs of disposition of such Collateral; (iii) such other distribution as necessary to satisfy the requirements of the Bankruptcy Code, including the surrender of any such collateral; or (iv) such other treatment as the Liquidating Trustee and such Other Secured Claim holder may agree.	Except to the extent that a holder of an Allowed Priority Non-Tax Claim against the Debtors agrees to a different treatment of such Claim, on the Effective Date, or as soon thereafter as is reasonably practicable, each holder of an Allowed Priority Non-Tax Claim shall receive the greater of one hundred present (100%) of the Allowed Amount of such Priority Non-Tax Claims or a Pro Rata Share of all remaining Cash after payment of all Allowed Claims in sections 2.1, 2.2, 2.3, 2.4, 2.5 and 4.2 of the Plan, on the Effective Date or as soon thereafter as is practicable. On the Effective Date, cash in the distributable amount of any Disputed Class 3 claim shall be paid into the Disputed Claim Reserve consistent with Sections 8.3 and 8.5 of the Plan.
Type of Claim or Equity Interest		Other Secured Claims	Priority Non-Tax Claims
Class		Class 2	Class 3

Class	Type of Claim or Equity Interest	<u>Treatment</u>	Estimated Recovery	<u>Impairment</u>	Voting
Class 4	General Unsecured Claims	The holders of Allowed General Unsecured Claims shall receive their Pro Rata Share of all remaining Cash, or instead with respect to Disputed Class 4 Claims, amounts shall be paid into the Disputed Claim Reserve consistent with Section 8.3 and 8.5 of the Plan, after payment in full of all Allowed Claims pursuant to sections 2.1, 2.2, 2.3, 2.4, 2.5, 4.1, 4.2 and 4.3 of the Plan at such times as the Trustee deems practicable, however, in the event that cash available for distribution to Allowed Class 4 Claims exceeds \$1.5 million an initial distribution to Allowed Class 4 Claims be made no later than ninety (90) days from the date on which the Trust Assets available for distribution to Class 4 Claims exceeds \$1.5 million.	30% (est.)	Impaired	Yes
Class 5	Equity Interests	Allowed Class 5 Equity Interests will receive no distribution 0% under the Plan. All Equity Interest issued by the Debtors shall be merged into Holdings (by operation of Section 5.1 of the Plan) on the Effective Date.	%0	Impaired	No

# **Estimated Allowed Claims and Projected Distributions**

Class and Type of Claim	Estimated Allowed Amounts	<b>Projected Distribution</b>
N/A Administrative (other than as specified below)	\$890,000 <sup>4</sup>	100%
N/A Compensation and Reimbursement Claims	\$680,000 <sup>5</sup>	100%
N/A U.S. Trustee Fees	-0- <sup>6</sup>	100%
N/A Wachovia Secured Claim	-0-	100%
N/A Priority Tax		100%
Class 1 Subordinated Participating Customers Secured Claims	\$2,300,000	100%
Class 2 Other Secured Claims	-0-	100%
Class 3 Priority Non-Tax Claims	Unknown <sup>7</sup>	100%
Class 4 General Unsecured Claims	\$30,000,000	30%8
Class 5 Equity Interests	N/A	-0-

<sup>&</sup>lt;sup>4</sup> Estimates based upon internal records and assumptions.

<sup>&</sup>lt;sup>5</sup> Through carveout period of December 31, 2006.

<sup>&</sup>lt;sup>6</sup> Estimated maximum.

<sup>&</sup>lt;sup>7</sup> Certain amounts may be due with respect to the Ohio Bureau of Workers Compensation that may be entitled to priority status, although the Co-Proponents dispute the validity of any such claims.

<sup>&</sup>lt;sup>8</sup> The estimated return to unsecured creditors based upon estimated Cash (without taking into account any potential value of Causes of Action) available for distribution to Allowed Unsecured Claims, estimated Allowed Claims with priority senior to General Unsecured Claims and estimated General Unsecured Claims.

#### III. GENERAL INFORMATION

#### A. Overview of Chapter 11

Chapter 11 is the primary business reorganization chapter of the Bankruptcy Code. Under chapter 11 of the Bankruptcy Code, a debtor is authorized to reorganize its business and affairs for itself, its creditors and its equity interest holders. In addition to permitting the rehabilitation of a debtor, another goal of chapter 11 is to promote equality of treatment for similarly situated creditors and similarly situated equity interest holders with respect to distributions of the value of a debtor's assets.

The commencement of a chapter 11 case creates a bankruptcy estate that is comprised of all of the legal and equitable interests of a debtor as of the Petition Date of the chapter 11 case. The Bankruptcy Code provides that a debtor may continue to operate its business and affairs and remain in possession of its property as a "debtor-in-possession".

The consummation of a plan of reorganization is the fundamental objective of a chapter 11 reorganization case. A plan of reorganization sets forth the means for restructuring a debtor's business and satisfying claims against and interests in a debtor. Confirmation of a plan of reorganization by a bankruptcy court binds the debtor, any issuer of securities under the plan, any person acquiring property under the plan, and any creditor or equity interest holder of a debtor. Pursuant to section 1123(a)(5), a debtor is permitted to distribute its property to those Persons with an interest in such property. Also, pursuant to section 1129(a)(11) of the Bankruptcy Code, a plan may provide for the liquidation of the debtor.

Certain holders of claims against and interests in a debtor are permitted to vote to accept or reject the plan. Prior to soliciting acceptances of the proposed plan, however, section 1125 of the Bankruptcy Code requires a debtor to prepare a disclosure statement containing adequate information of a kind, and in sufficient detail, to enable a hypothetical reasonable investor to make an informed judgment regarding the plan. The Plan Proponents are submitting this Disclosure Statement to holders of claims against and equity interests in the Debtors to satisfy the requirements of section 1125 of the Bankruptcy Code.

#### **B.** Description of Debtors

#### 1. The Debtors

The Debtors were formed as part of two separate purchase transactions on August 16, 2005 and December 20, 2005, respectively. As part of the August 16, 2005 transaction, the Debtors purchased the businesses associated with CEP and CEP Mexico (as defined herein) from Carlisle Engineered Products, Inc. (the "Carlisle Transaction"). As part of the December 20, 2005 transaction, the Debtors purchased the business associated with Thermoplastics (as defined herein) from Parker-Hannifin Corporation (the "Parker-Hannifin Transaction").

Creative Engineered Polymer Products, LLC ("CEP") is a limited liability company formed under the laws of the State of Ohio. CEP is wholly owned by CEP Holdings, LLC ("Holdings"), a privately-held limited liability company formed under the laws of the State of Ohio. Holdings is a holding company whose sole asset is its membership interests in CEP. CEP has three subsidiaries: (i) Composite Parts Mexico S.A. de C.V. ("CEP Mexico"), a Mexican corporation which is 99.9% owned by CEP and .01% owned by non-debtor The Reserve Capital Group, Ltd; (ii) Thermoplastics Acquisition, LLC ("Thermoplastics"), an Ohio limited liability company which is wholly owned by CEP and is a debtor in

the Cases; and (iii) CEP Latin America, LLC ("CEP LA"), a non-debtor Ohio limited liability company which is wholly owned by CEP. CEP LA was never funded and has no operations or debt. The principal place of business of the Debtors is 3560 West Market Street, Suite 340, Akron, Ohio 44333.

The Debtors, collectively with CEP Mexico, operated nine (9) manufacturing plants as of the Petition Date in Ohio, Michigan, Alabama, South Carolina and Mexico. CEP operated six plants in Ohio, Michigan and Alabama (one of which closed prior to the Petition Date). CEP Mexico operated two plants in Mexico. Thermoplastics operated one plant in Ohio and one in South Carolina.

CEP and its subsidiaries were custom molders and extruders of rubber and plastic products, primarily for the original equipment manufacturer ("OEM") automotive market.

Gross sales for the Debtors' businesses was approximately \$174,815,367 for fiscal 2006. As of the Petition Date, the Debtors employed approximately 1,106 hourly and salaried employees of whom approximately 455 were represented by collective bargaining units. In addition to the Debtors' manufacturing facilities in Ohio, Michigan, South Carolina, and Alabama, the Debtors also maintained a technical center in Livonia, Michigan which offered design assistance and program management services for the Debtors' businesses. CEP Mexico produced plastic products at two factories in Mexico.

As of December 31, 2006, the Debtors and CEP Mexico were no longer engaged in any ongoing business operations. As described more fully below, the Debtors utilized the chapter 11 process for the purpose of an orderly wind down and liquidation of their businesses, assets and affairs. The businesses and assets of CEP Mexico were likewise wound down and liquidated in conjunction with the Debtors' Cases.

#### IV. PREPETITION SECURED DEBT

In conjunction with the Carlisle Transaction, CEP Acquisition LLC n/k/a CEP entered into a Loan and Security Agreement, dated as of August 16, 2005, the CEP Loan Agreement, with Wachovia, as both Agent and Lenders thereunder. The CEP Loan Agreement provided two term loans and a revolving credit facility to CEP in the maximum amount of \$45 million (collectively, the "CEP Prepetition Loan"). The CEP Prepetition Loan funded the Carlisle Transaction. The CEP Prepetition Loan was secured by substantially all the assets of CEP, including, without limitation, all accounts, general intangibles, goods, inventory, equipment, real property, accounts receivable, other personal property and proceeds thereof (collectively, the "Prepetition CEP Collateral"). As of the Petition Date, the amount outstanding under the CEP Prepetition Loan was not less than \$21,693,507.60 (not taking into account pre-petition and post-petition interest, fees and expenses to which Agent may be entitled under the CEP Loan Agreement and applicable law).

In conjunction with the Parker-Hannifin Transaction, Thermoplastics entered into a Loan and Security Agreement, dated as of December 21, 2005 (the "Thermoplastics Loan Agreement"), with Wachovia, as both Agent and Lenders. The Thermoplastics Prepetition Loan (as defined herein) substantially funded the Parker-Hannifin Transaction. The Thermoplastics Loan Agreement provided a term loan and a revolving credit facility to Thermoplastics in the maximum amount of \$5 million (the "Thermoplastics Prepetition Loan", and together with the CEP Prepetition Loan, the "Prepetition Loans"). The Thermoplastics Prepetition Loan is secured by substantially all the assets of Thermoplastics, including, without limitation, all accounts, general intangibles, goods, inventory, equipment, accounts receivable, other personal property and proceeds thereof (collectively, the "Prepetition Thermoplastics Collateral", and together with the Prepetition CEP Collateral, the "Prepetition Collateral"). As of the

Petition Date, the amount outstanding under the Thermoplastics Prepetition Loan was not less than \$4,219,688.58 (not taking into account pre-petition and post-petition interest, fees and expenses to which Agent may be entitled under the Thermoplastics Loan Agreement and applicable law). The Prepetition Loans are cross-defaulted and cross-collateralized.

Although intended to be entity specific, in actuality, the CEP Prepetition Loan and the Thermoplastics Prepetition Loan were not, in practice, entity specific, as there were intercompany loans, transactions and other financial accommodations such that the CEP Prepetition Loan, in part funded operations of Thermoplastics and CEP Mexico and the Thermoplastics Prepetition Loan, in part funded operations of CEP and CEP Mexico.

As a result of severe liquidity constraints of the Debtors, prior to the Petition Date, Visteon, GM and Delphi (collectively, the "Subordinated Participating Customers") and Wachovia entered into a Subordinated Participation Agreement dated June 30, 2006 and a First Amendment to Subordination Participation Agreement dated August 18, 2006 pursuant to which the Subordinated Participating Customers purchased subordinated, last out participation interests (the "Participation Interests") in the Prepetition Loans. The Participating Customers purchased, in the aggregate, \$2.9 million of Participation Interests in the Prepetition Loans, the proceeds of which were used by the Debtors to fund their operations, including, inter alia, the building of the Subordinated Participating Customers' parts. In consideration of the purchase of Participation Interests by the Participating Customers, the Debtors were required to execute a Customer Agreement. The Customer Agreement required, among other things, the following: (i) the Participating Customers to (a) make Cash Infusions (as that term is defined in the Final DIP Order) to the Debtors for the funding of, among other things, Restructuring Charges and Wind Down Charges (as those terms are defined in the Final DIP Order); (b) purchase certain of the Debtors' (and CEP Mexico's) inventory; (c) purchase some or all of the Designated Equipment (as that term is defined in the Final DIP Order); and (d) continue accelerated payments terms to the Debtors on terms of "net immediate" (approximately 10 days); and (ii) the Debtors, on behalf of themselves and CEP Mexico, to (x) maintain production and produce parts banks requirements of the Participating Customers; and (y) assume the Access and Security Agreement (as that term is defined in the Final DIP Order) subject to certain modifications in the Customer Agreement, which provided the Participating Customers access to the Debtors' facilities and granted security in certain equipment.

#### V. THE CHAPTER 11 CASES

# A. Debtors' Chapter 11 Filings

The Debtors allege that the Debtors and other automotive suppliers and manufacturers faced a series of unanticipated operational and market challenges that adversely affected their operations and cash flows. These challenges allegedly impaired both the Debtors' suppliers and customers which in turn have severely affected the Debtors' operations and businesses. With respect to suppliers, the Debtors allege that the September 2005 hurricanes in the Gulf Coast region have disproportionately damaged manufacturers who rely on plastic resins. Shortly after the hurricanes, the Debtors began experiencing sharp increases in their principal raw materials (plastic resins) which increases were attributable to interrupted refining capacity. With prices already high due to increased global demand, insecurity and supply constraint issues, the hurricanes magnified the rise in the price of crude oil and natural gas. With respect to the Debtors' customers, the Debtors were unsuccessful in recovering much of these increases in raw material costs from their customers through price increases. The structure of the American automotive industry is such that it is difficult for some manufacturers such as the Debtors to pass rising material costs on to customers.

In addition to increased material costs, the Debtors allege that the general instability of the industry directly harmed the Debtors' liquidity. For example, the Debtors were impaired by the bankruptcy filing of several large OEM's, including Delphi, the Debtors' second largest customer. The bankruptcy filing of Delphi in October 2005 alone resulted in a cash loss to the Debtors of nearly \$1.7 million based on the Debtors' unpaid prepetition claim in that case. In addition to bankruptcy filings in the industry, the general credit downgrade led to delays and increasingly delinquent customer payments for approved tooling programs. These programs are typically managed and paid for by the Debtors for the benefit of a particular customer which subsequently reimbursed the Debtors. The increased delays and failure of customers to pay for these programs decreased the portion of accounts receivable against which Wachovia would lend under the CEP Loan Agreement and Thermoplastics Loan Agreement. This, the Debtors allege, further impaired the Debtors' liquidity.

The Debtors also allege that they experienced excess capacity at their plants due to certain operational decisions by their customers. For example, GM's transfer from the GMT800 platform to the GMT900 platform led to substantial idling of capacity. In late 2005, GM started phasing out the GMT800 platform, a manufacturing platform in which the Debtors were heavily involved. The Debtors allege that they were harmed by this action because (i) the Debtors had significant up front costs invested in the GMT800 platform and (ii) GM did not provide the Debtors with replacement work in the new GMT900 platform. Thus, the Debtors did not recover their costs associated with the GMT800 platform and operated at significantly lower capacity at several manufacturing plants due to a failure to receive work under the GMT900 platform.

Given the Debtors' liquidity issues, the Debtors worked with the Subordinated Participating Customers and Wachovia to allow the Debtors to formulate a restructuring plan that would reorganize the Debtors outside of a chapter 11 proceeding. As part of this plan, in May 2006 the Debtors entered into a series of forbearance, accommodation and access and security agreements with Wachovia and the Subordinated Participating Customers, which agreements provided a 120-day window for the Debtors to effectuate an out-of-court restructuring plan. During the out-of-court restructuring efforts by the Debtors, the Debtors caused an unofficial committee of trade creditors (the "Unofficial Trade Committee") to be organized, of which five (5) of the seven (7) members of the Committee were participants. The Unofficial Trade Committee as well as the Participating Customers rejected the Debtors' out-of-court restructuring proposals, and accordingly, the Debtors filed for bankruptcy protection on September 20, 2006. The Participating Customers determined to resource the work then being performed by the Debtors and CEP Mexico rather than work with the Debtors and CEP Mexico through a going concern restructuring. Without full support from the Participating Customers, the Committee determined that no feasible going concern restructuring plan could be proposed, and to this end; the Committee determined that return to unsecured creditors would be maximized through liquidation of the Debtors and CEP Mexico.

# **B.** Debtor-in-Possession Financing <sup>9</sup>

On the Petition Date, the Debtors filed the Motion of the Debtors and Debtors in Possession for Emergency Order Authorizing Debtors to: (A) Use Cash Collateral on an Emergency Basis; (B) Incur Postpetition Debt on an Emergency Basis; (C) Grant Adequate Protection and Provide Security and Other Relief to Wachovia Capital Finance Corporation (Central); and (D) Grant Certain Related Relief (the

-14-

<sup>&</sup>lt;sup>9</sup> Capitalized terms related to the Debtor-in-Possession financing that are not defined herein, shall have the meanings ascribed to them in the Plan or the Final DIP Order, in that order.

"DIP Motion"). The Unofficial Trade Committee filed an Objection to the DIP Motion. By Order dated September 26, 2006, the Bankruptcy Court granted the DIP Motion on an interim basis and authorized the Debtors to obtain interim postpetition financing on a secured, superpriority basis. Wachovia, the Subordinated Participating Customers and the Unofficial Trade Committee reach a resolution with respect to the Unofficial Trade Committee's objections to the DIP Motion; however, the Bankruptcy Court would not enter the order submitted by the Debtors in connection with the DIP Motion. Further negotiations by and among the Debtors, Wachovia, the Subordinated Participating Customers and the Committee led to a consensual resolution of the issues presented by the Unofficial Trade Committee and the Bankruptcy Court. Thereafter, on October 27, 2006, by Order of the Bankruptcy Court granted the DIP Motion on a final basis (the "Final DIP Order"), which authorized the Debtors to enter into the First Amendment to Postpetition Agreement, Customer Agreement and Postpetition Documents (as those terms are defined in the Final DIP Order) for an aggregate maximum principal amount of \$30,880,000.00. In accordance with the Final DIP Order, Wachovia was granted a security interest in substantially all of the Debtors assets on a superpriority basis for funding related to the Final DIP Order. Further, the Final DIP Order required the Subordinated Participating Customers to provide Cash Infusions (as that term is defined in the Final DIP Order) to the Debtors for the administration of the Estates.

Subject to the terms and conditions of the Plan, all amounts due and owing Wachovia and the Subordinated Participating Customers under the Final DIP Order will be paid in full or the Disputed Amount funded into the Disputed Claim Reserve on the Effective Date, if not sooner paid.

#### C. The Official Committee of Unsecured Creditors

On September 28, 2006, the Office of the United States Trustee filed its Notice of Appointment of Creditors' Committee and appointed five (5) members to serve on a single committee and to represent the interests of all general unsecured creditors of the Debtors. The five (5) members were (i) E.I. du Pont de Nemours and Company ("DuPont"); (ii) Rhodia, Inc.; (iii) The Brown Corporation of Greenville; (iv) Lanxess Corp.; and (v) Excel Polymers LLC. Thereafter, on October 12, 2006, the Office of the United States Trustee filed an Amended Appointment of Committee of Unsecured Creditors and appointed two (2) additional members to serve on a single committee and to represent the interests of all general unsecured creditors of the Debtors. The two (2) additional members were (i) Gold Key Processing Ltd.; and (ii) Washington Penn Plastic Co. Inc.

DuPont and Rhodia, Inc. serve as co-chairpersons of the Committee. The Committee's counsel is the law firm of McGuireWoods LLP. The financial advisor to the Committee is Grant Thornton LLP, as successor to Stout Risius Ross.

# D. Compliance with Bankruptcy Code, Bankruptcy Rules, Local Court Rules and U.S. Trustee Deadlines

On October 30, 2006, the Debtors filed their Statement of Financial Affairs, Schedules of Assets and Liabilities, and lists of Equity Security Holders, each as amended from time to time. On October 25, 2006, the Office of the United States Trustee conducted a meeting of creditors pursuant to section 341 of the Bankruptcy Code. Additionally, the Debtors have filed all monthly operating reports and timely paid all statutory quarterly fees as required by the Office of the U.S. Trustee. To the best of the Debtors' knowledge, information and belief, the Debtors have complied with all other applicable requirements of the Bankruptcy Code and Bankruptcy Rules, as well as local Bankruptcy Court rules and deadlines of the Office of the U.S. Trustee.

# E. Sale of Substantially All of the Debtors' Assets

#### 1. <u>Liquidating Facilities Sale.</u>

By Order dated November 21, 2006, the Bankruptcy Court approved the sale of substantially all of the Debtors' machinery and equipment located at Vandalia, Ohio; Bishopville, South Carolina; Crestline, Ohio; Canton, Ohio; Lapeer, Michigan; Belleville, Michigan; and Middlefield, Ohio (the "Liquidating Facilities Sale"). In furtherance of the Liquidating Facilities Sale, the Bankruptcy Court approved the Debtors retention of Biditup Auctions Worldwide, Inc. (the "Auctioneer") to facilitate the Liquidating Facilities Sale under an Asset Marketing Agreement by and between the Debtors and the Auctioneer, as amended (the "Asset Marketing Agreement"). Under the Asset Marketing Agreement, the Auctioneer guaranteed the net sum of \$5,015,000 (the "Guaranteed Minimum") for the auctions for the Debtors' machinery and equipment located at the following facilities: (i) Vandalia, Ohio; (ii) Belleville, Michigan; (iii) Canton, Ohio; (iv) Crestline, Ohio; and (vi) Lapeer, Michigan (the "First Auction Sites"). In connection with the sales of the First Auction Sites and the Auctioneer's Guaranteed Minimum, the Auctioneer caused to be issued for the benefit of the Estates and their secured lender, Wachovia, a letter of credit (the "Auctioneer Line of Credit") in the amount of \$5,015,000. The Auctioneer has successfully completed the sale of the First Auction Sites for the net sale amount of approximately \$5,602,000. Wachovia drew against the Auctioneer Line of Credit for satisfaction of the Guaranteed Minimum and applied all such amounts to the claim of Wachovia. Under the Asset Marketing Agreement, the Auctioneer was entitled to retain an amount equal to the Auctioneer Line of Credit (\$5,015,000), plus \$480,000, plus one-half of the Auctioneer's actual operating expenses for the month January 2007 related to the First Auction Sites. After application of the amounts related to the immediately proceeding sentence, the Auctioneer and the Debtors shared the remaining proceeds fifteen percent (15%) for the Auctioneer and eighty-five percent (85%) for the Debtors.

Under the Asset Marketing Agreement, the Auctioneer was also responsible for the sale of the machinery and equipment located at the following facilities: (i) Bishopville, South Carolina; and (ii) Middlefield, Ohio (the "Second Auction Sites"). With respect to the sales related to the Second Auction Sites, the Auctioneer was solely entitled to retain the following: (i) ten percent (10%) for bids made in person; (ii) fourteen percent (14%) for bids made via the internet; and (iii) three percent (3%) for all payments made in a method other than cash, cashier's check or money order (the "Buyer's Premium"). The sales at the Second Auction Sites occurred on January 30 and 31, 2007 and yielded approximately \$1.12 million.

## 2. <u>Tuscaloosa, Alabama Sale.</u>

The Estates sold substantially all of their assets (including machinery, equipment and inventory, but excluding accounts receivable) located at their Tuscaloosa, Alabama facility to Visteon for \$1,750,000.

## 3. CEP Mexico Sales.

The Debtors sold the assets located in the facilities (including machinery, equipment and inventory) of CEP Mexico pursuant to Orders of the Bankruptcy Court dated November 28, 2006 and November 29, 2006, respectively. The sale of assets located in the facilities of CEP Mexico were sold as follows: (i) Hermosillo, Mexico assets to Delphi Automotive Systems, LLC, or its Mexican maquiladora subsidiary, Alambrados Y Circuitos Electricos S.A. de C.V. (the "Hermosillo Sale"); and (ii) Chihuahua,

Mexico assets to Century Mold Mexico, LLC (the "Chihuahua Sale" together with the Hermosillo Sale, collectively the "CEP Mexico Sales").

The Estates hold in escrow \$5,408,597 in principal in an interest bearing account, which represents the net proceeds of the CEP Mexico Sales. See section VI (C) herein for a more detailed description of the dissolution of CEP Mexico and the disposition of the \$5,408,597 principal proceeds of the CEP Mexico Sales.

## 4. Real Estate Sales.

Real estate located at the Debtors' Lapeer, Michigan facility was sold with Bankruptcy Court approval to Marvin Engineering Co. and Daniel Schreiber in the amount of approximately \$2,042,000. Real estate located at the Debtors' Crestline, Ohio facility was also sold with Bankruptcy Court approval in the amount of approximately \$650,000.

# 5. Parker-Hannifin Real Estate Sales.

Under the terms of the Parker-Hannifin Transaction, the Debtors submit that they possess an equitable interest in the real estate located at the Debtors' Vandalia, Ohio and Bishopville, South Carolina facilities. Parker-Hannifin retained a purchase money mortgage in the Vandalia and Bishopville facilities as part of the Parker-Hannifin Transaction. Parker-Hannifin filed a motion for relief from the automatic stay with respect to the Vandalia and Bishopville properties. Each of Parker-Hannifin and the Plan Proponents obtained appraisals with respect to these properties. Parker-Hannifin thought that it had a buyer for the Vandalia property at a price acceptable to the Plan Proponents and as such a consensual order for relief was negotiated with respect to Vandalia, however, the proposed buyer for Vandalia withdrew its offer. Plan Proponents and Parker-Hannifin anticipate filing a joint motion to consensually resolve the Allowed Secured Claim of Parker-Hannifin, subject, however, to avoidance rights under section 502(d) of the Bankruptcy Code. The Committee is also investigating potential Causes of Action against Parker-Hannifin relating to the Parker-Hannifin Transaction.

## F. Assumption of Pension Plans

- (i) CEP by virtue of UNITE representation at its Middlefield, OH facility, is a participant in the Unite Textile Workers Pension Plan ("UNITE Plan"). CEP understands that the UNITE Plan is fully funded. CEP will terminate its participation in the UNITE Plan, and there is no expectation of withdrawal liability arising from CEP's termination of participation in the UNITE Plan.
- (ii) CEP assumed the Pension Plan for Bargaining Unit Employees of Carlisle Engineered Products-Crestline as part of the Carlisle Transaction. The Crestline Pension Plan is a defined benefit plan for members of the United Steelworkers of America Local 563-L. The Crestline Pension Plan is underfunded and as a result of the underfunding, it is believed that certain non-Debtor entities owned or controlled by the ultimate equity owners of the Debtors are part of the ERISA controlled group and accordingly have joint and several liability for underfunding of the Crestline Pension Plan in the event of termination of the Crestline Pension Plan. As discussed in more detail below, the Plan Proponents and The Reserve Group Management Company have negotiated accommodations such that the Crestline Pension Plan and the Canton Pension Plan will be assumed by a non-Debtor entity within the ERISA controlled group, Superior Fabrication Company, LLC.

- (iii) CEP assumed the pension plan for Bargaining Unit Employees of the Canton, Ohio Plan of Geauga Company Division of Carlisle Corporation as part of the Carlisle Transaction. The Canton Pension Plan is a defined benefit plan for members of the United Steelworkers of America Local 3610-04. The Canton Pension Plan is underfunded and as a result of the underfunding, it is believed that certain non-Debtor entities owned or controlled by the ultimate equity owner(s) of the Debtors are part of the ERISA controlled group and accordingly have joint and several liability for underfunding of the Canton Pension Plan in the event of termination of the Canton Pension Plan. The Plan Proponents and The Reserve Group Management Company have negotiated accommodations such that the Crestline Pension Plan and Canton Pension Plan will be assumed by non-Debtor entity within the ERISA controlled group, Superior Fabrication Company, LLC.
- (iv) The Plan Proponents and The Reserve Group Management Company contemplate filing a Bankruptcy Rule 9019 motion that seeks Bankruptcy Court approval of a settlement among the Debtors, Committee, The Reserve Group Management Company and affiliates that provides, inter alia, (a) Superior Fabrication Company, LLC, an affiliate of The Reserve Group Management Company, will assume the Pension Plans; (b) The Reserve Group Management Company will release all claims against the Debtors and their Estates, other than those to which The Reserve Group Management Company may, through subrogation to Washington Penn Plastics Company, become a party; (c) The Debtors, the Estates and the Committee will release The Reserve Group Management Company and its affiliates of claims and Causes of Action connected with their dealings and involvement with/on behalf of the Debtors; and (d) the Estates will pay (i) to the PBGC an amount not to exceed \$42,482 with respect to post-Petition Date premiums due under the Pension Plans; (ii) as a contribution to the Pension Plans an amount not to exceed \$29,776 with respect to normal costs charged to funding standard accounts under the Pension Plans post-Petition Date; and (iii) as reimbursement to the Pension Plans an amount not to exceed \$20,000 with respect to post-Petition Date in respect of actuarial services provided by Watson Wyatt to the Pension Plans.
- (v) The settlement referenced herein contains several conditions precedent including Bankruptcy Court approval; agreement by the USW to freeze accrual of benefits under the Pension Plans as of a date certain (contemplated to be June 30, 2007); and agreement by the PBGC to withdraw its proofs of claim for unfunded liabilities relating to the Pension Plans.
- (vi) The PBGC filed a total of eighteen (18) proofs of claim against the Debtors. In actuality, six identical claims are asserted by the PBGC against each of the three Debtors. Three claims are asserted with respect to the Canton Pension Plan as follows:
  - estimated contribution claim totaling \$20,060 of which \$4,414 is estimated to have administrative priority;
  - estimated general unsecured underfunding claim of \$1,325,313; and
  - estimated premiums claim totaling \$843,610 of which approximately \$18,172 represents an administrative priority claim for variable rate premiums; \$802,500 represents an administrative claim for termination premiums; and \$22,938 represents an unsecured claim for variable rate premiums.

Three claims are also asserted by the PBGC with respect to the Crestline Pension Plan as follows:

- estimated contribution claim totaling \$90,958 of which \$25,362 is estimated to have administrative status;
- estimated general unsecured underfunding claim of \$1,996,582; and
- estimated premiums claim totaling \$1,127,496 of which approximately \$24,310 represents an administrative priority claim for variable rate premiums; \$1,072,500 represents an administrative claim for termination premiums; and \$30,686 represents an unsecured claim for variable rate premiums.

With respect to unfunded benefit claims, the PBGC also asserts a lien for up to 30% of the collective net worth of all liable parties and administrative or alternatively unsecured priority status with respect to all such Claims. Assuming termination of the Pension Plans rather than assumption of the Pension Plans by Superior Fabrication Company, LLC as proposed in the settlement outlined above, and further assuming that the Claims asserted by the PBGC would be Allowed Claims (although not in triplicate and although the Claims of the PBGC may be subject to defenses including but not limited to the discount rate utilized by the PBGC in calculating its Claims), the Plan Proponents estimate that termination of the Pension Plans would cause a dilution in potential return to unsecured creditors other than the PBGC of approximately eleven cents (\$.11) or approximately thirty percent (30%). The Plan Proponents submit that litigating the Claims of the PBGC would be costly and time consuming such that litigating these claims may have a significant negative impact on returns to general unsecured creditors.

#### **G.** Termination of Collective Bargaining Agreements

The Debtors continue to engage in good faith effects bargaining with the unions representing former represented employees of the Debtors, in accordance with applicable non-bankruptcy law. Among the requests by the Debtors of the unions in such bargaining is the agreed termination of collective bargaining agreements. In the event, however, that agreed upon consensual termination of the collective bargaining agreements does not promptly occur, the Debtors will file one or more motions with the Bankruptcy Court pursuant to section 1113 of the Bankruptcy Code seeking authorization to terminate collective bargaining agreements effective as of the Confirmation Date. It is expected that termination – either voluntarily or by order of the Bankruptcy Court pursuant to section 1113 of the Bankruptcy Code – may give rise to termination claims of the respective unions the amounts that have not been fully calculated as of the date this Disclosure Statement. The United Steelworkers have filed two unliquidated proofs of claim—one specifically relating to claims for the Debtors' alleged violation of the WARN Act, and the other a catch-all for alleged unpaid wages, benefits and other considerations.

#### H. Extension of the Debtors' Exclusivity Periods

Section 1121(b) of the Bankruptcy Code provides for an initial period of 120 days after the filing of a voluntary petition for relief under chapter 11 during which a debtor has the exclusive right to file a plan of reorganization in the reorganization proceeding. Section 1121(c)(3) of the Bankruptcy Code provides that if a debtor files a plan of reorganization within such 120 day period, the debtor has a period of 180 days from the filing of its voluntary petition for relief to obtain acceptances of such plan, during which time competing plans may not be filed. The Debtors' initial exclusive periods were set to expire on January 18, 2007, and March 19, 2007, respectively. On December 22, 2006, the Debtors filed a motion

seeking to extend the exclusivity deadlines set forth in section 1121(b) of the Bankruptcy Code. The Committee has consented to an extension of the Debtors' exclusive period to file a plan until January 31, 2007. At this time, exclusivity has expired.

#### VI. PLAN OF LIQUIDATION

THE FOLLOWING DISCUSSION OF THE PLAN CONSTITUTES A SUMMARY ONLY AND IS QUALIFIED IN ITS ENTIRETY BY THE TERMS OF THE PLAN ITSELF. YOU SHOULD READ THE PLAN BEFORE DECIDING WHETHER TO VOTE TO ACCEPT OR REJECT THE PLAN. CHANGES MAY BE MADE TO THE PLAN. ANY SUCH CHANGES MADE TO THE PLAN WILL BE DESCRIBED AT THE CONFIRMATION HEARING. A copy of the Plan is attached as Exhibit A to this Disclosure Statement.

## A. General Description of the Plan

This Disclosure Statement has been submitted by the Plan Proponents in order to afford the Debtors the opportunity to confirm a Plan on or before July 31, 2007, that will be consummated on or before August 31, 2007.

#### **B.** Substantive Consolidation

The Plan contemplates "substantive consolidation" of the assets and liabilities of the Debtors. Substantive consolidation means that on the Effective Date (i) all assets (and all proceeds thereof) and liabilities of the Estate of Holdings, and, likewise, all assets (and proceeds thereof) of the Estates of CEP and Thermoplastics and all liabilities of CEP and Thermoplastics shall be deemed merged or treated as though they were merged into and with the assets and liabilities of the Estate of Holdings; (ii) no distributions shall be made under the Plan on account of inter-company claims among the Debtors and all such Claims be eliminated; (iii) all guarantees of the Debtors of the obligations of any other Debtors shall be deemed eliminated and extinguished such that any Claim against any Debtor and any guaranty thereof executed by any other Debtor and any joint or several liability of any of the Debtors shall be deemed to be one obligation of Consolidated Holdings; and (iv) each and every claim filed or to be filed in any of the Cases shall be deemed to be filed against Consolidated Holdings; provided, however, that (a) for the purposes of determining the availability of the right of set-off under section 553 of the Bankruptcy Code, the Debtors shall be treated as separate entities so that, subject to the other provisions of section 553 of the Bankruptcy Code debts due to any of the Debtors may not be set-off against the debts of any of the other Debtors, and (b) for purposes of determining the Plaintiff of, and availability of defenses to, Avoidance Actions, the Debtors may, at the option of the Trustee, be treated as separate entities. To the extent that a creditor is holding an Allowed Claim against more than one of the Debtors' Estates, such creditors shall be treated as a holder of an Allowed Claim as against Consolidated Holdings.

# C. Treatment of CEP's 99.9% Ownership Interest in CEP Mexico

Notwithstanding any provision of the Plan or Disclosure Statement to the contrary, CEP, through the Effective Date and, to the extent applicable, Consolidated Holdings thereafter shall continue to own a 100% interest in CEP Mexico. As 99.9% owner of the stock of CEP Mexico, CEP is in the process of causing the dissolution and liquidation of CEP Mexico to occur in accordance with applicable Mexican law. CEP understands that at this time, CEP Mexico has no creditors other than the contingent and unliquidated ERISA-controlled group liability claims of the PBGC and inter-company claims of the

Debtors. As part of the Mexican liquidation and dissolution process, a liquidator will be appointed on behalf of CEP Mexico, and this liquidator will replace the board of directors of CEP Mexico.

Given the Plan Proponents' understanding of claims (or actually, the lack thereof) against CEP Mexico, and CEP's understanding of the liquidation process for CEP Mexico under Mexican law, CEP has or will cause CEP Mexico to pay over all funds of CEP Mexico other than \$50,000, which is the maximum estimated cost of the dissolution process for CEP Mexico. Currently, a principal amount of \$5,408,597.00 is held in an interest bearing escrow by CEP pursuant to the applicable provisions of the Final DIP Order, which amount represents the proceeds of Bankruptcy Court authorized Mexican Sale Transactions. It is the position of the Plan Proponents that such proceeds represent the value of CEP's 99.9% ownership interest in CEP Mexico, and/or (b) such proceeds are proceeds of assets actually owned by CEP pursuant to a bailment agreement between CEP and CEP Mexico. Under the Plan, the proceeds referenced in this paragraph will be available for distribution to creditors of the Estates.

#### D. Unclassified Allowed Claims and Their Treatment

#### 1. Administrative Expense Claims.

Except to the extent that a holder of an Allowed Administrative Expense Claim agrees to a different treatment, on the Effective Date, or as soon thereafter as is reasonably practicable, the Liquidating Trustee shall pay to each holder of an Allowed Administrative Expense Claim, in full satisfaction of such Claim, an amount in Cash equal to the Allowed amount of such Claim.

# 2. <u>Bar Date For Administrative Expense Claims.</u>

THE DEADLINE TO FILE PROOFS OF ADMINISTRATIVE EXPENSE CLAIMS AND REQUESTS FOR PAYMENT OF ADMINISTRATIVE EXPENSE CLAIMS THAT AROSE PRIOR TO THE CONFIRMATION DATE IS NO LATER THAN THIRTY (30) DAYS FOLLOWING THE CONFIRMATION DATE. THIS DEADLINE, HOWEVER, DOES NOT INCLUDE CLAIMS FOR COMPENSATION AND REIMBURSEMENT WHICH SHALL BE TREATED PURSUANT TO SECTION VI (C)(3) HEREUNDER AND BANKRUPTCY CODE SECTION 503(b)(9) CLAIMS, WHICH WERE TO BE FILED ON OR BEFORE MARCH 1, 2007 PURSUANT TO THE BAR DATE ORDER.

Notwithstanding anything to the contrary herein, no proof of Administrative Expense Claim or application for payment of an Administrative Expense Claim that arose or arises after the Petition Date, need be filed for the allowance of any: (i) claims for professional fees; or (ii) fees of the U.S. Trustee arising under 28 U.S.C. § 1930. All Claims described in clause (i) and (ii) of the immediately preceding sentence shall be paid by the Liquidating Trustee in accordance with the terms of the Plan.

ANY PERSONS THAT FAILED TO FILE A PROOF OF ADMINISTRATIVE EXPENSE CLAIM OR REQUEST FOR PAYMENT THEREOF ON OR BEFORE THE APPLICABLE ADMINISTRATIVE BAR DATE AS REQUIRED HEREIN OR PURSUANT TO OTHER APPLICABLE ORDER OF THE BANKRUPTCY COURT SHALL BE FOREVER BARRED FROM ASSERTING SUCH CLAIM AGAINST ANY OF THE DEBTOR, THE ESTATE, OR ITS PROPERTY AND THE HOLDER THEREOF SHALL BE ENJOINED FROM COMMENCING OR CONTINUING ANY ACTION, EMPLOYMENT OF PROCESS OR ACT TO COLLECT, OFFSET OR RECOVER SUCH ADMINISTRATIVE EXPENSE CLAIM.

# 3. <u>Compensation and Reimbursement Claims.</u>

All Professionals or other Persons seeking an award by the Bankruptcy Court of compensation for services rendered or reimbursement of expenses incurred through and including the Confirmation Date under sections 327, 328, 330, 331, 503(b)(2), 503(b)(3), 503(b)(4), or 503(b)(5) of the Bankruptcy Code (i) shall file their respective final applications for allowance of compensation for services rendered and reimbursement of expenses incurred by the date that is thirty (30) days after the Confirmation Date, unless otherwise ordered by the Bankruptcy Court, (ii) shall be paid in full by the Liquidating Trustee in such amounts as are allowed by the Bankruptcy Court (A) upon the date upon which the order relating to any such Administrative Expense Claim is entered, or (B) upon such other terms as may be mutually agreed upon between the holder of such an Administrative Expense Claim and the Committee, and (iii) to the extent applicable, shall be subject to terms and provisions of the Final DIP Order relating to payment of Professionals.

#### 4. U.S. Trustee.

All fees payable in the Cases under 28 U.S.C. §1930, as agreed by the Debtors and Committee or as determined by the Bankruptcy Court, will, if not previously paid in full, be paid in Cash on the Effective Date and will continue to be paid by the Liquidating Trustee as required under 28 U.S.C. §1930 until such time as an order is entered by the Bankruptcy Court closing the Cases.

# 5. Wachovia Secured Claims.

Except to the extent that the Wachovia Secured Claims has been paid and satisfied prior to the Effective Date in accordance with the Final DIP Order, on or as soon as reasonably practicable after the Effective Date, Wachovia shall receive in full satisfaction of such Claim, Cash in an amount equal to one hundred percent (100%) of the unpaid amount of the Allowed Wachovia Secured Claim. The liens securing such Allowed Wachovia Secured Claim shall be deemed released at such time as the Wachovia Secured Claim is paid in full.

#### 6. Priority Tax Claims.

Except to the extent that a holder of an Allowed Priority Tax Claim agrees to a different treatment, on the Effective Date, or as soon thereafter as is reasonably practicable, the Liquidating Trustee shall pay to each holder of an Allowed Priority Tax Claim, in full satisfaction of such Claim, an amount in Cash equal to the Allowed amount of such Claim. To the extent that the holder of an Allowed Priority Tax Claim is not paid Cash equal to the Allowed amount of such Claim on the Effective Date, such holder shall be entitled to interest at a rate determined by applicable non-bankruptcy law.

# E. Classification and Treatment of Allowed Claims and Equity Interests

The Plan divides Allowed Claims against and Equity Interests in the Debtors into various classes that the Plan Proponents believe are in accordance with the classification requirements established by the Bankruptcy Code. Distribution of Cash to the holders of Allowed Claims under the Plan are in full satisfaction of such Allowed Claims (including any and all interest or costs accrued and allowable thereon). All Claims against and Equity Interests in the Debtors arising prior to the Petition Date will be discharged under the Plan on the Effective Date of the Plan to the extent provided in the Plan or the Confirmation Order.

A summary of the classification and treatment of Allowed Claims and Allowed Interests under the Plan is set forth below.

# 1. <u>Class 1: Subordinated Secured Participation Claims</u>

<u>Distribution</u>: To the extent that the Subordinated Secured Participation Claims have not been indefeasibly paid and satisfied prior to the Effective Date in accordance with the Final DIP Order on or as soon as reasonably practicable after the Effective Date, the Participating Customers shall receive in full satisfaction of their respective Subordinated Secured Participation Claims Cash in an amount equal to one hundred percent (100%) of the unpaid amount of the Allowed Subordinated Secured Participation Claims, plus interest accrued thereon from and after the Petition Date, less the amount of Agreed AR due from each respective Participating Customer. The Liens securing such Allowed Subordinated Secured Participation Claims shall be deemed released at such time as the Subordinated Secured Participation Claims are paid in full.

Class 1 is unimpaired under the Plan. Each holder of an Allowed Class 1 Claim is conclusively presumed to have accepted the Plan and, consequently, is not entitled to vote to accept or reject the Plan.

#### 2. Class 2: Other Secured Claims

<u>Distribution</u>: On or as soon as reasonably practicable after the Effective Date, each holder of an Allowed Other Secured Claim shall receive, at the option of the Trustee, and in full satisfaction of such Claim, either (i) Cash in an amount equal to one hundred percent (100%) of the Allowed Other Secured Claim (taking into account sections 506(a)(1) and 506(b) of the Bankruptcy Code), (ii) the proceeds of the sale or disposition of the Collateral securing such Allowed Other Secured Claim to the extent of the value of the holder's secured interest in such Collateral, net of the costs of disposition of such Collateral, (iii) such other distribution as necessary to satisfy the requirements of the Bankruptcy Code, including the surrender of any such collateral; or (iv) such other treatment as the Trustee and such holder other Secured Claim may agree. In the event the Debtor treats a Claim under clause (i) or (ii) of this section 3.2, the Liens securing such other Secured Claim shall be deemed released.

Class 2 is unimpaired under the Plan. Each holder of an Allowed Class 2 Claim is conclusively presumed to have accepted the Plan, and, consequently, is not entitled to vote to accept or reject the Plan.

#### 3. Class 3: Priority Non-Tax Claims

<u>Distribution</u>: Except to the extent that a holder of an Allowed Priority Non-Tax Claim against the Debtors agrees to a different treatment of such Claim, on the Effective Date, or as soon thereafter as is reasonably practicable, each holder of an Allowed Priority Non-Tax Claim shall receive the greater of 100% of the Allowed Amount of such Priority Non-Tax Claims or a Pro Rata Share of all remaining Cash after payment or treatment consistent with Section 8.3 and 8.5 of the Plan of all Allowed Claims in sections 2.1, 2.2, 2.3, 2.4, 2.5, 4.1 and 4.2 of the Plan, on the Effective Date or as soon thereafter as is practicable.

Class 3 is unimpaired under the Plan. Each holder of an Allowed Class 3 Claim is conclusively presumed to have accepted the Plan, and, consequently, is not entitled to vote to accept or reject the Plan.

# 4. <u>Class 4: General Unsecured Claims</u>

<u>Distribution</u>: The holders of Allowed General Unsecured Claims shall receive their Pro Rata Share of all remaining Cash after payment in full of all Allowed Claims, or treatment consistent with Sections 8.3 and 8.5 of the Plan, pursuant to sections 2.1, 2.2, 2.3, 2.4, 2.5, 4.1, 4.2 and 4.3 of the Plan, at such times as the Trustee deems practicable, however, in the event that cash available for distribution to Allowed Class 4 Claims exceeds \$1.5 million an initial distribution to Allowed Class 4 Claims be made no later than ninety (90) days from the date on which the Trust Assets available for distribution to Class 4 Claims exceeds \$1.5 million.

# Class 4 is impaired under the Plan. Holders of Allowed Class 4 Claims are entitled to vote to accept or reject the Plan

## 5. <u>Class 5: Equity Interests</u>

<u>Treatment</u>: All Equity Interests issued by the Debtors, other than Equity Interests in Holdings, shall be merged into Holdings on the Effective Date. Each holder of an Equity Interest shall neither receive nor retain any property or interest in property on account of such Equity Interest unless and until all Allowed Claims are paid in full with interest. It is not anticipated that any distributions will be made to Class 5 Equity Interest Holders.

Class 5 is impaired under the Plan, and the holders of Allowed Class 5 Equity Interests are deemed to reject the Plan. Consequently, holders of Allowed Class 5 Equity Interests are not entitled to vote to accept or reject the Plan.

#### F. Claims Bar Date

On December 15, 2006, the Bankruptcy Court entered the Order, Pursuant to Sections 501 and 502 of the Bankruptcy Code and Bankruptcy Rules 2002 and 3003(c)(3), Establishing Bar Dates and Filing Proofs of Claim and Approving Form and Manner of Notice Thereof (the "Bar Date Order"), which established a Bar Date for the filing of proofs of claim or interest for all creditors of the Debtors, including, without limitation, holders of executory contracts and unexpired leases of the Debtors (other than for Administrative Expense Claims) of March 1, 2007 ("Bar Date") and on or before March 19, 2007, for governmental units. Paragraph 13 of the Bar Date Order provides: "Pursuant to Bankruptcy Rule 3003(c)(2), any entity that is required to file a proof of claim in these Cases pursuant to the Bankruptcy Code, the Bankruptcy Rules or this Order with respect to a particular claim against a Debtor, but that fails to properly do so by the applicable Bar Date, shall be forever barred, estopped and enjoined from: (a) asserting any claim against the Debtors that the entity has that (i) is in an amount that exceeds the amount, if any, that is identified in the Schedules on behalf of such entity as undisputed, noncontingent and liquidated or (ii) is of a different nature or a different classification than any claim identified in the Schedules on behalf of such entity; or (b) voting upon, or receiving distributions under, the Plan or any other plan of reorganization in these Cases in respect of an Unscheduled Claim." Accordingly, to the extent a proof of claim was not filed by the applicable bar date, such Claim shall not be an "Allowed Claim" as defined in the Plan.

#### G. Treatment and Status of Certain Claims

#### 1. Environmental Claims

The Plan Proponents are not aware of any environmental claims that have been timely filed by any government agency or Person alleging claims based on environmental contamination or reclamation costs.

#### 2. Tort and Litigation Claims

The Debtors, or CEP Mexico, are a party to a variety of legal proceedings and administrative actions. The legal actions pending against the Debtors, or CEP Mexico, are summarized below:

## (a) Civil Actions.

(i) Washington Penn Plastic Company, Inc. Case No. 06 CV 1224. United States District Court for the Northern District of Ohio. Nature of Relief – Breach of Contract. This claim also names certain non-Debtor insiders of the Debtors as defendants.

(ii) Advanced Communications Co. Case No. 06CV-11340. Akron Municipal Court Small Claims Division Summit County, Ohio. Nature of Relief – Complaint on Account.

(iii) KW Plastics, Recycling Division. Case No. CV2006000124. Pike County Circuit Court, Troy, Alabama. Nature of Relief – Complaint on Security Agreement.

(b) Employee Related Litigation. Maria Monk has filed a workers compensation claim against CEP.

Unless otherwise stated in the Plan, such claims, when and if Allowed, will be included in Class 4.

#### 3. Insurance Claims.

Notwithstanding the foregoing, all Insurance Claims that arose prior to the Petition Date shall be deemed Allowed in the greater of (i) an amount equal to (x) the amount of any remaining self-insured retention under the Insurance Policy applicable to such Insurance Claim, divided by (y) the number of Insurance Claims to which the applicable Insurance Policy and self-insured retention applies; and (ii) \$0.00. The allowance of an Insurance Claim pursuant to this provision shall not be deemed an admission of liability by CEP, Thermoplastics and/or Holdings, and therefore shall not be conclusive as to the Debtors' liability in any proceeding by the holder of an Insurance Claim against an Insurer.

#### 4. Workers' Compensation Claims

The Debtors conducted business in Ohio under self-insured programs for workers compensation coverage. Prior to the Petition Date, the Debtors allegedly learned from the Bureau of Workers' Compensation of the State of Ohio that they were not authorized by the state to operate under the state's self-insurance program for workers' compensation coverage. The Bureau of Workers' Compensation asserts a claim for premiums for the entire period in which the Debtors conducted operations in Ohio as

though they were self-insured. Likewise, the Bureau of Workers' Compensation asserts claims against CEP for a period prior to its ownership and control in August 2005. It is the position of the Co-Proponents that the Debtors paid all workers compensation claims during the period at issue and allowing the Bureau of Workers Compensation a claim for unpaid premiums would unjustly enrich the Bureau of Workers Compensation and is otherwise subject to disallowance. The Plan Proponents will attempt to amicably resolve matters with the Bureau of Workers' Compensation, and in the event that such negotiations are unsuccessful, an objection to the claim of the Bureau of Workers' Compensation will be filed. The Debtors had third party workers' compensation coverage in all other states where they had employees and conducted business. The Debtors' third-party insurer – Hartford Fire Insurance Company – filed a late proof of claim to which an objection will be filed.

#### 5. Reclamation Claims

On October 13, 2006, the Bankruptcy Court entered the Order Establishing Procedures for Resolving Reclamation Claims Asserted Against the Debtors and Granting Certain Related Relief (the "Reclamation Order").

The Reclamation Order required the Debtors to file a report on or before January 18, 2007, identifying those reclamation demands it recognized as valid, and those that were invalid (the "Reclamation Report"). The Reclamation Report evidences the receipt by the Debtors of seven (7) reclamation claims. The Reclamation Report establishes that six of the reclamation claims are invalid in accordance with applicable non-bankruptcy law and one reclamation claim totaling approximately \$119,166.35 is subject to defenses in accordance with Sixth Circuit case law. It is therefore estimated that there are no valid reclamation claims against the Debtors.

# H. Confirmation Hearing

The Bankruptcy Court will hold the Confirmation Hearing at the following time and place:

## **Confirmation Hearing**

**Date and Time:** Commencing at 1:30 p.m. (Eastern time), on July 24, 2007.

**Place:** The United States Bankruptcy Court, Northern District of Ohio (Eastern Division), Bankruptcy Courtroom Number 260, U.S. Courthouse and Federal Building, 2 South Main Street, Akron, Ohio 44308.

**Judge:** The Honorable Marilyn Shea-Stonum, United States Bankruptcy Judge, Northern District of Ohio.

The Confirmation Hearing may be adjourned from time to time on announcement in the Bankruptcy Court on the scheduled date for the hearing. No further notice will be required to adjourn the hearing.

At the Confirmation Hearing, the Bankruptcy Court will:

• determine whether sufficient majorities in number and dollar amount, as applicable, from each Voting Class have delivered properly executed votes accepting the Plan to approve the Plan;

- hear and determine objections, if any, to the Plan and to confirmation of the Plan that have not been previously disposed of;
- determine whether the Plan meets the confirmation requirements of the Bankruptcy Code; and
- determine whether to confirm the Plan.

Any objection to confirmation of the Plan must be in writing and filed and served as required by the Bankruptcy Court under the order approving this Disclosure Statement. That order requires any objections to the confirmation of the Plan to be served so as to be received on or before 4:00 p.m. on **July 13, 2007 at 4:00 p.m.** (EST), by the following persons:

- Counsel for the Debtors, BAKER & HOSTETLER LLP, 3200 National City Center, 1900 East 9th Street, Cleveland, OH 44114, Attn: Joseph F. Hutchinson, Jr., Esquire;
- Counsel for the Committee, MCGUIREWOODS LLP, Dominion Tower, 23rd Floor, 625 Liberty Avenue, Pittsburgh, PA 15222, Attn: Mark E. Freedlander, Esquire;
- The Office of the United States Trustee for the Northern District of Ohio, H.M. Metzenbaum U.S. Courthouse, 201 Superior Avenue East, Suite 441, Cleveland, OH 44114-1240, Attn: Maria Giannirakis, Esquire;

# I. Cancellation of Existing Securities and Agreements

On the Effective Date, except as expressly provided in the Plan, the securities, promissory notes, trust indentures, security agreements, deeds of trust, collateral agency agreement and other instruments evidencing or securing a Claim shall be deemed cancelled without further act or action under any applicable agreement or Law, and the obligations of the Debtors under the agreements, instruments, trust indentures and certificates governing or securing such Claims, as the case may be, shall be discharged.

#### J. Means of Executing the Plan

#### 1. Liquidating Trust

- (a) <u>Execution of Agreement</u>. On the Effective Date, the Agreement shall be executed, and all other necessary steps shall be taken to establish the CEP Liquidating Trust and the beneficial interests therein, which shall be for the benefit of all Beneficiaries.
- (b) <u>Purpose of CEP Liquidating Trust</u>. The CEP Liquidating Trust shall be established and maintained for the sole purpose of liquidating and distributing assets of the Estates and proceeds thereof, including but not limited to the \$5,408,597 in principal plus interest relating to CEP Mexico contributed to the CEP Liquidating Trust (and constituting Trust Assets), in accordance with Treasury Regulation section 30.1.7701-4(d), and resolving and administering claims, with no objective to continue or engage in the conduct of a trade or business.

- Liquidating Trust Assets. On the Effective Date, right, title and interest to all assets as described in Section 7.1(c) of the Plan shall re-vest in the CEP Liquidating Trust as Trust Assets. On the Effective Date, the Estates shall be deemed to assign, set over, transfer, and convey to the CEP Liquidating Trust all of their right, title and interest in assets of the Estates. To the extent that certain assets of the Estates, because of their nature or because they will accrue subsequent to the Effective Date, cannot be assigned, set over, transferred or conveyed to the CEP Liquidating Trust on the Effective Date, such assets shall be deemed assigned, set over, transferred and conveyed to the CEP Liquidating Trust as soon as practical after the Effective Date. On or after the Effective Date, the Liquidating Trustee shall continue as a plaintiff in all litigation or Causes of Action (on behalf of the Beneficiaries) in which the Debtors were plaintiffs prior to the Effective Date. All recoveries and proceeds arising from litigation and Causes of Action shall be deemed assigned, set over, transferred and conveyed to the CEP Liquidating Trust upon receipt thereof. All fees and costs of the Liquidating Trustee (and agents thereof) arising from or relating to pursuing litigation or Causes of Action or other services performed at the request of the Liquidating Trustee on behalf of the CEP Liquidating Trust shall be paid as a first priority in right of distribution from the CEP Liquidating Trust. The transfer of assets of the Estates to the Liquidating Trust shall be made for the benefit of the Beneficiaries, in each case, but only to the extent the Beneficiaries are entitled to distributions under the Plan.
- (d) <u>Governance of CEP Liquidating Trust</u>. The CEP Liquidating Trust shall be governed by the Liquidating Trustee in accordance with the Agreement and consistent with the Plan.
- (e) <u>The Liquidating Trustee</u>. The Liquidating Trustee shall be designated on or before the Effective Date by the Committee. The designation of the Trustee shall be effective on the Effective Date without the need for a further order of the Bankruptcy Court. The Trustee shall exercise reasonable business judgment to administer Trust Assets and to make timely distributions from the CEP Liquidating Trust to its Beneficiaries.
- (f) <u>Nontransferability of CEP Liquidating Trust Interests</u>. The beneficial interests in the CEP Liquidating Trust shall not be certificated and are not transferable (except as otherwise provided in the Agreement).
- Assets (including any earnings thereon or proceeds therefrom) in short term overnight investments or longer term investments as permitted by section 345 of the Bankruptcy Code; provided, however, that such investments are investments permitted to be made by a liquidating trust within the meaning of Treasury Regulation section 301.7701-4(d), as reflected therein, or under applicable Internal Revenue Service guidelines, rulings, or other controlling authorities.
- (h) <u>Costs and Expenses of the Liquidating Trustee</u>. The costs and expenses of the CEP Liquidating Trust, including the fees and expenses of the Trustee and its retained professionals, shall be paid out of the CEP Liquidating Trust. Such costs and expenses shall be treated as administrative expenses of the Estates, having, however, a first priority in right of distribution, and without the need for Bankruptcy Court authorization or entry of a Final Order. The Liquidating Trustee shall retain such amounts as are reasonably necessary (at the discretion of the Liquidating Trustee) to meet the future expenses of administering the CEP Liquidating Trust in the Liquidation Expense Reserve Account.

- (i) <u>Compensation of the Liquidating Trustee</u>. The Liquidating Trustee shall be entitled to reasonable compensation, which shall be on economic terms agreeable to the Committee and established prior to the Effective Date. The compensation payable to the Liquidating Trustee shall be treated consistent with section 7.1(h) of the Plan.
- (j) <u>Distribution of Trust Assets</u>. The Liquidating Trustee shall distribute at least annually and in accordance with the Agreement, beginning on the Effective Date or as soon thereafter as is practicable, the Trust Assets on hand (including any Cash received from the Debtors and/or CEP Mexico on the Effective Date, and treating as Cash for purposes of section 7.1 of the Plan any permitted investments under section 7.1(g) of the Plan), except such amounts (i) as would be distributable to a holder of a Disputed Claim if such Disputed Claim had been Allowed prior to the time of such distribution (but only until such Claim is resolved); (ii) as are reasonably necessary to meet contingent liabilities and to maintain the value of the assets of the CEP Liquidating Trust during liquidation; (iii) to pay reasonable expenses (including, but not limited to, any taxes imposed on the assets of the CEP Liquidating Trust or in respect of the CEP Liquidating Trust); and (iv) to satisfy other liabilities incurred by the CEP Liquidating Trust in accordance with this Plan or the Agreement.
- (k) <u>Retention of Professionals by the Liquidating Trustee</u>. The Liquidating Trustee may retain and reasonably compensate counsel and other professionals to assist in its duties as Liquidating Trustee on such terms as the Trustee deems appropriate without Bankruptcy Court approval. The Liquidating Trustee may retain any professional who represented parties in interest (including the Debtors, CEP Mexico or the Committee) in the Cases.
- (l) <u>Authority to Grant Releases</u>. Without limiting the generality of section 7.6 of the Plan, in connection with the compromise and settlement of any Causes of Action, the Liquidating Trustee is authorized to release and discharge, to the fullest extent permitted by law, non-Debtor parties to Causes of Action from all Claims and Causes of Action that, the Committee or the Liquidating Trustee (as successor to the Debtors on behalf of the Beneficiaries) has or may have whether known or unknown against such Persons. Any settlement effectuated prior to the Confirmation Date, upon notice thereof to the Bankruptcy Court, shall deemed incorporated into the Plan and entry of the Confirmation Order including provisions of such settlement shall be deemed a settlement pursuant to section 1123(6)(3)(A) of the Bankruptcy Code.
- (m) <u>Attorney-Client Privilege</u>. Any attorney-client privilege, work-product privilege or other privilege or immunity that the Estates or Committee are entitled to assert in the Causes of Action shall vest in the Liquidating Trustee (and the attorneys and agents of the Liquidating Trustee) and the Liquidating Trustee shall be entitled to assert such privilege and immunity to the same extent that the Debtors and Committee were entitled to do so prior to the Effective Date.
- (n) Transfer of Claims of Debtors, the Estates to CEP Liquidating Trust. All assets of the Debtors and the Estates, including but not limited to the principal amount of \$5,408,547, plus interest accrued thereon, are to be transferred to and deemed property of the CEP Liquidating Trust for the benefit of the Beneficiaries on the Effective Date. Included among such assets are Causes of Action. A full investigation of potential Causes of Action has not been undertaken by either the Debtors or the Committee. It is, however, the Committee's belief that potential Causes of Action may exist as against, among others, (i) the Carlisle Companies and/or Parker-Hannifin Corporation and/or their respective subsidiaries, affiliates, agents or other Persons arising from or relating to the Carlisle Transaction and/or the Parker-Hannifin Transaction; (ii) Persons (and/or subsidiaries and affiliates thereof) identified in the Debtors' items 3(a) and 3(b) of the respective Debtors' Statements of Financial

Affairs with respect to Avoidance Actions (and other Persons subsequently discovered by the Committee or Trustee as having received payments that should have been listed and identified in item 3(a) or 3(b) of the respective Debtors' Statements of Financial Affairs and were not); (iii) unless otherwise settled, any one or more of the Participating Customers, other customers of the Debtors and/or CEP Mexico, or debtors of the Debtors or CEP Mexico, for failure to pay accounts receivable or other amounts due to the Debtors (including Cash Infusions) and/or CEP Mexico in accordance with, inter alia, the Final DIP Order and/or its related agreements, other contracts and course of conduct as between the Debtors, CEP Mexico and any customer of the Debtors or CEP Mexico with unpaid accounts due and owing to the Debtors or CEP Mexico; (iv) other Causes of Action identified by the Liquidating Trustee in the course of his post-Effective Date due diligence.

#### (o) Federal Income Tax Treatment of CEP Liquidating Trust.

(i) Trust Assets Treated as Owned by Creditors. For all federal income tax purposes, all parties (including, without limitation, the Debtors, the Liquidating Trustee, and the holders of beneficial interests in the CEP Liquidating Trust) shall treat the transfer of the Trust Assets to the CEP Liquidating Trust for the benefit of the Beneficiaries thereof, whether Allowed on or after the Effective Date, as (A) a transfer of the Trust Assets directly to the holders of satisfaction of such Claims (other than to the extent allocable to Disputed General Unsecured Claims) followed by (B) the transfer by such holders to the CEP Liquidating Trust of the Trust Assets in exchange for beneficial interests in the CEP Liquidating Trust. Accordingly, the Beneficiaries shall be treated for federal income tax purposes as the grantors and owners of their respective shares of the Trust Assets.

# (ii) <u>Tax Reporting</u>.

(A) The Liquidating Trustee shall file returns for the CEP Liquidating Trust as a grantor trust pursuant to Treasury Regulation section 1.671-1 through 4(a) and in accordance with this section 7.1(n) of the Plan. The Liquidating Trustee shall also annually send to each record holder of a beneficial interest a separate statement setting forth the holder's share of items of income, gain, loss, deduction, or credit and will instruct all such holders to report such items on their federal income tax returns or to forward the appropriate information to the beneficial holders with instructions to report such items on their federal income tax returns. The CEP Liquidating Trust's taxable income, gain, loss, deduction, or credit will be allocated (subject to section 7.1(n)(ii)(C) of the Plan, relating to Disputed General Unsecured Claims) to the holders of Allowed General Unsecured Claims in accordance with their relative beneficial interests in the CEP Liquidating Trust.

(B) As soon as possible after the Effective Date, but in no event later than ninety (90) days after the Effective Date, the Liquidating Trustee shall make a good faith valuation of the Trust Assets. Such valuation shall be made available from time to time, to the extent relevant, and used consistently by all parties (including, without limitation the Debtors, the Liquidating Trustee, and the holders of Allowed General Unsecured Claims) for all federal income tax purposes. The Liquidating Trustee shall also file (or cause to be filed) any other statements, returns, or disclosures relating to the CEP Liquidating Trust that are required by any governmental unit.

(C) Subject to definitive guidance from the Internal Revenue Service or a court of competent jurisdiction to the contrary (including the receipt by the Trustee of a private letter ruling if the Liquidating Trustee so requests one, or the receipt of an adverse determination by the Internal Revenue Service upon audit if not contested by the Liquidating Trustee), the Trustee shall (i) treat any Trust Assets allocable to, or retained on account of, Disputed Claims as held by one or more

discrete trusts for federal income tax purposes (the "Disputed Claim Reserve") in accordance with the trust provisions of the Tax Code (section 641 et seq.); (ii) treat as taxable income or loss of the Disputed Claim Reserve, with respect to any given taxable year, the portion of the taxable income or loss of the CEP Liquidating Trust that would have been allocated to the holders of Disputed Claims had such Claims been Allowed on the Effective Date (but only for the portion of the taxable year with respect to which such Claims are unresolved); (iii) treat as a distribution from any reserve established under CEP Liquidating Trust any increased amounts distributed by the CEP Liquidating Trust as a result of any Disputed Claims resolved earlier in the taxable year, to the extent such distributions relate to taxable income or loss of the CEP Liquidating Trust determined in accordance with the provisions hereof; and (iv) to the extent permitted by applicable law, report consistent with the foregoing for state and local income tax purposes. All holders of Allowed General Unsecured Claims shall report, for tax purposes, consistent with the foregoing.

(D) The Liquidating Trustee shall be responsible for payments, out of the Trust Assets, of any taxes imposed on the CEP Liquidating Trust or the Trust Assets, including any Disputed Claim Reserve established under the CEP Liquidating Trust. In the event, and to the extent, any Cash retained on account of Disputed General Unsecured Claims in any Disputed Claim Reserve established under the CEP Liquidating Trust is insufficient to pay the portion of any such taxes attributable to the taxable income arising from the assets allocable to, or retained on account of, Disputed General Unsecured Claims, such taxes shall be (i) reimbursed from any subsequent Cash amounts retained on account of Disputed General Unsecured Claims, or (ii) to the extent such Disputed General Unsecured Claims have subsequently been resolved, deducted from any amounts distributable by the Liquidating Trustee as a result of the resolutions of such Disputed General Unsecured Claims.

(E) The Liquidating Trustee may request an expedited determination of taxes of the CEP Liquidating Trust under section 505(b) of the Bankruptcy Code for all returns filed for, or on behalf of, the CEP Liquidating Trust for all taxable periods through the dissolution of the CEP Liquidating Trust.

shall be discharged or dissolved, as the case may be, at such time as (i) all Disputed General Unsecured Claims have been resolved, (ii) all Trust Assets have been liquidated and (iii) all distributions required to be made by the Liquidating Trustee under the Plan have been made, but in no event shall the Liquidating Trust be dissolved later than a period that would not adversely affect the status of the CEP Liquidating Trust as a liquidating trust for federal income tax purposes. The CEP Liquidating Trust shall only exist for a period as long as is necessary to facilitate or complete the recovery and liquidation of the Trust Assets and the disposition of their proceeds. The Liquidating Trustee shall not unduly prolong the duration of the CEP Liquidating Trust and shall at all times endeavor to resolve, settle or otherwise dispose of all claims that constitute Trust Assets and to effect the distribution of the Trust Assets in accordance with the terms hereof and terminate the CEP Liquidating Trust as soon as practicable. Prior to and upon termination of the CEP Liquidating Trust, the Trust Assets will be distributed to the Beneficiaries of CEP Liquidating Trust, pursuant to the provisions set forth in sections V.E and V.F of the CEP Liquidating Trust and Articles 2 and 4 of the Plan.

(q) <u>Indemnification of Liquidating Trustee</u>. The Liquidating Trustee, and the Liquidating Trustee's agents and professionals shall not be liable for actions taken or omitted in its capacity as, or on behalf of, the Trustee, except those acts arising out of its or their own willful misconduct, gross negligence, bad faith, self-dealing, breach of fiduciary duty or ultra vires acts, and each shall be entitled to indemnification and reimbursement for fees and expenses in defending any and all of

its actions or inactions in its capacity as, or on behalf of, the Liquidating Trustee, except for any actions or inactions involving willful misconduct, gross negligence, bad faith, self-dealing, breach of fiduciary duty; or ultra vires acts. Any indemnification claim of the Liquidating Trustee (and the other parties entitled to indemnification under this subsection 7.1(p) of the Plan) shall be satisfied first from the Trust Assets. The Liquidating Trustee shall be entitled to rely, in good faith, on the advice of its retained professionals.

#### 2. Distributions to Holders as of the Confirmation Date.

As of the close of business on the Confirmation Record Date, the Claims Register, the equity register and transfer and other registers as maintained by the Debtors and/or their respective agents, as applicable, will be closed and there will be no further changes in the record holder of any Claim or Equity Interest. The Liquidating Trustee will have no obligation to recognize any transfer of any Claim or Equity Interest occurring after the Confirmation Date. The Liquidating Trustee will instead be authorized and entitled to recognize and deal for all purposes under the Plan with only those record holders stated on the Claims Register and other registers as of the close of business on the Confirmation Date.

# 3. Closing of Cases by Charitable Gift.

If at any time the Liquidating Trustee determines with consent of the Committee that the expense of administering the CEP Liquidating Trust so as to make a final distribution to Allowed Claims is likely to exceed the value of the Trust Assets remaining in the CEP Liquidating Trust, the Liquidating Trustee shall apply to the Bankruptcy Court for authority to (i) reserve any amounts necessary to close the Cases, (ii) donate any balance to a charitable organization exempt from federal income tax under section 501(c)(3) of the Tax Code that is unrelated to the Debtors and any insider of the Debtors and (iii) close the Cases in accordance with the Bankruptcy Code and Bankruptcy Rules. Notice of such application shall be given electronically, to the extent practicable, to those parties who have filed requests for notices and whose electronic addresses remain current and operating.

#### 4. Release of Liens.

Except as otherwise specifically provided in or contemplated by the Plan or in any contract, instrument or other agreement or document created in connection with the Plan, (i) each holder of: (a) any purported Secured Claim and/or (b) any judgment, personal property or ad valorem tax, molder, warehouse or artisan or similar Lien Claim, in each case regardless of whether such Claim is an Allowed Claim, shall, on or immediately before the Effective Date and regardless of whether such Claim has been scheduled or proof of such Claim has been filed: (y) turn over and release to the Estates any and all property of the Debtors or Estates that secures or purportedly secures such Claim, or such Lien and/or Claim shall automatically, and without further action by the Debtors or the Liquidating Trustee, as the case may be, be deemed released and (z) execute such documents and instruments as the Debtors or Liquidating Trustee, as the case may be, requires to evidence the holder of a Claim's release of such property or Lien, and if such holder refuses to execute appropriate documents or instruments, the Debtors, CEP Mexico or the Liquidating Trustee, as the case may be, in its discretion, file a copy of the Confirmation Order in the appropriate recording office, which shall serve to release any holder of a Claim's rights in such property; and (ii) on the Effective Date, all right, title and interest in such property shall revert or be transferred to the CEP Liquidating Trust as Trust Assets, free and clear of all Claims, interests, and Liens of any kind.

# 5. Cancellation of Existing Securities and Agreements.

On the Effective Date, except as expressly provided in the Plan, the securities, promissory notes, trust indentures, share certificates, security agreements, deeds of trust, collateral agency agreements and other instruments evidencing or securing a Claim shall be deemed cancelled without further act or action under any applicable agreement or Law, and the obligations of the Debtors and the Liquidating Trustee, as successor to the Debtors under the agreements, instruments, trust indentures and certificates governing and securing such Claims and Equity Interests, as the case may be, shall be discharged.

# 6. <u>Liquidating Trustee's Post-Confirmation Role.</u>

All rights and obligations of the Debtors under this Plan that exist or continue on or after the Effective Date shall vest in the Liquidating Trustee and shall be rights and obligations exercisable by the Liquidating Trustee on and after the Effective Date. Further, the Liquidating Trustee shall perform each of the following acts as soon as practicable on or after the Effective Date:

- (a) <u>General Powers</u>. In furtherance of and consistent with the purpose of the CEP Liquidating Trust and the Plan, the Liquidating Trustee shall (A) have the power and authority to hold, manage, sell and distribute the Trust Assets in accordance with the Plan, (B) have the power and authority to prosecute, defend and resolve, in the name of the Debtors, CEP Mexico and/or the name of the Liquidating Trustee, Causes of Action, (C) have the power and authority to prosecute and resolve objections to Disputed Claims, (D) have the power and authority to perform such other functions as are provided in the Plan, and (E) have the power and authority to administer the closure of the Cases. The Liquidating Trustee shall be responsible for all decisions and duties with respect to the CEP Liquidating Trust and the Trust Assets. In all circumstances, the Liquidating Trustee shall act in the best interests of all beneficiaries of the CEP Liquidating Trust and in furtherance of the purpose of the CEP Liquidating Trust.
- (b) <u>Liquidating Payments and Transfers</u>. On the Effective Date, or as soon thereafter as is reasonably practicable, the Trustee, as disbursing agent, shall make payments and transfers to holders of Allowed Claims to claimants in the manner set forth at sections 2.1, 2.2, 2.3, 2.4, 2.5, 4.1, 4.2, 4.3 and 4.4 of the Plan, and in accordance with section 6.2 of the Plan.
- exercise all powers regarding the Debtors' tax matters, including filing tax returns, to the same extent as if the Liquidating Trustee were the debtor in possession. The Liquidating Trustee shall (A) complete and file within ninety (90) days of the filing for dissolution by CEP, Thermoplastics, Holdings and CEP Mexico, to the extent not previously filed, the Debtors' final federal, state, and local tax returns; (B) request an expedited determination of any unpaid tax liability of the Debtors under section 505(b) of the Bankruptcy Code for all tax periods of the Debtors ending after the Petition Date through the liquidation of the Debtors as determined under applicable tax laws, to the extent not previously requested; and (C) represent the interest and account of the Debtors before any taxing authority in all matters, including, but not limited to, any action, suit, proceeding, or audit.
- (d) <u>Claims Administration and Prosecution and Plan Distributions</u>. The Liquidating Trustee shall have the power and authority to prosecute and resolve objections to Disputed Subordinated Participating Customers Secured Claims, Disputed Administrative Expense Claims, Disputed Priority Tax Claims, Disputed Priority Non-Tax Claims and Disputed General Unsecured Claims. The Liquidating Trustee shall have the right, power and authority

to retain and assert all defenses, rights of setoff and counterclaims with respect to each of the foregoing. The Liquidating Trustee shall also have the power and authority to hold, manage and distribute Plan distributions to the holders of Allowed Claims consistent with applicable provisions of this Plan.

Notwithstanding the foregoing, all Insurance Claims that arose prior to the Petition Date shall be deemed Allowed in the greater of (i) an amount equal to (x) the amount of any remaining self-insured retention under the Insurance Policy applicable to such Insurance Claim, divided by (y) the number of Insurance Claims to which the applicable Insurance Policy and self-insured retention applies; and (ii) \$0.00. The allowance of an Insurance Claim pursuant to this provision shall not be deemed an admission of liability by CEP, Thermoplastics and/or Holdings, and therefore shall not be conclusive as to the Debtors' liability in any proceeding by the holder of an Insurance Claim against an Insurer.

#### 7. Books and Records.

Unless applicable non-bankruptcy law permits the distribution or destruction of certain of the Debtors' and CEP Mexico's business records at an earlier date, the Liquidating Trustee shall have the responsibility of storing and maintaining books and records until one year after the Effective Date, after which time such books and records may, at the sole discretion of the Liquidating Trustee, be abandoned or destroyed without further Bankruptcy Court order unless applicable non-bankruptcy law requires the retention and maintenance of any such books and records for a longer period, in which instance the Liquidating Trustee shall retain such books and records for at least the minimum period required by applicable non-bankruptcy law. For purposes of this section, books and records include computer generated or computer maintained books and records and computer data, as well as electronically generated or maintained books and records or data, along with books and records of the Debtors and CEP Mexico maintained by or in possession of third parties and all of the claims and rights of the Debtors and CEP Mexico in and to their books and records, wherever located.

#### 8. Corporate Action.

On the Effective Date, the Liquidating Trustee, shall perform each of the actions and effect each of the transfers required by the terms of the Plan in the time period allocated therefor, and all matters provided for under the Plan that would otherwise require approval of the stockholders, directors or comparable governing bodies of the Debtors shall be deemed to have occurred and shall be in effect from and after the Effective Date pursuant to the applicable general corporation law (or other applicable governing law) of the state in which the Debtors incorporated or organized, without any requirement of further action by the stockholders or directors (or other governing body) of the Debtors.

#### 9. Effectuating Documents and Further Transactions.

The Debtors and/or Liquidating Trustee, as the case may be, are authorized and directed to execute, deliver, file, or record such contracts, instruments, releases, indentures and other agreements or documents and take such actions as may be necessary or appropriate to effectuate and further evidence the terms and conditions of the Plan.

# 10. <u>Securities Exempt.</u>

The beneficial interests in the CEP Liquidating Trust satisfy the requirements of section 1145 of the Bankruptcy Code and, therefore, such interests are exempt from registration under the Securities Act of 1933, as amended, and any state or local law requiring registration.

#### 11. Distribution Procedures

- (a) <u>Effective Date Payments and Transfers by the Debtors</u>. On the Effective Date, the Debtors shall remit to the Liquidating Trustee all assets and property of the Estates, and the Liquidating Trustee shall pay as soon as practicable holders of Allowed Claims pursuant to sections 2.1, 2.2, 2.3, 2.4, 2.5, 4.1, 4.2 and 4.3 of the Plan an amount in Cash equal to the Allowed amount of such Claims or in such other amounts as set forth in the Plan. The transfer of the assets as described in the preceding sentence to the CEP Liquidating Trust shall be made for the benefit of the Beneficiaries, in each case, but only to the extent that the Beneficiaries are entitled to distributions under the Plan.
- Disbursing Agent. All distributions under the Plan will be made by the Liquidating Trustee. The Liquidating Trustee will not be required to give any bond or surety or other security for the performance of its duties unless otherwise ordered by the Bankruptcy Court, and in the event that the Disbursing Agent is otherwise so ordered, all costs and expenses of procuring any such bond or surety will be borne by the CEP Liquidating Trust. The Liquidating Trustee shall be empowered to (i) effect all actions and execute all agreements, instruments and other documents necessary to perform its duties under the Plan and/or CEP Liquidating Trust, (ii) make all distributions contemplated in the Plan, (iii) employ professionals to represent it with respect to its responsibilities under the Plan and (iv) exercise such other powers as may be vested in the Liquidating Trustee by order of the Bankruptcy Court, pursuant to the Plan and/or CEP Liquidating Trust, or as deemed by the Liquidating Trustee to be necessary and proper to implement the provisions of the Plan. Except as otherwise ordered by the Bankruptcy Court, the amount of any reasonable fees and expenses incurred by the Liquidating Trustee on or after the Effective Date (including without limitation, taxes) and any reasonable compensation and expense reimbursement claims (including, without limitation, reasonable attorneys' fees and other professional fees and expenses) made by the Liquidating Trustee shall be paid in Cash subject to the terms and conditions of the CEP Liquidating Trust.
- (c) <u>Distributions of Cash</u>. At the option of the Liquidating Trustee, any Cash payment to be made hereunder may be made by check or wire transfer or as otherwise required or provided in applicable agreements.
- Delivery of Distributions. Subject to Bankruptcy Rule 9010, unless (d) otherwise provided in the Plan, all distributions to any holder of an Allowed Claim will be made to the holder of each Allowed Claim at the address of such holder as listed in the Schedules, or on the books and records of the Debtors' claims agent BMC unless the Debtors or Liquidating Trustee, as the case may be, have been notified, in advance, in writing of a change of address, including, without limitation, by the timely filing of a proof of claim or interest by such holder that provides an address for such holder different from the address reflected in the Schedules or in the Debtors' books and records. In the event that any distribution to any holder is returned as undeliverable, no distribution to such holder will be made unless and until the Liquidating Trustee has been notified of the then current address of such holder, at which time or as soon as reasonably practicable thereafter, such distribution will be made to such holder without interest; provided, however, that, such undeliverable distributions will be deemed unclaimed property under section 347(b) of the Bankruptcy Code at the expiration of ninety (90) days after the date of distribution in accordance with the section on unclaimed distributions below. The Liquidating Trustee will have no obligation to attempt to locate any holder of an Allowed Claim other than by reviewing the Schedules and its books and records maintained by BMC (including any proofs of claim filed against the Debtors).

- (e) <u>Minimum Distributions</u>. No payment of Cash less than \$50 shall be made by the Liquidating Trustee. Any assets of the Estates and/or Trust Assets that are undistributable in accordance with this section 6.5 of the Plan shall vest in the CEP Liquidating Trust for the benefit of its Beneficiaries.
- (f) <u>Date of Distributions</u>. Unless otherwise expressly provided herein, any distributions or deliveries to be made under the Plan shall be made in accordance with the provisions of the Plan and the Agreement, in that order.

# 12. Transfer of Claims of Debtors, the Estates to CEP Liquidating Trust.

Included among assets and property of the Estates referred to in subsection V(I)(ii)(a) are Causes of Action. A full investigation of potential Causes of Action has not been undertaken by either the Debtors or the Committee. It is, however, the Committee's belief that potential Causes of Action may exist as against, among others, (i) the Carlisle Companies and/or Parker-Hannifin Corporation and/or their respective subsidiaries, affiliates, agents or other Persons arising from or relating to the Carlisle Transaction and/or the Parker-Hannifin Transaction; (ii) Persons (and/or subsidiaries and affiliates thereof) identified in the Debtors' items 3(a) and 3(b) of the respective Debtors' Statements of Financial Affairs with respect to Avoidance Actions (and other Persons subsequently discovered by the Committee or Liquidating Trustee as having received payments that should have been listed and identified in item 3(a) or 3(b) of the respective Debtors' Statements of Financial Affairs and were not); and (iv) other Causes of Action identified by the Committee or Liquidating Trustee in the course of their post-Effective Date due diligence.

# K. Treatment of Disputed, Contingent and Unliquidated Claims

## 1. <u>Disputed Claims</u>

The Debtors, and from and after the Effective Date the Liquidating Trustee, shall be entitled to object to Administrative Expense Claims, Priority Tax Claims, Subordinated Participating Customers Secured Claims, Other Secured Claims, Priority Non Tax Claims and General Unsecured Claims.

(a) <u>No Distribution Pending Allowance of Claims</u>. Notwithstanding any other provision hereof, if any portion of a Claim is a Disputed Claim, no payment or distribution provided hereunder shall be made on account of such Claim unless and until such Disputed Claim becomes an Allowed Claim.

#### (b) Reserve on Account of Disputed Claims.

(i) <u>Establishment and Maintenance of Reserve for Disputed Claims</u>. The Liquidating Trustee shall maintain a reserve (the "Disputed Claims Reserve") at an amount equal to the aggregate of 100% of the distributable amounts to which holders of such Disputed Claims would be entitled under the Plan if such Disputed Claims were Allowed Claims in their Disputed Claim Amounts or such lesser amount as required by a Final Order. For purposes of effectuating the provisions of this Section 8.13(a) and the distributions to holders of Allowed Claims, the Bankruptcy Court shall set, may fix or liquidate the amount of Disputed Claims pursuant to section 502(c) of the Bankruptcy Code, in which event the amounts so fixed or liquidated shall be deemed the amounts of the Disputed Claims for purposes of distribution under this Plan. In lieu of fixing or liquidating the amount of any Disputed Claim, the Bankruptcy Court may determine the amount to be reserved for such Disputed Claim or such

amount may be fixed by agreement in writing between the Liquidating Trustee and the holder of a Disputed Claim.

- (ii) <u>Distributions Upon Allowance of Disputed Claims</u>. The holder of a Disputed Claim that becomes an Allowed Claim shall receive distributions in Cash from the Disputed Claim Reserve as soon as practical following the date on which such Disputed Claim becomes an Allowed Claim pursuant to a Final Order. Such distributions shall be made in accordance with the Plan based upon the distributions that would have been made to such holder under the Plan if the Disputed Claim had been an Allowed Claim on or prior to the Effective Date. No holder of a Disputed Claim shall have any Claim against the Disputed Claim Reserve or the CEP Liquidating Trust with respect to such Claim until such Disputed Claim shall become an Allowed Claim, and no holder of a Disputed Claim shall have any right to interest on such Disputed Claim except for any proportional Reserve Income earned on the Disputed Claim Reserve.
- (c) <u>Resolution of Disputed Claims</u>. Unless otherwise ordered by the Bankruptcy Court after notice and a hearing, following the Effective Date, the Trustee shall have the right to the exclusion of all others (except as to applications for allowances of compensation and reimbursement of expenses under sections 328(a), 330 and 503 of the Bankruptcy Code) to make and file objections to Claims and shall serve a copy of each objection upon the holder of the Claim to which the objection is made as soon as practicable, but in no event later than one hundred eighty (180) days after the Effective Date (subject, however, to the right of the Liquidating Trustee to seek an extension of time to file such objections by seeking such extension with approval from the Bankruptcy Court).
- may at any time request that the Bankruptcy Court estimate any contingent, unliquidated or Disputed Claim pursuant to section 502(c) of the Bankruptcy Code regardless of whether the Debtors previously objected to such Claim, and the Bankruptcy Court will retain jurisdiction to estimate any Claim at any time during litigation concerning any objection to any Claim, including, without limitation, during the pendency of any appeal relating to any such objection. In the event that the Bankruptcy Court estimates any contingent, unliquidated or Disputed Claim, the amount so estimated shall constitute either the Allowed amount of such Claim or a maximum limitation on such Claim, as determined by the Bankruptcy Court. If the estimated amount constitutes a maximum limitation on the amount of such Claim, the Debtors or the Liquidating Trustee (as the case may be) may pursue supplementary proceedings to object to the allowance of such Claim. All of the aforementioned objection, estimation and resolution procedures are intended to be cumulative and not exclusive of one another. On and after the Confirmation Date, Claims that have been estimated may be compromised, settled, withdrawn or otherwise resolved subsequently, without further order of the Bankruptcy Court.
- (e) <u>Distributions to Holders of Allowed Claims Upon Disallowance of Disputed Claims</u>. Subject to section 7.2 of the Plan, upon disallowance of any Disputed Claim, each holder of an Allowed Claim in the same Class as the disallowed Disputed Claim will be entitled to its Pro Rata Share of Cash equal to the distribution that would have been made in accordance with the Plan to the holder of such Disputed Claim had such Disputed Claim been an Allowed Claim on or prior to the Effective Date. Such distributions on account of disallowed Disputed Claims will be made as soon as practicable after the fifteenth business day following allowance or disallowance of the last Disputed Claim. Upon allowance or disallowance of all or a portion of such Disputed Claims, the Debtor, or the Liquidating Trustee, as the case may be, will make appropriate distributions in accordance with the Plan.

#### L. The Committee

The Committee shall disband and be released of its duties and obligations on the Effective Date. Any rights or duties of the Committee under the Agreement shall be exercised by the co-chairs of the Committee as of the Confirmation Date, which Committee co-chairs shall continue to take all actions required of the Committee as of the Effective Date, consistent with section 1103 of the Bankruptcy Code.

## M. Vesting of Property and Discharge of Claims

## 1. <u>Vesting of Assets</u>

As of the Effective Date, all property of the Estates shall vest in the CEP Liquidating Trust, and shall constitute the Trust Assets.

As of the Effective Date, all assets of the Estates shall be free and clear of all Claims and Liens, except as provided in the Plan or the Confirmation Order.

## 2. Discharge of Claims and Debts and Termination of Equity Interests

Except as otherwise specifically provided herein or in the Confirmation Order, the rights afforded in the Plan and the payments and distributions to be made hereunder shall discharge all existing debts and Claims, and terminate all Equity Interests of any kind, nature or description whatsoever, other than Consolidated Holdings, against or in the Estates or any of their assets or properties to the fullest extent permitted by section 1141 of the Bankruptcy Code. Except as otherwise provided herein or in the Confirmation Order, upon the Effective Date, all existing Claims against the Estates shall be, and shall be deemed to be, discharged and terminated, and all holders of Claims and Equity Interests shall be precluded and enjoined from asserting against the Estates, or any of its assets or properties, any other or further Claim or Equity Interest based upon any act or omission, transaction or other activity of any kind or nature that occurred prior to the Effective Date, whether or not such holder has filed a proof of Claim or is listed in the Schedules.

# N. Exculpation

None of the Debtors, Committee nor any of their respective members, officers, directors, employees, advisors, professionals or agents, (collectively, "Exculpation Parties") shall have or incur any liability to any holder of a Claim or Equity Interest for any act or omission in connection with, related to, or arising out of the Cases, including, without limitation, negotiations regarding or concerning the Plan, the pursuit of confirmation of the Plan, the consummation of the Plan, or the administration of the Estates or the Plan or the property to be distributed under the Plan, except for willful misconduct or gross negligence, and, in all respects, the Exculpation Parties shall be entitled to rely upon the advice of counsel with respect to their duties and responsibilities under the Plan.

#### O. Releases

Various documents previously approved by or pending before the Bankruptcy Court provide for releases or contemplate (subject to Bankruptcy Court approval) providing releases on the part of the Debtors, the Estates and the Committee. The Final DIP Order releases Wachovia (and the Participating Customers with respect to certain Causes of Action) of pre-Petition Date claims. As described in section III (F), herein, the Debtors and Committee have agreed (subject to Bankruptcy Court approval) to release

The Reserve Group Management Company and certain related parties of claims of Causes of Action in consideration of the proposed assumption of the Pension Plans by Superior Fabrication Company, LLC. It is contemplated by the Plan Proponents that other releases may be proposed as additional issues in the Cases are resolved. Prior to the Confirmation Date, any and all such releases shall remain subject to approval by the Bankruptcy Court.

# P. Executory Contracts

On the Effective Date, all executory contracts and unexpired leases to which any one or more Debtors is a party shall be deemed rejected as of the Effective Date, except for an executory contract or unexpired lease that (i) has been assumed or rejected pursuant to Final Order of the Bankruptcy Court prior to the Effective Date, (ii) is a collective bargaining agreement governed by section 1113 of the Bankruptcy Code or an agreement providing for retiree benefits covered by section 1114 of the Bankruptcy Code, or (iii) is subject to separate motion to assume or reject (or terminate or modify, as the case may be) filed under sections 365, 1113 and/or 1114 of the Bankruptcy Code by the Debtors prior to the Effective Date

Notwithstanding the foregoing, any agreements, documents or instruments relating thereto (including Insurance Policies) that, are postpetition contracts shall continue to operate unaffected by the Plan, with the Insurers responsible for claims in accordance with the terms and provisions of such post petition contracts (including Insurance Policies) unless otherwise terminated by the Liquidating Trustee. The Debtors' Insurance Policies that have expired as of the Confirmation Date (whether entered into prior or subsequent to the Petition Date) are not executory contracts subject to assumption or rejection. The Insurers shall be responsible for continuing coverage obligations under such Insurance Policies regardless of the payment status of any retrospective or other insurance premiums.

Nothing contained in the Plan shall constitute or be deemed to be a waiver of any cause of action that the Debtors may hold against any Person, including, without limitation, any Insurer under any of the Debtors' Insurance Policies.

Entry of the Confirmation Order shall constitute the approval, pursuant to sections 365(a), 1113 and/or 1114 of the Bankruptcy Code, of the rejection of the executory contracts and unexpired leases rejected as of the Effective Date pursuant to the Plan.

Bar Date for Filing Proofs of Claim Relating to Executory Contracts and Unexpired Leases Rejected by Operation of the Plan. In the event that the rejection of an executory contract or unexpired lease by the Debtors pursuant to the Plan results in damages to the other party or parties to such contract or lease, a Claim for such damages, if not heretofore evidenced by a filed proof of claim, shall be forever barred and shall not be enforceable against the Debtors or any property to be distributed under the Plan unless a proof of claim is filed with the Bankruptcy Court and served upon the Trustee on or before the date that is thirty (30) days after the Confirmation Date. The foregoing sentence shall not, however, be applicable to any separate pre-Confirmation Date order of the Bankruptcy Court authorizing rejection of an executory contract or unexpired lease wherein a separate deadline by which rejection damages claims was established.

# Q. Retention of Jurisdiction of the Bankruptcy Court

The Bankruptcy Court shall retain jurisdiction of all matters arising under, arising out of, or related to the Cases and the Plan pursuant to, and for the purposes of, sections 105(a) and 1142 of the Bankruptcy Code and for, among other things, the following purposes: (a) to hear and determine motions for the assumption, assumption and assignment, or rejection of executory contracts or unexpired leases and the allowance of Claims resulting therefrom; (b) to determine any motion, adversary proceeding, application, contested matter, and other litigated matter pending on or commenced after the Confirmation Date, including, without limitation, any proceeding to recover a Cause of Action; (c) to ensure that distributions to holders of Allowed Claims are accomplished as provided herein; (d) to consider Claims or the allowance, classification, priority, compromise, estimation, or payment of any Claim; (e) to enter, implement, or enforce such orders as may be appropriate in the event the Confirmation Order is for any reason stayed, reversed, revoked, modified, or vacated; (f) to issue injunctions, enter and implement other orders, and take such other actions as may be necessary or appropriate to restrain interference by any person with the consummation, implementation, or enforcement of the Plan, the Confirmation Order, or any other order of the Bankruptcy Court; (g) to hear and determine any application to modify the Plan in accordance with section 1127 of the Bankruptcy Code, to remedy any defect or omission or reconcile any inconsistency in the Plan, the Disclosure Statement, or any order of the Bankruptcy Court, including the Confirmation Order, in such a manner as may be necessary to carry out the purposes and effects thereof; (h) to hear and determine all applications under sections 328, 330, 331, and 503(b) of the Bankruptcy Code for awards of compensation for services rendered and reimbursement of expenses incurred by Professionals prior to the Confirmation Date; (i) to hear and determine disputes arising in connection with the interpretation, implementation, or enforcement of the Plan, the Confirmation Order, any transactions or payments contemplated hereby, or any agreement, instrument, or other document governing or relating to any of the foregoing; (i) to hear and determine disputes arising in connection with the interpretation, implementation, or enforcement of the Agreement, and to hear and determine all matters involving or relating to the CEP Liquidating Trust or the Liquidating Trustee; (k) to take any action and issue such orders as may be necessary to construe, enforce, implement, execute, and consummate the Plan or to maintain the integrity of the Plan following consummation; (l) to recover all assets of the Debtors' property of the Estates, wherever located, which jurisdiction shall not be limited as a result of the transfer of such assets and property to the Liquidating Trust pursuant to the Plan; (m) to determine such other matters and for such other purposes as may be provided in the Confirmation Order; (n) to hear and determine matters concerning state, local, and federal taxes in accordance with sections 346, 505, and 1146 of the Bankruptcy Code (including, without limitation, matters with respect to any taxes payable by a trust or reserve established in furtherance of the Plan); (o) to hear and determine any other matters related hereto and not inconsistent with the Bankruptcy Code and title 28 of the United States Code; and (p) to enter a final decree closing the Cases.

#### R. Conditions to Effectiveness of the Plan

#### 1. Conditions

The following are conditions precedent to the Effective Date of the Plan:

- (a) No stay of the Confirmation Order shall then be in effect;
- (b) The Debtors shall have sufficient Cash to pay the sum of (i) Allowed Wachovia Secured Claim, Allowed Subordinated Participating Customers Secured Claims, Allowed Administrative Expense Claims, Allowed Priority Tax Claims, Allowed Other Secured Claims and

Allowed Priority Non-Tax Claims, and the Professional fees that have accrued but not been paid (to the extent that any of the foregoing are not otherwise payable under the Final DIP Order by "Cash Infusions", as the term is defined in the Final DIP Order); and/or (ii) an amount that would be required to distribute to the holders of Disputed Subordinated Participating Customers Secured Claims, Disputed Administrative Expense Claims, Disputed Priority Tax Claims, Disputed Other Secured Claims, and Disputed Priority Non-Tax Claims if all such Claims are subsequently Allowed, as set forth more fully in Article VII hereof; and

- (c) Substantial Consummation will or is likely to occur on or before August 31, 2007.
- (d) Superior Fabrication Company, LLC will assume the Pension Plans as described in more detail at Section V.F. of the Disclosure Statement.

#### S. Reserved

#### T. Other Miscellaneous Provisions

#### 1. Modification of the Plan

The Plan may be amended, modified, or supplemented by the Debtors and the Committee in the manner provided for by section 1127 of the Bankruptcy Code or as otherwise permitted by law without additional disclosure pursuant to section 1125 of the Bankruptcy Code, except as the Bankruptcy Court may otherwise direct. In addition, after the Confirmation Date, so long as such action does not materially adversely affect the treatment of holders of Claims or Equity Interests under the Plan, the Debtors and Committee may institute proceedings in the Bankruptcy Court to remedy any defect or omission or reconcile any inconsistencies in the Plan or the Confirmation Order, with respect to such matters as may be necessary to carry out the purposes and effects of the Plan. Prior to the Effective Date, the Debtors and Committee may make appropriate technical adjustments and modifications to the Plan without further order or approval of the Bankruptcy Court, *provided* that such technical adjustments and modifications do not adversely affect in a material way the treatment of holders of Claims or Equity Interests.

# 2. Revocation and Withdrawal of Plan

The Debtors and Committee reserve the right to revoke or withdraw the Plan at any time prior to the Confirmation Date. If the Debtors and Committee take such action, the Plan shall be deemed null and void. In such event, nothing contained herein shall be deemed to constitute a waiver or release of any Claim by or against the Debtors or any other person or to prejudice in any manner the rights of the Debtors or any other person in any further proceedings involving the Debtors.

#### 3. Cram-Down Reservation

If any impaired class votes to accept the Plan by the requisite statutory majorities provided in sections 1126(c) and 1126(d) of the Bankruptcy Code, as applicable, and/or if any impaired class is deemed to have rejected the Plan, the Debtor reserves the right to undertake to have the Bankruptcy Court confirm the Plan under section 1129(c) of the Bankruptcy Code and/or amend the Plan in accordance with Section 13.10 hereof to the extent necessary to obtain entry of a confirmation order.

#### VII. PROJECTED DISTRIBUTION TO CREDITORS

Attached as Exhibit C hereto is a distribution analysis. The distribution analysis was prepared by Huron Consulting, the financial advisor to the Debtors and agreed upon by Grant Thornton, the financial advisor to the Committee. It is important to note that distributions to unsecured creditors are substantially impacted by claims asserted against the Debtors. The claims bar date has passed, however, a complete claims analysis has not been finalized, and as such, the attached analysis is entirely premised on assumptions made with respect to claims in the Debtors cases (using the best information available). Actual distributions could be greater than projected in the event that Allowed Claims are less than estimated amounts and actual distributions could be less than projected in the event that Allowed Claims exceed estimated amounts. Likewise, the analysis attributes no value to Causes of Action due to their uncertain nature (solely for purposes of conservative estimation and which shall not be deemed an admission or waiver by the Debtors or Committee with respect to the value or viability of any Causes of Action).

SUBSTANTIAL EFFORT HAS BEEN MADE TO ENSURE THE ACCURACY OF THE ESTIMATED INFORMATION SUMMARIZED IN THE TABLE IN SECTION II HEREOF. ANTICIPATED ALLOWED AMOUNTS OF ALLOWED CLAIMS AND ALLOWED INTERESTS AND THE PROJECTED DISTRIBUTIONS SUMMARIZED IN THE TABLE IN SECTION II HEREOF ARE SUBJECT, HOWEVER, TO THE UNCERTAINTIES OF LITIGATION THAT MAY OCCUR WITH RESPECT TO CERTAIN CLAIMS AND OTHER FACTORS THAT MAY OR MAY NOT BE RESOLVED IN THE DEBTORS' FAVOR. THEREFORE, NO ASSURANCES CAN BE GIVEN THAT THE ESTIMATED AMOUNTS OF ALLOWED CLAIMS AND ALLOWED INTERESTS AND THE PROJECTED DISTRIBUTION WILL BE ACHIEVED.

#### VIII. UNITED STATES FEDERAL INCOME TAX CONSIDERATIONS

#### A. Introduction

The following discussion summarizes certain material United States federal income tax ("Federal Income Tax") consequences of the Plan to certain holders of Allowed Claims (the "Creditors") and the Debtors. This discussion does not address the Federal Income Tax consequences to: (i) Creditors whose claims are entitled to payment in full in cash, or are otherwise unimpaired under the Plan; and (ii) holders of equity interests or claims that are extinguished without a distribution. This discussion is based upon existing provisions of the Tax Code, Treasury regulations promulgated thereunder, judicial authorities and current administrative rulings and practices now in effect. No assurance can be given that future legislation, regulations, administrative pronouncements and/or judicial decisions will not change applicable law and affect the analysis described herein. Any such change could be applied retroactively in a manner that would adversely affect the creditors and the Debtors.

The Federal Income Tax consequences of certain aspects of the Plan are uncertain due to the lack of applicable legal authority and may be subject to administrative or judicial interpretations that differ from the discussion below. Counsel for the Debtors has not sought and will not seek any rulings from the Internal Revenue Service ("IRS") with respect to the Federal Income Tax consequences discussed below. Although the discussion below represents the best judgment as to the matters discussed herein, it does not in any way bind the IRS or the courts or in any way constitute an assurance that the Federal Income Tax consequences discussed herein will be accepted by the IRS or the courts.

The following discussion does not address state, local or foreign tax considerations that may be applicable to the Debtors or creditors and the discussion does not address the tax consequences of the Plan to certain types of creditors (including foreign persons, financial institutions, life insurance companies, tax-exempt organizations and taxpayers who may be subject to the alternative minimum tax) who may be subject to special rules not addressed herein.

THE FOLLOWING SUMMARY OF CERTAIN MATERIAL FEDERAL INCOME TAX CONSEQUENCES IS FOR GENERAL INFORMATIONAL PURPOSES ONLY AND IS NOT A SUBSTITUTE FOR CAREFUL TAX PLANNING AND ADVICE BASED UPON THE INDIVIDUAL CIRCUMSTANCES PERTAINING TO A HOLDER OF AN ALLOWED CLAIM OR EQUITY INTEREST. THE DEBTORS ARE NOT MAKING ANY REPRESENTATIONS REGARDING THE PARTICULAR TAX CONSEQUENCES OF CONFIRMATION AND CONSUMMATION OF THE PLAN WITH RESPECT TO THE DEBTOR OR THE HOLDERS OF ANY CLAIMS OR EQUITY INTERESTS, NOR ARE THE DEBTORS RENDERING ANY FORM OF LEGAL OPINION OR TAX ADVICE ON SUCH TAX CONSEQUENCES. ALL HOLDERS OF CLAIMS OR EQUITY INTERESTS ARE ADVISED TO CONSULT THEIR OWN TAX ADVISORS WITH RESPECT TO THE FEDERAL, STATE, LOCAL, FOREIGN AND OTHER TAX CONSEQUENCES OF THE IMPLEMENTATION OF THE PLAN IN GENERAL AND IN PARTICULAR, THE TIMING, CHARACTER AND AMOUNTS OF INCOME, GAIN, LOSS, DEDUCTION, CREDIT OR CREDIT RECAPTURE TO BE RECOGNIZED, AND ANY PROCEDURAL REQUIREMENTS WITH WHICH THE HOLDER MUST COMPLY.

#### B. Certain Material United States Federal Income Tax Consequences to the Debtors

The Debtors do not believe that there will be any material current cash United States federal income tax consequences to the Debtors as a result of the implementation of the Plan.

# C. Certain Material United States Federal Income Tax Consequences to Holders of Claims

#### 1. **General**

The tax treatment of holders of Claims and the character and amount of income, gain or loss recognized as a consequence of the Plan and the distributions provided for by the Plan will depend upon, among other things, (i) whether the Claim (or any portion thereof) constitutes a Claim for principal or interest; (ii) the type and classification of consideration received by the holder in exchange for the Claim; (iii) whether the holder is a resident of the United States for tax purposes (or falls into any special class of taxpayers, such as those that are excluded from this discussion as noted above); (iv) the manner in which a holder acquired a Claim; (v) the length of time the Claim has been held; (vi) whether the claim was acquired at a discount; (vii) whether the holder has taken a bad debt deduction with respect to the Claim (or any portion thereof) in the current or prior years; (viii) whether the holder has previously included accrued but unpaid interest with respect to the Claim; (ix) the method of tax accounting of the holder; (x) whether the claim is an installment obligation for United States federal income tax purposes; and (xi) whether the "market discount" rules are applicable to the holder. Therefore, holders of Claims should consult their tax advisors for information that may be relevant to their particular situation and circumstances and the particular tax consequences to them of the transactions contemplated by the Plan.

A holder of a Claim should generally recognize a gain (or loss) to the extent that the amount realized under the Plan in respect of the Claim exceeds (or is exceeded by) the holder's tax basis in the Claim. The holder's amount realized for this purpose will generally equal the amount of Cash the holder receives under the Plan in respect of its Claim. The timing and amount of income, gain or loss recognized as a consequence provided for by the Plan will depend on, among other things, whether the holder of a Claim receives multiple distributions pursuant to the Plan and whether the Debtors' obligation to make such payments is treated as a new debt for United States federal income tax purposes. Gain or loss may not currently be recognized if the property received does not have an ascertainable fair market value.

## 2. Market Discount

The market discount provisions of the Internal Revenue Code of 1986 (the "Tax Code") may apply to holders of certain Claims. In general, a debt obligation other than a debt obligation with a fixed maturity of one year or less that is acquired by a holder in the secondary market (or, in certain circumstances, upon original issuance) is a "market discount bond" as to that holder if its stated redemption price at maturity (or, in the case of a debt obligation having original issue discount, its revised issue price) exceeds the tax basis of the debt obligation will not be a "market discount bond" if such excess is less than a statutory de minimis amount. Gain recognized by the holder of a Claim with respect to a "market discount bond" will generally be treated as ordinary interest income to the extent of the market discount accrued on such bond during the holder's grace period of ownership, unless the holder elected to include accrued market discount in taxable income currently. A holder of a market discount bond that is required under the market discount rules of the Tax Code to defer deduction of all or a portion of the interest on indebtedness incurred or maintained to acquire or carry the bond may be allowed to deduct such interest, in whole or in part, on disposition of such bond.

## 3. <u>Information Reporting and Backup Withholding</u>

Certain payments, including the payments with respect to Claims pursuant to the Plan, are generally subject to information reporting by the payor (the Debtors) to the IRS. Moreover, such reportable payments are subject to backup withholding under certain circumstances. Under the Tax Code's backup withholding rules, a holder of a Claim may be subject to backup withholding with respect to distributions or payments made pursuant to the Plan, unless the holder: (1) comes within certain exempt categories (which generally include corporations) and, when required, demonstrates this fact, or (2) provides a correct United States taxpayer identification number and certifies under penalty of perjury that the taxpayer identification number is correct and that the taxpayer is not subject to backup withholding because of a failure to report all dividend and interest income.

Holders of Claims that are non-United States Persons and that receive payments or distributions under the Plan will not be subject to backup withholding, provided that such holders furnish certification of their status as non-United States Persons (and furnish any other required certification), or are otherwise exempt from backup withholding. Generally, such certification is provided on IRS Form W-8BEN.

Backup withholding is not an additional tax. Amounts withheld under the backup withholding rules may be credited against a holder's United States federal income tax liability and a holder may obtain a refund of any excess amounts withheld under the backup withholding rules by filing an appropriate claim for refund with the IRS (generally, a United States federal income tax return).

PERSONS CONCERNED WITH THE TAX CONSEQUENCES OF THE PLAN SHOULD CONSULT THEIR OWN ACCOUNTANTS, ATTORNEYS AND/OR ADVISORS. THE DEBTORS MAKE THE ABOVE-NOTED DISCLOSURE OF POSSIBLE TAX CONSEQUENCES FOR THE SOLE PURPOSE OF ALERTING READERS TO TAX ISSUES THEY MAY WISH TO CONSIDER.

#### IX. ALTERNATIVES TO THE PLAN

The Plan Proponents submit that the Plan is the best means of providing maximum recoveries to creditors. Alternatives to the Plan that have been considered and evaluated by the Plan Proponents during the course of the Cases and include (i) liquidation of the Debtors' assets under chapter 7 of the Bankruptcy Code, and (ii) an alternative chapter 11 plan. The Plan Proponents thorough consideration of these alternatives to the Plan have caused them each respectively to conclude that the Plan, in comparison, provides a greater recovery to creditors on a more expeditious timetable and in a manner which minimizes inherent risks in any other course of action available to the Debtors.

#### A. Other Plans of Reorganization

If the Plan is not confirmed, the Plan Proponents or any other party in interest (if the Debtors' exclusive period in which to file a chapter 11 plan has expired) could attempt to formulate an alternative chapter 11 plan that might provide for the liquidation of the Debtors' assets other than as provided in the Plan. However, since substantially all of the Debtors' assets (including assets located at the facilities of or relating to CEP Mexico) have already been sold to various parties, the Plan Proponents submit that any alternative chapter 11 plan will not provide greater or more rapid return to impaired creditors than the Plan. Any attempt to formulate an alternative chapter 11 plan would necessarily delay creditors' receipt of distributions and is highly unlikely to formulate a methodology to create greater value (return) to impaired creditors than the Plan. Accordingly, the Plan Proponents submit that the Plan will enable all creditors entitled to distributions to realize the greatest possible recovery on their respective Claims with the least possible delay.

## B. Liquidation Under Chapter 7 of the Bankruptcy Code

If the Plan or any other chapter 11 plan for the Debtors cannot be confirmed under section 1129(a) of the Bankruptcy Code, the Cases may be converted to cases under chapter 7 of the Bankruptcy Code, in which event a trustee would be appointed (or subsequently elected) to liquidate any remaining assets of the Debtors for distribution to creditors pursuant to chapter 7 of the Bankruptcy Code. If a trustee is appointed and the remaining assets of the Debtors are liquidated under chapter 7 of the Bankruptcy Code, all creditors holding Allowed Administrative Expense Claims, Allowed Priority Tax Claims and Allowed Priority Non-Tax Claims may receive distributions of a lesser value on account of their Allowed Claims and likely would have to wait a longer period of time to receive such distributions than they would under the Plan. A chapter 7 trustee would be required to invest substantial time and resources to investigate (i) the facts underlying the multitude of Claims filed against the Estates; (ii) inter Debtor and inter-company (including (CEP Mexico) claims; and (iii) ownership rights and interests as between the Debtors and CEP Mexico, thereby substantially increasing both costs and time necessary to fully administer the Estates. Likewise, in addition to fees of professionals retained by a chapter 7 trustee, the chapter 7 trustee would also charge a fee tied to the value of all assets administered by the chapter 7 trustee in accordance with section 326(a) of the Bankruptcy Code, which will not be charged by the Liquidating Trustee.

## X. CONFIRMATION REQUIREMENTS

## A. The Confirmation Hearing

The Bankruptcy Code requires the Bankruptcy Court, after notice, to hold a confirmation hearing before a plan of reorganization may be confirmed. The Confirmation Hearing to confirm the Plan has been scheduled for the date set forth in the attached notice of confirmation hearing before the Honorable Marilyn Shea-Stonum, United States Bankruptcy Judge in the United States Bankruptcy Court, Northern District of Ohio (Eastern Division), Bankruptcy Courtroom Number 260, U.S. Courthouse and Federal Building, 2 South Main Street, Akron, Ohio 44308. The Confirmation Hearing may be adjourned from time to time by the Bankruptcy Court without further notice except for an announcement of the adjourned date made at the Confirmation Hearing. Any objection to confirmation must be made in writing and specify in detail the name and address of the objector, all grounds for the objection and the amount of the claim or number and type of shares of equity security interests held by the objector. Any such objection must be filed with the Bankruptcy Court and served so that it is received by the Bankruptcy Court and certain other parties when and as set forth in the attached notice of confirmation hearing.

Objections to confirmation of the Plan are governed by Bankruptcy Rule 9014. At the hearing on the confirmation of the Plan, the Bankruptcy Court will confirm the Plan only if the requirements of the Bankruptcy Code, particularly those set forth in section 1129 of the Bankruptcy Code, have been satisfied.

## B. Acceptances Necessary to Confirm the Plan

At the Confirmation Hearing, the Bankruptcy Court must determine, among other things, whether the Plan has been accepted by the requisite amount and number of Allowed Claims and Allowed Interests in each impaired class. Under the Bankruptcy Code, a class of creditors or equity security holders is impaired if its legal, equitable or contractual rights are altered by a proposed plan of reorganization. If a class is not impaired, each creditor or equity security holder in such unimpaired class is conclusively presumed to have accepted the plan pursuant to section 1126(f) of the Bankruptcy Code. Classes 1, 2 and 3 are not impaired under the Plan and are, therefore, not entitled to vote on the Plan. Class 4 is impaired under the Plan and holders of Allowed Claims or Allowed Interests in such classes are entitled to vote for or against the Plan by completing and returning ballots mailed to them with this Disclosure Statement in the manner set forth in the ballots. Class 5 is impaired under the Plan, but is deemed to reject the Plan, and is not entitled to vote.

An impaired class of creditors and each holder of a claim in such class will be deemed to have accepted the Plan if the holders of at least two-thirds in amount and more than one-half of those in number of the Allowed Claims in such impaired class for which complete and timely ballots have been received have voted for acceptance of the Plan. An impaired class of equity securities and each holder of an interest in such class will be deemed to have accepted a plan if the plan has been accepted by at least two-thirds in amount of the interests in such class who actually vote on the Plan.

Because the equity interests held by the members of Class 5 is entirely eliminated under the Plan, Class 5 is deemed to have rejected the Plan, and the Debtors cannot satisfy the requirements of section 1129(a)(8) of the Bankruptcy Code. Accordingly, the Debtors intend to seek confirmation of the Plan pursuant to section 1129(b) of the Bankruptcy Code. Under section 1129(b), the Bankruptcy Court must determine, among other things, that the Plan does not discriminate unfairly and that it is fair and equitable

with respect to each class of impaired Allowed Claims and Allowed Interests that have not voted to accept the Plan.

## C. Best Interests of Creditors

The Bankruptcy Code requires that each holder of an impaired claim or equity interest either (i) accept the Plan, or (ii) receive or retain under the Plan property of a value, as of the Effective Date, that is not less than the value such holder would receive if the Debtors were liquidated under chapter 7 of the Bankruptcy Code.

The first step in determining whether this test has been satisfied is to determine the dollar amount that would be generated from the liquidation of the Debtors' assets and properties in the context of a chapter 7 liquidation case. The gross amount of cash that would be available for satisfaction of claims and equity interests would be the sum consisting of the proceeds resulting from the disposition of the unencumbered assets and properties of the Debtors, augmented by the unencumbered cash hold by the Debtors at the time of the commencement of the liquidation cases.

The next step is to reduce that gross amount by the costs and expenses of liquidation and by such additional administrative and priority claims that might result from the use of chapter 7 for the purposes of liquidation. Any remaining net cash would be allocated to creditors and shareholders in strict priority in accordance with section 726 of the Bankruptcy Code. Finally, the present value of such allocations (taking into account the time necessary to accomplish the liquidation) are compared to the value of the property that is proposed to be distributed under the Plan on the Effective Date.

The Debtors' costs of liquidation under chapter 7 would include the fees payable to a trustee in bankruptcy, as well as those fees that might be payable to attorneys and other professionals that such a trustee might engage. Other liquidation costs include the expenses incurred during the Cases and subsequently allowed in the chapter 7 cases, such as compensation for attorneys, financial advisors, appraisers, accountants and other professionals for the Debtors, and the Committee, and costs and expenses of members of the Committee, as well as other compensation claims.

The foregoing types of claims, costs, expenses, fees and such other claims that may arise in a liquidation case would be paid in full from the liquidation proceeds before the balance of those proceeds would be made available to pay prepetition priority and unsecured Claims.

The Plan Proponents submit that each impaired Class will receive under the Plan a recovery at least equal in value to the recovery such Class would receive pursuant to a liquidation of the Debtors under chapter 7 of the Bankruptcy Code, and almost certainly more due to the statutory fees to which a chapter 7 trustee is entitled for administering assets.

After consideration of the effects that a chapter 7 liquidation would have on the ultimate proceeds available for distribution to creditors in the Cases, including the chapter 7 trustee's investment of substantial time and resources to investigate the facts underlying the multitude of Claims filed against the Estates, and the chapter 7 trustee's investigation of inter-Debtor and inter-company (including (CEP Mexico) claims and ownership rights and interests as between the Debtors and CEP Mexico, the Plan Proponents have determined that confirmation of the Plan will provide each holder of an Allowed Claim with a recovery that is not less than such holder would receive pursuant to liquidation of the Debtors under chapter 7.

The Plan Proponents also submit that the value of any distributions to each Class of Allowed Claims in a chapter 7 case, including all Other Secured Claims, would be less than the value of distributions under the Plan because such distributions in a chapter 7 case would not occur for a substantial period of time. In the event litigation was necessary to resolve claims asserted in a chapter 7 case, the delay could be prolonged and administrative expenses increased.

## D. Feasibility

Section 1129(a)(11) of the Bankruptcy Code provides that a chapter 11 plan may be confirmed only if the Court finds that such plan is feasible. A feasible plan is one which will not lead to a need for further reorganization or liquidation of the debtor. Since the Plan provides for the complete liquidation of the Debtors, the Bankruptcy Court will find that the Plan is feasible if it determines that the Debtors will be able to satisfy the conditions precedent to the Effective Date and otherwise have sufficient funds to meet its post-Confirmation Date obligations to pay for the costs of administering and fully consummating the Plan and closing the Cases. The Plan Proponents submit that the Plan satisfies the financial feasibility requirement imposed by the Bankruptcy Code.

## E. Confirmation of the Plan

In the event the Bankruptcy Court determines that all of the requirements for the confirmation of the Plan are satisfied, the Bankruptcy Court will issue the Confirmation Order confirming the Plan pursuant to section 1129 of the Bankruptcy Code.

#### XI. CERTAIN RISK FACTORS TO BE CONSIDERED

HOLDERS OF IMPAIRED CLAIMS AGAINST OR INTERESTS IN THE DEBTORS ARE ENCOURAGED TO READ AND CONSIDER CAREFULLY THE FACTORS SET FORTH BELOW, AS WELL AS THE OTHER INFORMATION SET FORTH IN THIS DISCLOSURE STATEMENT (AND THE DOCUMENTS DELIVERED TOGETHER HEREWITH AND/OR INCORPORATED BY REFERENCE), PRIOR TO VOTING TO ACCEPT OR REJECT THE PLAN. THOSE RISK FACTORS SHOULD NOT, HOWEVER, BE REGARDED AS CONSTITUTING THE ONLY RISKS INVOLVED IN CONNECTION WITH THE PLAN AND ITS IMPLEMENTATION.

## A. Parties-In-Interest May Object to the Plan Proponents' Classification of Claims

Section 1122 of the Bankruptcy Code provides that a plan of reorganization may place a claim or an interest in a particular class only if such claim or interest is substantially similar to the other claims or interests of such class. The Plan Proponents submit that the classification of claims and interests under the Plan complies with the requirements set forth in the Bankruptcy Code. However, the Plan Proponents cannot give assurances that the Bankruptcy Court will reach the same conclusion.

## B. The Plan Proponents May Not Be Able to Secure Confirmation of the Plan

The Plan Proponents cannot assure you that the requisite acceptances to confirm the Plan will be received. Even if the requisite acceptances are received, the Plan Proponents cannot assure you that the Bankruptcy Court will confirm the Plan. A non-accepting creditor or equity security holder of the Debtors might challenge the balloting procedures and results as not being in compliance with the Bankruptcy Code or Bankruptcy Rules. Even if the Bankruptcy Court determined that the Disclosure Statement and the balloting procedures and results were appropriate, the Bankruptcy Court could still

decline to confirm the Plan if it found that any of the statutory requirements for confirmation had not been met. Section 1129 of the Bankruptcy Code sets forth the requirements for confirmation and requires, among other things, a finding by the Bankruptcy Court that the confirmation of the Plan is not likely to be followed by a liquidation or a need for further financial reorganization and that the value of distributions to non-accepting holders of claims and interests within a particular class under the Plan will not be less than the value of distributions such holders would receive if the Debtors were liquidated under chapter 7 of the Bankruptcy Code. While the Plan Proponents cannot give assurances that the Bankruptcy Court will conclude that these requirements have been met, the Plan Proponents submit that the Plan will not be followed by a need for further financial reorganization and that non-accepting holders within each class under the Plan will receive distributions at least as great as would be received following a liquidation under chapter 7 of the Bankruptcy Code when taking into consideration all administrative claims and the costs and uncertainty associated with any such chapter 7 case.

The confirmation and consummation of the Plan are also subject to certain conditions.

## C. The Plan Proponents and/or the Trustee May Object to the Amount or Classification of Your Claim

The Plan Proponents reserve the right to object to the amount or classification of any claim or interest. The estimates set forth in this Disclosure Statement cannot be relied on by any creditor whose claim or interest is subject to an objection. Any such claim or interest holder may not receive its specified share of the estimated distributions described in this Disclosure Statement. Such an objection to claim may be prosecuted by the Liquidating Trustee.

#### **D.** Insurance Claims

There is no guarantee that holders of Insurance Claims that are Allowed Claims will receive any payments on account of their Insurance Claims because such Claims may not be covered under the applicable Insurance Policies.

In addition, holders of Insurance Claims that are Allowed Claims, and that may otherwise be covered under the Insurance Policies, may not have access to or receive any proceeds under the Insurance Policies because the Insurers may deny coverage on otherwise valid claims. The Insurers have issued Insurance Policies to the Debtors that may provide coverage for Insurance Claims. The Insurers, however, may assert that the Plan may alter or violate the Insurers' rights under the Insurance Policies and applicable law and may potentially seek to avoid any coverage otherwise available under the Insurance Policies. In particular, the Insurers may assert that the Plan violates the Insurers' rights to (i) control the defense, investigation, and settlement of Insurance Claims; (ii) require the Debtors' compliance with the terms and conditions of the Insurance Policies; (iii) assert certain subrogation rights available to the Insurers under the Policies; and (iv) assert claims for setoff, contribution and recoupment. The Insurers may also assert that the Plan seeks to provide certain injunctive relief that may alter the Debtors' obligations under the Policies in such a way that would also void any available insurance coverage.

# E. Assumptions Forming the Basis for Calculating Estimated Returns to Creditors Could be Incorrect

The distribution analysis, at Exhibit C hereto, includes certain assumptions regarding, among others, (a) the costs of administering the CEP Liquidating Trust; (b) the amount of Allowed Claims; (c) the priority of certain Claims; and (d) the recoverable value of accounts receivable. In the event that any

one or more of the assumptions made relating to these items proves to be incorrect, the actual return to creditors, and in particular Allowed General Unsecured Claims, could be less than projected. The assumptions, however, for purposes of being conservative, attribute no value to Causes of Action which may have significant value, and as such, incorrect assumptions in respect of any of the foregoing may be offset by recoveries in respect of Causes of Action.

#### XII. WHERE YOU CAN OBTAIN MORE INFORMATION

Pursuant to the requirements of the Office of the U.S. Trustee, the Debtors are required to and have filed monthly operating reports for the postpetition period with the Bankruptcy Court. These monthly operating reports may be obtained at prescribed per page copy rates by writing to the Clerk of the United States Bankruptcy Court for the Northern District of Ohio (Eastern Division), 455 United States Courthouse, 2 South Main Street, Akron, Ohio, 44308; (telephone 330.252.6100) or BMC Group, Inc., CEP Holdings, LLC, et al., c/o BMC Group, P.O. Box 903, El Segundo, CA, 90245-0903 (telephone 888.909.0100), or on-line at the Bankruptcy Court's website: <a href="http://www.ohnb.uscourts.gov">http://www.ohnb.uscourts.gov</a>, or BMC's website: <a href="http://www.bmcgroup.com/cep">http://www.ohnb.uscourts.gov</a>, or BMC's website: <a href="http://www.bmcgroup.com/cep">http://www.bmcgroup.com/cep</a>.

## XIII. CONCLUSION AND RECOMMENDATION

The Plan Proponents believe that confirmation and implementation of the Plan is preferable to any of the alternatives described above because it will provide the greatest recoveries to holders of Claims. The Plan Proponents urge holders of Claims entitled to vote on the Plan to vote to accept the Plan

Dated: May 25, 2007

## CEP Holdings, LLC, Creative Engineered Polymer Products, LLC and Thermoplastics Acquisition, LLC

By: <u>/s/ Joseph Mallak</u>

Name: Joseph Mallak Title: President & CEO

The Official Committee of Unsecured Creditors

By: /s/ Bruce D. Tobiansky

Name: Bruce D. Tobiansky

E.I. du Pont de Nemours and Company

Committee Co-Chair

By: /s/ Doreen Bailey-Wilson

Name: Doreen Bailey-Wilson

Rhodia, Inc.

Committee Co-Chair

[THIS PAGE INTENTIONALLY LEFT BLANK]

## **EXHIBIT A**

First Amended Joint Plan of Liquidation Under Chapter 11 of the Bankruptcy Code Proposed by the Debtors and the Official Committee of Unsecured Creditors Dated May 25, 2007 [THIS PAGE INTENTIONALLY LEFT BLANK]

## UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF OHIO EASTERN DIVISION

IN RE:	)	Case No. 06-51848
	)	(Jointly Administered)
CEP HOLDINGS, LLC, et al.,1	)	Chapter 11
	)	_
Debtors.	)	Honorable Marilyn Shea-Stonum
	)	United States Bankruptcy Judge

FIRST AMENDED JOINT PLAN OF LIQUIDATION UNDER CHAPTER 11 OF THE BANKRUPTCY CODE PROPOSED BY THE DEBTORS AND THE OFFICIAL COMMITTEE OF UNSECURED CREDITORS DATED MAY 25, 2007

#### **BAKER & HOSTETLER LLP**

Joseph F. Hutchinson, Jr., Esquire Thomas M. Wearsch, Esquire Eric R. Goodman, Esquire 3200 National City Center 1900 East 9<sup>th</sup> Street Cleveland, OH 44114

Telephone: (216) 621-0200 Facsimile: (216) 696-0740

Counsel to the Debtors

-and-

MCGUIREWOODS LLP

Mark E. Freedlander, Esquire Sally E. Edison, Esquire William C. Price, Esquire Dominion Tower, 23rd Floor 625 Liberty Avenue Pittsburgh, PA 15222

Telephone: (412) 667-6000 Facsimile: (412) 667-6050

Counsel to the Official Committee of Unsecured

Creditors of the Debtors

<sup>1</sup> The Debtors include: CEP Holdings, LLC, Creative Engineered Polymer Products, LLC and Thermoplastics Acquisition, LLC.

## TABLE OF CONTENTS

ARTICLE I. D	EFINITIONS AND INTERPRETATION	1
1.1	Administrative Expense Claim.	1
1.2	Agreed AR	1
1.3	Agreement	1
1.4	Allowed	2
1.5	Authorized Agent.	2
1.6	Avoidance Actions.	2
1.7	Bankruptcy Code.	2
1.8	Bankruptcy Court	2
1.9	Bankruptcy Rules.	2
1.10	BBK.	2
1.11	BMC.	2
1.12	Beneficiaries.	2
1.13	Business Day	2
1.14	Canton Pension Plan.	2
1.15	Cases.	2
1.16	Cash.	2
1.17	Causes of Action.	3
1.18	CEP.	3
1.19	CEP Liquidating Trust.	3
1.20	CEP Loan Agreement.	3
1.21	CEP Mexico.	3
1.22	Claim	3
1.23	Claims Register.	3
1.24	Class	3
1.25	Collateral	3
1.26	Committee	3
1.27	Confirmation Date.	3
1.28	Confirmation Hearing	3
1.29	Confirmation Order	3
1.30	Consolidated Holdings	3

1.31	Crestline Pension Plan.	3
1.32	Debtor	4
1.33	Debtors	4
1.34	Delphi	4
1.35	Disbursing Agent.	4
1.36	Disclosure Statement.	4
1.37	Disputed	4
1.38	Disputed Claim Amount.	4
1.39	Disputed Claim Reserve.	4
1.40	Distribution Record Date.	4
1.41	Delphi.	4
1.42	Effective Date	4
1.43	Equity Interest.	4
1.44	Estates.	5
1.45	Final Order.	5
1.46	Final DIP Order.	5
1.47	General Unsecured Claim.	5
1.48	GM	5
1.49	Holdings	5
1.50	Impaired	5
1.51	Insurance Claim.	5
1.52	Insurance Policy.	5
1.53	Insurer.	5
1.54	Law	5
1.55	Lien.	5
1.56	Liquidating Trustee	5
1.57	Liquidation Expense Reserve Account.	6
1.58	Other Secured Claim.	6
1.59	Participating Customers	6
1.60	PBGC	6
1.61	Pension Plan	6
1.62	Person.	6
1.63	Petition Date.	6
1.64	Plan.	6
1.65	Plan Proponents.	6

1.66	Priority Non-Tax Claim.	6
1.67	Priority Tax Claim.	6
1.68	Pro Rata Share.	6
1.69	Professionals.	6
1.70	Reserve Income	6
1.71	Sale Transaction	6
1.72	Schedules.	7
1.73	Secured Claim.	7
1.74	Subordinated Participation Agreement.	7
1.75	Subordinated Participating Customers' Secured Claims.	7
1.76	Subordinated Participating Customers	7
1.77	Tax Code	7
1.78	Thermoplastics.	7
1.79	Thermoplastics Loan Agreement.	7
1.80	Trust Assets	7
1.81	Voting Class	7
1.82	Voting Procedures Order.	7
1.83	Visteon.	7
1.84	Wachovia	7
1.85	Wachovia Loan Agreements.	8
ARTICLE II.	ADMINISTRATIVE EXPENSE CLAIMS AND PRIORITY TAX CLAIMS	8
2.1	Administrative Expense Claims.	8
2.2	Compensation and Reimbursement Claims.	8
2.3	Fees Under 28 U.S.C. §1930.	8
2.4	Wachovia Secured Claim.	8
2.5	Priority Tax Claims	9
ARTICLE III	. CLASSIFICATION OF CLAIMS AND EQUITY INTERESTS	9
ARTICLE IV	T. TREATMENT OF CLAIMS AND EQUITY INTERESTS	9
4.1	Class 1 – Subordinated Secured Participation Claims	9
4.2	Class 2 – Other Secured Claims.	10
4.3	Class 3 - Priority Non-Tax Claims	10
4.4	Class 4 - General Unsecured Claims	10
4.5	Class 5 - Equity Interests	10
ARTICLE V.	SUBSTANTIVE CONSOLIDATION OF HOLDINGS, CEP, THERMOPLASTICS AND CEP MEXICO	10

5.1	Substantive Consolidation	10
ARTICLE V	I. PROVISIONS GOVERNING DISTRIBUTIONS	11
6.1	Distribution Record Date.	11
6.2	Method of Distributions Under the Plan.	11
6.3	Withholding and Reporting Requirements	12
6.4	Time Bar to Cash Payments.	12
6.5	Minimum Distributions	12
6.6	Setoffs	12
6.7	Transactions on Business Days	12
6.8	Allocation of Plan Distribution Between Principal and Interest.	13
ARTICLE V	III. MEANS FOR IMPLEMENTATION AND EXECUTION OF THE PLAN	13
7.1	The CEP Liquidating Trust	13
7.2	Distributions to Holders as of the Confirmation Date	17
7.3	Closing of Cases by Charitable Gift	17
7.4	Release of Liens	18
7.5	Cancellation of Existing Securities and Agreements	18
7.6	Liquidating Trustee's Post-Confirmation Role	18
7.7	Books and Records	19
7.8	Corporate Action.	19
7.9	Effectuating Documents and Further Transactions.	20
7.10	Securities Exempt	20
7.11	Pension Plans.	20
ARTICLE V	VIII. PROCEDURES FOR DISPUTED CLAIMS	20
8.1	Objections to Claims	20
8.2	No Distribution Pending Allowance	20
8.3	Reserve on Account of Disputed Claims.	20
8.4	Resolution of Disputed Claims	21
8.5	Estimation	21
8.6	Distributions to Holders of Allowed Claims Upon Disallowance of Disputed Claims	21
ARTICLE I	X. EXECUTORY CONTRACTS AND UNEXPIRED LEASES	22
9.1	Executory Contracts and Unexpired Leases	22
9.2	Approval of Rejection of Executory Contracts and Unexpired Leases	22
9.3	Rejection Claims	22
ARTICLE X	C EFFECTIVENESS OF THE PLAN	22

10.1	Condition Precedent to Confirmation of Plan	22
10.2	Conditions Precedent to Effective Date	23
10.3	Satisfaction of Conditions	23
10.4	Effect of Nonoccurrence of Conditions to Effective Date.	23
ARTICLE XI.	EFFECT OF CONFIRMATION	23
11.1	Vesting of Assets.	23
11.2	Release of Assets	23
11.3	Binding Effect	24
11.4	Term of Injunctions or Stays	24
11.5	Causes of Action	24
11.6	Injunction	24
11.7	Injunction Against Interference with Plan	24
ARTICLE XII	RETENTION OF JURISDICTION	24
12.1	Jurisdiction of Bankruptcy Court	24
ARTICLE XII	I. CRAMDOWN RESERVATION	26
13.1	Nonconsensual Confirmation	26
ARTICLE XIV	V. MISCELLANEOUS PROVISIONS	26
14.1	Disposition of Creditors' Committee	26
14.2	Record Date	26
14.3	Substantial Consummation	26
14.4	Exemption from Transfer Taxes	26
14.5	Exculpation	27
14.6	Discharge of Claims and Termination of Equity Interests	27
14.7	Post-Confirmation Date Fees and Expenses.	27
14.8	Payment of Statutory Fees	27
14.9	Modification of Plan.	27
14.10	Revocation or Withdrawal of Plan	28
14.11	Courts of Competent Jurisdiction	28
14.12	Severability	28
14.13	Governing Law	28
14.14	Exhibits	28
14.15	Successors and Assigns	28
14.16	Time	28
14 17	Notices	28

## UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF OHIO EASTERN DIVISION

IN RE:	)	Case No. 06-51848
	)	(Jointly Administered)
CEP HOLDINGS, LLC, et al., <sup>2</sup>	)	Chapter 11
	)	
Debtors.	)	Honorable Marilyn Shea-Stonum
	)	United States Bankruptcy Judge

## FIRST AMENDED JOINT PLAN OF LIQUIDATION UNDER CHAPTER 11 OF THE BANKRUPTCY CODE PROPOSED BY THE DEBTORS AND THE OFFICIAL COMMITTEE OF UNSECURED CREDITORS DATED MAY 25, 2007

CEP Holdings, LLC, Creative Engineered Polymer Products, LLC and Thermoplastics Acquisition, LLC, the above-captioned debtors and debtors-in-possession, propose jointly with the Official Committee of Unsecured Creditors the following chapter 11 plan of liquidation pursuant to section 1121(a) of title 11 of the United States Code:

#### ARTICLE I.

## **DEFINITIONS AND INTERPRETATION**

**<u>DEFINITIONS</u>**. The following terms used herein shall have the respective meanings defined below (such meanings to be equally applicable to both the singular and plural):

- **1.1** Administrative Expense Claim means any right to payment constituting a cost or expense of administration of any of the Cases allowed under sections 503(b), 507(a)(1), and 1114(e) of the Bankruptcy Code, including, without limitation, any actual and necessary costs and expenses of preserving the Estates; any actual and necessary costs and expenses of operating the Debtors' businesses; any indebtedness or obligations incurred or assumed by the Debtors, as debtors-in-possession, during the Cases, including, without limitation, for the acquisition or lease of property or an interest in property; or the rendition of services, any allowances of compensation and reimbursement of expenses to the extent allowed by Final Order under sections 328(a), 330 or 503 of the Bankruptcy Code, and any fees or charges assessed against the Estates under section 1930 of chapter 123 of title 28 of the United States Code.
- **1.2** Agreed AR means with respect to GM, \$56,003 (the "GM Agreed AR"), with respect to Visteon, \$600,000 (the "Visteon Agreed AR"), and with respect to Delphi, a net amount of \$354,944 (the "Delphi Agreed AR").
- **1.3** Agreement means the CEP Liquidating Trust Agreement substantially in the form of Exhibit 1 hereto.

1

<sup>&</sup>lt;sup>2</sup> The Debtors include: CEP Holdings, LLC, Creative Engineered Polymer Products, LLC and Thermoplastics Acquisition, LLC

- 1.4 Allowed means, with reference to any Claim, (i) any Claim against the Debtors that has been listed by the Debtors in the Schedules, as such Schedules may be amended by the Debtors from time to time in accordance with Bankruptcy Rule 1009, as liquidated in amount and not disputed or contingent and for which no contrary proof of claim has been filed, (ii) any timely filed Claim as to which no objection to allowance has been interposed in accordance with section 8.1 hereof or such other applicable period of limitation fixed by the Bankruptcy Code, the Bankruptcy Rules, or the Bankruptcy Court, or as to which any objection has been determined by a Final Order to the extent such objection is determined in favor of the respective holder, or (iii) any Claim expressly allowed by a Final Order or hereunder.
- **1.5** Authorized Agent means Joseph Mallak, as to be approved by the Confirmation Order, and where the term is used with respect to rights, powers or obligations on or after the Effective Date it shall mean the Liquidating Trustee.
- **1.6** Avoidance Actions means any actions commenced, or that may be commenced, before or after the Effective Date pursuant to sections 542, 544, 545, 547, 548, 549, 550, 551 or 553 of the Bankruptcy Code.
- **1.7 Bankruptcy Code** means title 11 of the United States Code, as amended from time to time, as applicable to these Cases.
- **1.8 Bankruptcy Court** means the United States District Court for the Northern District of Ohio (Eastern Division), having jurisdiction over these Cases and, to the extent of any reference made under section 157 of title 28 of the United States Code, the unit of such District Court having jurisdiction over these Cases under section 151 of title 28 of the United States Code.
- **1.9 Bankruptcy Rules** means the Federal Rules of Bankruptcy Procedure as promulgated by the United States Supreme Court under section 2075 of title 28 of the United States Code, as amended from time to time, applicable to these Cases, and any Local Rules of the Bankruptcy Court.
- **1.10 BBK** means BBK, Ltd., the agent under the Final DIP Order to the Participating Customers.
  - **1.11 BMC** means BMC Group, Inc.
- **1.12** <u>Beneficiaries</u> means all Persons entitled to receive distributions from the CEP Liquidating Trust pursuant to the Plan and the Agreement.
- **1.13** Business Day means any day other than a Saturday, a Sunday, or any other day on which banking institutions in New York, New York are required or authorized to close by law or executive order.
- **1.14** <u>Canton Pension Plan</u> means the Pension Plan for the bargaining unit employees of CEP's Canton, Ohio plant.
- **1.15** Cases means the within captioned jointly administered chapter 11 bankruptcy cases filed on the Petition Date.
  - **1.16** Cash means legal tender of the United States of America.

- 1.17 <u>Causes of Action</u> means any and all actions, causes of action, liabilities, obligations, rights, suits, damages, judgments, claims and demands whatsoever, including but not limited to Avoidance Actions, whether known or unknown, existing or hereafter arising, in law, equity, or otherwise, based in whole or in part upon any act or omission or other event occurring prior to the Petition Date or during the course of these Cases, arising out of, relating to or in connection with the pre-bankruptcy and/or post-bankruptcy Debtors.
  - **1.18** CEP means Creative Engineered Polymer Products, LLC.
- 1.19 <u>CEP Liquidating Trust</u> means that certain trust that will come into existence upon the Effective Date into which all of the assets, rights, powers and obligations of the Debtors will vest pursuant to the Plan, which trust shall be formed pursuant to and governed by the Agreement.
- **1.20** <u>CEP Loan Agreement</u> means that certain Loan and Security Agreement dated as of August 16, 2005 by and between CEP and Wachovia as amended, supplemented and modified from time to time, and including all schedules, attachments, exhibits and addendums thereto.
- **1.21** CEP Mexico means the non-debtor Mexican subsidiary of CEP, Composite Parts Mexico S.A. de C.V.
  - 1.22 <u>Claim</u> has the meaning set forth in section 101(5) of the Bankruptcy Code.
- **1.23** <u>Claims Register</u> means the list maintained by BMC listing all Claims as scheduled and/or filed in the Cases.
- **1.24** Class means any group of Claims or Equity Interests classified by the Plan in accordance with section 1122(a)(1) of the Bankruptcy Code.
- **1.25** <u>Collateral</u> means any property or interest in property of the Estates subject to a lien, charge or other encumbrance to secure the payment or performance of a Claim, which lien, charge or other encumbrance is not subject to avoidance under the Bankruptcy Code.
- **1.26** <u>Committee</u> means the statutory committee of unsecured creditors appointed by the Office of the United States Trustee in these Cases pursuant to section 1102 of the Bankruptcy Code, as the composition of the same has been or may be modified by the addition or removal of members from time to time.
- **1.27** <u>Confirmation Date</u> means the date on which the Clerk of the Bankruptcy Court enters the Confirmation Order.
- **1.28** <u>Confirmation Hearing</u> means the hearing to be held by the Bankruptcy Court regarding confirmation of the Plan, as such hearing may be adjourned or continued from time to time.
- **1.29** <u>Confirmation Order</u> means the order of the Bankruptcy Court confirming the Plan pursuant to section 1129 of the Bankruptcy Code.
- **1.30** <u>Consolidated Holdings</u> means the post-Effective Date surviving entity following the substantive consolidation of CEP and Thermoplastics with Holdings.
- **1.31** <u>Crestline Pension Plan</u> means the Pension Plan for the bargaining unit employees of CEP's Crestline, Ohio plant.

- **1.32 Debtor** means any one of the Debtors.
- **1.33** <u>Debtors</u> means CEP Holdings, LLC, Creative Engineered Polymer Products, LLC and Thermoplastics Acquisition, LLC, and where the term is used with respect to rights or obligations on or after the Effective Date, it shall mean the Liquidating Trustee.
  - **1.34 Delphi** means Delphi Automotive Systems, LLC and its affiliates and subsidiaries.
  - **1.35 Disbursing Agent** means the Liquidating Trustee.
- **1.36** <u>Disclosure Statement</u> means the disclosure statement relating to the Plan, including, without limitation, all exhibits, schedules and attachments thereto, as approved by the Bankruptcy Court pursuant to section 1125 of the Bankruptcy Code.
- **1.37** <u>Disputed</u> means, with respect to any Claim which has not been Allowed pursuant to the Plan or a Final Order:
- (a) if no proof of claim has been filed by the applicable deadline: (i) a Claim that has been or hereafter is listed on the Schedules as disputed, contingent or unliquidated; or (ii) a Claim that has been or hereafter is listed on the Schedules as other than disputed, contingent or unliquidated, but as to which the Debtors or any other party in interest has interposed an objection or request for estimation which has not been withdrawn or determined by a Final Order; or
- (b) if a proof of claim or request for payment of an Administrative Expense Claim has been filed by the applicable deadline: (i) a Claim for which no corresponding Claim has been or hereafter is listed on the Schedules; (ii) a Claim for which a corresponding Claim has been or hereafter is listed on the Schedules as other than disputed, contingent or unliquidated, but the nature or amount of the Claim as asserted in the proof of claim varies from the nature and amount of such Claim as listed on the Schedules; (iii) a Claim for which a corresponding Claim has been or hereafter is listed on the Schedules as disputed, contingent or unliquidated; or (iv) a Claim for which a timely objection or request for estimation is interposed by the Debtors which has not been withdrawn or determined by a Final Order.
- 1.38 <u>Disputed Claim Amount</u> means the amount set forth in the proof of claim relating to a Claim that is Disputed or an amount estimated pursuant to an order of the Bankruptcy Court in respect of a Claim that is Disputed in accordance with section 502(a) of the Bankruptcy Code and Bankruptcy Rule 3018.
  - **1.39 Disputed Claim Reserve** shall have the meaning set forth in Section 8.3 hereof.
  - **1.40 Distribution Record Date** means the Confirmation Date.
  - **1.41 Delphi** means Delphi Automotive Systems, LLC and its affiliates and subsidiaries.
- **1.42** Effective Date means a Business Day on or after the Confirmation Date specified by the Debtors on which (i) no stay of the Confirmation Order is in effect, and (ii) the conditions to the effectiveness of the Plan specified in section 10.2 hereof have been satisfied or waived.
- **1.43** Equity Interest means the interest of any holder of an equity security of the Debtors represented by any issued and outstanding shares of common or preferred stock or other instrument evidencing a present ownership interest in one or more of the Debtors, whether or not transferable, or any option, warrant or right, contractual or otherwise, to acquire any such interest.

- **1.44** Estates means the bankruptcy estates of CEP, Holdings, and Thermoplastics respectively, in these Cases pursuant to section 541 of the Bankruptcy Code.
- 1.45 Final Order means an order or judgment of the Bankruptcy Court entered by the Clerk of the Bankruptcy Court on the docket in the Cases that has not been reversed, vacated or stayed and as to which (i) the time to appeal, petition for certiorari or move for a new trial, reargument or rehearing has expired and as to which no appeal, petition for certiorari or other proceeding for a new trial, reargument or rehearing shall then be pending, or (ii) if an appeal, writ of certiorari, new trial, reargument or rehearing thereof has been sought, such order or judgment of the Bankruptcy Court shall have been affirmed by the highest court to which such order was appealed, or certiorari shall have been denied or a new trial, reargument or rehearing shall have been denied or resulted in no modification of such order, and the time to take any further appeal, petition for certiorari or move for a new trial, reargument or rehearing shall have expired.
- 1.46 Final DIP Order means that certain Final Order Authorizing Debtors to: (A) Use Cash Collateral; (B) Incur Postpetition Debt; (C) Grant Adequate Protection and Provide Security and Other Relief to Wachovia Capital Finance Corporation (Central); and (D) Grant Certain Related Relief entered by the Bankruptcy Court on October 27, 2006 in these Cases together with all agreements, exhibits or documents attached thereto.
- **1.47** General Unsecured Claim means any Claim against the Debtors that is (i) not a Wachovia Secured Claim, Subordinated Secured Participation Claim, Administrative Expense Claim, Priority Tax Claim, Other Secured Claim or Priority Non-Tax Claim, or (ii) otherwise determined by the Bankruptcy Court to be a General Unsecured Claim.
  - **1.48 GM** means General Motors Corporation and its affiliates and subsidiaries.
  - **1.49** Holdings means CEP Holdings, LLC.
  - **1.50 Impaired** shall have the meaning ascribed to it in section 1124 of the Bankruptcy Code.
- **1.51** <u>Insurance Claim</u> means any claim against one or more of the Debtors to the extent it may be covered and payable under any Insurance Policy of any Debtor and/or the Debtors.
- **1.52** <u>Insurance Policy</u> means any policy of insurance and agreements relating thereto covering the Debtors or that may be available to provide coverage for claims against the Debtors.
  - **1.53 Insurer** means any entity that has issued an Insurance Policy to the Debtor.
- **1.54** <u>Law</u> means any law, rule, regulation, order, decree or other requirement having the force of law and, where applicable, any interpretation thereof by an authority having jurisdiction with respect thereto or charged with administration thereof.
- 1.55 <u>Lien</u> means a judicial lien as defined in section 101(36) of the Bankruptcy Code; a lien as defined in section 101(37) of the Bankruptcy Code; a security interest as defined in section 101(51) of the Bankruptcy Code; a statutory lien as defined in section 101(53) of the Bankruptcy Code; and any other lien, interest, charge or encumbrance.
- **1.56** <u>Liquidating Trustee</u> means Shaun M. Martin, the trustee of the CEP Liquidating Trust appointed in accordance with the Agreement.

- **1.57** <u>Liquidation Expense Reserve Account</u> means the bank account or investment(s) established by the Liquidating Trustee to pay and reserve for all reasonable costs, expenses and fees incurred in connection with maintaining the CEP Liquidating Trust and the Trust Assets.
- 1.58 Other Secured Claim means any Secured Claim against one or more of the Debtors and/or CEP Mexico that is not a Wachovia Secured Claim or Subordinated Participating Customer Secured Claim and has not been paid in full in Cash out of the proceeds of a Sale Transaction in accordance with any agreement of sale and/or order of the Bankruptcy Court approving same.
  - **1.59** Participating Customers means GM, Delphi and Visteon.
- **1.60 PBGC** means the Pension Benefit Guaranty Corporation, an agency of the United States government created by the Employee Retirement Income Security Act of 1971, as amended, to administer the pension plan termination insurance program established under 29 U.S.C. §§ 1301(a)(13) and (14).
- **1.61** Pension Plan means each of the Canton Pension Plan and the Crestline Pension Plan, which collectively shall be the "Pension Plans".
  - **1.62 Person** has the meaning set forth in section 101(41) of the Bankruptcy Code.
  - **1.63** Petition Date means September 20, 2006.
- **1.64** Plan means this joint chapter 11 plan of liquidation, including exhibits, schedules and/or attachments annexed hereto, as the same may be amended or modified from time to time in accordance with the provisions of the Bankruptcy Code, the Bankruptcy Rules and the terms hereof.
  - **1.65 Plan Proponents** means the Debtors and the Committee.
- **1.66** Priority Non-Tax Claim means any Claim, other than an Administrative Expense Claim or a Priority Tax Claim, entitled to priority in payment as specified in section 507(a)(3), (4), (5), (6), (7) or (9) of the Bankruptcy Code.
- **1.67** Priority Tax Claim means any Claim of a governmental unit of the kind entitled to priority in payment as specified in sections 502(i) and 507(a)(8) of the Bankruptcy Code.
- 1.68 <u>Pro Rata Share</u> means a proportionate share, so that the ratio of the consideration distributed on account of an Allowed Claim in a Class to the amount of such Allowed Claim is the same as the ratio of the amount of the consideration distributed on account of all Allowed Claims in such Class to the amount of all Allowed Claims in such Class.
- **1.69 Professionals** means any Bankruptcy Court approved professional person employed by the applicable entity in these Cases at any time before the Confirmation Date.
- **1.70** Reserve Income means the interest earned on all funds maintained in the Disputed Claim Reserve on account of a Claim that is Disputed prior to the distribution of such funds (based upon the average rate of interest earned on all accounts and investments in which the Disputed Claim Reserve are invested as estimated by the Liquidating Trustee).
- 1.71 <u>Sale Transaction</u> means the sale of any assets of any of the Estates approved by the Bankruptcy Court by a Final Order prior to the Effective Date pursuant to section 363 of the Bankruptcy Code, or the sale, transfer or other disposition of the business and/or assets of CEP Mexico as of the

earlier to occur of (i) the Confirmation Date, or (ii) the appointment of a liquidator of CEP Mexico in accordance with applicable Law.

- 1.72 <u>Schedules</u> means the schedules of assets and liabilities and the statements of financial affairs filed by the Debtors under section 521 of the Bankruptcy Code, Bankruptcy Rule 1007 and the Official Bankruptcy Forms of the Bankruptcy Rules, as such schedules and statements have been or may be supplemented or amended from time to time through the Confirmation Date.
- 1.73 <u>Secured Claim</u> means a Claim (i) secured by Collateral, to the extent of the value of such Collateral (a) as set forth in the Plan, (b) as agreed to by the holder of such Claim and the Debtor, or (c) as determined by a Final Order, including the Final DIP Order, in accordance with section 506(a) of the Bankruptcy Code; or (ii) secured by the amount of any rights of setoff of the holder thereof under section 553 of the Bankruptcy Code.
- 1.74 <u>Subordinated Participation Agreement</u> means the agreement among Wachovia, the Subordinated Participating Customers, and BBK to which the Subordinated Participating Customers purchased a subordinated, last out participation in the Wachovia Loan Agreements dated June 30, 2006, as amended as follows: (i) First Amendment to Subordinated Participation Agreement, effective as of August 16, 2006; (ii) Second Amendment to Subordinated Participation Agreement, effective as of September 26, 2006; and (iii) Third Amendment to Subordinated Participation Agreement executed in connection with the Final DIP Order.
- 1.75 <u>Subordinated Secured Participation Claims</u> The principal amount of \$2,318,485.00 (plus accruing interest) allowed and allocated among the Participating Customers pursuant to the Subordinated Participation Agreement, as amended and in accordance with the Final DIP Order.
  - **1.76 Subordinated Participating Customers** means Delphi, GM and Visteon.
  - 1.77 <u>Tax Code</u> means title 26 of the United States Code, as amended from time to time.
  - **1.78** Thermoplastics means Thermoplastics Acquisition, LLC.
- **1.79** Thermoplastics Loan Agreement means that certain Loan and Security Agreement dated as of December 21, 2005 by and between Thermoplastics and Wachovia as amended, supplemented and modified from time to time, and including all schedules, attachments, exhibits and addendums thereto.
- **1.80** Trust Assets means all of the assets of the Debtors and property of their respective Estates, including, but not limited to, the Causes of Action, all of which shall vest in the CEP Liquidating Trust on the Effective Date pursuant to the Plan and in accordance with the Agreement.
  - **1.81 Voting Class** means Class 4 designated in Article III of the Plan.
- 1.82 <u>Voting Procedures Order</u> means that certain Order (A) Approving Proposed Disclosure Statement, (B) Establishing Procedures for Solicitation and Tabulation of Votes to Accept or Reject Proposed Joint Plan of Liquidation and (C) Scheduling Certain Dates in Connection Therewith entered by the Bankruptcy Court in these cases together with any exhibits or documents attached thereto.
  - **1.83** <u>Visteon</u> means Visteon Corporation and its affiliates and subsidiaries.
  - **1.84** <u>Wachovia</u> means Wachovia Capital Finance Corporation (Central).

**1.85** <u>Wachovia Loan Agreements</u> means collectively the CEP Loan Agreement and Thermoplastics Loan Agreement.

## INTERPRETATION; APPLICATION OF DEFINITIONS AND RULES OF CONSTRUCTION

The words "herein," "hereof," "hereto," "hereunder," and other words of similar import refer to the Plan as a whole and not to any particular section, subsection, or clause contained therein. The words "including" and "include" and other words of similar import shall be deemed to be followed by the phrase "without limitation." A term used herein that is not defined herein shall have the meaning assigned to that term in the Bankruptcy Code or Bankruptcy Rules. Wherever from the context it appears appropriate, each term stated shall include both singular and plural, and pronouns shall include the masculine, feminine and neuter regardless of how stated. The rules of construction contained in section 102 of the Bankruptcy Code shall apply to the Plan. The headings in the Plan are for convenience of reference only and shall not limit or otherwise affect the provisions hereof. The exhibits attached to this Plan are incorporated into and are part of this Plan as if fully set forth in this Plan.

#### ARTICLE II.

## ADMINISTRATIVE EXPENSE CLAIMS AND PRIORITY TAX CLAIMS

- **2.1** Administrative Expense Claims. Except to the extent that a holder of an Allowed Administrative Expense Claim agrees to a different treatment, on the Effective Date, or as soon thereafter as is reasonably practicable, the Liquidating Trustee shall pay to each holder of an Allowed Administrative Expense Claim, in full satisfaction of such Claim, an amount in Cash equal to the Allowed amount of such Claim.
- **2.2** Compensation and Reimbursement Claims. All Professionals or other Persons seeking an award by the Bankruptcy Court of compensation for services rendered or reimbursement of expenses incurred through and including the Confirmation Date under sections 327, 328, 330, 331, 503(b)(2), 503(b)(3), 503(b)(4) or 503(b)(5) of the Bankruptcy Code (i) shall file their respective final applications for allowance of compensation for services rendered and reimbursement of expenses incurred by the date that is thirty (30) days after the Confirmation Date, unless otherwise ordered by the Bankruptcy Court; and (ii) shall be paid in full by the Liquidating Trustee in such amounts as are allowed by the Bankruptcy Court (A) upon the date upon which the order relating to any such Administrative Expense Claim is entered, or (B) upon such other terms as may be mutually agreed upon between the holder of such an Administrative Expense Claim and the Committee.
- **2.3** Fees Under 28 U.S.C. §1930. All fees payable in the Cases under 28 U.S.C. §1930, as agreed by the Debtors and Committee or as determined by the Bankruptcy Court, will, if not previously paid in full, be paid in Cash on the Effective Date and will continue to be paid by the Liquidating Trustee as required under 28 U.S.C. §1930 until such time as an order is entered by the Bankruptcy Court closing the Cases.
- **2.4** <u>Wachovia Secured Claim.</u> Except to the extent that the Wachovia Secured Claims has been paid and satisfied prior to the Effective Date in accordance with the Final DIP Order, on or as soon as reasonably practicable after the Effective Date, Wachovia shall receive in full satisfaction of such Claim, Cash in an amount equal to one hundred percent (100%) of the unpaid amount of the Allowed Wachovia Secured Claim. The liens securing such Allowed Wachovia Secured Claim shall be deemed released at such time as the Wachovia Secured Claim is paid in full.

2.5 Priority Tax Claims. Except to the extent that a holder of an Allowed Priority Tax Claim agrees to a different treatment, on the Effective Date, or as soon thereafter as is reasonably practicable, the Liquidating Trustee shall pay to each holder of an Allowed Priority Tax Claim, in full satisfaction of such Claim, an amount in Cash equal to the Allowed amount of such Claim. To the extent that the holder of an Allowed Priority Tax Claim is not paid Cash equal to the Allowed Amount of such Claim on the Effective Date, such holder shall be entitled to interest at a rate determined under applicable non-bankruptcy Law

#### ARTICLE III.

## **CLASSIFICATION OF CLAIMS AND EQUITY INTERESTS**

The following table designates the Classes of Claims against and Equity Interests in the Debtors and specifies which of those Classes are (i) impaired or unimpaired by the Plan, (ii) entitled to vote to accept or reject the Plan in accordance with section 1126 of the Bankruptcy Code and (iii) deemed to reject the Plan:

Class	<u>Designation</u>	<u>Impairment</u>	Entitled to Vote
Class 1	Subordinated Participating Customers Secured Claims	Unimpaired	No
Class 2	Other Secured Claims	Unimpaired	No
Class 3	Priority Non-Tax Claims	Unimpaired	No
Class 4	General Unsecured Claims	Impaired	Yes
Class 5	Equity Interests	Impaired	No (deemed to reject)

For convenience of identification, the Plan classifies the Allowed Claims in Class 2 as a single Class. This Class is actually a group of subclasses, depending on the underlying property securing such Allowed Claims, and each subclass is treated hereunder as a distinct Class for voting and distribution purposes.

#### ARTICLE IV.

## TREATMENT OF CLAIMS AND EQUITY INTERESTS

4.1 <u>Class 1 – Subordinated Secured Participation Claims</u>. To the extent that the Subordinated Secured Participation Claims have not been indefeasibly paid and satisfied prior to the Effective Date in accordance with the Final DIP Order, on or as soon as reasonably practicable after the Effective Date, the Participating Customers shall receive in full satisfaction of their respective Subordinated Secured Participation Claims Cash in an amount equal to one hundred percent (100%) of the unpaid amount of the Allowed Subordinated Secured Participation Claims, plus interest accrued thereon from and after the Petition Date, less the amount of Agreed AR due from each respective Participating Customer. The Liens securing such Allowed Subordinated Secured Participation Claims shall be deemed released at such time as the Subordinated Secured Participation Claims are paid in full.

- 4.2 Class 2 Other Secured Claims. On or as soon as reasonably practicable after the Effective Date, each holder of an Allowed Other Secured Claim shall receive, at the option of the Liquidating Trustee, and in full satisfaction of such Claim, either (i) Cash in an amount equal to one hundred percent (100%) of the Allowed Other Secured Claim (taking into account sections 506(a)(1) and 506(b) of the Bankruptcy Code); (ii) the proceeds of the sale or disposition of the Collateral securing such Allowed Other Secured Claim to the extent of the value of the holder's secured interest in such Collateral, net of the costs of disposition of such Collateral; (iii) such other distribution as necessary to satisfy the requirements of the Bankruptcy Code, including the surrender of any such collateral; or (iv) such other treatment as the Liquidating Trustee and such holder other Secured Claim may agree. In the event that a holder of an Other Secured Claim is treated in accordance with clause (i) or (ii) of this section 4.2, the Liens securing such other Secured Claim shall be deemed released.
- 4.3 <u>Class 3 Priority Non-Tax Claims</u>. Except to the extent that a holder of an Allowed Priority Non-Tax Claim against the Debtors agrees to a different treatment of such Claim, on the Effective Date, or as soon thereafter as is reasonably practicable, each holder of an Allowed Priority Non-Tax Claim shall receive the greater of one hundred percent (100%) of the Allowed Amount of such Priority Non-Tax Claims or a Pro Rata Share of all remaining Cash after payment, or treatment consistent with sections 8.3 and 8.5 hereof, of all Allowed Claims in sections 2.1, 2.2, 2.3, 2.4, 2.5, 4.1 and 4.2, on the Effective Date or as soon thereafter as is practicable.
- 4.4 <u>Class 4 General Unsecured Claims.</u> The holders of Allowed General Unsecured Claims shall receive their Pro Rata Share of all remaining Cash after payment in full of all Allowed Claims, or treatment consistent with sections 8.3 and 8.5 hereof, pursuant to sections 2.1, 2.2, 2.3, 2.4, 2.5, 4.1, 4.2 and 4.3 at such times as the Liquidating Trustee deems practicable, however, in the event that Cash available for distribution to Allowed Class 4 Claims exceeds \$1.5 million, an initial distribution to Allowed Class 4 Claims shall be made no later than ninety (90) days from the date on which the Trust Assets available for distribution to Class 4 Claims exceeds \$1.5 million.
- 4.5 <u>Class 5 Equity Interests</u>. All Equity Interests issued by the Debtors other than Equity Interests in Holdings shall be merged by operation of section 5.1 hereof on the Effective Date. On the Effective Date, the Equity Interests of Holdings shall be transferred or conveyed to the CEP Liquidating Trust for the benefit of the Beneficiaries of the CEP Liquidating Trust. Each holder of an Equity Interest shall neither receive nor retain any property or interest in property on account of such Equity Interest unless and until all Allowed Claims are paid in full with interest.

## ARTICLE V.

# SUBSTANTIVE CONSOLIDATION OF HOLDINGS, CEP AND THERMOPLASTICS

5.1 <u>Substantive Consolidation</u>. On the Effective Date (i) all assets (and proceeds thereof) of the Estates of CEP and Thermoplastics and all liabilities of CEP and Thermoplastics shall be deemed merged or treated as though they were merged into and with the assets and liabilities of the Estate of Holdings; (ii) no distributions shall be made under the Plan on account of inter-company claims among the Debtors and all such Claims shall be eliminated; (iii) all guarantees of the Debtors of the obligations of any other Debtors shall be deemed eliminated and extinguished such that any Claim against any Debtor and any guaranty thereof executed by any other Debtor and any joint or several liability of any of the Debtors shall be deemed to be one obligation of Consolidated Holdings; and (iv) each and every claim filed or to be filed in any of the Cases shall be deemed to be filed against Consolidated Holdings; provided, however, that (a) for the purposes of determining the availability of the right of set-off under

section 553 of the Bankruptcy Code, the Debtors shall be treated as separate entities so that, subject to the other provisions of section 553 of the Bankruptcy Code debts due to any of the Debtors may not be set-off against the debts of any of the other Debtors, and (b) for purposes of determining the Plaintiff of, and availability of defenses to, Avoidance Actions, the Debtors may, at the option of the Liquidating Trustee, be treated as separate entities. To the extent that a creditor is holding an Allowed Claim against more than one of the Debtors' Estates, such creditors shall be treated as a holder of an Allowed Claim as against Consolidated Holdings.

#### ARTICLE VI.

## **PROVISIONS GOVERNING DISTRIBUTIONS**

**6.1 Distribution Record Date.** As of the close of business on the Confirmation Date, the various registers for each of the Classes of Claims or Equity Interests as maintained by the Debtors, or their agents, shall be deemed closed, and there shall be no further changes in the record holders of any of the Claims or Equity Interests unless otherwise specifically provided under the Plan. Neither the Debtors nor the CEP Liquidating Trustee shall have any obligation to recognize any transfer of the Claims or Equity Interests occurring on or after the Confirmation Date.

## 6.2 Method of Distributions Under the Plan.

- Date, the Debtors shall remit to the Liquidating Trustee, all assets and property of the Estates, and the Liquidating Trustee shall pay as soon as practicable holders of Allowed Claims pursuant to section 2.1, 2.2, 2.3, 2.4, 2.5, 4.1, 4.2 and 4.3 an amount in Cash equal to the Allowed amount of such Claims or in such other amounts as set forth in the Plan. The transfer of the assets as described in the preceding sentence to the CEP Liquidating Trust shall be made for the benefit of the Beneficiaries, in each case, but only to the extent that the Beneficiaries are entitled to distributions under the Plan.
- Disbursing Agent. All distributions under the Plan will be made by the (b) Liquidating Trustee. The Liquidating Trustee will not be required to give any bond or surety or other security for the performance of its duties unless otherwise ordered by the Bankruptcy Court, and in the event that the Liquidating Trustee is otherwise so ordered, all costs and expenses of procuring any such bond or surety will be borne by the CEP Liquidating Trust. The Liquidating Trustee shall be empowered to (i) effect all actions and execute all agreements, instruments and other documents necessary to perform its duties under the Plan and/or CEP Liquidating Trust, (ii) make all distributions contemplated in the Plan, (iii) employ professionals to represent it with respect to its responsibilities under the Plan and (iv) exercise such other powers as may be vested in the Liquidating Trustee by order of the Bankruptcy Court, pursuant to the Plan and/or Liquidating Trust, or as deemed by the Liquidating Trustee to be necessary and proper to implement the provisions of the Plan. Except as otherwise ordered by the Bankruptcy Court, the amount of any reasonable fees and expenses incurred by the Liquidating Trustee on or after the Effective Date (including without limitation, taxes) and any reasonable compensation and expense reimbursement claims (including, without limitation, reasonable attorneys' fees and other professional fees and expenses) made by the Liquidating Trustee shall be paid in Cash subject to the terms and conditions of the CEP Liquidating Trust.
- (c) <u>Distributions of Cash</u>. At the option of the Liquidating Trustee, any Cash payment to be made hereunder may be made by check or wire transfer or as otherwise required or provided in applicable agreements.

- **Delivery of Distributions.** Subject to Bankruptcy Rule 9010, unless otherwise (d) provided in the Plan, all distributions to any holder of an Allowed Claim will be made to the holder of each Allowed Claim at the address of such holder as listed in the Schedules, or on the books and records of the Debtors' claims agent BMC unless the Debtors or Liquidating Trustee, as the case may be, have been notified, in advance, in writing of a change of address, including, without limitation, by the timely filing of a proof of claim or interest by such holder that provides an address for such holder different from the address reflected in the Schedules or in the Debtors' books and records. In the event that any distribution to any holder is returned as undeliverable, no distribution to such holder will be made unless and until the Liquidating Trustee has been notified of the then current address of such holder, at which time or as soon as reasonably practicable thereafter, such distribution will be made to such holder without interest; provided, however, that, such undeliverable distributions will be deemed unclaimed property under section 347(b) of the Bankruptcy Code at the expiration of ninety (90) days after the date of distribution in accordance with the section on unclaimed distributions below. The Liquidating Trustee will have no obligation to attempt to locate any holder of an Allowed Claim other than by reviewing the Schedules and its books and records maintained by BMC (including any proofs of claim filed against the Debtors).
- **6.3** Withholding and Reporting Requirements. In connection with the Plan and the CEP Liquidating Trust and all instruments issued in connection therewith and distributed thereon, the Liquidating Trustee shall comply with all applicable withholding and reporting requirements imposed by any federal, state or local taxing authority, and all distributions under the Plan shall be subject to any such withholding or reporting requirements.
- Allowed Claims shall be null and void if not negotiated within sixty (60) days after the date of issuance thereof. Requests for reissuance of any check shall be made to the Liquidating Trustee by the holder of the Allowed Claim to whom such check was originally issued. Any Claim in respect of such a voided check shall be made on or before thirty (30) days after the expiration of the sixty (60) day period following the date of issuance of such check. Thereafter, the amount represented by such voided check shall irrevocably revert to the CEP Liquidating Trust and any Claim in respect of such voided check shall be discharged and forever barred from assertion against the Debtors and their property or the CEP Liquidating Trust, as the case may be.
- **6.5** Minimum Distributions. No payment of Cash less than \$50 shall be made by the Liquidating Trustee. Any Trust Assets that are undistributable in accordance with this section 6.5 shall vest in the CEP Liquidating Trust for the benefit of its Beneficiaries.
- 6.6 Setoffs. The Debtors or the Liquidating Trustee, as the case may be, may, but shall not be required, to set off against any Claim (for purposes of determining the Allowed amount of such Claim on which distribution shall be made), any claims of any nature whatsoever that the Debtors may have against the holder of such Claim, but neither the failure to do so nor the allowance of any Claim hereunder shall constitute a waiver or release by the Debtors or the Liquidating Trustee of any such claim the Debtors may have against the holder of such Claim.
- **6.7** Transactions on Business Days. If the Effective Date or any other date on which a transaction may occur under the Plan shall occur on a day that is not a Business Day, the transactions contemplated by the Plan to occur on such day shall instead occur on the next Business Day but shall be deemed to have been completed as of the required date.

**6.8** Allocation of Plan Distribution Between Principal and Interest. All distributions in respect of any Allowed Claim shall be allocated first to the principal amount of such Allowed Claim, as determined for federal income tax purposes, and thereafter, to the remaining portion of such Claim comprising interest, if any (but solely to the extent that interest is an allowable portion of such Allowed Claim).

#### ARTICLE VII.

## MEANS FOR IMPLEMENTATION AND EXECUTION OF THE PLAN

## 7.1 The CEP Liquidating Trust

- (a) <u>Execution of the Agreement</u>. On the Effective Date, the Agreement shall be executed, and all other necessary steps shall be taken to establish the CEP Liquidating Trust and the beneficial interests therein, which shall be for the benefit of all creditors and parties in interest entitled to receive distributions under the Plan. This section 7.1 sets forth certain of the rights, duties and obligations of the Liquidating Trustee. In the event of any conflict between the terms of section 7.1 of the Plan and the terms of the Agreement, the terms of the Agreement shall govern.
- (b) <u>Purpose of CEP Liquidating Trust</u>. The CEP Liquidating Trust shall be established and maintained for the sole purpose of liquidating and distributing assets of the Estates and proceeds thereof, including but not limited to the \$5,408,597 in principal, plus interest, relating to CEP Mexico contributed to the CEP Liquidating Trust (and constituting Trust Assets), in accordance with Treasury Regulation section 30.1.7701-4(d), and resolving and administering Claims, with no objective to continue or engage in the conduct of a trade or business.
  - (c) Transfer of Assets of the Estates to the CEP Liquidating Trust.
- (i) On the Effective Date, right, title and interest to all assets as described in section 6.2(a) hereof shall vest in the CEP Liquidating Trust as Trust Assets.
- (ii) On the Effective Date, the Estates shall be deemed to assign, set over, transfer and convey to the CEP Liquidating Trust all of their right, title and interest in assets of the Estates, including but not limited to the \$5,408,547 principal, plus interest, relating to CEP Mexico. To the extent that certain assets of the Estates, because of their nature or because they will accrue subsequent to the Effective Date, cannot be assigned, set over, transferred or conveyed to the CEP Liquidating Trust on the Effective Date, such assets shall be deemed assigned, set over, transferred and conveyed to the CEP Liquidating Trust as soon as practical after the Effective Date.
- (iii) On or after the Effective Date, the Liquidating Trustee shall continue as a plaintiff in all litigation or Causes of Action (on behalf of the Beneficiaries) in which the Debtors were plaintiffs prior to the Effective Date. All recoveries and proceeds arising from litigation and Causes of Action shall be deemed assigned, set over, transferred and conveyed to the CEP Liquidating Trust upon receipt thereof. All fees and costs of the Liquidating Trustee (and agents thereof) arising from or relating to pursuing litigation or Causes of Action or other services performed at the request of the Liquidating Trustee on behalf of the CEP Liquidating Trust shall be paid as a first priority in right of distribution from the CEP Liquidating Trust.

- (iv) The transfer of assets of the Estates, including but not limited to the \$5,408,597 principal, plus interest, relating to CEP Mexico to the Liquidating Trust shall be made for the benefit of the Beneficiaries, in each case, but only to the extent the Beneficiaries are entitled to distributions under the Plan.
- (d) <u>Governance of CEP Liquidating Trust</u>. The CEP Liquidating Trust shall be governed by the Liquidating Trustee in accordance with the Agreement and consistent with the Plan.
- (e) <u>The Liquidating Trustee</u>. Shaun M. Martin shall be the Liquidating Trustee. The designation of the Liquidating Trustee shall be effective on the Effective Date without the need for a further order of the Bankruptcy Court. The Liquidating Trustee shall exercise reasonable business judgment to administer Trust Assets and to make timely distributions from the CEP Liquidating Trust to its Beneficiaries.
- (f) <u>Nontransferability of Interests in the CEP Liquidating Trust</u>. The beneficial interests in the CEP Liquidating Trust shall not be certificated and are not transferable (except as otherwise provided in the Agreement).
- (g) <u>Cash</u>. The Liquidating Trustee may invest Cash consisting of the Trust Assets (including any earnings thereon or proceeds therefrom) in short term overnight investments or longer term investments as permitted by section 345 of the Bankruptcy Code; *provided*, *however*, that such investments are investments permitted to be made by a liquidating trust within the meaning of Treasury Regulation section 301.7701-4(d), as reflected therein, or under applicable Internal Revenue Service guidelines, rulings, or other controlling authorities.
- (h) <u>Costs and Expenses of the Liquidating Trustee</u>. The costs and expenses of the CEP Liquidating Trust, including the fees and expenses of the Liquidating Trustee and its retained professionals, shall be paid out of the CEP Liquidating Trust. Such costs and expenses shall be treated as administrative expenses of the Estates, having, however, a first priority in right of distribution, and without the need for Bankruptcy Court authorization or entry of a Final Order. The Liquidating Trustee shall retain such amounts as are reasonably necessary (at the discretion of the Liquidating Trustee) to meet the future fees and expenses incurred in administering the CEP Liquidating Trust in the Liquidation Expense Reserve Account.
- (i) <u>Compensation of the Liquidating Trustee</u>. The Liquidating Trustee shall be entitled to reasonable compensation, which shall be on economic terms agreeable to the Committee and established prior to the Effective Date. The compensation payable to the Liquidating Trustee shall be treated consistent with section 7.1(h) hereunder.
- (j) <u>Distribution of Trust Assets</u>. The Liquidating Trustee shall distribute at least annually and in accordance with the Agreement, beginning on the Effective Date or as soon thereafter as is practicable, the Trust Assets on hand (including any Cash received from the Debtors and CEP Mexico on the Effective Date, and treating as Cash for purposes of this section 7.1 any permitted investments under Section 7.1(g) hereof), except such amounts (i) as would be distributable to a holder of a Disputed Claim if such Disputed Claim had been Allowed prior to the time of such distribution (but only until such Claim is resolved); (ii) as are reasonably necessary to meet contingent liabilities and to maintain the value of the assets of the CEP Liquidating Trust during liquidation; (iii) to pay reasonable expenses (including, but not limited to, any taxes imposed on the assets of the CEP Liquidating Trust or in respect of the CEP Liquidating Trust); and (iv) to satisfy other liabilities incurred by the CEP Liquidating Trust in accordance with this Plan or the Agreement.

- (k) <u>Retention of Professionals by the Liquidating Trustee</u>. The Liquidating Trustee may retain and reasonably compensate counsel and other professionals to assist in its duties as Liquidating Trustee on such terms as the Liquidating Trustee deems appropriate without Bankruptcy Court approval. The Liquidating Trustee may retain any professional who represented parties in interest (including the Debtors, CEP Mexico or the Committee) in the Cases.
- (l) <u>Authority to Grant Releases</u>. Without limiting the generality of section 7.6 hereof, in connection with the compromise and settlement of any Causes of Action, the Liquidating Trustee is authorized to release and discharge, to the fullest extent permitted by law, non-Debtor parties to Causes of Action from all Claims and Causes of Action that, the Committee or the Liquidating Trustee (as successor to the Debtors on behalf of the Beneficiaries) has or may have whether known or unknown against such Persons. Any settlement effectuated prior to the Confirmation Date, upon notice thereof to the Bankruptcy Court, shall be deemed incorporated into the Plan and entry of the Confirmation Order including provisions of such settlement shall be deemed a settlement pursuant to section 1123(b)(3)(A) of the Bankruptcy Code.
- (m) <u>Attorney-Client Privilege</u>. Any attorney-client privilege, work-product privilege or other privilege or immunity that the Estates or Committee are entitled to assert in the Causes of Action shall vest in the Liquidating Trustee (and the attorneys and agents of the Liquidating Trustee) and the Liquidating Trustee shall be entitled to assert such privilege and immunity to the same extent that the Debtors and Committee were entitled to do so prior to the Effective Date.
- Transfer of Claims of Debtors and Their Estates to CEP Liquidating Trust. (n) Included among assets and property of the Estates referred to in section 6.2(a) are Causes of Action. A full investigation of potential Causes of Action has not been undertaken by either the Debtors, CEP Mexico or the Committee. It is, however, the Committee's belief that potential Causes of Action may exist as against, among others (i) the Carlisle Companies and/or Parker-Hannifin Corporation and/or their respective subsidiaries, affiliates, agents or other Persons arising from or relating to the Carlisle Transaction (as defined in the Disclosure Statement) and/or the Parker-Hannifin Transaction (as defined in the Disclosure Statement); (ii) Persons (and/or subsidiaries and affiliates thereof) identified in the Debtors' items 3(a) and 3(b) of the respective Debtors' Statements of Financial Affairs with respect to Avoidance Actions (and other Persons subsequently discovered by the Committee or Liquidating Trustee as having received payments that should have been listed and identified in item 3(a) or 3(b) of the respective Debtors' Statements of Financial Affairs and were not); (iii) any one or more of the Participating Customers for failure to cause all Cash Infusions required to be released to the Debtors in accordance with, inter alia, the Final DIP Order and/or its related agreements, to have been actually released to the Debtors; (iv) other Causes of Action identified by the Liquidating Trustee in the course of their post-Effective Date due diligence.
  - (o) Federal Income Tax Treatment of CEP Liquidating Trust.
- (i) <u>Trust Assets Treated as Owned by Creditors</u>. For all federal income tax purposes, all parties (including, without limitation, the Debtors, the Liquidating Trustee and the holders of beneficial interests in the CEP Liquidating Trust) shall treat the transfer of the Trust Assets to the CEP Liquidating Trust for the benefit of the Beneficiaries thereof, whether Allowed on or after the Effective Date, as (A) a transfer of the Trust Assets directly to the Beneficiaries (other than to the extent allocable to Disputed General Unsecured Claims) followed by (B) the transfer by the Beneficiaries to the CEP Liquidating Trust of the Trust Assets in exchange for beneficial interests in the CEP Liquidating Trust. Accordingly, Beneficiaries shall be treated for federal income tax purposes as the grantors and owners of their respective shares of the Trust Assets.

## (ii) <u>Tax Reporting</u>.

(A) The Liquidating Trustee shall file returns for the CEP Liquidating Trust as a grantor trust pursuant to Treasury Regulation section 1.671-1 through 4(a) and in accordance with this section 7.1(o). The Liquidating Trustee shall also annually send to each record holder of a beneficial interest a separate statement setting forth the holder's share of items of income, gain, loss, deduction or credit and will instruct all such holders to report such items on their federal income tax returns or to forward the appropriate information to the beneficial holders with instructions to report such items on their federal income tax returns. The CEP Liquidating Trust's taxable income, gain, loss, deduction or credit will be allocated (subject to section 7.1(o)(ii)(C) hereof, relating to Disputed General Unsecured Claims) to the holders of Allowed General Unsecured Claims in accordance with their relative beneficial interests in the CEP Liquidating Trust.

(B) As soon as possible after the Effective Date, but in no event later than permitted by applicable non-bankruptcy law, the Liquidating Trustee shall make a good faith valuation of the Trust Assets. Such valuation shall be made available from time to time, to the extent relevant, and used consistently by all parties (including, without limitation the Debtors, the Liquidating Trustee, and the holders of Allowed General Unsecured Claims) for all federal income tax purposes. The Liquidating Trustee shall also file (or cause to be filed) any other statements, returns or disclosures relating to the CEP Liquidating Trust that are required by any governmental unit.

Subject to definitive guidance from the Internal Revenue Service or a court of competent jurisdiction to the contrary (including the receipt by the Liquidating Trustee of a private letter ruling if the Liquidating Trustee so requests one, or the receipt of an adverse determination by the Internal Revenue Service upon audit if not contested by the Liquidating Trustee), the Liquidating Trustee shall (i) treat any Trust Assets allocable to, or retained on account of, Disputed Claims as held by one or more discrete trusts for federal income tax purposes (the "Disputed Claim Reserve") in accordance with the trust provisions of the Tax Code (section 641 et seq.); (ii) treat as taxable income or loss of the Disputed Claim Reserve, with respect to any given taxable year, the portion of the taxable income or loss of the CEP Liquidating Trust that would have been allocated to the holders of Disputed Claims had such Claims been Allowed on the Effective Date (but only for the portion of the taxable year with respect to which such Claims are unresolved); (iii) treat as a distribution from any reserve established under CEP Liquidating Trust any increased amounts distributed by the CEP Liquidating Trust as a result of any Disputed Claims resolved earlier in the taxable year, to the extent such distributions relate to taxable income or loss of the CEP Liquidating Trust determined in accordance with the provisions hereof; and (iv) to the extent permitted by applicable law, report consistent with the foregoing for state and local income tax purposes. All holders of Allowed General Unsecured Claims shall report, for tax purposes, consistent with the foregoing.

(D) The Liquidating Trustee shall be responsible for payments, out of the Trust Assets, of any taxes imposed on the CEP Liquidating Trust or the Trust Assets, including any Disputed Claim Reserve established under the CEP Liquidating Trust. In the event, and to the extent, any Cash retained on account of Disputed General Unsecured Claims in any Disputed Claim Reserve established under the CEP Liquidating Trust is insufficient to pay the portion of any such taxes attributable to the taxable income arising from the assets allocable to, or retained on account of, Disputed General Unsecured Claims, such taxes shall be (i) reimbursed from any subsequent Cash amounts retained on account of Disputed General Unsecured Claims, or (ii) to the extent such Disputed General Unsecured Claims have subsequently been resolved, deducted from any amounts distributable by the Liquidating Trustee as a result of the resolutions of such Disputed General Unsecured Claims.

- (E) The Liquidating Trustee may request an expedited determination of taxes of the CEP Liquidating Trust under section 505(b) of the Bankruptcy Code for all returns filed for, or on behalf of, the CEP Liquidating Trust for all taxable periods through the dissolution of the CEP Liquidating Trust.
- (p) <u>Dissolution</u>. The CEP Liquidating Trust and the Liquidating Trustee shall be discharged or dissolved, as the case may be, at such time as (i) all Disputed General Unsecured Claims have been resolved, (ii) all Trust Assets have been liquidated and (iii) all distributions required to be made by the Liquidating Trustee under the Plan have been made, but in no event shall the CEP Liquidating Trust be dissolved later than a period that would adversely affect the status of the CEP Liquidating Trust as a liquidating trust for federal income tax purposes. The CEP Liquidating Trust shall only exist for a period as long as is necessary to facilitate or complete the recovery and liquidation of the Trust Assets and distribution of their proceeds. The Liquidating Trustee shall not unduly prolong the duration of the CEP Liquidating Trust and shall at all times endeavor to resolve, settle or otherwise dispose of all claims that constitute Trust Assets and to effect the distribution of the Trust Assets in accordance with the terms hereof and terminate the CEP Liquidating Trust as soon as practicable. Prior to and upon termination of the CEP Liquidating Trust, the Trust Assets will be distributed to the Beneficiaries of CEP Liquidating Trust, pursuant to the provisions set forth in sections V.E and V.F of the CEP Liquidating Trust and Articles 2 and 4 of the Plan.
- Indemnification of Liquidating Trustee. The Liquidating Trustee and the Liquidating Trustee's agents and professionals shall not be liable for actions taken or omitted in its capacity as, or on behalf of, the Liquidating Trustee, except those acts arising out of its or their own willful misconduct, gross negligence, bad faith, self-dealing, breach of fiduciary duty or *ultra vires* acts, and each shall be entitled to indemnification and reimbursement for fees and expenses in defending any and all of its actions or inactions in its capacity as, or on behalf of, the Liquidating Trustee, except for any actions or inactions involving willful misconduct, gross negligence, bad faith, self-dealing, breach of fiduciary duty; or *ultra vires* acts. Any indemnification claim of the Liquidating Trustee (and the other parties entitled to indemnification under this subsection 7.1(q)) shall be satisfied first from the Trust Assets. The Liquidating Trustee shall be entitled to rely, in good faith, on the advice of its retained professionals.
- 7.2 <u>Distributions to Holders as of the Confirmation Date</u>. As of the close of business on the Confirmation Date, the Claims Register, the equity register and transfer and other registers as maintained by the Debtors and/or their respective agents, as applicable, will be closed and there will be no further changes in the record holder of any Claim or Equity Interest. The Liquidating Trustee will have no obligation to recognize any transfer of any Claim or Equity Interest occurring after the Confirmation Date. The Liquidating Trustee will instead be authorized and entitled to recognize and deal for all purposes under the Plan with only those record holders stated on the Claims Register and other registers as of the close of business on the Confirmation Date.
- 7.3 Closing of Cases by Charitable Gift. If at any time the Liquidating Trustee determines with consent of the Committee that the expense of administering the CEP Liquidating Trust so as to make a final distribution to Allowed Claims is likely to exceed the value of the Trust Assets remaining in the CEP Liquidating Trust, the Liquidating Trustee shall apply to the Bankruptcy Court for authority to (i) reserve any amounts necessary to close the Cases; (ii) donate any balance to a charitable organization exempt from federal income tax under section 501(c)(3) of the Tax Code that is unrelated to the Debtors and any insider of the Debtors; and (iii) close the Cases in accordance with the Bankruptcy Code and Bankruptcy Rules. Notice of such application shall be given electronically, to the extent practicable, to

those parties who have filed requests for notices and whose electronic addresses remain current and operating.

- 7.4 Release of Liens. Except as otherwise specifically provided in or contemplated by the Plan or in any contract, instrument or other agreement or document created in connection with the Plan, (i) each holder of: (a) any purported Secured Claim and/or (b) any judgment, personal property or ad valorem tax, molder, warehouse or artisan or similar Lien Claim, in each case regardless of whether such Claim is an Allowed Claim, shall, on or immediately before the Effective Date and regardless of whether such Claim has been scheduled or proof of such Claim has been filed: (y) turn over and release to the Estates any and all property of the Debtors or Estates that secures or purportedly secures such Claim, or such Lien and/or Claim shall automatically, and without further action by the Debtors or the Liquidating Trustee, as the case may be, be deemed released and (z) execute such documents and instruments as the Debtors or Liquidating Trustee, as the case may be, requires to evidence the holder of a Claim's release of such property or Lien, and if such holder refuses to execute appropriate documents or instruments, the Debtors or the Liquidating Trustee, as the case may be, in its discretion, file a copy of the Confirmation Order in the appropriate recording office, which shall serve to release any holder of a Claim's rights in such property; and (ii) on the Effective Date, all right, title and interest in such property shall revert or be transferred to the CEP Liquidating Trust as Trust Assets, free and clear of all Claims, interests, and Liens of any kind.
- **7.5** Cancellation of Existing Securities and Agreements. On the Effective Date, except as expressly provided in this Plan, the securities, promissory notes, trust indentures, share certificates, security agreements, deeds of trust, collateral agency agreements and other instruments evidencing or securing a Claim shall be deemed cancelled without further act or action under any applicable agreement or Law, and the obligations of the Debtors and the Liquidating Trustee, as successor to the Debtors under the agreements, instruments, trust indentures and certificates governing and securing such Claims, as the case may be, shall be discharged.
- 7.6 <u>Liquidating Trustee's Post-Confirmation Role</u>. All rights and obligations of the Debtors under this Plan that exist or continue on or after the Effective Date shall vest in the Liquidating Trustee and shall be rights and obligations exercisable by the Liquidating Trustee on and after the Effective Date. Further, the Liquidating Trustee shall perform each of the following acts as soon as practicable on or after the Effective Date:
- Liquidating Trust and the Plan, the Liquidating Trustee shall (A) have the power and authority to hold, manage, sell and distribute the Trust Assets in accordance with the Plan, (B) have the power and authority to prosecute and resolve, in the name of the Debtors, CEP Mexico and/or the name of the Liquidating Trustee, Causes of Action, (C) have the power and authority to prosecute and resolve objections to Disputed Claims, (D) have the power and authority to perform such other functions as are provided in the Plan and (E) have the power and authority to administer the closure of the Cases. The Liquidating Trustee shall be responsible for all decisions and duties with respect to the CEP Liquidating Trust and the Trust Assets. In all circumstances, the Liquidating Trustee shall act in the best interests of all beneficiaries of the CEP Liquidating Trust and in furtherance of the purpose of the CEP Liquidating Trust.
- (b) <u>Payments and Transfers</u>. On the Effective Date, or as soon thereafter as is reasonably practicable the Liquidating Trustee, shall make payments and transfers to holders of Allowed Claims to claimants in the manner set forth at sections 2.1, 2.2, 2.3, 2.4, 2.5, 4.1, 4.2, 4.3 and 4.4 hereof, and in accordance with section 6.2 hereof.

- (c) <u>Administration of Taxes</u>. The Liquidating Trustee shall be authorized to exercise all powers regarding the Debtors' and CEP Mexico's tax matters, including filing tax returns, to the same extent as if the Liquidating Trustee were the debtor in possession. The Liquidating Trustee shall (A) complete and file within ninety (90) days of the filing for dissolution by CEP, Thermoplastics, Holdings and CEP Mexico to the extent not previously filed, the Debtors' final federal, state, and local tax returns; (B) request an expedited determination of any unpaid tax liability of the Debtors under section 505(b) of the Bankruptcy Code for all tax periods of the Debtors ending after the Petition Date through the liquidation of the Debtors as determined under applicable tax laws, to the extent not previously requested; and (C) represent the interest and account of the Debtors before any taxing authority in all matters, including, but not limited to, any action, suit, proceeding or audit.
- Distributions. The Liquidating Trustee shall have the power and authority to prosecute and resolve objections to Disputed Subordinated Secured Participation Claims, Disputed Administrative Expense Claims, Disputed Priority Tax Claims, Disputed Other Secured Claims, Disputed Priority Non-Tax Claims and Disputed General Unsecured Claims. The Liquidating Trustee shall have the right, power and authority to retain and assert all defenses, rights of setoff, recoupment and counterclaims with respect to each of the foregoing. The Liquidating Trustee shall also have the power and authority to hold, manage and distribute Plan distributions to the holders of Allowed Claims consistent with applicable provisions of this Plan.

Notwithstanding the foregoing, all Insurance Claims that arose prior to the Petition Date shall be deemed Allowed in the greater of (i) an amount equal to (x) the amount of any remaining self-insured retention under the Insurance Policy applicable to such Insurance Claim, divided by (y) the number of Insurance Claims to which the applicable Insurance Policy and self-insured retention applies; and (ii) \$0.00. The allowance of an Insurance Claim pursuant to this provision shall not be deemed an admission of liability by CEP, Thermoplastics and/or Holdings, and therefore shall not be conclusive as to the Debtors' liability in any proceeding by the holder of an Insurance Claim against an Insurer.

- Pooks and Records. Unless applicable non-bankruptcy law permits the distribution or destruction of certain of the Debtors' and CEP Mexico's business records at an earlier date, the Liquidating Trustee shall have the responsibility of storing and maintaining books and records until one year after the Effective Date, after which time such books and records may, at the sole discretion of the Liquidating Trustee, be abandoned or destroyed without further Bankruptcy Court order, unless applicable non-bankruptcy law requires the retention and maintenance of any such books and records for a longer period, in which instance the Liquidating Trustee shall retain such books and records for at least the minimum period required by applicable non-bankruptcy law. For purposes of this section, books and records include computer generated or computer maintained books and records and computer data, as well as electronically generated or maintained books and records or data, along with books and records of the Debtors and CEP Mexico maintained by or in possession of third parties and all of the claims and rights of the Debtors and CEP Mexico in and to their books and records, wherever located.
- 7.8 Corporate Action. On the Effective Date, the Liquidating Trustee, shall perform each of the actions and effect each of the transfers required by the terms of the Plan in the time period allocated therefor, and all matters provided for under the Plan that would otherwise require approval of the stockholders, directors or comparable governing bodies of the Debtors shall be deemed to have occurred and shall be in effect from and after the Effective Date pursuant to the applicable general corporation law (or other applicable governing law) of the state in which the Debtors incorporated or organized, without any requirement of further action by the stockholders or directors (or other governing body) of the Debtors.

- **7.9** Effectuating Documents and Further Transactions. The Debtors and/or Liquidating Trustee, as the case may be, are authorized and directed to execute, deliver, file or record such contracts, instruments, releases, indentures, and other agreements or documents and take such actions as may be necessary or appropriate to effectuate and further evidence the terms and conditions of the Plan.
- **7.10** Securities Exempt. The beneficial interests in the CEP Liquidating Trust satisfy the requirements of section 1145 of the Bankruptcy Code and, therefore, such interests are exempt from registration under the Securities Act of 1933, as amended, and any state or local law requiring registration.
- Pension Plans. The Plan Proponents and The Reserve Group Management Company contemplate filing a Bankruptcy Rule 9019 motion that seeks Bankruptcy Court approval of a settlement among the Debtors, Committee, The Reserve Group Management Company and affiliates that provides, inter alia, (a) Superior Fabrication Company, LLC, an affiliate of The Reserve Group Management Company, will assume the Pension Plans; (b) The Reserve Group Management Company will release all claims against the Debtors and their Estates, other than those to which The Reserve Group Management Company may, through subrogation to Washington Penn Plastics Company, become a party; (c) The Debtors, the Estates and the Committee will release The Reserve Group Management Company and its affiliates of claims and Causes of Action connected with their dealings and involvement with/on behalf of the Debtors; and (d) the Estates will pay (i) to the PBGC an amount not to exceed \$42,482 with respect to post-Petition Date premiums due under the Pension Plans; (ii) as a contribution to the Pension Plans an amount not to exceed \$29,776 with respect to normal costs charged to funding standard accounts under the Pension Plans post-Petition Date; and (iii) as reimbursement to the Pension Plans an amount not to exceed \$20,000 with respect to post-Petition Date in respect of actuarial services provided by Watson Wyatt to the Pension Plans. The settlement referenced herein contains several conditions precedent including Bankruptcy Court approval; agreement by the USW to freeze accrual of benefits under the Pension Plans as of a date certain (contemplated to be June 30, 2007); and agreement by the PBGC to withdraw its proofs of claim for unfunded liabilities relating to the Pension Plans.

#### ARTICLE VIII.

## **PROCEDURES FOR DISPUTED CLAIMS**

- **8.1** Objections to Claims. The Debtors, Committee and from and after the Effective Date the Liquidating Trustee, shall be entitled to object to Administrative Expense Claims, Priority Tax Claims, Subordinated Secured Participation Claims, Other Secured Claims, Priority Non-Tax Claims and General Unsecured Claims.
- **8.2 No Distribution Pending Allowance.** Notwithstanding any other provision hereof, if any portion of a Claim is a Disputed Claim, no payment or distribution provided hereunder shall be made on account of such Claim unless and until such Disputed Claim becomes an Allowed Claim.

## **8.3** Reserve on Account of Disputed Claims

(a) Establishment and Maintenance of Reserve for Disputed Claims. The Liquidating Trustee shall maintain the Disputed Claim Reserve at an amount equal to the aggregate of 100% of the distributable amounts to which holders of such Disputed Claims would be entitled under the Plan if such Disputed Claims were Allowed Claims in their Disputed Claim Amounts or such lesser amount as required by a Final Order. For purposes of effectuating the provisions of this section 8.3(a) and the distributions to holders of Allowed Claims, the Bankruptcy Court shall set, may fix or liquidate

the amount of Disputed Claims pursuant to section 502(c) of the Bankruptcy Code, in which event the amounts so fixed or liquidated shall be deemed the amounts of the Disputed Claims for purposes of distribution under this Plan. In lieu of fixing or liquidating the amount of any Disputed Claim, the Bankruptcy Court may determine the amount to be reserved for such Disputed Claim or such amount may be fixed by agreement in writing between the Liquidating Trustee and the holder of a Disputed Claim.

- (b) <u>Distributions Upon Allowance of Disputed Claims</u>. The holder of a Disputed Claim that becomes an Allowed Claim shall receive distributions in Cash from the Disputed Claim Reserve as soon as practical following the date on which such Disputed Claim becomes an Allowed Claim pursuant to a Final Order. Such distributions shall be made in accordance with the Plan based upon the distributions that would have been made to such holder under the Plan if the Disputed Claim had been an Allowed Claim on or prior to the Effective Date. No holder of a Disputed Claim shall have any Claim against the Disputed Claim Reserve or the CEP Liquidating Trust with respect to such Claim until such Disputed Claim shall become an Allowed Claim, and no holder of a Disputed Claim shall have any right to interest on such Disputed Claim except for any Reserve Income earned thereon.
- **8.4** Resolution of Disputed Claims. Unless otherwise ordered by the Bankruptcy Court after notice and a hearing, following the Effective Date, the Liquidating Trustee shall have the right to the exclusion of all others (except as to applications for allowances of compensation and reimbursement of expenses under sections 328(a), 330 and 503 of the Bankruptcy Code) to make and file objections to Claims and shall serve a copy of each objection upon the holder of the Claim to which the objection is made as soon as practicable, but in no event later than one hundred eighty (180) days after the Effective Date (subject, however, to the right the Liquidating Trustee to seek an extension of time to file such objections by seeking such extension with approval from the Bankruptcy Court).
- 8.5 Estimation. The Debtors or the Liquidating Trustee (as the case may be) may at any time request that the Bankruptcy Court estimate any contingent, unliquidated or Disputed Claim pursuant to section 502(c) of the Bankruptcy Code regardless of whether the Debtors previously objected to such Claim, and the Bankruptcy Court will retain jurisdiction to estimate any Claim at any time during litigation concerning any objection to any Claim, including, without limitation, during the pendency of any appeal relating to any such objection. In the event that the Bankruptcy Court estimates any contingent, unliquidated or Disputed Claim, the amount so estimated shall constitute either the Allowed amount of such Claim or a maximum limitation on such Claim, as determined by the Bankruptcy Court. If the estimated amount constitutes a maximum limitation on the amount of such Claim, the Debtors or the Liquidating Trustee (as the case may be) may pursue supplementary proceedings to object to the allowance of such Claim. All of the aforementioned objection, estimation and resolution procedures are intended to be cumulative and not exclusive of one another. On and after the Confirmation Date, Claims that have been estimated may be compromised, settled, withdrawn or otherwise resolved subsequently, without further order of the Bankruptcy Court.
- 8.6 <u>Distributions to Holders of Allowed Claims Upon Disallowance of Disputed Claims</u>. Subject to section 7.2 hereof, upon disallowance of any Disputed Claim, each holder of an Allowed Claim in the same Class as the disallowed Disputed Claim will be entitled to its Pro Rata Share of Cash equal to the distribution that would have been made in accordance with the Plan to the holder of such Disputed Claim had such Disputed Claim been an Allowed Claim on or prior to the Effective Date. Such distributions on account of disallowed Disputed Claims will be made as soon as practicable after the fifteenth business day following allowance or disallowance of the last Disputed Claim. Upon allowance or disallowance of all or a portion of such Disputed Claims, the Debtor, or the Liquidating Trustee, as the case may be, will make appropriate distributions in accordance with the Plan.

# ARTICLE IX.

# EXECUTORY CONTRACTS AND UNEXPIRED LEASES

9.1 Executory Contracts and Unexpired Leases. On the Effective Date, all executory contracts and unexpired leases to which any one or more Debtors is a party shall be deemed rejected as of the Effective Date, except for an executory contract or unexpired lease that (i) has been assumed or rejected pursuant to Final Order of the Bankruptcy Court prior to the Effective Date, (ii) is a collective bargaining agreement governed by section 1113 of the Bankruptcy Code or an agreement providing for retiree benefits covered by section 1114 of the Bankruptcy Code, or (iii) is subject to separate motion to assume or reject (or terminate or modify, as the case may be) filed under sections 365, 1113 and/or 1114 of the Bankruptcy Code by the Debtors prior to the Effective Date.

Notwithstanding the foregoing, any agreements, documents or instruments relating thereto (including Insurance Policies) that, are postpetition contracts shall continue to operate unaffected by the Plan, with the Insurers responsible for claims in accordance with the terms and provisions of such postpetition contracts (including Insurance Policies) unless otherwise terminated by the Liquidating Trustee. The Debtors' Insurance Policies that have expired as of the Confirmation Date (whether entered into prior or subsequent to the Petition Date) are not executory contracts subject to assumption or rejection. The Insurers shall be responsible for continuing coverage obligations under such Insurance Policies, regardless of the payment status of any retrospective or other insurance premiums.

Nothing contained in the Plan shall constitute or be deemed to be a waiver of any cause of action that the Debtors may hold against any Person, including, without limitation, any Insurer under any of the Debtors' Insurance Policies.

- **9.2** Approval of Rejection of Executory Contracts and Unexpired Leases. Entry of the Confirmation Order shall constitute the approval, pursuant to sections 365(a), 1113 and/or 1114 of the Bankruptcy Code, of the rejection of the executory contracts and unexpired leases rejected as of the Effective Date pursuant to the Plan.
- **9.3** Rejection Claims. In the event that the rejection of an executory contract or unexpired lease by the Debtors pursuant to the Plan results in damages to the other party or parties to such contract or lease, a Claim for such damages, if not heretofore evidenced by a filed proof of claim, shall be forever barred and shall not be enforceable against the Debtors or any property to be distributed under the Plan unless a proof of claim is filed with the Bankruptcy Court and served upon the Liquidating Trustee on or before the date that is thirty (30) days after the Confirmation Date. The foregoing sentence shall not, however, be applicable to any separate pre-Confirmation Date order of the Bankruptcy Court authorizing rejection of an executory contract or unexpired lease wherein a separate deadline by which rejection damages claims was established.

#### ARTICLE X.

#### EFFECTIVENESS OF THE PLAN

- **10.1** Condition Precedent to Confirmation of Plan. The following are conditions precedent to the confirmation of the Plan:
- (a) The Bankruptcy Court shall have entered a Confirmation Order in form and substance satisfactory to the Committee; and

- (b) The Bankruptcy Court enters the Confirmation Order on or before July 31, 2007.
- **10.2** <u>Conditions Precedent to Effective Date</u>. The following are conditions precedent to the Effective Date of the Plan:
  - (a) No stay of the Confirmation Order shall then be in effect; and
- (b) The Debtors shall have sufficient Cash to pay the sum of (i) Allowed Wachovia Secured Claim, Allowed Subordinated Participating Customers Secured Claims, Allowed Administrative Expense Claims, Allowed Priority Tax Claims, Allowed Other Secured Claims and Allowed Priority Non-Tax Claims, and the Professional fees that have accrued but not been paid (to the extent that any of the foregoing are not otherwise payable under the Final DIP Order by "Cash Infusions", as the term is defined in the Final DIP Order); and/or (ii) an amount that would be required to distribute to the holders of Disputed Subordinated Secured Participation Claims, Disputed Administrative Expense Claims, Disputed Priority Tax Claims, Disputed Other Secured Claims, and Disputed Priority Non-Tax Claims if all such Claims are subsequently Allowed, as set forth more fully in Article VII hereof.
- (c) Substantial Consummation will or is likely to occur on or before August 31, 2007.
- (d) Superior Fabrication Company, LLC will assume the Pension Plans as described in more detail at Section V.F. of the Disclosure Statement.
- 10.3 <u>Satisfaction of Conditions</u>. Any actions required to be taken on the Effective Date shall take place and shall be deemed to have occurred simultaneously, and no such action shall be deemed to have occurred prior to the taking of any other such action. If the Committee decides that one of the conditions precedent set forth in section 10.2 hereof cannot be satisfied and the occurrence of such condition is not waived or cannot be waived, then the Committee shall file a notice of the failure of the Effective Date with the Bankruptcy Court.
- 10.4 Effect of Nonoccurrence of Conditions to Effective Date. If each of the conditions to consummation and the occurrence of the Effective Date has not been satisfied or duly waived on or before August 31, 2007, the Confirmation Order may be vacated by the Bankruptcy Court. If the Confirmation Order is vacated pursuant to this section 10.4, the Plan shall be null and void in all respects, and nothing contained in the Plan shall constitute a waiver or release of any Claims against the Debtors.

#### ARTICLE XI.

# **EFFECT OF CONFIRMATION**

# 11.1 <u>Vesting of Assets</u>.

- (a) As of the Effective Date, all property of the Estates (and CEP Mexico) shall vest in the CEP Liquidating Trust, and shall constitute the Trust Assets.
- (b) As of the Effective Date, all assets of the Estates (and CEP Mexico) shall be free and clear of all Claims and Liens, except as provided in the Plan or the Confirmation Order.
- 11.2 <u>Release of Assets.</u> Until the Effective Date, the Bankruptcy Court shall retain jurisdiction of the Debtors and their assets and properties. Thereafter, jurisdiction of the Bankruptcy Court shall be limited to the subject matter set forth in Article XI hereof.

- 11.3 <u>Binding Effect</u>. Except as otherwise provided in section 1141(d)(3) of the Bankruptcy Code, on and after the Confirmation Date, the provisions of the Plan shall bind any holder of a Claim against, or Equity Interest in any one or more of the Debtors and their respective successors and assigns, including, but not limited to, the CEP Liquidating Trust and the Liquidating Trustee, whether or not the Claim or Equity Interest of such holder is impaired under the Plan and whether or not such holder has accepted the Plan.
- 11.4 <u>Term of Injunctions or Stays</u>. Unless otherwise expressly provided herein, all injunctions or stays arising under or entered during the Cases under sections 105 or 362 of the Bankruptcy Code, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect until the closing of the Cases.
- Date, the Liquidating Trustee will have the exclusive right to enforce any and all Causes of Action against any person. The Liquidating Trustee may prosecute, defend, enforce, abandon, settle or release any or all Causes of Action as it deems appropriate, without the need to obtain approval or any other or further relief from the Bankruptcy Court. The Liquidating Trustee may, in its sole discretion, offset any such claim held against a person, against any payment due such person under the Plan; *provided, however*, that any claims of the Debtors arising before the Petition Date shall first be offset against Claims against the Debtors arising before the Petition Date, subject in each instance, however, to the limitations of section 6.6 hereof. All defenses and rights of avoidance of the Debtors shall be retained and may be exercised by the Liquidating Trustee.
- 11.6 <u>Injunction</u>. On and after the Confirmation Date, all persons are permanently enjoined from commencing or continuing in any manner any action or proceeding (whether directly, indirectly, derivatively, or otherwise) on account of or respecting any claim, debt, right or cause of action of the Debtors for which the Liquidating Trustee retains sole and exclusive authority to pursue in accordance with the Plan.
- 11.7 <u>Injunction Against Interference with Plan.</u> Upon the entry of the Confirmation Order, all holders of Claims and Equity Interests and other parties in interest, along with their respective present or former employees, agents, officers, directors or principals, shall be enjoined from taking any actions to interfere with the implementation or consummation of the Plan.

# ARTICLE XII.

# **RETENTION OF JURISDICTION**

- **12.1 Jurisdiction of Bankruptcy Court.** The Bankruptcy Court shall retain jurisdiction of all matters arising under, arising out of, or related to the Cases and the Plan pursuant to, and for the purposes of, sections 105(a) and 1142 of the Bankruptcy Code and for, among other things, the following purposes:
  - (a) To hear and determine motions for the assumption, assumption and assignment or rejection of executory contracts or unexpired leases and the allowance of Claims resulting therefrom;
  - (b) To determine any motion, adversary proceeding, application, contested matter, and other litigated matter pending on or commenced after the Confirmation Date, including, without limitation, any proceeding to recover a Cause of Action;

- (c) To ensure that distributions to holders of Allowed Claims are accomplished as provided herein;
- (d) To consider Claims or the allowance, classification, priority, compromise, estimation or payment of any Claim;
- (e) To enter, implement, or enforce such orders as may be appropriate in the event the Confirmation Order is for any reason stayed, reversed, revoked, modified or vacated;
- (f) To issue injunctions, enter and implement other orders, and take such other actions as may be necessary or appropriate to restrain interference by any person with the consummation, implementation or enforcement of the Plan, the Confirmation Order or any other order of the Bankruptcy Court;
- (g) To hear and determine any application to modify the Plan in accordance with section 1127 of the Bankruptcy Code, to remedy any defect or omission or reconcile any inconsistency in the Plan, the Disclosure Statement or any order of the Bankruptcy Court, including the Confirmation Order, in such a manner as may be necessary to carry out the purposes and effects thereof;
- (h) To hear and determine all applications under sections 328, 330, 331 and 503(b) of the Bankruptcy Code for awards of compensation for services rendered and reimbursement of expenses incurred by Professionals prior to the Confirmation Date;
- (i) To hear and determine disputes arising in connection with the interpretation, implementation, or enforcement of the Plan, the Confirmation Order, any transactions or payments contemplated hereby, or any agreement, instrument, or other document governing or relating to any of the foregoing;
- (j) To hear and determine disputes arising in connection with the interpretation, implementation or enforcement of the Agreement, and to hear and determine all matters involving or relating to the CEP Liquidating Trust or the Liquidating Trustee.
- (k) To take any action and issue such orders as may be necessary to construe, enforce, implement, execute and consummate the Plan or to maintain the integrity of the Plan following consummation;
- (l) To recover all assets of the Debtors' property of the Estates and CEP Mexico, wherever located, which jurisdiction shall not be limited as a result of the transfer of such assets and property to the Liquidating Trust pursuant to the Plan;

- (m) To determine such other matters and for such other purposes as may be provided in the Confirmation Order;
- (n) To hear and determine matters concerning state, local, and federal taxes in accordance with sections 346, 505 and 1146 of the Bankruptcy Code (including, without limitation, matters with respect to any taxes payable by a trust or reserve established in furtherance of the Plan);
- (o) To hear and determine matters relating to or arising from CEP's 99.9% ownership interest in CEP Mexico to the extent permitted by applicable Law;
- (o) To hear and determine any other matters related hereto and not inconsistent with the Bankruptcy Code and title 28 of the United States Code; and
- (p) To enter a final decree closing the Cases.

#### ARTICLE XIII.

#### **CRAMDOWN RESERVATION**

13.1 <u>Nonconsensual Confirmation</u>. If any impaired class votes to accept the Plan by the requisite statutory majorities provided in sections 1126(c) and 1126(d) of the Bankruptcy Code, as applicable, or if any impaired class is deemed to have rejected the Plan, the Debtors and the Committee each respectively reserve the right to undertake to have the Bankruptcy Court confirm the Plan under section 1129(c) of the Bankruptcy Code and/or amend the Plan in accordance with section 14.9 hereof to the extent necessary to obtain entry of the Confirmation Order.

#### ARTICLE XIV.

# MISCELLANEOUS PROVISIONS

- 14.1 <u>Disposition of Creditors' Committee</u>. The Committee shall disband and be released of its duties and obligations on the Effective Date. Any rights or duties of the Committee under the Agreement shall be exercised by the co-chairs of the Committee as of the Confirmation Date, which Committee co-chairs shall continue to take all actions required of the Committee as of the Effective Date.
- **14.2** Record Date. At the close of business on the Confirmation Date, the Claims Register shall be closed.
- **14.3** <u>Substantial Consummation</u>. On the Effective Date or as soon thereafter as practicable, but in no event, later than August 31, 2007, the Plan shall be deemed to be "substantially consummated" as contemplated under sections 1101 and 1127(b) of the Bankruptcy Code.
- 14.4 <u>Exemption from Transfer Taxes</u>. Subject to orders entered by the Bankruptcy Court authorizing certain sales of real property in connection with any sale transaction, pursuant to section 1146(c) of the Bankruptcy Code, the assignment or surrender of any lease or sublease, or the delivery of any deed or other instrument of transfer under, in furtherance of, or in connection with the Plan, including any deeds, bills of sale or assignments executed in connection with any disposition of assets contemplated

by the Plan shall not be subject to any stamp, real estate transfer, mortgage recording, sales, use, or other similar tax.

- 14.5 <u>Exculpation</u>. None of the Debtors, Committee nor any of their respective members, officers, directors, employees, advisors, professionals or agents, (collectively, "Exculpation Parties") shall have or incur any liability to any holder of a Claim or Equity Interest for any act or omission in connection with, related to, or arising out of the Cases, including, without limitation, negotiations regarding or concerning the Plan, the pursuit of confirmation of the Plan, the consummation of the Plan, or the administration of the Estates or the Plan or the property to be distributed under the Plan, except for willful misconduct or gross negligence, and, in all respects, the Exculpation Parties shall be entitled to rely upon the advice of counsel with respect to their duties and responsibilities under the Plan.
- 14.6 <u>Discharge of Claims and Termination of Equity Interests</u>. Except as otherwise specifically provided herein or in the Confirmation Order, the rights afforded in the Plan and the payments and distributions to be made hereunder shall discharge all existing debts and Claims, and terminate all Equity Interests of any kind, nature or description whatsoever, other than Consolidated Holdings, against or in the Estates or any of their assets or properties to the fullest extent permitted by section 1141 of the Bankruptcy Code. Except as otherwise provided herein or in the Confirmation Order, upon the Effective Date, all existing Claims against the Estates shall be, and shall be deemed to be, discharged and terminated, and all holders of Claims and Equity Interests shall be precluded and enjoined from asserting against the Estates, or any of its assets or properties, any other or further Claim or Equity Interest based upon any act or omission, transaction or other activity of any kind or nature that occurred prior to the Effective Date, whether or not such holder has filed a proof of Claim or is listed in the Schedules.

# 14.7 Post-Confirmation Date Fees and Expenses.

- (a) Fees and Expenses of Professionals. The Liquidating Trustee shall, in the ordinary course of business and without the necessity for any approval by the Bankruptcy Court, pay the reasonable fees and expenses, incurred after the Effective Date, of the professional persons employed by the Liquidating Trustee in connection with the implementation and consummation of the Plan, the claims reconciliation process and any other matters as to which such professionals may be engaged. The fees and expenses of such professionals shall be paid within ten (10) Business Days after submission of a detailed invoice therefor. If the Liquidating Trustee disputes the reasonableness of any such invoice, the Liquidating Trustee shall timely pay the undisputed portion of such invoice, and the Liquidating Trustee or the affected professional may submit such dispute to the Bankruptcy Court for a determination of the reasonableness of such invoice.
- 14.8 <u>Payment of Statutory Fees.</u> On the Effective Date, and thereafter as may be required, the Liquidating Trustee shall pay all fees payable pursuant to section 1930 of chapter 123 of title 28 of the United States Code.
- 14.9 <u>Modification of Plan</u>. The Plan may be amended, modified, or supplemented by the Debtors and the Committee in the manner provided for by section 1127 of the Bankruptcy Code or as otherwise permitted by law without additional disclosure pursuant to section 1125 of the Bankruptcy Code, except as the Bankruptcy Court may otherwise direct. In addition, after the Confirmation Date, so long as such action does not materially adversely affect the treatment of holders of Claims or Equity Interests under the Plan, the Debtors and Committee may institute proceedings in the Bankruptcy Court to remedy any defect or omission or reconcile any inconsistencies in the Plan or the Confirmation Order, with respect to such matters as may be necessary to carry out the purposes and effects of the Plan. Prior to

the Effective Date, the Debtors and Committee may make appropriate technical adjustments and modifications to the Plan without further order or approval of the Bankruptcy Court, *provided* that such technical adjustments and modifications do not adversely affect in a material way the treatment of holders of Claims or Equity Interests.

- **14.10** Revocation or Withdrawal of Plan. The Committee reserves the right to revoke or withdraw the Plan at any time prior to the Confirmation Date. If the Committee takes such action, the Plan shall be deemed null and void. In such event, nothing contained herein shall be deemed to constitute a waiver or release of any Claim by or against the Debtors or any other person or to prejudice in any manner the rights of the Debtors or any other person in any further proceedings involving the Debtors.
- **14.11** Courts of Competent Jurisdiction. If the Bankruptcy Court abstains from exercising, or declines to exercise, jurisdiction or is otherwise without jurisdiction over any matter arising out of the Plan, such abstention, refusal, or failure of jurisdiction shall have no effect upon and shall not control, prohibit, or limit the exercise of jurisdiction by any other court having competent jurisdiction with respect to such matter.
- 14.12 <u>Severability</u>. If, prior to the entry of the Confirmation Order, any term or provision of the Plan is held by the Bankruptcy Court to be invalid, void or unenforceable, the Bankruptcy Court, at the request of the Debtors and Committee, shall have the power to alter and interpret such term or provision to make it valid or enforceable to the maximum extent practicable, consistent with the original purpose of the term or provision held to be invalid, void, or unenforceable, and such term or provision shall then be applicable as altered or interpreted. Notwithstanding any such holding, alteration or interpretation, the remainder of the terms and provisions of the Plan will remain in full force and effect and will in no way be affected, impaired, or invalidated by such holding, alteration or interpretation. The Confirmation Order shall constitute a judicial determination and shall provide that each term and provision of the Plan, as it may have been altered or interpreted in accordance with the foregoing, is valid and enforceable pursuant to its terms.
- **14.13** Governing Law. Except to the extent that the Bankruptcy Code or other federal law is applicable, or to the extent an exhibit hereto provides otherwise, the rights, duties and obligations arising under the Plan shall be governed by, and construed and enforced in accordance with, the laws of the State of Ohio, without giving effect to the principles of conflicts of law thereof.
- **14.14 Exhibits.** All exhibits, schedules, addendum or other others annexed to the Plan are deemed incorporated into and are a part of the Plan as if set forth in full herein.
- 14.15 <u>Successors and Assigns</u>. All the rights, benefits and obligations of any person named or referred to in the Plan shall be binding on, and shall inure to the benefit of, the heirs, executors, administrators, successors and/or assigns of such person.
- **14.16** <u>Time</u>. In computing any period of time prescribed or allowed by the Plan, unless otherwise set forth herein or determined by the Bankruptcy Court, the provisions of Bankruptcy Rule 9006 shall apply.
- 14.17 <u>Notices</u>. All notices, requests and demands to or upon the Debtors to be effective shall be in writing and, unless otherwise expressly provided herein, shall be deemed to have been duly given or made when actually delivered or, in the case of notice by facsimile transmission, when received and telephonically confirmed, addressed as follows:

If to Debtor: BAKER & HOSTETLER LLP

Joseph F. Hutchinson, Jr., Esquire

3200 National City Center

1900 East 9<sup>th</sup> Street Cleveland, OH 44114

Telephone: (216) 621-0200 Facsimile: (216) 696-0740

If to the Committee: MCGUIREWOODS LLP

Mark E. Freedlander, Esquire Dominion Tower, 23rd Floor

625 Liberty Avenue Pittsburgh, PA 15222

Telephone: (412) 667-6000 Facsimile: (412) 667-6050

# Respectfully submitted,

Dated: May 25, 2007

CEP Holdings, LLC, Creative Engineered Polymer Products, LLC and Thermoplastics Acquisition, LLC

By: /s/ Joseph Mallak
Name: Joseph Mallak Title: President and CEO

The Official Committee of Unsecured Creditors

By: <u>/s/ Bruce D. Tobiansky</u>

Name: Bruce D. Tobiansky

E.I. du Pont de Nemours and Company

Committee Co-Chair

By: <u>/s/ Doreen Bailey-Wilson</u> Name: Doreen Bailey-Wilson

Rhodia, Inc.

Committee Co-Chair

# **PLAN EXHIBIT 1**

**Trust Agreement** 

[THIS PAGE INTENTIONALLY LEFT BLANK]

# IN THE UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF OHIO EASTERN DIVISION

		X	Case Nos. 06-51848
In re		:	(Jointly Administered)
		:	
CEP Holdings, LLC, Cre	ative Engineered	•	Chapter 11
Polymer Products, LLC a	and Thermoplastics	•	Honorable Marilyn Shea-Stonum
Acquisition, LLC,		:	
•	Debtors.	:	
		X	

# TRUST AGREEMENT

by and among

CEP HOLDINGS, LLC; CREATIVE ENGINEERED POLYMER PRODUCTS, LLC and THERMOPLASTICS ACQUISITION, LLC as Debtors and Debtors-in-possession,

and

OFFICIAL COMMITTEE OF UNSECURED CREDITORS as representatives of Beneficiaries

and

SHAUN M. MARTIN as Liquidating Trustee

Dated: , 200
--------------

# TABLE OF CONTENTS

		Page
SECTION	I THE CEP LIQUIDATING TRUST	2
A.	Creation.	2
В.	Objective and Purpose.	2
C.	Acceptance	2
D.	Intention of Parties to Establish Grantor Trust.	2
E.	Transfer of Assets to CEP Liquidating Trust	3
F.	Termination of CEP Liquidating Trust	3
SECTION	II LIQUIDATING TRUSTEE	4
A.	Appointment.	4
B.	Generally.	4
C.	Scope of Authority.	5
D.	Powers	5
E.	Additional Powers	7
F.	Limitation of Liquidating Trustee's Authority.	8
G.	Liability of Liquidating Trustee	9
H.	Reliance by Liquidating Trustee	9
I.	Authorization to Expend Trust Assets.	9
J.	Compensation of the Liquidating Trustee	10
K.	Exculpation; Indemnification	10
L.	Confidentiality.	11
M.	Final Decree	11
N.	Termination.	11
SECTION	III BENEFICIARIES	12
A.	Identification of Beneficiaries	12
B.	Withholding.	12
C.	Tax Identification Numbers.	12
SECTION	IV DISTRIBUTIONS	12
A.	Disputed Claim Reserve.	12
B.	Application of Trust Assets.	12

# TABLE OF CONTENTS

(continued)

		Page
C.	Distribution	13
D.	Undeliverable, Unclaimed and Returned Distributions.	13
E.	Rounding	
F.	Setoffs	
G.	De minimis Distributions	15
Н.	Taxes	15
I.	Compliance with Laws	15
J.	Final Disbursements of CEP Liquidating Trust and Unclaimed Property After the Final Disbursement Date	15
K.	Estate Representatives	15
SECTION	V SUCCESSOR TRUSTEES	15
A.	Removal	15
B.	Resignation	15
C.	Appointment of Successor Trustee	16
D.	Trust Continuance.	16
E.	Turnover of Documents.	16
SECTION	VI REPORTING	17
A.	Tax and Other Reports	17
B.	Federal Income Tax.	17
C.	Other	18
SECTION	VII TRANSFER OF BENEFICIARY'S INTERESTS	18
A.	Transfer of Beneficial Interests.	18
SECTION	VIII MISCELLANEOUS PROVISIONS	18
A.	Amendment; Waiver	18
B.	Intention of Parties to Establish Grantor Trust	18
C.	Preservation of Privilege	18
D.	Cooperation	19
E.	Laws as to Construction	19
F.	Relationship to Plan.	19
G.	Severability	19

# TABLE OF CONTENTS (continued)

		Page
Н.	Disposition of the Committee.	19
I.	Professionals of the Liquidating Trustee.	19
J.	Retention of Jurisdiction; Venue.	19
K.	Notices	19
L.	Notices if to a Beneficiary	20
M.	Headings	20
N.	Counterparts	20

THIS AGREEMENT ("Agreement") is made this	day of _	, 2007,
by and among CEP Holdings, LLC, Creative Engineered	Polymer	Products, LLC and
Thermoplastics Acquisition, LLC, debtors and debtors-in-poss	session (co	llectively "Debtors"),
the Official Committee of Unsecured Creditors ("Committee")	and Shaur	n M. Martin (together
with any successors, the "Liquidating Trustee") under the Plan (a	as defined l	pelow).

# **RECITALS:**

A.	On September 20, 2006, each of the Debtors filed voluntary petitions for relief
under chapte	r 11 of title 11 of the United States Code (the "Bankruptcy Code") in the United
States Bankr	ruptcy Court for the Northern District of Ohio, Eastern Division ("Bankruptcy
Court"), whi	ch cases are pending before the Honorable Marilyn Shea-Stonum and are being
jointly admin	istered at Case No. 06-51848 under the caption In re CEP Holdings, LLC et al. (the
"Cases").	

- B. By order, dated [\_\_\_\_\_\_, 2007], the Bankruptcy Court confirmed the Joint Plan of Liquidation (as same may have been or may be amended or modified, the "Plan").
- C. Article 7 of the Plan calls for, <u>inter alia</u>, the creation of the CEP Liquidating Trust to hold and reduce to Cash all Trust Assets, including but not limited to the \$5,408,597 in principal, plus interest, relating to Composite Parts Mexico S.A. de C.V. ("CEP Mexico"), for the benefit of the Beneficiaries, to resolve and object to Claims, and to distribute proceeds of Trust Assets to Allowed Claims in accordance with the Plan.
- E. The CEP Liquidating Trust is created on behalf of, and for the benefit of the Beneficiaries.
- F. The CEP Liquidating Trust is created pursuant to, and to effectuate, the Plan for the primary purpose of liquidating the Trust Assets, including but not limited to the \$5,408,597 principal, plus interest, relating to CEP Mexico, for the benefit of the Beneficiaries as a liquidating trust, in accordance with Treasury Regulation § 301.7701-4(d), and Revenue Procedure 94-45 I.R.B. 1994-28, 124, (July 11, 1994) ("Revenue Procedure 94-45") (or any subsequent Revenue Procedures that may be issued), with no objective to continue or engage in the conduct of a trade or business except to the extent reasonably necessary to, and consistent with, the liquidating purpose of the CEP Liquidating Trust.
- G. The CEP Liquidating Trust provides that the Beneficiaries of the CEP Liquidating Trust will be treated as the grantors of the CEP Liquidating Trust and deemed owners of the Trust Assets. The CEP Liquidating Trust requires the Liquidating Trustee to file returns for the CEP Liquidating Trust as a grantor trust pursuant to Treasury Regulation § 1.671-1 through 4(a).
- H. The CEP Liquidating Trust is intended to qualify as a "grantor trust" for federal income tax purposes with the Beneficiaries treated as the grantors and owners of the trust.

<sup>&</sup>lt;sup>1</sup> Capitalized terms used herein without definition shall have the respective meanings assigned to such terms in the Plan.

- I. All of the CEP Liquidating Trust's income and/or recoveries are to be treated as subject to tax on a current basis to the Beneficiaries who will be responsible for payment of any tax due.
- J. Subject to Section I.F hereof, the CEP Liquidating Trust contains a fixed determinable termination date that is not more than five years from the date of creation of the CEP Liquidating Trust and that is reasonable based on all the facts and circumstances.
- K. The investment powers of the Liquidating Trustee other than those reasonably necessary to maintain the value of the Trust Assets and to further the liquidating purpose of the CEP Liquidating Trust, are limited to powers to invest in demand and time deposits, such as short-term certificates (or overnight swaps) of deposit, in banks or other savings institutions, or other temporary, liquid investments, such as Treasury bills.
- L. The CEP Liquidating Trust is required to distribute to the Beneficiaries its net income plus all net proceeds from the sale or other liquidation of assets, except that the CEP Liquidating Trust may retain an amount of net proceeds or net income reasonably necessary to maintain the value of the Trust Assets, to fund the Liquidation Expense Reserve Account and the Disputed Claim Reserve.

NOW, THEREFORE, in consideration of the promises and the mutual covenants and agreements contained herein and in the Plan, the Debtors, the Committee, and the Liquidating Trustee agree as follows:

# **SECTION I**

# THE CEP LIQUIDATING TRUST

- **A.** Creation. In accordance with section 7.1 of the Plan, on the Effective Date of the Plan, the CEP Liquidating Trust is hereby created, including but not limited to the \$5,408,597 principal, plus interest, relating to CEP Mexico.
- **B.** Objective and Purpose. The CEP Liquidating Trust shall be established and maintained for the sole purpose of liquidating and distributing the Trust Assets in accordance with Treasury Regulation 30.1.7701 4(d) and resolving and administering Claims, with no objection to continue or engage in the conduct of a trade or business.
- **C. Acceptance**. The Liquidating Trustee accepts the terms, provisions and conditions of this Agreement and the Plan and agrees to observe, perform and abide hereby and thereby.
- **D.** Intention of Parties to Establish Grantor Trust. This Agreement is intended to create a grantor trust for the United States federal income tax purposes, and to the extent provided by law, shall be governed and construed in all respects as such a grantor trust. The CEP Liquidating Trust is not intended to constitute nor shall the formation of the CEP Liquidating Trust be interpreted as creating an association, partnership or joint venture of any kind.

# E. Transfer of Assets to CEP Liquidating Trust.

- 1. On the Effective Date, right, title and interest to all assets of the Debtors and the Estates, including but not limited to the \$5,408,597 in principal, plus interest, relating to CEP Mexico, shall vest in the CEP Liquidating Trust as Trust Assets consistent with section 7.1(c) of the Plan.
- 2. On the Effective Date, the Estates shall be deemed to assign, set over, transfer and convey to the CEP Liquidating Trust all of their right, title and interest in assets of the Estates and CEP Mexico, including, but not limited to the \$5,408,597 in principal, plus interest, relating to CEP Mexico. To the extent that certain assets of the Estates (and CEP Mexico), because of their nature or because they will accrue subsequent to the Effective Date, cannot be assigned, set over, transferred or conveyed to the CEP Liquidating Trust on the Effective Date, such assets shall be deemed assigned, set over, transferred and conveyed to the CEP Liquidating Trust as soon as practical after the Effective Date.
- 3. On or after the Effective Date, the Liquidating Trustee shall continue as a plaintiff in all litigation or Causes of Action (on behalf of the Beneficiaries) in which the Debtors were plaintiffs prior to the Effective Date. All recoveries and proceeds arising from litigation and Causes of Action shall be deemed assigned, set over, transferred and conveyed to the CEP Liquidating Trust upon receipt thereof. All fees and costs of the Liquidating Trustee (and agents thereof) arising from or relating to pursuing litigation or Causes of Action or other services performed at the request of the Liquidating Trustee on behalf of the CEP Liquidating Trust shall be paid as a first priority in right of distribution from the CEP Liquidating Trust.
- 4. The transfer of assets of the Estates, including but not limited to the \$5,408,597 in principal, plus interest, relating to CEP Mexico, to the Liquidating Trust shall be made for the benefit of the Beneficiaries, in each case, but only to the extent the Beneficiaries are entitled to distributions under the Plan
- 5. For all purposes of the Internal Revenue Code of 1986 ("IRS Code"), as amended, all Persons (including, without limitation, the Debtors, the Liquidating Trustee and the Beneficiaries) shall treat the transfer of the Trust Assets to the CEP Liquidating Trust through the Debtors and/or agents of the Debtors as set forth in this Section I.E, as the deemed transfer of such assets to the Beneficiaries followed by the deemed transfer of such assets by the Beneficiaries to the CEP Liquidating Trust. Thus, the Beneficiaries shall be treated as the grantors and owners of the CEP Liquidating Trust for federal income tax purposes.
- 6. The Liquidating Trustee and the Beneficiaries will value the Trust Assets as set forth in section 7.1(o)(ii)(B) of the Plan and such valuations will be consistently used for all federal income tax purposes.

# F. Termination of CEP Liquidating Trust.

1. The CEP Liquidating Trust will terminate no later than the fifth (5th) anniversary of the Effective Date; <u>provided</u>, <u>however</u>, on or prior to the date six (6) months prior to such termination, the Bankruptcy Court, upon motion by a party in interest, may extend the term of the CEP Liquidating Trust for a finite period if it is necessary or appropriate in furthering

the liquidating purpose thereof. Multiple extensions can be obtained so long as Bankruptcy Court approval is obtained at least six (6) months prior to the expiration of each extended term; provided, however, that the Liquidating Trustee receives an opinion of counsel or a favorable ruling from the Internal Revenue Service ("IRS") that any further extension would not adversely affect the status of the trust as a liquidating trust and grantor trust for federal income tax purposes.

- Assets will be distributed to the Beneficiaries in accordance with the Plan and subject to this Agreement. If at any point prior to such termination date, the Liquidating Trustee determines that further disbursements of the Trust Assets are unwarranted, the Liquidating Trustee shall provide notice of the Liquidating Trustee's intention to make a final disbursement to the Beneficiaries. Absent the filing of a timely objection thereto, the Liquidating Trustee shall thereafter be authorized to and shall make the final disbursement of the Trust Assets to the Beneficiaries in accordance with their distribution rights under the Plan and this Agreement. Ninety (90) days after the final disbursement, the Liquidating Trustee shall stop payment on any check(s) remaining unpaid and such unclaimed funds shall thereafter be used to satisfy any outstanding expenses of administering the CEP Liquidating Trust and the remaining balance, if any, shall thereafter be paid to a charity to be selected by the Liquidating Trustee in accordance with section 7.3 of the Plan.
- 3. The Liquidating Trustee shall not unduly prolong the duration of the CEP Liquidating Trust and shall at all times endeavor to resolve, settle or otherwise dispose of all claims that constitute Trust Assets and to effect the distribution of the Trust Assets to the Beneficiaries in accordance with the terms hereof and terminate the CEP Liquidating Trust as soon as practicable.

# **SECTION II**

# LIQUIDATING TRUSTEE

- A. Appointment. The Committee has determined to appoint Shaun M. Martin, a director of Huron Consulting Group, to serve as the initial Liquidating Trustee under the Plan, and the Liquidating Trustee hereby accepts such appointment and agrees to serve in such capacity, upon the Effective Date of the Plan. A Successor Trustee (as defined in Section V.C hereof) may be appointed in the event that the Liquidating Trustee is removed or resigns pursuant to this Agreement or the Liquidating Trustee otherwise vacates the position.
- **B.** Generally. The Liquidating Trustee's powers are exercisable solely in a fiduciary capacity consistent with, and in furtherance of, the purposes of the CEP Liquidating Trust and not otherwise, except that the Liquidating Trustee may administer the Trust Assets as permitted by the provisions of Section II.D hereof. The Liquidating Trustee shall have the authority to bind the CEP Liquidating Trust, but shall for all purposes hereunder be acting in the capacity as Liquidating Trustee and not personally or individually. The Liquidating Trustee shall not be obligated to give any bond, surety or other security for the performance of any duties or obligations hereunder. The Liquidating Trustee shall be entitled to engage in all such activities as the Liquidating Trustee deems necessary and appropriate, in the Liquidating Trustee's sole

discretion, except where Committee approval may be required as set forth herein; <u>provided</u>, <u>however</u>, that such activities do not conflict with the purpose of the CEP Liquidating Trust and are in the best interests of the Beneficiaries; and <u>provided</u>, <u>further</u>, that the Liquidating Trustee shall devote such time and effort as is necessary to timely and properly fulfill all such duties.

Scope of Authority. The responsibilities and authority of the Liquidating Trustee shall include: (1) administering and liquidating the Trust Assets; (2) facilitating the prosecution or settlement of Causes of Action, including but not limited to Avoidance Actions; (3) facilitating the prosecution or settlement of objections to and estimations of Claims against the Debtors; (4) calculating and implementing all distributions in accordance with the Plan and this Agreement; (5) filing all required tax returns and paying taxes and all other obligations on behalf of the CEP Liquidating Trust from funds held by the CEP Liquidating Trust; (6) periodic reporting to the Bankruptcy Court, the Committee and parties in interest of the status of the Claims resolution process, distributions on account of Allowed Claims against the Debtors; and (7) such other responsibilities as may be vested in the Liquidating Trustee pursuant to the Plan, this Agreement, any Final Order of the Bankruptcy Court or as may be necessary and proper to carry out the provisions of the Plan.

#### D. Powers.

- 1. In connection with the administration of the CEP Liquidating Trust, except as otherwise set forth in this Agreement or the Plan, the Liquidating Trustee is authorized to perform any and all acts necessary and reasonable to accomplish the purposes of the CEP Liquidating Trust, including but not limited to acts relating to CEP Mexico Without limiting, but subject to the foregoing, and subject in all respects to the other terms and conditions of this Agreement, the Liquidating Trustee shall be expressly authorized, but shall not be required, to:
  - (a) pursue and prosecute Causes of Action or to refrain from pursuing any Cause of Action, based upon the Liquidating Trustee's assessment of the net benefit expected to be received by the CEP Liquidating Trust in connection therewith (taking into account the costs and expenses projected to be incurred in connection therewith, the likelihood of success on the merits, and the range of potential recoveries to be received by the CEP Liquidating Trust);
  - (b) compromise, settle, resolve and collect on any Causes of Action upon such terms and conditions as the Liquidating Trustee deems appropriate and in the best interests of the CEP Liquidating Trust without further approval of the Bankruptcy Court. All compromises and settlements shall be deemed final and binding upon all parties in interest in the applicable Cases. The Liquidating Trustee shall have the right, power and authority to retain and assert all defenses, rights to setoff, recoupment, avoidance rights and counterclaims with respect to and Cause of Action;
  - (c) review, analyze and, as appropriate, prosecute objections to Claims and any claims arising after the Petition Date ("<u>Post-Petition Date Claims</u>"), including any Administrative Expense Claims;

- (d) compromise, settle and resolve any Disputed Claims upon such terms and conditions as the Liquidating Trustee deems appropriate and in the best interests of the CEP Liquidating Trust without further approval of the Bankruptcy Court. All compromises and settlements shall be deemed final and binding upon all parties in interest in the Cases;
- (e) review, litigate and settle (without further approval of the Bankruptcy Court) or otherwise resolve Claims and Post-Petition Date Claims (including any Administrative Expense Claims), Causes of Action or other disputes involving or arising out of alleged environmental contamination and/or remediation;
- (f) maintain all accounts, make distributions to Beneficiaries of the CEP Liquidating Trust from the Trust Assets and take other actions consistent with the Plan in the name of the CEP Liquidating Trust, provided, however, that the Liquidating Trustee shall not be entitled to use the Debtors' bank accounts that are in existence as of the Effective Date and shall be required to open new bank or other depository accounts in the name of the CEP Liquidating Trust as may be necessary or appropriate in the discretion of the Liquidating Trustee to enable the Liquidating Trustee to administer the CEP Liquidating Trust in accordance with the terms of this Agreement;
- (g) the power to invest funds in and withdraw, make distributions and pay expenses, taxes, and other obligations owed by the CEP Liquidating Trust from funds held by the Liquidating Trustee in accordance with the Plan and this Agreement;
- (h) maintain, in respect of the CEP Liquidating Trust and the Beneficiaries, books and records relating to the assets and income of the CEP Liquidating Trust and the payment of expenses of, and liabilities of, claims against or assumed by, the CEP Liquidating Trust in such detail and for such period of time as may be necessary to enable him to make full and proper accounting in respect thereof in accordance with this and to comply with Except as provided herein, nothing in this applicable provisions of law. Agreement requires the CEP Liquidating Trust or the Liquidating Trustee to file any accounting or seek approval of any court with respect to the administration of the CEP Liquidating Trust, or as a condition for making any payment or distribution out of the Trust Assets. Beneficiaries shall have the right upon thirty (30) days' prior written notice delivered to the Liquidating Trustee to inspect such books and records, provided that, if so requested, such Beneficiary shall have entered into a confidentiality agreement satisfactory in form and substance to the Liquidating Trustee.;
- (i) retain or engage, without the necessity of obtaining any approval from the Bankruptcy Court, such employees, independent contractors, professional persons, and agents as are appropriate, necessary or desirable to complete (i) disbursements to Beneficiaries of the CEP Liquidating Trust and (ii)

the general administration of the CEP Liquidating Trust as required by this Agreement and applicable law;

- (j) incur any reasonable and necessary expenses on behalf of the CEP Liquidating Trust in carrying out the Liquidating Trustee's obligations and responsibilities on behalf of the CEP Liquidating Trust and make ordinary and reasonable disbursements from Trust Assets to pay the ordinary and necessary expenses of administering the CEP Liquidating Trust, without the necessity of providing any notice or seeking or obtaining any approval of the Bankruptcy Court with respect to such disbursements;
- (k) make interim and final disbursements of the Trust Assets to the Beneficiaries of the CEP Liquidating Trust in accordance with the terms of this Agreement, the Plan and applicable law;
- (l) execute, deliver, file and/or record such contracts, instruments, releases, indentures, and other agreements or documents, and to take such actions, as may be necessary, desirable or appropriate to administer the CEP Liquidating Trust;
- (m) prepare and file tax and informational returns on behalf of the CEP Liquidating Trust as required by applicable federal, state and local law, and in accordance with the terms of this Agreement;
- (n) file with the Bankruptcy Court and/or the Office of the United States Trustee the reports and other documents required by the Plan or otherwise required to close the Cases;
- (o) take any and all actions required in connection with the Pension Plans, including, but not limited to, their standard termination or distressed termination, as necessary;
- (p) notwithstanding any other provisions of the Plan or this Agreement, to pay any and all fees of the Office of the United States Trustee as and when such fees become due, which payments shall be paid by the Liquidating Trustee from the Trust Assets; and
- (q) take all other actions not inconsistent with the provisions of the Plan or the Agreement that the Liquidating Trustee deems reasonably necessary or desirable in connection with the administration of the CEP Liquidating Trust.
- **E.** Additional Powers. Except as otherwise set forth in this Agreement or in the Plan, and subject to the retained jurisdiction of the Bankruptcy Court as provided for in the Plan, but without prior or further authorization, the Liquidating Trustee may control and exercise authority over the Trust Assets, including without limitation the \$5,408,597 in principal, plus interest, relating to CEP Mexico, and over the protection, conservation and disposition thereof. No person dealing with the CEP Liquidating Trust shall be obligated to inquire into the authority of the Liquidating Trustee in connection with the protection, conservation or disposition of the

Trust Assets. It is intended that a signed copy of this Agreement serve as adequate proof of the Liquidating Trustee's authority to act if such proof is required for any reason by any third party.

# F. Limitation of Liquidating Trustee's Authority.

- 1. <u>No Trade or Business</u>. The Liquidating Trustee shall not, and shall not be authorized to, engage in any trade or business with respect to the Trust Assets or any proceeds therefrom except to the extent reasonably necessary or appropriate to, and consistent with, the liquidating purpose of the CEP Liquidating Trust and the Liquidating Trustee shall take such actions consistent with the prompt orderly liquidation of the Trust Assets as are required by applicable law and consistent with the treatment of the CEP Liquidating Trust as a CEP Liquidating Trust under Treasury Regulation § 301.7701-4(d) and Revenue Procedure 94-45 (or any subsequent Revenue Procedures that may be issued), and such actions permitted herein.
- 2. <u>Released Claims</u>. The Liquidating Trustee shall not have any authority to pursue any Claims and/or Causes of Action waived, exculpated or released in accordance with the provisions of the Plan or Final Order of the Bankruptcy Court.
- Investment and Safekeeping of the Trust Assets. All Trust Assets received 3. by the CEP Liquidating Trust shall, until distributed or paid over as herein provided, be held in trust for the benefit of the Beneficiaries, but need not be segregated unless and to the extent required by applicable law or by the Plan. The Liquidating Trustee shall be under no liability for interest or producing income on any Trust Assets received by the CEP Liquidating Trust hereunder and held for distribution or payment to the Beneficiaries. Investments of any Trust Assets held by the CEP Liquidating Trust shall be administered in a manner in which individuals of ordinary prudence, discretion and judgment would act in the management of their own affairs; provided, however, that the right and power of the Liquidating Trustee to invest the Trust Assets, the proceeds thereof, or any income earned by the CEP Liquidating Trust, shall be limited to the right and power to invest such assets (pending periodic distributions in accordance with this Section II.F.3 hereof) in demand and time deposits, such as short-term certificates of deposit, in banks or other savings institutions, or other temporary liquid investments, such as Treasury bills; and, provided, further, that the scope of any such permissible investments shall be limited to include only those investments that a CEP Liquidating Trust, within the meaning of Treasury Regulation § 301.7701-4(d) and Revenue Procedure 94-45 (or any subsequent Revenue Procedures that may be issued), may be permitted to hold, pursuant to the Treasury Regulations, or any modification in the IRS guidelines, whether set forth in IRS rulings, other IRS pronouncements or otherwise.
- 4. <u>Limiting Transfers</u>. The Liquidating Trustee shall not take, or cause the CEP Liquidating Trust to take, any action that would cause the interests in the CEP Liquidating Trust to be considered readily tradable on a secondary market (or a substantial equivalent thereof) within the meaning of Section 7704(b)(2) of the IRS Code, and Treasury Regulations § 1.7704-1(c), and the Liquidating Trustee shall not permit any transfer of an interest in the CEP Liquidating Trust if it would cause the CEP Liquidating Trust (were it be classified as a partnership rather than a grantor trust) to be treated as a "publicly traded partnership" as defined in IRS Code § 7704.

- Liability of Liquidating Trustee. In no event shall the Liquidating Trustee, the Liquidating Trustee's officers, employees, independent contractors or any of the Liquidating Trustee's professionals, agents or representatives (or their designees) be held personally liable for any claim asserted against the CEP Liquidating Trust, the Liquidating Trustee, the Liquidating Trustee's officers, employees, independent contractors and/or any of the Liquidating Trustee's professionals, agents or representatives (or their designees), except to the extent occasioned by or based their upon their own fraud, willful misconduct or gross negligence. Specifically, the Liquidating Trustee, the Liquidating Trustee's officers, employees, independent contractors or any of the Liquidating Trustee's professionals, agents or representatives (or their designees) shall not be liable for any negligence or any error of judgment in either case made in good faith, in the exercise of its or their business judgment or with respect to any action taken or omitted to be taken in good faith, except to the extent that the action taken or omitted to be taken by the Liquidating Trustee, the Liquidating Trustee's officers, employees, independent contractors or any of the Liquidating Trustee's professionals, agents or representatives (or their designees) is determined by a Final Order of a court of competent jurisdiction to be due to their own respective fraud, gross negligence or willful misconduct.
- **H. Reliance by Liquidating Trustee**. Except as otherwise provided in Section II.F hereof:
- 1. The Liquidating Trustee may rely, and shall be protected in acting upon, any resolution, certificate, statement, instrument, opinion, report, notice, request, consent, order or other paper or document believed by the Liquidating Trustee to be genuine and to have been signed or presented by the proper party or parties;
- 2. The Liquidating Trustee may consult with legal counsel, financial or accounting advisors and other professionals, and the Liquidating Trustee shall not be liable for any action taken or omitted to be taken by him in accordance with the advice thereof; and
- 3. All persons or entities hired by, retained by, doing business with or otherwise dealing with the Liquidating Trustee shall look only to the Trust Assets to satisfy any liability incurred by the Liquidating Trustee on behalf of the CEP Liquidating Trust to such persons or entities in carrying out the terms of this Agreement, and the Liquidating Trustee shall have no personal obligation to satisfy any such liability, except to the extent such liability or obligation arises as a result of the fraud, gross negligence or willful misconduct of the Liquidating Trustee as determined in accordance with Section II.G of this Agreement, in which case the Trust Assets shall not be subject to such claims or liabilities.
- I. Authorization to Expend Trust Assets. The Liquidating Trustee may expend the Trust Assets (i) to pay expenses of administration of the CEP Liquidating Trust (including, but not limited to, the fees and expenses of the Liquidating Trustee, its officers, employees, and independent contractors and any of its professionals, agents or representatives, any taxes imposed on the CEP Liquidating Trust or in respect of the Trust Assets, and any fees and expenses incurred in connection with any Causes of Action), and (ii) to satisfy other liabilities incurred or assumed by the CEP Liquidating Trust (or to which the assets are otherwise subject) in accordance with this Agreement or the Plan.

# J. Compensation of the Liquidating Trustee.

- 1. The costs and expenses of the CEP Liquidating Trust, including the fees and expenses of the Liquidating Trustee and its retained professionals, shall be paid out of the CEP Liquidating Trust. Such costs and expenses shall be treated as administrative expenses of the Estates, having, however, a first priority in right of distribution, and without the need for Bankruptcy Court authorization or entry of a Final Order. The Liquidating Trustee shall retain such amounts as are reasonably necessary (at the discretion of the Liquidating Trustee) to meet the future fees and expenses of interest in administering the CEP Liquidating Trust in the Liquidation Expense Reserve Account.
- 2. The Liquidating Trustee shall be entitled to reasonable compensation, which shall be on economic terms agreeable to the Committee and established prior to the Effective Date. The compensation payable to the Liquidating Trustee shall be treated consistent with this section. The Liquidating Trustee, and the officers, employees and independent contractors of the Liquidating Trustee shall be entitled to receive compensation at their standard hourly rates, which rates currently range from \$200 per hour to \$450 per hour and which rates are subject to periodic adjustments, for services rendered on behalf of the CEP Liquidating Trust.
- 3. All compensation and other amounts payable to the Liquidating Trustee shall be paid from the Trust Assets. If the cash in the CEP Liquidating Trust shall be insufficient to compensate and reimburse the Liquidating Trustee, as the case may be, for any amounts to which he is entitled hereunder, then the Liquidating Trustee is hereby authorized to reduce to cash in a commercially reasonable manner that portion of the Trust Assets necessary so as to effect such compensation and reimbursement.
- K. **Exculpation**; Indemnification. From and after the Effective Date, the Liquidating Trustee, the Liquidating Trustee's officers, employees, independent contractors, or any of the Liquidating Trustee's professionals, agents or representatives (or their designees) shall be and hereby are exculpated by all persons, and entities including, without limitation, holders of Claims against the Debtors and other parties in interest, from any and all claims, causes of action and other assertions of liability arising out of the discharge of the powers and duties conferred upon the Liquidating Trustee by this Agreement, the Plan or any order of the Bankruptcy Court entered pursuant to or in furtherance of the Plan or this Agreement, or applicable law or otherwise, except only for actions or omissions to act only to the extent determined by a Final Order of a court of competent jurisdiction to be due to the fraud, gross negligence or willful misconduct of the Liquidating Trustee, the Liquidating Trustee's officers, employees, independent contractors, or any of the Liquidating Trustee's professionals, agents or representatives (or their designees) after the Effective Date. No holder of a Claim against the Debtors or any other party in interest will have or be permitted to pursue any claim or cause of action against the Liquidating Trustee, the CEP Liquidating Trust or the Liquidating Trustee's officers, employees, independent contractors, professionals, agents or representatives (or their designees) for making payments in accordance with the Plan or this Agreement or for implementing the provisions of the Plan or this Agreement except in cases of fraud, gross negligence or willful misconduct. The CEP Liquidating Trust shall indemnify, defend and hold harmless the Liquidating Trustee, the Liquidating Trustee's officers, employees, independent contractors and any of the Liquidating Trustee's professionals, agents or representatives (or their

designees) from and against any and all claims, causes of action, liabilities, obligations, losses, damages or expenses (including attorneys' fees) (except to the extent determined by a Final Order of a court of competent jurisdiction to arise out of the fraud, gross negligence or willful misconduct of the Liquidating Trustee, or any of the Liquidating Trustee's officers, employees, independent contractors, or any of the Liquidating Trustee's professionals, agents or representatives (or their designees) but this parenthetical shall only apply to those individuals or entities to which such finding or findings of fraud, gross negligence or willful misconduct shall have been made) to the fullest extent permitted by applicable law. Any action taken or omitted to be taken with the approval of the Bankruptcy Court will conclusively be deemed not to constitute fraud, gross negligence or willful misconduct.

- L. Confidentiality. The Liquidating Trustee shall, and shall cause the Liquidating Trustee's officers, employees, independent contractors or any of the Liquidating Trustee's professionals, agents or representatives (or their designees) to, during the period that the Liquidating Trustee serves as Liquidating Trustee under this Agreement hold strictly confidential and not use for personal gain any material, non-public information of or pertaining to the Debtors or any Beneficiary or any entity to which any of the Trust Assets relate or of which the Liquidating Trustee has become aware in its capacity as such, other than in connection with the Liquidating Trustee carrying out the objectives and purposes of the CEP Liquidating Trust.
- **M. Final Decree.** It shall be the duty of the Liquidating Trustee to seek and obtain a final decree or decrees from the Bankruptcy Court in the Cases upon full administration of the CEP Liquidating Trust, and with respect to the other Debtors as soon as reasonably practicable after the Effective Date of the Plan.
- **N. Termination.** The duties, responsibilities and powers of the Liquidating Trustee will terminate on the date the CEP Liquidating Trust is dissolved under applicable law in accordance with the Plan and this Agreement, or by an Order of the Bankruptcy Court or by entry of a final decree closing the Cases, whichever date is later

#### SECTION III

# **BENEFICIARIES**

- A. Identification of Beneficiaries. To determine the actual names, addresses and tax identification numbers of the Beneficiaries, the Liquidating Trustee shall be entitled to conclusively rely on the names, addresses and tax identification numbers in the records maintained by BMC. If a Beneficiary has not filed a proof of claim or request for Administrative Expense Claim, or provided written notice indicating such information, the Liquidating Trustee shall be entitled to conclusively rely on the names, addresses and tax identification numbers reflected in the applicable Schedules or, if more recent, contained in the Debtors' records. Each Beneficiary's right to distribution from the CEP Liquidating Trust shall be that accorded to such Beneficiary under the Plan. Each distribution by the Liquidating Trustee to the Beneficiaries shall be made in accordance with the Plan and terms set forth herein.
- **B.** Withholding. The Liquidating Trustee shall withhold from the amounts distributable to the Beneficiaries from the Trust Assets at any time such sum or sums as may be required to be withheld under the applicable income tax laws of the United States or of any state or political subdivision thereof. Notwithstanding any other provision of this Agreement or the Plan, each Beneficiary receiving a distribution or payment from the CEP Liquidating Trust shall have sole and exclusive responsibility for the satisfaction and payment of any tax obligations imposed by any governmental unit on account of such distribution or payment.
- **C.** Tax Identification Numbers. The Liquidating Trustee shall require any Beneficiary to furnish to the Liquidating Trustee its tax identification number as assigned by the IRS and the Liquidating Trustee may condition any distribution to any Beneficiary upon receipt of such tax identification number.

#### **SECTION IV**

#### **DISTRIBUTIONS**

- A. Disputed Claim Reserve. The Liquidating Trustee shall maintain, in accordance with the Liquidating Trustee's powers and responsibilities under the Plan and this Agreement, a Disputed Claim Reserve; provided, however, that the Liquidating Trustee may not retain cash or cash equivalents in excess of a reasonable amount to meet any claims or contingent liabilities (including the Disputed Claims), to maintain the value of the Trust Assets during liquidation or to pay unpaid and estimated or anticipated future expenses of the CEP Liquidating Trust. Such amounts (net of any expenses, including any taxes, of the escrow relating thereto) shall be distributed, as provided herein and in the Plan, after such Disputed Claims are resolved.
- **B.** Application of Trust Assets. The Liquidating Trustee on behalf of the CEP Liquidating Trust shall apply all other cash constituting Trust Assets and any proceeds therefrom in the order and reflecting the priorities set forth below:

FIRST, to pay all the costs and expenses of the CEP Liquidating Trust including, without limitation, funding of the Liquidation Expense Reserve Account for the

post-confirmation fees and expenses and any and all costs, expenses and liabilities incurred by the Liquidating Trustee in connection with the performance of his duties under this Agreement, including the payment of any professionals, agents or other representatives (and designees) retained by the Liquidating Trustee.

**SECOND**, to the holders of Allowed Claims, classified and unclassified under the Plan in accordance with the terms and conditions of the Plan.

Notwithstanding anything to the contrary in this Section IV, prior to making any distribution to Class 4 Claims under the Plan, the Liquidating Trustee may retain such amounts (i) to pay estimated expenses of administration (including, but not limited to, the fees and expenses of the Liquidating Trustee, any taxes imposed on the CEP Liquidating Trust or in respect of the assets of the CEP Liquidating Trust, and fees and expenses in connection with litigation), (ii) to satisfy other liabilities incurred or assumed by the CEP Liquidating Trust (or to which the Trust Assets are otherwise subject), all for the term of the CEP Liquidating Trust and in accordance with this Agreement or the Plan, and (iii) to satisfy the post-confirmation fees and expenses detailed in the Plan; provided, however, that, from the net amount distributable, the Liquidating Trustee may reserve, in accordance with the provisions of Sections IV.D and IV.E hereof, such amounts necessary for the Disputed Claim Reserve.

The CEP Liquidating Trust hereby grants to the Liquidating Trustee a first-priority lien on and security interest in the Trust Assets to secure the payment of all amounts owed to, accrued or reserved on account of, the Liquidating Trustee or any persons or entities, including, but not limited to, professionals, to be retained by the Liquidating Trustee hereunder or otherwise due hereunder. The CEP Liquidating Trust agrees to take such actions and execute such documents as the Liquidating Trustee deems appropriate to perfect the Liquidating Trustee's lien and security interest hereunder. The Liquidating Trustee is authorized to execute and deliver all documents on behalf of the CEP Liquidating Trust and the Liquidating Trustee to accomplish the purposes of this Agreement and the Plan.

C. Distribution. The CEP Liquidating Trust shall distribute at least annually and in accordance with the Plan and the Agreement to the Beneficiaries all net cash recoveries plus all net cash proceeds from the liquidation of the Trust Assets (including as cash for this purpose, all cash equivalents) in accordance with this Section, except that the CEP Liquidating Trust may retain an amount of net proceeds or net income reasonably necessary to maintain the Liquidation Expense Reserve Account and the Disputed Claim Reserve. Notwithstanding any other provision of this Agreement or the Plan to the contrary, distributions or payments may be deferred or delayed in the discretion of the Liquidating Trustee for a reasonable time in the event that such deferral is necessary to permit investments to reach maturity, in the event that additional time is needed to make a proper distribution or payment, or in the event that the receipt of additional funds is necessary to make meaningful payments.

- **D.** Undeliverable, Unclaimed and Returned Distributions.
- 1. If any distribution or payment to a Beneficiary is returned to the CEP Liquidating Trust as undeliverable, no further distributions or payments will be made to such Beneficiary, and such Beneficiary will not be entitled to any further distributions or payments, unless and until the Liquidating Trustee is notified in writing of such Beneficiary's correct address. If any check issued to a Beneficiary in respect of a distribution or payment from the CEP Liquidating Trust remains outstanding for more than ninety (90) days following the date of such check, no further distributions or payments will be made to such Beneficiary, and such Beneficiary will not be entitled to any further distributions or payments. Nothing contained in this Agreement or the Plan shall require the Debtors, the Liquidating Trustee, or any of their respective agents, employees, attorneys or professionals to attempt to locate any Beneficiary.
- 2. Notwithstanding anything in this Agreement or the Plan to the contrary, any Beneficiary of an undeliverable distribution or payment who does not assert an entitlement to such distribution or payment before ninety (90) days after the later of (i) the Effective Date and (ii) the date of mailing of the undelivered Distribution or payment shall have its remaining unpaid Claim discharged and shall be forever barred from asserting any such Claim against the CEP Liquidating Trust, the Debtors or the Estates. Any unclaimed Cash held by the CEP Liquidating Trust on account of such unclaimed, undeliverable distributions or payments and any investment yields thereon, as of the expiration of the time limit set forth in this paragraph shall become the property of the CEP Liquidating Trust.
- 3. **No Interest on Claims.** Except as set forth in a Final Order of the Bankruptcy Court, no Beneficiary shall be entitled to interest accruing on or after the Petition Date on a Claim, however, funds held by the Liquidating Trustee for the benefit of Beneficiaries will accrue interest and such interest will inure to the benefit of the Beneficiaries. Except as otherwise provided in a Final Order of the Bankruptcy Court, interest shall not accrue or be paid on any Disputed Claim with respect to the period from the Petition Date to the date a final distribution or payment is made thereon if and after any such Disputed Claim, or any part thereof, becomes an Allowed Claim, provided, however, that Reserve Income will inure to the benefit of holders of Disputed Claims that become Allowed Claims.
- **E. Rounding.** Whenever any payment of a fraction of a dollar would otherwise be called for, the actual payment shall reflect a rounding of such fraction down to the nearest whole dollar.
- F. Setoffs. Pursuant to section 553 of the Bankruptcy Code and applicable non-bankruptcy law, the Liquidating Trustee may set off or recoup against any Allowed Claim and the distributions to be made thereon pursuant to this Agreement and the Plan, the claims, rights and Causes of Action of any nature that the Estates, the Debtors or the CEP Liquidating Trust may hold against such Beneficiary prior to any distribution on account of such Allowed Claim. The Liquidating Trustee's election not to exercise such setoff or recoupment rights, pursuant to this provision, shall not constitute a waiver of or in any way affect such claims, rights and Causes of Action which the Estates, the Debtors or the Liquidating Trustee may possess against such Beneficiary. To the extent that the amount of the Estates', the Debtors' or the Liquidating Trustee's claim, right or Cause of Action exceeds the amount of the Allowed Claim

or distribution against which it is set off, the Liquidating Trustee shall reserve their right to recover the full amount of such excess from such Beneficiary.

- **G. De minimis Distributions.** If the amount due any Beneficiary would result in a distribution or payment in an amount less than fifty dollars (\$50.00), such Beneficiary shall not receive such distribution or payment.
- H. Taxes. The Liquidating Trustee will comply with all tax withholding and reporting requirements imposed by all governmental entities, and all distributions or payments pursuant to this Agreement and the Plan will, to the extent applicable, be subject to such withholding and reporting requirements. The Liquidating Trustee shall be authorized to take any and all actions that may be necessary or appropriate to comply with such withholding and reporting requirements. Notwithstanding any other provision of this Agreement or the Plan, each entity receiving a distribution or payment pursuant to this Agreement or the Plan shall have sole and exclusive responsibility for the satisfaction and payment of any tax obligations imposed by any governmental unit, including income, withholding and other tax obligations, on account of such distribution or payment. Pending the implementation of satisfactory arrangements and/or mechanisms to comply with the requirements of this Section, any distribution or payment to be made pursuant to the Plan shall be treated as undeliverable.
- I. Compliance with Laws. Any and all distributions or payments of Trust Assets shall be in compliance with applicable laws, including, but not limited to, applicable federal and state securities laws.
- J. Final Disbursements of CEP Liquidating Trust and Unclaimed Property After the Final Disbursement Date. The Liquidating Trustee is authorized to terminate the CEP Liquidating Trust in accordance with the terms of this Agreement and the Plan when the Liquidating Trustee determines that no further disbursements of the respective assets of the CEP Liquidating Trust are possible, among other conditions.
- **K.** Estate Representatives. Pursuant to section 1123(b)(3)(B) of the Bankruptcy Code, the Liquidating Trustee shall serve as the representative of the Estates, CEP Mexico and the CEP Liquidating Trust on and following the Effective Date.

# **SECTION V**

#### **SUCCESSOR TRUSTEES**

- A. Removal. The Liquidating Trustee may only be removed by order of the Bankruptcy Court, for cause, including: (a) fraud, gross negligence or willful misconduct in connection with the affairs of the CEP Liquidating Trust; (b) physical or mental disability that substantially prevents the Liquidating Trustee from performing the duties as Liquidating Trustee hereunder; (c) breach of fiduciary duty; or (d) failure, in good faith judgment of the Committee, to reasonably perform the duties as Liquidating Trustee hereunder.
- **B.** Resignation. The Liquidating Trustee may resign by giving not less than sixty (60) days' prior written notice thereof to the Committee and the Bankruptcy Court. Such resignation shall become effective on the later to occur of (i) the date specified in such notice and

- (ii) the selection of a Successor Trustee (defined herein) and the acceptance by such Successor Trustee of such appointment, unless otherwise ordered by the Bankruptcy Court.
- C. Appointment of Successor Trustee. In the event of the death, resignation or removal of the Liquidating Trustee, the Committee shall appoint a successor to the Liquidating Trustee ("Successor Trustee"). Notice of any Successor Trustee shall be filed with the Bankruptcy Court and provided to the Beneficiaries. Any Successor Trustee appointed hereunder shall execute an instrument accepting such appointment hereunder and shall file such acceptance with the CEP Liquidating Trust's records. Thereupon, such Successor Trustee shall, without any further act, become vested with all the Trust Assets, properties, rights, powers, trusts and duties of his or her or its predecessor in the CEP Liquidating Trust with like effect as if originally named herein; provided, however, that a removed or resigning trustee shall, nevertheless, when requested in writing by the Successor Trustee, execute and deliver an instrument or instruments conveying and transferring to such Successor Trustee under the CEP Liquidating Trust all of the rights in connection with the Trust Assets, properties, rights, powers, trusts and duties of such predecessor trustee.
- **D.** Trust Continuance. The death, resignation or removal of the Liquidating Trustee shall not operate to terminate the CEP Liquidating Trust created by this Agreement or revoke any existing agency (other than the agency of such Liquidating Trustee as Liquidating Trustee) created pursuant to the terms of this Agreement or invalidate any action therefore taken by the Liquidating Trustee. The Liquidating Trustee agrees that the provisions of this Agreement shall be binding on or inure to the benefit of any Successor Trustee.
- E. Turnover of Documents. Upon the resignation or removal of the Liquidating Trustee, the Liquidating Trustee shall promptly: (a) execute and deliver, by the effective date of resignation or removal, all such documents, instruments, and other writings as may be required to effect the termination of the Liquidating Trustee's capacity under this Agreement and conveyance of the Trust Assets there held by the Liquidating Trustee to the Successor Trustee; (b) deliver to the Successor Trustee all documents, instruments, books, records and other writings relating to the CEP Liquidating Trust as may be in the possession or under control of the Liquidating Trustee; and (c) otherwise assist and cooperate in effecting the assumption of the rights, powers, duties and obligations under this Agreement by the Successor Trustee.

#### SECTION VI

# REPORTING

A. Tax and Other Reports. As soon as practicable after the end of each calendar year, and as soon as practicable upon termination of the CEP Liquidating Trust, the Liquidating Trustee shall submit to the Bankruptcy Court a written report including: (i) financial statements of the CEP Liquidating Trust at the end of such calendar year or period and the receipts and disbursements of the CEP Liquidating Trust for such period; (ii) a description of any action taken by the Liquidating Trustee in the performance of his duties, which materially and adversely affects the CEP Liquidating Trust and of which notice has not previously been given to the Beneficiaries; and (iii) subject to Section VI.B hereof, a separate statement for each Beneficiary setting forth the holder's share of items of income, gain, loss, deduction or credit and instructing all such holders to report such items on their federal income tax returns. The Liquidating Trustee shall promptly submit additional appropriate reports to the Bankruptcy Court and whenever an adverse material event or change occurs which affects either the CEP Liquidating Trust or the rights of the Beneficiaries hereunder.

#### **B.** Federal Income Tax.

- 1. <u>Grantor Trust Status</u>. Subject to definitive guidance from the IRS or a court of competent jurisdiction to the contrary (including the issuance of applicable Treasury Regulations, the receipt by the Liquidating Trustee of a private letter ruling if the Liquidating Trustee so requests one, or the receipt of an adverse determination by the IRS upon audit if not contested by the Liquidating Trustee), the Liquidating Trustee shall file returns for the CEP Liquidating Trust as a grantor trust pursuant to Treasury Regulation § 1.671-1 through 4(a).
- Allocations of CEP Liquidating Trust Taxable Income. All of the CEP Liquidating Trust's income is subject to tax on a current basis, regardless of whether the Liquidating Trustee has established a reserve for Disputed Claims. Subject to the provisions of Section VI.B.1 hereof, allocations of CEP Liquidating Trust taxable income among Beneficiaries shall be determined by reference to the manner in which an amount of cash equal to such taxable income would be distributed (without regard to any restriction on distributions described herein) if, immediately prior to such deemed distribution, the CEP Liquidating Trust had distributed all of its other assets (valued for this purpose at their tax book value) to Beneficiaries (treating to the extent determined by the Liquidating Trustee in his sole discretion, any holder of a Disputed Claim against the Debtors, for this purpose, as a current Beneficiary entitled to distributions), taking into account all prior and concurrent distributions from the CEP Liquidating Trust (including all distributions held in reserve pending the resolution of Disputed Claims). Similarly, taxable losses of the CEP Liquidating Trust will be allocated among Beneficiaries by reference to the manner in which an economic loss would be borne immediately after a liquidating distribution of the remaining Trust Assets. The tax book value of the Trust Assets on the Effective Date for this purpose shall be as set forth in the liquidation analysis attached to the Disclosure Statement, or, if acquired later, the fair market value of such assets on the date such assets were acquired by the CEP Liquidating Trust, adjusted in either case in accordance with tax accounting principles prescribed by the IRS Code, the regulations promulgated thereunder and other applicable administrative and judicial authorities and pronouncements.

**C. Other.** The Liquidating Trustee shall also file (or cause to be filed) any other statements, returns or disclosures relating to the CEP Liquidating Trust, that are required to be filed by any applicable governmental unit or under applicable law, guidelines, rules and regulations.

# **SECTION VII**

# TRANSFER OF BENEFICIARY'S INTERESTS

A. Transfer of Beneficial Interests. Each of the Beneficiaries may sell, transfer, assign or otherwise dispose of directly or indirectly ("Transfer"), the beneficial interests owned by it; provided, however, that (a) such Transfer does not subject the CEP Liquidating Trust to the resignation requirements of either the Securities and Exchange Act of 1934, as amended, or the Investment Company Act of 1940, as amended; (b) any such Transfer is not carried out on a national security exchange or system of automated dissemination of quotations of securities prices; (c) the CEP Liquidating Trust will not participate in the establishment of a market for its beneficial interests or the inclusions beneficial interests thereon, nor will it recognize any transaction made on such market for any purpose that could subject the CEP Liquidating Trust to entitle level taxation for state and federal income tax purposed or affect the CEP Liquidating Trust tax status for state or federal income tax purposes; and (d) upon the closing of any such Transfer, in writing, of such transfer and a new address for the transfer, provided, however, that failing such notice, the Liquidating Trustee shall not have any legal obligation to recognize the transferee as the legal owner of the beneficial interest.

#### **SECTION VIII**

# MISCELLANEOUS PROVISIONS

- **A.** Amendment; Waiver. This Agreement cannot be amended, waived, modified or changed after the Effective Date without prior consent of the Liquidating Trustee and the Committee; <u>provided</u>, <u>however</u>, that in no event shall a change be made to this Agreement that would be inconsistent with the Plan or adversely affect the federal income tax status of the CEP Liquidating Trust as a "grantor trust."
- **B.** Intention of Parties to Establish Grantor Trust. This Agreement is intended to create a "liquidating trust", as defined in Treasury Regulations § 301.7701-4(d), to be taxed as a grantor trust for federal income tax purposes and, to the extent provided by law, shall be governed and construed in all respects as a grant or trust.
- C. Preservation of Privilege. In connection with the rights, claims, and Causes of Action that constitute the Trust Assets, any attorney-client privilege, work-product privilege, or other privilege or immunity attaching to any documents, work-product or communications (whether written or oral) transferred to the CEP Liquidating Trust shall vest in the CEP Liquidating Trust and its representatives, and the Debtors and the Liquidating Trustee are authorized to take all necessary actions to effectuate the transfer of such privileges.

- **D.** Cooperation. The Debtors shall each respectively provide the Liquidating Trustee with copies of such of their books and records as the Liquidating Trustee shall reasonably require for the purpose of performing his duties and exercising his powers hereunder.
- **E.** Laws as to Construction. This Agreement shall be governed by and construed in accordance with the laws of the State of Ohio, without giving effect to rules governing the conflict of law.
- **F.** Relationship to Plan. The principal purpose of this Agreement is to aid in the implementation and effectuation of the Plan, and therefore, is subject to the provisions of the Plan. In the event of a conflict between the Plan and this Agreement, this Agreement shall control.
- **G. Severability.** If any provision of this Agreement or the application thereof to any person or circumstance shall be finally determined by a court of competent jurisdiction to be invalid or unenforceable to any extent, the remainder of this Agreement, or the application of such provision to persons or circumstances other than those as to which it is held invalid or unenforceable, shall not be affected thereby, and such provision of this Agreement shall be valid and enforced to the fullest extent permitted by law unless this Agreement, as modified, will no longer effectuate the intent of the parties hereto in all material respects.
- H. Disposition of the Committee. The Committee shall disband and be released of its duties and obligations on the Effective Date. Any rights or duties of the Committee under this Agreement shall be exercised by the co-chairs of the Committee existing as of the Confirmation Date, and which Committee co-chairs shall continue to take all actions required of the Committee as of the Effective Date. The Liquidating Trustee (on behalf of the CEP Liquidating Trust) shall be permitted to retain the professionals of the Committee.
- I. Professionals of the Liquidating Trustee. The Liquidating Trustee shall retain McGuireWoods LLP as counsel to the CEP Liquidating Trust. The Liquidating Trustee may employ and retain such other professionals as the Liquidating Trustee deems necessary and appropriate, including, without limitation, Baker Hostetler. Such professionals shall be paid from the Liquidation Expense Reserve Account in accordance with this Agreement.
- J. Retention of Jurisdiction; Venue. The Bankruptcy Court shall retain jurisdiction over any dispute pertaining to this Agreement to the fullest extent permitted under applicable law and the Plan. In the event that the Bankruptcy Court lacks jurisdiction or fails or refuses to exercise jurisdiction, the parties agree to bring any claims or causes of action arising out of, relating to or in connection with the Agreement in the United States District Court for the Northern District of Ohio, and hereby irrevocably submit to the jurisdiction of such Court. The parties hereto irrevocably waive any objection to the laying of venue in any action, suit or proceeding arising out of, relating to or in connection with this Agreement that is brought in the aforementioned Court and further waive any claim of forum non conveniens.
- **K. Notices.** Any notice or other communication hereunder shall be in writing and shall be deemed to have been sufficiently given, for all purposes, if deposited, postage prepaid,

in a post office or letter box addressed to the person for whom such notice is intended at such address as set forth below or such other address as filed with the Bankruptcy Court.

If to the Debtors, the CEP Liquidating Trust or the Liquidating Trustee, or the Committee:

#### Debtors:

c/o Joseph F. Hutchinson, Jr., Esq. BAKER & HOSTETLER LLP 3200 National City Center 1900 East 9<sup>th</sup> Street Cleveland, OH 44114

CEP Liquidating Trust or the Liquidating Trustee:

Shaun M. Martin Huron Consulting Group 623 Fifth Avenue, 15<sup>th</sup> Floor New York, NY 10022

# Committee:

c/o Mark E. Freedlander, Esq. Sally E. Edison, Esq. McGuireWoods LLP Dominion Tower, 23<sup>rd</sup> Floor 625 Liberty Avenue Pittsburgh, PA 15222

- L. Notices if to a Beneficiary. Any notice or other communication hereunder shall be in writing and shall be deemed to have been sufficiently given, for all purposes, if deposited, postage prepaid, in a post office or letter box addressed to the person for whom such notice is intended to the name and address determined in accordance with Section III.A hereof.
- **M. Headings.** The section headings contained in this Agreement are solely for convenience of reference and shall not affect the meaning or interpretation of this Agreement or of any term or provision hereof.
- **N.** Counterparts. This Agreement may be executed in multiple counterparts, each of which shall be deemed an original but all of which taken together shall constitute one and the same instrument.

[Remainder of this page intentionally left blank – signatures on following pages]

IN WITNESS WHEREOF, the parties hereto have either executed and acknowledged this Agreement, or caused it to be executed and acknowledged on their behalf by their duly authorized officers or members all as of the date first above written.

### CEP HOLDINGS, LLC

By:
Its:
CREATIVE ENGINEERING POLYMER PRODUCTS, LLC
By: Its:
THERMOPLASTICS ACQUISTION, LLC
By: Its:
COMPOSITE PARTS MEXICO S.A. de C.V.
By: Its:
OFFICIAL COMMITTEE OF UNSECURED CREDITORS
By: Its:
LIQUIDATING TRUSTEE:
By: Its:

\4339882.8

[THIS PAGE INTENTIONALLY LEFT BLANK]

### **EXHIBIT B**

Order (A) Approving Proposed Disclosure Statement, (B) Establishing Procedures for Solicitation and Tabulation of Votes to Accept or Reject Proposed Joint Plan of Liquidation and (C) Scheduling Certain Dates in Connection Therewith [THIS PAGE INTENTIONALLY LEFT BLANK]

### IT IS SO ORDERED.

Dated: 04:56 PM June 06 2007

MAETLYN SHEA-STONUM U.S. Bankruptcy Judge

### UNITED STATES BANKRUPTCY COURT NORTHERN DISTRICT OF OHIO EASTERN DIVISION

	$\mathbf{v}$	
In re:	:	Case No. 06-51848
CEP HOLDINGS, LLC, et al., 1	:	(Jointly Administered)
CEP HOLDINGS, LLC, <u>et ai</u> .,	:	Chapter 11
Debtors.	:	
	:	Honorable Marilyn Shea-Stonum
	:	
	X	

ORDER (A) APPROVING PROPOSED DISCLOSURE STATEMENT, (B) ESTABLISHING PROCEDURES FOR SOLICITATION AND TABULATION OF VOTES TO ACCEPT OR REJECT PROPOSED JOINT PLAN OF LIQUIDATION AND (C) SCHEDULING CERTAIN DATES IN CONNECTION THEREWITH

Upon the motion (Docket No. 572) (the "Motion")<sup>2</sup> of CEP Holdings, LLC and its affiliated debtors and debtors in possession (each a "Debtor" and collectively, the "Debtors" or "CEP") and the Official Committee of Unsecured Creditors (the "Creditors' Committee') in

The Debtors are: CEP Holdings, LLC, Creative Engineered Polymer Products, LLC and Thermoplastics Acquisition, LLC.

<sup>&</sup>lt;sup>2</sup> Capitalized terms used but not otherwise defined herein shall have the meanings given to them in the Motion.

the above-captioned Chapter 11 cases (the "Cases"), for entry of an order, pursuant to sections 502, 1125, 1126 and 1128 of the Bankruptcy Code and Bankruptcy Rules 2002, 3003, 3016, 3017, 3018 and 3020, (i) approving the Disclosure Statement (as such term is defined below), (ii) establishing procedures for the solicitation and tabulation of votes to accept or reject the Plan and (iii) scheduling a hearing on the confirmation of the Plan and approving related notice procedures; the Court having reviewed the Motion, having received evidence and having heard the statements of counsel in support of the relief requested therein at a hearing before the Court (the "Hearing"); and the Court having determined that the legal and factual bases set forth in the Motion and at the Hearing establish just cause for the relief granted herein;

### THE COURT FINDS THAT:

- A. The Court has jurisdiction over this matter pursuant to 28 U.S.C. §§ 157 and 1334.
  - B. This is a core proceeding pursuant to 28 U.S.C. § 157(b)(2).
- C. Notice of the Motion and the Hearing, made in the manner described in the Motion, was sufficient and appropriate under the circumstances and complied with the requirements of the Bankruptcy Code and the Bankruptcy Rules.
- D. The First Amended Disclosure Statement to Accompany First Amended Joint Plan of Liquidation Under Chapter 11 of the Bankruptcy Code Proposed by the Debtors and the Official Committee of Unsecured Creditors Dated May 25, 2007 (Docket No. 591) (the "Disclosure Statement"), contains adequate information within the meaning of section 1125 of the Bankruptcy Code.
- E. The relief requested in the Motion and granted herein is warranted under the circumstances and is in the best interests of the Debtors' respective estates.

- F. The form of the ballot attached hereto as **Exhibit 1** (collectively and including the Instructions set forth therein, the "**Ballot**") (1) is consistent with Official Form No. 14, (2) adequately addresses the particular needs of these Cases, (3) is appropriate for the class of claims entitled to vote to accept or reject the Plan and (4) complies with Bankruptcy Rule 3017(d).
- G. Ballots need not be provided to holders of claims and interest in Classes 1, 2, 3 and 5 under the Plan because: (1) Classes 1, 2 and 3 are unimpaired and are conclusively presumed to accept the Plan in accordance with section 1126(f) of the Bankruptcy Code; and (2) holders of claims and interests in Class 5 under the Plan will not retain or receive any property under the Plan, and therefore, this class is deemed to reject the Plan in accordance with section 1126(g) of the Bankruptcy Code.
- H. Because the Plan calls for the substantive consolidation of the Debtors for purposes of voting on the Plan and distributions under the Plan, solicitation and balloting on the Plan on a consolidated basis is appropriate.
- I. The period during which the Debtors may solicit votes to accept or reject the Plan, as established by this Order, provides a sufficient time for creditors to make informed decisions to accept or reject the Plan and submit timely Ballots.
- J. The procedures for the solicitation and tabulation of votes to accept or reject the Plan, as approved herein, provide a fair and equitable voting process and are consistent with section 1126 of the Bankruptcy Code.
- K. The contents of the Solicitation Packages and the procedures for providing notice of the Confirmation Hearing and the other matters set forth in the Confirmation Hearing Notice comply with the Bankruptcy Rules 2002 and 3017 and constitute sufficient notice to all interested parties in accordance with the Bankruptcy Code and the Bankruptcy Rules.

### IT IS HEREBY ORDERED THAT:

- 1. The Motion is GRANTED in its entirety.
- 2. The Disclosure Statement is APPROVED, pursuant to section 1125 of the Bankruptcy Code.
- 3. The Debtors and the Creditors' Committee are granted authority to make non-substantive modifications to the Disclosure Statement and the Plan prior to dissemination of the Disclosure Statement and the Plan.
- 4. The Ballot substantially in the form attached hereto as **Exhibit 1**, including the Instructions attached to the Ballot, is APPROVED. The Ballot shall be distributed to holders of claims in the class entitled to vote to accept or reject the Plan.
- 5. To be counted as votes to accept or reject the Plan, all Ballots must be properly executed, completed and delivered to BMC, the Debtors' solicitation and tabulation agent, by either (a) mail in the return envelope provided with each Ballot, (b) overnight courier or (c) personal delivery so that, in each case, such Ballots are received by BMC no later than 5:00 p.m., Eastern Time, on July 13, 2007 (the "Voting Deadline").
- 6. Solely for purposes of voting to accept or reject the Plan, and not for the purpose of the allowance of, or distribution on account of, a claim and without prejudice to the rights of the Debtors or the Creditors' Committee in any other context, each claim within the class of claims entitled to vote to accept or reject the Plan shall be temporarily allowed in accordance with the following rules (the "**Tabulation Rules**"):

- a. Unless one or more of the Tabulation Rules set forth below applies and provides otherwise, a claim will be deemed temporarily allowed for voting purposes in an amount equal to: (i) the noncontingent, liquidated and undisputed amount of such claim, as set forth in the Schedules; or (ii) if a proof of claim has been timely filed in respect of such claim, the noncontingent, liquidated and undisputed amount set forth in such proof of claim
- b. If a claim is deemed allowed in accordance with the Plan, such claim will be temporarily allowed for voting purposes in the deemed allowed amount set forth in the Plan.
- c. If a claim for which a proof of claim has been timely filed and has not been disallowed is listed or marked or otherwise referenced on its face as contingent, unliquidated or disputed, either in whole or in part, only the noncontingent, liquidated and undisputed portion, if any, of such claim will be deemed temporarily allowed for voting purposes, subject to the other Tabulation Rules, and the remaining portion of such claim will be disallowed for voting purposes.
- d. If a claim for which a proof of claim has been timely filed is marked as a priority claim, either in whole or in part, but is listed in the Schedules as a nonpriority claim or as a priority claim only in part, such claim will be temporarily allowed for voting purposes as a nonpriority claim in an amount equal to the lesser of (i) the entire amount of such claim as set forth in the proof of claim or (ii) the nonpriority claim set forth in the Schedules, provided that such claim is not listed in the Schedules or marked on the proof of claim as contingent, unliquidated or disputed.
- e. If a claim has been allowed pursuant to a stipulation approved by the Court, such claim will be deemed temporarily allowed for voting purposes in the amount set forth in the stipulation.
- f. If a claim has been estimated or otherwise allowed for voting purposes by order of the Court, such claim will be temporarily allowed for voting purposes in the amount so estimated or allowed by the Court.
- g. With respect to a claim as to which (i) the claim is (A) listed in the Schedules as contingent, unliquidated or disputed or in a zero amount or (B) not listed in the Schedules and (ii) a proof of claim was not timely filed and no stipulation allowing the claim has been approved by the Court, such claim will be disallowed for voting purposes.

- h. If the Debtors have filed and served an objection to a claim on or before the date that is 10 days after the date of service of the Confirmation Hearing Notice, such claim will be disallowed for voting purposes in accordance with the relief sought in the objection, unless the objection only seeks disallowance of part of such claim, in which event only the amount subject to the objection will be disallowed for voting purposes.
- i. If a claim holder identifies a claim amount on its Ballot that is less than the amount otherwise calculated in accordance with the Tabulation Rules, the claim will be temporarily allowed for voting purposes in the lesser amount identified on such Ballot.
- 7. If any claimant seeks to challenge the allowance of its claim for voting purposes in accordance with the Tabulation Rules, such claimant must file a motion, pursuant to Bankruptcy Rule 3018(a), for an order temporarily allowing such claim in a different amount or classification for purposes of voting to accept or reject the Plan (a "Rule 3018 Motion") and serve such motion on counsel for the Debtors (at the address set forth in the Plan) so that it is received no later than the later of (a) the date of service of the Confirmation Hearing Notice and (b) the date of service of a notice of an objection, if any, to the underlying claim. The Debtors shall have 7 days to file and serve any responses to Rule 3018 Motions. In accordance with Bankruptcy Rule 3018, any Ballot submitted by a creditor that files a Rule 3018 Motion shall be counted solely in accordance with the Tabulation Rules and the other applicable provisions contained herein unless and until the underlying claim is temporarily allowed by the Court for voting purposes in a different amount, after notice and a hearing. The Debtors shall coordinate with the Court to adjudicate and resolve all pending Rule 3018 Motions and any responses thereto in a timely fashion and, to the extent possible, at a single hearing in advance of the Confirmation Hearing.
- 8. In tabulating the Ballots, the following additional procedures shall be utilized: (a) if a creditor casts more than one Ballot voting the same Claim before the Voting Deadline, the latest dated properly executed Ballot will be deemed to reflect the voter's intent and thus will

supersede any prior Ballots; (b) creditors will be required to vote all of their claims within a particular class under the Plan either to accept or reject the Plan and may not split their votes; thus, a Ballot (or a group of Ballots within a Plan class received from a single creditor) that partially rejects and partially accepts the Plan will not be counted; and (c) for the purposes of determining whether the numerosity and claim amount requirements of sections 1126(c) and 1126(d) of the Bankruptcy Code have been satisfied, the Tabulation Agent will tabulate only those Ballots received by the Voting Deadline.

- 9. The Confirmation Hearing is scheduled to be conducted on July 24, 2007 at 1:30 p.m. Eastern Time. The Confirmation Hearing may be continued from time to time by the Court without further notice other than the announcement of the adjourned date(s) at the Confirmation Hearing or any continued hearing.
- 10. Objections, if any, to confirmation of the Plan must: (a) be in writing; (b) state the name and address of the objecting party and the nature of the claim or interest of such party; (c) state with particularity the basis and nature of any objection to the confirmation of the Plan; and (d) be filed with the Court and served so that they are received no later than 4:00 p.m., Eastern Time, on the same date as the Voting Deadline (the "Confirmation Objection Deadline") on (i) Joseph F. Hutchinson, Esq., Baker & Hostetler LLP, 3200 National City Center, 1900 East 9th Street, Cleveland, OH 44114-3485; (ii) Mark E. Freedlander, Esq., McGuireWoods LLP, 625 Liberty Avenue, 23rd Floor Dominion Tower, Pittsburgh, PA 15222; and (iii) Maria D. Giannirakis, Esq., Office of the U.S. Trustee, Howard M. Metzenbaum U.S. Courthouse, 201 Superior Ave., East, Suite 441, Cleveland, OH 44114. The Debtors or the Creditors' Committee shall file and serve on the Interested Parties and any party that files an objection to the Plan any reply to any such objections within 2 days of the Confirmation Hearing.

A pre-hearing conference on any objection to the confirmation of the Plan that is filed by the Confirmation Objection Deadline shall be held on July 19, 2007 at 3:30 p.m., Eastern Time.

- 11. The Confirmation Hearing Notice in substantially the form attached hereto as **Exhibit 2** is APPROVED. The Debtors shall serve copies of the Confirmation Hearing Notice, along with the other materials comprising the Solicitation Package, in accordance with the procedures set forth below.
- 12. Pursuant to Bankruptcy Rule 3017(d), June 6, 2007 shall be the record date for purposes of determining which creditors and equity holders are entitled to receive Solicitation Packages and, where applicable, vote on the Plan (the "**Record Date**").
- 3. With respect to any transferred claim, the transferee shall be entitled to receive a Solicitation Package and cast a Ballot on account of such claim only if: (a) all actions necessary to effect the transfer of the claim pursuant to Bankruptcy Rule 3001(e) have been completed by the Record Date; or (b) the transferee files, no later than the Record Date, (i) the documentation required by Bankruptcy Rule 3001(e) to evidence the transfer and (ii) a sworn statement of the transferor supporting the validity of the transfer. Each transferee shall be treated as a single creditor for purposes of the numerosity requirement in section 1126(c) of the Bankruptcy Code and the other voting and solicitation procedures set forth herein.
- 14. The Debtors are authorized to mail or cause to be mailed Solicitation Packages containing copies of: (a) the Confirmation Hearing Notice; (b) the Disclosure Statement (together with the exhibits thereto, that have been filed with the Court before the date of the mailing); (c) the Plan (together with the exhibits thereto, that have been filed with the Court before the date of the mailing), as an exhibit to the Disclosure Statement; (d) letters from the Debtors and the Creditors' Committee recommending acceptance of the Plan (Docket Nos. 582)

& 586); and (e) for Solicitation Packages sent to holders of claims in classes entitled to vote to accept or reject the Plan, an appropriate form of Ballot and a Ballot return envelope.

- 15. Except as described below, Solicitation Packages will be mailed no later than 7 days after the entry of this Order to: (a) all persons or entities that have filed proofs of claim or equity interests on or before the Record Date (or their transferees in accordance with the procedures set forth above); (b) all persons or entities listed in the Schedules as holding liquidated, noncontingent, undisputed claims as of the Record Date (or their transferees in accordance with the procedures set forth above); (c) all other known holders of claims or equity interests against the Debtors, if any, as of the Record Date; (d) all governmental units identified in the Schedules as creditors or potential creditors; (e) all non-Debtor parties to executory contracts and unexpired leases identified in the Schedules or any motion filed with the Court regarding the Debtors' assumption, assumption and assignment or rejection or executory contracts or unexpired leases; (f) all parties in interest that have filed requests for notice in accordance with Bankruptcy Rule 2002 in the Debtors' Cases on or before the Record Date; (g) the U.S. Trustee; (h) the Securities and Exchange Commission; and (i) all other parties identified on the General Service List established by the Case Management Order. Except as otherwise provided above, entities identified in the Schedules as holding contingent, unliquidated or disputed claims that did not file a timely proof of claim will be served only with a copy of the Confirmation Hearing Notice.
- 16. If the Debtors and the Creditors' Committee, in their sole discretion, deem it appropriate not to attempt to obtain accurate addresses for those entities for which the Debtors only have Undeliverable Addresses, based on returned mail in connection with the mailing of the Disclosure Statement Notice or earlier mailings in these Cases, the Debtors are excused from

mailing Solicitation Packages to such entities, unless the Debtors are provided with accurate addresses for such entities, in writing, on or before the Record Date. Failure to mail the Solicitation Packages to such entities shall not constitute inadequate notice of the Confirmation Objection Deadline, the Confirmation Hearing, the Voting Deadline or any other matter.

- 17. The Debtors and BMC are authorized to take any action necessary or appropriate to implement the terms of, and the relief granted in, this Order without further order of the Court.
- 18. This Court shall retain jurisdiction to hear and determine all matters arising from or relating to the implementation of this Order.

IT IS SO ORDERED.

###

### Respectfully submitted by:

/s/ Joseph F. Hutchinson, Jr.

Joseph F. Hutchinson, Jr. (0018210) Thomas M. Wearsch (0078403) Eric R. Goodman (0076035) BAKER & HOSTETLER LLP 3200 National City Center 1900 East 9th Street Cleveland, Ohio 44114-3485

Phone: 216.621.0200 Fax: 216.696.0740

Counsel for the Debtors and Debtors-in-Possession

/s/ Mark Freedlander

Mark Freedlander Sally Edison MCGUIRE WOODS Dominion Tower 625 Liberty Avenue 23rd Floor Pittsburgh, PA 15222

Counsel to the Official Committee of Unsecured Creditors

## **EXHIBIT C**

# **Distribution Analysis**

[THIS PAGE INTENTIONALLY LEFT BLANK]

#### Estimated Funds Available for Administrative and General Unsecured Claims As of May 25, 2007

Projected Proceeds:					
Cash on Hand	[1]	\$	1,477,833		
Deposits related to Sales of Assets in Trust account	[2]		190,000		
Funds held by Wachovia	[3]		5,817,358		
Cash Excrow - Proceeds from Sale of Mexican Facilities	[4]		5,408,597		
Cash Excrow - Interest on Mexican Proceeds	[5]		125,625		
Total Cash				\$	13,019,414
Projected Proceeds from Assets Disposition Liquidation of A/R - Estimate	[6]				1,010,948
Refund from Dupont - Excess of CIA payments over shipments Refund from AIR- Excess Post Petiton Payments to be refunded	[7] [8]				76,000 25,000
Total Proceeds Available					14,131,361
Less: Projected Payoffs and Expenses					
Customer Jr Participation Loan	[9]				2,317,483
Refunds to Customers	[10]				479,132
Wind Down Budget Professional Fees outstanding thru April 30, 2007 Wind Down Costs - Health Care Wind Down Costs - General Wind Down Costs - Professional Fees	[11] [12] [13] [14]	-	480,159 90,000 200,000 500,000		
Total Wind Down Costs					1,270,159
Total Projected Payoffs and Expenses					4,066,773
Sub-total				<u></u>	10,064,588

### Notes:

- [1] Based on cash balances as of May 24, 2007.
- [2] Deposits from Sale of Crestline and the Mexican Facilities
- [3] Based on Wachovia's balance as of May 24, 2007
- [4] Proceeds from the sale of Hermosillo and Chihuahua are in an escrow account at Wachovia
- [5] Interest earned on the Mexican Escrow account goes into a special account with Wachovia
- [6] AR Settlements in place Delphi \$355K, GM \$56K, Visteon \$600K
- [7] Dupont was paid on a CIA basis during the bankruptcy period. Payments were based on estimates. They did a reconciliation and show approx \$76,000 over payment.
- [8] ARI reflects on theoir statement an overpayment of \$25K
- [9] Original Customer Jr particiation was \$2,917,483. This number was reduced by \$600,000 which represents the portion of the Customer Jr. Participation Loan that is unsecured. Accrued interest, if any, is not included
- [10] Several customers continued to make payments on account after a payoff settlement was reached which need to be refunded
- [11] Professional fees that have billed and outstanding to include holdbacks
- [12] Estimated residual health care run-off claims
- [13] Estimated And expenses to include outstanding Workers Comp and Pension Administration Expenses
- [14] Estimate

[THIS PAGE INTENTIONALLY LEFT BLANK]

## **EXHIBIT 2**

### UNITED STATES BANKRUPTCY COURT NORTHERN DISTRICT OF OHIO EASTERN DIVISION

In re:

Case No. 06-51848

(Jointly Administered)

CEP HOLDINGS, LLC, et al.,

Chapter 11

Debtors.

Honorable Marilyn Shea-Stonum

Hearing Date: 7/24/07 at 1:30 p.m.

Objection Deadline: 7/13/07 at 4:00 p.m.

# NOTICE OF (A) DEADLINE FOR CASTING VOTES TO ACCEPT OR REJECT PROPOSED JOINT PLAN OF LIQUIDATION, (B) HEARING TO CONSIDER CONFIRMATION OF PROPOSED JOINT PLAN OF LIQUIDATION AND (C) RELATED MATTERS

### PLEASE TAKE NOTICE OF THE FOLLOWING:

1. On May 25, 2007, the above-captioned debtors and debtors in possession (collectively, the "Debtors") and the Official Committee of Unsecured Creditors (the "Creditors' Committee") filed: (a) the First Amended Joint Plan of Liquidation Under Chapter 11 of the Bankruptcy Code Proposed by the Debtors and the Official Committee of Unsecured Creditors Dated May 25, 2007 (as it may be amended or modified, the "Plan");<sup>2</sup> and (b) the related First Amended Disclosure Statement to Accompany First Amended Joint Plan of Liquidation Under Chapter 11 of the Bankruptcy Code Proposed by the Debtors and the Official Committee of Unsecured Creditors Dated May 25, 2007 (Docket No. 591) (as it may be

The Debtors are: CEP Holdings, LLC, Creative Engineered Polymer Products, LLC and Thermoplastics Acquisition, LLC.

<sup>&</sup>lt;sup>2</sup> Capitalized terms used but not otherwise defined herein shall have the meanings given to them in the Plan.

amended or modified, the "**Disclosure Statement**") under section 1125 of title 11 of the United States Code (the "**Bankruptcy Code**").

- 2. Pursuant to an order of the Court dated June 6, 2007 (the "**Disclosure Statement Order**"), the Disclosure Statement and certain related materials (collectively, the "**Solicitation Materials**") have been approved for solicitation of votes to accept or reject the Plan. In accordance with the Disclosure Statement Order, certain Solicitation Materials have been provided with this Notice, including a copy of the Disclosure Statement and the Plan (including all exhibits thereto filed with the Court).
- 3. A hearing to consider the confirmation of the Plan (the "Confirmation Hearing") will be held before the Honorable Marilyn Shea-Stonum, United States Bankruptcy Judge, in the Judge's usual courtroom at United States Bankruptcy Court, 455 U.S. Courthouse, 2 South Main Street, Akron, Ohio 44308, at 1:30 p.m., Eastern Time, on July 24, 2007.
- 4. Pursuant to the Disclosure Statement Order, the Court approved certain procedures for the tabulation of votes to accept or reject the Plan. If you are the holder of a claim against one of the Debtors as of **June 6, 2007** (the record date established in the Disclosure Statement Order) in a class entitled to vote on the Plan, you have received with this Notice a ballot form (a "**Ballot**") and voting instructions appropriate for your claim. The following procedures apply with respect to voting your claim:
  - a. Except as provided in subparagraph (b) below, for your vote to accept or reject the Plan to be counted, you must complete all required information on the Ballot, execute the Ballot and return the completed Ballot to the address indicated on the Ballot so that it is received by 5:00 p.m., Eastern Time, on July 13, 2007 (the "Voting Deadline"). Any failure to follow the voting instructions included

with the Ballot or to return a properly completed Ballot so that it is received by the Voting Deadline may disqualify your Ballot and your vote.

- b. Your claim has been temporarily allowed solely for purposes of voting to accept or reject the Plan in accordance with the following tabulation rules approved by the Court in the Disclosure Statement Order (the "Tabulation Rules"):
  - i. Unless one or more of the Tabulation Rules set forth below applies and provides otherwise, a claim will be deemed temporarily allowed for voting purposes in an amount equal to: (A) the noncontingent, liquidated and undisputed amount of such claim, as set forth in the Schedules; or (B) if a proof of claim has been timely filed in respect of such claim, the noncontingent, liquidated and undisputed amount set forth in such proof of claim.
  - ii. If a claim is deemed allowed in accordance with the Plan, such claim will be temporarily allowed for voting purposes in the deemed allowed amount set forth in the Plan.
  - iii. If a claim for which a proof of claim has been timely filed and has not been disallowed is listed or marked or otherwise referenced on its face as contingent, unliquidated or disputed, either in whole or in part, only the noncontingent, liquidated and undisputed portion, if any, of such claim will be deemed temporarily allowed for voting purposes, subject to the other Tabulation Rules, and the remaining portion of such claim will be disallowed for voting purposes.
  - iv. If a claim for which a proof of claim has been timely filed is marked as a priority claim, either in whole or in part, but is listed in the Schedules as a nonpriority claim or as a priority claim only in part, such claim will be temporarily allowed for voting purposes as a nonpriority claim in an amount equal to the lesser of (A) the entire amount of such claim as set forth in the proof of claim or (B) the nonpriority claim set forth in the Schedules, provided that such claim is not listed in the Schedules or marked on the proof of claim as contingent, unliquidated or disputed.

- v. If a claim has been allowed pursuant to a stipulation approved by the Court, such claim will be deemed temporarily allowed for voting purposes in the amount set forth in the stipulation.
- vi. If a claim has been estimated or otherwise allowed for voting purposes by order of the Court, such claim will be temporarily allowed for voting purposes in the amount so estimated or allowed by the Court.
- vii. With respect to a claim as to which (A) the claim is (I) listed in the Schedules as contingent, unliquidated or disputed or in a zero amount or (II) not listed in the Schedules and (B) a proof of claim was not timely filed and no stipulation allowing the claim has been approved by the Court, such claim will be disallowed for voting purposes.
- viii. If the Debtors have filed and served an objection to a claim on or before the date that is 10 days after the date of service of the Confirmation Hearing Notice, such claim will be disallowed for voting purposes in accordance with the relief sought in the objection, unless the objection only seeks disallowance of part of such claim, in which event only the amount subject to the objection will be disallowed for voting purposes.
- ix. If a claim holder identifies a claim amount on its Ballot that is less than the amount otherwise calculated in accordance with the Tabulation Rules, the claim will be temporarily allowed for voting purposes in the lesser amount identified on such Ballot.
- c. Temporary allowance or disallowance of your Claim for voting purposes does not constitute an allowance or disallowance of your Claim for purposes of receiving distributions under the Plan and is without prejudice to the rights of the Debtors in any other context, including the right of the Debtors to contest the amount or validity of any claim for purposes of allowance and distribution under the Plan. If you wish to challenge the allowance or disallowance of your Claim for voting purposes, you must file a motion, pursuant to Bankruptcy Rule 3018(a), for an order temporarily allowing your Claim in a different amount or classification for purposes of voting to

accept or reject the Plan and serve such motion on the counsel to the Debtors so that it is received no later than 10 days after the later of (i) the date of service of this Notice and (ii) the date of service of a notice of an objection, if any, to the underlying claim. The Debtors shall have 7 days to file and serve any responses to such motions. Unless the Court orders otherwise, your Claim will not be counted for voting purposes in excess of the amount as determined in accordance with the Tabulation Rules.

- 5. Objections, if any, to the confirmation of the Plan must: (a) be in writing; (b) state the name and address of the objecting party and the nature of the claim or interest of such party; (c) state with particularity the basis and nature of any objection to the confirmation of the Plan; and (d) be filed with the Court and served so that they are received no later than **4:00** p.m., Eastern Time, on July 13, 2007 by (i) Joseph F. Hutchinson, Esq., Baker & Hostetler LLP, 3200 National City Center, 1900 East 9th Street, Cleveland, OH 44114-3485; (ii) Mark E. Freedlander, Esq., McGuireWoods LLP, 625 Liberty Avenue, 23rd Floor Dominion Tower, Pittsburgh, PA 15222; and (iii) Maria D. Giannirakis, Esq., Office of the U.S. Trustee, Howard M. Metzenbaum U.S. Courthouse, 201 Superior Ave., East, Suite 441, Cleveland, OH 44114. A pre-hearing conference on any objection to the confirmation of the Plan that is filed by the Confirmation Objection Deadline will be held on July 19, 2007 at 3:30 p.m., Eastern Time.
- 6. Requests for additional copies of the Disclosure Statement and the Plan (including all exhibits thereto filed with the Court) by parties in interest may be made in writing or orally to BMC Group, Inc., CEP HOLDINGS, LLC, et al., c/o BMC Group, PO Box 903, El Segundo, CA 90245-0903 (telephone number (888) 909-0100). Copies of the Disclosure Statement and the Plan (with all exhibits filed with the Court) also are available for review, without charge, at http://www.bmcgroup.com/cep.

7. The Confirmation Hearing may be continued from time to time without further notice, other than the announcement of the adjourned date(s) at the Confirmation Hearing or any continued hearing.

Dated: June 13, 2007 Cleveland, OH

> CEP HOLDINGS, LLC, et al., Debtors and Debtors-in-Possession

By: /s/ Joseph F. Hutchinson, Jr.
One of Their Attorneys

Joseph F. Hutchinson, Jr. (0018210) Thomas M. Wearsch (0078403) Eric R. Goodman (0076035) BAKER & HOSTETLER LLP 3200 National City Center 1900 East 9th Street Cleveland, Ohio 44114-3485

Phone: 216.621.0200 Fax: 216.696.0740

Counsel for the Debtors and Debtors-in-Possession

# OFFICIAL COMMITTEE OF UNSECURED CREDITORS

By: /s/ Mark E. Freedlander
One of Their Attorneys

Mark E. Freedlander Sally E. Edison MCGUIREWOODS LLP 625 Liberty Avenue 23rd Floor Dominion Tower Pittsburgh, PA 15222

Phone: 412.667.6000 Fax: 412.667-6050

Counsel for the Creditors' Committee

## **EXHIBIT 3A**

### UNITED STATES BANKRUPTCY COURT NORTHERN DISTRICT OF OHIO EASTERN DIVISION

	Х	
In re:	:	Case No. 06-51848
	:	(Jointly Administered)
CEP HOLDINGS, LLC, <u>et</u> <u>al</u> ., <sup>1</sup>	:	
	:	Chapter 11
Debtors.	:	
	:	Honorable Marilyn Shea-Stonum
	:	
	X	

### BALLOT FOR ACCEPTING OR REJECTING THE FIRST AMENDED JOINT PLAN OF LIQUIDATION OF CEP HOLDINGS, LLC AND ITS AFFILIATED DEBTORS AND DEBTORS IN POSSESSION

**CLASS 4: GENERAL UNSECURED CLAIMS** 

THE VOTING DEADLINE TO ACCEPT OR REJECT THE PLAN IS 5:00 P.M., EASTERN TIME, ON JULY 13, 2007

This Ballot is submitted to you to solicit your vote to accept or reject the *First Amended Joint Plan of Liquidation Under Chapter 11 of the Bankruptcy Code Proposed by the Debtors and the Official Committee of Unsecured Creditors Dated May 25, 2007* (as it may be amended or modified, the "**Plan**")<sup>2</sup> described in the accompanying first amended disclosure statement, dated May 25, 2007 (Docket No. 591) (as it may be amended or modified, the "**Disclosure Statement**"). The Court has approved the Disclosure Statement with respect to the Plan. The Disclosure Statement provides information to assist you in deciding how to vote this Ballot. If you do not have a copy of the Disclosure Statement, you may obtain a copy from BMC Group, Inc. (the "**Tabulation Agent**") by writing to CEP HOLDINGS, LLC, et al., c/o BMC Group, 1330 E. Franklin Ave., El Segundo, CA 90245 or by visiting the website maintained by the Tabulation Agent at www.bmcgroup.com/cep.

The Plan can be confirmed by the Bankruptcy Court and thereby made binding on you if it is accepted by the holders of at least two-thirds in amount and more than one-half in number of the

<sup>&</sup>lt;sup>1</sup> The Debtors are: CEP Holdings, LLC, Creative Engineered Polymer Products, LLC and Thermoplastics Acquisition, LLC.

<sup>&</sup>lt;sup>2</sup> Capitalized terms used in this Ballot and the attached Instructions that are not otherwise defined herein have the meanings given to them in the Plan.

Claims in each impaired Class that vote on the Plan and if the Plan otherwise satisfies the applicable requirements of section 1129(a) of the Bankruptcy Code. If the requisite acceptances are not obtained, the Bankruptcy Court may confirm the Plan if it finds that the Plan: (a) provides fair and equitable treatment to, and does not unfairly discriminate against, the Class or Classes rejecting the Plan; and (b) otherwise satisfies the requirements of section 1129(b) of the Bankruptcy Code. If the Plan is confirmed by the Bankruptcy Court it will be binding on you whether or not you vote.

To have your vote counted, you must complete, sign and return this Ballot to the Tabulation Agent so that it is received by the Voting Deadline at the following address:

By regular U.S. mail: CEP HOLDINGS, LLC, et al. c/o BMC Group PO Box 903 El Segundo, CA 90245-0903 By messenger or overnight courier: CEP HOLDINGS, LLC, et al. c/o BMC Group 1330 E. Franklin Ave. El Segundo, CA 90245

If your Ballot is not received by the Voting Deadline, your vote will not count as either an acceptance or rejection of the Plan.

PLEASE READ THE ATTACHED VOTING INFORMATION, INSTRUCTIONS, DISCLOSURE STATEMENT AND PLAN BEFORE COMPLETING THIS BALLOT. YOU MAY WISH TO SEEK LEGAL ADVICE CONCERNING THE PLAN AND YOUR CLASSIFICATION AND TREATMENT UNDER THE PLAN. YOUR CLAIM HAS BEEN PLACED IN CLASS 4 UNDER THE PLAN.

[Continued on Next Page]

PLEASE COMPLETE ITEMS 1 AND 2. IF THIS BALLOT IS NOT SIGNED ON THE APPROPRIATE LINES BELOW, THIS BALLOT WILL NOT BE VALID OR COUNTED AS HAVING BEEN CAST.

**Item 1.** Class Vote. The undersigned, a holder of a Claim against the Debtors in Class 4 in the amount set forth below, votes to (check <u>one</u> box):

ACCEPT THE PLAN	REJECT THE PLAN
☐ CHECK HERE	☐ CHECK HERE
*0010566*	*9910566*
Item 2. Acknowledgments. By signing this Bathe Disclosure Statement and the other applicable undersigned is the claimant or has the power and behalf of the claimant.	e solicitation materials and certifies that the
Name of Holder: A & M PALLET CO	
Claim Amount: \$ 639.60	
15340 GEORGIA RD MIDDLEFIELD, OH 44062	Name Signature
	If by Authorized Agent, Name and Title
	Social Security or Federal Tax I.D. No. (optional)
	Street Address
	City, State, Zip Code
_	Telephone Number
_	Date Completed

# VOTING INFORMATION AND INSTRUCTIONS FOR COMPLETING THE BALLOT

1. In the boxes provided in Item 1 of the Ballot, please indicate acceptance or rejection of the Plan. Complete the Ballot by providing all the information requested and sign, date and return the Ballot to BMC Group, Inc. (the "**Tabulation Agent**") at the following address:

By regular U.S. mail: CEP HOLDINGS, LLC, et al. c/o BMC Group PO Box 903 El Segundo, CA 90245-0903 By messenger or overnight courier: CEP HOLDINGS, LLC, et al. c/o BMC Group 1330 E. Franklin Ave. El Segundo, CA 90245

- 2. Ballots must be received by the Tabulation Agent by 5:00 p.m., Eastern Time, on July 13, 2007 (the "Voting Deadline"). If a Ballot is received after the Voting Deadline, it will not be counted. An envelope addressed to the Tabulation Agent is enclosed for your convenience. Ballots submitted by facsimile will not be accepted.
- 3. Each Ballot you receive is for voting only your Claim described on the Ballot. The attached Ballot is designated only for voting General Unsecured Claims in Class 4. Please complete and return each Ballot you receive. You must vote all of your Claims within a single Class under the Plan to either accept or reject the Plan. Accordingly, if you return more than one Ballot voting different Claims within a single Class under the Plan and the Ballots are not voted in the same manner, those Ballots will not be counted. An otherwise properly executed Ballot that attempts to partially accept and partially reject the Plan will not be counted.
- 4. Your Claim has been temporarily allowed solely for purposes of voting to accept or reject the Plan in accordance with certain tabulation rules approved by the Bankruptcy Court (the "Tabulation Rules"). The Tabulation Rules are set forth in Notice of (A) Deadline for Casting Votes to Accept or Reject Proposed Joint Plan of Liquidation, (B) Hearing to Consider Confirmation of Proposed Joint Plan of Liquidation and (C) Related Matters, which is enclosed with the solicitation materials you receive along with this Ballot. The temporary allowance of your Claim for voting purposes does not constitute an allowance of your Claim for purposes of distribution under the Plan and is without prejudice to the rights of the Debtors or the Trustee in any other context (e.g., the right of the Debtors or the Trustee to contest the amount or validity of any Claim for purposes of allowance under the Plan). If you wish to challenge the temporary allowance of your Claim for voting purposes, you must file a motion, pursuant to Rule 3018(a) of the Federal Rules of Bankruptcy Procedure, for an order temporarily allowing your Claim in a different amount or classification for purposes of voting to accept or reject the Plan and serve such motion on the Debtors so that it is received by the later of June 25, 2007 and 10 days after service of any objection to the underlying claim. Unless the Bankruptcy

Court orders otherwise, your Claim will not be counted as a vote in excess of the amount as determined in accordance with the Tabulation Rules.

- 5. The Ballot does not constitute and shall not be deemed a proof of Claim or Equity Interest or an assertion of a Claim or Equity Interest.
- 6. If you cast more than one Ballot voting the same Claim prior to the Voting Deadline, the last dated properly executed Ballot received by the Tabulation Agent before the Voting Deadline will supersede any prior Ballots.

A PERSON ENTITLED TO VOTE ON THE PLAN MUST COMPLETE AND RETURN THE BALLOT BY THE VOTING DEADLINE TO HAVE THE VOTE COUNT. PLEASE RETURN YOUR BALLOT PROMPTLY.

THE TABULATION AGENT WILL NOT ACCEPT BALLOTS BY FACSIMILE TRANSMISSION.

IF YOU HAVE RECEIVED A DAMAGED BALLOT OR HAVE LOST YOUR BALLOT, OR IF YOU HAVE ANY QUESTIONS CONCERNING THIS BALLOT OR THE VOTING PROCEDURES, PLEASE CALL THE TABULATION AGENT AT (888) 909-0100.

### II. APPLICABLE LAW AND RULES

On approval of the disclosure statement, plan proponents may solicit holders of claims or interests for acceptance of a plan. 11 U.S.C. § 1125(b). Unless the court orders otherwise regarding any unimpaired classes of creditors or equity security holders, the debtor in possession, trustee, proponent of the plan, or clerk, as ordered by the court, must mail to all creditors and equity security holders, and to the U.S. trustee, (1) the plan or court approved summary of the plan, (2) the approved disclosure statement, (3) notice of the time within which to file acceptance and rejection of the plan, and (4) other information as the court may direct, including any opinion of the court approving the disclosure statement or court approved summary of the opinion. Rule 3017(d) of the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules" or "Fed R. Bankr. P."). In addition, notice of the time fixed for filing objections to confirmation of the plan and notice of the hearing on confirmation must be mailed to all creditors and equity security holders, and a ballot must be mailed to those entitled to vote on the plan. Fed. R. Bankr. P. 2002(b), 3017(d).

Section 1123(a)(1) provides that a chapter 11 plan must designate classes of claims and interests for treatment under the reorganization.

Under section 1126(c) of the Bankruptcy Code, an entire class of claims is considered to have accepted the plan if the plan has been accepted by creditors that hold: (1) at least two-thirds in amount, and (2) more than one-half in number of allowed claims of the class held by creditors that have accepted or rejected the plan, *i.e.*, creditors that have voted on the plan. A class of equity security interests is considered to have accepted a plan if the plan has been accepted by the holders of two-thirds in amount of such allowed interests that have voted on the plan.

11 U.S.C. § 1126(d). Under section 1129(a)(10) of the Bankruptcy Code, if there are impaired classes of claims, the court cannot confirm the plan unless it has been accepted by at least one class of non-insiders who hold impaired claims (*i.e.*, claims that are not going to be paid completely or in which some legal, equitable, contractual right is altered). Moreover, under section 1126(f) of the Bankruptcy Code, holders of unimpaired claims are considered to have accepted the plan.

Entities entitled to accept or reject the plan must do so within the time fixed by the court. Fed. R. Bankr. P. 3018(a). Subject to Bankruptcy Rule 3018(b), a creditor or equity security holder whose claim is based on a security of record is not entitled to vote on a plan unless they were the holder of record of such security on the date the order approving the disclosure statement was entered. Fed. R. Bankr. P. 3018(a). An acceptance or rejection must be in writing; identify the plan; be signed by the creditor, equity security holder or an authorized agent; and conform to the Official Form. More than one plan may be accepted or rejected by the voting person, and if more than one plan is accepted, the voting person may designate a preference or preferences among the plans. Fed. R. Bankr. P. 3018(c). A creditor holding an allowed claim that is partly secured and partly unsecured is entitled to accept or reject the plan in both capacities. Fed. R. Bankr. P. 3018(d).

Bankruptcy Rule 3018(b) allows the acceptance or rejection of the plan before commencement of the case if certain solicitation requirements were satisfied under Bankruptcy Rule 3018(b) and section 1126 of the Bankruptcy Code.

Section 1127(a) of the Bankruptcy Code provides that the plan proponent may modify the plan at any time before confirmation, and the modified plan will become the plan. But the plan as modified must meet all the requirements of chapter 11. Bankruptcy Rule 3019 provides that, when there is a proposed modification after balloting has been conducted and the court finds after a hearing that the proposed modification does not adversely affect the treatment of any creditor who has not accepted the modification in writing, the modification shall be deemed to have been accepted by all creditors who previously accepted the plan. If it is determined that the proposed modification does have an adverse effect on the claims of the nonconsenting creditors, then another balloting must take place.

Because more than one plan may be submitted to the creditors for approval, Bankruptcy Rule 3016(b) requires that every proposed plan and modification be dated and identified with the name of the entity or entities submitting such plan or modification. When competing plans are presented and meet the requirements for confirmation, the court must consider the preferences of the creditors and equity security holders in determining which plan to confirm.

The Bankruptcy Code requires the court, after notice, to hold a hearing on the confirmation of a plan. 11 U.S.C. § 1128. At or after the confirmation hearing, the court may confirm the plan and make it binding on all creditors and equity security holders, if it is accepted by the holders or two-thirds in amount and more than one-half in number of claims in each class of creditors and the holders of two-thirds in amount of equity security interests in each class voting on the plan. Before confirmation can be granted, the court also must be satisfied that there has been compliance with the other requirements of confirmation set forth in section 1129 of the Bankruptcy Code, even in the absence of any objections. In order to confirm the plan, the

court must find that the plan is feasible, is proposed in good faith, and that the plan and the proponent of the plan are in compliance with the Bankruptcy Code. In addition, the court must find the confirmation of the plan is not likely to be followed by liquidation or the need for further financial reorganization, unless it is proposed in the plan. 11 U.S.C. § 1129(a)(11). For a complete list of requirements for confirmation of a plan, parties in interest may refer to section 1129 of the Bankruptcy Code.

In the event the required acceptances are not obtained, the court may nevertheless confirm the plan if the court finds that the plan accords fair and equitable treatment to the class or classes rejecting it and otherwise satisfies the requirements of section 1129(b) of the Bankruptcy Code. This procedure is sometimes referred to as a "cramdown."

In addition to the Bankruptcy and Federal Rules of Bankruptcy Procedure, each district may have local bankruptcy court rules that may contain further requirements. Local rules may be obtained at the bankruptcy clerk's office or at the court's website at http://www.ohnb.uscourts.gov.

### 11724800001323 \*11724800001323\*

A & M PALLET CO 15340 GEORGIA RD MIDDLEFIELD, OH 44062



## **EXHIBIT 3B**

### UNITED STATES BANKRUPTCY COURT NORTHERN DISTRICT OF OHIO EASTERN DIVISION

	Χ	
In re:	:	Case No. 06-51848
	:	(Jointly Administered)
CEP HOLDINGS, LLC, <u>et al</u> ., <sup>1</sup>	:	
	:	Chapter 11
Debtors.	:	
	:	Honorable Marilyn Shea-Stonum
	:	
	X	

### BALLOT FOR ACCEPTING OR REJECTING THE FIRST AMENDED JOINT PLAN OF LIQUIDATION OF CEP HOLDINGS, LLC AND ITS AFFILIATED DEBTORS AND DEBTORS IN POSSESSION

**CLASS 4: GENERAL UNSECURED CLAIMS** 

THE VOTING DEADLINE TO ACCEPT OR REJECT THE PLAN IS 5:00 P.M., EASTERN TIME, ON JULY 13, 2007

This Ballot is submitted to you to solicit your vote to accept or reject the *First Amended Joint Plan of Liquidation Under Chapter 11 of the Bankruptcy Code Proposed by the Debtors and the Official Committee of Unsecured Creditors Dated May 25, 2007* (as it may be amended or modified, the "**Plan**")<sup>2</sup> described in the accompanying first amended disclosure statement, dated May 25, 2007 (Docket No. 591) (as it may be amended or modified, the "**Disclosure Statement**"). The Court has approved the Disclosure Statement with respect to the Plan. The Disclosure Statement provides information to assist you in deciding how to vote this Ballot. If you do not have a copy of the Disclosure Statement, you may obtain a copy from BMC Group, Inc. (the "**Tabulation Agent**") by writing to CEP HOLDINGS, LLC, <u>et al.</u>, c/o BMC Group, 1330 E. Franklin Ave., El Segundo, CA 90245 or by visiting the website maintained by the Tabulation Agent at www.bmcgroup.com/cep.

The Plan can be confirmed by the Bankruptcy Court and thereby made binding on you if it is accepted by the holders of at least two-thirds in amount and more than one-half in number of the Claims in each impaired Class that vote on the Plan and if the Plan otherwise satisfies the

The Debtors are: CEP Holdings, LLC, Creative Engineered Polymer Products, LLC and Thermoplastics Acquisition, LLC.

Capitalized terms used in this Ballot and the attached Instructions that are not otherwise defined herein have the meanings given to them in the Plan.

applicable requirements of section 1129(a) of the Bankruptcy Code. If the requisite acceptances are not obtained, the Bankruptcy Court may confirm the Plan if it finds that the Plan: (a) provides fair and equitable treatment to, and does not unfairly discriminate against, the Class or Classes rejecting the Plan; and (b) otherwise satisfies the requirements of section 1129(b) of the Bankruptcy Code. If the Plan is confirmed by the Bankruptcy Court it will be binding on you whether or not you vote.

To have your vote counted, you must complete, sign and return this Ballot to the Tabulation Agent so that it is received by the Voting Deadline at the following address:

By regular U.S. mail: CEP HOLDINGS, LLC, et al. c/o BMC Group PO Box 903 El Segundo, CA 90245-0903 By messenger or overnight courier: CEP HOLDINGS, LLC, et al. c/o BMC Group 1330 E. Franklin Ave. El Segundo, CA 90245

If your Ballot is not received by the Voting Deadline, your vote will not count as either an acceptance or rejection of the Plan.

PLEASE READ THE ATTACHED VOTING INFORMATION, INSTRUCTIONS, DISCLOSURE STATEMENT AND PLAN BEFORE COMPLETING THIS BALLOT. YOU MAY WISH TO SEEK LEGAL ADVICE CONCERNING THE PLAN AND YOUR CLASSIFICATION AND TREATMENT UNDER THE PLAN. YOUR CLAIM HAS BEEN PLACED IN CLASS 4 UNDER THE PLAN.

[Continued on Next Page]

**REJECT** THE PLAN

PLEASE COMPLETE ITEMS 1 AND 2. IF THIS BALLOT IS NOT SIGNED ON THE APPROPRIATE LINES BELOW, THIS BALLOT WILL NOT BE VALID OR COUNTED AS HAVING BEEN CAST.

**Item 1.** Class Vote. The undersigned, a holder of a Claim against the Debtors in Class 4 in the amount set forth below, votes to (check <u>one</u> box):

**ACCEPT** THE PLAN

☐ CHECK HERE	☐ CHECK HERE
*0010032*	*9910032*
Item 2. Acknowledgments. By signing this Bathe Disclosure Statement and the other applicable undersigned is the claimant or has the power and behalf of the claimant.	
Name of Holder: <b>A-1 VENDING</b> Claim Amount: \$	
A-1 VENDING 713 FENTON RD FLINT, MI 48503	Name Signature
	If by Authorized Agent, Name and Title
	Social Security or Federal Tax I.D. No. (optional)  Street Address
	City, State, Zip Code Telephone Number
_	Date Completed

# VOTING INFORMATION AND INSTRUCTIONS FOR COMPLETING THE BALLOT

1. In the boxes provided in Item 1 of the Ballot, please indicate acceptance or rejection of the Plan. Complete the Ballot by providing all the information requested and sign, date and return the Ballot to BMC Group, Inc. (the "**Tabulation Agent**") at the following address:

By regular U.S. mail: CEP HOLDINGS, LLC, et al. c/o BMC Group PO Box 903 El Segundo, CA 90245-0903 By messenger or overnight courier: CEP HOLDINGS, LLC, et al. c/o BMC Group 1330 E. Franklin Ave. El Segundo, CA 90245

- 2. Ballots must be received by the Tabulation Agent by 5:00 p.m., Eastern Time, on July 13, 2007 (the "Voting Deadline"). If a Ballot is received after the Voting Deadline, it will not be counted. An envelope addressed to the Tabulation Agent is enclosed for your convenience. Ballots submitted by facsimile will not be accepted.
- 3. Each Ballot you receive is for voting only your Claim described on the Ballot. The attached Ballot is designated only for voting General Unsecured Claims in Class 4. Please complete and return each Ballot you receive. You must vote all of your Claims within a single Class under the Plan to either accept or reject the Plan. Accordingly, if you return more than one Ballot voting different Claims within a single Class under the Plan and the Ballots are not voted in the same manner, those Ballots will not be counted. An otherwise properly executed Ballot that attempts to partially accept and partially reject the Plan will not be counted.
- 4. Your Claim has been temporarily allowed solely for purposes of voting to accept or reject the Plan in accordance with certain tabulation rules approved by the Bankruptcy Court (the "Tabulation Rules"). The Tabulation Rules are set forth in Notice of (A) Deadline for Casting Votes to Accept or Reject Proposed Joint Plan of Liquidation, (B) Hearing to Consider Confirmation of Proposed Joint Plan of Liquidation and (C) Related Matters, which is enclosed with the solicitation materials you receive along with this Ballot. The temporary allowance of your Claim for voting purposes does not constitute an allowance of your Claim for purposes of distribution under the Plan and is without prejudice to the rights of the Debtors or the Trustee in any other context (e.g., the right of the Debtors or the Trustee to contest the amount or validity of any Claim for purposes of allowance under the Plan). If you wish to challenge the temporary allowance of your Claim for voting purposes, you must file a motion, pursuant to Rule 3018(a) of the Federal Rules of Bankruptcy Procedure, for an order temporarily allowing your Claim in a different amount or classification for purposes of voting to accept or reject the Plan and serve such motion on the Debtors so that it is received by the later of June 25, 2007 and 10 days after service of any objection to the underlying claim. Unless the Bankruptcy

Court orders otherwise, your Claim will not be counted as a vote in excess of the amount as determined in accordance with the Tabulation Rules.

- 5. The Ballot does not constitute and shall not be deemed a proof of Claim or Equity Interest or an assertion of a Claim or Equity Interest.
- 6. If you cast more than one Ballot voting the same Claim prior to the Voting Deadline, the last dated properly executed Ballot received by the Tabulation Agent before the Voting Deadline will supersede any prior Ballots.

A PERSON ENTITLED TO VOTE ON THE PLAN MUST COMPLETE AND RETURN THE BALLOT BY THE VOTING DEADLINE TO HAVE THE VOTE COUNT. PLEASE RETURN YOUR BALLOT PROMPTLY.

THE TABULATION AGENT WILL NOT ACCEPT BALLOTS BY FACSIMILE TRANSMISSION.

IF YOU HAVE RECEIVED A DAMAGED BALLOT OR HAVE LOST YOUR BALLOT, OR IF YOU HAVE ANY QUESTIONS CONCERNING THIS BALLOT OR THE VOTING PROCEDURES, PLEASE CALL THE TABULATION AGENT AT (888) 909-0100.

### II. APPLICABLE LAW AND RULES

On approval of the disclosure statement, plan proponents may solicit holders of claims or interests for acceptance of a plan. 11 U.S.C. § 1125(b). Unless the court orders otherwise regarding any unimpaired classes of creditors or equity security holders, the debtor in possession, trustee, proponent of the plan, or clerk, as ordered by the court, must mail to all creditors and equity security holders, and to the U.S. trustee, (1) the plan or court approved summary of the plan, (2) the approved disclosure statement, (3) notice of the time within which to file acceptance and rejection of the plan, and (4) other information as the court may direct, including any opinion of the court approving the disclosure statement or court approved summary of the opinion. Rule 3017(d) of the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules" or "Fed R. Bankr. P."). In addition, notice of the time fixed for filing objections to confirmation of the plan and notice of the hearing on confirmation must be mailed to all creditors and equity security holders, and a ballot must be mailed to those entitled to vote on the plan. Fed. R. Bankr. P. 2002(b), 3017(d).

Section 1123(a)(1) provides that a chapter 11 plan must designate classes of claims and interests for treatment under the reorganization.

Under section 1126(c) of the Bankruptcy Code, an entire class of claims is considered to have accepted the plan if the plan has been accepted by creditors that hold: (1) at least two-thirds in amount, and (2) more than one-half in number of allowed claims of the class held by creditors that have accepted or rejected the plan, *i.e.*, creditors that have voted on the plan. A class of equity security interests is considered to have accepted a plan if the plan has been accepted by the holders of two-thirds in amount of such allowed interests that have voted on the plan.

11 U.S.C. § 1126(d). Under section 1129(a)(10) of the Bankruptcy Code, if there are impaired classes of claims, the court cannot confirm the plan unless it has been accepted by at least one class of non-insiders who hold impaired claims (*i.e.*, claims that are not going to be paid completely or in which some legal, equitable, contractual right is altered). Moreover, under section 1126(f) of the Bankruptcy Code, holders of unimpaired claims are considered to have accepted the plan.

Entities entitled to accept or reject the plan must do so within the time fixed by the court. Fed. R. Bankr. P. 3018(a). Subject to Bankruptcy Rule 3018(b), a creditor or equity security holder whose claim is based on a security of record is not entitled to vote on a plan unless they were the holder of record of such security on the date the order approving the disclosure statement was entered. Fed. R. Bankr. P. 3018(a). An acceptance or rejection must be in writing; identify the plan; be signed by the creditor, equity security holder or an authorized agent; and conform to the Official Form. More than one plan may be accepted or rejected by the voting person, and if more than one plan is accepted, the voting person may designate a preference or preferences among the plans. Fed. R. Bankr. P. 3018(c). A creditor holding an allowed claim that is partly secured and partly unsecured is entitled to accept or reject the plan in both capacities. Fed. R. Bankr. P. 3018(d).

Bankruptcy Rule 3018(b) allows the acceptance or rejection of the plan before commencement of the case if certain solicitation requirements were satisfied under Bankruptcy Rule 3018(b) and section 1126 of the Bankruptcy Code.

Section 1127(a) of the Bankruptcy Code provides that the plan proponent may modify the plan at any time before confirmation, and the modified plan will become the plan. But the plan as modified must meet all the requirements of chapter 11. Bankruptcy Rule 3019 provides that, when there is a proposed modification after balloting has been conducted and the court finds after a hearing that the proposed modification does not adversely affect the treatment of any creditor who has not accepted the modification in writing, the modification shall be deemed to have been accepted by all creditors who previously accepted the plan. If it is determined that the proposed modification does have an adverse effect on the claims of the nonconsenting creditors, then another balloting must take place.

Because more than one plan may be submitted to the creditors for approval, Bankruptcy Rule 3016(b) requires that every proposed plan and modification be dated and identified with the name of the entity or entities submitting such plan or modification. When competing plans are presented and meet the requirements for confirmation, the court must consider the preferences of the creditors and equity security holders in determining which plan to confirm.

The Bankruptcy Code requires the court, after notice, to hold a hearing on the confirmation of a plan. 11 U.S.C. § 1128. At or after the confirmation hearing, the court may confirm the plan and make it binding on all creditors and equity security holders, if it is accepted by the holders or two-thirds in amount and more than one-half in number of claims in each class of creditors and the holders of two-thirds in amount of equity security interests in each class voting on the plan. Before confirmation can be granted, the court also must be satisfied that there has been compliance with the other requirements of confirmation set forth in section 1129 of the Bankruptcy Code, even in the absence of any objections. In order to confirm the plan, the

court must find that the plan is feasible, is proposed in good faith, and that the plan and the proponent of the plan are in compliance with the Bankruptcy Code. In addition, the court must find the confirmation of the plan is not likely to be followed by liquidation or the need for further financial reorganization, unless it is proposed in the plan. 11 U.S.C. § 1129(a)(11). For a complete list of requirements for confirmation of a plan, parties in interest may refer to section 1129 of the Bankruptcy Code.

In the event the required acceptances are not obtained, the court may nevertheless confirm the plan if the court finds that the plan accords fair and equitable treatment to the class or classes rejecting it and otherwise satisfies the requirements of section 1129(b) of the Bankruptcy Code. This procedure is sometimes referred to as a "cramdown."

In addition to the Bankruptcy and Federal Rules of Bankruptcy Procedure, each district may have local bankruptcy court rules that may contain further requirements. Local rules may be obtained at the bankruptcy clerk's office or at the court's website at http://www.ohnb.uscourts.gov.

11724801006053 \*11724801006053\*

A-1 VENDING 713 FENTON RD FLINT, MI 48503



# **EXHIBIT 3C**

## UNITED STATES BANKRUPTCY COURT NORTHERN DISTRICT OF OHIO EASTERN DIVISION

	X	
In re:	:	Case No. 06-51848
	:	(Jointly Administered)
CEP HOLDINGS, LLC, et al., 1	:	
	:	Chapter 11
Debtors.	:	
	:	Honorable Marilyn Shea-Stonum
	:	
	X	

## BALLOT FOR ACCEPTING OR REJECTING THE FIRST AMENDED JOINT PLAN OF LIQUIDATION OF CEP HOLDINGS, LLC AND ITS AFFILIATED DEBTORS AND DEBTORS IN POSSESSION

### **CLASS 4: GENERAL UNSECURED CLAIMS**

THE VOTING DEADLINE TO ACCEPT OR REJECT THE PLAN IS 5:00 P.M., EASTERN TIME, ON JULY 13, 2007

This Ballot is submitted to you to solicit your vote to accept or reject the *First Amended Joint Plan of Liquidation Under Chapter 11 of the Bankruptcy Code Proposed by the Debtors and the Official Committee of Unsecured Creditors Dated May 25, 2007* (as it may be amended or modified, the "**Plan**")<sup>2</sup> described in the accompanying first amended disclosure statement, dated May 25, 2007 (Docket No. 591) (as it may be amended or modified, the "**Disclosure Statement**"). The Court has approved the Disclosure Statement with respect to the Plan. The Disclosure Statement provides information to assist you in deciding how to vote this Ballot. If you do not have a copy of the Disclosure Statement, you may obtain a copy from BMC Group, Inc. (the "**Tabulation Agent**") by writing to CEP HOLDINGS, LLC, et al., c/o BMC Group, 1330 E. Franklin Ave., El Segundo, CA 90245 or by visiting the website maintained by the Tabulation Agent at www.bmcgroup.com/cep.

The Plan can be confirmed by the Bankruptcy Court and thereby made binding on you if it is accepted by the holders of at least two-thirds in amount and more than one-half in number of the

The Debtors are: CEP Holdings, LLC, Creative Engineered Polymer Products, LLC and Thermoplastics Acquisition, LLC.

<sup>&</sup>lt;sup>2</sup> Capitalized terms used in this Ballot and the attached Instructions that are not otherwise defined herein have the meanings given to them in the Plan.

Claims in each impaired Class that vote on the Plan and if the Plan otherwise satisfies the applicable requirements of section 1129(a) of the Bankruptcy Code. If the requisite acceptances are not obtained, the Bankruptcy Court may confirm the Plan if it finds that the Plan: (a) provides fair and equitable treatment to, and does not unfairly discriminate against, the Class or Classes rejecting the Plan; and (b) otherwise satisfies the requirements of section 1129(b) of the Bankruptcy Code. If the Plan is confirmed by the Bankruptcy Court it will be binding on you whether or not you vote.

To have your vote counted, you must complete, sign and return this Ballot to the Tabulation Agent so that it is received by the Voting Deadline at the following address:

By regular U.S. mail: CEP HOLDINGS, LLC, et al. c/o BMC Group PO Box 903 El Segundo, CA 90245-0903 By messenger or overnight courier: CEP HOLDINGS, LLC, et al. c/o BMC Group 1330 E. Franklin Ave. El Segundo, CA 90245

If your Ballot is not received by the Voting Deadline, your vote will not count as either an acceptance or rejection of the Plan.

PLEASE READ THE ATTACHED VOTING INFORMATION, INSTRUCTIONS, DISCLOSURE STATEMENT AND PLAN BEFORE COMPLETING THIS BALLOT. YOU MAY WISH TO SEEK LEGAL ADVICE CONCERNING THE PLAN AND YOUR CLASSIFICATION AND TREATMENT UNDER THE PLAN. YOUR CLAIM HAS BEEN PLACED IN CLASS 4 UNDER THE PLAN.

[Continued on Next Page]

PLEASE COMPLETE ITEMS 1 AND 2. IF THIS BALLOT IS NOT SIGNED ON THE APPROPRIATE LINES BELOW, THIS BALLOT WILL NOT BE VALID OR COUNTED AS HAVING BEEN CAST.

**Item 1. Class Vote**. The undersigned, a holder of a Claim against the Debtors in Class 4 in the amount set forth below, votes to (check <u>one</u> box):

ACCEPT THE PLAN	REJECT THE PLAN
☐ CHECK HERE	☐ CHECK HERE
*0011311*	*9911311*
Item 2. Acknowledgments. By signing this Bathe Disclosure Statement and the other applicable undersigned is the claimant or has the power and behalf of the claimant.	e solicitation materials and certifies that the
Name of Holder: ARGO PARTNERS	
Claim Amount: \$318,089.60	
12 WEST 37TH ST 9TH FL NEW YORK, NY 10018	Name Signature
	If by Authorized Agent, Name and Title
	Social Security or Federal Tax I.D. No. (optional)
	Street Address
	City, State, Zip Code
_	Telephone Number

Date Completed

# VOTING INFORMATION AND INSTRUCTIONS FOR COMPLETING THE BALLOT

1. In the boxes provided in Item 1 of the Ballot, please indicate acceptance or rejection of the Plan. Complete the Ballot by providing all the information requested and sign, date and return the Ballot to BMC Group, Inc. (the "**Tabulation Agent**") at the following address:

By regular U.S. mail: CEP HOLDINGS, LLC, et al. c/o BMC Group PO Box 903 El Segundo, CA 90245-0903 By messenger or overnight courier: CEP HOLDINGS, LLC, et al. c/o BMC Group 1330 E. Franklin Ave. El Segundo, CA 90245

- 2. Ballots must be received by the Tabulation Agent by 5:00 p.m., Eastern Time, on July 13, 2007 (the "Voting Deadline"). If a Ballot is received after the Voting Deadline, it will not be counted. An envelope addressed to the Tabulation Agent is enclosed for your convenience. Ballots submitted by facsimile will not be accepted.
- 3. Each Ballot you receive is for voting only your Claim described on the Ballot. The attached Ballot is designated only for voting General Unsecured Claims in Class 4. Please complete and return each Ballot you receive. You must vote all of your Claims within a single Class under the Plan to either accept or reject the Plan. Accordingly, if you return more than one Ballot voting different Claims within a single Class under the Plan and the Ballots are not voted in the same manner, those Ballots will not be counted. An otherwise properly executed Ballot that attempts to partially accept and partially reject the Plan will not be counted.
- 4. Your Claim has been temporarily allowed solely for purposes of voting to accept or reject the Plan in accordance with certain tabulation rules approved by the Bankruptcy Court (the "**Tabulation Rules**"). The Tabulation Rules are set forth in *Notice of (A)* Deadline for Casting Votes to Accept or Reject Proposed Joint Plan of Liquidation, (B) Hearing to Consider Confirmation of Proposed Joint Plan of Liquidation and (C) Related Matters, which is enclosed with the solicitation materials you receive along with this Ballot. The temporary allowance of your Claim for voting purposes does not constitute an allowance of your Claim for purposes of distribution under the Plan and is without prejudice to the rights of the Debtors or the Trustee in any other context (e.g., the right of the Debtors or the Trustee to contest the amount or validity of any Claim for purposes of allowance under the Plan). If you wish to challenge the temporary allowance of your Claim for voting purposes, you must file a motion, pursuant to Rule 3018(a) of the Federal Rules of Bankruptcy Procedure, for an order temporarily allowing your Claim in a different amount or classification for purposes of voting to accept or reject the Plan and serve such motion on the Debtors so that it is received by the later of June 25, 2007 and 10 days after service of any objection to the underlying claim. Unless the Bankruptcy

Court orders otherwise, your Claim will not be counted as a vote in excess of the amount as determined in accordance with the Tabulation Rules.

- 5. The Ballot does not constitute and shall not be deemed a proof of Claim or Equity Interest or an assertion of a Claim or Equity Interest.
- 6. If you cast more than one Ballot voting the same Claim prior to the Voting Deadline, the last dated properly executed Ballot received by the Tabulation Agent before the Voting Deadline will supersede any prior Ballots.

A PERSON ENTITLED TO VOTE ON THE PLAN MUST COMPLETE AND RETURN THE BALLOT BY THE VOTING DEADLINE TO HAVE THE VOTE COUNT. PLEASE RETURN YOUR BALLOT PROMPTLY.

THE TABULATION AGENT WILL NOT ACCEPT BALLOTS BY FACSIMILE TRANSMISSION.

IF YOU HAVE RECEIVED A DAMAGED BALLOT OR HAVE LOST YOUR BALLOT, OR IF YOU HAVE ANY QUESTIONS CONCERNING THIS BALLOT OR THE VOTING PROCEDURES, PLEASE CALL THE TABULATION AGENT AT (888) 909-0100.

### II. APPLICABLE LAW AND RULES

On approval of the disclosure statement, plan proponents may solicit holders of claims or interests for acceptance of a plan. 11 U.S.C. § 1125(b). Unless the court orders otherwise regarding any unimpaired classes of creditors or equity security holders, the debtor in possession, trustee, proponent of the plan, or clerk, as ordered by the court, must mail to all creditors and equity security holders, and to the U.S. trustee, (1) the plan or court approved summary of the plan, (2) the approved disclosure statement, (3) notice of the time within which to file acceptance and rejection of the plan, and (4) other information as the court may direct, including any opinion of the court approving the disclosure statement or court approved summary of the opinion. Rule 3017(d) of the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules" or "Fed R. Bankr. P."). In addition, notice of the time fixed for filing objections to confirmation of the plan and notice of the hearing on confirmation must be mailed to all creditors and equity security holders, and a ballot must be mailed to those entitled to vote on the plan. Fed. R. Bankr. P. 2002(b), 3017(d).

Section 1123(a)(1) provides that a chapter 11 plan must designate classes of claims and interests for treatment under the reorganization.

Under section 1126(c) of the Bankruptcy Code, an entire class of claims is considered to have accepted the plan if the plan has been accepted by creditors that hold: (1) at least two-thirds in amount, and (2) more than one-half in number of allowed claims of the class held by creditors that have accepted or rejected the plan, *i.e.*, creditors that have voted on the plan. A class of equity security interests is considered to have accepted a plan if the plan has been accepted by the holders of two-thirds in amount of such allowed interests that have voted on the plan.

11 U.S.C. § 1126(d). Under section 1129(a)(10) of the Bankruptcy Code, if there are impaired classes of claims, the court cannot confirm the plan unless it has been accepted by at least one class of non-insiders who hold impaired claims (*i.e.*, claims that are not going to be paid completely or in which some legal, equitable, contractual right is altered). Moreover, under section 1126(f) of the Bankruptcy Code, holders of unimpaired claims are considered to have accepted the plan.

Entities entitled to accept or reject the plan must do so within the time fixed by the court. Fed. R. Bankr. P. 3018(a). Subject to Bankruptcy Rule 3018(b), a creditor or equity security holder whose claim is based on a security of record is not entitled to vote on a plan unless they were the holder of record of such security on the date the order approving the disclosure statement was entered. Fed. R. Bankr. P. 3018(a). An acceptance or rejection must be in writing; identify the plan; be signed by the creditor, equity security holder or an authorized agent; and conform to the Official Form. More than one plan may be accepted or rejected by the voting person, and if more than one plan is accepted, the voting person may designate a preference or preferences among the plans. Fed. R. Bankr. P. 3018(c). A creditor holding an allowed claim that is partly secured and partly unsecured is entitled to accept or reject the plan in both capacities. Fed. R. Bankr. P. 3018(d).

Bankruptcy Rule 3018(b) allows the acceptance or rejection of the plan before commencement of the case if certain solicitation requirements were satisfied under Bankruptcy Rule 3018(b) and section 1126 of the Bankruptcy Code.

Section 1127(a) of the Bankruptcy Code provides that the plan proponent may modify the plan at any time before confirmation, and the modified plan will become the plan. But the plan as modified must meet all the requirements of chapter 11. Bankruptcy Rule 3019 provides that, when there is a proposed modification after balloting has been conducted and the court finds after a hearing that the proposed modification does not adversely affect the treatment of any creditor who has not accepted the modification in writing, the modification shall be deemed to have been accepted by all creditors who previously accepted the plan. If it is determined that the proposed modification does have an adverse effect on the claims of the nonconsenting creditors, then another balloting must take place.

Because more than one plan may be submitted to the creditors for approval, Bankruptcy Rule 3016(b) requires that every proposed plan and modification be dated and identified with the name of the entity or entities submitting such plan or modification. When competing plans are presented and meet the requirements for confirmation, the court must consider the preferences of the creditors and equity security holders in determining which plan to confirm.

The Bankruptcy Code requires the court, after notice, to hold a hearing on the confirmation of a plan. 11 U.S.C. § 1128. At or after the confirmation hearing, the court may confirm the plan and make it binding on all creditors and equity security holders, if it is accepted by the holders or two-thirds in amount and more than one-half in number of claims in each class of creditors and the holders of two-thirds in amount of equity security interests in each class voting on the plan. Before confirmation can be granted, the court also must be satisfied that there has been compliance with the other requirements of confirmation set forth in section 1129 of the Bankruptcy Code, even in the absence of any objections. In order to confirm the plan, the

court must find that the plan is feasible, is proposed in good faith, and that the plan and the proponent of the plan are in compliance with the Bankruptcy Code. In addition, the court must find the confirmation of the plan is not likely to be followed by liquidation or the need for further financial reorganization, unless it is proposed in the plan. 11 U.S.C. § 1129(a)(11). For a complete list of requirements for confirmation of a plan, parties in interest may refer to section 1129 of the Bankruptcy Code.

In the event the required acceptances are not obtained, the court may nevertheless confirm the plan if the court finds that the plan accords fair and equitable treatment to the class or classes rejecting it and otherwise satisfies the requirements of section 1129(b) of the Bankruptcy Code. This procedure is sometimes referred to as a "cramdown."

In addition to the Bankruptcy and Federal Rules of Bankruptcy Procedure, each district may have local bankruptcy court rules that may contain further requirements. Local rules may be obtained at the bankruptcy clerk's office or at the court's website at http://www.ohnb.uscourts.gov.



# 11724803900013 \*11724803900013\*

ARGO PARTNERS 12 WEST 37TH ST 9TH FL NEW YORK, NY 10018



Transferee	Claim No.	Transferor	Class 4 Amount
ARGO	s2432	AAR-KEL MOULDS LTD.	\$14,830.00
ARGO	s2439	AIC EQUIPMENT & CONTROLS INC.	\$5,849.79
ARGO	s1944	BAY STATE POLYMER DISTRIBUTION, INC	\$33,347.32
ARGO	450	BOLLHOFF RIV-NUT	\$30,639.33
ARGO	s1880	CLAIRON METALS CORP.	\$6,315.40
ARGO	s1947	DELTA MOLD, INC.	\$62,066.66
ARGO	s1899	FGH SYSTEMS, INC.	\$23,263.00
ARGO	s1432	FIDELIS AUTOMATION, INC.	\$10,712.00
ARGO	s1666	K. WM BEACH MFG., INC.	\$9,479.12
ARGO	196-A	METRO MECHANICAL SERVICES INC	\$15,389.43
ARGO	249	OPTIVA SYSTEMS INC	\$8,382.00
ARGO	576	PIQUA TECHNOLOGIES LLC	\$22,192.33
ARGO	s1940	PORTAGE PRECISION POLYMERS	\$11,794.74
ARGO	s2010	RATLIFF & TAYLOR, INC.	\$5,000.00
ARGO	s1956	SIJL DEVELOPMENT COMPANY, LLC	\$7,925.00
ARGO	s2151	SYMMCO INC.	\$5,914.60
ARGO	412	TAYLOR IND SERVICES (HPM DIV)	\$4,195.65
ARGO	s1969	TBDN TENNESSEE COMPANY	\$4,172.00
ARGO	s1494	THE RESIN ENTERPRISE	\$5,491.25
ARGO	s1986	UNIVERSITY BUSINESS INSTALLATIONS	\$9,375.00
ARGO	241-A	WELCH PACKAGING DETROIT INC	\$21,754.98
			\$318,089.60

# **EXHIBIT A**

# for CEP HOLDINGS, LLC, et al.

Total number of parties: 6033

## **Exhibit A - Solicitation**

Svc Lst	Name and Address of Served Party	Mode of Service
24746	BEVERLY H SHIDELER, (RE: IBM CORP), 2 LINCOLN CENTER, OAKBROOK TERRACE, IL, 60181	US Mail (1st Class)
24746	BRIAN ILLION VP AND GC, (RE: ABC GROUP OF COMPANIES), 2 NORELCO CR, TORONTO, ON, M9L 2X6CANADA	US Mail (1st Class)
24746	PEYTON C COCHRANE, (RE: TAX COLLECTOR), 714 GREENSBORO AVE RM 124, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24746	ROSA DOMINY, (RE: IOS CAPITAL INC), 1738 BASS RD, PO BOX 13708, MACON, GA, 31208-3708	US Mail (1st Class)
24799	3560 W MARKET LLC (JOHN DELLAGNESE &, ASSOCIATES (JDA)), EMBASSY CORPORATE PARK, 4000 EMBASSY PKWY, STE 400, AKRON, OH, 44333-8322	US Mail (1st Class)
24847	7TH AVENUE MANAGEMENT, ATTN: MARY MCGEE, 100 SEVENTH AVENUE, SUITE 125, CHARDON, OH, 44024 served 6-18-07	US Mail (1st Class)
24799	A T & T, ACCT: 216 T56-3796 210 8; 330 454 2890 772 3;, AND 330 456 2481 833 3, N17W24300 RIVERWOOD, WAUKESHA, WI, 53187	US Mail (1st Class)
24799	A-1 VENDING, 713 FENTON RD., FLINT, MI, 48503	US Mail (1st Class)
24799	ACCOUNTEMPS, 12400 COLLECTIONS CENTER DRIVE, CHICAGO, IL, 60693	US Mail (1st Class)
24746	ACCOUNTING DEPARTMENT, DAVID ENGLISH MANAGER, (RE: TRIDON), 3200 PARKER DR, ST AUGUSTINGE, FL, 32084	US Mail (1st Class)
24799	ACE IMAGING SOLUTIONS, 6599 GRANGER RD, INDEPENDENCE, OH, 44131	US Mail (1st Class)
24847	ACE/AMERICAN COPY EQUIPMENT, 6599 GRANGER ROAD, CLEVELAND, OH, 44131 served 6-18-07	US Mail (1st Class)
24799	ACOM SOLUTIONS, INC., 2850 E 29TH STREET, LONG BEACH, CA, 90806	US Mail (1st Class)
24847	ADT SECURITY SERVICES, INC, 5590 LAUBY RD, N CANTON, OH, 44720 served 6-18-07	US Mail (1st Class)
24799	AEROTEK INC, 3689 COLLECTION CTR. DR., CHICAGO, IL, 60693	US Mail (1st Class)
24799	AL ANDERSEN COMPANY, 114 W MAIN ST, WAUNAKEE, WI, 53597	US Mail (1st Class)
24799	ALABAMA DEPT OF REVENUE, 50 N RIPLEY ST, MONTGOMERY, AL, 36130-1001	US Mail (1st Class)
24847	ALBIN BUSINESS CENTER AA, 4343 CONCOURSE SUITE 160, ANN ARBOR, MI, 48108 served 6-18-07	US Mail (1st Class)
24799	ALL OHIO CRANE INC, PO BOX 520, CANAL FULTON, OH, 44614	US Mail (1st Class)
24799	AMANO BUSINESS CREDIT, 2012 A MORTON DR, CHARLOTTESVILLE, VA, 22903-6803	US Mail (1st Class)
24799	AQUA PERFECT, INC., 4812 PETER PLACE ,, CINCINATTI, OH, 45246-1037	US Mail (1st Class)
24799	ARAMARK, 1101 MARKET ST, PHILADELPHIA, PA, 19107	US Mail (1st Class)
24746	ARGO PARTNERS, (RE: TRANSFER AGENT), 12 WEST 37TH ST, 9TH FL, NEW YORK, NY, 10018	US Mail (1st Class)
24746	ARGO PARTNERS, PAUL S BERG, (RE: TRANSFER AGENT), 12 WEST 37TH ST, 9TH FL, NEW YORK, NY, 10018	US Mail (1st Class)
24799	ASCOM HASLER/GE CAP PROG, 1010 THOMAS EDISON BLVD SW, CEDAR RAPIDS, IA, 52404	US Mail (1st Class)
24746	ASM, ADAM MOSKOWITZ, (RE: TRANSFER AGENT), 7600 JERICHO TURNPIKE, STE 302, WOODBURY, NY, 11797	US Mail (1st Class)
24746	ASST ATTORNEY GENERAL, VICTORIA A REARDON, (RE: MICHIGAN DEPT OF TREASURY), CADILLAC PLACE, 3030 W GRAND BLVD STE 10-200, DETROIT, MI, 482002	US Mail (1st Class)
24799	AT&T - UNIVERSAL BILLER, ACCT: 171 788 0794 436, ONE AT&T WAY, BEDMINSTER, NJ, 07921-2693	US Mail (1st Class)
24746	ATRADIUS TRADE CREDIT INS, DANA SANTILLI, (RE: TRANSFER AGENT`), 5026 CAMPBELL BLVD, STE C, BALTIMORE, MD, 21236	US Mail (1st Class)
24799	ATTORNEY GENERAL OF THE STATE OF OHIO, (RE: OHIO DEPARTMENT OF TAXATION), COLLECTIONS ENFORCEMENT, 150 E GAY STREET, 21ST FLOOR, COLUMBUS, OH, 43215	US Mail (1st Class)
24746	ATTORNEY GENERAL, MICHAEL A COX, (RE: MICHIGAN DEPT OF TREASURY), CADILLAC PLACE, 3030 W GRAND BLVD STE 10-200, DETROIT, MI, 482002	US Mail (1st Class)
24799	AUTOMOTIVE RENTALS, INC., 9000 MIDLANTIC DR, MT LAUREL, NJ, 08054	US Mail (1st Class)

Svc Lst	st Name and Address of Served Party	
24746	BAKER & HOSTETLER LLP, JOSEPH F HUTCHINSON, JR, NATIONAL CITY CTR, 1900 E 9TH ST STE 3200, CLEVELAND, OH, 44114-3482	US Mail (1st Class)
24746	BAKER & HOSTETLER LLP, NORMA MILLER, NATIONAL CITY CTR, 1900 E 9TH ST STE 3200, CLEVELAND, OH, 44114-3482	US Mail (1st Class)
24746	BAKER & HOSTETLER LLP, THOMAS WEARSCH, NATIONAL CITY CTR, 1900 E 9TH ST STE 3200, CLEVELAND, OH, 44114-3482	US Mail (1st Class)
24746	BALCH & BINGHAM LLP, ERIC T RAY, (RE: ALABAMA POWER CO), PO BOX 306, BIRMINGHAM, AL, 35201	US Mail (1st Class)
24746	BALCH & BINGHAM LLP, W CLARK WATSON, (RE: ALABAMA POWER CO), PO BOX 306, BIRMINGHAM, AL, 35201	US Mail (1st Class)
24746	BANKRUPTCY ADMINISTRATION, (RE: IKON FINANCIAL SERVICES), 1738 BASS RD, PO BOX 13708, MACON, GA, 31208-3708	US Mail (1st Class)
24746	BECKET AND LEE LLP, GILBERT B WEISMAN, (RE: AMERICAN EXPRESS TRAVEL), PO BOX 3001, MALVERN, PA, 19355-0701	US Mail (1st Class)
24746	BENESCH FRIEDLANDER COPLAN, DAVID M NEUMANN, (RE: WACHOVIA CAPITAL FINANCE CORP), 2300 BP TOWER, 200 PUBLIC SQUARE, CLEVELAND, OH, 44114-2378	US Mail (1st Class)
24746	BENESCH FRIEDLANDER COPLAN, MARK A PHILLIPS, (RE: WACHOVIA CAPITAL FINANCE CORP), 2300 BP TOWER, 200 PUBLIC SQUARE, CLEVELAND, OH, 44114-2378	US Mail (1st Class)
24746	BENESCH FRIEDLANDER COPLAN, STUART A LAVEN JR, (RE: WACHOVIA CAPITAL FINANCE CORP), 2300 BP TOWER, 200 PUBLIC SQUARE, CLEVELAND, OH, 44114-2378	US Mail (1st Class)
24799	BERKHEIMER ASSOCIATES, EARNED INCOME TAX ADMIN, 50 N 7TH ST, BANGOR, PA, 18013	US Mail (1st Class)
24847	BIDITUP AUCTIONS WORLDWIDE, INC, ATTENTION: STEVEN MATTES, PRESIDENT, 11426 VENTURA BLVD, STUDIO CITY, CA, 91604 served 6-18-07	US Mail (1st Class)
24799	BROWN AND SHARPE, 250 CIRCUIT DR, NORTH KINGSTOWN, RI, 02852	US Mail (1st Class)
24746	BUCKLEY KING LPA, JEFFREY C TOOLE, (RE: CHASE EQUIPMENT LEASING INC), 600 SUPERIOR AVE E, STE 1400, CLEVELAND, OH, 44114	US Mail (1st Class)
24799	BUREAU OF WORKER'S COMPENSATION, JAMES CONRAD, ADMIN, 30 W SPRING ST, COLUMBUS, OH, 43215-2256	US Mail (1st Class)
24746	BUTZEL LONG, TOM B RADOM ESQ, (RE: DELPHI AUTOMOTIVE SYSTEMS LLC), 100 BLOOMFIELD HILLS PKWY STE 200, BLOOMFIELD HILLS, MI, 48304	US Mail (1st Class)
24746	CALFEE HALTER & GRISWOLD LLP, JEAN R ROBERTSON, (RE: DEBTOR/DEBTOR IN POSSESSION), 1400 MCDONALD INVESTMENT CENTER, 800 SUPERIOR AVE, CLEVELAND, OH, 44114-2688	US Mail (1st Class)
24799	CARLISLE ENGINEERED PRODUCTS INC, DREW PAROBEK & CARRIE BROSIUS, VORYS SATER SEYMOUR & PEASE LLP, 2100 ONE CLEVELAND CENTER, 1375 EAST NINTH ST, CLEVELAND, OH, 44114	US Mail (1st Class)
24799	CARLISLE ENGINEERED PRODUCTS INC, CARRIE M BROSIUS, VORYS SATER SEYMOUR & PEASE LLP, 2100 ONE CLEVELAND CENTER, 1375 EAST NINTH ST, CLEVELAND, OH, 44114	US Mail (1st Class)
24799	CARPENTER SHELLY, PATTI, 2354 MARK ST, CROSWELL, MI, 48422-9640	US Mail (1st Class)
24799	CASCADE TOWNSHIP, MARY J CARPENTER, TREASURER, 2865 THORNHILLS AVE SE, GRAND RAPIDS, MI, 49546-7192	US Mail (1st Class)
24799	CENTRAL COLLECTION AGENCY, NASSIM LYNCH, TAX ADMIN, 1701 LAKESIDE AVE E, CLEVELAND, OH, 44114	US Mail (1st Class)
24799	CENTRAL TAX BUREAU, INC, 141 RANDALL AVE, GIRARD, PA, 16417	US Mail (1st Class)
24746	CEP HOLDINGS, LLC, JAMES D VAN TIEM, 3560 W MARKET ST, STE 300, AKRON, OH, 44333	US Mail (1st Class)
24746	CEP HOLDINGS, LLC, JOSEPH MALLAK, PRES, 3560 W MARKET ST STE 300, FAIRLAWN, OH, 44333-2687	US Mail (1st Class)
24799	CEP INVESTMENTS OF OHIO, INC., C/O JAMES D. VAN TIEM, 3560 W. MARKET STREET, SUITE 300, AKRON, OH, 44333	US Mail (1st Class)
24799	CFD TRUST NO. 8, AN OHIO TRUST, (SEVENTH AVENUE PROPERTIES), 100 SEVENTH AVE. ,STE 150, CHARDON, OH, 44024	US Mail (1st Class)
24847	CHASE EQUIPMENT LEASING INC, 1111 POLARIS PKWY, SUITE A3, COLUMBUS, OH, 43240 served 6-18-07	US Mail (1st Class)
24799	CHASE LEASING CORPORATION FKA BANC ONE LEASING CRP, 1111 POLARIS PKWY, STE A3, COLUMBUS, OH, 43240	US Mail (1st Class)
24799	CINTAS CORPORATION, 6800 CINTAS BLVD., MASON, OH, 45040	US Mail (1st Class)
24799	CITICAPITAL COMMERCIAL CORPORATION, 3950 REGENT BLVD, MAIL STOP S2B-230, IRVING, TX, 75063	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24847	CITICORP DEALER FINANCE, 450 MAMARONECK AVENUE, HARRISON, NY, 10528 served 6-18-07	US Mail (1st Class)
24799	CITICORP DEL-LEASE, 450 MAMARONECK AVE., HARRISON, NY, 10528	US Mail (1st Class)
24799	CITICORP LEASING, INC., 450 MAMARONECK AVE, HARRISON, NY, 10528-2400	US Mail (1st Class)
24847	CITICORP LEASING, INC., 450 MAMARONECK AVE, HARRISON, NY, 10528-2400 served 6-18-07	US Mail (1st Class)
24799	CITY OF ASHTABULA, 4717 MAIN AVE, ASHTABULA, OH, 44004	US Mail (1st Class)
24799	CITY OF IMLAY CITY, 150 N MAIN STREET, IMLAY CITY, MI, 48444	US Mail (1st Class)
24799	CITY OF TUSCALOOSA REVENUE DEPARTMENT, 2201 UNIVERSITY AVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24799	CITY OF VANDALIA, 333 JAMES BOHANAN DR, VANDALIA, OH, 45377-2319	US Mail (1st Class)
24799	CLEANNET OF GREATER MICHIGAN INC., 901 WILSHIRE DR, STE 480, TROY, MI, 48084	US Mail (1st Class)
24799	CO-ACTIV CAPITAL PARTNERS LLC FKA MAC FIN, C/O ANDREW L UNTERLACK ESQUIRE, LAMM RUBENSTONE LESAVOY BUTZ & DAVID, 3600 HORIZON BLVD, STE 200, TREVOSE, PA, 19053	US Mail (1st Class)
24847	COHEN & GRIGSBY, ATTN: THOMAS D MAXSON, 11 STANWIX STREET, 15TH FLOOR, PITTSBURGH, PA, 15222-1319  served 6-18-07	US Mail (1st Class)
24746	COMMONWEALTH OF PENNSYLVANIA, SHARON L ROYER, (RE: UNEMPLOYMENT COMP TAX SERVICES), DEPT OF LABOR & INDUSTRY, 1171 S CAMERON ST RM 312, HARRISBURG, PA, 17104-2513	US Mail (1st Class)
24799	CONTINUED CARE ADMIN., INC, 272 VILLAGE, CLEVELAND, OH, 44147	US Mail (1st Class)
24746	CONTRARIAN FUNDS LLC, (RE: TRANSFER AGENT), 411 WEST PUTNAM ABE, STE 225, GREENWICH, CT, 06830	US Mail (1st Class)
24799	COUNTY TREASURER, MONTGOMERY COUNTY AUDITOR, COUNTY ADMINISTRATION BLDG, 451 WEST THIRD ST, DAYTON, OH, 45422-1033	US Mail (1st Class)
24799	DANKA OFFICE IMAGING COMPANY, 11208 DANKA BLVD, ST. PETERSBURG, FL, 33716	US Mail (1st Class)
24799	DARLING, WENDY M, 67 BETTY LOU DR, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24746	DBA REVENUE MANAGEMENT, LIQUIDITY SOLUTIONS, (RE: TRANSFER AGENT), ONE UNIVERSITY PLZ, STE 312, HACKENSACK, NJ, 07601	US Mail (1st Class)
24847	DELL COMPUTER CORPORATION, ONE DELL WAY, ROUND ROCK, TX, 78682  served 6-18-07	US Mail (1st Class)
24799	DELL FINANCIAL SERVICES L.P., 12234 N. I-35, AUSTIN, TX, 78753-1705	US Mail (1st Class)
24746	DICKINSON WRIGHT PLLC, DANIEL F GOSCH, (RE: VISTEON CORP), 200 OTTAWA AVE NW, STE 900, GRAND RAPIDS, MI, 49503-2427	US Mail (1st Class)
24746	DICKINSON WRIGHT PLLC, JAMES A PLEMMONS, (RE: DAIMLERCHRYSLER CORP), 500 WOODWARD AVE STE 4000, DETROIT, MI, 48226	US Mail (1st Class)
24746	DICKINSON WRIGHT PLLC, KRISTI A KATSMA, (RE: VISTEON CORP), 500 WOODWARD AVE, STE 4000, DETROIT, MI, 48226	US Mail (1st Class)
24746	DICKINSON WRIGHT PLLC, MICHAEL C HAMMER, (RE: VISTEON CORP), 301 E LIBERTY ST STE 500, ANN ARBOR, MI, 48104-2262	US Mail (1st Class)
24799	DIVISION OF TAXATION, 3487 S SMITH RD, FAIRLAWN, OH, 44333	US Mail (1st Class)
24847	DOVE LEASING, 16851 ELEVEN MILE ROAD, ROSEVILLE, MI, 48066 served 6-18-07	US Mail (1st Class)
24799	DOWDLE GAS COMPANY, 2413 HIGHWAY 45 NORTH, P.O. BOX 9129, COLUMBUS, MS, 39705	US Mail (1st Class)
24746	DUPONT, BRUCE D TOBIANSKY, (RE: COMMITTEE OF UNSECURED CREDITORS), BARLEY MILL PLAZA BMP26-2174, 4417 LANCASTER PIKE, WILMINGTON, DE, 19805	US Mail (1st Class)
24847	EDMAC COMPRESSOR COMPANY, P O BOX 10489, WINSTON-SALEM, NC, 27108 served 6-18-07	US Mail (1st Class)
24799	ERIE COUNTY TAX CLAIM BUREAU, 140 W 6TH STREET, RM 110, ERIE, PA, 16501-1011	US Mail (1st Class)
24799	ERVIN LEASING COMPANY, 3893 RESEARCH PARK DR, ANN ARBOR, MI, 48108	US Mail (1st Class)
24746	EXCEL POLYMERS LLC, KATHY MCDONALD, (RE: COMMITTEE OF UNSECURED CREDITORS), 6521 DAVIS INDUSTRIAL PKWY, SOLON, OH, 44139	US Mail (1st Class)
24799	FABNET ASSOCIATES INC ET AL, DARRELL A CLAY, WALTER & HAVERFIELD LLP, THE TOWER AT ERIEVIEW, 1301 EAST 9TH ST, STE 3500, CLEVELAND, OH, 44113-2253	US Mail (1st Class)
24746	FIRTH JOHNSTON MARTINEZ, VICTOR M FIRTH, (RE: DURABOX DE CHIHUAHUA SA), 415 NORTH MESA, 3RD FLOOR, EL PASO, TX, 79901	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24746	FOLEY & LARDNER LLP, NICOLE Y LAMB-HALE, (RE: KAUTEX TEXTRON), 500 WOODWARD AVE, STE 2700, DETROIT, MI, 48226	US Mail (1st Class)
24799	FORESIGHT TECHNOLOGY IBM CSO, PO BOX 12195, RESEARCH TRIANGLE, NC, 27709-2195	US Mail (1st Class)
24847	FTC BUSINESS SOLUTIONS, PO BOX 1823, SUMTER, SC, 29150 served 6-18-07	US Mail (1st Class)
24799	FTC BUSINESS SOLUTIONS, 631 N PIKE W, SUMTER, SC, 29150	US Mail (1st Class)
24847	FULBRIGHT & JAWORSKI LLP, ATTN: MR HEATH D ESTERAK, 300 CONVENT STREET, SUITE 2200, SAN ANTONIO, TX, 78205  served 6-18-07	US Mail (1st Class)
24746	FULBRIGHT & JAWORSKI LLP, MICHAEL M PARKER, (RE: NL VENTURES V CARLISLE LP), 300 COVENT ST STE 2200, SAN ANTONIO, TX, 78205	US Mail (1st Class)
24847	GE BETZ, INC, 4636 SOMERTON ROAD, TREVOSE, PA, 19053 served 6-18-07	US Mail (1st Class)
24799	GE CAPITAL, 1010 THOMAS EDISON BLVD SW, CEDAR RAPIDS, IA, 52404	US Mail (1st Class)
24799	GE INFRASTRUCTURE WATER & PROCESS TECHNOLOGIES (GE, BETZ, INC.), 4636 SOMERTON RD, TREVOSE, PA, 19053	US Mail (1st Class)
24799	GEAUGA COUNTY TREASURER, 231 MAIN ST, STE 1B, CHARDON, OH, 44024-1249	US Mail (1st Class)
24799	GENERAL ELECTRIC CAPITAL CORPORATION, 4333 EDGEWOOD RD NE ,, CEDAR RAPIDS, IA, 52499	US Mail (1st Class)
24799	GENERAL ELECTRIC CAPITAL CORPORATION, 1010 THOMAS EDISON BLVD SW, CEDAR RAPIDS, IA, 52404	US Mail (1st Class)
24847	GENERAL ELECTRIC CORPORATION, 3000 LAKESIDE DRIVE, BANNOCKBURN, IL, 60015 served 6-18-07	US Mail (1st Class)
24746	GOLD KEY PROCESSING LTD, ROBERT TOTH, (RE: COMMITTEE OF UNSECURED CREDITORS), 14910 MADISON RD, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24746	GOLDBERG KOHN BELL BLACK, SHIRA ISENBERG, (RE: WACHOVIA CAPITAL FINANCE CORP), 55 EAST MONROE STREET, STE 3700, CHICAGO, IL, 60603	US Mail (1st Class)
24746	GOLDBERG KOHN, ALAN P SOLOW, (RE: WACHOVIA CAPITAL FINANCE CORP), 55 E MONROE ST STE 3700, CHICAGO, IL, 60603-5806	US Mail (1st Class)
24746	GOLDBERG KOHN, JEREMY M DOWNS, (RE: WACHOVIA (SECURED LENDERS)), 55 E MONROE ST STE 3700, CHICAGO, IL, 60603-5806	US Mail (1st Class)
24799	GRANT THORNTON LLP, KIMBERLY DAVIS RODRIGUEZ, PRINCIPAL, 27777 FRANKLIN ROAD, SUITE 800, SOUTHFIELD, MI, 48034	US Mail (1st Class)
24799	GREAT LAKES WASTE SERVICE, 3328 TORREY RD, FLINT, MI, 48507-3252	US Mail (1st Class)
24799	GRUPO 7 S.A. DE C.V., CALLE 2A, NO. 4 ,COL. CENTRO 31000, CHIAHUAHUA, CHIH., MX, MEXICO	US Mail (1st Class)
24799	HASLER LEASING, A PROGRAM OF GECC, 3000 LAKESIDE DRIVE, SUITE 200N, BANNOCKBURN, IL, 60015	US Mail (1st Class)
24746	HONIGMAN MILLER SCHWARTZ & COHN LLP, DONALD F BATY JR ESQ, (RE: GENERAL MOTORS CORP), FIRST NATIONAL BLDG, 660 WOODWARD AVE STE 2290, DETROIT, MI, 48226-3506	US Mail (1st Class)
24746	HONIGMAN MILLER SCHWARTZ & COHN LLP, E TODD SABLE ESQ, (RE: VALEO INC & VALEO ELECTRICAL SYS), 2290 FIRST NATIONAL BLDG, 660 WOODWARD AVE, DETROIT, MI, 48226	US Mail (1st Class)
24746	HONIGMAN MILLER SCHWARTZ & COHN, AARON M SILVER ESQ, (RE: GENERAL MOTORS CORP), 2290 FIRST NATIONAL BLDG, 660 WOODWARD AVE, DETROIT, MI, 48226	US Mail (1st Class)
24746	HONIGMAN MILLER SCHWARTZ & COHN, SARAH HILTZ SEEWER, (RE: GENERAL MOTORS CORP), 2290 FIRST NATIONAL BLDG, 660 WOODWARD AVE, DETROIT, MI, 48226	US Mail (1st Class)
24746	HOWARD M METZENBAUM US COURTHOUSE, MARIA D GIANNIRAKIS, (RE: US TRUSTEE), 201 SUPERIOR AVE EAST, STE 441, CLEVELAND, OH, 44114	US Mail (1st Class)
24746	HUNTER & SCHANK CO LPA, JOHN J HUNTER JR, (RE: FAURECIA AUTOMOTIVE SEATING), ONE CANTON SQUARE, 1700 CANTON AVE, TOLEDO, OH, 43624	US Mail (1st Class)
24799	HURON CONSULTING GROUP(FKA GLASS & ASSOCIATES INC), 4571 STEPHEN CIRCLE, NW, SUITE 130, CANTON, OH, 44718	US Mail (1st Class)
24799	HYLANT GROUP, 6225 OAK TREE BLVD, CLEVELAND, OH, 44131	US Mail (1st Class)
24847	IKON FINANCIAL SERVICES, 1738 BASS ROAD, MACON, GA, 31210 served 6-18-07	US Mail (1st Class)
24799	IKON OFFICE SOLUTIONS, 70 VALLEY STREAM PKWY, MALVERN, PA, 19355-1407	US Mail (1st Class)
24799	IMLAY CITY, CITY OF, 150 N MAIN ST, IMLAY CITY, MI, 48444	US Mail (1st Class)
24799	INCOME TAX DEPARTMENT, 16 NORTH MAIN STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24799	INTERNAL REVENUE SERVICE, DEPARTMENT OF THE TREASURY, 1240 E 9TH ST, RM 457, CLEVELAND, OH, 44199	US Mail (1st Class)
24847	IOS CAPITAL, PO BOX 9115, MACON, GA, 31208-9115 served 6-18-07	US Mail (1st Class)
24847	JAVITCH, BLOCK, & RATHBONE, ATTN:THEODORE A KONSTANTINOPOULOS, 1100 SUPERIOR AVENUE, 19TH FLOOR, CLEVELAND, OH, 44114-9971 served 6-18-07	US Mail (1st Class)
24799	JOHNSON CONTROLS, INC., 5757 N GREEN BAY AVE, GLENDALE, WI, 53209-4408	US Mail (1st Class)
24746	LAMM RUBENSTONE LESAVOY, ANDREW L UNTERLACK ESQ, (RE: CO ACTIV CAPITAL CORP), 3600 HORIZON BLVD, STE 200, TREVOSE, PA, 19053	US Mail (1st Class)
24746	LANXESS CORP, ALAN P RUGH, (RE: COMMITTEE OF UNSECURED CREDITORS), 111 RIDC PARK WEST DR, PITTSBURG, PA, 15275-1112	US Mail (1st Class)
24847	LANXESS CORPORATION, JILL LOCNIKAR BRADLEY, 111 RIDC PARK WEST DRIVE, PITTSBURGH, PA, 15275-1112 served 6-18-07	US Mail (1st Class)
24799	LAPEER, CITY OF, 576 LIBERTY ST, LAPEER, MI, 48446	US Mail (1st Class)
24847	LARRY FELLENGER, 2816 OAKWOOD AVE, SOUTHINGTON, OH, 44470 served 6-18-07	US Mail (1st Class)
24746	LEECH TISHMAN FUSCALDO & LAMPL, KIMBERLY A COLEMAN ESQ, (RE: WASHINGTON PENN PLASTIC CO), 525 WILLIAM PENN PLACE, 30TH FLOOR, PITTSBURGH, PA, 15219	US Mail (1st Class)
24799	LIVONIA, CITY OF, LINDA GRIMSBY TREASURER, 33000 CIVIC CENTER DR, LIVONIA, MI, 48154	US Mail (1st Class)
24799	LUDWIG, DAVID E, 10731 ETTER RD, WATERFORD, PA, 16441	US Mail (1st Class)
24799	MATHEWS, DAVID G, 4523 CARDINAL DR, RICHMOND, IN, 47374	US Mail (1st Class)
24799	MATTEC CORPORATION, 1301 MATTEC DR, LOVELAND, OH, 45140-7300	US Mail (1st Class)
24746	MCDONALD HOPKINS CO LPA, JEAN R ROBERTSON, (RE: DEBTOR/DEBTOR IN POSSESSION), 600 SUPERIOR AVE E, SUITE 2100, CLEVELAND, OH, 44114	US Mail (1st Class)
24746	MCDONALD HOPKINS CO LPA, SHAWN M RILEY, (RE: DEBTOR/DEBTOR IN POSSESSION), 600 SUPERIOR AVE E, SUITE 2100, CLEVELAND, OH, 44114	US Mail (1st Class)
24746	MCDONALD HOPKINS CO, LPA, ELIZABETH WAMBSGAND, (RE: DEBTOR/DEBTOR IN POSSESSION), 600 SUPERIOR AVE E, SUITE 2100, CLEVELAND, OH, 44114	US Mail (1st Class)
24746	MCGUIRE WOODS LLP, MARK E FREEDLANDER ESQ, (RE: COMMITTEE OF UNSECURED CREDITORS), 625 LIBERTY AVE, DOMINION TOWER 23RD FL, PITTSBURGH, PA, 15222	US Mail (1st Class)
24746	MCGUIRE WOODS LLP, SALLY E EDISON ESQ, (RE: COMMITTEE OF UNSECURED CREDITORS), 625 LIBERTY AVE, DOMINION TOWER 23RD FL, PITTSBURGH, PA, 15222	US Mail (1st Class)
24746	MCGUIRE WOODS LLP, WILLIAM PRICE, (RE: COMMITTEE OF UNSECURED CREDITORS), 625 LIBERTY AVE, DOMINION TOWER 23RD FL, PITTSBURGH, PA, 15222	US Mail (1st Class)
24746	MCGUIREWOODS LLP, MICHAEL J ROESCHENTHALER ESQ, (RE: COMMITTEE OF UNSECURED CREDITORS), 625 LIBERTY AVE, DOMINION TOWER 23RD FL, PITTSBURGH, PA, 15222	US Mail (1st Class)
24847	MCMASTER-CARR SUPPLY COMPANY, ATTN: STEPHANIE PHIFER, P O BOX 7690, CHICAGO, IL, 60680-7690 served 6-18-07	US Mail (1st Class)
24799	MEDINA COUNTY TREASURER, JOHN A BURKE, 144 N BROADWAY STREET, MEDINA, OH, 44256	US Mail (1st Class)
24799	MENG, LAN, 17222 ORCHARD RIDGE, NORTHVILLE, MI, 48168	US Mail (1st Class)
24799	MICHIGAN DEPT OF TREASURY DEPT 77003, CADILLAC PLACE, 3060 W GRAND BLVD STE 2-200, DETROIT, MI, 48277-0003	US Mail (1st Class)
24799	MILLCREEK TAX COLLECTOR, MILLCREEK MUNICIPAL BUILDING, 3608 W 26TH STREET, ERIE, PA, 16506	US Mail (1st Class)
24799	MT BUSINESS TECHNOLOGIES, INC., 8200 E. PLEASANT VALLEY ROAD, INDEPENDENCE, OH, 44131	US Mail (1st Class)
24799	NEOPOST LEASING, INC., 30955 HUNTWOOD AVE, HAYWARD, CA, 94544	US Mail (1st Class)
24799	NL VENTURES V CARLISLE LP, MICHAEL M PARKER, FULBRIGHT & JAWORSKI LLP, 300 CONVENT ST, STE 2200, SAN ANTONIO, TX, 78205-3792	US Mail (1st Class)
24799	NL VENTURES V CARLISLE, LP, C/O AIC VENTURES, 8080 NORTH CENTRAL EXPWY, STE 1080, DALLAS, TX, 75206	US Mail (1st Class)
24799	NORTEL NETWORKS, 8200 DIXIE RD, STE 100, BRAMPTON, ON, L6T 5P6CANADA	US Mail (1st Class)
24799	NORTHWEST REGIONAL SYSTEMS OF MANSFIELD, INC., 485 OAK ST, MANSFIELD, OH, 44907	US Mail (1st Class)
24799	OAKRIDGE LAWN & SNOW, 1665 FISH LAKE RD., LAPEER, MI, 48446	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24746	OFFICE OF CHIEF COUNSEL, GENNICE D BRICKHOUSE, (RE: PENSION BENEFIT GUARANTY CORP), 1200 K ST NW, STE 340, WASHINGTON, DC, 20005-4026	US Mail (1st Class)
24799	OFFICE OF UNEMPLOYMENT COMPENSATION, SEVENTH AND FORSTER STREETS, HARRISBURG, PA, 17121	US Mail (1st Class)
24799	OHIO BUREAU OF WORKERS COMPENSATION, LAW SECTION BANKRUPTCY UNIT, PO BOX 15567, 30 W SPRING ST, COLUMBUS, OH, 43215-0567	US Mail (1st Class)
24799	OHIO DEPARTMENT OF TAXATION, BANKRUPTCY DIVISION, ATTN REBECCA L DAUM, PO BOX 530, COLUMBUS, OH, 43216-0530	US Mail (1st Class)
24799	OHIO DEPT JOB & FAMILY SERVICES, 30 E BROAD ST 32ND FL, COLUMBUS, OH, 43215-3414	US Mail (1st Class)
24799	OHIO SCHOOL DISTRICE INCOME TAX, TAX COMMISSIONERS OFFICE, 30 E BROAD ST 22ND FL, COLUMBUS, OH, 43215-3414	US Mail (1st Class)
24799	ONLINE EMPLOYMENT SERVICES INC., 6258 WEST SAGINAW HIGHWAY, LANSING, MI, 48917	US Mail (1st Class)
24799	OPTIVA SYSTEMS, INC., 23399 COMMERCE DRIVE, SUITE B-5A, FARMINGTON HILLS, MI, 48335	US Mail (1st Class)
24847	PACKAGING CORPORATION OF AMERICA, ATTN: KAREN E MCGRILL, 900 E DIEHL ROAD, SUITE #131, NAPERVILLE, IL, 60563  served 6-18-07	US Mail (1st Class)
24746	PBGC, JEFFREY B COHEN CHIEF COUNSEL, (RE: PENSION BENEFIT GUARANTY CORP), 1200 K ST NW, STE 340, WASHINGTON, DC, 20005-4026	US Mail (1st Class)
24746	PEAR SPERLING EGGAN & DANIELS PC, KEVIN SUMMERS, (RE: ATS AUTOMATION TECHNOLOGY SCHWOPE), 1349 S HURON ST, STE 1, YPSILANTI, MI, 48197	US Mail (1st Class)
24799	PITNEY BOWES, 1 ELMCROFT ROAD, STAMFORD, CT, 06926-0700	US Mail (1st Class)
24799	PITNEY BOWES, 1545 KEYSTONE AVE ,, LANSING, MI, 48911-4039	US Mail (1st Class)
24799	PITNEY BOWES, 1225 CORPORATE DR ,, HOLLAND, OH, 43528	US Mail (1st Class)
24847	PITNEY BOWES CREDIT CORPORATION, 1313 N ATLANTIC ST FL3, SPOKANE, WA, 99201-2303 served 6-18-07	US Mail (1st Class)
24799	PITNEY BOWES CREDIT CORPORATION, 2225 AMERICAN DRIVE ,, NEENAH, WI, 54956-1005	US Mail (1st Class)
24847	PITNEY BOWES, INC, 4517 HONEYWELL COURT, DAYTON, OH, 45424-5781 served 6-18-07	US Mail (1st Class)
24799	PQ SYSTEMS, 10468 MIAMISBURG SPRINGBORO RD, MIAMISBURG, OH, 45342-4834	US Mail (1st Class)
24799	PRIMA TECHNOLOGIES INC, LYNN M BRIMER, STROBL & SHARP PC, 300 E LONG LAKE RD, STE 200, BLOOMFIELD HILLS, MI, 48304-2376	US Mail (1st Class)
24847	PRO-COPY LEASING, 4720 GLENDALE MILFORD ROAD, CINCINNATI, OH, 45242 served 6-18-07	US Mail (1st Class)
24799	QUENCH/DOLPHIN CAPITAL CORP, PO BOX 605, MOBERLY, MO, 65270	US Mail (1st Class)
24799	REHAB CTR (PROGRESS IND) THE, 270 STERKEL BLVD, MANSFIELD, OH, 44907	US Mail (1st Class)
24746	RHODIA, DOREEN WILSON-BAILEY, (RE: COMMITTEE OF UNSECURED CREDITORS), 8 CEDAR BROOK DR, CRANBURY, NJ, 08512	US Mail (1st Class)
24799	RICHLAND COUNTY TREASURER, DANIEL F. SMITH, 50 PARK AVE EAST, MANSFIELD, OH, 44902	US Mail (1st Class)
24847	RS OFFICE SOLUTIONS, PO BOX 697, SANDUSKY, OH, 44870 served 6-18-07	US Mail (1st Class)
24799	RS OFFICE SOLUTIONS, 3911 VENICE RD, SANDUSKY, OH, 44870	US Mail (1st Class)
24847	SAFEWAY PACKAGING, INC, ATTN: KEVIN MANOR, 245 BOULDER LANE, MINSTER, OH, 45865 served 6-18-07	US Mail (1st Class)
24847	SAFEWAY PACKAGING, INC, ATTN: KEVIN MANOR, P O BOX 54, MINSTER, OH, 45865 served 6-18-07	US Mail (1st Class)
24799	SANTA MARIA INDUSTRIAL PARTNERS, LP, CALZ. DEL VALLE #110 OTE. ,COL. DEL VALLE, GARZA GARCIA, NUEVO LEON, MX, MEXICO	US Mail (1st Class)
24799	SC DEPT OF REVENUE WITHHOLDING, COLUMBIA MILLS BUILDING, 301 GERVAIS ST, COLUMBIA, SC, 29214-0004	US Mail (1st Class)
24746	SCHIRRMEISTER DIAZ-ARRASTIA BREM, GEORGE R DIAZ-ARRASTIA, (RE: DU-PONT MEXICO), PENNZIL PLACE NORTH TOWER, 700 MILAM 10TH FL, HOUSTON, TX, 77002	US Mail (1st Class)
24746	SCHWARZWALD & MCNAIR LLP, DAVID M FUSCO, (RE: UNITED STEEL PAPER FORESTRY ET AL), PENTON MEDIA BLDG, 1300 EAST NINTH ST STE 616, CLEVELAND, OH, 44114-1503	US Mail (1st Class)
24799	SECURITIES AND EXCHANGE COMMISSION, CHICAGO REGIONAL OFFICE, MERRI JO GILLETTE, REGIONAL DIRECTOR, 175 W. JACKSON BOULEVARD, SUITE 900, CHICAGO, IL, 60604	US Mail (1st Class)
24799	SECURITIES AND EXCHANGE COMMISSION, HEADQUARTERS, 100 F STREET, NE, WASHINGTON, DC, 20549	US Mail (1st Class)

## **Exhibit A - Solicitation**

Svc Lst	Name and Address of Served Party	Mode of Service
24799	SIJL DEVELOPMENT COMPANY, LLC, 26111 W. 14 MILE RD., STE. LL-4, FRANKLIN, MI, 48025	US Mail (1st Class)
24799	SOUTH CAROLINA EMPLOYMENT SECURITY COMMISSION, 1550 GADSDEN ST., COLUMBIA, SC, 29202	US Mail (1st Class)
24799	SPECIALTY RISK SERVICES LLC, RICHARD COSTELLO, BANKRUPTCY UNIT T-1-55, HARTFORD PLAZA - TOWER, HARTFORD, CT, 06115	US Mail (1st Class)
24799	STARK COUNTY TREASURER, GARY D ZIEGLER, 110 CENTRAL PLAZA SOUTH STE 250, CANTON, OH, 44702-1410	US Mail (1st Class)
24799	STATE OF ALABAMA, 534 14TH STREET, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24799	STATE OF ALABAMA DEPT OF INDUSTRIAL RELATIONS, UNEMPLOYMENT AGENCY, JAMES A WEBB, 649 MONROE ST, MONTGOMERY, AL, 36131	US Mail (1st Class)
24746	STROBL & SHARP PC, LYNN M BRIMER, (RE: PRIMA TECHNOLOGIES INC.), 300 E LONG LAKE RD, STE 200, BLOOMFIELD, MI, 48304-2376	US Mail (1st Class)
24799	SUMMIT COUNTY AUDITORS OFC, 175 S MAIN ST, #414, AKRON, OH, 44308-1311	US Mail (1st Class)
24799	SUNGARD RECOVERY SERVICES, 680 E SWEDESFORD RD, WAYNE, PA, 19087-1605	US Mail (1st Class)
24799	SUPPLYON AG, DEP. FINANCE, LUDWIGSTRASSE 49, HALLBERGMOOS, 85399GERMANY	US Mail (1st Class)
24847	TAFT, STETTINIUS & HOLLISTER LLP, ATTN: RICHARD L FERRELL, 425 WALNUT STREET, SUITE 1800, CINCINNATI, OH, 45202-3957 served 6-18-07	US Mail (1st Class)
24799	TEMTRON INC, 7530 MARSHALL AVE, ST LOUIS, MO, 63119	US Mail (1st Class)
24746	THE BROWN CORP OF GREENVILLE, MARK FERDERBER, (RE: COMMITTEE OF UNSECURED CREDITORS), 1927 N THEOBALD ST, GREENVILLE, MS, 38703	US Mail (1st Class)
24746	THE NANTUMCDONALDCKET BUILDING, JOHNATHAN P BLAKELY ESQ, (RE: BERNLOHR WERTZ LLP), 23 S MAIN ST, THRID FLOOR, AKON, OH, 44308	US Mail (1st Class)
24799	THE PERFECT WATER CO., LLC, 51410 MILANO DR ,STE 110, MACOMB, MI, 48042-4015	US Mail (1st Class)
24746	THOMPSON HINE LLP, ANDREW L TURSCAK JR, (RE: FORD MOTOR COMPANY), 3900 KEY CENTER, 127 PUBLIC SQUARE, CLEVELAND, OH, 44114-1216	US Mail (1st Class)
24746	THOMPSON HINE LLP, JENNIFER A LESNY FLEMING, (RE: FORD MOTOR COMPANY), 3900 KEY CENTER, 127 PUBLIC SQUARE, CLEVELAND, OH, 44114-1216	US Mail (1st Class)
24847	THRASHER, DINSMORE & DOLAN, ATTN: MATTHEW J DOLAN, 100 SEVENTH AVENUE, SUITE 150, CHARDON, OH, 44024  served 6-18-07	US Mail (1st Class)
24799	TOYOTA FINANCIAL SERVICES, COMMERCIAL FINANCIAL GROUP, PO BOX 3457, TORRANCE, CA, 90510-3457	US Mail (1st Class)
24847	TOYOTA MOTOR CREDIT CORPORATION, PO BOX 3457, TORRANCE, CA, 90510-3467 served 6-18-07	US Mail (1st Class)
24799	TRADEBEAM, INC., TWO WATERS PARK DR., SUITE 200,, SAN MATEO, CA, 94403	US Mail (1st Class)
24799	TREASURER OF CRAWFORD COUNTY, GARY COLE, 112 E MANSFIELD ST, BUCYRUS, OH, 44820-0150	US Mail (1st Class)
24799	TREASURER OF STATE OF OHIO, JENNETTE B BRADLEY, 30 E BROAD ST 9TH FL, COLUMBUS, OH, 43266-0421	US Mail (1st Class)
24799	TREASURER OF THE STATE OF OHIO, OFFICE OF FISCAL ADMINISTRATION, 30 E BROAD ST 9TH FL, COLUMBUS, OH, 43266-0421	US Mail (1st Class)
24799	TREASURER, CITY OF CANTON (400), ROBERT SCHIRACK, TREASURER, 218 CLEVELAND AVE SW, CANTON, OH, 44702	US Mail (1st Class)
24799	TREASURER, CITY OF DETROIT, 128 CITY-COUNTY BLDG, DETROIT, MI, 48226	US Mail (1st Class)
24799	TREASURER, CITY OF FLINT, 1ST FLOOR CITY HALL, 1101 S SAGINAW ST, FLINT, MI, 48502	US Mail (1st Class)
24799	TRIPLE S CONSTRUCTION, 8521 SILVERWOOD RD, SILVERWOOD, MI, 48760	US Mail (1st Class)
24799	TRIPPLEHORN'S PROFESSIONAL LAWN & SNOW SERVICE INC, 2896 KLAM RD, LAPEER, MI, 48446	US Mail (1st Class)
24799	TUSCALOOSA COUNTY, 2201 UNIVERSITY AVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24799	TUSCALOOSA COUNTY, ALABAMA, PEYTON C COCHRANE TAX COLLECTOR, 714 GREENSBORO AVE, RM 124, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24799	UNITED STATES TREASURY, INTERNAL REVENUE SERVICE, 550 MAIN ST, CINCINNATI, OH, 45999-0039	US Mail (1st Class)
24799	VAN BUREN TOWNSHIP, SHARRY A BUDD, TREASURER, 46425 TYLER ROAD, BELLEVILLE, MI, 48111	US Mail (1st Class)

## **Exhibit A - Solicitation**

Svc Lst	Name and Address of Served Party	Mode of Service
24746	VORYS SATER SEYMOUR AND PEASE LLP, CARRIE M BROSIUS, (RE: CARLISLE ENGINEERED PRODUCTS INC), ONE CLEVELAND CENTER, 1375 EAST NINTH ST STE 2100, CLEVELAND, OH, 44114	US Mail (1st Class)
24746	VORYS SATER SEYMOUR AND PEASE LLP, DREW T PAROBEK ESQ, (RE: CARLISLE ENGINEERED PRODUCTS INC), ONE CLEVELAND CENTER, 1375 EAST NINTH ST STE 2100, CLEVELAND, OH, 44114	US Mail (1st Class)
24746	VORYS SATER SEYMOUR AND PEASE LLP, ROBERT J SIDMAN ESQ, (RE: HONDA OF AMERICA MFG), PO BOX 1008, COLUMBUS, OH, 43216-1008	US Mail (1st Class)
24746	VORYS SATER SEYMOUR AND PEASE LLP, ROBERT J SIDMNA ESQ, (RE: HONDA OF AMERICA MFG), PO BOX 1008, COLUMBUS, OH, 43216-1008	US Mail (1st Class)
24847	VORYS, SATER, SEYMOUR & PEASE, LLP, ATTENTION: SCOTT DORAN, ESQ, 52 E GAY STREET, COLUMBUS, OH, 43215 served 6-18-07	US Mail (1st Class)
24746	WASHINGTON PENN PLASTIC CO INC, BOB STOUGH, (RE: OAKBROOK TERRACE), PO BOX 236, 450 RACETRACK RD, WASHINGTON, PA, 15301	US Mail (1st Class)
24799	WAYNE COUNTY CCD, ATTN: MICHAEL DOTSON, 801 W FORT STREET, DETROIT, MI, 48226	US Mail (1st Class)
24847	WELLS FARGO FINANCIAL LEASING, INC, ATTENTION: ANGELA ALLISON, 800 WALNUT STREET, DES MOINES, IA, 50309 served 6-18-07	US Mail (1st Class)
24847	WINEGARDEN HALEY LINDHOLM & ROBERTSON PLC, ATTN: DENNIS M HALEY, G-9460 S SAGINAW STREET, SUITE A, GRAND BLANC, MI, 48439 served 6-18-07	US Mail (1st Class)
24799	XEROX, 800 LONG RIDGE RD, STAMFORD, CT, 06902-1227	US Mail (1st Class)
24847	YALE FINANCIAL SERVICES, PO BOX 3083, CEDAR RAPIDS, IA, 52406 served 6-18-07	US Mail (1st Class)

Subtotal for this group: 241

# **EXHIBIT B**

Svc Lst	Name and Address of Served Party	Mode of Service
24846	2 HEARTS CAR DETAILING, 130 HOLLY DR., MANSFIELD, OH, 44903	US Mail (1st Class)
24846	2002 MAPICS USER CONFERENCE, 427 UNIVERSITY AVE., NORWOOD, MA, 02062-2636	US Mail (1st Class)
24846	21ST CENTURY NEWSPAPER, 1551 E LINCOLN STE 142, MADISON HEIGHTS, MI, 48071-4159	US Mail (1st Class)
24846	28TH DISTRICT COURT, 14720 REAUME PARKWAY, SOUTHGATE, MI, 48195	US Mail (1st Class)
24846	2ULTI-MATIX, HWY 7 SOUTH, PO BOX 38, CENTERVILLE, AR, 72829	US Mail (1st Class)
24846	3D POLYMERS, INC., 1070 LIVERNOIS, TROY, MI, 48083	US Mail (1st Class)
24846	3M COMPANY, 3M CENTER, SAINT PAUL, MN, 55144-1000	US Mail (1st Class)
24846	3P SOLUTIONS LLC, 429 PROVIDENCE STREET, WOONSOCKET, RI, 02895	US Mail (1st Class)
24846	4-H JUNIOR FAIR AWARDS, 119 SOUTH BARRON STREET, EATON, OH, 45320	US Mail (1st Class)
24846	4IMPRINT, 101 COMMERCE ST, OSHKOSH, WI, 54901-4864	US Mail (1st Class)
24846	7TH AVENUE MANAGEMENT, 100 SEVENTH AVE., SUITE 150, CHARDON, OH, 44024	US Mail (1st Class)
24846	A A EXECUTIVE CATERING, INC., 5211 W. TUSCARAWAS STREET, CANTON, OH, 44708	US Mail (1st Class)
24846	A C WILSON ENTERPRISES, INC., 15035 BERSHIRE INDUSTRIAL PKY, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	A G VONK CONSULTING SERVICES, 9040 BALDWIN STREET, ASHBURN, ON, L0B 1A0CANADA	US Mail (1st Class)
24846	A I M, THE CARMICHAEL CENTER, 4001 CARMICHAEL RD, SUITE 115, MONTGOMERY, AL, 36106	US Mail (1st Class)
24846	A P TOOL & GAUGE, 3005 DEZIEL DRIVE, WINDSOR, ON, N8W 5A5CANADA	US Mail (1st Class)
24846	A PERFECT SORT, 39675 SOUTHPOINTE, HARRISON TWP., MI, 48045	US Mail (1st Class)
24846	A S B INDUSTRIES, INC., 1031 LAMBERT STREET, BARBERTON, OH, 44203	US Mail (1st Class)
24846	A S T M, 100 BARR HARBOR DR, W. CONSHOHECKEN, PA, 19428	US Mail (1st Class)
24846	A SCHULMAN, 3550 W MARKET ST, FAIRLAWN, OH, 44333-2658	US Mail (1st Class)
24846	A SCHULMAN INC., 3550 W MARKET ST, AKRON, OH, 44333	US Mail (1st Class)
24846	A T & T, ACCT: 216 T56-3796 210 8; 330 454 2890 772 3;, AND 330 456 2481 833 3, N17W24300 RIVERWOOD, WAUKESHA, WI, 53187	US Mail (1st Class)
24846	A T & T, ACCT: 810 R01 5129 015 7; 171 788 0806 605;, 810 664 7212 029 6; AND 030 487 8410 001, N17W24300 RIVERWOOD, WAUKESHA, WI, 53187	US Mail (1st Class)
24846	A TO Z TOTAL HEATING & COOLING, 24614 VAN BORN RD., DEARBORN HEIGHTS, MI, 48125	US Mail (1st Class)
24846	A&A MFG. CO., INC., 2300 S CALHOUN RD, NEW BERLIN, WI, 53151-2708	US Mail (1st Class)
24846	A&M ASSOCIATES, 1525 CORPORATE WOODS PKWY, STE 100, UNIONTOWN, OH, 44685	US Mail (1st Class)
24846	A&M FIRE AND SAFETY EQUIP. INC., 715 COTTAGE ST, ASHLAND, OH, 44805	US Mail (1st Class)
24846	A&R PACKAGING &, DISTRIBUTION SERVICES, 37386 EAGLE WAY, CHICAGO, IL, 60678-1373	US Mail (1st Class)
24846	A&R TRANSPORT, INC., 8440 TABLER RD., MORRIS, IL, 60450	US Mail (1st Class)
24846	A.A.A. BASKETBALL, ATTN: MIKE ELLIOTT, 317 FENTON AVE. N.E., CANTON, OH, 44704	US Mail (1st Class)
24846	A.O. KINZER AND ASSOCIATES, 35 LATOUR WAY, GREER, SC, 29650-4553	US Mail (1st Class)
24846	A.T. INFORMATION SYSTEM, 575 CORPORATE DR, MANWAH, NJ, 07430	US Mail (1st Class)
24846	A.T.C.O. INDUSTRIES INC., 7200 FIFTEEN MILE RD., STERLING HEIGHTS, MI, 48312-4524	US Mail (1st Class)
24846	AAA LAND SURVEYING INC., 100 CENTER STREET, SUITE 150, CHARDON, OH, 44024	US Mail (1st Class)
24846	AALBERS TOOL & MOLD, INC., 1525 MORO DRIVE, OLDCASTLE, ON, NOR 1L0CANADA	US Mail (1st Class)
24846	AARDVARK CONTROL EQUIPMENT, 19571 PROGRESS DR., STRONGSVILLE, OH, 44136	US Mail (1st Class)
24846	AAR-KEL MOULDS LTD., 17 ELM, WALLACEBURG, ON, N8A 1A1CANADA	US Mail (1st Class)
24846	AARON HAYCOOK, 900 S WILEY STREET, PO BOX 320, CRESTLINE, OH, 44827-0320	US Mail (1st Class)
24846	ABANAKI CORP., 17387 MUNN RD, CHAGRIN FALLS, OH, 44023	US Mail (1st Class)
24846	ABAQUS GREAT LAKES, INC., 14500 SHELDON ROAD, SUITE 160, PLYMOUTH, MI, 48170-2408	US Mail (1st Class)
24846	ABBEON CAL, INC., 123-303F GRAY AVE., SANTA BARBARA, CA, 93101-1809	US Mail (1st Class)
24846	ABBOTT VALVE & FITTING, 6090 COCHRAN RD, SOLON, OH, 44139-3304	US Mail (1st Class)
24846	ABC CLIMATE CONTROL SYSTEMS INC., 54 BETHRIDGE ROAD, TORONTO, ON, M9W 1N1CANADA	US Mail (1st Class)
24846	ABC DIV-LAW ENFORCEMENT OFC, 6050 MIMOSA CIR, TUSCALOOSA, AL, 35405-6525	US Mail (1st Class)
24846	ABC FASTENER GROUP, INC., ALMA BOLT COMPANY, 925 EAST SUPERIOR ST., ALMA, MI, 48801-0099	US Mail (1st Class)
24846	ABCLEANING, LLC, 32710 SCHOOLCRAFT RD., SUITE 100, LIVONIA, MI, 48150	US Mail (1st Class)
24846	ABER'S TOWING & CRANE / TRUCK CTR., 1722 CLAREMONT AVE., ASHLAND, OH, 44805	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	ABF, G 1315 W HILL RD., FLINT, MI, 48507	US Mail (1st Class)
24846	ABF FREIGHT SYSTEM INC., 1321 GAIL BORDEN PLACE, EL PASO, TX, 79935-5301	US Mail (1st Class)
24846	ABILITY FASTENERS, INC., 685 FARGO AVENUE, ELK GROVE VILLAGE, IL, 60007	US Mail (1st Class)
24846	ABRASIVE SUPPLY, 25240 STATE ROUTE 172, MINERVA, OH, 44657	US Mail (1st Class)
24846	ABSOLUTE GAGE CONTROL, 12936 E STONECREEK DRIVE, PICKERINGTON, OH, 43147	US Mail (1st Class)
24846	ABSOLUTE GUTTERS, 21800 MARTINSVILLE RD., BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	ABSORB TECH, 3900 W WILLIAM RICHARDSON DR., SOUTH BEND, IN, 46628	US Mail (1st Class)
24846	ACCU CHECK INSTRUMENT SERVICE, 3160 WEST FAIR AVENUE, LANCASTER, OH, 43130	US Mail (1st Class)
24846	ACCU-MEASURE, 4069A ROUND BOTTOM RD, CINCINNATI, OH, 45244	US Mail (1st Class)
24846	ACCURATE DOOR SYSTEMS, 6331 WHIPPLE AVE NW, NORTH CANTON, OH, 44720	US Mail (1st Class)
24846	ACCU-TECH INC, 43718 UTICA ROAD, STERLING HEIGHTS, MI, 48314	US Mail (1st Class)
24846	ACE INDUSTRIAL PRODUCTS, ASSOCIATED EQUIPMENT CORP., PO BOX 7356, ST. LOUIS, MO, 63177-7356	US Mail (1st Class)
24846	ACE LOADING DOCK, 5996 VERNON, DEARBORN HEIGHTS, MI, 48127	US Mail (1st Class)
24846	ACE MACHINE SHOP, 137 EAST PEEBLES, MEMPHIS, TN, 38109	US Mail (1st Class)
24846	ACE POLYETHYLENE BAG CO., 9801 WALFORD AVE, CLEVELAND, OH, 44102	US Mail (1st Class)
24846	ACI PLASTICS INC., 2945 DAVISON ROAD, FLINT, MI, 48506	US Mail (1st Class)
24846	ACM PLASTIC PRODUCTS, 507 ST. JOSEPH STREET, PO BOX 580, STURGIS, MI, 49091-0580	US Mail (1st Class)
24846	ACOUSTISEAL INC., 8485 PROSPECT AVE, KANSAS CITY, MO, 64132-2376	US Mail (1st Class)
24846	ACROLAB INSTRUMENT LTD, 1423 FRANKLIN STREET, UNIT F 143, DETROIT, MI, 48207	US Mail (1st Class)
24846	ACROTECH MIDWEST INC., ACR-1301, PO BOX 1683, MINNEAPOLIS, MN, 55480-1683	US Mail (1st Class)
24846	ACRYLON PLASTICS, 122 PAQUIN ROAD, WINNIPEG, MB, R2J 3V4CANADA	US Mail (1st Class)
24846	ACS CUSTOMER SERVICE GROUP, 140 RODDY AVE., SOUTH ATTLEBORO, MA, 02703	US Mail (1st Class)
24846	ACTCO METROLOGY SERVICES, 202 WESTVIEW DRIVE, MEADVILLE, PA, 16335	US Mail (1st Class)
24846	ACTEK MFG & ENGINEERING INC., 17181 GALE AVE. UNIT C, CITY OF INDUSTRY, CA, 91745	US Mail (1st Class)
24846	ACTIFY, 60 SPEAR STREET, SAN FRANCISCO, CA, 94105	US Mail (1st Class)
24846	ACTION ENVIRONMENTAL, 14830 AL HWY 91, HANCEVILLE, AL, 35077	US Mail (1st Class)
24846	ACTION MANAGEMENT SERVICE, 6055 ROCKSIDE WOODS BLVD, CLEVELAND, OH, 44131	US Mail (1st Class)
24846	ACTION PACKAGING, 425 36TH STREET SW, GRAND RAPIDS, MI, 49548	US Mail (1st Class)
24846	ACTIVE AERO CHARTER, 2068 E ST, BELLEVILLE, MI, 48111-1262	US Mail (1st Class)
24846	AD VISIBLES, INC., 36420 CHURCHILL DR, SOLON, OH, 44139-2658	US Mail (1st Class)
24846	ADA SUPPLY CO., 239 S EDGEWOOD AVE., HAMILTON, OH, 45013	US Mail (1st Class)
24846	ADDIS INC., 3721 LEE ROAD, CLEVELAND, OH, 44120	US Mail (1st Class)
24846	ADELL COMPOUNDING, 4530 ANNAPOLIS ROAD, BALTIMORE, MD, 21227	US Mail (1st Class)
24846	ADELPHIA, 3300 LAKESIDE AVE, CLEVELAND, OH, 44111	US Mail (1st Class)
24846	ADHESIVE SERVICES, INC., 5310 W 161ST ST, BROOK PARK, OH, 44142-1632	US Mail (1st Class)
24846	ADT SECURITY SERVICES, 1400 EAST AVIS DR., MADISON HEIGHTS, MI, 48071	US Mail (1st Class)
24846	ADVANCE INDUSTRIES, 7523 S STATE RD # 338, GOODRICH, MI, 48438-9219	US Mail (1st Class)
24846	ADVANCE MOLD COMPANY, 12690 STOUT STREET, DETROIT, MI, 48223-3323	US Mail (1st Class)
24846	ADVANCED AUTO CARE & TIRE, 10625 BELLEVILLE RD., BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	ADVANCED BLOW MOLDING SERVICES, 12955 FAIRVIEW RD SE, NEWARK, OH, 43056	US Mail (1st Class)
24846	ADVANCED CALIBRATION SERVICES, 24170 SHERWOOD, CENTER LINE, MI, 48015	US Mail (1st Class)
24846	ADVANCED CARPET CLEANING, 2390 VENEZIA DR, DAVISON, MI, 48423-8774	US Mail (1st Class)
24846	ADVANCED COMPOSITES, SIDNEY OHIO PLANT, PO BOX 633895, CINCINNATI, OH, 45263-3895	US Mail (1st Class)
24846	ADVANCED COMPOSITES, CORPORATE OFFICE, 1062 S 4TH AVE, SIDNEY, OH, 45365-8977	US Mail (1st Class)
24846	ADVANCED COMPOSITES, 1062 FOURTH AVE., BOX 747, SIDNEY, OH, 45365-0747	US Mail (1st Class)
24846	ADVANCED CONTROLS, INC., PO BOX 310, 15229 S. STATE ST., MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	ADVANCED MACHINERY, 4530 WADSWORTH RD, DAYTON, OH, 45414	US Mail (1st Class)
24846	ADVANCED MANUFACTURING, 3110 SEXTON ROAD, DECATUR, AL, 35603	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	ADVANCED MATERIALS, INC, 3303 LEE PKWY STE 105, DALLAS, TX, 75219-5116	US Mail (1st Class)
24846	ADVANCED PLASTIC &, MATERIAL TESTING INC., 42 DUTCH MILL ROAD, ITHACA, NY, 14850	US Mail (1st Class)
24846	ADVANCED PLASTIC SYSTEMS, 733 WINTER PARK DRIVE, MARS, PA, 16046	US Mail (1st Class)
24846	ADVANCED SATELLITE, 9282 GENERAL DRIVE, PLYMOUTH, MI, 48170	US Mail (1st Class)
24846	ADVANCED TOOL & SUPPLY, 4530 WADSWORTH RD., DAYTON, OH, 45414	US Mail (1st Class)
24846	ADVANTAGE, 700 ENERGY CENTER BLVD, SUITE 405, NORTHPORT, AL, 35473	US Mail (1st Class)
24846	ADVANTAGE ENG INC., 2461 RELIABLE PKWY, CHICAGO, IL, 60686-0024	US Mail (1st Class)
24846	ADVANTAGE MARKETING INCORP., 221 CHURCH STREET, SUITE 5, ASHLAND, OH, 44805	US Mail (1st Class)
24846	AEA TECHNOLOGY, ENGINEERING SOFTWARE, INC., OMEGA CORPORATE CENTER, PITTSURGH, PA, 15205	US Mail (1st Class)
24846	AEARO COMPANY/AOSAFETY, 2519 RELIABLE PARKWAY, CHICAGO, IL, 60686-0025	US Mail (1st Class)
24846	AEC INC, C/O AL BELROSE CO., 4835 CASCADE RD. SE, GRAND RAPIDS, MI, 49546	US Mail (1st Class)
24846	AEC WHITLOCK, 801 AEC DR, WOOD DALE, IL, 60191-1123	US Mail (1st Class)
24846	AERIAL VIEWS, 39 NORTH PLATT STREET, ALBION, NY, 14411	US Mail (1st Class)
24846	AERO CHEM INC., 741 MC CLURG ROAD, YOUNGSTOWN, OH, 44512	US Mail (1st Class)
24846	AERO COMPOSITES, 3400 SPANGLER ROAD, PO BOX 404, MEDWAY, OH, 45341	US Mail (1st Class)
24846	AERO MOTIVE COMPANY, 3 PARKWAY N STE 550, DEERFIELD, IL, 60015-2529	US Mail (1st Class)
24846	AEROQUIP AUTOMOTIVE OPERATIONS, PO BOX 12 THORNS ROAD, BRIERLY HILL, WEST MIDLANDS, DY52LBUNITED KINGDOM	US Mail (1st Class)
24846	AEROTEK INC., ATTN: A/P, PO BOX 591, HANOVER, MD, 21076	US Mail (1st Class)
24846	AETNA INC., AETNA-MIDDLETOWN, PO BOX 70966, CHICAGO, IL, 60673-0966	US Mail (1st Class)
24846	AETNA PLASTICS CORP., 1702 SAINT CLAIR AVE NE, CLEVELAND, OH, 44114-2008	US Mail (1st Class)
24846	AFC TOOL COMPANY, INC., 4900 WEBSTER ST, DAYTON, OH, 45414-4831	US Mail (1st Class)
24846	AFLAC, WORLDWIDE HEADQUARTER, COLUMBUS, GA, 31999	US Mail (1st Class)
24846	AGGREKO, INC., ATTN: REGIONAL LOCKBOX TX1-0006, 14800 ERYE ROAD, FORT WORTH, TX, 76155	US Mail (1st Class)
24846	AGIE LTD, 9009 PERIMETER WOODS DR STE G, CHARLOTTE, NC, 28216-0041	US Mail (1st Class)
24846	AGILITY, INC., 7000 FITZWATER RD, SUITE 4, BRECKSVILLE, OH, 44141	US Mail (1st Class)
24846	AGRI-INDUSTRIAL PLASTICS CO., 301 N 22ND STREET, FAIRFIELD, IA, 52556	US Mail (1st Class)
24846	AIAG, 26200 LAHSER RD STE 200, SOUTHFIELD, MI, 48033-7156	US Mail (1st Class)
24846	AIC EQUIPMENT & CONTROLS INC., 51760 GRAND RIVER AVE., WIXOM, MI, 48393	US Mail (1st Class)
24846	AIC VENTURES, WACHOVIA TEMPLATE, ADDRESS UNAVAILABLE,	US Mail (1st Class)
24846	AIM FOR THE PIN GOLF LEAGUE, C/O ERIC HEIN, AFTP TREASURER, 959 SCHANTZ AVE., DAYTON, OH, 45419	US Mail (1st Class)
24846	AIM NATIONAL LEASE, 1500 TRUMBULL RD, GIRARD, OH, 44420	US Mail (1st Class)
24846	AIN PLASTICS, 1750 E HEIGHTS ST, MADISON HEIGHTS, MI, 48071-4235	US Mail (1st Class)
24846	AIR ENERGY PRODUCTS CO., 5562 PLEASANT VIEW RD, MEMPHIS, TN, 38134-6503	US Mail (1st Class)
24846	AIR LIQUIDE AMERICA L.P., 2700 POST OAK BLVD STE 1800, HOUSTON, TX, 77056-5797	US Mail (1st Class)
24846	AIR POWER OF OHIO, 6607 CHITTENDEN RD, HUDSON, OH, 44236	US Mail (1st Class)
24846	AIR TECHNOLOGIES INC., 1900 JETWAY BLVD, COLUMBUS, OH, 43219-1681	US Mail (1st Class)
24846	AIRBORNE EXPRESS, DHL EXPRESS, 1200 S PINE ISLAND RD STE 600, PLANTATION, FL, 33324-4465	US Mail (1st Class)
24846	AIRCONNECT, 35253 W WARREN, WESTLAND, MI, 48185	US Mail (1st Class)
24846	AIRGAS SAFETY INC., W3645 AIRGAS SAFETY, PO BOX 7777, PHILADELPHIA, PA, 19175-3645	US Mail (1st Class)
24846	AIR-O-DYNAMIX, 320 FRANKLIN DR, DOYLESTOWN, OH, 44230-1533	US Mail (1st Class)
24846	AIRSPECIALISTS INC., 598 KIRK RD NW, MARIETTA, GA, 30060-1046	US Mail (1st Class)
24846	AIRTECH CONTROLS COMPANY, 46919 LIBERTY DR, PO BOX 930345, WIXOM, MI, 48393-0345	US Mail (1st Class)
24846	AITP, 401 N MICHIGAN AVE STE 2400, CHICAGO, IL, 60611-4227	US Mail (1st Class)
24846	AKCROS CHEMICALS AMERICA, 500 JERSEY AVE, NEW BRUNSWICK, NJ, 08901-3502	US Mail (1st Class)
24846	AKROCHEM CORPORATION, 255 FOUNTAIN ST., AKRON, OH, 44304	US Mail (1st Class)
24846	AKRON INCOME TAX, 1 CASCADE PLZ FL 11, AKRON, OH, 44308-1100	US Mail (1st Class)
24846	AKRON MUNICIPAL COURT AUDITOR, 217 S HIGH ST, AKRON, OH, 44308-1611	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	AKRON POLYMER LABORATORIES, 1480 ENGLEWOOD, AKRON, OH, 44305	US Mail (1st Class)
24846	AKRON QUALITY FEEDSCREWS, 399 COMMERCE STREET, TALLMADGE, OH, 44278-2134	US Mail (1st Class)
24846	AKRON RUBBER DEV. LAB.INC, 300 KENMORE BOULEVARD, AKRON, OH, 44301	US Mail (1st Class)
24846	AKRON TREASURER, 161 S HIGH ST, AKRON, OH, 44308-1602	US Mail (1st Class)
24846	A-LAB CORP., 3050 DRYDEN RD., DAYTON, OH, 45439	US Mail (1st Class)
24846	ALABAMA AUTOMOTIVE MFG. ASSOC., 500 BEACON PARKWAY WEST, BIRMINGHAM, AL, 35209	US Mail (1st Class)
24846	ALABAMA CONVEYOR COMPANY, 7103 JUNIPER RD., FAIRVIEW, TN, 37062	US Mail (1st Class)
24846	ALABAMA EMPLOYMENT LETTER, 5201 VIRGINIA WAY, BRENTWOOD, TN, 37024-5094	US Mail (1st Class)
24846	ALABAMA FLAG AND BANNER, 2315 BOB WALLACE AVE SW, HUNTSVILLE, AL, 35805-4714	US Mail (1st Class)
24846	ALABAMA MACHINE & WELDING, INC., PO BOX 801, FAIRFIELD, AL, 35064-0801	US Mail (1st Class)
24846	ALABAMA PRECISION MOLD, 14120 HWY 11 NORTH, COTTONDALE, AL, 35453	US Mail (1st Class)
24846	ALABAMA SLING CENTER, 1103 HOKE AVE, DOLOMITE, AL, 35061	US Mail (1st Class)
24846	ALADDIN'S CARPET CLEANING, 852 S MAPLE LEAF RD., LAPEER, MI, 48446	US Mail (1st Class)
24846	ALAMO IRON WORKS, INC., 943 AT AND T CENTER PKWY, SAN ANTONIO, TX, 78219-3107	US Mail (1st Class)
24846	ALAMO IRON WORKS, INC., 943 AT AND T CENTER PKWY, SAN ANTONIO, TX, 78219-3107	US Mail (1st Class)
24846	ALBAR INDUSTRIES, INC., 780 WHITNEY DR., LAPEER, MI, 48446-2570	US Mail (1st Class)
24846	ALBINO'S MEATS & CATERING, 2316 LAKE AVE, ASHTABULA, OH, 44004	US Mail (1st Class)
24846	ALBIS CORPORATION, (AKA) ALBIS PLASTICS CORP, 19901 SOUTHWEST FWY, SUGAR LAND, TX, 77479-6538	US Mail (1st Class)
24846	ALDON, 6001 SHELLMOUND STREET, SUITE 600, EMERYVILLE, CA, 94608-1901	US Mail (1st Class)
24846	ALESSANDRO ROMAGNOLI, 1865 EASTPORT ST., TROY, MI, 48083	US Mail (1st Class)
24846	ALEX PRODUCTS, INC., 19-911 ROAD T, PO BOX 26, RIDGEVILLE CORNERS, OH, 43555	US Mail (1st Class)
24846	ALEXANDER HAMILTON INST., 70 HILLTOP ROAD, RAMSEY, NJ, 07446-1119	US Mail (1st Class)
24846	ALI MED INC, 297 HIGH ST, DEDHAM, MA, 02026-2852	US Mail (1st Class)
24846	ALL AERIALS, LLC, 4945 BRECKSVILLE RD, RICHFIELD, OH, 44286	US Mail (1st Class)
24846	ALL CONTROL ENTERPRISES, INC., 1644 CAMBRIDGE DR., ELGIN, IL, 60123	US Mail (1st Class)
24846	ALL OHIO CRANE, INC., 2381 UNIT 5 LOCUST ST S, CANAL FULTON, OH, 44614	US Mail (1st Class)
24846	ALL PRO INC, 6770 5TH STREET, NORTHPORT, AL, 35476	US Mail (1st Class)
24846	ALL SIZE POLY BAGS, RKS PLASTICS INC., PO BOX 836, NEW BRUNSWICK, NJ, 08903	US Mail (1st Class)
24846	ALL STATE SALES &, ADMINISTRATIVE SERVICES, PO BOX 401, SYRACUSE, NY, 13209-0401	US Mail (1st Class)
	ALLEN L. HIBBLER, 8280 BOWERS ROAD, IMLAY CITY, MI, 48444	US Mail (1st Class)
24846	ALLEN PUMP COMPANY INC., PO BOX 1030, AKRON, OH, 44309-1030	US Mail (1st Class)
24846	ALLIANCE AUTOMATION SYSTEMS, 400 TRABOLD RD, ROCHESTER, NY, 14624-2529	US Mail (1st Class)
24846	ALLIANCE INDUSTRIES, INC., 15721 LEONE DRIVE, MACOMB TOWNSHIP, MI, 48042	US Mail (1st Class)
24846	ALLIED CAPITAL PARTNERS, L.P., PRESTONWOOD TOWER, 5151 BELT LINE RD STE 500, DALLAS, TX, 75254-7520	US Mail (1st Class)
24846	ALLIED POWER TRANSMISSION, 27000 RICHMOND ROAD, SUITE 3, SOLON, OH, 44139-1016	US Mail (1st Class)
24846	ALLIED SUPPLY COMPANY, 1100 E. MONUMENT AVE., DAYTON, OH, 45402	US Mail (1st Class)
24846	ALLIED TELEPHONE DIRECTORIES, 2000 CORPORATE SQUARE BLVD, JACKSONVILLE, FL, 32216-0326	US Mail (1st Class)
24846	ALLIED VAN LINES INC., 700 OAKMONT LN, WESTMONT, IL, 60559-5551	US Mail (1st Class)
24846	ALLIED WASTE SERVICE #23, 15880 N GREENWAY HAYDEN LOOP, SCOTTSDALE, AZ, 85260-1648	US Mail (1st Class)
24846	ALLIED WASTE SERVICES, 15880 N GREENWAY HAYDEN LOOP, SCOTTSDALE, AZ, 85260-1648	US Mail (1st Class)
24846	ALL-RITE INDUSTRIES, 470 OAKWOOD RD., LAKE ZURICH, IL, 60047	US Mail (1st Class)
24846	ALL-SOUTH SUBCONTRACTORS, INC., 2678 QUEENSTOWN RD, BIRMINGHAM, AL, 35210-5428	US Mail (1st Class)
24846	ALLTECH SUPPORT INC., 17527 CHATSWORTH STREET, GRANDHILLS, CA, 91344	US Mail (1st Class)
24846	ALMO MANIFOLD & TOOL, 777 AULERICH RD, EAST TAWAS, MI, 48730-9565	US Mail (1st Class)
24846	ALPHA COATINGS, INC, PO BOX 1006, FOSTORIA, OH, 44830-1006	US Mail (1st Class)
24846	ALPHA EXPRESS INC., 3275 KENT RD. SUITE 202, STOW, OH, 44224	US Mail (1st Class)
24846	ALPHABET GROUP, 4100 W 159TH ST, CLEVELAND, OH, 44135	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	ALPHA-OMEGA ASSEMBLY, & PACKAGING, INC., 8315 BROWN BEAR CIRCLE NW, MASSILLON, OH, 44646	US Mail (1st Class)
24846	ALTA GREENHOUSE & GARDEN, 935 SOUTH HOME ROAD, MANSFIELD, OH, 44906	US Mail (1st Class)
24846	ALTAIR ENGINEERING, INC., LOCKBOX #8847, 8847 RELIABLE PARKWAY, CHICAGO, IL, 60686-0088	US Mail (1st Class)
24846	ALTEMUS MD, PO BOX 1045, MIDDLEFIELD, OH, 44062-1045	US Mail (1st Class)
24846	ALTERNATIVE FLASH INC, 274 MAIN ST., WADSWORTH, OH, 44281	US Mail (1st Class)
24846	ALTERNATIVE RUBBER, 200 CREEKSIDE DRIVE, AMHERST, NY, 14228	US Mail (1st Class)
24846	ALTERNATIVE STAFFING, INC., PO BOX 890265, CHARLOTTE, NC, 28289-0265	US Mail (1st Class)
24846	ALVAN MOTOR FREIGHT INC., 3600 ALVAN ROAD, KALAMAZOO, MI, 49001	US Mail (1st Class)
24846	AMANDA HAUN, 1401 INDUSTRIAL DRIVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	AMANO BUSINESS CREDIT, 2081 SE OCEAN BLVD, STE 2A, STUART, FL, 34996	US Mail (1st Class)
24846	AMBASSADOR BANQUET & CONFERENCE CTR, 7794 PEACH STREET, ERIE, PA, 16509	US Mail (1st Class)
24846	AMBERTEK SYSTEMS, INC., 3533 OLD CONEJO RD STE 114, NEWBURY PARK, CA, 91320-6170	US Mail (1st Class)
24846	AMERI PAK, 62 BAODWIN, PONTIAC, MI, 48342	US Mail (1st Class)
24846	AMERICAN ALERT, 4640 NORTH RIDGE EAST, GENEVA, OH, 44041-0336	US Mail (1st Class)
24846	AMERICAN ASSOC LAB ACCRED, 5301 BUCKEYSTOWN PIKE, SUITE 350, FREDERICK, MD, 21704-8307	US Mail (1st Class)
24846	AMERICAN ASSOCIATION OF, INDUSTRIAL MGMT., STEARNS BLDG./SUITE 206, 293 BRIDGE STREET, SPRINGFIELD, MA, 01103	US Mail (1st Class)
24846	AMERICAN CANCER SOCIETY, 1599 CLIFTON RD., ATLANTA, GA, 30329	US Mail (1st Class)
24846	AMERICAN CANCER SOCIETY, 2353 S LINDEN RD., SUITE B, FLINT, MI, 48532	US Mail (1st Class)
24846	AMERICAN COMMODITIES INC., 2945 DAVISON RD., FLINT, MI, 48506	US Mail (1st Class)
24846	AMERICAN EXPRESS, ATTN: RPC, 300 RIVERSIDE PLAZA, CHICAGO, IL, 60606	US Mail (1st Class)
24846	AMERICAN EXPRESS, 200 VESEY STREET, 50TH FLOOR, NEW YORK, NY, 10285-1000	US Mail (1st Class)
24846	AMERICAN EXPRESS, SUITE 0005, CHICAGO, IL, 60679-0005	US Mail (1st Class)
24846	AMERICAN EXPRESS, TRAVEL RELATED SERVICES CO INC, PO BOX 360001, FORT LAUDERDALE, FL, 33336-0001	US Mail (1st Class)
24846	AMERICAN EXPRESS, SUITE 0003, CHICAGO, IL, 60679	US Mail (1st Class)
24846	AMERICAN FENCE CO., 6403 MCFARLAND BLVD. E, TUSCALOOSA, AL, 35405	US Mail (1st Class)
24846	AMERICAN FINANCIAL, REF NO. 337139, PO BOX 41601, PHILADELPHIA, PA, 19101-1601	US Mail (1st Class)
24846	AMERICAN HANDLING EQUIP, C/O BANK AMERICA, PO BOX 281772, ATLANTA, GA, 30384-1772	US Mail (1st Class)
24846	AMERICAN HEART ASSOCIATION, 2901 7TH STREET, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	AMERICAN HEAT TREATING, 1346 MORRIS AVE, DAYTON, OH, 45408	US Mail (1st Class)
24846	AMERICAN INTERNATIONAL, INC., 4080 LONYO, PO BOX 10098, DETROIT, MI, 48210	US Mail (1st Class)
24846	AMERICAN INTL GROUP CLAIM SERVICE, MORRIS CORPORATE CENTER III, 400 INTERPACE PARKWAY, BLDG D, 3RD FLOOR, PARSIPPANY, NJ, 07054	US Mail (1st Class)
24846	AMERICAN LEGION, 216 N. SELTZER, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	AMERICAN MANAGEMENT, 1601 BROADWAY, NEW YORK, NY, 10019-7434	US Mail (1st Class)
24846	AMERICAN MANAGEMENT ASSOC, 1601 BROADWAY, NEW YORK, NY, 10019-7434	US Mail (1st Class)
24846	AMERICAN METALWORKS, 3726 INPARK CIRCLE, DAYTON, OH, 45414	US Mail (1st Class)
24846	AMERICAN MFG./TOLEDO PLT.(GM), 2375 DORR STREET, TOLEDO, OH, 43607	US Mail (1st Class)
24846	AMERICAN PAYROLL ASSOCIATION, 660 N. MAIN AVE., SUITE 100, SAN ANTONIO, TX, 78205-1271	US Mail (1st Class)
24846	AMERICAN PERSONNEL, INC., 4119 WHIPPLE AVE NW, SUITE B, CANTON, OH, 44718	US Mail (1st Class)
24846	AMERICAN RED CROSS, 3747 EUCLID AVE., CLEVELAND, OH, 44115	US Mail (1st Class)
24846	AMERICAN RED CROSS, 1401 S GRAND TRAVERSE, FLINT, MI, 48503	US Mail (1st Class)
24846	AMERICAN RED CROSS, CRESTLINE AREA, 102 N THOMAN STREET, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	AMERICAN RED CROSS, ASHTABULA COUNTY CHAPTER, 433 CENTER STREET, ASHTABULA, OH, 44004	US Mail (1st Class)
24846	AMERICAN RED CROSS, WEST ALABAMA CHAPTER, TUSCALOOSA, AL, 35404	US Mail (1st Class)
24846	AMERICAN RED CROSS, 525 MARKET AVE., NORTH, CANTON, OH, 44702	US Mail (1st Class)
24846	AMERICAN RESEARCH KEMICALS, 3410 GRIFFIN RD, DAVIE, FL, 33329	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	AMERICAN ROLLER COMPANY, 135 S. LASALLE, DEPT. 3697, CHICAGO, IL, 60674-3697	US Mail (1st Class)
24846	AMERICAN RUBBER PRODUCTS CORP., 795 WURLITZER DRIVE, N TONAWANDA, NY, 14120	US Mail (1st Class)
24846	AMERICAN SAFETY & EQUIPMENT INC., 3268 ADVENTURE LANE, OXFORD, MI, 48371	US Mail (1st Class)
24846	AMERICAN SIGN & BANNER, 439 EAST MAIN STREET, CRESTLINE, OH, 44827-1119	US Mail (1st Class)
24846	AMERICAN SPEEDY PRINTING, 700 S. MAIN ST., SUITE 108, LAPEER, MI, 48446	US Mail (1st Class)
24846	AMERICAN STEEL RULE DIE, INC., 23900 REEDY DR., ELKHART, IN, 46514	US Mail (1st Class)
24846	AMERICAN SYNERGISTICS, INC., 35355 SCHOOLCRAFT ROAD, LIVONIA, MI, 48150	US Mail (1st Class)
24846	AMERICAN TOOL & ENGINEERING, 102 INDUSTRIAL PARKWAY, PO BOX 599, GREENE, IA, 50636-0599	US Mail (1st Class)
24846	AMERIGAS, 14900 DOWNEY ROAD, CAPEC, MI, 48014	US Mail (1st Class)
24846	AMERIMOLD INC., 595A E. WATERLOO ROAD, MOGADORE, OH, 44260	US Mail (1st Class)
24846	AMERIPOL SYNPOL COMPANY, DIV. OF UNIROYAL GOODRICH, PO BOX 95956, CHICAGO, IL, 60694	US Mail (1st Class)
24846	AMERITECH, ADVANCED DATA SERVICES, PO BOX 1838, SAGINAW, MI, 48605-1838	US Mail (1st Class)
24846	AMERITECH CREDIT CORP., DBA SBC CAPITAL SERVICES, PO BOX 71614, CHICAGO, IL, 60694	US Mail (1st Class)
24846	AMES DEPARTMENT STORE, 2418 MAIN ST, ROCKY HILL, CT, 06067-2550	US Mail (1st Class)
24846	AMPEX METAL PRODUCTS, 5581 W 164TH ST, CLEVELAND, OH, 44142	US Mail (1st Class)
24846	AMR, 50320 PATRICIA AVE., CHESTERFIELD, MI, 48051	US Mail (1st Class)
24846	AMSPEC CHEMICAL CORP., 751 WATER STREET, GLOUCESTER CITY, NJ, 08030	US Mail (1st Class)
24846	AMSTERDAM PRINTING & LITHO, 166 WALLINS CORNERS RD, AMSTERDAM, NY, 12010-1817	US Mail (1st Class)
24846	AMY TOPPER, 3515 MASSILON ROAD, SUITE 320, UNIONTOWN, OH, 44685	US Mail (1st Class)
24846	AMYE HOUSE, 1401 INDUSTRIAL DRIVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	ANALYST INTERNATIONAL, 3252 UNIVERSITY DR., SUITE 200, AUBURN HILLS, MI, 48326	US Mail (1st Class)
24846	ANALYSTS INTERNATIONAL, SEQUOIA SERVICES GROUP, 36266 TREASURY CENTER, CHICAGO, IL, 60694-6200	US Mail (1st Class)
24846	ANASTASIA SIMMONS, ADDRESS UNAVAILABLE, CHARDON, OH, 44024	US Mail (1st Class)
24846	ANDERSON BOLDS, 24050 COMMERCE PARK RD, CLEVELAND, OH, 44122	US Mail (1st Class)
24846	ANDERSON TOOL & EQUIPMENT, 7901 NORTH LAPEER RD., FOSTORIA, MI, 48435	US Mail (1st Class)
24846	ANDERSON TRANSFER, 922 S GUIGNARD DRIVE, SUMTER, SC, 29150	US Mail (1st Class)
24846	ANKARA INDUSTRIES, INC., 56359 NORTH BAY DRIVE, CHESTERFIELD, MI, 48051	US Mail (1st Class)
24846	ANNE BROECKER, 5650 HASCO ROAD, MILLINGTON, MI, 48746	US Mail (1st Class)
24846	ANNETTA HALL, 3131 COLUMBUS RD NE, CANTON, OH, 44705	US Mail (1st Class)
24846	ANNIE DANIELS, 1401 INDUSTRIAL PK DR., TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	ANTHEM BLUE CROSS, 120 MONUMENT CIR, INDIANAPOLIS, IN, 46204-4906	US Mail (1st Class)
24846	ANTHEM LIFE INSURANCE, COMPANY OF INDIANA, DEPARTMENT L-880, COLUMBUS, OH, 43260-0880	US Mail (1st Class)
24846	ANTHONY CLAYTON, 2324 REFSET DRIVE, JANESVILLE, WI, 53545	US Mail (1st Class)
24846	ANTHONY G. STURM, 9000 MURPHY ROAD, LYONS, MI, 48851	US Mail (1st Class)
24846	ANTHONY J. MURRU, 3560 WEST MARKET STREET, SUITE 340, AKRON, OH, 44333	US Mail (1st Class)
24846	ANWC CAR, 1155 GRATIOT AVE, DETROIT, MI, 48207-2732	US Mail (1st Class)
24846	ANXEBUSINESS CORP, 2000 TOWN CENTER, SUITE 2050, SOUTHFIELD, MI, 48075	US Mail (1st Class)
24846	APA GREATER CLEVELAND CHAPTER, ATTN: MARTHA DIAMOND, PO BOX 1753, MENTOR, OH, 44061	US Mail (1st Class)
24846	APEX PLASTICS, INC., 3305 NASSAU STREET WEST, EAST CANTON, OH, 44730	US Mail (1st Class)
24846	APICS, 5301 SHAWNEE RD, ALEXANDRIA, VA, 22312-2312	US Mail (1st Class)
24846	A-PLUS LAWN CARE, 308 W HIGH STREET, PO BOX 253, ANSONIA, OH, 45303	US Mail (1st Class)
24846	APPLICATION SYSTEMS USER GROUP, PO BOX 31353, CLEVELAND, OH, 44131	US Mail (1st Class)
24846	APPLIED ERGONOMICS, 5070 COVINGTON HWY, DECATUR, GA, 30035	US Mail (1st Class)
24846	APPLIED HANDLING, 5359 DIXIE HWY, SAGINAW, MI, 48601-5599	US Mail (1st Class)
24846	APPLIED HANDLING INC., 15200 CENTURY DRIVE, PO BOX 217, DEARBORN, MI, 48121	US Mail (1st Class)
24846	APPLIED TECHNOLOGIES & RESOURC, 587 MCCORMICK DR, LAPEER, MI, 48446-2575	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	AQP, ACCOUNTS RECEIVABLE, PO BOX 555, MILWAUKEE, WI, 53201-0555	US Mail (1st Class)
24846	AQUA TECH ENVIROMENTAL LABS, 1776 MARION WALDO RD, MARION, OH, 43302-7428	US Mail (1st Class)
24846	AQUAPERFECT INC., 5684 W 74TH ST, INDIANAPOLIS, IN, 46278-1752	US Mail (1st Class)
24846	AQUA-TECH ENVIRONMENTAL, INC., PO BOX 1886, COLUMBUS, OH, 43271-0001	US Mail (1st Class)
24846	ARAMARK, 3900 WILDLIFE WAY, CLEVELAND, OH, 44109	US Mail (1st Class)
24846	ARAMARK UNIFORM SERVICES, 115 N 1ST ST, BURBANK, CA, 91502-1856	US Mail (1st Class)
24846	ARC INC., COMMERCE SQUARE, 210 WALNUT ST. ROOM 46, LOCKPORT, NY, 14094	US Mail (1st Class)
24846	ARC RESIN CORP., 1500 GREEN ST, MARIETTA, OH, 45750	US Mail (1st Class)
24846	ARCH, PO BOX 4330, CAROL STREAM, IL, 60197-4330	US Mail (1st Class)
24846	ARCH PAGING, 6677 RICHMOND HWY, ALEXANDRIA, VA, 22306-6647	US Mail (1st Class)
24846	ARCON EQUIPMENT INC., 20468 KRICK RD WEST, WALTON HILLS, OH, 44146	US Mail (1st Class)
24846	ARD TRUCKING CO., INC., PO DRAWER 504, DARLINGTON, SC, 29540	US Mail (1st Class)
24846	ARDEN FASTENERS, INC., A PARK OHIO COMPANY, PO BOX 72370, CLEVELAND, OH, 44192-0370	US Mail (1st Class)
24846	ARDEN NASON, 2715 GREENACRE DRIVE, ERIE, PA, 16506	US Mail (1st Class)
24846	ARGENT, 6 FOREST PARK DR, FARMINGTON, CT, 06032-1480	US Mail (1st Class)
24846	ARGON TOOL & MFG., 32309 MILTON AVENUE, MADISON HEIGHTS, MI, 48071	US Mail (1st Class)
24846	ARLEEN ORTIZ SANTIAGO, 1733 BLUE JAY CIRCLE, ASHTABULA, OH, 44004	US Mail (1st Class)
24846	ARMOLOY OF OHIO INC., 1950 E LEFFEL LANE, SPRINGFIELD, OH, 45505	US Mail (1st Class)
24846	ARMOTEC MATERIALS CORP., 3825 W. 150TH ST., CLEVELAND, OH, 44111	US Mail (1st Class)
24846	ARNE O. SANDNESS, 10047 KEYSTONE DRIVE, LAKE CITY, PA, 16423	US Mail (1st Class)
24846	ARNTZ SAWING TECHNOLOGY, 320 INTERNATIONAL CIRCLE, SUMMERVILLE, SC, 29483	US Mail (1st Class)
24846	ARROW FREIGHT MANAGEMENT, 1721 ADDISON, EL PASO, TX, 79928	US Mail (1st Class)
24846	ARROWHEAD INDUSTRIES CANADA, 4015 BRODEUR, SHERBROOKE, QC, J1L 1K4CANADA	US Mail (1st Class)
24846	ART IRON INC., 860 CURTIS ST, TOLEDO, OH, 43609-2304	US Mail (1st Class)
24846	ARTHUR WRIGHT & ASSOC., 12205 OLD BIG BEND RD., KIRKWOOD, MO, 63122-6803	US Mail (1st Class)
24846	ARTI-GRAF, JOSE ELEUTERIO GZZ NO. 681, SANTA CATARINA, 66350MEXICO	US Mail (1st Class)
24846	ARTX LTD, 1770 WLEXINGTON AVE., CINCINNATI, OH, 45212	US Mail (1st Class)
24846	ASA SALES, INC., 2501 A WEST 237TH STREET, TORRANCE, CA, 90505	US Mail (1st Class)
24846	ASAHI THERMOFIL, INC., DEPT. 77600, PO BOX 77000, DETROIT, MI, 48277	US Mail (1st Class)
24846	ASAHI THERMOFIL, INC., ONE THERMOFIL WAY, FOWLERVILLE, MI, 48836	US Mail (1st Class)
24846	ASAP SYSTEMS, 819 1/2 PACIFIC AVE. #2, SANTA CRUZ, CA, 95060	US Mail (1st Class)
24846	ASSE, 2122 FIFTEEN MILE RD, SUITE F, STERLING HEIGHTS, MI, 48310	US Mail (1st Class)
24846	ASCOM HASLER LEASING, PNC BANK, LOCKBOX 802585, STE 105, 160 HANSEN CT, WOOD DALE, IL, 60191	US Mail (1st Class)
24846	ASE EDUCATIONAL FOUNDATION, 23815 NORTHWESTERN HIGHWAY, SOUTHFIELD, MI, 48075	US Mail (1st Class)
24846	ASE-EDUCATIONAL FOUNDATION, DEPARTMENT 238901, PO BOX 67000, DETROIT, MI, 48267-2389	US Mail (1st Class)
24846	ASHLAND CHEMICAL, 5200 BLAZER PKWY, DUBLIN, OH, 43017-5309	US Mail (1st Class)
24846	ASHLAND CHEMICAL CO., GENERAL POLYMERS DIV., 12001 TOEPFER RD., WARREN, MI, 48089	US Mail (1st Class)
24846	ASHLAND INCORPORATED, COLLECTION DEPT DS3, PO BOX 2219, COLUMBUS, OH, 43216	US Mail (1st Class)
24846	ASHLAND PUBLISHING, L.L.C., 40 EAST SECOND STREET, ASHLAND, OH, 44805	US Mail (1st Class)
24846	ASHLAND, INC, 50 E RIVERCENTER BLVD, COVINGTON, KY, 41011-1683	US Mail (1st Class)
24846	ASHTON AUTOMOTIVE TESTING INC., 2100 CENTERWOOD DRIVE, WARREN, MI, 48091	US Mail (1st Class)
24846	ASI CHARITY GOLF, AMERICAN SHOWA, INC., 707 WEST CHERRY STREET, SUNBURY, OH, 43074-6595	US Mail (1st Class)
24846	ASQ, 600 N PLANKINTON AVE, MILWAUKEE, WI, 53203-2914	US Mail (1st Class)
24846	ASQ, 600 N PLANKINTON AVE, MILWAUKEE, WI, 53203	US Mail (1st Class)
24846	ASQ SECTION 0812/, BARD ENDOSCOPIC TECHNOLOGIES, 6091 HEISLEY RD., MENTOR, OH, 44060	US Mail (1st Class)
24846	ASQC, 600 N PLANKINTON AVE, MILWAUKEE, WI, 53203-2914	US Mail (1st Class)
24846	ASSOCIATED BAG COMPANY, P.O BOX 3036, MILWAUKEE, WI, 53201-3036	US Mail (1st Class)
24846	ASSURANCE PACKAGING & MFG., 2005 LIBERTY AVE, LAWRENCEBURG, TN, 38464-4604	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	ASTRO NETICS, 1780 E 14 MILE ROAD, MADISON HGTS., MI, 48071	US Mail (1st Class)
24846	AT SALES PROMOTIONS, 146 COUNTRYVIEW DR, HARRISON, OH, 45030	US Mail (1st Class)
24846	AT&T, ACCT: 050 402 8290 001, ATTN: EXECUTIVE OFFICE 220 N MERIDIAN ST 9 RM 861, INDIANAPOLIS, IN, 46204	US Mail (1st Class)
24846	AT&T, ACCT: 734 367 1400 800 7; 734 762 0831 485 D;, AND 734 261 5509 781 5, N17W24300 RIVERWOOD, WAUKESHA, WI, 53187	US Mail (1st Class)
24846	AT&T, ONE AT&T WAY, BEDMINSTER, NJ, 07921-2693	US Mail (1st Class)
24846	AT&T, ACCT: 051 484 7303 001, ATTN: EXECUTIVE OFFICE 220 N MERIDIAN ST 9 RM 861, INDIANAPOLIS, IN, 46204	US Mail (1st Class)
24846	AT&T 22, ACCT: 030 515 2449 001, ATTN: EXECUTIVE OFFICER, 220 N MERIDIAN ST 9, RM 861, INDIANAOPOLIS, IN, 46204	US Mail (1st Class)
24846	AT&T CORP, 1355 UNIVERSITY DR, MESA, AZ, 85201	US Mail (1st Class)
24846	AT&T ONENET SERVICE, ACCT: 1000 682 2851, ONE AT&T WAY, BEDMINSTER, NJ, 07921-2693	US Mail (1st Class)
24846	AT&T TELECONFERENCE SERVICES, ACCT: 31446040-00011, ONE AT&T WAY, BEDMINSTER, NJ, 07921-2693	US Mail (1st Class)
24846	AT&T UNIVERSAL BILLER, ACCT: 03100 18101, ONE AT&T WAY, BEDMINSTER, NJ, 07921-2693	US Mail (1st Class)
24846	AT&T UNIVERSAL BILLER, ACCT: 171 788 0806 605; 171 788 0917 511, ONE AT&T WAY, BEDMINSTER, NJ, 07921-2693	US Mail (1st Class)
24846	AT&T UNIVERSAL BILLER, ACCT: 171 788 0894 266, ONE AT&T WAY, BEDMINSTER, NJ, 07921-2693	US Mail (1st Class)
24846	AT&T UNIVERSAL BILLER, ACCT: 171 788 0792 420; 171 788 0885 102, ONE AT&T WAY, BEDMINSTER, NJ, 07921-2693	US Mail (1st Class)
24846	AT&T UNIVERSAL BILLER, ACCT: 17 788 0801 540, ONE AT&T WAY, BEDMINSTER, NJ, 07921-2693	US Mail (1st Class)
24846	AT&T WIRELESS SERVICES, ONE AT&T WAY, BEDMINSTER, NJ, 07921-2693	US Mail (1st Class)
24846	ATHENA CONTROLS, INC., 5145 CAMPUS DRIVE, PLYMOUTH MEETING, PA, 19462-1129	US Mail (1st Class)
24846	ATHLETIC WORLD ADVERTISING, 3340 N COLLEGE AVE, FAYETTEVILLE, AR, 72703-3815	US Mail (1st Class)
24846	ATI INDUSTRIAL AUTO, 1031 GOODWORTH DRIVE, PINNACLE PARK, APAX, NC, 27539	US Mail (1st Class)
24846	ATI, INC., 100 ATI PARKWAY, BIRMINGHAM, AL, 35244	US Mail (1st Class)
24846	ATLAS FLUID COMPONENTS, INC., 135 E ASCOT LANE, PO BOX 4040, CUYAHOGA FALLS, OH, 44223-4050	US Mail (1st Class)
24846	ATLAS PEN & PENCIL CORP, 3040 NORTH 29TH AVE, HOLLYWOOD, FL, 33020	US Mail (1st Class)
24846	ATLAS VAN LINES, 2578 COLLECTIONS CENTER DR., CHICAGO, IL, 60693	US Mail (1st Class)
24846	ATLAS VAN LINES, 1212 ST GEORGE RD, PO BOX 509, EVANSVILLE, IN, 47703-0509	US Mail (1st Class)
24846	ATLAS WEATHERING SERVICES GRP., 45601 N 47TH AVE, PHOENIX, AZ, 85087-7042	US Mail (1st Class)
24846	ATOMIC PRINTING & IMAGING INC., 595 S MAIN STREET, PLYMOUTH, MI, 48170	US Mail (1st Class)
24846	ATTORNEY GENERAL OF OHIO, ACCT# 4601633, 150 E GAY STREET, 21ST FLOOR, COLUMBUS, OH, 43215	US Mail (1st Class)
24846	AUBURN FENCE CORPORATION, 18096 MUNN RD., CHAGRIN FALLS, OH, 44023	US Mail (1st Class)
24846	AUDIO VISUAL MEDIA, 1141 LEXINGTON AVE., PO BOX 3526, MANSFIELD, OH, 44907	US Mail (1st Class)
24846	AUGUST INDUSTRIAL SUPPLY, 5534 WEST LAKE RD., ERIE, PA, 16505	US Mail (1st Class)
24846	AURORA AIR PRODUCTS, INC., 231C EOLA ROAD, AURORA, IL, 60504	US Mail (1st Class)
24846	AURORA RYERSON, 290 MCCORMICK ST., LAPEER, MI, 48446	US Mail (1st Class)
24846	AUTO BUSINESS LTD., NATIONAL WESTMINSTER BANK, 52 HIGH ST., STAMFORD, LINCS, PE9 2BDUNITED KINGDOM	US Mail (1st Class)
24846	AUTO HYGIENE, 1492 STATE RTE. 598, GALION, OH, 44833	US Mail (1st Class)
24846	AUTO PALLETS - BOXES, INC., 28000 SOUTHFIELD, LATHRUP VILLAGE, MI, 48076	US Mail (1st Class)
24846	AUTO TECHNOLOGY CO., 20026 PROGRESS DRIVE, STRONGSVILLE, OH, 44149	US Mail (1st Class)
24846	AUTOMATED MACHINE, 10525 CHESTER RD STE A, CINCINNATI, OH, 45215-1254	US Mail (1st Class)
24846	AUTOMATED MAILING SOLUTIONS, 36921 SCHOOLCRAFT ROAD, LIVONIA, MI, 48150	US Mail (1st Class)
24846	AUTOMATIC DATA PROCESSING, 7007 E.PLEASANT VALLEY RD, CLEVELAND, OH, 44131	US Mail (1st Class)
24846	AUTOMATIC ICE & BEVERAGE INC., 1400 TUSCALOOSA AVE SW, BIRMINGHAM, AL, 35211-2639	US Mail (1st Class)
24846	AUTOMATION & MODULAR, 2250 N. OPDYKE ROAD, AUBURN HILLS, MI, 48326	US Mail (1st Class)
24846	AUTOMATION TECHNOLOGY SCHWOPE, 37679 SCHOOLCRAFT RD., LIVONIA, MI, 48150	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	AUTOMODULAR ASSEMBLIES INC., 200 MONTECORTE STREET, WHITBY, ON, L1N 9V8CANADA	US Mail (1st Class)
24846	AUTOMOTIVE ACCENTS, CAR CARE, INC., 17499 KINSMAN RD., MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	AUTOMOTIVE COMPONENT CARRIERS, 4405 CONTINENTAL DR., FLINT, MI, 48507	US Mail (1st Class)
24846	AUTOMOTIVE COMPONENTS INSPECTION, PO BOX 41, NORTH WEBSTER, IN, 46555	US Mail (1st Class)
24846	AUTOMOTIVE FORECASTERS, 23 MARION PLACE, HUNTINGTON STATION, NY, 11746	US Mail (1st Class)
24846	AUTOMOTIVE GAUGE & FIXTURE, 5270 BURKE STREET, WINDSOR, ON, N9A 6J3CANADA	US Mail (1st Class)
24846	AUTOMOTIVE IND ACTION GR, 26200 LAHSER RD STE 200, SOUTHFIELD, MI, 48033-7156	US Mail (1st Class)
24846	AUTOMOTIVE INDUSTRY, ACTION GROUP, DEPT.77839 P.O. BOX 77000, DETROIT, MI, 48277	US Mail (1st Class)
24846	AUTOMOTIVE INDUSTRY ACTION GRP, 26200 LAHSER ROAD, SUITE 200, SOUTHFIELD, MI, 48034	US Mail (1st Class)
24846	AUTOMOTIVE NEWS, SUBSCRIBER SERVICES, DEPARTMENT 77940, DETROIT, MI, 48277-0940	US Mail (1st Class)
24846	AUTOTRIM INC., 860 COLLISTON ROAD, ANN ARBOR, MI, 48105	US Mail (1st Class)
24846	AUTOWEB COMMUNICATIONS, INC., 1680 MOMENTUM PLACE, CHICAGO, IL, 60689-5316	US Mail (1st Class)
24846	AUTOWEB COMMUNICATIONS, INC., DIVISION OF ISD, PO BOX 8066, HOLLAND, MI, 49422-8066	US Mail (1st Class)
24846	AVA SERVICES, INC., 1711 W PALMETTO ST, FLORENCE, SC, 29501	US Mail (1st Class)
24846	AVALON VISION SOLUTIONS, 422 THORNTON ROAD, SUITE 104, LITHIA SPRINGS, GA, 30122	US Mail (1st Class)
24846	AVERY DENNISON IPD, 12303 COLLECTION CENTER DRIVE, CHICAGO, IL, 60693	US Mail (1st Class)
24846	AVI FOODSYSTEMS, INC., 2590 ELM ROAD, N.E., WARREN, OH, 44483	US Mail (1st Class)
24846	AVIS, 5500 AIRLINE DRIVE, BIRMINGHAM, AL, 35212	US Mail (1st Class)
24846	AVON AUTOMOTIVE, 805 W 13TH ST, CADILLAC, MI, 49601-9281	US Mail (1st Class)
24846	AWT METAL SPECIALTIES, 16590 13 MILE ROAD, ROSEVILLE, MI, 48066-0082	US Mail (1st Class)
24846	B & L PACKAGING, INC., 1752 STANLEY AVE, DAYTON, OH, 45404-1117	US Mail (1st Class)
24846	B & M AIR COMPRESSOR, 1310 E 49TH STREET, CLEVELAND, OH, 44114	US Mail (1st Class)
24846	B C WIRE ROPE & RIGGING, 4060 E MICHIGAN STREET, TUCSON, AZ, 85714	US Mail (1st Class)
24846	B D I, 11800 BELDEN COURT, LIVONA, MI, 48150	US Mail (1st Class)
24846	B O P P, C/O TERRI BERMAN, 11379 TAYLOR WELLS RD., CHARDON, OH, 44024	US Mail (1st Class)
24846	B W ROGERS CO., 7900 EMPIRE PKWY, MACEDONIA, OH, 44056	US Mail (1st Class)
24846	B.C. MACHINING, P.O. BOX 644, CONNEAUT, OH, 44030	US Mail (1st Class)
24846	B.C. MACHINING, INC., PO BOX 644, CONNEAUT, OH, 44030	US Mail (1st Class)
24846	B.F. JARVIS INC., 7327 SCENIC DR, ASHTABULA, OH, 44004-9371	US Mail (1st Class)
24846	BAILEY MANUFACTURING, BENNETT STATE ROAD, PO BOX 356, FORESTVILLE, NY, 14062	US Mail (1st Class)
24846	BAKER & HOSTETLER LLP, WIRE ONLY, HUNTINGTON NATIONAL BANK, CLEVELAND, OH, 44115	US Mail (1st Class)
24846	BAKER & MCKENZIE, P TRIUNFO DE LA REPUBLICA 3304, PISO 2, PARTIDO ESCOBEDO, JUAREZ, CHIHUAHUA, 32330MEXICO	US Mail (1st Class)
24846	BAKER VEHICLE SYSTEMS, 9035 FREEWAY DR, MACEDONIA, OH, 44056-1508	US Mail (1st Class)
24846	BALANCE TECHNOLOGY INC., 120 ENTERPRISE DRIVE, ANN ARBOR, MI, 48103	US Mail (1st Class)
24846	BALDWIN COOKE, PO BOX 312, GLOVERSVILLE, NY, 12078	US Mail (1st Class)
24846	BALTIMORE AIRCOIL COMPANY, 7600 DORSEY RUN RD, JESSOP, MD, 20794	US Mail (1st Class)
24846	BAMBERGER POLYMERS INC., 1334 SOLUTIONS CENTER, CHICAGO, IL, 60677-1003	US Mail (1st Class)
24846	BAMTECH, 1351 S CLEVELAND MASSILLON RD, COPLEY, OH, 44321-1662	US Mail (1st Class)
24846	BANK OF MOUNDVILLE, ATTN: TONY O. LEE, PO BOX 158, MOUNDVILLE, AL, 35474	US Mail (1st Class)
24846	BANK ONE, NA, 600 SUPERIOR AVENUE, PO BOX 91308, CLEVELAND, OH, 44101-9990	US Mail (1st Class)
24846	BANK ONE, NA, ATTN: AMY JAY, FIRST VICE PRESIDENT, 611 WOODWARD AVENUE, DETROIT, MI, 48226	US Mail (1st Class)
24846	BARBARA MORAN, 985 FALLS CREEK DR., VANDALIA, OH, 45377	US Mail (1st Class)
24846	BARBARA ROHARIK, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	BARBARA WHALEN, ADDRESS UNAVAILABLE, CHARDON, OH, 44024	US Mail (1st Class)
24846	BARBE AMERICA, WEST WHITE ROAD, PO BOX 69, FLOWERY BRANCH, GA, 30542	US Mail (1st Class)
24846	BARBER COLEMAN, INDUSTRIAL INSTRUMENT DIV., 1354 CLIFFORD AVE., LOVES PARK, IL, 61111	US Mail (1st Class)
24846	BARCO PRODUCTS CO., 11 NORTH BATAVIA, BATAVIA, IL, 60510	US Mail (1st Class)
24846	BARCODE SYSTEMS TECHNOLOGIES, 3769 COMMERCE CT., WAYNE, MI, 48184	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	BARKSDALE WARRIOR PAPER, 2401 CLEMENTS RD, COTTONDALE, AL, 35453	US Mail (1st Class)
24846	BARLOWORLD, 2749 SOUTHSIDE DR, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	BARNES VAN LINES, 3615 HWY 61, CARROLLTON, GA, 30116	US Mail (1st Class)
24846	BARNETS INC., 1619 BARNETS MILL ROAD, CAMDEN, OH, 45311	US Mail (1st Class)
24846	BART A. HEDRICK, 430 GEM AVE., MANSFIELD, OH, 44907	US Mail (1st Class)
24846	BART'S CONCRETE WORK, 523 S WILEY STREET, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	BARTA'S PRECISION SURFACE, PLATE CO., 3357 ORMOND ROAD, CLEVELAND, OH, 44118	US Mail (1st Class)
24846	BASELL POLYOLEFINS, 912 APPLETON RD, ELKTON, MD, 21921-3920	US Mail (1st Class)
24846	BASF/SUNOCO NORTH AMERICA, 100 CAMPUS DR, SLORHAM PARK, NJ, 07932	US Mail (1st Class)
24846	BASIC SERVICE CORPORATION, 2525 IMLAY CITY RD., LAPEER, MI, 48446-3215	US Mail (1st Class)
24846	BASS LAKE INN, 1180 BELL RD., CHAGRIN FALLS, OH, 44022	US Mail (1st Class)
24846	BAUDVILLE, 5380 52ND STREET S.E., GRAND RAPIDS, MI, 49512-9765	US Mail (1st Class)
24846	BAY STATE POLYMER DISTRIBUTION, INC, 37540 DETROIT RD, WESTLAKE, OH, 44145	US Mail (1st Class)
24846	BBK, LTD., 300 GALLERIA OFFICENTRE, SUITE 103, SOUTHFIELD, MI, 48034	US Mail (1st Class)
24846	BBU SERVICES, INC., 4318 LOGAN-LANCASTER, PO BOX 2541, LANCASTER, OH, 43130	US Mail (1st Class)
24846	BEARE'S RESTAURANT, 13 E DAYTON ST, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	BEARING DISTRIBUTORS INC., 8000 HUB PKWY, CLEVELAND, OH, 44125-5731	US Mail (1st Class)
24846	BEARING SERVICE COMPANY, RIDC PARK, 630 ALPHA DR, PITTSBURGH, PA, 15238-2802	US Mail (1st Class)
24846	BEATRICE M. MCFADDEN, 14819 ROSELAWN, DETROIT, MI, 48238	US Mail (1st Class)
24846	BEAVER VALLEY MFG. INC., 781 FACTORY RD., BEAVERCREEK, OH, 45434	US Mail (1st Class)
24846	BECKER & COMPANY, 27 W MAIN ST, CARNEGIE, PA, 15106-2404	US Mail (1st Class)
24846	BECKER ELECTRIC SUPPLY, 1341 E 4TH ST, DAYTON, OH, 45402-2235	US Mail (1st Class)
24846	BECKER'S ELECTRIC SUPPLY, 1341 E FOURTH ST, DAYTON, OH, 45402	US Mail (1st Class)
24846	BECKWITH MACHINERY CO., 4565 WILLIAM PENN HWY, MURRYSVILLE, PA, 15668	US Mail (1st Class)
24846	BECKY VALENTIN, 272 SARAH FERN CT., LAPEER, MI, 48446	US Mail (1st Class)
24846	BEEBE OIL CO., 4999 22 MILE ROAD, UTICA, MI, 48317	US Mail (1st Class)
24846	BEHCO, INC., 32613 FOLSOM ROAD, FARMINGTON HILLS, MI, 48336	US Mail (1st Class)
24846	BEHM INDUSTRIAL PRODUCTS, 20801 SALISBURY ROAD, BEDFORD, OH, 44146	US Mail (1st Class)
24846	BEHR CLIMATE SYSTEMS INC., 5020 AUGUSTA DRIVE, FORT WORTH, TX, 76106	US Mail (1st Class)
24846	BEHR GMBH & COMPANY, MAUSERSTRABE 3, STUTTGART 30 09 20, D-70469GERMANY	US Mail (1st Class)
24846	BEKAERT CORPORATION, 6000 N. BAILEY AVENUE, SUITE 9, AMHERST, NY, 14226	US Mail (1st Class)
24846	BEKUM, 1140 W GRAND RIVER, WILLIAMSTON, MI, 48895-0567	US Mail (1st Class)
24846	BELCAN SERVICES GROUP, LOCKBOX 771428, 1428 SOLUTIONS CENTER, CHICAGO, IL, 60677-1004	US Mail (1st Class)
24846	BELDEN VILLAGE MALL, 4230 BELDEN VILLAGE MALL, CANTON, OH, 44718	US Mail (1st Class)
24846	BELLEVILLE CITY TREASURER, 6 MAIN ST, BELLEVILLE, MI, 48111-2736	US Mail (1st Class)
24846	BELLEVILLE PRINT SHOP, 10665 BELLEVILLE ROAD, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	BELLSOUTH, ACCT: 205-333-9854; 205-333-0703;, AND 205-333-0001, 1155 PEACHTREE ST NE, ATLANTA, GA, 30309-7629	US Mail (1st Class)
24846	BELMONT EQUIPMENT CO., 32055 EDWARD AVE, MADISON HEIGHTS, MI, 48071-1419	US Mail (1st Class)
24846	BEN WILEY, 103 PLEASANT PARK, WARREN, OH, 44481	US Mail (1st Class)
24846	BENEFIT MANAGEMENT SERV., 9066 E. MARKET STREET, SUITE A, WARREN, OH, 44484	US Mail (1st Class)
24846	BENJAMIN STEEL CO., INC., 777 BENJAMIN DR, SPRINGFIELD, OH, 45502	US Mail (1st Class)
24846	BENNETT-URSEM COMPANY, 10314 BRECKSVILLE ROAD, BRECKSVILLE, OH, 44141-3338	US Mail (1st Class)
24846	BERG CHILLING SYSTEMS INC., 51 NANTUCKET BLVD., TORONTO, ON, M1P 2N5CANADA	US Mail (1st Class)
24846	BERGER TRANSFER & STORAGE, 2950 LONG LAKE RD, SAINT PAUL, MN, 55113-1050	US Mail (1st Class)
24846	BERKHEIMER ASSOCIATES, EARNED INCOME TAX ADMIN, PO BOX 900, BANGOR, PA, 18013-0900	US Mail (1st Class)
24846	BERNARD J. WIERCKZ, 184 CENTENNIAL, SPARTA, MI, 49345	US Mail (1st Class)
24846	BERNICE THOMAS BLAIR, 4203 20TH STREET NE, TUSCALOOSA, AL, 35404	US Mail (1st Class)
24846	BERNITA BAAS, 8707 SAMUEL BARTON DR., BELLEVILLE, MI, 48111	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	BEST FOAM FABRICATORS,INC., 9633 SOUTH COTTAGE GROVE RD, CHICAGO, IL, 60628	US Mail (1st Class)
24846	BEST MOLD & MFG INC., 1546 EAST TURKEYFOOT LAKE RD, AKRON, OH, 44312	US Mail (1st Class)
24846	BETA/LASERMIKE, INC., 21522 NETWORK PLACE, CHICAGO, IL, 60673-1215	US Mail (1st Class)
24846	BETHEL ENGINEERING & EQUIPMENT, 13830 MCBETH RD, NEW HAMPSHIRE, OH, 45870	US Mail (1st Class)
24846	BETTY JOYCE FANTIN, 567 STATE ROUTE 314 N, MANSFIELD, OH, 44903-9366	US Mail (1st Class)
24846	BEULAH INC., 375 N 475 W, VALPARAISO, IN, 46385	US Mail (1st Class)
24846	BEVERLY J. SCHULTZ, 834 CANTERBURY STREET SW, WYOMING, MI, 49509	US Mail (1st Class)
24846	BIBBY FINANCIAL SERVICES, ANCHOR RUBBER, DEPT CH 17360, PALATINE, IL, 60055-7360	US Mail (1st Class)
24846	BIG BROTHERS BIG SISTERS, OF NORTHEAST OHIO, INC., 2010 E PROSPECT STREET, ASHTABULA, OH, 44005-2683	US Mail (1st Class)
24846	BIJUR LUBRICATING, BOX 74727, CLEVELAND, OH, 44194-0810	US Mail (1st Class)
24846	BILL BOLLMAN, 2227 CHIPPENHAM CHASE, ROCHESTER HILLS, MI, 48306-3909	US Mail (1st Class)
24846	BILL CORNN, 119 EAST DAYTON STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	BILL CURRENCE, 8707 SAMUEL BARTON DRIVE, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	BILL METZCAR, 119 EAST DAYTON STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	BILL TESSO, 900 S. WILEY ST., PO BOX 320, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	BILL WARING, 2109 S CHESANING S.E., GRAND RAPIDS, MI, 49506	US Mail (1st Class)
24846	BILL'S CONCRETE WORK, 523 SOUTH WILEY STREET, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	BILLY CAMPBELL, PO BOX 1327, UNIONTOWN, AL, 36786	US Mail (1st Class)
24846	BKR INTERNATIONAL, S.C., AV LAZARO CARDENAS 2400, PTE EDIFICIO LOSOLES C-31-7, GARZA GARCIA, NL, 66296MEXICO	US Mail (1st Class)
24846	BLACK & WHITE TECH, PO BOX 489, 212 EAST WESTBROOK RD., BROOKVILLE, OH, 45309-0489	US Mail (1st Class)
24846	BLACK BOX CORPORATION, 1000 PARK DR, LAWRENCE, PA, 15055-1018	US Mail (1st Class)
24846	BLAKE'S PAINTING COMPANY, 1402 SPRINGWELLS, DETROIT, MI, 48209	US Mail (1st Class)
24846	BLAZING PROMOTIONS, 118 ALLENWOOD DR., HUNTSVILLE, AL, 35811	US Mail (1st Class)
24846	BLEVINS TECHNICAL, 1780 SOUTH METRO PARKWAY, DAYTON, OH, 45409	US Mail (1st Class)
24846	BLIND SITE CUSTOM WINDOW FASHION, 749 YORKSHIRE RD., TIPP CITY, OH, 45371	US Mail (1st Class)
24846	BLOW MOLDED SPECIALLTIES, INC., 222 BRONDER DR., PO BOX 310, FOLEY, MN, 56329-0310	US Mail (1st Class)
24846	BLUE CROSS BLUE SHIELD, OF ALABAMA, PO BOX 360037, BIRMINGHAM, AL, 35236	US Mail (1st Class)
24846	BLUE M ELECTRIC, 304 HART ST, WATERTOWN, WI, 53094-6616	US Mail (1st Class)
24846	BLUE TECH INC., 2251 SAN DIEGO AVE STE B155, SAN DIEGO, CA, 92110-2926	US Mail (1st Class)
24846	BLUE WATER PRINTING, 645 MCCORMICK DR., PO BOX 241, LAPEER, MI, 48446	US Mail (1st Class)
24846	BLUMERICH COMMUNICATIONS SERVICES, 6403 W PIERSON RD., FLUSHING, MI, 48433	US Mail (1st Class)
24846	BMC, C/O CITIBANK, ADDRESS UNAVAILABLE,	US Mail (1st Class)
24846	BMI GENERAL LICENSING, PO BOX 406741, ATLANTA, GA, 30384-6741	US Mail (1st Class)
24846	BMS BURNS/J.H. BURNS INC., 1061 INDUSTRIAL PKWY, MEDINA, OH, 44256	US Mail (1st Class)
24846	BOHL EQUIPMENT CO., 534 W. LASKEY CO, TOLEDO, OH, 43612	US Mail (1st Class)
24846	BOHLER UDDEHOLM CORP., 548 CLAYTON COURT, WOOD DALE, IL, 60191	US Mail (1st Class)
24846	BOISE CASCADE OFFICE PROD, 13301 STEPHENS RD, WARREN, MI, 48089	US Mail (1st Class)
24846	BOLLHOFF RIV-NUT, 2705 MARION DR, KENDALLVILLE, IN, 46755	US Mail (1st Class)
24846	BOLLHOFF RIV-NUT, 2705 MARION DR, KENDALLVILLE, IN, 46755	US Mail (1st Class)
24846	BOND FLUIDAIRE INC., 5506 36TH STREET SE, GRAND RAPIDS, MI, 49512	US Mail (1st Class)
24846	BONITA ARNOLD, 38 BALDWIN AVE #3, MANSFIELD, OH, 44906	US Mail (1st Class)
24846	BONNIE QUAY, 8844 MIDDLE RD, LAKE CITY, PA, 16423-2130	US Mail (1st Class)
24846	BONNY KIPP, 900 W. WILEY ST., PO BOX 320, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	BORBOLLA METROLOGY, 417 EAST COMA ST, HIDALGO, TX, 78557	US Mail (1st Class)
24846	BORG WARNER AIR/FLUID SYS, PO BOX 727, HIGHWAY 32 EAST, WATER VALLEY, MS, 38965	US Mail (1st Class)
24846	BOSANOVA, INC., 2021 W. ROSE GARDEN LANE, PHOENIX, AZ, 85027	US Mail (1st Class)
24846	BOSTON MATTHEWS, INC., 57 OAK STREET, NORWOOD, NJ, 07648	US Mail (1st Class)
24846	BOSTWICK-BRAUN CO., 5000 GROVE AVE, LORAIN, OH, 44055	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	BPR/RICO EQUIPMENT, INC., 691 WEST LIBERTY STREET, MEDINA, OH, 44256	US Mail (1st Class)
24846	BRAD STANLEY, 13357 NEFF RD, CLIO, MI, 48420	US Mail (1st Class)
24846	BRAD SZABO, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	BRADLEY DESIGN SYSTEM, 197 FRONTENAC, HOWELL, MI, 48843	US Mail (1st Class)
24846	BRADY INDUSTRIAL SUPPLY, INC., 41100 BRIDGE STREET, NOVI, MI, 48375	US Mail (1st Class)
24846	BRAKE PRODUCTS INC., 16751 HILLTOP PARK PL, CHAGRIN FALLS, OH, 44023-4500	US Mail (1st Class)
24846	BRANCH SYSTEMS & EQUIPMENT, 545 N BROAD, SUITE 3, CANFIELD, OH, 44406	US Mail (1st Class)
24846	BRANDENBURG LAWN CARE, 5847 ST RTE 127, CAMDEN, OH, 45311	US Mail (1st Class)
24846	BRANDY VADNEY, 10047 KEYSTONE DRIVE, LAKE CITY, PA, 16423	US Mail (1st Class)
24846	BRECHBUHLER SCALES, INC., 1424 SCALE ST SW, CANTON, OH, 44706-3096	US Mail (1st Class)
24846	BRENDA G. NOLDEN, 8016 BAPTIST, CAMPGROUND ROAD, NORTHPORT, AL, 35473	US Mail (1st Class)
24846	BRENDA WHITE, PREBLE COUNTY TREASURER, PO BOX 341, EATON, OH, 45320	US Mail (1st Class)
24846	BRENNTAG NORTHEAST, 1085 ALLEGHENY AVE, OAKMONT, PA, 15139	US Mail (1st Class)
24846	BRENT BROWN, 290 MCCORMICK STREET, LAPEER, MI, 48446	US Mail (1st Class)
24846	BRETT E. DUBY, 104 AVERLON AVE APT 9, ERIE, PA, 16509-1470	US Mail (1st Class)
24846	BRIAN CUNNINGHAM, 1701 INDUSTRIAL PARK DRIVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	BRIAN E. KERNS, 3023 30TH STREET NE, CANTON, OH, 44705	US Mail (1st Class)
24846	BRIAN GERMAINE, 9344 BRISTOL, DAVISON, MI, 48423	US Mail (1st Class)
24846	BRIAN SCHLECTY, PO BOX 36, 109 WEST CROSS ST., PALESTINE, OH, 45352	US Mail (1st Class)
24846	BRIAN THORNBURG, 7894 NICODEMUS RD, RAVENNA, OH, 44266-8320	US Mail (1st Class)
24846	BRIAN WINDBIGLER PHOTOGRAPHY &, DESIGN, 1183 SOUTH BELVOIR BLVD., SOUTH EUCLID, OH, 44121	US Mail (1st Class)
24846	BRIGHTON ANALYTICAL, 2105 PLESS DRIVE, BRIGHTON, MI, 48114	US Mail (1st Class)
24846	BRJ CHEMICALS, PO BOX 944, REYNOLDSBURG, OH, 43068	US Mail (1st Class)
24846	BROADCAST MUSIC, INC., 10 MUSIC SQUARE EAST, NASHVILLE, TN, 37203-4399	US Mail (1st Class)
24846	BROADWAY SAND & GRAVEL, 2000 SANDRIDGE DR, DAYTON, OH, 45439	US Mail (1st Class)
24846	BROCKMAN, COATS, GEDALIAN, 1735 MERRIMAN ROAD, AKRON, OH, 44313-9007	US Mail (1st Class)
24846	BROCKMAN, COATS, GEDELIAN & CO, 1735 MERRIMAN ROAD, AKRON, OH, 44313-9007	US Mail (1st Class)
24846	BRONER GLOVE AND SUPPLY, 1750 HARMON RD, AUBURN HILLS, MI, 48326	US Mail (1st Class)
24846	BRONIEC ASSOCIATES,INC., 4855 PEACHTREE INDUSTRIAL BLVD STE 215, NORCROSS, GA, 30092-3014	US Mail (1st Class)
24846	BRONZE HEADQUARTERS INC., SUB. OF HEADCO INDUSTRY, 10220-30 BEREA ROAD, CLEVELAND, OH, 44102	US Mail (1st Class)
24846	BROOKHOLLOW EVERYDAY CARD, 1 STATIONERY PLACE, REXBURG, ID, 83441	US Mail (1st Class)
24846	BROOKVILLE GLOVE MFG. CO. INC., C/O BSI DIVERSIFIED LLC, PO BOX 280, REYNOLDSVILLE, PA, 15851	US Mail (1st Class)
24846	BROOKVILLE STAR, 14 MULBERRY STREET, PO BOX 100, BROOKVILLE, OH, 45309-0100	US Mail (1st Class)
24846	BROTHER INTERNATIONAL CORP, 100 SOMERSET CORPORATE BLVD, BRIDGEWATER, NJ, 08807	US Mail (1st Class)
24846	BROUGH CARPETS, 1938 S DORT HWY, FLINT, MI, 48503	US Mail (1st Class)
24846	BROUSE MCDOWELL, P.O. BOX 75579, CLEVELAND, OH, 44101-4755	US Mail (1st Class)
24846	BROWN & SHARPE, INC. MSD, 1742 SOLUTIONS CENTER, CHICAGO, IL, 60677-1007	US Mail (1st Class)
24846	BROWN DERBY 'ROAD HOUSE', 3985 PARK AVENUE WEST, PO BOX 223, ONTARIO, OH, 44862	US Mail (1st Class)
24846	BROWN PUBLISHING, 10222 ALLIANCE RD, CINCINNATI, OH, 45242-4710	US Mail (1st Class)
24846	BROWN PUBLISHING CO., 10222 ALLIANCE RD, CINCINNATI, OH, 45242-4710	US Mail (1st Class)
24846	BROWN SPRINKLER SERVICE, INC., 2458 STATE ROUTE 193, DORSET, OH, 44032	US Mail (1st Class)
24846	BROWNLEE SERVICES INC., 123 W ASH ST, PIQUA, OH, 45356	US Mail (1st Class)
24846	BRUCE FASSETT, 31572 INDUSTRIAL RD., SUITE #300, LIVONIA, MI, 48150	US Mail (1st Class)
24846	BRUCE TAYLOR, 290 MCCORMICK, LAPEER, MI, 48446	US Mail (1st Class)
24846	BRUCE WANDYEZ, 983 RIVEROAK AVE., BIRMINGHAM, MI, 48009	US Mail (1st Class)
24846	BRUMBAUGH CONSTRUCTION, 3520 STATE ROUTE 49, PO BOX 309, ARCANUM, OH, 45304	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	BRUSKE PRODUCTS, PO BOX 669, 7447 DUVAN DRIVE, TINLEY PARK, IL, 60477-0669	US Mail (1st Class)
24846	BRY AIR INC., BOX 2731, PO BOX 8500, PHILADELPHIA, PA, 19178-2731	US Mail (1st Class)
24846	BSF INC, 8895 NORTH DIXIE DR, DAYTON, OH, 45414	US Mail (1st Class)
24846	BUCCANEER CAMPSITES, PO BOX 352, 1408 ST. RT. 307W, JEFFERSON, OH, 44047	US Mail (1st Class)
24846	BUCKEYE AIR COMPRESSOR CO., 151 HEID AVE., DAYTON, OH, 45404	US Mail (1st Class)
24846	BUCKEYE BUSINESS PRODUCTS, INC, BOX 92340, CLEVELAND, OH, 44193	US Mail (1st Class)
24846	BUCKEYE DUPLICATING, 286 BRUBAKER DR., PO BOX 131, NEW CARLISLE, OH, 45344	US Mail (1st Class)
24846	BUCKHORN INC, 55 W TECHNE CENTER DR, MILFORD, OH, 45150-8901	US Mail (1st Class)
24846	BUCKHORN MHG. INC., 1397 PIEDMONT, SUITE 500, TROY, MI, 48083	US Mail (1st Class)
24846	BUD HIPSHER, 108 MAYNARD AVE, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	BUDGET MOLDERS SUPPLY INC, 8303 CORPORATE PARK DR, MACEDONIA, OH, 44056	US Mail (1st Class)
24846	BUDZAR INDUSTRIES, 38241 WILLOUGHBY PKWY, WILLOUGHBY, OH, 44094-7582	US Mail (1st Class)
24846	BUILD A MOLD LTD, 34845 MOUND RD, STERLING HEIGHTS, MI, 48310	US Mail (1st Class)
24846	BULKMATIC TRANSPORT CO., 5275 PAYSPHERE CIRCLE, CHICAGO, IL, 60674	US Mail (1st Class)
24846	BULLDOG FABRICATING CORP., 50 ENTERPRISE DR, ANN ARBOR, MI, 48103	US Mail (1st Class)
24846	BULLSEYE ASSEMBLY SOLUTIONS, 140 ENGELWOOD DR., SUITE G, ORION, MI, 48359	US Mail (1st Class)
24846	BULLSEYE TELECOM, 25900 GREENFIELD ROAD, SUITE 330, OAK PARK, MI, 48237	US Mail (1st Class)
24846	BUNTING MAGNETICS CANADA, 2424 FINCH AVENUE WEST, UNIT 4B, TORONTO, ON, M3M 2E3CANADA	US Mail (1st Class)
24846	BUNTING MAGNETICS CO., 500 S SPENCER RD, NEWTON, KS, 67114-4109	US Mail (1st Class)
24846	BURCH CORPORATION, 216 OPORTO MADRID BLVD S, BIRMINGHAM, AL, 35206-4850	US Mail (1st Class)
24846	BUREAU OF BUSINESS, PRACTICE, CHICAGO, IL, 60673-0845	US Mail (1st Class)
24846	BUREAU OF MOTOR VEHICLES, 1101 S FRONT ST, HARRISBURG, PA, 17104	US Mail (1st Class)
24846	BUREAU OF MOTOR VEHICLES, NON EPA REGISTRATION RENEWAL, PO BOX 183089, COLUMBUS, OH, 43218-3089	US Mail (1st Class)
24846	BUREAU OF MOTOR VEHICLES, NONEPA REG RENEWAL, PO BOX 182228, COLUMBUS, OH, 43218-2228	US Mail (1st Class)
24846	BUREAU OF MOTOR VEHICLES, NONEPA LEASE RENEWAL, PO BOX 183087, COLUMBUS, OH, 43218-3087	US Mail (1st Class)
24846	BUREAU OF WORKER'S COMPENSATION, PO BOX 710977, COLUMBUS, OH, 43271-0977	US Mail (1st Class)
24846	BURKES MECHANICAL, INC., # 2 INDUSTRIAL RD., BRENT, AL, 35034	US Mail (1st Class)
24846	BURKHARDT GMBH, ROBERT-KOCK - STR, 15, 71665 VAIHINGEN, ENSINGEN, GERMANY	US Mail (1st Class)
24846	BURNS MOLD SUPPLY, 1061 INDUSTRIAL PKWY, MEDINA, OH, 44256-2448	US Mail (1st Class)
24846	BURR & FORMAN LLP, LAURA ROBERTS, 420 NORTH TWENTIETH STREET, SUITE 3100, BIRMINGHAM, AL, 35203	US Mail (1st Class)
24846	BURTON BLUEPRINTING, INC., 5252 EXCHANGE DR, FLINT, MI, 48507-2934	US Mail (1st Class)
24846	BURTON PRESS COMPANY, 2156 AVON INDUSTRIAL DRIVE, ROCHESTER HILLS, MI, 48309	US Mail (1st Class)
24846	BURTON SHEET METAL, 17741 CLARIDON TROY RD., BURTON, OH, 44021	US Mail (1st Class)
24846	BUSCH INC, 516 VIKING DR, VIRGINIA BEACH, VA, 23452	US Mail (1st Class)
24846	BUSH INDUSTRIAL TIRE, 1515 J.P. HENNESSY DRIVE, LAVERGNE, TN, 37086	US Mail (1st Class)
24846	BUSHMAN EQUIPMENT,INC., WEST 133 N 4960 CAMPBELL DR, MENONOMEE FALLS, WI, 53051	US Mail (1st Class)
24846	BUSHONG RESTAURANT SUPPLY, 422 EAST BACON STREET, DAYTON, OH, 45402	US Mail (1st Class)
24846	BUSINESS & LEGAL REPORTS, 141 MILL ROCK ROAD EAST, OLD SAYBROOK, CT, 06475	US Mail (1st Class)
24846	BUSINESS 21 PUBLISHING, 477 BALTIMORE PIKE, SPRINGFIELD, PA, 19064	US Mail (1st Class)
24846	BUSINESS COMMUNICATIONS, INC., THOMAS REGISTER, G.P.O. 9352, NEW YORK, NY, 10087-9352	US Mail (1st Class)
24846	BUSINESS COUNCIL OF ALABAMA, 2 NORTH JACKSON ST, MONTGOMERY, AL, 36104	US Mail (1st Class)
24846	BUSINESS HEALTH SERVICES, ATTN: JACKIE SCHANK, 36616 PLYMOUTH ROAD, LIVONIA, MI, 48150	US Mail (1st Class)
24846	BUSINESS WEEK, 1221 AVENUE OF THE AMERICAS, NEW YORK, NY, 10020-1001	US Mail (1st Class)
24846	BUTLER COUNTY COMMON PLEAS COURT, 315 HIGH STREET, HAMILTON, OH, 45011	US Mail (1st Class)
24846	BWC STATE INSURANCE FUND, CORPORATE PROCESSING DEPT., PO BOX 710977, COLUMBUS, OH, 43271-0977	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	BYK GARDNER USA, 9104 GUILFORD, COLUMBIA, MD, 21046	US Mail (1st Class)
24846	BYOKANE USA CORPORATION, 3400 KASHIWA STREET, TORRANCE, CA, 90505	US Mail (1st Class)
24846	BYRNES TOOL COMAPNY, 870 WHITNEY DRIVE, LAPEER, MI, 48446	US Mail (1st Class)
24846	BYTEWISE MEASUREMENT SYS., 1150 BROOKSTONE CTR PKWY, COLUMBUS, GA, 31904	US Mail (1st Class)
24846	C & A TOOL, 111 S. MAIN ST., CHURUBUSCO, IN, 46723	US Mail (1st Class)
24846	C & F TRUCKING & TIRE SERV, 111 SMITH AVENUE, MANSFIELD, OH, 44905	US Mail (1st Class)
24846	C & H DISTRIBUTORS, 770 S 70TH ST, MILWAUKEE, WI, 53214-3109	US Mail (1st Class)
24846	C & H DISTRIBUTORS LLC, 22133 NETWORK PLACE, CHICAGO, IL, 60673-1133	US Mail (1st Class)
24846	C & J BRAFASCO, 13400 STARK ROAD, LIVONIA, MI, 48150	US Mail (1st Class)
24846	C & S INDUSTRIAL SUPPLIES, 6342 E STATE HIGHWAY 402, LOVELAND, CO, 80537-8815	US Mail (1st Class)
24846	C D & R SORTING, 3247 WEST STATE RD 229, BATESVILLE, IN, 47006	US Mail (1st Class)
24846	C H ROBINSON INC., 8100 MITCHELL RD, EDEN PRAIRIE, MN, 55344-2178	US Mail (1st Class)
24846	C P HALL COMPANY, 120 S RIVERSIDE PLZ RM 1620, CHICAGO, IL, 60606-3911	US Mail (1st Class)
24846	C P HALL COMPANY, POBOX 99404, CHICAGO, IL, 60693	US Mail (1st Class)
24846	C T GAUL BUILDER, 3793 MAXWELL ROAD RTE.14, LEXINGTON, OH, 44904	US Mail (1st Class)
24846	C V A INCORPORATED, 4513 SOUTH BEECH DALY, DEARBORN HGHTS, MI, 48125	US Mail (1st Class)
24846	C&E SALES, INC., 677 CONGRESS PARK DR, DAYTON, OH, 45459	US Mail (1st Class)
24846	C&J MACHINE CO., 403 STATE RT. 44, HARTVILLE, OH, 44632	US Mail (1st Class)
24846	C.E.P. CARLISEL (BUNDY), 1901 WAGER ROAD, ERIE, PA, 16514	US Mail (1st Class)
24846	C.H. RACHES, INC., 1100 N. OPDYKE RD., SUITE #200, AUBURN HILLS, MI, 48326	US Mail (1st Class)
24846	C.P. SERVICES, 5025 S ASH ST. STE B17, TEMPE, AZ, 85282	US Mail (1st Class)
24846	C.S.D.D., 8609 W GRAND RIVER AVE., SUITE 103, BRIGHTON, MI, 48116	US Mail (1st Class)
24846	C.W. WARD, 2323 IMLAY CITY RD, LAPEER, MI, 48446	US Mail (1st Class)
24846	C/L MACHINE INC., 826 CLEVELAND RD, RAVENNA, OH, 44266	US Mail (1st Class)
24846	CADCAM-E.COM, 6 ARROW ROAD, SUITE 20B, RAMSEY, NJ, 07746	US Mail (1st Class)
24846	CADILLAC ELECTRIC &AUTOMATION, C/O FRANKLIN BANK, PO BOX 44047, DETROIT, MI, 48244-0047	US Mail (1st Class)
24846	CADY REPORTING SERVICES, 55 PUBLIC SQUARE, SUITE 1225, CLEVELAND, OH, 44113	US Mail (1st Class)
24846	CAE SERVICES CORPORATION, 280 BELLEVIEW LANE, BATAVIA, IL, 60510	US Mail (1st Class)
24846	CAIN STEEL & SUPPLY, INC., 2650 20TH ST, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	CALDWELL & BLOOR CO., 1283 S TRIMBLE RD, MANSFIELD, OH, 44907	US Mail (1st Class)
24846	CALGRAPH TECHNOLOGY SVS. INC., 577 BURNING TREE RD., FULLERTON, CA, 92833	US Mail (1st Class)
24846	CALLAHAN DOOR SALES, INC., 7855 FREEDOM AVENUE N.W., NORTH CANTON, OH, 44720	US Mail (1st Class)
24846	CALSAC CORPORATION, PO BOX 514698, LOS ANGELES, CA, 90051-4698	US Mail (1st Class)
24846	CALVIN SPIESS, TREASURER, P.O. BOX 546, NAPOLEAN, OH, 43545-0546	US Mail (1st Class)
24846	CAM INC., 1525 CORPORATE WOODS PKWY, UNIONTOWN, OH, 44685	US Mail (1st Class)
24846	CAM TECHNOLOGIES INC., 3933 KAULOOSA AVE, TUSCALOOSA, AL, 35401-7031	US Mail (1st Class)
24846	CAMBRIDGE TOOL & DIE, BOX 901285, CLEVELAND, OH, 44190-1285	US Mail (1st Class)
24846	CAMCAR DIV. OF TEXTRON, TEXTRON FASTENING SYSTEMS, 24201 NETWORK PLACE, CHICAGO, IL, 60673-1242	US Mail (1st Class)
24846	CAMCAR PLASTICS INC., 1204 W WESTERN, MUSKEGON, MI, 49441	US Mail (1st Class)
24846	CAMPBELL OIL COMPANY, 611 ERIE ST S, MASSILLON, OH, 44646-6711	US Mail (1st Class)
24846	CAMPFIRE INTERACTIVE, INC., 320 MILLER RD., SUITE A1, ANN ARBOR, MI, 48104	US Mail (1st Class)
24846	CAN.IMPERIAL BANK OF COMM, DRAWER 33018, TRIM TRENDS/HARVARD IND., DETROIT, MI, 48232	US Mail (1st Class)
24846	CANADIAN FEED SCREWS MFG. INC., 80 VENTURE DRIVE, SCARBOROUGH, ON, M1B 3L6CANADA	US Mail (1st Class)
24846	CANFAB INC., 5621 12TH AVE. E, TUSCALOOSA, AL, 35405	US Mail (1st Class)
24846	CANNONBURGH MOVING COMPANY, PO BOX 1151, MURFREESBORO, TN, 37133-1151	US Mail (1st Class)
24846	CANON COMMUNICATIONS LLC, 11444 W OLYMPIC BLVD, SUITE 900, LOS ANGELES, CA, 90064-1549	US Mail (1st Class)
24846	CANTEEN SERVICE COMPANY, 5202 COMMERCE ROAD, FLINT, MI, 48507	US Mail (1st Class)
24846	CANTON CITY AUDITOR, 218 CLEVELAND AVE SW, CANTON, OH, 44702-1906	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	CANTON CTY TREASURER INCOME TX, 424 MARKET AVE N, CANTON, OH, 44702-1544	US Mail (1st Class)
24846	CANTON ERECTORS, INC., 2009 QUIMBY AVE. S.W., CANTON, OH, 44706	US Mail (1st Class)
24846	CANTON STERILIZED WIPING, CLOTH, PO BOX 7227, CANTON, OH, 44705	US Mail (1st Class)
24846	CAP COMPANY, 10355 AUBURN RD, CHARDON, OH, 44024-8620	US Mail (1st Class)
24846	CAPITAL TELEPHONE, LLC, 7182 HWY 14, SUITE 918, MIDDLETON, WI, 53562	US Mail (1st Class)
24846	CAPITOL REPRODUCTIONS INC, 215 E TWELVE MILE ROAD, PO BOX 71484, MADISON HEIGHTS, MI, 48071	US Mail (1st Class)
24846	CAPRO INC., LOCKBOX 3355, PO BOX 8500, PHILADELPHIA, PA, 19178-3355	US Mail (1st Class)
24846	CARBIDE SPECIALTIES, INC., 3941 EASTERN SOUTHEAST, GRAND RAPIDS, MI, 49508	US Mail (1st Class)
24846	CARBIDE TOOLS & ABRASIVES INC., 2206 13TH ST SE, CANTON, OH, 44705	US Mail (1st Class)
24846	CARCO INC., PO BOX 13859, 10333 SHOEMAKER, DETROIT, MI, 48213	US Mail (1st Class)
24846	CARDINAL HIGH SCHOOL, 14785 THOMPSON AVE, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	CAREER TRACK INC., 9757 METCALF AVE, OVERLAND PARK, KS, 66212-2219	US Mail (1st Class)
24846	CAREERTRACK, 9757 METCALF AVE, OVERLAND PARK, KS, 66212-2219	US Mail (1st Class)
24846	CARGO EXPRESS, 8401 GATEWAY WEST E-1, EL PASO, TX, 79925	US Mail (1st Class)
24846	CARI EARLY, 900 S WILEY STREET, PO BOX 320, CRESTLINE, OH, 44827-0320	US Mail (1st Class)
24846	CARLENE PARKER, 8707 SAMUEL BARTON DRIVE, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	CARLISLE ENG.PRD.LAKE CIT, 10047 KEYSTONE DRIVE, LAKE CITY, PA, 16423	US Mail (1st Class)
24846	CARLISLE INSURANCE COMPANY, ATTN: MELISSA HANCOCK, 100 BANK STREET, SUITE 610, BURLINGTON, VT, 05401	US Mail (1st Class)
24846	CARLISLE SYNTEC, 22721 NETWORK PLACE, CHICAGO, IL, 60673-1227	US Mail (1st Class)
24846	CARLISLE TIRE & WHEEL, 601 EAST WISE STREET, TRENTON, SC, 29847	US Mail (1st Class)
24846	CARLISLE TIRE & WHEEL, 601 E WISE ST, TRENTON, SC, 29847	US Mail (1st Class)
24846	CARLSON CATERING COMPANY, 44850 N. I-94 SERVICE DRIVE, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	CARMELITA CALUAG, 1401 INDUSTRIAL PARK DRIVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	CARNEGIE TEXTILE, 1734 IVANHOE RD, CLEVELAND, OH, 44112	US Mail (1st Class)
24846	CARO PUBLISHING, 344 N STATE ST, CARO, MI, 48723-1538	US Mail (1st Class)
24846	CAROL STARRE-KMIECIK, 1445 WYANDOTTE AVENUE, LAKEWOOD, OH, 44107	US Mail (1st Class)
24846	CAROLE K. FALLOT, 3131 COLUMBUS ROAD NE, CANTON, OH, 44705	US Mail (1st Class)
24846	CAROLINA COLOR CORP., 100 EAST 17TH STREET, SALISBURY, NC, 28145-0486	US Mail (1st Class)
24846	CAROLINA COLOR CORP. OF OHIO, 100 COLOMET DR, DELAWARE, OH, 43015-3846	US Mail (1st Class)
	CAROLINA INDUSTRIAL EQUIPMENT, 6210 DIXIE RD, CHARLOTTE, NC, 28278-9516	US Mail (1st Class)
24846	CAROTHERS EXTERMINATING, PO BOX 1048, GALION, OH, 44833	US Mail (1st Class)
24846	CARRIER TRANSICOLD DIV., REC'G. DEPT. BLDG. TR 20, THOMPSON ROAD, SYRACUSE, NY, 13221	US Mail (1st Class)
24846	CARRIG & ASSOC., 11565 PEARL RD STE 101, STRONGSVILLE, OH, 44136-3356	US Mail (1st Class)
24846	CARTER LUMBER, 14601 KINSMAN RD., MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	CARTER LUMBER, 3835 MANSFIELD RD., SHELBY, OH, 44875-9474	US Mail (1st Class)
24846	CASCAN TOOL ENG. & MFG., R.R. NO. 1, RUSCOM, ON, NOR 1ROCANADA	US Mail (1st Class)
24846	CASTER CONNECTION, 745 SOUTH ST, CHARDON, OH, 44024	US Mail (1st Class)
24846	CATERPILLAR FINANCIAL SERVICES, 2120 W END AVE, NASHVILLE, TN, 37203-5258	US Mail (1st Class)
24846	CATHERINE BORTON, 985 FALLS CREEK DR, VANDALIA, OH, 45377	US Mail (1st Class)
24846	CATHY COUSINS, 11526 TAMARACK RD, WATERFORD, PA, 16441	US Mail (1st Class)
24846	CAVERT WIRE CO., 620 FORUM PKWY, RURAL HALL, NC, 27045-8934	US Mail (1st Class)
24846	CAVERT WIRE CO., INC., 620 FORUM PKWY, RURAL HALL, NC, 27045-8934	US Mail (1st Class)
24846	CBS PERSONNEL SERVICES, LOCATION 00464, CINCINNATI, OH, 45264-0464	US Mail (1st Class)
24846	CCA-DIVISION OF TAXATION, 1701 LAKESIDE AVE., CLEVELAND, OH, 44114	US Mail (1st Class)
24846	CCC ENDOWMENT FUND, ATTN: GOLF, PO BOX 354, CONNERSVILLE, IN, 47331	US Mail (1st Class)
24846	CCI THERMAL TECHNOLOGIES, INC., 1420 W MAIN ST, GREENSBURG, IN, 47240-9730	US Mail (1st Class)
24846	CCM RENTAL CENTERS, 15613 W. HIGH ST., MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	CD&R COMPONENTS INC, DBA CD&R SORTING, 3247 WEST STATE RD 229, BATESVILLE, IN, 47006	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	CD&R COMPONENTS INC., 3247 W STATE RD 229, BATESVILLE, IN, 47006	US Mail (1st Class)
24846	CDI TORQUE PRODUCTS, 19220 SAN JOSE AVE, CITY OF INDUSTRY, CA, 91748-1497	US Mail (1st Class)
24846	CDW COMPUTER CENTERS, INC., 200 N MILWAUKEE AVE, VERNON HILLS, IL, 60061-1577	US Mail (1st Class)
24846	CDW DIRECT LLC, 200 N MILWAUKEE AVE, VERNON HILLS, IL, 60061	US Mail (1st Class)
24846	CECILIA S. HYNES, 1038 BURNS ST, MT MORRIS, MI, 48458	US Mail (1st Class)
24846	CEDAR POINT, ACCOUNTING DEPT., ONE CEDAR POINT DRIVE, SANDUSKY, OH, 44870-5559	US Mail (1st Class)
24846	CELLULAR CONNECTION, 50 W BROAD ST, COLUMBUS, OH, 43215	US Mail (1st Class)
24846	CELLULAR SYSTEMS, 1241 MAIN STREET, CUYAHOGA FALLS, OH, 44221	US Mail (1st Class)
24846	CEM CORPORATION, 3100 SMITH FARM RD., MATTHEWS, NC, 28105	US Mail (1st Class)
24846	CEMENT PRODUCTS, INC., 389 PARK AVENUE EAST, MANSFIELD, OH, 44905-2842	US Mail (1st Class)
24846	CENTENNIAL MOLD INC., 44038 PHOENIX, STERLING HEIGHTS, MI, 48314	US Mail (1st Class)
24846	CENTER FOR CORPORATE HEALTH, 2420 LAKE AVE., ASHTABULA, OH, 44004	US Mail (1st Class)
24846	CENTERPOINT INDUSTRIES, 1449 BOWERS RD., LAPEER, MI, 48446	US Mail (1st Class)
24846	CENTRAL COMPUTER SUPPLIES, 139 HAMPSHIRE DR, TROY, MI, 48085-3226	US Mail (1st Class)
24846	CENTRAL MICHIGAN LUMBER, 1245 IMLAY CITY RD., LAPEER, MI, 48446	US Mail (1st Class)
24846	CENTRAL OHIO POWER TOOLS, 1042 DUBLIN ROAD, COLUMBUS, OH, 43217	US Mail (1st Class)
24846	CENTRAL SUPPLY, PO DRAWER 2209, TUSCALOOSA, AL, 35403	US Mail (1st Class)
24846	CENTRAL TAX BUREAU, INC., 141 RANDALL AVE., GIRARD, PA, 16417	US Mail (1st Class)
24846	CENTRAL TRANSPORT, 6599 HARPER AVE, DETROIT, MI, 48211-2404	US Mail (1st Class)
24846	CENTRAL-ALLIED ENTERPRISES INC, 2609 COLUMBUS RD NE, CANTON, OH, 44705-3764	US Mail (1st Class)
24846	CENTREX CORPORATION, 308 COLUMBIA RD, NEW KENSINGTON, PA, 15068	US Mail (1st Class)
24846	CENTURY CARRIERS, INC., P.O BOX 159, COTTONDALE, AL, 35453	US Mail (1st Class)
24846	CEP HOLDINGS, LLC, 3560 W. MARKET ST., SUITE 340, AKRON, OH, 44333	US Mail (1st Class)
24846	CEP PRODUCTS, LLC, 3560 W. MARKET STREET, AKRON, OH, 44333	US Mail (1st Class)
24846	CEP PRODUCTS/BELLEVILLE, 8707 SAMUEL BARTON DRIVE, BELLEVILLE, MI, 48112	US Mail (1st Class)
24846	CEP PRODUCTS/CANTON, 3131 COLUMBUS RD NE, CANTON, OH, 44705	US Mail (1st Class)
24846	CEP PRODUCTS/CHARDON, 100 SEVENTH AVE, SUITE 100, CHARDON, OH, 44024	US Mail (1st Class)
24846	CEP PRODUCTS/W. ALEX, 119 EAST DAYTON STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	CEP PRODUCTS-HERM/MEX, BLVD. FUTURA # 120, DYNATECH SOUTH INDUSTRIAL PARK, HERMOSILLO SONORA, 83296MEXICO	US Mail (1st Class)
24846	CEP-CHI/MEX S. A. DE C.V., AV. VICTOR HUGO #330, COMPLEJO INDUSTRIAL CHIHUAHUA, CHIHUAHUA CHIH, CP, 31109MEXICO	US Mail (1st Class)
24846	CERTIFIED COMPARATOR PRODUCTS, 810 N. ORCHARD LANE, SUITE 102, BEAVERCREEK, OH, 45434	US Mail (1st Class)
24846	CHAMBERS INDUSTRIAL, PRODUCTS COMPANY, 8040 AQUADALE DRIVE, YOUNGSTOWN, OH, 44512	US Mail (1st Class)
24846	CHANNEL POLYMERS, 11956 JESSE DR, EDINBORO, PA, 16412-4014	US Mail (1st Class)
24846	CHAPMAN GASES & WELDING, 1313 CHESTNUT ST., ERIE, PA, 16501	US Mail (1st Class)
24846	CHARDON FINANCE DEPT, 111 WATER ST, CHARDON, OH, 44024-1201	US Mail (1st Class)
24846	CHARDON LABORATORIES, ACCOUNTS RECEIVABLE, DEPT1A, PO BOX 1004, COLUMBUS, OH, 43216-1004	US Mail (1st Class)
24846	CHARDON LABORATORIES,INC., 7300 TUSSING RD, REYNOLDSBURG, OH, 43068	US Mail (1st Class)
24846	CHARDON PLASTICS MACHINE, 11680 BUTTERNUT RD., CHARDON, OH, 44024	US Mail (1st Class)
24846	CHARLES A. RODRIGUEZ, 3131 COLUMBUS RD., CANTON, OH, 44705	US Mail (1st Class)
24846	CHARLES F. BRACKNEY, 8707 SAMUEL BARTON DRIVE, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	CHARLES FOSTER, 2924 EUTAW STREET, AIKEN, SC, 29801	US Mail (1st Class)
24846	CHARLES RAMSEY, 12143 PLOWMAN RD., GORDO, AL, 35466	US Mail (1st Class)
24846	CHARLES S. BIGGS, 6952 DARLIE LANE, DELTON, MI, 49046	US Mail (1st Class)
24846	CHARLES WOLFFIS, 1485 DELYNN DRIVE, CENTERVILLE, OH, 45459	US Mail (1st Class)
24846	CHEM SERV, 3205 MAVERICK DR, KILGORE, TX, 75662	US Mail (1st Class)
24846	CHEM TREND INC., LOCK BOX # 22002, 22002 NETWORK PLACE, CHICAGO, IL, 60673-1220	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	CHEMBAR, INC., 1089 CLAYCRAFT ROAD, GAHANNA, OH, 43230	US Mail (1st Class)
24846	CHEMCENTRAL, 7050 W 71ST ST, CHICAGO, IL, 60638-5902	US Mail (1st Class)
24846	CHEM-DRY OF RICHLAND CO., 121 W. COOK RD., MANSFIELD, OH, 44907	US Mail (1st Class)
24846	CHEMETALL OAKITE, OAKITE PRODUCTS, INC., 22040 NETWORK PL., CHICAGO, IL, 60673-1220	US Mail (1st Class)
24846	CHEMIR/POLYTECH LABS, 2672 METRO BLVD, MARYLAND HEIGHTS, MO, 63043	US Mail (1st Class)
24846	CHEM-PAK, 242 CORNING WAY, MARTINSBURG, WV, 25401	US Mail (1st Class)
24846	CHEMSEARCH, 2727 CHEMSEARCH BLVD, IRVING, TX, 75062-6454	US Mail (1st Class)
24846	CHEM-TREND, INC., 6600 RELIABLE PKWY., CHICAGO, IL, 60686	US Mail (1st Class)
24846	CHERYL WESTFALL, 2601 S. RIDGE RD EAST, ASHTABULA, OH, 44004	US Mail (1st Class)
24846	CHEVRON PHILLIPS, 10001 SIX PINES RD, THE WOODLANDS, TX, 77380-1498	US Mail (1st Class)
24846	CHEVRON PHILLIPS CHEMICAL, 1001 SIX PINES DR., THE WOODLANDS, TX, 77380	US Mail (1st Class)
24846	CHEVRON PHILLIPS CHEMICAL CO., 4358 COLLECTION DRIVE, CHICAGO, IL, 60693	US Mail (1st Class)
24846	CHICAGO RIVET & MACHINE, 901 FRONTENAC RD, NAPERVILLE, IL, 60563-1744	US Mail (1st Class)
24846	CHISSO AMERICA INC, 1920 THOREAU DR N, SCHAUMBURG, IL, 60173-4176	US Mail (1st Class)
24846	CHRIS ORT, 290 MCCORMICK ST., LAPEER, MI, 48446	US Mail (1st Class)
24846	CHRIS TERRASI, 4411 OAK ARBOR COURT NE, GRAND RAPIDS, MI, 49525	US Mail (1st Class)
24846	CHRISTIAN CLEANING SERVICE, 329 E NEPESSING ST APT 3, LAPEER, MI, 48446-2398	US Mail (1st Class)
24846	CHRISTIE RANDALL, 290 MCCORMICK STREET, LAPEER, MI, 48446	US Mail (1st Class)
24846	CHRISTINA MOUTON, 8707 SAMUEL BARTON DRIVE, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	CHRISTINE HESS, 985 FALLS CREEK DR., VANDALIA, OH, 45377	US Mail (1st Class)
24846	CHRISTINE KALICKY, 10047 KEYSTONE DRIVE, LAKE CITY, PA, 16423	US Mail (1st Class)
24846	CHRISTINE LIPPERT, 119 E DAYTON STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	CHRISTINE VASSEN, 9976 BATEMAN AVE., CRANSVILLE, PA, 16410	US Mail (1st Class)
24846	CHRISTOPHER ORT, 1821 BROOKFIELD, APT 4, LAPEER, MI, 48446	US Mail (1st Class)
24846	CHRISTOPHER R. HURLEY, 12225 LONG LAKE DRIVE, SPARTA, MI, 49345	US Mail (1st Class)
24846	CHROMPARTS INC., 828 HALL AVE., DAYTON, OH, 45404	US Mail (1st Class)
24846	CHUCK OHL, 12577 G.A.R. HIGHWAY, CHARDON, OH, 44024	US Mail (1st Class)
24846	CHURCHES LUMBER YARDS, 107 S. SQUIRREL ROAD, AUBURN HILLS, MI, 48326-3254	US Mail (1st Class)
24846	CICCHINI ENTERPRISES, 5239 FULTON DR NW, CANTON, OH, 44718	US Mail (1st Class)
24846	CII TECHNOLOGIES, 1396 CHARLOTTE HWY, FAIRVIEW, NC, 28730-8544	US Mail (1st Class)
24846	CIMMETRY SYSTEMS CORP., 13694 COLLECTIONS CENTER DR., CHICAGO, IL, 60693	US Mail (1st Class)
24846	CINCINNATI FAN & VENTILATOR CO.,INC, PO BOX 640338, CINCINNATI, OH, 45264-0338	US Mail (1st Class)
24846	CINCINNATI TEST SYSTEMS, 5555 DRY FORK ROAD, CLEVES, OH, 45002	US Mail (1st Class)
24846	CINDY BRUMBAUGH, 3560 W. MARKET ST., STE. 340, AKRON, OH, 44333	US Mail (1st Class)
24846	CINDY WOODIE, 14715 WINDFIELD DR., NOVELTY, OH, 44072	US Mail (1st Class)
24846	CINDY'S CLEANING SERVICE, 116 N. HIGHLAND AVE., BUCYRUS, OH, 44820	US Mail (1st Class)
24846	CINTAS CORPORATION, 1300 BOLTONFIELD ST, COLUMBUS, OH, 43228	US Mail (1st Class)
24846	CINTAS CORPORATION, 39145 WEBB DR., WESTLAND, MI, 48185	US Mail (1st Class)
24846	CINTRON SCALE, INC., 3570 INTERCITY DR, DAYTON, OH, 45424-5124	US Mail (1st Class)
24846	CISCO, INC., 4565 HERMAN S.W., GRAND RAPIDS, MI, 49509	US Mail (1st Class)
24846	CITICAPITAL (SM), 1026 WILLIAMSBURG CIR, GRAYSLAKE, IL, 60030-7903	US Mail (1st Class)
24846	CITICORP VENDOR FINANCE, INC., 1 INTERNATIONAL BLVD, MAHWAH, NJ, 07495-0027	US Mail (1st Class)
24846	CITY OF ASHTABULA, SERVICE BILLING, PO BOX 2210, ASHTABULA, OH, 44005-2210	US Mail (1st Class)
24846	CITY OF CRESTLINE, 100 N SELTZER STREET, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	CITY OF GALION, RECREATION DEPT, C/O BILL BAUER, GALION, OH, 44833	US Mail (1st Class)
24846	CITY OF GREEN FINANCE DEPT, 5383 MASSILLON RD, CANTON, OH, 44720-1335	US Mail (1st Class)
24846	CITY OF LAPEER, 576 LIBERTY ST., LAPEER, MI, 48446	US Mail (1st Class)
24846	CITY OF LAPEER, 576 LIBERTY PARK, LAPEER, MI, 48446	US Mail (1st Class)
24846	CITY OF LAPEER, ACCT: 4-45280-05/ 4-45290-01, 576 LIBERTY ST., LAPEER, MI, 48446	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	CITY OF LIVONIA, LINDA GRIMSBY, TREASURER, 33000 CIVIC CENTER DR, LIVONIA, MI, 48154	US Mail (1st Class)
24846	CITY OF LIVONIA, 33000 CIVIC CENTER DRIVE, LIVONIA, MI, 48154	US Mail (1st Class)
24846	CITY OF LIVONIA, 33000 CIVIC CENTER DR, LIVONIA, MI, 48154-3060	US Mail (1st Class)
24846	CITY OF VANDALIA (2610), PO BOX 727, VANDALIA, OH, 45377	US Mail (1st Class)
24846	CITY OF VANDALIA, WATER, ACCT: 19-6020-3, 333 JAMES E. BOHANAN MEMORIAL DR., VANDALIA, OH, 45377-2394	US Mail (1st Class)
24846	CLAIRON METALS CORP., 11194 ALCOVY RD., COVINGTON, GA, 30014	US Mail (1st Class)
24846	CLARENCE B. REYNOLDS, 232 EAST MAIN ST., APT 2, GIRARD, PA, 16417	US Mail (1st Class)
24846	CLARK ENGINEERING CO.,INC., 700 MCMILLIN, DETROIT, MI, 48867	US Mail (1st Class)
24846	CLARK FIRE & SAFETY INC., 619 SOUTH WASHINGTON STREET, OWOSSO, MI, 48867	US Mail (1st Class)
24846	CLARK FOWLER ELECTRIC, 510 W HENRY, WOOSTER, OH, 44691	US Mail (1st Class)
24846	CLARK-MCKIBBEN SAFETY, 2001 HAMPTON RD., PO BOX 3244, ERIE, PA, 16508	US Mail (1st Class)
24846	CLARKSON FUND, CLARKSON UNIVERSITY, PO BOX 5525, POTSDAM, NY, 13699	US Mail (1st Class)
24846	CLAUDE SINTZ, INC., 300 S. CHESTNUT STREET, DESHLER, OH, 43516-1043	US Mail (1st Class)
24846	CLAUDE W. WIERCZK JR., 9670 E DAVID HIGHWAY, PENWAMD, MI, 48873	US Mail (1st Class)
24846	CLAUDIA DEISCH, 9006 COUNTY RD 30, GALION, OH, 44833	US Mail (1st Class)
24846	CLAUDIA LITTELL, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	CLAUDIO PINHEIRO, 31572 INDUSTRIAL RD., SUITE 300, LIVONIA, MI, 48150	US Mail (1st Class)
24846	CLAYTON GROUP SERVICES, INC., DEPARTMENT #187201, PO BOX 67000, DETROIT, MI, 48267-1872	US Mail (1st Class)
24846	CLEANING MANAGEMENT SYSTEMS, 4551 WILLOW PARKWAY, CLEVELAND, OH, 44125	US Mail (1st Class)
24846	CLEMENT COMMUNICATIONS, 10 LACRUE AVE, CONCORD, PA, 19331	US Mail (1st Class)
24846	CLEMMER MOVING & STORAGE, INC., 425 SCHOOLHOUSE RD., TELFORD, PA, 18969	US Mail (1st Class)
24846	CLEVELAND GASKET & MFG CO, 1602-10 WEST 116TH ST., CLEVELAND, OH, 44102	US Mail (1st Class)
24846	CLEVELAND INDIANS, JACOBS FIELD, 2401 ONTARIO STREET, CLEVELAND, OH, 44115-4003	US Mail (1st Class)
24846	CLEVELAND MACHINERY SALES, 5000 BROOKPARK RD., CLEVELAND, OH, 44134	US Mail (1st Class)
24846	CLEVELAND METROPARKS ZOO, 3900 WILDLIFE WAY, CLEVELAND, OH, 44109	US Mail (1st Class)
24846	CLEVELAND STATE UNIVERSITY, OFFICE OF BURSAR, 2121 EUCLID AVE., ROOM 460, CLEVELAND, OH, 44115-2214	US Mail (1st Class)
24846	CLEVELAND TIME CLOCK, 4750 STATE ROAD, CLEVELAND, OH, 44109-5251	US Mail (1st Class)
24846	CLIFF`S MOBILE LOCKSMITH SVC, 579 CRANWOOD CIRCLE, NEW LEBANON, OH, 45345	US Mail (1st Class)
24846	CLIFTON AUTOMATIC INC., 9937 SAMPSON AVENUE, PO BOX 421, LAKE CITY, PA, 16423-1599	US Mail (1st Class)
24846	CLINTON ALUMINUM &, STAINLESS STEEL, PO BOX 75702, CLEVELAND, OH, 44101-4755	US Mail (1st Class)
24846	CLINTON ALUMINUM & STAINLESS, 6270 VAN BUREN RD, CLINTON, OH, 44216	US Mail (1st Class)
24846	CLINTON ENERGY, MANAGEMENT SERVICES, INC., PO BOX 846053, DALLAS, TX, 75284-6053	US Mail (1st Class)
24846	CLINTON MCENTYRE MD, 2330 UNIVERSITY BLVD S, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	CMA ETTWORTH, INC., 1020 W EAU GALLIE BLVD, SUITE A, MELBOURNE, FL, 32935	US Mail (1st Class)
24846	CMS OIL COMPANY, 30 N SAGINAW STREET, PO BOX 730, LAPEER, MI, 48446	US Mail (1st Class)
24846	CNC LOGISTICS, 205 NORTH CAPITAL AVE., SUITE A, CORYDON, IN, 47112	US Mail (1st Class)
24846	COAST COMPUTER PRODUCTS, 1740 W KAELLA AVE, SUITE K, ORANGE, CA, 92867	US Mail (1st Class)
24846	COAST EQUIPMENT & SVCS. CO., 23561 YOUNG STREET, SEAFORD, DE, 19973	US Mail (1st Class)
24846	COAST TO COAST, 16912 PARTHENIA ST, NORTHRIDGE, CA, 91343	US Mail (1st Class)
24846	COASTAL TRAINING TECHNOLOGIES, 500 STUDIO DR, VIRGINIA BEACH, VA, 23452	US Mail (1st Class)
24846	COBANE CORPORATION, 19640 HARPER AVE, GROSSE POINTE WOODS, MI, 48236-1966	US Mail (1st Class)
24846	COBRA SYSTEMS, INC., 1230 LEXINGTON AVE, MANSFIELD, OH, 44907	US Mail (1st Class)
24846	CODY WARREN, 985 FALLS CREEK DR., VANDALIA, OH, 45377	US Mail (1st Class)
24846	COLE CHEMICAL & DIST, 900 THREADNEEDLE ST STE 350, HOUSTON, TX, 77079-2945	US Mail (1st Class)
24846	COLE PARMER INSTRUMENT CO., DEPT. 77-6391, CHICAGO, IL, 60678-6391	US Mail (1st Class)
24846	COLLABORATIVE TESTING SER, 21331 GENTRY DR, STERLING, VA, 20166	US Mail (1st Class)
24846	COLLINS & AIKMAN, GLOBAL CARPET & ADHESIVES, GLENN ROAD, TROY, NC, 27371	US Mail (1st Class)
24846	COLORITE POLYMERS, 101 RAILROAD AVE, RICHFIELD, NJ, 07657	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	COLORTRONICS INC., 155 9TH AVE, SUITE A, RUNNEMEDE, NJ, 08078	US Mail (1st Class)
24846	COLUMBIA MARKING TOOLS, 42600 EXECUTIVE DRIVE, HARRISSON TWP., MI, 48045	US Mail (1st Class)
24846	COLUMBIAN CHEMICAL CO., 1800 W O'COMMONS CT, MARIETTA, GA, 30062	US Mail (1st Class)
24846	COLUMBIANA COUNTY CSEA, 126 EAST CHESTNUT, LISBON, OH, 44432-0491	US Mail (1st Class)
24846	COLUMBUS GASKET CO., 1875 LONE EAGLE STREET, COLUMBUS, OH, 43228	US Mail (1st Class)
24846	COMALLOY INTERNATIONAL, 481 ALLIED DR, NASHVILLE, TN, 37211-3316	US Mail (1st Class)
24846	COMCAST METROPHONE, 1500 MARKET ST, PHILADELPHIA, PA, 19102-2100	US Mail (1st Class)
24846	COMMERCIAL DOOR, 20 BROADVIEW, TUSCALOOSA, AL, 35405	US Mail (1st Class)
24846	COMMERCIAL GROUP-DAYTON, 9955 GRAND RIVER AVE, DETROIT, MI, 48204	US Mail (1st Class)
24846	COMMERCIAL SOUND SYSTEMS, 2875 BLACK SEA RD, JEFFERSON, OH, 44047	US Mail (1st Class)
24846	COMMON CENTS WORKS, 157 NORTH MAIN STREET, ALBION, PA, 16401	US Mail (1st Class)
24846	COMMONWEALTH OF PA, DEPT OF TRANSPORTATION, BUREAU OF MOTOR VEHICLES, HARRISBURG, PA, 17104-2516	US Mail (1st Class)
24846	COMMUNITY SAFETY NE, 276 CAVALIER STREET, PEMBINA, ND, 58271	US Mail (1st Class)
24846	COMMUNITY SOUP BOWL, TUSCALOOSA MORNING ROTARY, 630 UNIVERSITY BLVD, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	COMPANY B INC, 255 SPINNAKER WAY, UNIT 2, CONCORD, ON, L4K 4J1CANADA	US Mail (1st Class)
24846	COMPASS PACKAGING, 18360 INDUSTRIAL CIRCLE, PARKMAN, OH, 44080	US Mail (1st Class)
24846	COMPDATA SURVEYS, 1713 EAST 123RD STREET, OLATHE, KS, 66061	US Mail (1st Class)
24846	COMPETITION CHEMICALS INC., 715 RAILROAD ST, IOWA FALLS, IA, 50126-2110	US Mail (1st Class)
24846	COMPLETE SURFACE TECH, 21338 CARLO DRIVE, CLINTON TWP, MI, 48038	US Mail (1st Class)
24846	COMPLETE TRAFFIC SERVICE INC., 8358 MUNSON RD., SUITE 102, MENTOR, OH, 44060	US Mail (1st Class)
24846	COMPONENTS FOR INDUSTRY, 210-12 TERRACE DRIVE, MUNDELEIN, IL, 60060	US Mail (1st Class)
24846	COMPOSITE SOLUTIONS, 23300 MERCANTILE RD., BOX 221112, BEACHWOOD, OH, 44122	US Mail (1st Class)
24846	COMPREHENSIVE LOGISTICS, 365 VICTORIA ROAD, AUSTINTOWN, OH, 44415	US Mail (1st Class)
24846	COMPRESSOR ENGR. CORP., 13330 FOLEY, DETROIT, MI, 48227	US Mail (1st Class)
24846	COMPRESSOR TECHNOLOGIES INC., 5800 SAFETY DRIVE, BELMONT, MI, 49306	US Mail (1st Class)
24846	COMPTROL INC, 9505 MIDWEST AVE., CLEVELAND, OH, 44125	US Mail (1st Class)
24846	COMPUNET CLINICAL LAB, 2308 SANDRIDGE DR, DAYTON, OH, 45439	US Mail (1st Class)
24846	COMPUSA, 14951 DALLAS PKWY, DALLAS, TX, 75254	US Mail (1st Class)
24846	COMPUSTAR SYSTEMS INC., 1335 E BRADFORD PKWY, SPRINGFIELD, MO, 65804-4265	US Mail (1st Class)
24846	COMPU-TECH, 7235 19 MILE ROAD, STERLING HEIGHTS, MI, 48314	US Mail (1st Class)
	COMPUTER ASSISTANCE INC., 8173 EMBURY ROAD, GRAND BLANC, MI, 48439	US Mail (1st Class)
24846	COMPUTER INTEGRATED SERVICES, 7938 COOLEY LAKE ROAD, SUITE 700W, WATERFORD, MI, 48327	US Mail (1st Class)
24846	COMPUTER PRINTER RESOURCES INC., 29760 SHIAWASSEE RD., FARMINGTON HILLS, MI, 48336	US Mail (1st Class)
24846	COMPUWARE CORPORATION, DRAWER # 64376, DETROIT, MI, 48264-0376	US Mail (1st Class)
24846	COMSOURCE, 2130 AUSTIN DRIVE, ROCHESTER HILLS, MI, 48309	US Mail (1st Class)
24846	CONCERTIAL PEALITY L. C. 4077 PIONEED DR. WALLED LAKE MI. 40300	US Mail (1st Class)
24846	CONCEPTUAL REALITY L.L.C., 4077 PIONEER DR., WALLED LAKE, MI, 48390	US Mail (1st Class)
24846	CONDITIONALITY ASSURANCE, 9950 S PURAND RD, DURAND MI 48420	US Mail (1st Class)
24846	CONDIT QUALITY ASSURANCE, 8950 S DURAND RD., DURAND, MI, 48429	US Mail (1st Class)
24846	CONNEAUT LAKE DADK 12292 CENTED STREET CONNEAUT LAKE DA 16216	US Mail (1st Class)
24846	CONNEAUT LAKE PARK, 12382 CENTER STREET, CONNEAUT LAKE, PA, 16316 CONNECTIONS, 14770 MAPLEWOOD DR., SUITE 107, BURTON, OH, 44021	US Mail (1st Class) US Mail (1st Class)
24846	CONNIE BOGER, 290 MCCORMICK, LAPEER, MI, 48446	US Mail (1st Class)
24846 24846	CONNIE BOGER, 290 MCCORMICK, LAPEER, MI, 46446  CONNIE MILLER, 10047 KEYSTONE DRIVE, LAKE CITY, PA, 16423	US Mail (1st Class)
24846	CONRAD KACSIK INSTRUMENT SYS. INC., 30925 AURORA RD, SOLON, OH, 44139-2731	US Mail (1st Class)
24846	CONREM, INC., 7349 PEPPERMILL PKWY STE G, N CHARLESTON, SC, 29418-7413	US Mail (1st Class)
24040	CONTREM, 1140., 1070   ELLI ENWILL   INVIT STE G, IN CHARLESTON, 30, 23410-1413	50 man (15t 01855)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	CONSTELLATION NEW ENERGY, ACCT: 4389, CORP HEADQUARTERS 750 EAST PRATT ST 14 FL, BALTIMORE, MD, 21202	US Mail (1st Class)
24846	CONSTELLATION NEWENERGY, GAS DIVISION, LLC, PO BOX 2059, CAROL STREAM, IL, 60132-2059	US Mail (1st Class)
24846	CONSTITUTION MOLD & ENG., 56568 N BAY DR, CHESTERFIELD, MI, 48051-3738	US Mail (1st Class)
24846	CONSUMER CREDIT, COUNSELING SERVICE, 320 HIGH STREET SUITE A, WARREN, OH, 44481	US Mail (1st Class)
24846	CONTINENTAL PAPER, PO BOX 64628, DETROIT, MI, 48264-0123	US Mail (1st Class)
24846	CONTRACT PROFESSIONALS, 4141 WALTON BLVD, WATERFORD TOWNSHIP, MI, 48329	US Mail (1st Class)
24846	CONTRACT PROFESSIONALS, INC., 4141 W WALTON BLVD, WATERFORD, MI, 48329-4179	US Mail (1st Class)
24846	CONTROL COMPONENTS, 2275 LOWER WETUMPKA RD., MONTGOMERY, AL, 36110	US Mail (1st Class)
24846	CONTROL RISKS GROUP, 1749 OLD MEADOW ROAD, SUITE 120, MCLEAN, VA, 22102	US Mail (1st Class)
24846	CONTROL SYSTEMS, 3822 HWY 25, MONTEVALLO, AL, 35115	US Mail (1st Class)
24846	CONTROLCO, INC., 30903 VIKING PKWY, CLEVELAND, OH, 44145-1054	US Mail (1st Class)
24846	CONTROLLED PRECISION, 37560 ENTERPRISE COURT, FARMINGTON HILLS, MI, 48331	US Mail (1st Class)
24846	CONTROLLED TEMPERATURE, INC., 1897 BEST DRIVE, WALLED LAKE, MI, 48390	US Mail (1st Class)
24846	CONVERTED PRODUCTS, 8501 WEST TOWER AVE., MILWAUKEE, WI, 53224	US Mail (1st Class)
24846	CONVERTERS, INC., 1617 REPUBLIC RD, HUNTINGDON VALLEY, PA, 19006-1807	US Mail (1st Class)
24846	CONVOY, INC., 1811 20TH ST NE, CANTON, OH, 44714-2123	US Mail (1st Class)
24846	CON-WAY CENTRAL EXPRESS, 110 PARKLAND PLZ, ANN ARBOR, MI, 48103-6201	US Mail (1st Class)
24846	CON-WAY TRANSPORTATION SERVICE, 5555 RUFE SNOW DR, SUITE 5515, FORT WORTH, TX, 76180	US Mail (1st Class)
24846	COOPER AUTOMATION & MACHINE CO. INC, 260 INDUSTRIAL DR, OXFORD, AL, 36203-2058	US Mail (1st Class)
24846	COOPER'S TOOLS, 8419 N. GENESEE ROAD, MT. MORRIS, MI, 48458	US Mail (1st Class)
24846	COOPER-STANDARD AUTOMOTIVE INC, 39550 ORCHARD HILL PL, NOVI, MI, 48375-5329	US Mail (1st Class)
24846	COPECO, INC., 7835 FREEDOM AVE NW, NORTH CANTON, OH, 44720-6907	US Mail (1st Class)
24846	COPELEY HILL INC., 5951 HAVENS RD., DRYDEN, MI, 48428	US Mail (1st Class)
24846	CORA M. MASSEY, 1117 14TH ST NE, CANTON, OH, 44705-1201	US Mail (1st Class)
24846	CORNELL SUPPLY CO, 5625 ENTERPRISE BLVD, TOLEDO, OH, 43612	US Mail (1st Class)
24846	CORNERSTONE STAFFING INC., 10801 S SAGINAW ST, STE A, GRAND BLANC, MI, 48439-8126	US Mail (1st Class)
24846	CORPAD CO., INC., 555 PARK AVE E, MANSFIELD, OH, 44905-2871	US Mail (1st Class)
24846	CORPORATE CONSULTING SERVICE, 1145 HIGHBROOK AVE #500, AKRON, OH, 44301	US Mail (1st Class)
24846	CORPORATE EXPRESS OFFICE PDTS., 1301 INTERNATIONAL PARKWAY, WOOLRIDGE, IL, 60517	US Mail (1st Class)
24846	CORPORATE GLASS ETCHING, 2640 WEST 17TH STREET, ERIE, PA, 16505	US Mail (1st Class)
24846	CORRUGATED CONTAINERS, INC., 2125 LANDMARK DR., CAYCE, SC, 29033	US Mail (1st Class)
24846	COSTELLO ENTERPRISES, 56358 PRECISION DRIVE, CHESTERFIELD TWP, MI, 48051	US Mail (1st Class)
24846	COT-PURITECH, 2993 PERRY DRIVE, SW, CANTON, OH, 44706	US Mail (1st Class)
24846	COULTER PAINTING, 475 S COURT STREET, LAPEER, MI, 48446	US Mail (1st Class)
24846	COUNTRY CLUB OF TUSCALOOSA, 3700 6TH STREET, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	COUNTRY TREASURES FLORIST, PO BOX 1, BROOKVILLE, OH, 45309-0001	US Mail (1st Class)
24846	COUNTY SEWER DEPT., ACCT: 10-02615-00-1, 1701 MAHONING RD NE, CANTON, OH, 44705	US Mail (1st Class)
24846	COURTNEY SHUMAC, 17187 N LAUREL PK DR., SUITE #208, LIVONIA, MI, 48152	US Mail (1st Class)
24846	COURTYARD MARRIOTT, 4115 COURTNEY DR., TUSCALOOSA, AL, 35405	US Mail (1st Class)
24846	COX MIP, PO BOX 5635, BRANDON, MS, 39047	US Mail (1st Class)
24846	COXCO INC., 3603 PINE LANE SE, BESSEMER, AL, 35022	US Mail (1st Class)
24846	COX-COLVIN & ASSOC., 7750 CORPORATE BLVD, PLAIN CITY, OH, 43064	US Mail (1st Class)
24846	CPR/HEALTH INFORMATION, SERVICES, PLLC, 9320 MAYFLOWER COURT, PLYMOUTH, MI, 48170	US Mail (1st Class)
24846	CRAIG`S SIGNS, 1498 N. SAGINAW STREET, SUITE 2, LAPEER, MI, 48446	US Mail (1st Class)
24846	CRAIG-RICHARD, INC., PROMOTIONAL PRODUCTS, 24151 TELEGRAPH ROAD, SOUTHFIELD, MI, 48034	US Mail (1st Class)
24846	CRAIN'S DETROIT BUSINESS, DEPARTMENT 77940, DETROIT, MI, 48277-0940	US Mail (1st Class)
24846	CRAMERS` INC., 4944 SOUTHWAY S.W, CANTON, OH, 44706	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	CRANE AMERICA SERVICES, 88048 EXPEDITE WAY, CHICAGO, IL, 60695-0001	US Mail (1st Class)
24846	CRANE AMERICA SERVICES INC., 88048 EXPEDITE WAY, CHICAGO, IL, 60695-0001	US Mail (1st Class)
24846	CRANE PRO SERVICES, REGION AMERICAS CORPORATE OFFICE, 4401 GATEWAY BLVD, SPRINGFIELD, OH, 45502-9339	US Mail (1st Class)
24846	CRAWFORD COUNTY SANITARY, LANDFILL DIST. NO. 1, 5128 LINCOLN HWY EAST, BUCYRUS, OH, 44820	US Mail (1st Class)
24846	CRAWFORD COUNTY TREASURER, GARY COLE, COURT HOUSE, P.O. BOX 565, BUCYRUS, OH, 44820	US Mail (1st Class)
24846	CRAWFORD CTY BOARD, OF COMMISSIONERS, C/O CITY OF CRESTLINE, 100 N SELTZER STREET, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	CRAWFORD CTY MUNICIPAL COURT, 301 HARDING WAY EAST, GALION, OH, 44833	US Mail (1st Class)
24846	CRAWFORD RESOURCES INC., 105 WEST MARKET ST., SUITE 100, SANDUSKY, OH, 44870	US Mail (1st Class)
24846	CREATIVE ENG. POLYMER PROD. LLC, 15332 OLD STATE ROAD, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	CREATIVE ENG. POLYMER PROD. LLC, 900 SOUTH WILEY STREET, CRESTLINE, OH, 44827-0320	US Mail (1st Class)
24846	CREATIVE ENGINEERED POLYMER PRODUCTS, LLC, 3560 WEST MARKET ST., SUITE 340, AKRON, OH, 44333	US Mail (1st Class)
24846	CREATIVE FOAM CORP., 300 N ALLOY DR, FENTON, MI, 48430-2648	US Mail (1st Class)
24846	CREATIVE IMPRINT SYSTEMS, INC., 2670 W. 11TH ST., ERIE, PA, 16505-4168	US Mail (1st Class)
24846	CREST PRODUCTS, INC., 75 REMITTANCE DRIVE, SUITE #6791, CHICAGO, IL, 60675-6791	US Mail (1st Class)
24846	CRESTLINE ACTIVITY CENTER, 216 N STELTZER ST, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	CRESTLINE ADVOCATE, PO BOX 226, 312 NORTH SELTZER STREET, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	CRESTLINE AREA CHAMBER, OF COMMERCE, PO BOX 355, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	CRESTLINE C.E.P. PETTY CASH, JERRY SCHARDT, 900 S. WILEY ST., CRESTLINE, OH, 44827-0320	US Mail (1st Class)
24846	CRESTLINE FIRE DEPARTMENT, ATTN: WILLIAM SHARP, 100 N. SELTZER STREET, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	CRESTLINE HARVEST-ANTIQUE FEST, C/O CRESTLINE CITY COUNCIL, 100 N. SELTZER ST., CRESTLINE, OH, 44827	US Mail (1st Class)
24846	CRESTLINE HIGH SCHOOL, 7854 OLDFIELD ROAD, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	CRESTLINE JR. FOOTBALL, 308 BAUER AVE., CRESTLINE, OH, 44827	US Mail (1st Class)
24846	CRESTLINE POLICE DEPT., 100 N. SELTZER, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	CRESTLINE SPECIALTIES, INC., 70 MOUNT HOPE AVE, LEWISTON, ME, 04240-1021	US Mail (1st Class)
24846	CRESTLINE WATER WORKS, 100 N SELTZER ST, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	CRESTLINE WATER WORKS, ACCT: 01-42-1; 01-124-1, 100 N. SELTZER ST., CRESTLINE, OH, 44827	US Mail (1st Class)
24846	CRESTLINE WOOD PRODUCTS, 8511 SHILOH NORWALK RD., SHILOH, OH, 44878	US Mail (1st Class)
24846	CRITTER CONTROL, 12740 DEVONSON STREET, ALLIANCE, OH, 44601	US Mail (1st Class)
24846	CROCKER LIMITED, 1510 N. MAIN STREET, THREE RIVERS, MI, 49093	US Mail (1st Class)
24846	CROSBY MOOK OFFICE SUP., 558 HIGH ST., WARREN, OH, 44483	US Mail (1st Class)
24846	CROSS SALES, PO BOX 65486, CHARLOTTE, NC, 28265-0486	US Mail (1st Class)
24846	CROUCH FIRE & SAFETY, 8210 EXPANSION WAY, DAYTON, OH, 45424	US Mail (1st Class)
24846	CROWN LIFT TRUCKS, 44 S WASHINGTON ST, NEW BREMEN, OH, 45869-1247	US Mail (1st Class)
24846	CROWN TOOL & SUPPLY CO., 30845 SOLON RD. UNIT #2, PO BOX 39337, SOLON, OH, 44139	US Mail (1st Class)
24846	CRP INDUSTRIES, INC., 1 MINUE STREET, CARTERET, NJ, 07008-1198	US Mail (1st Class)
24846	CRUCIBLE SPECIALTY METAL, 575 STATE FAIR BLVD, SYRACUSE, NY, 13209-1560	US Mail (1st Class)
24846	CRUM MANUFACTURING, 1265 WATERVILLE-MONCLOVA, PO BOX 28, WATERVILLE, OH, 43566	US Mail (1st Class)
24846	CRYOGENIC SYSTEMS & PARTS, 3595 CADILLAC AVE., COSTA MESA, CA, 92626	US Mail (1st Class)
24846	CSI INK, 1335 E BRADFORD PKWY, SPRINGFIELD, MO, 65805	US Mail (1st Class)
24846	CTJ BUSINESS SERVICES, PO BOX 100, COTTONDALE, AL, 35453	US Mail (1st Class)
24846	CTR SYSTEMS, 555 KEYSTONE DR, WARRENDALE, PA, 15086-7538	US Mail (1st Class)
24846	CTX, INC., PO BOX 830, DEARBORN HEIGHTS, MI, 48127	US Mail (1st Class)
24846	CULLIGAN SOFT WATER SERV, 1 CULLIGAN PKWY, NORTHBROOK, IL, 60062-6209	US Mail (1st Class)
24846	CULLIGAN WATER, 1 CULLIGAN PKWY, NORTHBROOK, IL, 60062-6209	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	CUMBERLAND ENGINEERING CORP., SPARE PARTS EXPRESS, PO BOX 846034, BOSTON, MA, 02284-6034	US Mail (1st Class)
24846	CUMBERLAND ENGINEERING CORP., 100 RODDY AVE, SOUTH ATTLEBORO, MA, 02703-7900	US Mail (1st Class)
24846	CURBELL PLASTICS, INC., 28455 SCHOOLCRAFT RD #5, LIVONIA, MI, 48150	US Mail (1st Class)
24846	CURT CLARK, 2070 S. ALMONT AVE, IMLAY CITY, MI, 48444	US Mail (1st Class)
24846	CURTIS 1000 INC., 1725 BRECKINRIDGE PKWY STE 500, DULUTH, GA, 30096-8994	US Mail (1st Class)
24846	CURTIS STEEL AND SUPPLY, INC., 1210 CURTIS STREET, AKRON, OH, 44301	US Mail (1st Class)
24846	CUSTOM CHROME PLATING, 828 HALL AVE., DAYTON, OH, 45404	US Mail (1st Class)
24846	CUSTOM COMPONENTS CORP., 820 MONROE AVENUE. NW, SUITE 215, GRAND RAPIDS, MI, 49503	US Mail (1st Class)
24846	CUSTOM ETCH INC., 1813 WEST STATE STREET, NEW CASTLE, PA, 16101	US Mail (1st Class)
24846	CUSTOM FLORAL DESIGN, 2193 ST. RT. 305, CORTLAND, OH, 44410	US Mail (1st Class)
24846	CUSTOM NICKEL PLATING, 45 N CLINTON ST., DAYTON, OH, 45402	US Mail (1st Class)
24846	CUSTOM WELD CORP., 1500 HENRY AVE. SW, CANTON, OH, 44706	US Mail (1st Class)
24846	CUTRITE, 2201 COMMERCE DR., FREMONT, OH, 43420	US Mail (1st Class)
24846	CUTTER CORPORATION, 4903 DAWN AVE., EAST LANSING, MI, 48823	US Mail (1st Class)
24846	CYBERMETRICS CORPORATION, 16100 NORTH GREENWAY-HADYEN LOOP, SUITE 100, SCOTTSDALE, AZ, 85260	US Mail (1st Class)
24846	CYPRESS INN, 501 RICE MINE ROAD, NORTHPORT, AL, 35476	US Mail (1st Class)
24846	D & D MACHINERY MOVERS INC., 24200 GROESBECK HWY, WARREN, MI, 48089	US Mail (1st Class)
24846	D & H FIRE PROTECTION CO., 9162 GALE RD, OTISVILLE, MI, 48463	US Mail (1st Class)
24846	D & W AUTO GLASS, INC., PO BOX 426, 55 ASHLAND ROAD, MANSFIELD, OH, 44901	US Mail (1st Class)
24846	D & W PAINT & EQUIPMENT, 325 E. 3RD STREET, MANSFIELD, OH, 44902	US Mail (1st Class)
24846	D S TECHNOLOGY, RR 1, WOODSTOCK, ON, N4S 7V6CANADA	US Mail (1st Class)
24846	D&D MANUFACTURING, 49 MITCHELL RD, IPSWICH, MA, 01938-1218	US Mail (1st Class)
24846	D&E QUALITY,INC., 1603 MAHOGANY RUN DRIVE, LAGRANGE, KY, 40031	US Mail (1st Class)
24846	D&F CORPORATION, 42455 MERRILL ROAD, STERLING HEIGHTS, MI, 48314	US Mail (1st Class)
24846	D. T. FOWLER MFG. CO. INC, 101 N MAPLELEAF ROAD, PO BOX 70, LAPEER, MI, 48446	US Mail (1st Class)
24846	D.A. CAMPBELL AND COMPANY, 30285 BRUCE INC. PKW., SUITE A, SOLON, OH, 44139	US Mail (1st Class)
24846	D.C. SYSTEMS, INC., 1251 INDUSTRIAL PKWY N, BRUNSWICK, OH, 44212	US Mail (1st Class)
24846	D.T. FOWLER MFG. CO. INC., 101 N MAPLELEAF RD, LAPEER, MI, 48446	US Mail (1st Class)
24846	DA STIRLING INC, 2740 HUDSON DR, CUYAHOGA FALLS, OH, 44221	US Mail (1st Class)
	DA STIRLING, INC., 2740 HUDSON DR., CUYAHOGA FALLS, OH, 44221	US Mail (1st Class)
24846	DADCO INC., 43850 PLYMOUTH OAKS BLVD., PLYMOUTH, MI, 48170	US Mail (1st Class)
24846	DAIMLER CHRYSLER, BEHR DAYTON THERMAL LL 05407, 1600 WEBSTER STREET, DAYTON, OH, 45404	US Mail (1st Class)
24846	DAIMLER CHRYSLER, WARREN TRUCK ASSY., PLANT 02452, 21500 MOUND RD., WARREN, MI, 48901	US Mail (1st Class)
24846	DAIMLERCHRYSLER, C/O EXCEL-TOLEDO, 1717 MATZINGER RD., TOLEDO, OH, 43612	US Mail (1st Class)
24846	DAISY BEDDELL, 1581 PEKIN DR SE, LOT #2, EAST CANTON, OH, 44730-9548	US Mail (1st Class) US Mail (1st Class)
24846	DALIN MOLD, 2915 INDUSTRIAL PARK DR., AUSTINBURG, OH, 44010  DALIN MOLD, PO BOX 220, 2915 INDUSTRIAL PARK DR, AUSTINBURG, OH, 44010	US Mail (1st Class)
24846 24846	DAN BEVIER, 119 EAST DAYTON STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	DAN HOWARD, 207 LARIAT DRIVE, LAPEER, MI, 48446	US Mail (1st Class)
24846	DANA BARRE, 5166 AMHERST ROAD, ERIE, PA, 16506	US Mail (1st Class)
24846	DANBOISE MECHANICAL INC., 31625 GRAND RIVER AVE., FARMINGTON, MI, 48336	US Mail (1st Class)
24846	DANIEL ELECTRIC CO., 3000 AUBURN RD., AUBURN HILLS, MI, 48326	US Mail (1st Class)
24846	DANIEL F. SMITH, RICHLAND COUNTY TREASURER, 50 PARK AVE, EAST, MANSFIELD, OH, 44902-	US Mail (1st Class)
	1895	
24846	DANIEL FIELD, 12616 W. LAKE ROAD, EAST SPRINGFIELD, PA, 16411	US Mail (1st Class)
24846	DANIEL MANN, 6339 E CARPENTER ROAD, FLINT, MI, 48506	US Mail (1st Class)
24846	DANIEL R OZIO, 10000 SETTLEMENT HOUSE, CENTERVILLE, OH, 45458	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	DANKA OFFICE IMAGING, 4388 COLLECTIONS CENTER DRIVE, CHICAGO, IL, 60693	US Mail (1st Class)
24846	DANNY POSEY, 1401 INDUSTRIAL PK DR, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	DANSER, INC., PO BOX 4098, MURPHYTOWN RD., PARKERSBURG, WV, 26104	US Mail (1st Class)
24846	DANTHORPE CORP./LEXCO ENG., 3434 UNIONVILLE PIKE, HATFIELD, PA, 19440-1829	US Mail (1st Class)
24846	DAPRA MARKING SYSTEMS, 66 GRANBY ST, BLOOMFIELD, CT, 06002-3512	US Mail (1st Class)
24846	DARIC I LONG, 9135 STATION ROAD, NORTH EAST, PA, 16428	US Mail (1st Class)
24846	DARING COMPANY, PO BOX 8, OXFORD, MI, 48371-0008	US Mail (1st Class)
24846	DARLENE GERBER, 5240 S RIVER, GENEVA, OH, 44041	US Mail (1st Class)
24846	DARLENE HART, 2271 DORHILL STREET, NW, UNIONTOWN, OH, 44685	US Mail (1st Class)
24846	DARLENE TUCKER, 3131 COLUMBUS RD., CANTON, OH, 44705	US Mail (1st Class)
24846	DARREN HAWK, 1401 INDUSTRIAL PARK DRIVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	DAR-TECH, INC., 16485 ROCKSIDE RD, CLEVELAND, OH, 44137	US Mail (1st Class)
24846	DARYL L. MOWEN, 4109 SKINNER ROAD, LAPEER, MI, 48446	US Mail (1st Class)
24846	DATA COMM WAREHOUSE, 7077 COLLECTION CENTER DR., CHICAGO, IL, 60693-0072	US Mail (1st Class)
24846	DATA STREAM, 50 DATASTREAM PLAZA, GREENVILLE, SC, 29605	US Mail (1st Class)
24846	DAUBERT CHEMICAL CO., 4700 S CENTRAL AVE, CHICAGO, IL, 60638-1531	US Mail (1st Class)
24846	DAVE HALL, INC., 2589 VAN DYKE, MARLETTE, MI, 48453	US Mail (1st Class)
24846	DAVE TOBIAS, PO BOX 402, GRATIS, OH, 45330	US Mail (1st Class)
24846	DAVID BALTER, 985 FALLS CREEK DR., VANDALIA, OH, 45377	US Mail (1st Class)
24846	DAVID E LUDWIG, 10731 ETTER ROAD, WATERFORD, PA, 16441	US Mail (1st Class)
24846	DAVID E. TROYER, 14380 OLD STATE RD., MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	DAVID F. DICK, 1241 LONG LAKE COURT, BRIGHTON, MI, 48116	US Mail (1st Class)
24846	DAVID F. DICK, 1241 LONG LAKE CT., BRIGHTON, MI, 48116	US Mail (1st Class)
24846	DAVID G. MATHEWS, 4523 CARDINAL DR., RICHMOND, IN, 47374	US Mail (1st Class)
24846	DAVID HERRON, 3022 DAVISON AVE., ERIE, PA, 16504	US Mail (1st Class)
24846	DAVID J. BEANS, 1317 SHORB AVE NW, CANTON, OH, 44703	US Mail (1st Class)
24846	DAVID KUNSCHKE, 756 PEPPERMILL CR, LAPEER, MI, 48446	US Mail (1st Class)
24846	DAVID L. DIXON, 315 PERSHING NE, N CANTON, OH, 44720	US Mail (1st Class)
24846	DAVID L. ROHALEY, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	DAVID M. BUTLER, 985 FALLS CREEK DR., VANDALIA, OH, 45377	US Mail (1st Class)
24846	DAVID PRICE METAL SERVICES, INC., 360 EASTPARK DRIVE, NORWALK, OH, 44857	US Mail (1st Class)
24846	DAVID R BABB, 1743 ERIE AVE NW, MASSILLON, OH, 44646	US Mail (1st Class)
24846	DAVID SIMPSON, 2159 PARKTON RD., MT. MORRIS, MI, 48458	US Mail (1st Class)
24846	DAVID SUNSHEIN, 566 EAST DOROTHY LANE, KETTERING, OH, 45419	US Mail (1st Class)
24846	DAVID TRUXALL, 27504 BUCKINGHAM, LIVONIA, MI, 48154	US Mail (1st Class)
24846	DAVID WINTERS, 2601-A S. RIDGE RD. EAST, ASHTABULA, OH, 44004	US Mail (1st Class)
24846	DAVID YOUNG, 119 E DAYTON ST, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	DAVID ZUCKER, 900 S. WILEY ST., PO BOX 320, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	DAVIS INSTRUMENTS LLC, 4701 MOUNT HOPE DR., BALTIMORE, MD, 21215	US Mail (1st Class)
24846	DAVO INC., 3300 MERRITTVILLE HWY, UNIT 2, THOROLD, ON, L2V 4Y6CANADA	US Mail (1st Class)
24846	DAWN GUINTHER, 8707 SAMUEL BARTON DR., BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	DAWNE SNYDER, C/O CEP PRODUCTS, LLC, PO BOX 320, CRESTLINE, OH, 44827-0320	US Mail (1st Class)
24846	DAYS INN OF IMLAY CITY, DAYS INN C/O UB-S GROUP INC., 6692 NEWARK RD., IMLAY CITY, MI, 48444	US Mail (1st Class)
24846	DAYTIMERS, INC., 1 WILLOW LANE, EAST TEXAS, PA, 18046	US Mail (1st Class)
24846	DAYTON AREA CHAMBERS, 1 CHAMBER PLAZA, DAYTON, OH, 45402	US Mail (1st Class)
24846	DAYTON BLUE PRINT CO, 222 NORTH ST. CLAIR ST., DAYTON, OH, 45402-1230	US Mail (1st Class)
24846	DAYTON DAILY NEWS, 45 S LUDLOW ST, DAYTON, OH, 45402	US Mail (1st Class)
24846	DAYTON DOOR SALES INC., 1112 SPRINGFIELD ST, DAYTON, OH, 45403	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	DAYTON DRILL BUSHING CO., 1920 STANLEY AVE, DAYTON, OH, 45404-1121	US Mail (1st Class)
24846	DAYTON EMPLOYMENT GUIDE, 5301 ROBIN HOOD RD, NORFOLK, VA, 23513-2419	US Mail (1st Class)
24846	DAYTON FIRE PROTECTION, 7616 N MAIN ST, DAYTON, OH, 45415	US Mail (1st Class)
24846	DAYTON FREIGHT LINES, INC, 6450 POE AVE STE 311, DAYTON, OH, 45414-2647	US Mail (1st Class)
24846	DAYTON PRECISION PUNCH,IN, 1875 RADIO ROAD, PO BOX 710954, CINCINNATI, OH, 45271	US Mail (1st Class)
24846	DAYTON WINDUSTRIAL CO, 137 EAST HELENA STREET, DAYTON, OH, 45404	US Mail (1st Class)
24846	DCH CENTER FOR OCCUPATIONAL, HEALTH, 701 UNIVERSITY BLVD E, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	DCH REGIONAL MEDICAL CTR, 809 UNIVERSITY BLVD E, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	DCI CALIBRATION SERVICE INC., 717 HWY 67 S. STE 16, PRICEVILLE, AL, 35603	US Mail (1st Class)
24846	DE FABRIECK SA DE CV, CIPRES 218, COL DEL VALLE, SALTILLO COAH, 25160MEXICO	US Mail (1st Class)
24846	DEARING COMPRESSOR & PUMP CO., PO BOX 6044, YOUNGSTOWN, OH, 44501-6044	US Mail (1st Class)
24846	DEBORAH CIRINCIONE, 3560 W. MARKET ST., STE. 340, AKRON, OH, 44333	US Mail (1st Class)
24846	DEBRA WOODS, 8707 SAMUEL BARTON DR., BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	DECISIONONE CORPORATION, 426 W LANCASTER AVE, DEVON, PA, 19333-1510	US Mail (1st Class)
24846	DECK VENDING SERVICE, 1031 CHAMBERLAIN BLVD., CONNEAUT, OH, 44030	US Mail (1st Class)
24846	DECOMA - MYTOX, 600 WILSHIRE DR, TROY, MI, 48084-1625	US Mail (1st Class)
24846	DEE STREASICK, 3225 MONTANA AVE, FLINT, MI, 48506	US Mail (1st Class)
24846	DEFIANCE TESTING &, ENGINEERING SERVICES, DRAWER #67-480A, DETROIT, MI, 48267	US Mail (1st Class)
24846	DEHN ENGINEERING INC., 633 APPOLINE CT., ROCHESTER, MI, 48307	US Mail (1st Class)
24846	DELAGE LANDEN FINANCIAL SVS., 1111 OLD EAGLE SCHOOL RD, WAYNE, PA, 19087-1453	US Mail (1st Class)
24846	DELAWARE VALLEY CORP., 500 BROADWAY, LAWRENCE, MA, 01841-3002	US Mail (1st Class)
24846	DELL FINANCIAL SERVICES, PAYMENT PROCESS CENTER, 4293 COLLECTION CENTER DR., CHICAGO, IL, 60693	US Mail (1st Class)
24846	DELL MARKETING L.P., C/O DELL USA L.P., PO BOX 643561, PITTSBURGH, PA, 15264-3561	US Mail (1st Class)
24846	DELL MARKETING, LP, C/O DELL USA, PO BOX 534418, ATLANTA, GA, 30353-4118	US Mail (1st Class)
24846	DELL MARKING SYSTEMS, INC, 721 WANDA DRIVE, FERNDALE, MI, 48220	US Mail (1st Class)
24846	DELLAGNESE COMPANIES, 4000 EMBASSY PARKWAY, SUITE 400, AKRON, OH, 44333	US Mail (1st Class)
24846	DELOITTE TAX, LLP, 13943 COLLECTIONS CENTER DR., CHICAGO, IL, 60693	US Mail (1st Class)
24846	DELORES CHATMAN, 26642 ANDOVER ST, INKSTER, MI, 48141	US Mail (1st Class)
24846	DELPHI HARRISON THERMAL, GENERAL MOTORS CORP., PO BOX 71384, CHICAGO, IL, 60694-1384	US Mail (1st Class)
24846	DELPHI HARRISON THERMAL SYS, C/O PAR INDUSTRIES INC., 6505 ROCKSIDE RD, SUITE 125, INDEPENDENCE, OH, 44131-2361	US Mail (1st Class)
24846	DELPHI PACKARD ELECTRIC, SYSTEMS GEN. MOTORS CORP., PO BOX 71405, CHICAGO, IL, 60694	US Mail (1st Class)
24846	DELPHI PACKARD ELECTRIC SYSTEMS, RAPID PROTOTYPING LAB, PO BOX 431 MAIL STATION 93P, WARREN, OH, 44486	US Mail (1st Class)
24846	DELTA CONTROLS INC., 15 STANLEY AVE., THOMASVILLE, NC, 27360	US Mail (1st Class)
24846	DELTA ELECTRIC SUPPLY CO., 489 E HURON RIVER DR., BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	DELTA INDUSTRIAL SERVICES, BATALLA DE IZUCAR, 118, COL. JOSE MARIA MORELOS, GUADALUPE, NL, 67150MEXICO	US Mail (1st Class)
24846	DELTA MOLD, INC., 9415 STOCKPORT PL, CHARLOTTE, NC, 28273	US Mail (1st Class)
24846	DELTA PRECISION INC., 664 AJAX DR., MADISON HEIGHTS, MI, 48071	US Mail (1st Class)
24846	DELTA TOOLING COMPANY, 1350 HARMON ROAD, AUBURN HILLS, MI, 48326	US Mail (1st Class)
24846	DEMAG PLASTICS GROUP, 88049 EXPEDITE WAY, CHICAGO, IL, 60695-8049	US Mail (1st Class)
24846	DEMAG PLASTICS GROUP, 88049 EXPEDITE WAY, CHICAGO, IL, 60695-8049	US Mail (1st Class)
24846	DEMAG PLASTICS GROUP CORP., C/O JPMC LOCKBOX #88049, 550 W VAN BUREN 14TH FLOOR, CHICAGO, IL, 60607	US Mail (1st Class)
24846	DEMO TECHNIC, SAS, 281 RUE ISAAC NEWTON, CHAMBLY, 60230FRANCE	US Mail (1st Class)
24846	DEMPSEY INDUSTRIES, INC., 802 NORTH FOURTH STREET, MIAMISBURG, OH, 45342	US Mail (1st Class)
24846	DENISE R DUNCAN, 524 SOUTH STREET, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	DENISE TOLEN, 290 MCCORMICK AVE., LAPEER, MI, 48446	US Mail (1st Class)
24846	DENNIS CARMEDY, 2335 ROLL DRIVE, SUITE 4-82, SAN DIEGO, CA, 92154	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	DENNIS CREEDON, 31572 INDUSTRIAL RD., SUITE #300, LIVONIA, MI, 48150	US Mail (1st Class)
24846	DENNIS DAUER, 119 EAST DAYTON STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	DENNIS JOHNSON, 119 EAST DAYTON STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	DENNIS SMITH, PO BOX 524, UNION CITY, PA, 16438	US Mail (1st Class)
24846	DEPRAG, INC., 640 HEMBRY ST, LEWISVILLE, TX, 75057-4777	US Mail (1st Class)
24846	DEPUTY SHERIFFS ASSOC. OF MI, 191 CADY CENTRE, NORTHVILLE, MI, 48167-1121	US Mail (1st Class)
24846	DERBY FABRICATING INC., 4500 PRODUCE RD, LOUISVILLE, KY, 40218-3058	US Mail (1st Class)
24846	DERBY FABRICATING LLC, 4500 PRODUCE RD, LOUISVILLE, KY, 40218	US Mail (1st Class)
24846	DERBY FABRICATING,LLC, 4500 PRODUCE RD., LOUISVILLE, KY, 40218	US Mail (1st Class)
24846	DERBY INDUSTRIES, 24350 STATE ROAD 23, SOUTH BEND, IN, 46614-9593	US Mail (1st Class)
24846	DERENDA CALLAWAY, 100 SEVENTH AVENUE, SUITE 100, CHARDON, OH, 44024-1077	US Mail (1st Class)
24846	DESAHZO CRANE, 190 AIRPARK INDUSTRIAL RD, ALABASTER, AL, 35007-9597	US Mail (1st Class)
24846	DESCHNER CORP., 3211 WEST HARVARD ST, SANTA ANA, CA, 92704	US Mail (1st Class)
24846	DESIGNER SIGNS & TEE'S, 9955 SHEPHERD RD. LOT B#12, LOCKBOURNE, OH, 43137	US Mail (1st Class)
24846	DESTINY PLASTICS, 6975 E SANILAC RD, PORT SANILAC, MI, 48469	US Mail (1st Class)
24846	DETAIL BORING INC., 190 COY CT, OXFORD, MI, 48371-1644	US Mail (1st Class)
24846	DETROIT ECONOMIC CLUB, 211 WEST FORT ST., SUITE 505, DETROIT, MI, 48226	US Mail (1st Class)
24846	DETROIT EMPLOYMENT GUIDE, 5301 ROBIN HOOD RD, NORFOLK, VA, 23513-2419	US Mail (1st Class)
24846	DETROIT NEWSPAPERS, DEPT. 77708, DETROIT, MI, 48277-0708	US Mail (1st Class)
24846	DETROIT RED WINGS, ATTN: BOX OFFICE, 600 CIVIC CENTER DR, DETROIT, MI, 48226	US Mail (1st Class)
24846	DETROIT RIGGERS, INC., 3519 LAPEER ROAD, METAMORA, MI, 48455	US Mail (1st Class)
24846	DETROIT TESTING LAB., INC, 7111 EAST ELEVEN MILE ROAD, WARREN, MI, 48092	US Mail (1st Class)
24846	DETROIT TESTING LABORATORY,INC, DEPARTMENT #264201, PO BOX 67000, DETROIT, MI, 48267-2642	US Mail (1st Class)
24846	DETROIT WIRE ROPE, 528 ROBBINS DR, TROY, MI, 48083-4514	US Mail (1st Class)
24846	DEUBLIN COMPANY, DEPT. 77-5136, CHICAGO, IL, 60678-5136	US Mail (1st Class)
24846	DHL DANZAS AIR & OCEAN, 1200 S PINE ISLAND RD STE 140-145, PLANTATION, FL, 33324-4413	US Mail (1st Class)
24846	DHL WORLDWIDE EXPRESS, 515 W GREENS RD, HOUSTON, TX, 77067	US Mail (1st Class)
24846	DIAGRAPH, 5307 MEDOWLAND PARKWAY, MARION, IL, 62959	US Mail (1st Class)
24846	DIAGRAPH OF NORTHERN, 15400 INDUSTRIAL PARKWAY, CLEVELAND, OH, 44135	US Mail (1st Class)
24846	DIAMLER CHRYSLER, WINDSOR ASSY 09103 RS DK, 2199 CHRYSLER CENTRE, WINDSOR, ON, N9A 4H6CANADA	US Mail (1st Class)
24846	DIAMOND AMERICA CORP., 399 COMMERCE STREET, TALLMADGE, OH, 44278	US Mail (1st Class)
24846	DIANA ANDERSON, ADDRESS UNAVAILABLE, UNIONTOWN, OH, 44685	US Mail (1st Class)
24846	DIANA CANTRELL, 608 IVY COURT, EATON, OH, 45320	US Mail (1st Class)
24846	DIANE ELLIS, 3187 HIRED HAND ROAD, LAPEER, MI, 48446	US Mail (1st Class)
24846	DIANES TRAILER RENTAL, 3187 HIRED HAND RD., LAPEER, MI, 48446	US Mail (1st Class)
24846	DICAR, 3888 AINA LANE, SOUTH LYON, MI, 48178	US Mail (1st Class)
24846	DICK'S SPORTING GOODS, ATTN: ACCT. RECEIVABLE, 200 INDUSTRY DRIVE, PITTSBURGH, PA, 15275	US Mail (1st Class)
24846	DICKMAN INDUSTRIAL, 1991 ST MARYS RD, SIDNEY, OH, 45365-9395	US Mail (1st Class)
24846	DICKSON/UNIGAGE INC., 930 SOUTH WESTWOOD AVE., ADDISON, IL, 60101-4917	US Mail (1st Class)
24846	DIE STAMPCO INC., 1301 NORTH LINCOLN STREET, BAY CITY, MI, 48708	US Mail (1st Class)
24846	DIECO PRODUCTO RING, 2980 TURNER RD, JAMESTOWN, NY, 14701-9024	US Mail (1st Class)
24846	DIFFLEY WRIGHT CORP., 1891 PORTER LAKE DR, SUITE 106, SARASOTA, FL, 34240	US Mail (1st Class)
24846	DIGITAL SOURCE 360 LLC, ATTN: LOGISTIC SPECIALIST, 10405-A GRANITE STREET, CHARLOTTE, NC, 28273	US Mail (1st Class)
24846	DILLON/QUALITY PLUS, INC., 3201 CORTE MALPASO STE 311, CAMARILLO, CA, 93012-8075	US Mail (1st Class)
24846	DILMAR OIL CO., INC., 1951 W DARLINGTON ST, FLORENCE, SC, 29565	US Mail (1st Class)
24846	DIMENSION PRODUCTS CORP., 13746 172ND AVENUE, GRAND HAVEN, MI, 49417	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	DION HAMPTON, 1401 INDUSTRIAL PARK, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	DIRECT SAFETY CO., 3202 LATHAM DR, MADISON, WI, 53713	US Mail (1st Class)
24846	DIRECT SEARCH SYSTEMS, 205 NORTH STREET, SUITE 201, CHARDON, OH, 44024	US Mail (1st Class)
24846	DIRECT SOURCE, 1126 ETHEL ROAD, MIAMISBURG, OH, 45342	US Mail (1st Class)
24846	DISTRIBUTOR SUPPLY CO., INC, PO BOX 1934, ANNISTON, AL, 36202	US Mail (1st Class)
24846	DISTRICT COURT 52-3, ROCHESTER HILL, 700 BARCLAY CIRCLE, ROCHESTER HILLS, MI, 48307	US Mail (1st Class)
24846	DIVERSIFIED AIR SYS.,INC., 4760 VAN EPPS ROAD, CLEVELAND, OH, 44131	US Mail (1st Class)
24846	DIVERSIFIED TRANSPORT, 15051 WHITE ROAD, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	DIVERSITY PRODUCTS, 32031 HOWARD STREET, MADISON HEIGHTS, MI, 48071	US Mail (1st Class)
24846	DIVISION OF TAXATION, PO BOX 5433, FAIRLAWN, OH, 44334	US Mail (1st Class)
24846	DIXIE INDUSTRIAL, PO BOX 110038, BIRMINGHAM, AL, 35211-0038	US Mail (1st Class)
24846	DLF CONTAINER SYSTEMS, 1450 SOUTH 35TH STREET, GALESBURG, MI, 49053	US Mail (1st Class)
24846	DLH INDUSTRIES, INC, 2422 LEO AVE., S.W., CANTON, OH, 44706	US Mail (1st Class)
24846	DOALL DAYTON, 4436 PAYSPHERE CIRCLE, CHICAGO, IL, 60674	US Mail (1st Class)
24846	DOBBY JOHNSON, 9455 EAST ROAD, BURT, MI, 48417	US Mail (1st Class)
24846	DOCUMENT CONCEPTS, INC., 1835 DUEBER AVE SW, CANTON, OH, 44706	US Mail (1st Class)
24846	DO-IT-FAST DISTRIBUTORS, 567 NOTTINGHILL RD., LONDON, ON, N6K 1R3CANADA	US Mail (1st Class)
24846	DOMINION EAST OHIO, ACCT: 6-4205-0007-8440, ATTN: CREDIT DEPT, 701 E CARY, 18TH FLOOR, RICHMOND, VA, 23219	US Mail (1st Class)
24846	DOMINO'S PIZZA, INC., 30 FRANK LLOYD WRIGHT DR, ANN ARBOR, MI, 48105-9757	US Mail (1st Class)
24846	DON MAILNOWSKI, 3305 REGIS DR., ERIE, PA, 16510	US Mail (1st Class)
24846	DON MALINOWSKI, 3305 REGIS DR., ERIE, PA, 16510	US Mail (1st Class)
24846	DONALD C. MANN, M.D., 1611 S GREEN RD. #203, S. EUCLID, OH, 44121	US Mail (1st Class)
24846	DONALD J. STOLARSKI, 25 EAST 37TH STREET, ERIE, PA, 16504	US Mail (1st Class)
24846	DONALD M. ALEMAN, 1230 STACY DRIVE, CANTON, MI, 48188	US Mail (1st Class)
24846	DONALD PATTERSON, 8200 OLD MARION RD., LOT 16, TUSCALOOSA, AL, 35405	US Mail (1st Class)
24846	DONALD WINANS, 3152 EASTGATE, BURTON, MI, 48519	US Mail (1st Class)
24846 24846	DONELCO, INC., 3795 STATE ROUTE 39, SHELBY, OH, 44875 DONNA BRYHN, 900 S. WILEY ST., PO BOX 320, CRESTLINE, OH, 44827	US Mail (1st Class) US Mail (1st Class)
24846	DONNELLY BROTHERS INC., 100 GARLINGTON, LAURENS, SC, 29360	US Mail (1st Class)
24846	DORAL GOLF RESORT AND SPA, 4400 N.W. 87TH AVENUE, MIAMI, FL, 33178	US Mail (1st Class)
24846	DORIS T. TURNER, CIRCUIT CLERK 3RD FLOOR, TUSC. CO. COURTHOUSE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	DORMAN & ASSOCIATES, 600 NORTH FORSYTHE ST., FRANKLIN, IN, 46131	US Mail (1st Class)
24846	DOROTHY HICKS, 8707 SAMUEL BARTON DR., BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	DOROTHY NOLAN, 3389 ROBINET RD., NORTH BRANCH, MI, 48461-8924	US Mail (1st Class)
24846	DOUBLE TECHNOLOGY INC., 6211 YOUNGSTOWN-KINGSVILLE RD., CORTLAND, OH, 44410	US Mail (1st Class)
24846	DOUG HORNAK, 100 SEVENTH AVE., SUITE 100, CHARDON, OH, 44024	US Mail (1st Class)
24846	DOUG STOKES, 290 MCCORMICK, LAPEER, MI, 48446	US Mail (1st Class)
24846	DOVE DIE & STAMPING, 15665 BROOKPARK ROAD, CLEVELAND, OH, 44142	US Mail (1st Class)
24846	DOVE LEASING, 16851 ELEVEN MILE ROAD, ROSEVILLE, MI, 48066	US Mail (1st Class)
24846	DOVEBID VALUATION SERVICES, 10877 WILSHIRE BLVD STE 600, LOS ANGELES, CA, 90024-4348	US Mail (1st Class)
24846	DOVEBID VALUATION SERVICES, ATTN: DVS A/R, 1241 E. HILLSDALE BLVD., FOSTER CITY, CA, 94404	US Mail (1st Class)
24846	DOVER HYDRAULICS, 2996 PROGRESS STREET, PO BOX 2239, DOVER, OH, 44622	US Mail (1st Class)
24846	DOW CHEMICAL COMPANY, 7719 COLLECTION CENTER DR., CHICAGO, IL, 60693	US Mail (1st Class)
24846	DOW CORNING, 2200 W SALZBURG RD, MIDLAND, MI, 48686	US Mail (1st Class)
24846	DOW CORNING CORPORATION, 2200 SALZBURG ST, MIDLAND, MI, 48640-8531	US Mail (1st Class)
24846	DOW SCREW PRODUCTS INC., 3810 PAULE AVE, SAINT LOUIS, MO, 63125-1718	US Mail (1st Class)
24846	DOW-CORNING STI, 111 S PROGRESS DR E, KENDALLVILLE, IN, 46755-3268	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	DOX CALIBRATION, 13938 FERMOY AVE, ORLAND PARK, IL, 60467-7410	US Mail (1st Class)
24846	DOZIER EQUIPMENT COMPANY, BOX 88154, MILWAUKEE, WI, 53288-0154	US Mail (1st Class)
24846	DP BROWN OF DETROIT INC., 1646 CHAMPAGNE DR N, SAGINAW, MI, 48604	US Mail (1st Class)
24846	DPS ENTERPRISES, 7191 GREEN VALLEY DRIVE, MENTOR, OH, 44060	US Mail (1st Class)
24846	DR DAVID STADNICK, 536 S. TRIMBLE ROAD, MANSFIELD, OH, 44905	US Mail (1st Class)
24846	DR. CLARENCE R. DEITSCH, 1201 SOUTH C.R. 725 W., YORKTOWN, IN, 47396	US Mail (1st Class)
24846	DR. ZENKO J HRYNKIW MDPC, 720 MONTCLAIR RD, SUITE 201, BIRMINGHAM, AL, 35213	US Mail (1st Class)
24846	DRAKE HEATING & AIR, 73 N MAIN STREET, WEST ALEXANDRIA, OH, 45381-3118	US Mail (1st Class)
24846	DRAKE-STATE AIR INC., 71 N MAIN STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	DRILLING SYSTEMS LLC, 2845 S. REDBUD TRAIL, NILES, MI, 49120	US Mail (1st Class)
24846	DRIVE SYSTEMS TECHNOLOGY, 9158 ANTIOCH RD., SALEM, WI, 53168	US Mail (1st Class)
24846	DRS INDUSTRIES, 1067 HAMILTON DRIVE, HOLLAND, OH, 43528	US Mail (1st Class)
24846	DRS. FOSTER & SMITH, PET WAREHOUSE, 2253 AIR PARK ROAD, PO BOX 100, RHINELANDER, WI, 54501-0100	US Mail (1st Class)
24846	DRUMM & O'HARAH TOOL & DESIGN, 8343 EDINBORO RD, ERIE, PA, 16509	US Mail (1st Class)
24846	DRUMMOND RAINBOW, CARPET CLEAN, 15 BELLVIEW DR, TUSCALOOSA, AL, 35405-4155	US Mail (1st Class)
24846	DSM ENGINEERING PLASTICS, 233 ARVIN AVENUE, STONEY CREEK, ON, L8E 2L9CANADA	US Mail (1st Class)
24846	DSM THERMOPLASTIC, 4101 PAYSPHERE CIRCLE, CHICAGO, IL, 60674-0041	US Mail (1st Class)
24846	DSM THERMOPLASTIC ELASTOMERS, 135 S LASALLE, DEPT. 4101, CHICAGO, IL, 60674-4101	US Mail (1st Class)
24846	DU PONT MEXICO SA DE CV, HOMERO 206 15, CHAPULTEPEC MORALES, MEXICO DF, 11570MEXICO	US Mail (1st Class)
24846	DUKANE CORP., BOX 95080, CHICAGO, IL, 60694-5080	US Mail (1st Class)
24846	DUKANE ULTRASONICS, 2900 DUKANE DR., ST. CHARLES, IL 60174, SAINT CHARLES, IL, 60174	US Mail (1st Class)
24846	DUNAVAN'S TRIM & ASSEM., 9096 CENTRALIA, REDFORD, MI, 48239	US Mail (1st Class)
24846	DUNHILL PROFESSIONALS, 2815 BROADLAWN DR, ERIE, PA, 16506-1809	US Mail (1st Class)
24846	DUNN BUILDINGS, LLC, 2700 HWY 280 S STE 110, BIRMINGHAM, AL, 35223	US Mail (1st Class)
24846	DUPONT COMPANY, CORPORATE HEADQUARTERS 1007 MARKET STREET, WILMINGTON, DE, 19898-0001	US Mail (1st Class)
24846	DUPONT DENEMOURS & CO., 222 MERCHANDISE MART PLZ STE 10-111, CHICAGO, IL, 60654-1183	US Mail (1st Class)
24846	DUPONT DOW ELASTOMERS LLC, 21088 NETWORK PLACE, CHICAGO, IL, 60673-1210	US Mail (1st Class)
24846	DUPONT/ENGINEERING POLYMERS, 1007 MARKET STREET, WILMINGTON, DE, 19898-0001	US Mail (1st Class)
24846	DURABOX-RAM DE MEXICO, BANK OF THE WEST, ATTN: JHERI BUSTILLOS, 500 NORTH MESA STREET, EL PASO, TX, 79901	US Mail (1st Class)
24846	DURAND RECYCLING LLC, 9919 E MILLER, DURAND, MI, 48429	US Mail (1st Class)
24846	DURR MARKETING ASSOC.,INC, 1300 LOWER RODI ROAD, TURTLE CREEK, PA, 15145	US Mail (1st Class)
24846	DWIGHT FRY, 6915 HOLLANSBURG-SAMPSON RD., ARCANUM, OH, 45304	US Mail (1st Class)
24846	DWYER INSTRUMENTS INC., 102 INDIANA HIGHWAY 212, MICHIGAN CITY, IN, 46360-1956	US Mail (1st Class)
24846	DYER COMPANY, 1500 MCGOVERNVILLE RD, PO BOX 4966, LANCASTER, PA, 17604	US Mail (1st Class)
24846	DYNALAB CORPORATION, 350 COMMERCE DR, ROCHESTER, NY, 14623-3508	US Mail (1st Class)
24846	DYNAMAX, INC,, 637 E BIG BEAVER RD., SUITE 209, TROY, MI, 48083-1425	US Mail (1st Class)
24846	DYNAMIC CONVEYOR CORP., 5980 GRAND HAVEN RD., MUSKEGON, MI, 49441	US Mail (1st Class)
24846	DYNAMIC METAL TREATING INC., 7784 RHONDA DRIVE, CANTON TOWNSHIP, MI, 48187	US Mail (1st Class)
24846	DYNAMIC TECHNOLOGY INC., 1200 N OLD 23, HARTLAND, MI, 48353	US Mail (1st Class)
24846	DYNA-MIX, 705 S LYONS AVE, TYLER, TX, 75702-6857	US Mail (1st Class)
24846	DYNEON LLC, NW - 8984, PO BOX 1450, MINNEAPOLIS, MN, 55485-8984	US Mail (1st Class)
24846	DYNISCO INSTRUMENTS, 38 FORGE PKWY, FRANKLIN, MA, 02038-3134	US Mail (1st Class)
24846	DYNISCO POLYMER TEST, 38 FORGE PKWY, FRANKLIN, MA, 02038-3134	US Mail (1st Class)
24846	E & E MANUFACTURING, 231290 MOMENTUM PLACE, CHICAGO, IL, 60689-5311	US Mail (1st Class)
24846	E & L SERVICES-TOWING & RECOV, 920 DEMILLE RD., LAPEER, MI, 48446	US Mail (1st Class)
24846	E I DUPONT DE NEMOURS, 1007 MARKET STREET, WILMINGTON, DE, 19898-0001	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	E&M EQUIPMENT CO., 4285 EAST MAIN STREET, COLUMBUS, OH, 43213	US Mail (1st Class)
24846	E.B. BROWN OPTICIANS, 1549 E. 30TH STREET, CLEVELAND, OH, 44114	US Mail (1st Class)
24846	E.C. MORRIS CORPORATION, 201 QUADRAL DR, WADSWORTH, OH, 44281-9571	US Mail (1st Class)
24846	E.I. DUPONT, 950 STEVENSON HWY, TROY, MI, 48083	US Mail (1st Class)
24846	E.I.S. FIBER COATING, INC, 616 E. MAIN TS., LOGANSPORT, IN, 46947	US Mail (1st Class)
24846	E.O.A. SYSTEMS, INC., 3305 WILEY POST RD, CARROLLTON, TX, 75006-5113	US Mail (1st Class)
24846	E.P.B.S., 3303 S MERIDIAN AVE, OKLAHOMA CITY, OK, 73119-1026	US Mail (1st Class)
24846	E.T. PRINTING, 7524 TYLER BLVD., MENTOR, OH, 44060	US Mail (1st Class)
24846	EAGLE FENCE, 3836 TWIN TOWNSHIP RD, LEWISBURG, OH, 45338	US Mail (1st Class)
24846	EAGLE MARK IV EQUIPMENT, INC., 330 ASHLAND RD, MANSFIELD, OH, 44905	US Mail (1st Class)
24846	EAGLE REGISTRATIONS INC., 2242 W SCHANTZ AVE., DAYTON, OH, 45409	US Mail (1st Class)
24846	EAP ASSOCIATES, INC., 1903 W. 8TH STREET #142, ERIE, PA, 16505	US Mail (1st Class)
24846	EARL HARMON, 5195 OLD STATE ROAD, NORTH BRABCH, MI, 48461	US Mail (1st Class)
24846	EARLE M.JORGENSEN COMPANY, 10650 ALAMEDA ST, LYNWOOD, CA, 90262	US Mail (1st Class)
24846	EARLY, KEVIN, 777 PORTLAND WAY N, GALION, OH, 44833	US Mail (1st Class)
24846	EAST OF CHICAGO PIZZA, 147 SOUTH SELTZER STREET, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	EASTDALE OVERHEAD DOOR, 529 FRONT AVE, ASHTABULA, OH, 44004-3553	US Mail (1st Class)
24846	EASTERN MICHIGAN UNIVERSITY, CENTER FOR CORPORATE TRAINING, 2000 HURON RIVER DR. STE 101, YPSILANTI, MI, 48197-1699	US Mail (1st Class)
24846	EASTERN OIL CO., 590 S PADDOCK, PONTIAC, MI, 48341	US Mail (1st Class)
24846	EASTERN SINTERED ALLOYS, 126 ACCESS RD., PO BOX 708, ST.MARYS, PA, 15857	US Mail (1st Class)
24846	EATON-PREBLE CHAMBER, 110 W MAIN ST STE 3, EATON, OH, 45320-1746	US Mail (1st Class)
24846	ECONO LODGE & SUITES, 1930 MCFARLAND BLVD., NORTHPORT, AL, 35476	US Mail (1st Class)
24846	ECS SERVICES, 26993 REAUME ROAD, WOODHAVEN, MI, 48183	US Mail (1st Class)
24846	ECS SERVICES, 26993 REAUME RD, WOODHAVEN, MI, 48183	US Mail (1st Class)
24846	ED FIFIELD, 7851 RICH RD, FOSTORIA, MI, 48435	US Mail (1st Class)
24846	EDGE DISTRIBUTION, 766 W. ALGONQUIN ROAD, ARLINGTON HEIGHTS, IL, 60005	US Mail (1st Class)
24846	EDICT SYSTEMS, INC., 2680 INDIAN RIPPLE RD., DAYTON, OH, 45440-3605	US Mail (1st Class)
24846	EDM ZAP PARTS INC., 1108 FRONT STREET, LISLE, IL, 60532-2258	US Mail (1st Class)
24846	EDS, LABEL ERROR PROOFING, ATTN: CAROL JANSEN, PO BOX 7923, FLINT, MI, 48507-7923	US Mail (1st Class)
24846	EDS & EFT, EDS CORP., PO BOX 14947, ST. LOUIS, MO, 63150-4947	US Mail (1st Class)
24846	EDWARD CLEMENTS, 119 EAST DAYTON STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	EDWARD HOWARD & CO., 1360 EAST 9TH ST. 7TH FLOOR, CLEVELAND, OH, 44114-1716	US Mail (1st Class)
24846	EDWARD JONES, 8707 SAMUEL BARTON DR., BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	EDWARD R. CROWE, 900 S. WILEY ST., PO BOX 320, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	EFFECTIVE TRAINING INC., 2116 S WAYNE RD., WESTLAND, MI, 48186	US Mail (1st Class)
24846	EFFICIENT QUALITY SERVICES INC, 4456 WEBB RD STE 300, SIMPSONVILLE, KY, 40067-6488	US Mail (1st Class)
24846	EFO, INC., 977 WATERMAN AVE, EAST PROVIDENCE, RI, 02914-1342	US Mail (1st Class)
24846	EIHRA, PO BOX 2184, RICHMOND, IN, 47375	US Mail (1st Class)
24846	EIS, INC. FILE 98059, SUPPLIERS, INC., PO BOX 98059, CHICAGO, IL, 60693-8059	US Mail (1st Class)
24846	EISC, INC., 5555 AIRPORT HIGHWAY, SUITE 100, TOLEDO, OH, 43615-7320	US Mail (1st Class)
24846	EL PASO MOLD SUPPLY, 11394 JAMES WATT DR STE 416, EL PASO, TX, 79936-6441	US Mail (1st Class)
24846	ELBER SUPPLY, 9361 HAMILTON DR, MENTOR, OH, 44060-4559	US Mail (1st Class)
24846	ELDRIDGE & COMPANY, 4148 PARK AVE W, MANSFIELD, OH, 44903-8589	US Mail (1st Class)
24846	ELECTRIC MACHINERY & REPAIR, 125 E. BAKER STREET, PO BOX 428, GREEN SPRINGS, OH, 44836	US Mail (1st Class)
24846	ELECTRIC MOTOR SALES & SERVICE, 232 ALABAMA ST, COLUMBUS, MS, 39702-3204	US Mail (1st Class)
24846	ELECTRODES INC., 252 DEPOT RD, MILFORD, CT, 06460-3807	US Mail (1st Class)
24846	ELECTRO-DYNAMICS, INC., 318 THIRD STREET, PITTSBURG, PA, 15215	US Mail (1st Class)
24846	ELECTROMATIC EQUIPMENT CO., 600 OAKLAND AVE., CEDARHURST, NY, 11516	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	ELECTROMATIC PRODUCTS, 23409 INDUSTRIAL PARK, FARMINGTON HILLS, MI, 48336-2849	US Mail (1st Class)
24846	ELECTRO-MATIC PRODUCTS, INC., DEPARTMENT 117401, PO BOX 67000, DETROIT, MI, 48267-1174	US Mail (1st Class)
24846	ELECTRO-MATIC PRODUCTS, INC., 23409 INDUSTRIAL PARK COURT, FARMINGTON HILLS, MI, 48335	US Mail (1st Class)
24846	ELECTRONIC CENTER, INC., 7480 MARKET AVE N, CANTON, OH, 44721-1979	US Mail (1st Class)
24846	ELECTRO-TECHNICAL, 4642 NORTH RAVENSWOOD, CHICAGO, IL, 60640	US Mail (1st Class)
24846	ELIOKEM/MATTESON-RIDOLFI, 14450 KING ROAD, PO BOX 2129, RIVERVIEW, MI, 48192	US Mail (1st Class)
24846	ELITE ENTERPRISES, INC., 2701 S. COLISEUM BLVD., SUITE 1158, FORT WAYNE, IN, 46803	US Mail (1st Class)
24846	ELIZABETH CHAMBERS, 2048 MIDDLE ROAD, ASHTABULA, OH, 44004	US Mail (1st Class)
24846	ELIZABETH SAUERS, 8707 SAMUEL BARTON DR., BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	ELLIOTT`S, 12021 CLARK RD, CHARDON, OH, 44024	US Mail (1st Class)
24846	ELLZEY & BROOKS, LLC., PO BOX 11612, COLUMBIA, SC, 29211	US Mail (1st Class)
24846	ELWOOD CORP-MARK 2, 195 WEST RYAN ROAD, OAK CREEK, WI, 53154	US Mail (1st Class)
24846	EMERGENCY FILM GROUP, 140 COOKE ST, EDGARTOWN, MA, 02539-5915	US Mail (1st Class)
24846	EMERGENCY PROFESSIONAL, SERVICES INC., 2620 RIDGEWOOD RD., AKRON, OH, 44313	US Mail (1st Class)
24846	EMERINE'S TOWING INC., 3779 MAHONING AVE NW, WARREN, OH, 44483	US Mail (1st Class)
24846	EMERY CUSTOMS BROKERS, PO BOX 1067, SCRANTON, PA, 18577-0067	US Mail (1st Class)
24846	EMERY EXPEDITE, PO BOX 1994, SCRANTON, PA, 18501	US Mail (1st Class)
24846	EMHART, PO BOX 73151, CHICAGO, IL, 60673	US Mail (1st Class)
24846	EMHART FASTENING, 12332 COLLECTIONS CENTER, CHICAGO, IL, 60693	US Mail (1st Class)
24846	EMI CRANES & RIGGING, 6640 OLD MONTGOMERY HWY, TUSCALOOSA, AL, 35405	US Mail (1st Class)
24846	E-MOTION LLC, 4790 ERHART ROAD, MEDINA, OH, 44256	US Mail (1st Class)
24846	EMPAQUES,FLEJES Y EQUIPOS, CALLE 39 NO 1003-B COL OBRERA, CHIHUAHUA, MEXICO	US Mail (1st Class)
24846	EMPIRE COFFEE CO., PO BOX 1544, PAINESVILLE, OH, 44077	US Mail (1st Class)
24846	EMPLOYEE MOTIVATION &INCENTIVE, 525 CAYUGA STREET, STORM LAKE, IA, 50588	US Mail (1st Class)
24846	EMPLOYEE PARTICIPATION COUNCIL, OF NCO, LONGVIEW CENTER, MANSFIELD, OH, 44906	US Mail (1st Class)
24846	EMPLOYERS RESOURCE COUNCIL, 1382 WEST 9TH ST. SUITE 400F, CLEVELAND, OH, 44113-1231	US Mail (1st Class)
24846	ENERGY SOLUTIONS CO, 6830 HIGHLAND RD, WHITE LAKE, MI, 48383-2845	US Mail (1st Class)
24846	ENGEL CANADA INC., 5573 N SHORE DR, DULUTH, MN, 55804	US Mail (1st Class)
24846	ENGEL CANADA INC., 545 ELMIRA ROAD, GUELPH, ON, N1K 1C2CANADA	US Mail (1st Class)
24846	ENGEL MACHINERY INC., 3740 BOARD RD., RD. #5, YORK, PA, 17402	US Mail (1st Class)
24846	ENGINEERED EQUIP. SALES, 72 HARMON AVE, PAINESVILLE, OH, 44077	US Mail (1st Class)
24846	ENGINEERED MACHINE SYSTEMS, 210 HAYES DRIVE UNIT F, CLEVELAND, OH, 44131-1036	US Mail (1st Class)
24846	ENGIS EQUIPMENT CO., 105 W HINTZ ROAD, WHEELING, IL, 60090	US Mail (1st Class)
24846	ENGLE, 3740 VORD RD., YORK, PA, 17402	US Mail (1st Class)
24846	ENRON, C/O NCO FINANCIAL SYSTEMS, 507 PRUDENTIAL ROAD, HORSHAM, PA, 19044	US Mail (1st Class)
24846	ENRON ENERGY SERVICE, INC., FOUR HOUSTON CENTER, 1221 LAMAR ST STE 1600, HOUSTON, TX, 77010-3039	US Mail (1st Class)
24846	ENSINE CORPORATION, 310-B WEST HIGHTOWER DRIVE, DAWSONVILLE, GA, 30534	US Mail (1st Class)
24846	ENTEC POLYMERS LLC, 1900 SUMMIT TOWER BLVD STE 900, ORLANDO, FL, 32810-5925	US Mail (1st Class)
24846	ENTELA INC., DEPARTMENT # 251601, PO BOX 67000, DETROIT, MI, 48267-2516	US Mail (1st Class)
24846	ENTERPRISE RENT-A-CAR, 111 W GENESEE STREET, LAPEER, MI, 48446	US Mail (1st Class)
24846	ENTROPEK, 1271 LOGAR ST., SARNIA, ON, N75 5N5CANADA	US Mail (1st Class)
24846	ENTROPEX, 1271 LOUGAR AVE, SARNIA, ON, N7S 5N5CANADA	US Mail (1st Class)
24846	ENVIRO-CONTROL INC., 6959 PROMWAY NW, NORTH CANTON, OH, 44720	US Mail (1st Class)
24846	ENVIRONMENTAL AFFAIRS MGMT., 1313 DUBLIN RD, COLUMBUS, OH, 43215-7000	US Mail (1st Class)
24846	ENVIRONMENTAL COORDINATION, SERVICES & RECYCLING, 3237 US HIGHWAY 19, COCHRANTON, PA, 16314	US Mail (1st Class)
24846	ENVIRONMENTAL RECYCLING, 527 E WOODLAND CIRCLE, PO BOX 167, BOWLING GREEN, OH, 43402	US Mail (1st Class)
24846	ENVIRONMENTAL TECHNOLOGY, 6748 WARNER ROAD, MADISON, OH, 44057	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	ENVIROSAFE TECHNOLOGIES, 11201 ST. JOHNS INDUSTRIAL PARKWAY, JACKSONVILLE, FL, 32246	US Mail (1st Class)
24846	EPCO MACHINERY LLC, DEPT. 161601, PO BOX 67000, DETROIT, MI, 48267-1616	US Mail (1st Class)
24846	EPIC EXPRESS, 19-5425 DIXIE ROAD, MISSISSAUGA, ON, L4W 2A2CANADA	US Mail (1st Class)
24846	EQUITY TECHNOLOGIES, 2301 PERIMETER ROAD, MOBILE, AL, 36615-1135	US Mail (1st Class)
24846	ER INDUSTRIAL SALES INC., PO BOX 77000, DETROIT, MI, 48277-0403	US Mail (1st Class)
24846	ERA INDUSTRIAL SALES, 80 MODULAR AVE., COMMACK, NY, 11725	US Mail (1st Class)
24846	ERIC L. JOHNSON, 3006 W. RUNYAN, LAKE ODESSA, MI, 48849	US Mail (1st Class)
24846	ERIC OLINGER, 4121 SHOCK DRIVE, NEW LEBANON, OH, 45345	US Mail (1st Class)
24846	ERIC TOUCHETTE, 10113 E MOUNT MORRIS RD, DAVISON, MI, 48423-9333	US Mail (1st Class)
24846	ERIC WHITE, 1401 INDUSTRIAL PARK DR., TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	ERIE BEARINGS COMPANY, 1432 E 12TH ST, ERIE, PA, 16503-1719	US Mail (1st Class)
24846	ERIE HARD CHROME INC., 1570 EAST 12TH ST., ERIE, PA, 16511	US Mail (1st Class)
24846	ERIE MILL & PRESS, 953 E 12TH ST, ERIE, PA, 16503-1441	US Mail (1st Class)
24846	ERIE RIGHT TO KNOW, 360 YOUNG ROAD, ERIE, PA, 16509	US Mail (1st Class)
24846	ERIEZ MANUFACTURING CO., POBOX 641890, PITTSBURGH, PA, 15264-1890	US Mail (1st Class)
24846	ERNST BOWLING LANES, 10651 GRANT STREET, CHARDON, OH, 44024	US Mail (1st Class)
24846	ERNST FLUID POWER, 3815 WYSE RD, DAYTON, OH, 45414-2542	US Mail (1st Class)
24846	ERVINS GROUP, DEPARTMENT 227201, PO BOX 67000, DETROIT, MI, 48267-2272	US Mail (1st Class)
24846	EST TESTING SOLUTIONS, INC., 345 EAST 48TH STREET, HOLLAND, MI, 49423	US Mail (1st Class)
24846	ESTABROOK CORP., 700 W. BAGLEY RD., PO BOX 804, BEREA, OH, 44017	US Mail (1st Class)
24846	ESTES EXPRESS LINES, 3318 ROBT T LONGWAY, FLINT, MI, 48506	US Mail (1st Class)
24846	EUREKA EDUCATORS, 630 N WAYNE RD, WESTLAND, MI, 48185-3631	US Mail (1st Class)
24846	EUROTHERM GROUP COMPANIES, 741-F MILLER DR SE, LEESBURG, VA, 20175-8920	US Mail (1st Class)
24846	EVANS MIDDLEFIELD MED CENTER, PO BOX 368, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	EVER KOLD REFRIGERATION, 12017 WOODBINE, REDFORD, MI, 48239	US Mail (1st Class)
24846	EVERCLEAR, 3700 OAKWOOD AVE, AUSTINTOWN, OH, 44515	US Mail (1st Class)
24846	EWIKON HOTRUNNER SYSTEM, 1051 E MAIN STREET, UNIT 101, EAST DUNDEE, IL, 60118	US Mail (1st Class)
24846	EXAIR CORPORATION, LOCATION 00766, CINCINNATI, OH, 45264-0766	US Mail (1st Class)
24846	EXCEL GLOBAL LOGISTICS INC., 980 LONE OAK ROAD, SUITE NO. 108, EACAN, MN, 55121	US Mail (1st Class)
24846	EXCEL POLYMERS, LLC, 6521 DAVIS INDUSTRIAL PKWY, SOLON, OH, 44139-3549	US Mail (1st Class)
24846	EXCEL QUALITY SERVICES, INC., ACCOUNTS RECEIVABLE, 25925 GLENDALE, REDFORD, MI, 48239	US Mail (1st Class)
24846	EXCELL PARTNERSHIP, 464 HERITAGE RD STE 1, SOUTHBURY, CT, 06488-3863	US Mail (1st Class)
24846	EXECUTIVE CONFERENCE MGMT., PO BOX 700272, 11696 AMHERST CT., PLYMOUTH, MI, 48170	US Mail (1st Class)
24846	EXECUTIVE FOCUS INTERNATIONAL, 5099 HIGHWAY A1A, VERO BEACH, FL, 32963	US Mail (1st Class)
24846	EXECUTIVE HOUSE, 40 CENTRE SQUARE, SUITE 301, 40 CENTRE STREET, CHATHAM, ON, N7M 5W3CANADA	US Mail (1st Class)
24846	EXEL GLOBAL LOGISTICS INC., 980 LONE OAK ROAD, SUITE NO. 108, EACAN, MN, 55121	US Mail (1st Class)
24846	EXONIC SYSTEMS, BW ROGERS CO, 380 WATER ST, AKRON, OH, 44308-1045	US Mail (1st Class)
24846	EXOTIC, DEPT.# 233601, PO BOX 67000, DETROIT, MI, 48267-2336	US Mail (1st Class)
24846	EXPRESS PERSONNEL SERVICES, INTERNATIONAL HEADQUARTERS, 8516 NW EXPRESSWAY ST, OKLAHOMA CITY, OK, 73162-6010	US Mail (1st Class)
24846	EXPRESS PRINTING, 815 HARDING WAY WEST, GALION, OH, 44833	US Mail (1st Class)
24846	EXTENDED STAY AMERICA, 7851 LOIS CIRCLE, DAYTON, OH, 45459	US Mail (1st Class)
24846	EXTRAHELP STAFFING SERVICE, 120 WEST SECOND STREET, DAYTON, OH, 45402	US Mail (1st Class)
24846	EXXON CHEMICAL AMERICAS, 13501 KATY FWY, HOUSTON, TX, 77079-1305	US Mail (1st Class)
24846	EXXONMOBIL CHEMICAL, 37567 INTERCHANGE DR., FARMINGTON HILLS, MI, 48335	US Mail (1st Class)
24846	EZE RENTAL CENTER, 1524 WEST FOURTH ST, MANSFIELD, OH, 44906	US Mail (1st Class)
24846	F & L HYDRAULICS SERVICES, 1410 E. 6TH STREET EXT., ASHTABULA, OH, 44004	US Mail (1st Class)
24846	F & S TOOL INC., 2300 POWELL AVE., ERIE, PA, 16506	US Mail (1st Class)
24846	F.A.S. SALES LLC, 6074 STATE RD., CLEVELAND, OH, 44134	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	FABER PUMP & EQUIPMENT INC., 6027 SCHUMACHER PARK DR., WEST CHESTER, OH, 45069	US Mail (1st Class)
24846	FABNET ASSOCIATES, INC., 48 BRIGGSBORO LANE, PO BOX 128, FAIRPORT, NY, 14450	US Mail (1st Class)
24846	FABRICATING UNLIMITED, 8463 SIMONS RD., WILLIAMSFIELD, OH, 44093	US Mail (1st Class)
24846	FADHIL AL OTHMAN, 6531 MANSFIELD, DETROIT, MI, 48228	US Mail (1st Class)
24846	FAIRBORN EQUIPMENT CO.INC., 225 TARHE TRL, UPPER SANDUSKY, OH, 43351-8700	US Mail (1st Class)
24846	FAIRFIELD INN CLEVELAND-STREETSBORO, 9783 STATE RT. 14, STREETSBORO, OH, 44241	US Mail (1st Class)
24846	FAIRLAWN TAX DEPT, 3487 S SMITH RD, FAIRLAWN, OH, 44333-3007	US Mail (1st Class)
24846	FAIRLAWN, CITY OF, FAIRLAWN TAX DEPT, 3487 S SMITH RD, FAIRLAWN, OH, 44333-3007	US Mail (1st Class)
24846	FALCON SALES COMPANY, 370 BREWSTER, ROCHESTER HILLS, MI, 48309	US Mail (1st Class)
24846	FALCON TOOL COMPANY, INC., 7500 HUB PARKWAY, CLEVELAND, OH, 44125	US Mail (1st Class)
24846	FALLSWAY EQUIPMENT CO., 1277 DEVALERA ST, AKRON, OH, 44310-2454	US Mail (1st Class)
24846	FAM, INC., PO BOX 71941, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	FAMILY TIRE CENTER, 1227 HARDING WAY EAST, GALION, OH, 44833	US Mail (1st Class)
24846	FANUC ROBOTICS, DRAWER 5739, PO BOX 79001, DETROIT, MI, 48279-5739	US Mail (1st Class)
24846	FANUC ROBOTICS, DRAWER #5739, PO BOX 79001, DETROIT, MI, 48279-5739	US Mail (1st Class)
24846	FANUC ROBOTICS MEXICO, CIRCUITO AGUASCALIENTES, NORTE #136, PARQUE IND. DEL VALLE DE A, AGUAS CALIENTES, AGS, 20140MEXICO	US Mail (1st Class)
24846	FARGO MACHINE COMPANY, 998 STEVENSON ROAD, ASHTABULA, OH, 44004	US Mail (1st Class)
24846	FARNAM CUSTOM PRODUCTS, INC., 90 BRADLEY BRANCH RD., PO BOX 1377, ARDEN, NC, 28704	US Mail (1st Class)
24846	FAST HEAT, INC., 135 S LASALLE, DEPT. 4723, CHICAGO, IL, 60674-4723	US Mail (1st Class)
24846	FAST TEK GROUP LLC, 280 W MORLEY DR, SAGINAW, MI, 48601	US Mail (1st Class)
24846	FASTI USA, INC., 927 NORTH STATE STREET, ELGIN, IL, 60123	US Mail (1st Class)
24846	FAURECIA AUTOMOTIVE, 6141 VIPOND DRIVE, MISSISSAUGA, ON, L5T 2B2CANADA	US Mail (1st Class)
24846	FAURECIA INTERIOR SYSTEMS, 2380 MEIJER DRIVE, TROY, MI, 48084	US Mail (1st Class)
24846	FAWZI ALI-SALEH, 8707 SAMUEL BARTON DR., BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	FAY INDUSTRIES, INC., 17200 FOLTZ PKWY, STRONGSIVLLE, OH, 44149	US Mail (1st Class)
24846	FAYETTE MEDICAL CENTER, PO DRAWER 710, FAYETTE, AL, 35555	US Mail (1st Class)
24846	FBR DISTRIBUTING LLC, 689 WARDS CORNER RD, LOVELAND, OH, 45140	US Mail (1st Class)
24846	FCL PRECISION WELD, 76 E THIRD STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	FEATHERS SECURITY INC., 219 BULLENS LANE, WOODLYN, PA, 19094	US Mail (1st Class)
24846	FEDERAL EXPRESS CORP., 3 GALLERIA TOWER, 13155 NOEL RD , SUITE 1600, DALLAS, TX, 75240	US Mail (1st Class)
24846	FEDERAL FLOORING, 3200 KETTERING BLVD., DAYTON, OH, 45439	US Mail (1st Class)
24846	FEDERAL RESERVE BANK OF C, 717 GRANT ST., PITTSBURGH, PA, 15233	US Mail (1st Class)
24846	FEDERAL TRAINING CENTER, PMB 317, 105 S STATE ST, OREM, UT, 84058-5419	US Mail (1st Class)
24846	FEDEX FREIGHT WEST INC., 1715 AARON BRENNER DR, SUITE 600, MEMPHIS, TN, 38120	US Mail (1st Class)
24846	FEDEX GROUND, 3875 AIRWAYS, MODULE H3, DEPARTMENT 4634, MEMPHIS, TN, 38116	US Mail (1st Class)
24846	FEIT ELECTRIC COMPANY, 4901 GREGG RD., PICO RIVERA, CA, 90660-2108	US Mail (1st Class)
24846	FERGUSON CONSTRUCTION CO., 400 CANAL ST, SIDNEY, OH, 45365	US Mail (1st Class)
24846	FERGUSON ENTERPRISES, 1111 JAMES P COLE BLVD., FLINT, MI, 48503	US Mail (1st Class)
24846	FERRIS STATE UNIVERSITY, 330 OAK STREET, WEST 100, BIG RAPIDS, MI, 49307-2031	US Mail (1st Class)
24846	FERRO CORPORATION, 1000 LAKESIDE AVE E, CLEVELAND, OH, 44114-1117	US Mail (1st Class)
24846	FESTO CORPORATION, 395 MORELAND RD, HAUPPAUGE, NY, 11788-3909	US Mail (1st Class)
24846	FET ENGINEERING, INC., 903 NUTTER DRIVE, BARDSTOWN, KY, 40004	US Mail (1st Class)
24846	FGH SYSTEMS, INC., 2 RICHWOOD PLACE, DENVILLE, NJ, 07834	US Mail (1st Class)
24846	FIDELIS AUTOMATION, INC., 201 E 5TH STREET, MANSFIELD, OH, 44902	US Mail (1st Class)
24846	FIES SCALES & SYSTEMS, INC, 570 LEO ST., DAYTON, OH, 45404-1506	US Mail (1st Class)
24846	FIKE CORPORATION, 12230 COLLECTIONS CENTER DRIVE, CHICAGO, IL, 60693	US Mail (1st Class)
24846	FILTRATION GROUP, INC., 912 E WASHINGTON ST, JOLIET, IL, 60433-1268	US Mail (1st Class)
24846	FINDLEY DAVIES, INC., 300 MADISON AVE STE 1000, TOLEDO, OH, 43604-1567	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	FINLEY TECH, 1415 S CHESTNUT ST, OWOSSO, MI, 48867	US Mail (1st Class)
24846	FINN & CONWAY/GLATLE, 7524 W 98TH PL, BRIDGEVIEW, IL, 60455-2312	US Mail (1st Class)
24846	FINNEY REFRIGERATION INC., 1101 2ND STREET NE, CANTON, OH, 44704	US Mail (1st Class)
24846	FIRE PROTECTION SERVICE, 1022 DUEBER AVENUE SW, CANTON, OH, 44706	US Mail (1st Class)
24846	FIRST CALL QUALITY SVS, 6960 HILLSDALE CT, INDIANAPOLIS, IN, 46250	US Mail (1st Class)
24846	FIRST ENERGY SERVICES CORP., DEPT. 652, COLUMBUS, OH, 43265-0652	US Mail (1st Class)
24846	FIRST INDEX, 26 PARSIPPANY RD., WHIPPANY, NJ, 07981	US Mail (1st Class)
24846	FIRSTCREDIT INTERNATIONAL, PO BOX 13283, AKRON, OH, 44334-8683	US Mail (1st Class)
24846	FISCAL OFFICER - COUNTY OF SUMMIT, JOHN A DONOFRIO, 175 S MAIN STREET- RM 211, AKRON, OH, 44308	US Mail (1st Class)
24846	FISHER ELECTRIC, PC JET ENTERPRISES INC., 2025 DUEBER AVE. SW, CANTON, OH, 44706	US Mail (1st Class)
24846	FISHER SCIENTIFIC, 3970 JOHNS CREEK COURT, SUITE 500, SUWANEE, GA, 30024	US Mail (1st Class)
24846	FISHER SCIENTIFIC, DEPT 414328-01, PITTSBURGH, PA, 15230	US Mail (1st Class)
24846	FISHER SCIENTIFIC, ACCT# 096952-001, P. O. BOX 360153, PITTSBURGH, PA, 15250-6153	US Mail (1st Class)
24846	FISHER SCIENTIFIC, DEPT 294820, PO BOX 360153, PITTSBURGH, PA, 15250-6153	US Mail (1st Class)
24846	FLAMBEAU CORPORATION, DRAWER 78306, MILWAUKEE, WI, 53278-0306	US Mail (1st Class)
24846	FLAMBEAU JAPAN CP LTD., 13-114 KUMANO-MACHI 8-CHOME, NISHINOMIYA HYOGO, JAPAN	US Mail (1st Class)
24846	FLAMBEAU PRODUCTS CORP., DRAWER 78345, MILWAUKEE, WI, 53278	US Mail (1st Class)
24846	FLEX TECHNOLOGIES, DEPT. L-1259, COLUMBUS, OH, 43260	US Mail (1st Class)
24846	FLEXCON CORPORATION, 673 MORRIS AVENUE, SPRINGFIELD, NJ, 07081-1590	US Mail (1st Class)
24846	FLEX-TEMP WARREN, INC., PO BOX 2517, SANDUSKY, OH, 44871-2517	US Mail (1st Class)
24846	FLINT JOURNAL, ATTN: CLASSIFIED AD, PO BOX 386, FLINT, MI, 48501	US Mail (1st Class)
24846	FLINT MACHINE TOOLS, 3710 HEWATT COURT, SNELLVILLE, GA, 30039	US Mail (1st Class)
24846	FLORIDA OPERATORS, 33100 TAVERES AVE, LEESBURG, FL, 34788	US Mail (1st Class)
24846	FLOW DYNAMICS, 1024 11TH CT W, BIRMINGHAM, AL, 35204-1808	US Mail (1st Class)
24846	FLOWER CART FLORIST, 531 HARDINGWAY WEST, GALION, OH, 44833	US Mail (1st Class)
24846	FLOWERS WELDING & MACH CO, CHASSIN TECHNOLOGY,INC., 706 TUSCARAWAS ST., E., CANTON, OH, 44707	US Mail (1st Class)
24846	FLOWTRONICS, INC., 10435 ORTONVILLE RD, CLARKSTON, MI, 48348	US Mail (1st Class)
24846	FLOYD DELONG & SON EXCAVATING, 1295 MORRIS ROAD, LAPEER, MI, 48446	US Mail (1st Class)
24846	FLUID POWER & BEARINGS, 830 S. PIKE ROAD #5, SUMTER, SC, 29150	US Mail (1st Class)
	FLUID POWER TECH, 145 N FOURTH STREET, BOX 1273, LEWISTON, NY, 14092	US Mail (1st Class)
24846	FLUIDRAULICS, INC., 9140 RAVENNA ROAD, TWINSBURG, OH, 44087-2441	US Mail (1st Class)
24846	FLUKE CORPORATION, 7272 COLLECTION CENTER DRIVE, CHICAGO, IL, 60693	US Mail (1st Class)
24846	FLYNN BURNER CORP., 425 FIFTH AVE., PO BOX 431, NEW ROCHELLE, NY, 10801	US Mail (1st Class)
24846	FOAMEY, 074 CANDUCKY DD. ADOADE, NY, 44005	US Mail (1st Class)
24846	FOAMEX, 274 SANDUSKY RD, ARCADE, NY, 14005	US Mail (1st Class)
24846	FOAMEX, PO DRAWER CS-100324, ATLANTA, GA, 30384-0324	US Mail (1st Class)
24846	FOCUSED SOLUTIONS, 101 STONECREST, GREER, SC, 29650	US Mail (1st Class)
24846	FOODTOWN PLUS, 100 SELTZER STATION, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	FOR YOUR HEALTH, 10411 FREDERICK PIKE, VANDALIA, OH, 45377	US Mail (1st Class)
24846	FOR YOUR INFORMATION, PO BOX 647, NORTHVILLE, MI, 48167-0647	US Mail (1st Class)
24846	FORESIGHT TECHNOLOGY GROUP, 6670 WEST SNOWVILLE ROAD, UNIT 9, BRECKSVILLE, OH, 44141	US Mail (1st Class)
24846	FOREST CITY TECHNOLOGIES, 299 CLAY ST, WELLINGTON, OH, 44090	US Mail (1st Class)
24846	FOREVER SHINE CLEANING SERVICES INC, PO BOX 1476, MEDINA, OH, 44258	US Mail (1st Class)
24846	FORMOSA PLASTICS, 9 PEACH TREE HILL RD., LIVINGSTON, NJ, 07039	US Mail (1st Class)
24846	FORREST SCREW MACHINE, 139 ILLINOIS AVE S, MANSFIELD, OH, 44905	US Mail (1st Class)
24846	FORRESTER-SMITH INC., 213 HOBBS ST, TAMPA, FL, 33619-8033	US Mail (1st Class)
24846	FORUM DISTRIBUTION, 12536 POINTER PLACE, FISHERS, IN, 46038	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	FORWARD TECHNOLOGY, 13500 COUNTY ROAD 6, MINNEAPOLIS, MN, 55441	US Mail (1st Class)
24846	FOST CORP., PO BOX 145, LAPEER, MI, 48446	US Mail (1st Class)
24846	FOSTER BLUE WATER OIL, LLC, 7430 LAKESHORE RD, LEXINGTON, MI, 48450	US Mail (1st Class)
24846	FOUR POINTS HOTEL, 320 BRYANT DRIVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	FOUR SLIDE PRODUCTS, 752 BANGOR RD., RTE 191, NAZARETH, PA, 18064	US Mail (1st Class)
24846	FOWLER PRODUCTS INC., 810 COLBY ROAD, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	FOX PLUMBING & HEATING, 126 N. WASHINGTON ST., GALION, OH, 44833	US Mail (1st Class)
24846	FOX VALLEY MARKING SYST., 640 INDUSTRIAL DRIVE, CARY, IL, 60013	US Mail (1st Class)
24846	FPS TECHNOLOGIES, INC., 1417 FORESTDALE BLVD, BIRMINGHAM, AL, 35214	US Mail (1st Class)
24846	FRANCES PORTER, 14625 STAGECOACH DR, RD 2, GARRETTSVILLE, OH, 44231	US Mail (1st Class)
24846	FRANCIS SULLIVAN, 39286 DILLINGHAM, WESTLAND, MI, 48186-3726	US Mail (1st Class)
24846	FRANCISCO BRICIO, COMMERCE BANK, 12633 DORSETT RD., MARYLAND HEIGHTS, MO, 63043-2199	US Mail (1st Class)
24846	FRANK HERTZEL, 329 GENESSEE RD, IRMO, SC, 29063	US Mail (1st Class)
24846	FRANKENMUTH IND. SERVICES, 310 LIST ST., PO. BOX 357, FRANKENMUTH, MI, 48734-0357	US Mail (1st Class)
24846	FRANKLIN COVEY CO., POBOX 31456, SALT LAKE CITY, UT, 84131	US Mail (1st Class)
24846	FRATERNAL ORDER OF POLICE, GALION LODGE NO. 71, INC., PO BOX 572, GALION, OH, 44833	US Mail (1st Class)
24846	FRAZA EQUIPMENT, INC., 15725 E 12 MILE RD, ROSEVILLE, MI, 48066-1844	US Mail (1st Class)
24846	FREA LLC, 4000 EMBASSY PKWY, STE 400, AKRON, OH, 44333	US Mail (1st Class)
24846	FRED BLACK, 2935 CROWN POINT, CORTLAND, OH, 44410	US Mail (1st Class)
24846	FRED J. BREITEN LUMBER COMPANY, 44250 NORTH AVE, CLINTON TOWNSHIP, MI, 48036-1168	US Mail (1st Class)
24846	FRED MCFARLAND, 892 UNITY DR., MANSFIELD, OH, 44905	US Mail (1st Class)
24846	FREEDOM TECHNOLOGIES, CORP, 10370 CITATION DRIVE, SUITE 200, BRIGHTON, MI, 48116	US Mail (1st Class)
24846	FREEMAN COMPANY, 911 GRAHAM DR, FREMONT, OH, 43420-4086	US Mail (1st Class)
24846	FREEMAN ELECTRIC COMPANY, 1616 25TH AVE, TUSCALOOSA, AL, 35401-4521	US Mail (1st Class)
24846	FREEMAN MFG. & SUPPLY CO., BOX 931234, CLEVELAND, OH, 44193-1358	US Mail (1st Class)
24846	FREEMAN PATTERN & SUPPLY, PO BOX 931234, CLEVELAND, OH, 44193-1358	US Mail (1st Class)
24846	FREEMAN TOOL, 911 GRAHAM DR., FREMONT, OH, 43420	US Mail (1st Class)
24846	FREMONT PLASTIC MOLDS, 100 PARK PL, CHAGRIN FALLS, OH, 44022-4442	US Mail (1st Class)
24846	FREMONT PLASTIC MOLDS, DIV OF MIDWEST TOOLING GROUP, 4711 N STATE ROUTE 19, FREMONT, OH, 43420	US Mail (1st Class)
24846	FREUDENBERG NOK, BOX 73229, CHICAGO, IL, 60673	US Mail (1st Class)
24846	FREUDENBERG NOK, BOX 73229, CHICAGO, IL, 60673	US Mail (1st Class)
24846	FREUND CAN COMPANY, 36690 TREASURY CENTER, CHICAGO, IL, 60694-6600	US Mail (1st Class)
24846	FRIEND ENGINEERING & MACH CO., 67 S. ILLINOIS AVE., MANSFIELD, OH, 44905	US Mail (1st Class)
24846	FUN FACTORY BOUNCING PARTIES, 2741 SOUTHSIDE DR., TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	G & L SUPPLY COMPANY, 455 LONGVIEW AVENUE W, MANSFIELD, OH, 44903	US Mail (1st Class)
24846	G & W DELIVERY SERVICE, PO BOX 91313, CLEVELAND, OH, 44101	US Mail (1st Class)
24846	G.B.I.E. INC., 4461 4TH CON. RD., R.R. #4, AMHERSTBURG, ON, N9V 2Y9CANADA	US Mail (1st Class)
24846	G.F. GOODMAN & SON, INC., 2 IVYBROOK BLVD, WARMINSTER, PA, 18974	US Mail (1st Class)
24846	G.F. GOODMAN & SON, INC., TWO IVYBROOK BLVD, IVYLAND, PA, 18974	US Mail (1st Class)
24846	G.J. SUNBURG & ASSOC., 23629 LIBERTY STREET, FARMINGTON, MI, 48335	US Mail (1st Class)
24846	G.L.A. WATER INC., 17802 RAPIDS RD., HIRAM, OH, 44234	US Mail (1st Class)
24846	G.T. TOOLS LTD., COXMOOR ROAD, SUTTON-IN-ASHFIELD, NOTTINGHAM, NG17 4NEUNITED KINGDOM	US Mail (1st Class)
24846	G/B SALES & SERVICE, 39550 SCHOOLCRAFT, PLYMOUTH, MI, 48170	US Mail (1st Class)
24846	GALA LANES, 739 PORTLAND WAY SOUTH, GALION, OH, 44833	US Mail (1st Class)
24846	GALAXY MOLD & MACHINE INC., 1064 GRANT STREET, AKRON, OH, 44311	US Mail (1st Class)
24846	GALION AUTO & TRUCK, LTD., 540 WOOD STREET, GALION, OH, 44833	US Mail (1st Class)
24846	GALION BOOSTERS CLUB, 207 HARDING WAY WEST, GALION, OH, 44833	US Mail (1st Class)
24846	GALION CANVAS PRODUCTS, 385 SOUTH MARKET STREET, GALION, OH, 44833	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	GALION CO-ED SOFTBALL LEAGUE, C/O TOM FORTON, 339 FIRST AVENUE, GALION, OH, 44833	US Mail (1st Class)
24846	GALION COMM. CTR. YMCA, 500 GILL AVE., GALION, OH, 44833	US Mail (1st Class)
24846	GALION COUNTRY CLUB, INC., 4810 STATE RTE 309, GALION, OH, 44833	US Mail (1st Class)
24846	GALION HIGH SCHOOL, 200 NORTH UNION ST., GALION, OH, 44833	US Mail (1st Class)
24846	GALION INQUIRER, 378 NORTH MARKET, PO BOX 648, GALION, OH, 44833	US Mail (1st Class)
24846	GAMMAFLUX, 113 EXECUTIVE DR, SUITE 106, STERLING, VA, 20166	US Mail (1st Class)
24846	GANDERT DOOR CO., 3077 PARK AVE, PO BOX 143, ONTARIO, OH, 44862	US Mail (1st Class)
24846	GARBER ELECTRICAL CONTRACTORS, 6234 WEBSTER ST, DAYTON, OH, 45414	US Mail (1st Class)
24846	GARON PRODUCTS INC., 2430 STATE HWY NO 34, WALL TOWNSHIP, NJ, 08736	US Mail (1st Class)
24846	GARRIE M BUTLER, 1821 10TH AVENUE, APT 11A, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	GARY ALLRED, 4808 LEEWARD AVE, NORTHPORT, AL, 35473-1979	US Mail (1st Class)
24846	GARY CAMPBELL, 8707 SAMUEL BARTON DRIVE, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	GARY COLE, TREASURER OF CRAWFORD COUNTY, PO BOX 565, BUCYRUS, OH, 44820-0565	US Mail (1st Class)
24846	GARY DEGREEN, 1401 INDUSTRIAL DRIVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	GARY HONEYCUTT, 1401 INDUSTRIAL PARK, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	GATESMCDONALD, 215 N FRONT ST, COLUMBUS, OH, 43215-2255	US Mail (1st Class)
24846	GATEWAY, 7565 IRVINE CENTER DR, IRVINE, CA, 92618-2930	US Mail (1st Class)
24846	GATEWAY COMPANIES INC., 7565 IRVINE CENTER DR, IRVINE, CA, 92618-2930	US Mail (1st Class)
24846	GATEWAY SAFETY NET PUB., 276 CAVALIER STREET, PO BOX 559, PEMBINA, ND, 58271-0559	US Mail (1st Class)
24846	GAUTHIER'S OFFICE EQUIP SERV, 4736 BEECHWOOD AVE. NE, PARIS, OH, 44669	US Mail (1st Class)
24846	GBJ GROUP INC., 30600 TELEGRAPH RD., SUITE 3250, BINGHAM FARMS, MI, 48025	US Mail (1st Class)
24846	GCJFS, 12480 RAVENSWOOD DRIVE, CHARDON, OH, 44024	US Mail (1st Class)
24846	GE CANADA POLMERLAND, 1 PLASTICS AVE, PITTSFIELD, MA, 01201-3662	US Mail (1st Class)
24846	GE COMMERCIAL MATERIALS, AV CHURUBUSCO 3900, MONTERREY, NL, 64517MEXICO	US Mail (1st Class)
24846	GE INFRASTRUCTURE WATER & PROCESS TECHNOLOGIES (GE, BETZ, INC.), 7796 COLLECTION CENTER DRIVE, CHICAGO, IL, 60693-0077	US Mail (1st Class)
24846	GE PANAMETRICS, BANK OF AMERICA, LOCKBOX 848502, DALLAS, TX, 75202	US Mail (1st Class)
24846	GE PLASTICS, 9930 KINCEY AVE, HUNTERSVILLE, NC, 28078	US Mail (1st Class)
24846	GE POLYERSHAPES, 4168 COLLECTIONS CENTER, CHICAGO, IL, 60693	US Mail (1st Class)
24846	GE POLYMERLAND, 12200 HERBERT WAYNE CT., SUITE 150, HUNTERSVILLE, NC, 28078	US Mail (1st Class)
24846	GE POLYMERLAND, 9930 KINCEY AVE., HUNTERSVILLE, NC, 28078	US Mail (1st Class)
24846	GE SILICONES, GE PLASTICS, PO BOX 640959, PITTSBURG, PA, 15264	US Mail (1st Class)
24846	GE SUPPLY, 2 CORPORATE DR, SHELTON, CT, 06484-6238	US Mail (1st Class)
24846	GEAUGA CLEANERS, 15615 W HIGH STREET, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	GEAUGA COUNTY AUDITOR, 231 MAIN ST # 2, CHARDON, OH, 44024-1235	US Mail (1st Class)
24846	GEAUGA COUNTY AUDITOR, 231 MAIN ST, #2, CHARDON, OH, 44024-1235	US Mail (1st Class)
24846	GEAUGA COUNTY CIC, 470 CENTER PLACE, BLDG 1, CHARDON, OH, 44024	US Mail (1st Class)
24846	GEAUGA COUNTY COURT OF COMMON PLEAS, 100 SHORT COURT, CHARDON, OH, 44024	US Mail (1st Class)
24846	GEAUGA COUNTY TREASURER, CHRISTOPHER P. HITCHCOCK, 211 MAIN ST, STE 1B, CHARDON, OH, 44024-1249	US Mail (1st Class)
24846	GEAUGA COUNTY TREASURER, CHRISTOPHER P HITCHCOCK, 211 MAIN STREET, CHARDON, OH, 44024-1249	US Mail (1st Class)
24846	GEAUGA GLASS SERVICE, 14333 KINSMAN RD #3, BURTON, OH, 44021-9666	US Mail (1st Class)
24846	GEAUGA LOCK & KEY, 13069 MADISON RD., MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	GEAUGA NUISANCE CONTROL, 11493 CLARIDON TROY RD., CHARDON, OH, 44024	US Mail (1st Class)
24846	GEAUGA VISION, 14901 BOX 1210, PO BOX 1210, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	GEBHART'S OFFICE MACHINES, 611 BEER ROAD, MANSFIELD, OH, 44906	US Mail (1st Class)
24846	GEDDES TOOL & GAUGE, 3096 DEVON DR. UNIT #25, WINDSOR, ON, N8X 4L2CANADA	US Mail (1st Class)
24846	GEFRAN ISI, INC., ATTN: ACCOUNTS RECEIVABLE, 8 LOWELL AVENUE, WINCHESTER, MA, 01890- 1130	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	GEHRING`S INDUSTRIAL, SERVICE, PO BOX 121, HUBBARD, OH, 44425	US Mail (1st Class)
24846	GEIGER TECHNIK GMBH, POSTFACH 1354, PLZ82453, BREITENAU, GARMISCH-PARTENKIRCHEN, D-82467GERMANY	US Mail (1st Class)
24846	GEM CITY ENGINEERING CO., 401 LEO ST, DAYTON, OH, 45404-1009	US Mail (1st Class)
24846	GENE LARY'S TROPHY SHOP, 2724 LURLEEN WALLACE BLVD, NORTHPORT, AL, 35476	US Mail (1st Class)
24846	GENERAL BINDING CORP., 70 E LAKE ST FL 1200, CHICAGO, IL, 60601-5915	US Mail (1st Class)
24846	GENERAL COLOR & CHEMICAL, 604 VALLEY ST, MINERVA, OH, 44657-1547	US Mail (1st Class)
24846	GENERAL DATA COMPANY INC., 4354 FERGUSON DR, CINCINNATI, OH, 45245-1667	US Mail (1st Class)
24846	GENERAL DATA COMPANY,INC., LOCATION #0558, 4354 FERGUSON DRIVE, CINCINNATI, OH, 45245	US Mail (1st Class)
24846	GENERAL FASTENERS, DRAWER #5735, PO BOX 79001, DETROIT, MI, 48279-5735	US Mail (1st Class)
24846	GENERAL FASTENERS CO., DEPT. 77308, PO BOX 77000, DETROIT, MI, 48277-0308	US Mail (1st Class)
24846	GENERAL POLYMERS, ASHLAND CHEMICAL CO., PO BOX 371002, PITTSBURGH, PA, 15250	US Mail (1st Class)
24846	GENERAL RENTAL, 7047 WHIPPLE AVE. NW, NORTH CANTON, OH, 44720	US Mail (1st Class)
24846	GENERAL WAREHOUSE & INSPECTION, 61 MCKNIGHT DR., LEXINGTON, TN, 38351	US Mail (1st Class)
24846	GENESEE PACKING, 2010 N DORT HWY, FLINT, MI, 48506-2937	US Mail (1st Class)
24846	GENEX, 10003 WASHINGTON BLVD, CULVER CITY, CA, 90232-2727	US Mail (1st Class)
24846	GEOLOGISTICS, 5496 PAYSPHERE CIRCLE, CHICAGO, IL, 60674	US Mail (1st Class)
24846	GEORGE JOHNSON, 818 3RD AVENUE, LAKE ODESSA, MI, 48849	US Mail (1st Class)
24846	GEORGE PASQUARELLO SR., 5126 KEVIN DRIVE, MCKEAN, PA, 16426	US Mail (1st Class)
24846	GEORGE YOUNG, 900 S. WILEY, PO BOX 320, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	GERALD SCHARDT, C/O CEP PRODUCTS, LLC, PO BOX 320, CRESTLINE, OH, 44827-0320	US Mail (1st Class)
24846	GERARD DANIEL WORLDWIDE, 34 BARNHART DR, HANOVER, PA, 17331-9586	US Mail (1st Class)
24846	GERRY WEIMANN, 109 COUNTRYSIDE DRIVE, CHAGRIN FALLS, OH, 44022	US Mail (1st Class)
24846	GESSWEIN, PO BOX 3998, BRIDGEPORT, CT, 06605	US Mail (1st Class)
24846	GEYER'S FRESH FOODS, 100 SELTZER STATION, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	GHESQUIERE PLASTICS, 20450 HARPER AVE, HARPER WOODS, MI, 48225	US Mail (1st Class)
24846	GI PLASTEK L.P., 3 PERKINS WAY, NEWBURYPORT, MA, 01950-4007	US Mail (1st Class)
24846	GIACOMIN GROUP, INC., 24725 WEST 12-MILE ROAD, SUITE 308, SOUTHFIELD, MI, 48034	US Mail (1st Class)
24846	GIANT EAGLE INC., 907 W MAPLE ST, HARTVILLE, OH, 44632	US Mail (1st Class)
24846	GIBBS MAINTENANCE SERVICES, 15784 OLD STATE RD., SUITE 83, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	GIERKE-ROBINSON, 3929 W RIVER DR, DAVENPORT, IA, 52802-2415	US Mail (1st Class)
24846	GIFT CERTIFICATES.COM, 11510 BLONDO ST., SUITE 103, OMAHA, NE, 68164	US Mail (1st Class)
24846	GILBERT KERR, 628 RANVEEN DRIVE, WHITE LAKE, MI, 48085	US Mail (1st Class)
24846	GINNIE HICKS, 119 EAST DAYTON STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	GIRL SCOUTS OF HERITAGE TRAILS, 35 NORTH PARK, MANSFIELD, OH, 44902	US Mail (1st Class)
24846	GITTLEN CANCER RESEARCH INSTITUTE, C/O CARLISLE SYNTEC, INC., PO BOX 7000, CARLISLE, PA, 17013	US Mail (1st Class)
24846	GJ SUNBURG & ASSOC, 23629 LIBERTY ST, FARMINGTON, MI, 48335	US Mail (1st Class)
24846	GKS INSPECTION SERVICES, 4533 FIVE MILE RD, PLYMOUTH, MI, 48170	US Mail (1st Class)
24846	GLASER SOFTWATER, INC., 939 WEST MAIN ST., TIPP CITY, OH, 45371	US Mail (1st Class)
24846	GLASS & ASSOCIATES, INC., 4571 STEPHEN CIRCLE N.W., STE. 130, CANTON, OH, 44718	US Mail (1st Class)
24846	GLASS CITY PLASTICS INC., 971 HAMILTON DRIVE, HOLLAND, OH, 43528	US Mail (1st Class)
24846	GLEASON REEL CORPORATION, 600 S CLARK ST, MAYVILLE, WI, 53050-1821	US Mail (1st Class)
24846	GLENOIT CORP. FABRIC DIV., 3001 N MAIN ST, TARBORO, NC, 27886-1922	US Mail (1st Class)
24846	GLOBAL ENG. DOCUMENTS, DEPT #1501, DENVER, CO, 80291-1501	US Mail (1st Class)
24846	GLOBAL ENG. DOCUMENTS, 15 INVERNESS WAY EAST, ENGLEWOOD, CO, 80112	US Mail (1st Class)
24846	GLOBAL INDUSTRIAL EQUIPMENT, 120 SATELLITE BLVD., SUWANEE, GA, 30024	US Mail (1st Class)
24846	GLOBAL INDUSTRIAL EQUIPMENT, 11 HARBOR PARK DR, PORT WASHINGTON, NY, 11050-4656	US Mail (1st Class)
24846	GLOBAL INSIGHT, 1000 WINTER ST, WALTHAM, MA, 02451	US Mail (1st Class)
24846	GLOBAL QUALITY INDUSTRIES, LLC, 6401 MILLER ROAD, DEARBORN, MI, 48126	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	GLOBAL SOURCE LINK, 2401 COOPER ST, FORT SCOTT, KS, 66701-3033	US Mail (1st Class)
24846	GLS CORPORATION, 4028 PAYSPHERE CIRCLE, CHICAGO, IL, 60674	US Mail (1st Class)
24846	GLYCON CORPORATION, 912 INDUSTRIAL DRIVE, TECUMSEH, MI, 49286	US Mail (1st Class)
24846	GM ARLINGTON PLT.18008, 2525 EAST ABRAM DOCK A, ARLINGTON, TX, 76010	US Mail (1st Class)
24846	GM JANESVILLE LIGHT, LIGHT DUTY PLANT #1, JANESVILLE, WI, 53546	US Mail (1st Class)
24846	GM NAO SMALL CAR GROU, LANSING, 300 RENAISSANCE CENTER, DETROIT, MI, 48265-0001	US Mail (1st Class)
24846	GM TRUCK & BUS 48063, C/O JOHNSON CONTROLS INC., 1207 BELMAR DRIVE, BELCAMP, MD, 21017	US Mail (1st Class)
24846	GOLD KEY PROCESSING, LTD., 14910 MADISON RD, MIDDLEFIELD, OH, 44062-8403	US Mail (1st Class)
24846	GOLDEN CIR. CREDIT UNION, 1215 GREENFIELD AVE., SW, PO BOX 6300, CANTON, OH, 44706	US Mail (1st Class)
24846	GOOD & FAIR MOVING SERVICE,INC, 100 SPACELAB COURT, HUNTSVILLE, AL, 35806	US Mail (1st Class)
24846	GOOD NEWS, 14909 NORTH STATE AVE, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	GOODWAY TECHNOLOGIES CORP, 420 WEST AVENUE, STAMFORD, CT, 06902	US Mail (1st Class)
24846	GOODWILL INDUSTRIES, 621 GOODWILL DRIVE, PO BOX 2926, ASHTABULA, OH, 44005-2926	US Mail (1st Class)
24846	GORDON COOK, 8249 KENSINGTON BLVD, DAVISON, MI, 48423	US Mail (1st Class)
24846	GORDON FLESCH COMPANY, INC., 2675 RESEARCH PARK DR, FITCHBURG, WI, 53711-4906	US Mail (1st Class)
24846	GORTRAC, 2300 SOUTH CALHOUN RD., NEW BERLIN, WI, 53151	US Mail (1st Class)
24846	GOSHEN RUBBER CO., INC., DEPT. 77781, DETROIT, MI, 48277	US Mail (1st Class)
24846	GOVDOCS.COM, 380 JACKSON STREET, SUITE 550, ST. PAUL, MN, 55101-2903	US Mail (1st Class)
24846	GPK ENTERPRISES, PO BOX 1374, BRIGHTON, MI, 48116	US Mail (1st Class)
24846	GRAB ON PRODUCTS INC., 350 EAST BEECH, WALLA WALLA, WA, 99362	US Mail (1st Class)
24846	GRAHAM ENGINEERING, 1203 EDEN RD, YORK, PA, 17402-1965	US Mail (1st Class)
24846	GRAINGER, DEPT. 825120991, PALATINE, IL, 60038	US Mail (1st Class)
24846	GRAINGER, DEPT. 080-821302312, PALATINE, IL, 60038-0001	US Mail (1st Class)
24846	GRAINGER, DEPT. 825120991, PALATINE, IL, 60038-0001	US Mail (1st Class)
24846	GRAINGER COMPANY, DIVISION OF W.W. GRAINGER, DEPT 825120991, PALATINE, IL, 60038-0001	US Mail (1st Class)
24846	GRAINGER PARTS OPERATIONS, DEPT 232-813301918, PALATINE, IL, 60038-0001	US Mail (1st Class)
24846	GRAINGER PARTS OPERATIONS, DEPT. 970-801190000, PALATINE, IL, 60038	US Mail (1st Class)
24846	GRAINGER PARTS OPERATIONS, DEPT. 175 - 802207837, PALATINE, IL, 60038	US Mail (1st Class)
24846	GRAINGER, W.W., INC., DEPT. 825120991, PALATINE, IL, 60038-0001	US Mail (1st Class)
24846	GRANBY MOLD, 4380 HAGGERTY ROAD, WALLED LAKE, MI, 48390	US Mail (1st Class)
24846	GRAND TRAVERSE PLASTICS, 5780 MOORE ROAD, WILLIAMSBURG, MI, 49690	US Mail (1st Class)
24846	GRANDVIEW INDUSTRIAL LEAGUE, PO BOX 361, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	GRANDVIEW INN, BOX 1075, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	GRANT INDUSTRIES, DEPT #78069, PO BOX 78000, DETROIT, MI, 48278-0069	US Mail (1st Class)
24846	GRANUTEC, INC., 100 DAVIS ST, DOUGLAS, MA, 01516-2310	US Mail (1st Class)
24846	GRANVILLE ASSOCIATES, D/B/A VISION BUSINESS, 600 LOGAN STREET, CARNEGIE, PA, 15106	US Mail (1st Class)
24846	GRAYBAR ELECTRIC CO, 12444 COLLECTIONS CTR. DR., CHICAGO, IL, 60693	US Mail (1st Class)
24846	GRAYBAR ELECTRIC COMPANY, INC., 12444 COLLECTIONS CENTER DR., CHICAGO, IL, 60693	US Mail (1st Class)
24846	GRAYBILL & COMPANY, 3755 GEORGETOWN RD NE, CANTON, OH, 44704-2625	US Mail (1st Class)
24846	GREAT LAKES AUTOMATION, SERVICES, INC., 2001 PENINSULA DRIVE, SUITE #20, ERIE, PA, 16506	US Mail (1st Class)
24846	GREAT LAKES BANK, 9365 MENTOR AVE., MENTOR, OH, 44060	US Mail (1st Class)
24846	GREAT LAKES CONTROLS, PO BOX 75549, CLEVELAND, OH, 44101	US Mail (1st Class)
24846	GREAT LAKES ELECTRONIC RECYCLING, 19030 FITZPATRICK STREET, DETROIT, MI, 48228	US Mail (1st Class)
24846	GREAT LAKES ELECTRONICS CORP., 12600 GREENFIELD, DETROIT, MI, 48227	US Mail (1st Class)
24846	GREAT LAKES FILTERS, 301 ARCH AVENUE, HILLSDALE, MI, 49242	US Mail (1st Class)
24846	GREAT LAKES POWER, 7455 TYLER BLVD., MENTOR, OH, 44060	US Mail (1st Class)
24846	GREAT LAKES PRINTING, 2926 LAKE AVE., ASHTABULA, OH, 44004	US Mail (1st Class)
24846	GREAT LAKES SERVICE & SALES, 12525 HALE, RIVERVIEW, MI, 48192	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	GREAT LAKES SUPPLY CO., 4115 EAST 116TH STREET, CLEVELAND, OH, 44105	US Mail (1st Class)
24846	GREATER ERIE INDUSTRIAL, DEVELOPMENT CORPORATION, 5240 KNOWLEDGE PARKWAY, ERIE, PA, 16510-4658	US Mail (1st Class)
24846	GREEN OUTDOORS YARD, 1844 BIDDLE ROAD, GALION, OH, 44833	US Mail (1st Class)
24846	GREENBELT INDUSTRIES, INC., 45 COMET STREET, BUFFALO, NY, 14216-1710	US Mail (1st Class)
24846	GREENVILLE DOOR SALES, 320 CHESTNUT STREET, GREENVILLE, OH, 45331	US Mail (1st Class)
24846	GREG ROMANCHOK, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	GREG SMITH, 17187 N LAUREL PK DR, SUITE #208, LIVONIA, MI, 48152	US Mail (1st Class)
24846	GREG SWEET CHEVROLET, 286 E MAIN RD, CONNEAUT, OH, 44030	US Mail (1st Class)
24846	GREGG & ASSOCIATES, INC., 1159 HIGHWAY 25, GALLATIN, TN, 37066-6104	US Mail (1st Class)
24846	GREKO PRINTING & IMAGING, 1052 W. ANN ARBOR RD., PLYMOUTH, MI, 48170	US Mail (1st Class)
24846	GRIFFCO QUALITY SOLUTIONS, INC, 12300 OLD TESSON ROAD, SUITE 400A, ST. LOUIS, MO, 63128	US Mail (1st Class)
24846	GRILLIOT'S HEATING & COOLING, 6801 WENGERLAWN RD, BROOKVILLE, OH, 45309	US Mail (1st Class)
24846	GRINNELL FIRE PROTECTION, SYSTEMS CO., PO BOX 371170 M, PITTSBURGH, PA, 15251	US Mail (1st Class)
24846	GROSS LUMBER COMPANY, 645 CENTRAL AVENUE, CARLISLE, OH, 45005	US Mail (1st Class)
24846	GROWTH AND OPPORTUNITY, 525 S COURT ST, LAPEER, MI, 48446	US Mail (1st Class)
24846	GRUBER, 21439 N 2ND AVE., PHOENIX, AZ, 85027	US Mail (1st Class)
24846	G-TECH SERVICES INC., 17101 MICHIGAN AVE., DERABORN, MI, 48126-2736	US Mail (1st Class)
24846	GUELPH TOOL INC., C/O 24150 GIBSON, WARREN, MI, 48089	US Mail (1st Class)
24846	GUENTHER MECHANICAL INC., PO BOX 97, 1248 MIDDLE ROWSBURG RD., ASHLAND, OH, 44805	US Mail (1st Class)
24846	GUILLERMO RODRIGUEZ, OCOTILLOS NO. 3644, CHIHUAHUA, 31125MEXICO	US Mail (1st Class)
24846	GUNZELMAN INDUSTRIAL SERVICES, 3696 STATE ROUTE 193, KINGSVILLE, OH, 44048-9724	US Mail (1st Class)
24846	GUY WHITTINGTON, 16088 ALPINE DR., LIVONIA, MI, 48154-2542	US Mail (1st Class)
24846	GWI, 61 MCKNIGHT DR., LEXINGTON, TN, 38351	US Mail (1st Class)
24846	GWYER REPROGRAPHICS, 25153 DEQUINDRE RD, MADISON HEIGHTS, MI, 48071	US Mail (1st Class)
24846	GYRO-SPIN INC., 100 FIRE TOWER DRIVE, FIRE TOWER INDUSTRIAL PARK, BUFFALO, NY, 14150-5889	US Mail (1st Class)
24846	H & H HARDWARE, 14980 S STATE ST, MIDDLEFIELD, OH, 44062-7601	US Mail (1st Class)
24846	H & H MACHINED PRODUCTS, 2540 MANCHESTER RD, ERIE, PA, 16506	US Mail (1st Class)
24846	H & L INDUSTRIAL SUPPLY,INC, 55 S. MAIN ST., PMB145, OBERLIN, OH, 44074	US Mail (1st Class)
24846	H & L TOOL COMPANY, 2120 WARFEL AVENUE, ERIE, PA, 16503	US Mail (1st Class)
24846	H A INDUSTRIES, INC., 5510 GATEWOOD ST., STERLING HEIGHTS, MI, 48310	US Mail (1st Class)
24846	H B CHEMICAL CORPORATION, 140 E ASCOT LN, CUYAHOGA FALLS, OH, 44223-3768	US Mail (1st Class)
24846	H B E MACHINE COMPANY, 1100 STATE ROUTE 61, S. MONROEVILLE, OH, 44847	US Mail (1st Class)
24846	H MUEHLSTEIN, 800 CONNECTICUT AVE, NORWALK, CT, 06854-1631	US Mail (1st Class)
24846	H S DIE, DRAWER # 641587, PO BOX 64000, DETROIT, MI, 48264-1587	US Mail (1st Class)
24846	H&H MACHINING, INC., PO BOX 116, OLD ROUTE 66 SOUTH, LEXINGTON, IL, 61753	US Mail (1st Class)
24846	H&H TUBE MANUFACTURING CO., 2525 COLLECTIONS CENTER DRIVE, CHICAGO, IL, 60693	US Mail (1st Class)
24846	H. LEFF ELECTRIC COMPANY, 1163 E 40TH ST, CLEVELAND, OH, 44114-3801	US Mail (1st Class)
24846	H.E.S.C., PO BOX 402125, ATLANTA, GA, 30384-2125	US Mail (1st Class)
24846	H.J. KIRBY CORP., 29325 CLEMENS RD., WESTLAKE, OH, 44145-1093	US Mail (1st Class)
24846	H.P. PRODUCTS, INC., 512 W. GORGAS ST., LOUISVILLE, OH, 44641-3912	US Mail (1st Class)
24846	H.R. DIRECT, GRAYARC-HR DIRECT, %K NELSON TAYLOR CO, PO BOX 3669, MANKATO, MN, 56002-3669	US Mail (1st Class)
24846	H.S. DIE & ENGINEERING, INC., DRAWER #641587, PO BOX 64000, DETROIT, MI, 48264	US Mail (1st Class)
24846	HAB-OBT, PO BOX 906, BANGOR, PA, 18013	US Mail (1st Class)
24846	HAGERTY PRECISION TOOL, 12250 EUREKA ROAD, EDINBORO, PA, 16412-1276	US Mail (1st Class)
24846	HAHN SYSTEMS, 32423 WEST EIGHT MILE ROAD, LIVONIA, MI, 48152-1301	US Mail (1st Class)
24846	HAIDET'S GLASS, 1655 30TH ST. NE, CANTON, OH, 44714	US Mail (1st Class)
24846	HALE CHROME SERVICES, 2282 ALBION ST., TOLEDO, OH, 43608	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	HALEE M. VATTER, 452 CHURCH STREET, APT 9, CAMBRIDGE SPRINGS, PA, 16403	US Mail (1st Class)
24846	HALLMARK TOOLS, 2187 HURON CHURCH ROAD, WINDSOR, ON, N9C 3Y6CANADA	US Mail (1st Class)
24846	HALSTAB-HAMMOND GROUP, 2690 RELIABLE PARKWAY, CHICAGO, IL, 60686-0026	US Mail (1st Class)
24846	HAMILIN, R. MARK, 3560 WEST MARKET STREET, SUITE 300, AKRON, OH, 44333	US Mail (1st Class)
24846	HAMILTON AVTEC, INC., 260 BRUNEL ROAD, MISSISSAUGA, ON, L4Z 1T5CANADA	US Mail (1st Class)
24846	HAMILTON-RYKER COMPANY, 1801 S HARPER RD STE 1C # 4, CORINTH, MS, 38834-6768	US Mail (1st Class)
24846	HAMMERTEK CORPORATION, 885 S CHIQUES RD, MANHEIM, PA, 17545	US Mail (1st Class)
24846	HAMMOND DRIVES & EQUIPMENT, 8527 MIDLAND RD., FREELAND, MI, 48623	US Mail (1st Class)
24846	HAMPTON INN, 5335 BROADMOOR CIRCLE, CANTON, OH, 44709	US Mail (1st Class)
24846	HAMPTON INN, 3041 WEST 12TH STREET, ERIE, PA, 16505	US Mail (1st Class)
24846	HAMPTON INN, 600 HARPER LEE DRIVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	HANK W. MCBROOM, 201 HESS AVENUE, ERIE, PA, 16507	US Mail (1st Class)
24846	HANNAH TOOL CO. INC., PO BOX 207, WINDSOR, OH, 44099-0207	US Mail (1st Class)
24846	HANNON ELECTRIC, 11321 CONCORD-HAMBDEN, PAINESVILLE, OH, 44077	US Mail (1st Class)
24846	HANSALOY CORPORATION, 820 W 35TH ST, DAVENPORT, IA, 52806	US Mail (1st Class)
24846	HANSER GARDNER PUBLICATIONS, 6915 VALLEY AVE, CINCINNATI, OH, 45244-3029	US Mail (1st Class)
24846	HANSON PUBLICATIONS INC., PO BOX 574, ANSONIA, CT, 06401-0574	US Mail (1st Class)
24846	HAPCO, INC., 390 PORTAGE BLVD., KENT, OH, 44240	US Mail (1st Class)
24846	HARBOR FREIGHT TOOLS, 3491 MISSION OAKS BLVD., PO BOX 6010, CAMARILLO, CA, 93012	US Mail (1st Class)
24846	HARCROS CHEMICALS, INC., BR 0033, PO BOX 74583, CHICAGO, IL, 60690	US Mail (1st Class)
24846	HARDY`S CATERING, 4918 M. L. KING JR. BLVD, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	HARKINS SAFETY INC., 400 HASTINGS ST., PITTSBURGH, PA, 15206	US Mail (1st Class)
24846	HARRIS TRICITY CONSTRUCT, 2177 STATE ROUTE 19, BUCYRUS, OH, 44820	US Mail (1st Class)
24846	HARRISON CONSTRUCTION CO. INC., 5870 CHARLIE SHIRLEY RD, NORTHPORT, AL, 35473-8042	US Mail (1st Class)
24846	HARRISON PIPE, 38777 SCHOOLCRAFT RD, LIVONIA, MI, 48150-1033	US Mail (1st Class)
24846	HARRY TRIMBLE, 105 HIGH RIDGE CT., GOODLETTSVILLE, TN, 37072	US Mail (1st Class)
24846	HARTLEY OIL COMPANY, INC., RR 1, RAVENSWOOD, WV, 26164-9801	US Mail (1st Class)
24846	HARTMAN SPRENG CO., INC., BANK ONE, NA, MAIL SUITE OH2-5170, 528 S. MAIN ST., AKRON, OH, 44311-1058	US Mail (1st Class)
24846	HARTVILLE TRUE VALUE, 940 W. MAPLE STREET, HARTVILLE, OH, 44632	US Mail (1st Class)
24846	HARVARD FACTORY AUTOMATION INC, 3 NORTH LINCOLN, HARVARD, IL, 60033	US Mail (1st Class)
24846	HASLER, INC., 19 FOREST PKWY, SHELTON, CT, 06484	US Mail (1st Class)
24846	HAWK PRECISION COMPONENTS, ALLEGHENY CLEARFIELD DIV., PO BOX 88149, CHICAGO, IL, 60695-1149	US Mail (1st Class)
24846	HAWKEYE MOLDING ENGINEERS, INC., PO BOX 216, HWY 5 SOUTH, ALBIA, IA, 52531	US Mail (1st Class)
24846	HAWKINS INDUSTRIAL SERVICE, 244 N MILL ST, WAYNESBURG, OH, 44688-9122	US Mail (1st Class)
24846	HAWKPRECISION COMPANY, 200 PUBLIC SQ STE 1500, CLEVELAND, OH, 44114-2301	US Mail (1st Class)
24846	HAYS COMPANIES, NCB-88, PO BOX 1414, MINNEAPOLIS, MN, 55840-1414	US Mail (1st Class)
24846	HAZARD LABEL ADVISOR, 2230 GLENDENNING RD., KALAMAZOO, MI, 49001	US Mail (1st Class)
24846	HBE MACHINE, INC., 1100 ST. RT. 61, S. MONROEVILLE, OH, 44847	US Mail (1st Class)
24846	HEALTHAMERICA PA, INCPGH COM, 11 STANWIX ST, PITTSBURGH, PA, 15222-1312	US Mail (1st Class)
24846	HEALTHCARE FOR BUSINESS, 150 MENTOR AVENUE #5, PAINESVILLE, OH, 44077	US Mail (1st Class)
24846	HEALTHPLUS OF MICHIGAN, DEPARTMENT 77028, PO BOX 77000, DETROIT, MI, 48277-0028	US Mail (1st Class)
24846	HEARTAED,LLC, 2274 DAVISTA, HIGHLAND, MI, 48356	US Mail (1st Class)
24846	HEAT EQUIPMENT & TECHNOLGY INC, 9805 PRINCETON GLENDALE ROAD, CINCINNATI, OH, 45246- 1017	US Mail (1st Class)
24846	HEAT EQUIPMENT & TECHNOLOGY, 951 W. NORTH BEND ROAD, CINCINNATI, OH, 45224	US Mail (1st Class)
24846	HEAT SOURCE INC., 10486 OLYMPIC DRIVE, SUITE 101, DALLAS, TX, 75220	US Mail (1st Class)
24846	HEAT TECH SYSTEMS, 1496 CLEARVIEW DRIVE, ACWORTH, GA, 30102	US Mail (1st Class)
24846	HEATING AND CONTROLS, INC., 21420 W NORTH AVE., BROOKFIELD, WI, 53045	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	HEAVENLY HAM, 220 MILLCREEK PLAZA, ERIE, PA, 16565	US Mail (1st Class)
24846	HELEN RIDDLE, 1437 S LINDEN, ALLIANCE, OH, 44601	US Mail (1st Class)
24846	HELLO DIRECT, 5893 RUE FERRARI, SAN JOSE, CA, 95138-1858	US Mail (1st Class)
24846	HELPING HANDS & MORE, 212 EAST WESTBROOK RD, BROOKVILLE, OH, 45309	US Mail (1st Class)
24846	HENKLE SURFACE, PO BOX 101432, ATLANTA, GA, 30392	US Mail (1st Class)
24846	HENN, STEINEBACH 18, DORNBIRN, 6850AUSTRIA	US Mail (1st Class)
24846	HENRY A. SELINSKY INC., 4015 23RD ST SW, CANTON, OH, 44706-2313	US Mail (1st Class)
24846	HENRY GOODEN, 1401 INDUSTRIAL DRIVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	HERBERT WOOD PRODUCTS, INC., 15089 WHITE RD., MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	HERCULES CEDCO, DEPT. 77707, BOX 77000, DETROIT, MI, 48277-0707	US Mail (1st Class)
24846	HERDRICH LUBRICANTS LLC, 276 NORTH 150 WEST, PO BOX 157, RUSHVILLE, IN, 46173	US Mail (1st Class)
24846	HERE'S YOUR SIGN, PO BOX 204, 6549 S. MAIN STREET, NORTH KINGSVILLE, OH, 44068	US Mail (1st Class)
24846	HERITAGE FLAG & BANNER, 2600 GLEN VALLEY DR., LEONARD, MI, 48367	US Mail (1st Class)
24846	HERITAGE NEWSPAPERS, 109 E MAIN ST, MANCHESTER, MI, 48158-9312	US Mail (1st Class)
24846	HERITAGE-CRYSTAL CLEAN, 2175 POINT BLVD, SUITE 375, ELGIN, IL, 60123	US Mail (1st Class)
24846	HERMANN SCHMIDT CO., 247 BURNHAM STREET, EAST HARTFORD, CT, 06108	US Mail (1st Class)
24846	HERMANN SCHMIDT CO., 250 NUTMEG RD. S, SUITE J, SOUTH WINDSOR, CT, 06074	US Mail (1st Class)
24846	HERTZ RENTAL EQUIPMENT, 5222 N BRYANT AVE, OKLAHOMA CITY, OK, 73121	US Mail (1st Class)
24846	HEYDEN MOLD & BENCH CO., 233 S. THOMAS ROAD, PO BOX 441, TALLMADGE, OH, 44278	US Mail (1st Class)
24846	HIBBITT, KARLSSON & SORENSEN, 14500 SHELDON ROAD, SUITE 160, PLYMOUTH, MI, 48170-2408	US Mail (1st Class)
24846	HICKS BBQ, 4400 FAYETTE HWY, NORTHPORT, AL, 35475	US Mail (1st Class)
24846	HICKS PLUMBING & HEATING INC., 2406 GREENSBORO AVE., TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	HICKS ROOFING, INC., 2162 PLEASANT VALLEY ROAD NE, NEW PHILADELPHIA, OH, 44663	US Mail (1st Class)
24846	HIGH TECH SYSTEMS & EQUIPMENT, 7950 MORGAN ROAD, CLEVES, OH, 45002	US Mail (1st Class)
24846	HIGHLAND PLASTICS INC., 525 N.SECOND STREET, PO BOX 0099, SHEPHERD, MI, 48883-0099	US Mail (1st Class)
24846	HIGHPOINT SALES, 643 YOUNG ROAD, ERIE, PA, 16509	US Mail (1st Class)
24846	HI-LITE MAINTENANCE, INC., 10739 KINSMAN RD., PO BOX 345, NEWBURY, OH, 44065	US Mail (1st Class)
24846	HILL HEATING & COOLING, 2816 BULLOCK RD., MEXAMORA, MI, 48455	US Mail (1st Class)
24846	HILL, BARTH & KING LLC, PARK PLACE SOUTH, 155 SOUTH PARK AVENUE, WARREN, OH, 44481- 9974	US Mail (1st Class)
24846	HILLSDALE HIGH SCHOOL, 485 TWP. ROAD 1902, JEROMESVILLE, OH, 44840	US Mail (1st Class)
24846	HISA INTERNATIONAL, 2931 CENTRAL AVE, PMB#136, EL PASO, TX, 79905	US Mail (1st Class)
24846	HITACHI CREDIT AMERICA CORP., 9625 PRESERVE PL, DAYTON, OH, 45458-3934	US Mail (1st Class)
24846	HI-TECH COATINGS, INC., 24660 INDUSTRIAL HIGHWAY, WARREN, MI, 48089	US Mail (1st Class)
24846	HI-TECH MOLD & ENGINEERING INC, 2775 COMMERCE DR, ROCHESTER HILLS, MI, 48309-3815	US Mail (1st Class)
24846	HITEK ENGINEERING INC., 8914 STATE ROUTE 60, WAKEMAN, OH, 44889	US Mail (1st Class)
24846	HI-WATT INC., 5068 PAYSPHERE CIRCLE, CHICAGO, IL, 60674	US Mail (1st Class)
24846	HI-WAY TRANSPORT, PO BOX 71133, TUSCALOOSA, AL, 35407	US Mail (1st Class)
24846	HLT, INC., 12460 PLAZA DR, CLEVELAND, OH, 44130-1057	US Mail (1st Class)
24846	HM HEALTH SERVICES SJ, PO BOX 901887, CLEVELAND, OH, 44190-1887	US Mail (1st Class)
24846	HMC TECHNOLOGIES, 1112 WESSON TATE DR, NEW ALBANY, MS, 38652-2006	US Mail (1st Class)
24846	HOCKING COLLEGE, ATTN: CASHIER'S OFFICE, 3301 HOCKING PKW, NELSONVILLE, OH, 45764- 9704	US Mail (1st Class)
24846	HOECHST CELANESE CORP., PO BOX 96332, CHICAGO, IL, 60693	US Mail (1st Class)
24846	HOEHN PLASTICS INC., HWY 165 N CR 925 S, POSEYVILLE, IN, 47633	US Mail (1st Class)
24846	HOFF ENGINEERING CO., 764 FOX RIVER DR., BLOOMFIELD HILLS, MI, 48304	US Mail (1st Class)
24846	HOFFMAN RIGGERS, INC., 1221 WALNUT STREET, ERIE, PA, 16501	US Mail (1st Class)
24846	HOFMAN'S SHOE STORE, 324 W NEPESSING STREET, LAPEER, MI, 48446	US Mail (1st Class)
24846	HOHMAN PLATING & MFG. INC., 814 HILLROSE AVE., DAYTON, OH, 45404-1132	US Mail (1st Class)
24846	HOHNER CORPORATION, 5536 REGIONAL ROAD 81, BEAMSVILLE, ON, LOR 1B3CANADA	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	HOLCKER HARDWARE, 320 SELTZER ST, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	HOLIDAY INN-MANSFIELD, C/O ALLIANCE HOSP. INC., 600 ENTERPRISE DR., LEWIS CENTER, OH, 43035-9432	US Mail (1st Class)
24846	HOLLIE J. VANHOOSE, 2660 COPPER SMITH AVENUE, DAYTON, OH, 45414	US Mail (1st Class)
24846	HOLLO PLASTICS EQUIPMENT INC., 115 EAST AURORA RD, SUITE 102, NORTHFIELD, OH, 44067	US Mail (1st Class)
24846	HOME ACCENTS, 2300 MCFARLAND BLVD, NORTHPORT, AL, 35476	US Mail (1st Class)
24846	HOME DEPOT, 1500 SUMMIT, LAPEER, MI, 48446	US Mail (1st Class)
24846	HOMER PATTERSON, 1401 INDUSTRIAL PK DR, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	HOMETOWN HEALTH NETWORK, 100 LILLIAN GISH BLVD, MASSILON, OH, 44647	US Mail (1st Class)
24846	HONDA OF AMERICA MFG., ATTN:KAREN BRANDEL-SUPPLIER SUPPORT, 24000 HONDA PARKWAY, MARYSVILLE, OH, 43040-9251	US Mail (1st Class)
24846	HONEYWELL INTERNATIONAL INC., 101 COLUMBIA RD, MORRISTOWN, NJ, 07960-4640	US Mail (1st Class)
24846	HOPE GLOBAL, 50 MARTIN ST, CUMBERLAND, RI, 02864	US Mail (1st Class)
24846	HORAUF & KOHLER GMBH, PETER-DORFLER-STR.27, POSTFACH 220140, AUGSBURG, 86181GERMANY	US Mail (1st Class)
24846	HORNING BUILDERS SUPPLY, P.O. BOX 21, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	HORST WITTE GERATEBAU BARSKAMP, HORNDORFER WEG 26-28, BLECKEDE, 21354GERMANY	US Mail (1st Class)
24846	HOSFORD INTERNATIONAL,INC, 3939 W RIDGE RD, SUITE A-120, ERIE, PA, 16506	US Mail (1st Class)
24846	HOT MELT TECHNOLOGIES,INC, 1723 W. HAMLIN RD, PO BOX 80067, ROCHESTER, MI, 48309	US Mail (1st Class)
24846	HOTSET CORP, 1045 HARTS LAKE RD, PO BOX 2404, BATTLE CREEK, MI, 49016	US Mail (1st Class)
24846	HOVAIR, 6912 S 220TH STREET, KENT, WA, 98032	US Mail (1st Class)
24846	HOWARD KELLEY, 17187 N LAUREL PK DR, SUITE #208, LIVONIA, MI, 48152	US Mail (1st Class)
24846	HR DIRECT, GRAYARC-HR DIRECT, %K NELSON TAYLOR CO, PO BOX 3669, MANKATO, MN, 56002-3669	US Mail (1st Class)
24846	HR ONE, 720 INTERNATIONAL PKWY, SUNRISE, FL, 33325-6219	US Mail (1st Class)
24846	HRONE.COM, 720 INTERNATIONAL PKWY, SUNRISE, FL, 33325-6219	US Mail (1st Class)
24846	HRS HOT RUNNER SYSTEM, NA, 4570 RHODES DRIVE, UNIT 100, WINDSOR, ON, N8W 5C2CANADA	US Mail (1st Class)
24846	HS TECHNOLOGIES, 0-159 FIRST COURT, GRAND RAPIDS, MI, 49544-3357	US Mail (1st Class)
24846	HTG RIMES LOGISTIC SERVICES, 1101 E 55TH ST, CLEVELAND, OH, 44103-1026	US Mail (1st Class)
24846	HUDSON BASEBALL ASSOCIATION, P.O. BOX 2219, HUDSON, OH, 44236	US Mail (1st Class)
24846	HUGH QUILL, TREASURER, MONTGOMERY COUNTY, 451 W. THIRD ST., DAYTON, OH, 45422-0476	US Mail (1st Class)
24846	HULITT BOYD, 149 PIONEER PLACE, AIKEN, SC, 29801	US Mail (1st Class)
24846	HUMBERT SANITARY SERVICE, 1581 APPLEGROVE ROAD NW, PO BOX 2126, NORTH CANTON, OH, 44720	US Mail (1st Class)
24846	HUME SALES LTD, 1000 SUITE H, TAYLOR STATION RD., GAHANNA, OH, 43230	US Mail (1st Class)
24846	HUMILITY OF MARY FAMILY MEDIC, PO BOX 71 1871, COLUMBUS, OH, 43271-1871	US Mail (1st Class)
24846	HUNKAR LABORATORIES, INC., 7007 VALLEY AVE, CINCINNATI, OH, 45244	US Mail (1st Class)
24846	HURON PLUMBING, 3956 DUNNING ROAD, ROCHESTER HILLS, MI, 48309	US Mail (1st Class)
24846	HURON TECHNOLOGIES INC., 415 INDUSTRIAL DRIVE, PO BOX 537, LESLIE, MI, 49251	US Mail (1st Class)
24846	HUSKY INJECTION MOLDING, 55 AMHERST VILLA RD, CHEEKTOWAGA, NY, 14225	US Mail (1st Class)
24846	HUSKY PARTS DIVISION, 55 AMHERST VILLA RD, CHEEKTOWAGA, NY, 14225	US Mail (1st Class)
24846	HUSSAIN J. AL-THABHAWI, PO BOX 1161, ERIE, PA, 16512	US Mail (1st Class)
24846	HUSSMAN REFRIGERATION ENGINEERING, 2089 EASY STREET, WALLED LAKE, MI, 48390	US Mail (1st Class)
24846	HUTCHINSON MOVING, 716 E 2ND STREET, TIFTON, GA, 31794	US Mail (1st Class)
24846	HYCOA, 200 WILLIAMS DRIVE, ZELIENOPLE, PA, 16063	US Mail (1st Class)
24846	HYDRADYNE HYDRAULICS, PO BOX 760, HARVEY, LA, 70059-0760	US Mail (1st Class)
24846	HYDRA-POWER SYSTEMS INC., MAIL STOP 95, PO BOX 5087, PORTLAND, OR, 97208	US Mail (1st Class)
24846	HYDRAULIC FITTINGS CO.INC, 5606 TULIP ST, PHILADELPHIA, PA, 19124	US Mail (1st Class)
24846	HYDRECLAIM, DEPT. 59904, MILWAUKEE, WI, 53259-0904	US Mail (1st Class)
24846	HYDRO MECHANICAL, PO BOX 5965, SAGINAW, MI, 48603-0965	US Mail (1st Class)
24846	HYDROTECH, CCG CANADA, 135 S LASALLE DPT 2646, CHICAGO, IL, 60674-2646	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	HYMAN COHEN CO., LPA, COUNSELOR AT LAW, PO BOX 22360, CLEVELAND, OH, 44122	US Mail (1st Class)
24846	HY-TEK MATERIAL HANDLING INC., 2222 PORT RD, COLUMBUS, OH, 43217-1130	US Mail (1st Class)
24846	I/O CONNECTIONS INC, 175 E HARRIS ST, OGDEN, UT, 84401	US Mail (1st Class)
24846	IBERIA MACHINE SHOP, 8402 COUNTY ROAD 30, PO BOX 205, IBERIA, OH, 43325	US Mail (1st Class)
24846	IBM, 91222 COLLECTIONS CTR DR, CHICAGO, IL, 60693	US Mail (1st Class)
24846	IBM CORP., 1 NEW ORCHARD RD, ARMONK, NY, 10504-1722	US Mail (1st Class)
24846	IBM CORPORATION, 91222 COLLECTIONS CTR. DR., CHICAGO, IL, 60693-1222	US Mail (1st Class)
24846	ICI DULUX PAINTS, 21033 NETWORK PLACE, CHICAGO, IL, 60673-1210	US Mail (1st Class)
24846	ID AUTOMATION.COM, 550 N REO ST STE 230, TAMPA, FL, 33609-1062	US Mail (1st Class)
24846	IDEA ART, INC., 2434 ATRIUM WAY, NASHVILLE, TN, 37214-5102	US Mail (1st Class)
24846	IDEAL STITCHER COMPANY, DIV. OF W.R. PABICH MFG. CO., 2323 NORTH KNOX AVE., CHICAGO, IL, 60639-3484	US Mail (1st Class)
24846	IDEMITSU CHEMICALS, 2000 TOWN CENTER, SUITE 1450, SOUTHFIELD, MI, 48075	US Mail (1st Class)
24846	IDG USA LLC-MIDWEST DIV., 950 E PACES FERRY RD NE STE 1575, ATLANTA, GA, 30326-1383	US Mail (1st Class)
24846	IFM EFECTOR, INC., 782 SPRINGDALE DR, EXTON, PA, 19341-2850	US Mail (1st Class)
24846	IHS, 15 INVERNESS WAY EAST, ENGLEWOOD, CO, 80112-5704	US Mail (1st Class)
24846	IKON OFFICE SOLUTIONS, GREAT LAKES WEST DISTRICT, PO BOX 802558, CHICAGO, IL, 60680-2558	US Mail (1st Class)
24846	IKON OFFICE SOLUTIONS, GREAT LAKES DISTRICT, PO BOX 802558, CHICAGO, IL, 60680-2558	US Mail (1st Class)
24846	ILT TOYOTA-LIFT, 5667 E SCHAAF RD, CLEVELAND, OH, 44131-1305	US Mail (1st Class)
24846	IMADA INC., 3100 DUNDEE RD., SUITE 707, NORTHBROOK, IL, 60062	US Mail (1st Class)
24846	IMERYS PIGMENTS & ADDITIVES GP, 100 MANFELL CT E, SUITE 300, ROSWELL, GA, 30076	US Mail (1st Class)
24846	IMH PRODUCTS, INC., 1121 E 46TH ST, INDIANAPOLIS, IN, 46205-2016	US Mail (1st Class)
24846	IMOGENE SMITH, 12953 GOLD DRIVE, NORTHPORT, AL, 35473	US Mail (1st Class)
24846	IMPACT EMPLOYMENT SOLUTIONS, INC., 312 RADER RD., MCCOMB, OH, 45858	US Mail (1st Class)
24846	IMPACT NETWORK SOLUTIONS, INC., IMPACT LONG DISTANCE, PO BOX 76521, CLEVELAND, OH, 44101-6500	US Mail (1st Class)
24846	IMPACT TRAINING SERVICES, INC., 149 CHAPLIN RD, BLOOMFIELD, KY, 40008	US Mail (1st Class)
24846	IMPERIAL INDUSTRIES, INC., 505 INDUSTRIAL PARK AVENUE, MOSINEE, WI, 54455-7917	US Mail (1st Class)
24846	IMPEX GLS, INC, 17244 S. MAIN STREET, GARDENA, CA, 90248	US Mail (1st Class)
24846	IMPROVED PARTS & SERVICE, 5430 W 70TH PL, CHICAGO, IL, 60638-6321	US Mail (1st Class)
24846	IMS CO./THRIFT PRODUCTS, 10373 STAFFORD RD, CHAGRIN FALLS, OH, 44023	US Mail (1st Class)
24846	IMS SERVICE DIVISION, PO BOX 58094, CINCINNATI, OH, 45258	US Mail (1st Class)
24846	IN PACK, CALLE 10 MAYO # 105, CHIHUAHUA, MEXICO	US Mail (1st Class)
24846	INCAT SYSTEMS, INC., DEPT. 123301, PO BOX 67000, DETROIT, MI, 48267-1233	US Mail (1st Class)
24846	INCOE CORPORATION, 1740 EAST MAPLE ROAD, PO BOX 485, TROY, MI, 48099-0485	US Mail (1st Class)
24846	INCOME TAX DEPT, 100 N SELTZER ST, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	INCONTROL, INC., 117 HILLWOOD CIRCLE, NEWNAN, GA, 30263	US Mail (1st Class)
24846	INDEPENDENT CONSULTANT GROUP, LLC, 8829 N. CONGRESS #918, KANSAS CITY, MO, 64153	US Mail (1st Class)
24846	INDEPENDENT PLASTIC, 6611 PETROPARK DR, HOUSTON, TX, 77041-4924	US Mail (1st Class)
24846	INDIANA UNIVERSITY EAST, 2325 CHESTER, RICHMOND, IN, 47374	US Mail (1st Class)
24846	INDUSTRIAL APPLICATION SERVICES INC, 13453 WOODBRIER LANE, GRAND RAPIDS, OH, 43522	US Mail (1st Class)
24846	INDUSTRIAL CABLE & SLING, 5740 FEDERAL STREET, DETROIT, MI, 48209	US Mail (1st Class)
24846	INDUSTRIAL CLEANING, PO BOX 535, LAPEER, MI, 48446	US Mail (1st Class)
24846	INDUSTRIAL CONTROLS, DISTRIBUTORS LLC, PO BOX 827058, PHILADELPHIA, PA, 19182-7058	US Mail (1st Class)
24846	INDUSTRIAL ELECTRIC REWINDING CO., 730 E LIBERTY ST, SUMTER, SC, 29153	US Mail (1st Class)
24846	INDUSTRIAL ELECTRICAL, LOCK BOX 740995, ATLANTA, GA, 30374-0995	US Mail (1st Class)
24846	INDUSTRIAL ELECTRICAL SERVICES, 6065 PERIMETER PARKWAY, MONTGOMERY, AL, 36116	US Mail (1st Class)
24846	INDUSTRIAL ENERGY RESOURCES, 5828 LAPASEO ST, HOUSTON, TX, 77087	US Mail (1st Class)
24846	INDUSTRIAL EQUIPMENT SALES, 372 JENKS CIRCLE, CORONA, CA, 92880-1742	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	INDUSTRIAL HEARING SOL, 2533 MAGNOLIA PLACE, BIRMINGHAM, AL, 35242	US Mail (1st Class)
24846	INDUSTRIAL LAWN MAINTENANCE, MICKEY JONES, PO BOX 253, NORTHPORT, AL, 35476	US Mail (1st Class)
24846	INDUSTRIAL LOSS CONTROL, 613 PELHAM RD. SOUTH, SUITE 1, JACKSONVILLE, AL, 36265	US Mail (1st Class)
24846	INDUSTRIAL MAINTENANCE INSTALLATION, 6830 HIGHLAND ROAD, WHITE LAKE, MI, 48383	US Mail (1st Class)
24846	INDUSTRIAL OPPORTUNITIES, PO BOX 12541, FT. WAYNE, IN, 46863	US Mail (1st Class)
24846	INDUSTRIAL PNEUMATIC & SU, 1715 BROOKPARK ROAD, CLEVELAND, OH, 44109	US Mail (1st Class)
24846	INDUSTRIAL REPAIR SERVICE, INC., 2650 BUSINESS DRIVE, CUMMING, GA, 30040	US Mail (1st Class)
24846	INDUSTRIAL SORTING SVS. INC., 109 B SEVENTY-SIX BLVD., BEREA, KY, 40403	US Mail (1st Class)
24846	INDUSTRIAL SUPPLY, 1439 PETTYJOHN RD, SHANNON, AL, 35142	US Mail (1st Class)
24846	INDUSTRIAL TEMP.CONTROL, 23419 FORD RD., DEARBORN, MI, 48128	US Mail (1st Class)
24846	INDUSTRIAL TIRE SERVICE, PO BOX 102003, IRONDALE, AL, 35210	US Mail (1st Class)
24846	INDUSTRIAL TOOL CO., INC., BOX 20289, CANTON, OH, 44701	US Mail (1st Class)
24846	INDUSTRIAL TUBE CORP., 293 VALLEY ROAD, HILLSBOROUGH, NJ, 08844	US Mail (1st Class)
24846	INDUSTRIAL WASTE CONTROL, INC., 240 SINTER COURT, YOUNGSTOWN, OH, 44510	US Mail (1st Class)
24846	INDUSTRIAL WATER SERVICE, 4500 TURF ROAD, EL PASO, TX, 79938	US Mail (1st Class)
24846	INFORM SOLUTIONS, 1815 LOVE ROAD, GRAND ISLAND, NY, 14072	US Mail (1st Class)
24846	INFORMATION HANDLING SERVICES, PO BOX 34960, SEATTLE, WA, 98124-1960	US Mail (1st Class)
24846	INFRARED SERVICES, 5730 FALLS DRIVE, SUITE 100, FORT WAYNE, IN, 46804	US Mail (1st Class)
24846	INGENIERIA DE EMPAQUE, AVE UNIVERSIDAD # 2305, CHIHUAHUA, MEXICO	US Mail (1st Class)
24846	INGERSOLL-RAND CO., 13551 MERRIMAN RD, LIVONIA, MI, 48150	US Mail (1st Class)
24846	INNOVASYS, LLC, 35520 MOUND ROAD, STERLING HEIGHTS, MI, 48310	US Mail (1st Class)
24846	INNOVATIVE DESIGN (ENL), 13208 ARTIC CIRLCE, SANTA FE SPRINGS, CA, 90670	US Mail (1st Class)
24846	INNOVATIVE TRANSPORTATION SERV, 12201 TECUMSEH RD EAST, WINDSOR, ON, N8N 1M3CANADA	US Mail (1st Class)
24846	INNOVATIVE WATER TECHNOLOGIES, 3010 SHAFFER, SE, STE. 1, KENTWOOD, MI, 49512	US Mail (1st Class)
24846	INNOVENE USA LLC, 13536 COLLECTIONS CENTER DR., CHICAGO, IL, 60693-0001	US Mail (1st Class)
24846	INNOVENE USA LLC, 13536 COLLECTIONS CENTER DR., CHICAGO, IL, 60693	US Mail (1st Class)
24846	INOVENT ENGINEERING, PO BOX 39541, 5905 B HARPER, SOLON, OH, 44139	US Mail (1st Class)
24846	INPRAX PERFORMANCE RESOURCES, 5345 OLD DIXIE HWY, FOREST PARK, GA, 30297	US Mail (1st Class)
24846	INSIDE VERSION 5, PO BOX 9405, GAITHERSBURG, MD, 20897-9824	US Mail (1st Class)
24846	INSIGHT, 6820 S HARL AVE, TEMPE, AZ, 85283	US Mail (1st Class)
24846	INSPECTION ENGINEERING, 29313 CLEMENS ROAD, SUITE 2A, CLEVELAND, OH, 44145	US Mail (1st Class)
24846	INSTITUTE OF BUSINESS PUBLICATIONS, 626 THORNCROFT DR, WEST CHESTER, PA, 19380-6442	US Mail (1st Class)
24846	INSTITUTE OF INDUSTRIAL ENGR., 25 TECHNOLOGY PARK, PO BOX 740018, ATLANTA, GA, 30371-9941	US Mail (1st Class)
24846	INSTITUTE OF MANAGEMENT ACCOUNTANTS, 10 PARAGON DRIVE, MONTVALE, NJ, 07645-1760	US Mail (1st Class)
24846	INSTITUTE OF MGMT. ACCOUNTANTS, 10 PARAGON DR, MONTVALE, NJ, 07645-1773	US Mail (1st Class)
24846	INSTRON CORPORATION, 825 UNIVERSITY AVE, NORWOOD, MA, 02062-2643	US Mail (1st Class)
24846	INSTRUCON INC., 3415 PRECISION DRIVE, ROCKFORD, IL, 61009	US Mail (1st Class)
24846	INSTRUCON, INC, 3415 PRECISION DRIVE, ROCKFORDE, IL, 61009	US Mail (1st Class)
24846	INSTRUMENT SERV. & EQUIP., 10100 ROYALTON RD., CLEVELAND, OH, 44133	US Mail (1st Class)
24846	INSTRUMENT SERV. & EQUIP., INC, 417 TOMAHAWK DRIVE, MAUMEE, OH, 43537	US Mail (1st Class)
24846	INTEC TST, 1700 BUSHA HWY, MARYSVILLE, MI, 48040 INTEGRA, 3345 COLLECTIONS CENTER, CHICAGO, IL, 60693	US Mail (1st Class) US Mail (1st Class)
24846		US Mail (1st Class)
24846 24846	INTEGRATED CONTROL, 28 BRIDGE AVE, SCITUATE, MA, 02066-4331 INTEGRATED LOGISTIC SOL./ARDEN, 23000 EUCLID AVE, CLEVELAND, OH, 44117	US Mail (1st Class)
24846	INTEGRATED LOGISTIC SOL./ARDEN, 23000 EUCLID AVE, CLEVELAND, OH, 44117  INTELLIGENT MAILING SOLUTIONS, 10200 BRECKSVILLE ROAD, BRECKSVILLE, OH, 44141-3208	US Mail (1st Class)
24846	INTERMEC, DEPT. CH 10696, PALATINE, IL, 60055-0696	US Mail (1st Class)
24846	INTERMEC TECHNOLOGIES CORP., 22095 NETWORK PLACE, CHICAGO, IL, 60673-1220	US Mail (1st Class)
24846	INTERNAL REVENUE SERVICE, ADDRESS UNAVAILABLE,	US Mail (1st Class)
<u> </u>	INTERNAL REVEROE OFFICE, ADDITEOU DIVAVAIDADEE,	50 man (15t 01055)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	INTERNAL REVENUE SERVICE, IRS, CINCINNATI, OH, 45999-0046	US Mail (1st Class)
24846	INTERNATIONAL ASSOCIATION OF, ADMINISTRATIVE PROFESSIONALS, PO BOX 413740, KANSAS CITY, MO, 64141-3740	US Mail (1st Class)
24846	INTERNATIONAL BUSINESS, DIRECTORIES, 10200 N.W. 25TH STREET, MIAMI, FL, 33172	US Mail (1st Class)
24846	INTERNATIONAL CONSULTING GROUP, 1551 WINTER SPRINGS BLVD., WINTER SPRINGS, FL, 32708	US Mail (1st Class)
24846	INTERNATIONAL HEALTH, AWARENESS CENTER, INC., PO BOX 33881, DETROIT, MI, 48232-8128	US Mail (1st Class)
24846	INTERNATIONAL MOLD STEEL, 6796 POWERLINE DR, FLORENCE, KY, 41042	US Mail (1st Class)
24846	INTERNATIONAL OUTSOURCING, 1600 W. BLOOMFIELD RD., BLOOMINGTON, IN, 47403	US Mail (1st Class)
24846	INTERNATIONAL SPORTS, MANAGEMENT, INC., 700 WEST VIRGINIA STREET, MILWAUKEE, WI, 53204	US Mail (1st Class)
24846	INTERNATIONAL TECHNE GRP, LOCATION 001, 5303 DUPONT CIRCLE, MILFORD, OH, 45150	US Mail (1st Class)
24846	INTERNETWORK, ENGINEERING TECHNOLOGIES, 8117 KINMORE ST., DEARBORN HEIGHTS, MI, 48127	US Mail (1st Class)
24846	INTERSTATE CAPITAL CORP., 1255 COUNTRY CLUB RD STE D, SANTA TERESA, NM, 88008-9707	US Mail (1st Class)
24846	INTERSTATE LIFT TRUCKS, INC., 5667 SCHAAF RD ,, CLEVELAND, OH, 44131-1305	US Mail (1st Class)
24846	INTERSTATE OPTICAL CO., 680 LINDAIRE LANE, PO BOX 308, MANSFIELD, OH, 44901	US Mail (1st Class)
24846	INTERSTATE PUMP COMPANY, INC., 33370 WINONA RD, SALEM, OH, 44460-9011	US Mail (1st Class)
24846	INTERSTATE QUALITY CONSULTANTS, 137 NORTH MAIN STREET, SUITE 600, DAYTON, OH, 45402	US Mail (1st Class)
24846	INTERTEK ETL ENTELA, DEPARTMENT #251601, PO BOX 67000, DETROIT, MI, 48267-2516	US Mail (1st Class)
24846	INTERTEX WORLD RESOURCES, LTD., 4000 EMBASSY PARKWAY, SUITE 205, AKRON, OH, 44333	US Mail (1st Class)
24846	IONIC TECHNOLOGIES INC., 207 FAIRFOREST WAY, GREENVILLE, SC, 29607	US Mail (1st Class)
24846	IPM, INC., 5252 SOUTHWAY S.W., CANTON, OH, 44706	US Mail (1st Class)
24846	IPSWITCH INC., 10 MAGUIRE RD STE 220, LEXINGTON, MA, 02421-3120	US Mail (1st Class)
24846	IRC, 4222 S STAPLES ST, CORPUS CHRISTI, TX, 78411	US Mail (1st Class)
24846	IRENE LONG, 161 KELSO DR APT 1, ERIE, PA, 16505-1982	US Mail (1st Class)
24846	IRISH ROSE FLORIST, 33608 W. SEVEN MILE, LIVONIA, MI, 48152	US Mail (1st Class)
24846	IRMA CAROL ZEPEDA FLORES, AV. JOSEFA ORTIZ, DE DOMINGUEZ 68-2, QUERETARO, 76900MEXICO	US Mail (1st Class)
24846	IRON MOUNTAIN OFF-SITE, DATA PROTECTION, PO BOX 27129, NEW YORK, NY, 10087-7129	US Mail (1st Class)
24846	IRON WORKS, 5514 CHARLESTON RD, NEWTON FALLS, OH, 44444	US Mail (1st Class)
24846	ISBELL INSULATION, 2519 28TH STREET, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	ISOCAL LABORATORIES, INC., 16 TOBEY ROAD, BLOOMFIELD, CT, 06002	US Mail (1st Class)
24846	ITW DELPRO, 75 REMITTANCE DR., SUITE 1631, CHICAGO, IL, 60675-1631	US Mail (1st Class)
24846	ITW DELTAR, 21555 S HARLEM AVE, FRANKFORT, IL, 60423	US Mail (1st Class)
24846	ITW DELTAR CANADA, 80 SANTE DRIVE, CONCORD, ON, L4K 3C4CANADA	US Mail (1st Class)
24846	ITW DELTAR TEKFAST, 1700 1ST AVE, CHIPPEWA FALLS, WI, 54729-1417	US Mail (1st Class)
24846	ITW DELTAR-ENGINEERED FASTENER, 1700 1ST AVE, CHIPPEWA FALLS, WI, 54729	US Mail (1st Class)
24846	ITW MEDALIST, 2700 YORK ROAD, ELK GROVE VILLAGE, IL, 60007	US Mail (1st Class)
24846	ITW SHAKEPROOF AUTO., 850 STEPHENSON HWY, SUITE 500, TROY, MI, 48083	US Mail (1st Class)
24846	ITW SUPERB PRODUCTS, 75 REMITTANCE DRIVE, SUITE 1438, CHICAGO, IL, 60675-1438	US Mail (1st Class)
24846	IVAN HARRIS, 168 BRECKENRIDGE DR., N. AUGUSTA, SC, 29841	US Mail (1st Class)
24846	IWATA BOLT USA INC., 7131 ORANGEWOOD AVE., GARDEN GROVE, CA, 92841-1409	US Mail (1st Class)
24846	J & D SURPLUS INC., DBA SURPLUS JUNK STORE, 2126 COLUMBUS ROAD NE, CANTON, OH, 44705	US Mail (1st Class)
24846	J & H MACHINE TOOLS, 4345 MORRIS PARK DR, CHARLOTTE, NC, 28227	US Mail (1st Class)
24846	J & J JANITORIAL, 11420 ASHLEY AVE., NORTHPORT, AL, 35475	US Mail (1st Class)
24846	J & J TELEPHONE SERVICE, 3925 ANDREWS ST. EAST, TUSCALOOSA, AL, 35405	US Mail (1st Class)
24846	J & L SIGNS, 512 HARGROVE ROAD EAST, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	J & M WELDING, 2640 E. WATER, ROCK CREEK, OH, 44084	US Mail (1st Class)
24846	J I T RESHIPPABLES, 1250 PAGE RD., PO BOX 218, AURORA, OH, 44202	US Mail (1st Class)
24846	J&K RENAISSANCE CONSULTING LLC, 21951 LYON TRAIL N., SOUTH LYON, MI, 48178	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	J&L ELECTRIC, INC., 300 CUSTER DRIVE, BATTLE CREEK, MI, 49017	US Mail (1st Class)
24846	J.B.& CO., INC. ROOFING, 1480 S. CO. RD. 594, PO BOX 520, TIFFIN, OH, 44883	US Mail (1st Class)
24846	J.F. GOOD CO., 109 N UNION ST, AKRON, OH, 44304	US Mail (1st Class)
24846	J.F. GOOD CO., 905 MIAMI ST, AKRON, OH, 44311-2363	US Mail (1st Class)
24846	J.L. CLEMSON, INC., 3435 S MAIN ST, ROCK CREEK, OH, 44084	US Mail (1st Class)
24846	J.P. PAINTING, 16107 LAKEVIEW DR., BURTON, OH, 44021	US Mail (1st Class)
24846	J.R. CAMPBELL EQUIPMENT CO., 3659 PARKWAY LANE, HILLIARD, OH, 43026	US Mail (1st Class)
24846	JA MEYER LLC, 11221 SPRINGVILLE HWY, ONSTED, MI, 49265	US Mail (1st Class)
24846	JAC BELLS REALTY COMPANY, 2172 WEST 32ND STREET, ERIE, PA, 16508	US Mail (1st Class)
24846	JACK BREWER, 119 EAST DAYTON STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	JACK GRIFFIN, 7110 ALLEGAN DR., DAVISON, MI, 48423	US Mail (1st Class)
24846	JACK'S HEATING & COOL, 587 S COURT ST, LAPEER, MI, 48446-2552	US Mail (1st Class)
24846	JACK'S WELDING SERVICE, 260 LUCKNOW RD., BISHOPVILLE, SC, 29010	US Mail (1st Class)
24846	JACOB GRANT, 24040 ROSEWOOD, OAK PARK, MI, 48237	US Mail (1st Class)
24846	JACOB OGLESBY, 21227 LAKEVIEW DR., MCCALLA, AL, 35111	US Mail (1st Class)
24846	JACOB PRINTING CO., 34101 JEFFERSON AVE., ST. CLAIR SHORES, MI, 48082	US Mail (1st Class)
24846	JAKE WEAVER, 15078 GEORGIA RD., MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	JAMCO PRODUCTS, INC., ONE JAMCO CT., PO BOX 66, S. BELOIT, IL, 61080	US Mail (1st Class)
24846	JAMES BURNS, 14356 SHILOH ROAD, RALPH, AL, 35480	US Mail (1st Class)
24846	JAMES C. SMITH CONSTRUCTION, 123 AUBURN DRIVE, BOARDMAN, OH, 44512	US Mail (1st Class)
24846	JAMES DRYDEN, 1401 INDUSTRIAL PK DR, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	JAMES E. RIMMEL CO., LPA, 44 N BROAD ST, CANFIELD, OH, 44406-1171	US Mail (1st Class)
24846	JAMES GEARHART, 1348 OX YOKE DR., FLINT, MI, 48532	US Mail (1st Class)
24846	JAMES HUBBARD, 5001 MADISON ROAD, APT 2307, NORTHPORT, AL, 35475	US Mail (1st Class)
24846	JAMES LEONELLO, 2542 SAMIRA ROAD, STOW, OH, 44224	US Mail (1st Class)
24846	JAMES MCKINNON, 8707 SAMUEL BARTON DR., BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	JAMES R. MCCAULEY, 3729 SHANABRUCK NW, CANTON, OH, 44705	US Mail (1st Class)
24846	JAMES W. MCKIAN, 1265 NORRIS ROAD, MIDDLEVILLE, MI, 49333	US Mail (1st Class)
24846	JAMES W. WOOLEY, 8420 HIGHWAY 82 EAST, DUNCANVILLE, AL, 35456	US Mail (1st Class)
24846	JAMES WRIGHTSMAN, 1000 EVERETT-HULL RD, CORTLAND, OH, 44410	US Mail (1st Class)
24846	JAMIE WELCH & JIM FRITZ, C/O MUNZ-PIRNSTILL FUNERAL HOME, 215 N WALNUT ST., BUCYRUS, OH, 44820	US Mail (1st Class)
24846	JANET MERRIMAN, 1401 INDUSTRIAL PK DR, TUSCALOOSA, AL, 35402	US Mail (1st Class)
24846	JANICE A. BURANDT, 4400 MASSILLON ROAD, N. CANTON, OH, 44720	US Mail (1st Class)
24846	JANICE A. BURANDT, 3131 COLUMBUS ROAD NE, CANTON, OH, 44705	US Mail (1st Class)
24846	JANICE K WHITE, 3131 COLUMBUS RD NE, CANTON, OH, 44705	US Mail (1st Class)
24846	JAS STEEL, 5645 IMLAY CITY RD., ATTICA, MI, 48412	US Mail (1st Class)
24846	JASAN CARPET, HIGHWAY 82, REFORM, AL, 35481	US Mail (1st Class)
24846	JASON SLACKTA, 6214 POTTER, BURTON, MI, 48509	US Mail (1st Class)
24846	JASON STALKER, 119 EAST DAYTON STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	JASON STILES, 2423 DEWEY ROAD, ASHTABULA, OH, 44004	US Mail (1st Class)
24846	JAY ELECTRIC, 1515 17TH AVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	JAY ENN CORP, 33943 DEQUINDRE RD., TROY, MI, 48084-4692	US Mail (1st Class)
24846	JAY INDUSTRIAL TECHNOLOGIES, GROUP, INC., PO BOX 641341, CINCINNATI, OH, 45264-1341	US Mail (1st Class)
24846	JAY'S SEPTIC TANK SERVICE, 2787 GREENWOOD RD., LAPEER, MI, 48446	US Mail (1st Class)
24846	JBC TECHNOLOGIES, INC., 7887 BLISS PKWY, N RIDGEVILLE, OH, 44039-3475	US Mail (1st Class)
24846	JBE, INC., 512 HARTLAND DR, HARTSVILLE, SC, 29550	US Mail (1st Class)
24846	JC MALONE, DEPT. # 5097, PO BOX 740041, LOUISVILLE, KY, 40201-7441	US Mail (1st Class)
24846	JCI CONTRACTORS, 529 W PROSPECT RD., SUITE 101, ASHTABULA, OH, 44004	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	JEAN MCENTARFER, 119 EAST DAYTON ST., WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	JEANETTE DAVIS, 1401 INDUSTRIAL PARK, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	JEANETTE K. MUNCY, 8707 SAMUEL BARTON DRIVE, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	JEFF DEUNGER, 3131 COLUMBUS RD., CANTON, OH, 44705	US Mail (1st Class)
24846	JEFF LARY, 1401 INDUSTRIAL, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	JEFF LAZARSKI, 4219 ILLINOIS SW, WYOMING, MI, 49509	US Mail (1st Class)
24846	JEFF RAMSELL, 3353 GONDER AVE SE, CANTON, OH, 44707-1530	US Mail (1st Class)
24846	JEFF SIMS, 1401 INDUSTRIAL DRIVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	JEFF SOUTHWOOD, VISTEON CORP., 45000 HELMS STREET, PLYMOUTH, MI, 48170	US Mail (1st Class)
24846	JEFF SZINK, 3131 COLUMBUS RD., CANTON, OH, 44705	US Mail (1st Class)
24846	JEFFCO MACHINE SERVICE, 460 VANESSA DRIVE, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	JEFFREY BRIDGES, 16104 CHARTER OAKS, APT 4, DAVISON, MI, 48423	US Mail (1st Class)
24846	JEFFREY W STACK, C/O CEP PRODUCTS, LLC, PO BOX 320, CRESTLINE, OH, 44827-0320	US Mail (1st Class)
24846	JENNIFER HOLDEN, 100 SEVENTH AVE, SUITE 100, CHARDON, OH, 44024	US Mail (1st Class)
24846	JENNIFER KASTL, 21819 MIDDLEBELT RD., FARMINGTON HILLS, MI, 48336	US Mail (1st Class)
24846	JENNIFER WATSON, C/O GREEN HAINES SGAMBATI CO.,L.P.A, 120 WESTCHESTER DR., STE. A, PO BOX 4338, YOUNGSTOWN, OH, 44515	US Mail (1st Class)
24846	JEREMEY BROWN, 1446 WEST 10TH ST, #1, ASHTABULA, OH, 44004	US Mail (1st Class)
24846	JEREMY GILDNER, 4340 MAPLE VALLEY RD., MARLETTE, MI, 48453	US Mail (1st Class)
24846	JERMAINE HARKINS, 4222 ELM DR NE, TUSCALOOSA, AL, 35404	US Mail (1st Class)
24846	JERRY B. SELLMAN, ESQ., ARBITRATOR & ATTORNEY AT LAW, 88 EAST BROAD STREET, COLUMBUS, OH, 43215	US Mail (1st Class)
24846	JERRY RIPPEL, 3131 COLUMBUS RD., CANTON, OH, 44705	US Mail (1st Class)
24846	JETS PIZZA, SUMMIT DRIVE, LAPEER, MI, 48446	US Mail (1st Class)
24846	JH AUTOMATION, 3295 E. CHARMWOOD DR., PORT HURON, MI, 48060	US Mail (1st Class)
24846	JIFFY PRINTS, INC. OF LAPEER, 1975 W GENESEE STREET, LAPEER, MI, 48446	US Mail (1st Class)
24846	JIM BOYNS, 703 N. WILDER RD., LAPEER, MI, 48446	US Mail (1st Class)
24846	JIM SLOAN, INC., C/O BARSA CONSULTING, LLC, 2900 WESTCHESTER AVENUE, PURCHASE, NY, 10577	US Mail (1st Class)
24846	JIM SUMMERS, 929 ROLLING HILLS LANE/APT. 1, LAPEER, MI, 48446	US Mail (1st Class)
24846	JIM SWARTZ, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	JIM TAYLOR, 8707 SAMUEL BARTON DR., BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	JIMMY NORRIS, 1701 INDUSTRIAL PARK DR., TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	JIMMY TUCKER, 175 FLYNN ROAD, FAYETTE, AL, 35555-6528	US Mail (1st Class)
24846	JIMOLA WILSON, 7445 TIMBERLEA CT., FLINT, MI, 48532	US Mail (1st Class)
24846	JLB & ASSOCIATES, INC., 771 BOSTON POST RD. EAST # 162, MARLBORO, MA, 01752	US Mail (1st Class)
24846	JLC FABRICATING, 1828 N SAGINAW ST, LAPEER, MI, 48446-7741	US Mail (1st Class)
24846	JLS ROOFING SALES, 535 WACO RD, TUSCALOOSA, AL, 35405	US Mail (1st Class)
24846	JLW INSTRUMENTS, INC., 452 N SANGAMON, CHICAGO, IL, 60622	US Mail (1st Class)
24846	JM ASSEMBLY, PO BOX 773, GREENVILLE, OH, 45331	US Mail (1st Class)
24846	JM MOLD SOUTH, 807 SHEFFIELD RD, EASLEY, SC, 29642-3555	US Mail (1st Class)
24846	JM POLYMERS, LLC, 10286 RAMA CT, CLARKSTON, MI, 48348	US Mail (1st Class)
24846	JNO J. DISCH CO., 1616 COUTANT AVE, CLEVELAND, OH, 44107	US Mail (1st Class)
24846	JOANN SMITH, 8211 NORMAN ROAD, IMLAY CITY, MI, 48444	US Mail (1st Class)
24846	JOANNE E. MURRAY, PO BOX 87, 201 E. ORCHARD ST., W. MANCHESTER, OH, 45382-0087	US Mail (1st Class)
24846	JOANNE S. EDINGER, 3823 DAVIDSON AVE, ERIE, PA, 16504	US Mail (1st Class)
24846	JODIE TURNER, 900 S. WILEY ST., PO BOX 320, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	JODY L. HEETER, 662 BORRADALE RD, EATON, OH, 45320-8543	US Mail (1st Class)
24846	JODY MC MILLIAN, 3256 DAVIS LAKE RD, LAPEER, MI, 48446	US Mail (1st Class)
24846	JOE BILLOTTE, 900 S. WILEY ST., PO BOX 320, CRESTLINE, OH, 44827	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	JOE SYKORA, ADDRESS UNAVAILABLE, CHARDON, OH, 44024	US Mail (1st Class)
24846	JOE VIOLAND, 2905 3RD ST. N.W., CANTON, OH, 44708	US Mail (1st Class)
24846	JOEL GREGG, INSIDE OUT PAINTING, 1804 DOVER-ZOAR RD NE, BOLIVAR, OH, 44612	US Mail (1st Class)
24846	JOHN A BECKER CO., 1341 E 4TH ST, DAYTON, OH, 45402	US Mail (1st Class)
24846	JOHN BINDER, 617 WILLOW DRIVE, CONNERSVILLE, IN, 47331	US Mail (1st Class)
24846	JOHN CARPENTER, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	JOHN GALE, 11211 PLYMOUTH AVE, CLEVELAND, OH, 44125-2711	US Mail (1st Class)
24846	JOHN KRAUS, 125 TONAWANDA TRAIL SW, HARTVILLE, OH, 44632	US Mail (1st Class)
24846	JOHN L. BEEVER, 206 S ADAMS ST, MT VERNON, OH, 43050	US Mail (1st Class)
24846	JOHN LEBEAU, 3131 COLUMBUS ROAD NE, CANTON, OH, 44705	US Mail (1st Class)
24846	JOHN MCCULLOUGH, 3131 COLUMBUS ROAD, CANTON, OH, 44705	US Mail (1st Class)
24846	JOHN MCINERNEY, 37785 INGLESIDE DR, CLINTON TWP, MI, 48036	US Mail (1st Class)
24846	JOHN RUE, 417 BROADWAY STREET, PIQUA, OH, 45356	US Mail (1st Class)
24846	JOHN THOMAS FLOOR COVERING, 42177 E ANN ARBOR RD, PLYMOUTH, MI, 48170	US Mail (1st Class)
24846	JOHN TOWNSEND TREE TRIMMING, 3433 KALURAH, CANTON, OH, 44721	US Mail (1st Class)
24846	JOHN W. MILLER, 703 WEST CHURCH STREET, GALION, OH, 44833	US Mail (1st Class)
24846	JOHN W. MILLER, 3708 BLACKBERRY LANE, NORTHPORT, AL, 35473	US Mail (1st Class)
24846	JOHNNIE FRANKLIN, 31572 INDUSTRIAL RD., SUITE 300, LIVONIA, MI, 48150	US Mail (1st Class)
24846	JOHNNY NOBLE, TUSCALOOSA PLANT, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	JOHNNY RAY'S BAR-B-QUE, 100 MCFARLAND BLVD., NORTHPORT, AL, 35476	US Mail (1st Class)
24846	JOHNSON CONTROLS, 9410 BUNSEN PKWY, SUITE 100B, LOUISVILLE, KY, 40220	US Mail (1st Class)
24846	JOHNSON CONTROLS INC., DEPT. 77012, PO BOX 77000, DETROIT, MI, 48277-0012	US Mail (1st Class)
24846	JOHNSON CONTROLS, INC, 120 C LINE, ORANGEVILLE, ON, L9W 2Z5CANADA	US Mail (1st Class)
24846	JOHNSON CONTROLS, INC., DRAWER 242, MILWAUKEE, WI, 53278	US Mail (1st Class)
24846	JOHNSON CONTROLS, INC., 49200 HALYARD DR, PLYMOUTH, MI, 48170-2465	US Mail (1st Class)
24846	JOHNSON CONTROLS, INC., 47912 HALYARD DR., PLYMOUTH, MI, 48170	US Mail (1st Class)
24846	JOHNSON SYSTEM INC., 18999 OLD 27 NORTH, MARSHALL, MI, 49068	US Mail (1st Class)
24846	JOLEENE MOSS, 2474 EUGENE STREET, BURTON, MI, 48519	US Mail (1st Class)
24846	JONATHAN JACKSON, 119 EAST DAYTON STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	JONATHAN WILLIAMS, 8695 NOTTINGHAM, YPSILANTI, MI, 48198	US Mail (1st Class)
24846	JONES PRINTING, 20 LEXINGTON AVE, MANSFIELD, OH, 44907-1217	US Mail (1st Class)
24846	JORDAN POWER & EQUIP. CO., 281 SOUTHWEST AVE., TALLMADGE, OH, 44278	US Mail (1st Class)
24846	JOSEPH A. NIEDENBERGER, 2601A S. RIDGE RD. EAST, ASHTABULA, OH, 44004	US Mail (1st Class)
24846	JOSEPH MALLAK, 3560 W. MARKET ST., STE. 340, AKRON, OH, 44333	US Mail (1st Class)
24846	JOSEPH POOR, 18031 CLARIDON TROY, HIRAM, OH, 44234	US Mail (1st Class)
24846	JOSEPH R.O. SMITH, 15 MYRTLE DRIVE, BISHOPVILLE, SC, 29010	US Mail (1st Class)
24846	JOSEPH RAY MAURINA, 5118 BERNEDA, FLINT, MI, 48506	US Mail (1st Class)
24846	JOSEPH T. RYERSON & SON INC., 33959 TREASURY CENTER, CHICAGO, IL, 60694-3900	US Mail (1st Class)
24846	JOSEPH T. RYERSON & SON, INC., 2621 W 15TH PL, CHICAGO, IL, 60608-1712	US Mail (1st Class)
24846	JOSEPH VIOLAND, 3131 COLUMBUS ROAD NE, CANTON, OH, 44705	US Mail (1st Class)
24846	JSL & ASSOCIATES INC., 3077 MARKLE DR., SILVER LAKE, OH, 44224	US Mail (1st Class)
24846	JUGENHEIMER INDUSTRIAL SUPPLIE, 6863 COMMERCE DR, HUBBARD, OH, 44425	US Mail (1st Class)
24846	JULIE ROSSWURM, 8707 SAMUEL BARTON DRIVE, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	JUNE OLSON, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	JUNIOR ACHIEVEMENT, OF NORTH CENTRAL OHIO, 777 N TRIMBLE RD., MANSFIELD, OH, 44901	US Mail (1st Class)
24846	JUST JOKIN', 602 W. BUCYRUS ST., CRESTLINE, OH, 44827	US Mail (1st Class)
24846	K & H TRUCKING, INC., 5007 BENEFIT AVE, ASHTABULA, OH, 44004-6530	US Mail (1st Class)
24846	K L INDUSTRIES, 5944 PAYSPHERE CIRCLE, CHICAGO, IL, 60674	US Mail (1st Class)
24846	K&B PRODUCTS, 812 CHARLES STREET, GALION, OH, 44833	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	K. WM BEACH MFG., INC., 4655 URBANA RD, SPRINGFIELD, OH, 45502-9503	US Mail (1st Class)
24846	KAHL'S ELECTRIC, 2740 LEXINGTON AVE, MANSFIELD, OH, 44904	US Mail (1st Class)
24846	KANAWHA SCALES, ROCK BRANCH INDUSTRIAL PARK, POCA, WV, 25159	US Mail (1st Class)
24846	KANETEC USA CORP, 124 W. DEVON AVE., BENSENVILLE, IL, 60106-1148	US Mail (1st Class)
24846	KANTUS CORP., MSC 410367, PO BOX 41500, NASHVILLE, TN, 37241-5000	US Mail (1st Class)
24846	KAREN ROMEO, 2630 NETHERLAND DR., #312, BEAVERCREEK, OH, 45431	US Mail (1st Class)
24846	KAREN SHENBERGER, C/O CEP PRODUCTS, LLC, PO BOX 320, CRESTLINE, OH, 44827-0320	US Mail (1st Class)
24846	KARL'S JEWELRY, 15962 E HIGH ST, MIDDLEFIELD, OH, 44062-9474	US Mail (1st Class)
24846	KARR METROLOGY INSTRUMENTS INC, 5725 DRAGON WAY, SUITE 104, CINCINNATI, OH, 45227	US Mail (1st Class)
24846	KASPAR SCALE, INC., 30311 CLEMENS, CLEVELAND, OH, 44145	US Mail (1st Class)
24846	KATHLEEN DANA, 1276 SHERMAN DR., LAKE CITY, PA, 16423	US Mail (1st Class)
24846	KATHLEEN MURAUSKAS, 6714 CONNECTICUT COLONY CIRCLE, MENTOR, OH, 44060	US Mail (1st Class)
24846	KATHRYN B. MITCHELL, 11241 EXMOOR DR, CONCORD, OH, 44077	US Mail (1st Class)
24846	KATHRYN M. SPEARS, 900 S. WILEY ST., PO BOX 320, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	KATHY CLARK, 3538 DAVISON RD., LAPEER, MI, 48446	US Mail (1st Class)
24846	KAUTEX TEXTRON, 25036 NETWORK PLACE, CHICAGO, IL, 60673-1250	US Mail (1st Class)
24846	KAWASAKI ROBOTICS, INC., 28140 LAKEVIEW DR, WIXOM, MI, 48393	US Mail (1st Class)
24846	KAY'S KARPET KARE, (AKA( BEACON CLEANING SERVICES, 3501 WATERMELON RD, NORTHPORT, AL, 35473-5136	US Mail (1st Class)
24846	KBC TOOLS & MACHINERY, 5344 JACKMAN RD, TOLEDO, OH, 43613	US Mail (1st Class)
24846	K-BIN SUBSID.OF SHINTECH, 3 E GREENWAY PLAZA, STE 1150, HOUSTON, TX, 77046	US Mail (1st Class)
24846	KBZ ELECTRIC, INC., 725 LONGVIEW AVE., PO BOX 277, MANSFIELD, OH, 44901	US Mail (1st Class)
24846	KEENER CORPORATION, STATE ROAD ONE NORTH, 419 EAST TENTH STREET, CONNERSVILLE, IN, 47331	US Mail (1st Class)
24846	KEHN KRANES, 20800 N. LIBERTY ROAD, BUTLER, OH, 44822	US Mail (1st Class)
24846	KEITH BUYCK, 8707 SAMUEL BARTON DRIVE, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	KEITH CHAMBERS, 8707 SAMUEL BARTON DRIVE, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	KEITH PANTER, 31572 INDUSTRIAL RD., SUITE 300, LIVONIA, MI, 48150	US Mail (1st Class)
24846	KELCO BUILDERS, 7640 OLDFIELD RD., CRESTLINE, OH, 44827	US Mail (1st Class)
24846	KELLER GRAIN & FEED, INC., 7977 MAIN STREET-HORATIO, GREENVILLE, OH, 45331	US Mail (1st Class)
24846	KELLER RIGGING & CONST, 1247 EASTWOOD AVENUE, PO BOX 357, TALLMADGE, OH, 44278-0357	US Mail (1st Class)
24846	KELLY J. BOYLE, 6823 MONROE, NORTH BRANCH, MI, 48461	US Mail (1st Class)
24846	KELLY SALES AND ENGINEERING, PO BOX 194, GRAND HAVEN, MI, 49417	US Mail (1st Class)
24846	KELLY TEMPORARY SERVICE, 999 W BIG BEAVER RD, TROY, MI, 48084	US Mail (1st Class)
24846	KELM ACUBAR, 1055 NORTH SHORE DRIVE, BENTON HARBOR, MI, 49022	US Mail (1st Class)
24846	KELMAR, INC., 830 NEW YORK AVE, TOLEDO, OH, 43611-3333	US Mail (1st Class)
24846	KEN NOH, 3131 COLUMBUS RD NE, CANTON, OH, 44705	US Mail (1st Class)
24846	KEN PARRISH, 119 EAST DAYTON STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	KENDALE INDUSTRIES INC., 7600 HUB PARKWAY, VALLEY VIEW, OH, 44125	US Mail (1st Class)
24846	KEN-KOAT INC., 801 GARRETT PKWY, LEWISBURG, TN, 37091-3557	US Mail (1st Class)
24846	KENMORE LABEL & TAG INC., 8810 TYLER BLVD, MENTOR, OH, 44060	US Mail (1st Class)
24846	KENNAMETAL, 1600 TECHNOLOGY WAY, LATROBE, PA, 15650	US Mail (1st Class)
24846	KENNEDY EXCAVATING, 4345 N STATE RD, DAVISON, MI, 48423	US Mail (1st Class)
24846	KENNETH BERTRAM, 9044 ONTARIO ST. NW, MASSILLON, OH, 44646	US Mail (1st Class)
24846	KENNETH WASHINGTON, 303 BROWN RD, ETHELSVILLE, AL, 35461	US Mail (1st Class)
24846	KERR PUMP & SUPPLY, DRAWER 64185, DETROIT, MI, 48264	US Mail (1st Class)
24846	KETTERING UNIVERSITY, CORP. RELATIONS/SHARON BARBEE, 1700 WEST THIRD, FLINT, MI, 48504-4898	US Mail (1st Class)
24846	KETTERING WORKERS CARE, 2023 SPRINGBORO W., DAYTON, OH, 45439	US Mail (1st Class)
24846	KEVIN BURTON, 3223 34TH AVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	KEVIN DOMINIC, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	KEVIN EARLY, 900 S. WILEY ST., PO BOX 320, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	KEVIN EDWARDS, 701 W COMMONS ST NE, CANTON, OH, 44721-3917	US Mail (1st Class)
24846	KEVIN J. CHAMBERS, 1401 INDUSTRIAL PARK, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	KEVIN KONCZAK, 348 BROOK STREET, ROMEO, MI, 48065	US Mail (1st Class)
24846	KEVIN MYERS, 5604 BUCKHEAD DR, NORTHPORT, AL, 35475	US Mail (1st Class)
24846	KEVIN SHRIDER, 900 S. WILEY STREET, PO BOX 320, CRESTLINE, OH, 44827-0320	US Mail (1st Class)
24846	KEY EQUIPMENT FINANCE, 1000 S MCCASLIN BLVD, SUPERIOR, CO, 80027-9456	US Mail (1st Class)
24846	KEY LOGISTICS, 140 MUSHROOM BLVD, ROCHESTER, NY, 14623	US Mail (1st Class)
24846	KEY PLASTICS, 40 TIDWORTH SQUARE, SCARBOROUGH, ON, M1S 2V3CANADA	US Mail (1st Class)
24846	KEY PLASTICS INC., 201 EAST MAIN STREET, PO BOX 140, EL PASO, TX, 79980-0001	US Mail (1st Class)
24846	KEYE PRODUCTIVITY CENTER, PO BOX 4725, BUFFALO, NY, 14240-4725	US Mail (1st Class)
24846	KEY-PLAS, 10 WITHAY DRIVE, AJAX, ON, L1T 3J6CANADA	US Mail (1st Class)
24846	KEY-PLAS, 10 WHITAY DRIVE, AJAX, ON, L1T 3J6CANADA	US Mail (1st Class)
24846	KIMASTLE, 28291 KEHRIG DRIVE, CHESTERFIELD, MI, 48047	US Mail (1st Class)
24846	KIMBERLY A. LAPPING, 5400 WEST GATE DRIVE, GIRARD, PA, 16417	US Mail (1st Class)
24846	KIMBERLY CONNER, 1012 23RD AVE EAST, TUSCALOOSA, AL, 35404	US Mail (1st Class)
24846	KIMBERLY KENNEDY, 1401 INDUSTRIAL DR., TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	KIMBERLY MCCOLLUM, 603 HARGROVE RD E, APT 107, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	KIMBERLY-CLARK, 351 PHELPS DR, IRVING, TX, 75038	US Mail (1st Class)
24846	KINCAID CUSTOM MACHINING &, MOLD, PO BOX 295, NEWBURY, OH, 44065	US Mail (1st Class)
24846	KINDT - COLLINS COMPANY, 12651 ELMWOOD AVE, CLEVELAND, OH, 44111	US Mail (1st Class)
24846	KINE POWER, DIV. OF KINEFAC, PO BOX 612, LEOMINSTER, MA, 01453	US Mail (1st Class)
24846	KINETICO INC., 11015 KINSMAN RD, NEWBURY, OH, 44065	US Mail (1st Class)
24846	KINETICS NOISE CONTROL, 6300 IRELAN PLACE, PO BOX 655, DUBLIN, OH, 43017-0655	US Mail (1st Class)
24846	KING HUGES FASTENER CO., 550 FOURTH ST., IMLAY CITY, MI, 48444	US Mail (1st Class)
24846	KING TOOL & GAGE, 114 1ST ST SW, FAYETTE, AL, 35555-2706	US Mail (1st Class)
24846	KINKO'S INC., 942 S SHADY GROVE RD, MEMPHIS, TN, 38120-4117	US Mail (1st Class)
24846	KINNEY VACUUM DIVISION, TUTHILL CORP., PO BOX 92694, CHICAGO, IL, 60675-2694	US Mail (1st Class)
24846	KINZER, ALLEN, 35 LATOUR WAY, GREER, SC, 29650-4553	US Mail (1st Class)
24846	KIPLINGER WASHINGTON LETTER, 1729 H ST NW, WASHINGTON, DC, 20006-3901	US Mail (1st Class)
24846	KIRBY PLUMBING & HEATING, 604 CLOVERDALE AVE, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	KISTER PROPERTIES, 2024 AETNA RD, ASHTABULA, OH, 44004-6260	US Mail (1st Class)
24846	KITTY HAWK CHARTERS, INC., 1640 W 23RD ST, DALLAS, TX, 75261-4024	US Mail (1st Class)
24846	KITTY TILLEY, 1401 INDUSTRIAL DRIVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	KMI PRINTING, 112 PARKER CT, CHARDON, OH, 44024-1112	US Mail (1st Class)
24846	KML INDUSTRIAL SUPPLY INC., 3312 WEST 22ND STREET, ERIE, PA, 16506	US Mail (1st Class)
24846	KNIGHT ERGONOMIC &, ASSEMBLY SYSTEMS, PO BOX 75393, CLEVELAND, OH, 44101-2199	US Mail (1st Class)
24846	KNIGHT MARKETING OF ALABAMA, 1041 HULL RD, SUMITON, AL, 35148	US Mail (1st Class)
24846	KNOX COMPANY, 17672 ARMSTRONG AVE, IRVINE, CA, 92614	US Mail (1st Class)
24846	KNOX MCLAUGHLIN GORNALL, 120 W 10TH STREET, ERIE, PA, 16501-1461	US Mail (1st Class)
24846	KOBER SALES, 29 S AURELIUS RD, MASON, MI, 48854-9504	US Mail (1st Class)
24846	KOCH SUPPLIES, INC., LOCKBOX, PO BOX 95087, CHICAGO, IL, 60694-5087	US Mail (1st Class)
24846	KOLLMORGEN CORPORATION, INDUSTRIAL DRIVES DIV., 3606 COLLECTIONS CENTER DR., CHICAGO, IL, 60693	US Mail (1st Class)
24846	KOMAR SCREW CORP., 7790 N. MERRIMAC, NILES, IL, 60714	US Mail (1st Class)
24846	KOOLTRONICS INC., 30 PENNINGTON HOPEWELL RD, PENNINGTON, NJ, 08534	US Mail (1st Class)
24846	KOTLER MARKETING GROUP, 925 15TH STEET NW, 4TH FLOOR, WASHINGTON, DC, 20005	US Mail (1st Class)
24846	KOTTER METAL PRODUCTS, 1595 LOST NATION ROAD, WILLOUGHBY, OH, 44094-4633	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	KP ASSOCIATES INC., 3951 RED BANK ROAD, CINCINNATI, OH, 45227	US Mail (1st Class)
24846	KRAFT FLUID SYSTEMS, INC., 14300 FOLTZ PARKWAY, STRONGSVILLE, OH, 44149	US Mail (1st Class)
24846	KRAIG MCINALLY, 290 MCCORMICK, LAPEER, MI, 48446	US Mail (1st Class)
24846	KREIDER CORPORATION, 2000 S YELLOW SPRINGS ST., SPRINGFIELD, OH, 45506	US Mail (1st Class)
24846	KRIS CAM PALLETS, 1506 WESLEY AVE, ERIE, PA, 16510-1676	US Mail (1st Class)
24846	KRISTAL IVANIC, 31125 PORTSIDE DR APT 6301, NOVI, MI, 48377-4203	US Mail (1st Class)
24846	KROGERS CO., ADDRESS UNAVAILABLE,	US Mail (1st Class)
24846	KUHN TOOL & DIE CO, 373 DAVIS ST, PO BOX 574, MEADVILLE, PA, 16335	US Mail (1st Class)
24846	KUHNLEIN & MARTIN, INC., 33001 SOLON RD, SUITE 220, SOLON, OH, 44139	US Mail (1st Class)
24846	KURTZMAN SANITATION, 725 W MAIN, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	KVT TECHNOLOGIES INC., 2199 BLACKACRE DRIVE, OLDCASTLE, ON, NOR 1L0CANADA	US Mail (1st Class)
24846	K-W PLASTICS RECYCLING DIVISION, PIKE COUNTY LAKE ROAD, TROY, AL, 36079	US Mail (1st Class)
24846	KYLE JACKSON, 102B FAIRWAY DR., CHARDON, OH, 44024	US Mail (1st Class)
24846	L & M MACHINE LIMITED, 6825 CONC.2, RR#1, STONEY POINT, ON, NOR 1N0CANADA	US Mail (1st Class)
24846	L & P MACHINE COMPANY, 8488 ROUTE 305, GARRETTSVILLE, OH, 44231	US Mail (1st Class)
24846	L.T. ASSOCIATES, INC., 1250 BEDFORD S.W., CANTON, OH, 44710	US Mail (1st Class)
24846	LA MAISON, 1755 JAMESTOWN DRIVE, MANSFIELD, OH, 44906	US Mail (1st Class)
24846	LA ROCA INDUSTRIES, 416 FERRY DR STE B, ASHTABULA, OH, 44004-3573	US Mail (1st Class)
24846	LA ROS EQUIPMENT CO., 8278 SHAVER ROAD, PORTAGE, MI, 49002	US Mail (1st Class)
24846	LAB SAFETY SUPPLY INC., 401 S WRIGHT RD, JANESVILLE, WI, 53546	US Mail (1st Class)
24846	LAB SAFETY SUPPLY INC., 401 S WRIGHT ROAD, PO BOX 5004, JANESVILLE, WI, 53547-5004	US Mail (1st Class)
24846	LABEL WORLD, 29 JET VIEW DRIVE, ROCHESTER, NY, 14624	US Mail (1st Class)
24846	LABELMASTER, AN AMERICAN LABELMARK COMPANY, PO BOX 46402, CHICAGO, IL, 60646-0402	US Mail (1st Class)
24846	LABELTAPE, INC., 4558 50TH STREET S.E., GRAND RAPIDS, MI, 49512	US Mail (1st Class)
24846	LABOR DEPT-BOILER DIV, 31708 BRETTON ST, LIVONIA, MI, 48152-1534	US Mail (1st Class)
24846	LABOR WORKS, 2600 PRESTON HWY, LOUISVILLE, KY, 40217-2433	US Mail (1st Class)
24846	LADDER MAN INC., 3025 SILVER DR, COLUMBUS, OH, 43224	US Mail (1st Class)
24846	LAGANOWSKI & ASSOCIATES, INC., 784 ST. MARY'S PKWY, BUFFALO GROVE, IL, 60089	US Mail (1st Class)
24846	LAIRD PLASTICS, 6800 BROKEN SOUND PKWY STE 150, BOCA RATON, FL, 33487-2788	US Mail (1st Class)
24846	LAKE BUSINESS PRODUCTS, 38322 APOLLO PARKWAY, WILLOUGHBY, OH, 44094	US Mail (1st Class)
24846	LAKE ERIE IRON & METAL CO, 26 STAGE AVE, PAINESVILLE, OH, 44077-3050	US Mail (1st Class)
24846	LAKE ERIE VIDEO PRODUCTS INC., 101 SOUTH ST, CHARDON, OH, 44024-1336	US Mail (1st Class)
24846	LAKESHORE, 2000 FASAN DRIVE, OLDCASTLE, ON, NOR 1L0CANADA	US Mail (1st Class)
24846	LAKESHORE TOOL & MOLD, 5220 OUTER DR., WINDSOR, ON, N9A 6J3CANADA	US Mail (1st Class)
24846	LAKESIDE SUPPLY COMPANY, 3000 W. 117TH STREET, CLEVELAND, OH, 44111-1667	US Mail (1st Class)
24846	LAKEWOOD AUTOMATION, INC., 27911 CLEMENS ROAD, WESTLAKE, OH, 44145	US Mail (1st Class)
24846	LAMKO TOOL, 105 TOWERLINE PLACE, LONDON, ON, N6E 2T3CANADA	US Mail (1st Class)
24846	LAN MENG, 43077 BROOKSTONE DR., NOVI, MI, 48377	US Mail (1st Class)
24846	LAND INSTRUMENTS, 10 FRIENDS LANE, NEWTOWN, PA, 18940	US Mail (1st Class)
24846	LANDSTAR RANGER INC, 13410 SUTTON PARK DR S, JACKSONVILLE, FL, 32224	US Mail (1st Class)
24846	LANDSTAR RANGER INC, 13410 SUTTON PARK DR SOUTH, JACKSONVILLE, FL, 32224	US Mail (1st Class)
24846	LANGUA TRANSLATIONS INC, 30150 TELEGRAPH ROAD, SUITE 300, BINGHAM FARMS, MI, 48025	US Mail (1st Class)
24846	LANXESS CORP., ATTN DEBBIE KRAUSE, 111 RIDC PARK WEST DR, PITTSBURGH, PA, 15275-1112	US Mail (1st Class)
24846	LAPEER AREA CHAMBER OF, COMMERCE, LAPEER, MI, 48446	US Mail (1st Class)
24846	LAPEER CARTAGE COMPANY, 3187 HIRED HAND RD., LAPEER, MI, 48446	US Mail (1st Class)
24846	LAPEER CITY FINANCE DEPT, 576 LIBERTY ST, LAPEER, MI, 48446-2185	US Mail (1st Class)
24846	LAPEER COUNTRY CLUB, 3786 HUNT ROAD, LAPEER, MI, 48446	US Mail (1st Class)
24846	LAPEER COUNTY PARKS &, RECREATION DEPT., LAPEER, MI, 48446	US Mail (1st Class)
24846	LAPEER COUNTY SPORTSMAN CLUB, 1227 N LAKE GEORGE ROAD, ATTICA, MI, 48412	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	LAPEER COUNTY VISION CTR, 129 W. NEPESSING STREET, LAPEER, MI, 48446	US Mail (1st Class)
24846	LAPEER DEVELOPMENTS CORP, 449 MC CORMICK ST., LAPEER, MI, 48446	US Mail (1st Class)
24846	LAPEER GRAIN CO., 155 SOUTH SAGINAW, LAPEER, MI, 48446	US Mail (1st Class)
24846	LAPEER INCOME TAX DEPT, 576 LIBERTY ST, LAPEER, MI, 48446-2185	US Mail (1st Class)
24846	LAPEER KIWANI'S CLUB, KIWANIS INTERNATIONAL, 3636 WOODVIEW TRCE, INDIANAPOLIS, IN, 46268-3196	US Mail (1st Class)
24846	LAPEER LOCK & SAFE, 343 PINE, LAPEER, MI, 48446	US Mail (1st Class)
24846	LAPEER REGIONAL HOSPITAL, 1375 NORTH MAIN, LAPEER, MI, 48446	US Mail (1st Class)
24846	LAPEER RENT-ALL, INC., 1399 IMLAY CITY RD., LAPEER, MI, 48446	US Mail (1st Class)
24846	LAPEER TEAM WORK, 655 MCCORMICK, LAPEER, MI, 48446	US Mail (1st Class)
24846	LAPEER WEST HIGH SCHOOL, C/O PROJECT GRADUATION, LAPEER, MI, 48446	US Mail (1st Class)
24846	LAPMASTER INT'L, 6400 W OAKTON ST, MORTON GROVE, IL, 60053	US Mail (1st Class)
24846	LAPP ROOFING & SHEET, 2860 OME AVENUE, DAYTON, OH, 45414	US Mail (1st Class)
24846	LARAE A. BYRD, 8938 MACARTHUR BLVD, YPSILANTI, MI, 48198	US Mail (1st Class)
24846	LARCO INDUSTRIES, INC., 9833 5 MILE RD, NORTHVILLE, MI, 48168	US Mail (1st Class)
24846	LAROSE RF SYSTEMS, 150 DOVER RD., PO BOX 158, MILLIS, MA, 02054-0158	US Mail (1st Class)
24846	LAROSE TOOL CO., INC., 3407B KUEMERIE AVE. NE, CANTON, OH, 44705	US Mail (1st Class)
24846	LARRICK'S WAREHOUSE OUTLET, 23 MACKOIL AVE., DAYTON, OH, 45403	US Mail (1st Class)
24846	LARRY BUTTERFIELD, 31572 INDUSTRIAL RD, SUITE #300, LIVONIA, MI, 48150	US Mail (1st Class)
24846	LARRY GARDNER, 3131 COLUMBUS RD. NE, CANTON, OH, 44705	US Mail (1st Class)
24846	LARRY J ROOT, 43162 WEBSTER RD, WELLINGTON, OH, 44090-9614	US Mail (1st Class)
24846	LARRY LITTLE, 3131 COLUMBUS RD NE, CANTON, OH, 44705	US Mail (1st Class)
24846	LARRY SMITH, 1401 INDUSTRIAL PARK, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	LASERCOMP, INC., 12878 FARMINGTON RD., LIVONIA, MI, 48150	US Mail (1st Class)
24846	LASTAR, 1501 WEBSTER ST, DAYTON, OH, 45404-1559	US Mail (1st Class)
24846	LAURA NOLD, 7893 REMINGER, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	LAURALEE SWEENEY, 31572 INDUSTRIAL RD., SUITE #300, LIVONIA, MI, 48150	US Mail (1st Class)
24846	LAUREL OFFICE PARK LIMITED, 17197 N LAUREL PARK DR, SUITE 171, LIVONIA, MI, 48152	US Mail (1st Class)
24846	LAURIE M. RIDLON, 8707 SAMUEL BARTON DRIVE, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	LAVAL TOOL & MOULD LTD., 49658 CONCESSION, MAIDSTONE, ON, NOR 1A0CANADA	US Mail (1st Class)
24846	LAVALLA PROTOTYPE, 29370 JOHN R, MADISON HEIGHTS, MI, 48071	US Mail (1st Class)
24846	LAW OFFICES OF BRIKHO &, KALLABAT, P.C., 29992 NORTHWESTERN HIGHWAY, FARMINGTON HILLS, MI, 48334	US Mail (1st Class)
24846	LAWRENCE & FREDERICK, 411 E NORTH AVE., STREAMWOOD, IL, 60107	US Mail (1st Class)
24846	LAWRENCE L. KRAMMIN, 4889 COATS GROVE ROAD, HASTINGS, MI, 49058	US Mail (1st Class)
24846	LAWRENCE M FELLENGER, 824 FERNCLIFF AVE NW, WARREN, OH, 44483-2125	US Mail (1st Class)
24846	LEA AND SACHS, 1345 E GOLF ROAD, PO BOX 1667, DES PLAINES, IL, 60017	US Mail (1st Class)
24846	LEAH RAUSCHER, C/O BUNDY, 1901 WAGER RD., ERIE, PA, 16509	US Mail (1st Class)
24846	LEAN ENTERPRISE INSTITUTE,INC., ONE CAMBRIDGE CENTER, CAMBRIDGE, MA, 02142	US Mail (1st Class)
24846	LEAR CORPORATION, 21557 TELEGRAPH RD, SOUTHFIELD, MI, 48033-4248	US Mail (1st Class)
24846	LEAR ELECTRIC INC., 16604 OLD STATE RD, MIDDLEFIELD, OH, 44062-8213	US Mail (1st Class)
24846	LEBLOND LATHE PARTS LTD, DEPARTMENT 134101, PO BOX 67000, DETROIT, MI, 48267-1341	US Mail (1st Class)
24846	LECTRO COMPONENTS, INC., 154 EASY STREET, CAROL STREAM, IL, 60188	US Mail (1st Class)
24846	LEE A. HALLMAN, LICENSE COMMISSIONER, PO BOX 020737, TUSCALOOSA, AL, 35402	US Mail (1st Class)
24846	LEE COUNTY CHAMBER OF COMMERCE, 221 N MAIN ST, BISHOPVILLE, SC, 29010-1415	US Mail (1st Class)
24846	LEE COUNTY TREASURER, PO BOX 309, BISHOPVILLE, SC, 29010-0309	US Mail (1st Class)
24846	LEE HECHT HARRISON LLC, DEPT CH #10544, PALATINE, IL, 60055-0544	US Mail (1st Class)
24846	LEE SPRING COMPANY, 1462-62ND STREET, BROOKLYN, NY, 11219-5448	US Mail (1st Class)
24846	LEGATO SYSTEMS, INC., PO BOX 58199, SANTA CLARA, CA, 95052-8199	US Mail (1st Class)
24846	LENZKES CLAMPING TOOLS INC, 825 RADFORD ST, CHRISTIANSBURG, VA, 24073-3306	US Mail (1st Class)

24946   LEON ENTERPRISE INSTITUTE, 651 WASHINGTON ST. BROOKLINE, MA. 02446-4517   US Mail (1st Class)   24946   LEONARD ENTERPRISES, INC., 302 N ELIZABETH, CORONA, CA, 91720-2504   US Mail (1st Class)   24946   LEPPERT MACHINE CO., INC., 580 WAYNE ST. MANSFIELD, 0H, 44906   US Mail (1st Class)   24946   LESUEJ, LAYTON, 192 SOLOMON ROAD, ALICEVILLE, AL, 35442   US Mail (1st Class)   24946   LESUEJ, LAYTON, 192 SOLOMON ROAD, ALICEVILLE, AL, 35442   US Mail (1st Class)   24946   LEWELLYN TECHNOLOGY, INC., 1375 W ROAD, LINTON, IN, 47441   US Mail (1st Class)   24946   LIBERTY COMPANIES, 1534 RAGE ST. CINCINNATI, OH, 45202-7012   US Mail (1st Class)   24946   LIBERTY SOLVENTS AND, CHEMICALS, 2928 PAVENNAR AROD, TWINSBURG, OH, 44087   US Mail (1st Class)   24946   LIFESAPE SERVICES-CLEVELAND, 5971 POWERS AVE, SUITE B, JACKSONVILLE, FL, 32217   US Mail (1st Class)   24946   LINC SYSTEMS, 16540 SOUTH-RARK DR, WESTFIELD, IN, 46074   US Mail (1st Class)   24946   LINC SYSTEMS, 16540 SOUTH-RARK DR, WESTFIELD, IN, 46074   US Mail (1st Class)   24946   LINC SYSTEMS, 16540 SOUTH-RARK DR, WESTFIELD, IN, 46074   US Mail (1st Class)   24946   LINC SYSTEMS, 16540 SOUTH-RARK DR, WESTFIELD, IN, 46074   US Mail (1st Class)   24946   LINC SOUTH-RANK DR, WESTFIELD, IN, 46074   US Mail (1st Class)   24946   LINC SOUTH-RANK DR, COSTORIA MI, 49435   US MAIl (1st Class)   24946   LINC SOUTH-RANK DR, T, 7751 HOWARD AVENUE, MCCREGOR, ON, NOR 1JUCANADA   US Mail (1st Class)   24946   LINC SOUTH-RANK DR, T, 7751 HOWARD AVENUE, MCCREGOR, ON, NOR 1JUCANADA   US Mail (1st Class)   24946   LINC SOUTH-RANK DR, T, 7751 HOWARD AVENUE, MCCREGOR, ON, NOR 1JUCANADA   US Mail (1st Class)   24946   LING SOUTH-RANK DR, T, 7751 HOWARD AVENUE, MCCREGOR, ON, NOR 1JUCANADA   US Mail (1st Class)   24946   LING SOUTH-RANK DR, T, 7751 HOWARD AVENUE, MCCREGOR, ON, NOR 1JUCANADA   US Mail (1st Class)   24946   LINC SOUTH-RANK DR, SOUTH-RANK DR, AVENT DR	Svc Lst	Name and Address of Served Party	Mode of Service
24846         LEPPERT MACHINE CO., INC., 650 WAYNE ST, MANSFIELD, OH, 44905         US Mail (1st Class)           24846         LESSUEJ, J. LAYTON. 192 SOLOMON ROAD. ALICEVILLE, AL., 35442         US Mail (1st Class)           24846         LEWSELLWN TECHNOLOGY, INC., 1375 W ROAD, LINTONI, NI, 47441         US Mail (1st Class)           24846         LEWSELLWN TECHNOLOGY, INC., 1375 W ROAD, LINTONI, NI, 47441         US Mail (1st Class)           24846         LIBERTY COMPANIES, 1534 RACE ST, CINCINNATI, OH, 45022-7012         US Mail (1st Class)           24846         LIBERSTY SOLVENTS AND, CHEMICALS, 9429 RAVENINA ROAD, TWINSBURG, OH, 44087         US Mail (1st Class)           24846         LIFESAFE SERVICES-CLEVELAND, 5971 POWERS AVE. SUITE 8, JACKSONVILLE, FL, 32217         US Mail (1st Class)           24846         LIFESAFE SERVICES-CLEVELAND, 5971 POWERS AVE. SUITE 8, JACKSONVILLE, FL, 32217         US Mail (1st Class)           24846         LINDATE, NIC., 1550 AIRPORT RD, CALLATIN, TN, 37068         US Mail (1st Class)           24846         LINDE, ANDER SERVICES-CLEVELAND, 5971 PLANTER, DR, WESTFIELD, IN, 46074         US Mail (1st Class)           24846         LINDE, DRIVE TRANSPORT, 7781 HOWARD AVENUE, MCCREGOR, ON, NOR 1JOCANADA         US Mail (1st Class)           24846         LINDE, DRIVE REVERSOR SIN RAILE ADAM ROAD, LEAVITTSBURG, OH, 44430         US Mail (1st Class)           24846         LINDA CORNER, CREMENT, 470 PORTAGE LAKES DR R 201	24846	LEON ENTERPRISE INSTITUTE, 651 WASHINGTON ST, BROOKLINE, MA, 02446-4517	US Mail (1st Class)
24846         LESLE J. LAYTON, 192 SOLOMOR ROAD, ALICEVILLE, AL, 35442         US Mail (1st Class)           24846         LESSONS IN ILEADERSHIP, SECTION 210, LOUISVILLE, KY, 40289         US Mail (1st Class)           24846         LEWELLYN TECHNOLOGY, INC., 1378 NORD, LINTON, IN, 47441         US Mail (1st Class)           24846         LIBERTY COMPANIES, 1534 RACE ST. CINCINNATI, OH, 45202-7012         US Mail (1st Class)           24846         LIBERTY SOLVENTS AND, CHEMICALS, 9429 RAVENNA ROAD, TWINSBURG, OH, 44087         US Mail (1st Class)           24846         LIBERTY SOLVENTS AND, CHEMICALS, 9429 RAVENNA ROAD, TWINSBURG, OH, 44087         US Mail (1st Class)           24846         LIBERTY COMPANIES, 1534 RACE ST. CINCINNATIO, OH, 4417         US Mail (1st Class)           24846         LINCS SYSTEMS, 16540 SOUTHPARK DR, WESTFIELD, IN, 46074         US Mail (1st Class)           24846         LINDA HAWE, 266 GOODRICH ROAD, FOSTORIA, MI, 46435         US Mail (1st Class)           24846         LINDERORNE TUBE COMPANY, 1500 E 219TH, CEVELAND, OH, 44117         US Mail (1st Class)           24846         LINDERORNE TUBE COMPANY, 1500 E 219TH, CEVELAND, OH, 44117         US Mail (1st Class)           24846         LINDERORNE TUBE COMPANY, 1500 E 219TH, CEVELAND, OH, 44117         US Mail (1st Class)           24846         LINDA CONTERNATION AND AREA SOLORE, BIRMINGHAM BUSINESS PARK, BIRMINGHAM, B37         US Mail (1st Class)	24846	LEONARD ENTERPRISES, INC., 302 N ELIZABETH, CORONA, CA, 91720-2504	US Mail (1st Class)
24846         LESSONS IN LEADERSHIP, SECTION 210, LOUISVILLE, KY, 40289         US Mail (1st Class)           24846         LEWELLYN TECHNOLOGY, LOY, 375 W ROAD, LINTON, IN, 47441         US Mail (1st Class)           24846         LIBERTY COMPANIES, 1534 RACE ST, CINCINNATI, OH, 45202-7012         US Mail (1st Class)           24846         LIBERTY SOLVENTS AND, CHEMICALS, 9429 RAVENNA ROAD, TWINSBURG, OH, 44087         US Mail (1st Class)           24846         LIFESARE SERVICES-CLEVELAND, 5871 POWERS AVE, SUITE 8, JACKSONVILLE, FL, 32217         US Mail (1st Class)           24846         LIFESARE SERVICES-CLEVELAND, 5871 POWERS AVE, SUITE 8, JACKSONVILLE, FL, 32217         US Mail (1st Class)           24846         LINDEREY, 1550 AIRCORT RD, GALLATIN, TN, 37086         US Mail (1st Class)           24846         LINDEREY, 1566 AIRCORT RD, CALLATIN, TN, 37086         US Mail (1st Class)           24846         LINDERINE TUBE COMPANY, 1500 E 218TH, CLEVELAND, OH, 44117         US Mail (1st Class)           24846         LINDERINE TUBE COMPANY, 1500 E 218TH, CLEVELAND, OH, 44117         US Mail (1st Class)           24846         LINDERINE TUBE COMPANY, 1500 E 218TH, CLEVELAND, OH, 44117         US Mail (1st Class)           24846         LINDER NEW E TRANSPORT, 7781 HOWARD AVENUE, MCREGOR, ON, NOR LOCANDA         US Mail (1st Class)           24846         LINDER NEW E TRANSPORT, 7781 HOWARD AVENUE, MCREGOR, ON, NOR LOCANDA         US Mail (1st Clas	24846	LEPPERT MACHINE CO., INC., 650 WAYNE ST, MANSFIELD, OH, 44905	US Mail (1st Class)
24846         LEWELLYN TECHNOLOGY, INC. 1375 W ROAD, LINTON, IN, 47441         US Mail (1st Class)           24846         LIBERTY COMPANIES, 1534 RACE ST. CINCINNATI, OH, 45202-7012         US Mail (1st Class)           24846         LIBERTY SOLVENTS AND, CHEMICALS, 9262 RAVENNA ROAD, TWINSBURG, OH, 44087         US Mail (1st Class)           24846         LIFESAFE SERVICES-CLEVELAND, 5971 POWERS AVE, SUITE 8, JACKSONVILLE, FL, 32217         US Mail (1st Class)           24846         LIREX, INC., 1550 AIRPORTS CH, GALLATIN, TN, 3760         US Mail (1st Class)           24846         LINCATKEY, INC., 1550 AIRPORTS CH, GALLATIN, TN, 3760         US Mail (1st Class)           24846         LINDA HAWE, 266 GOODRICH ROAD, FOSTORIA, MI, 48435         US Mail (1st Class)           24846         LINDA HAWE, 266 GOODRICH ROAD, FOSTORIA, MI, 48435         US Mail (1st Class)           24846         LINDER MET LUBE COMPANY, 1500 E 219TH, CLEVELAND, OH, 44117         US Mail (1st Class)           24846         LINDA CINTERNALS, 3160 PARK SOUARE, BIRNINGHAM BUSINESS PARK, BIRNINGHAM, B37         US Mail (1st Class)           24846         LINDA CINTERNALS, 3160 PARK SOUARE, BIRNINGHAM BUSINESS PARK, BIRNINGHAM, B37         US Mail (1st Class)           24846         LISA SHANK, 5021 N RIDGE RD EAST, ASHTABULA, OH, 44004         US Mail (1st Class)           24846         LISA SHANK, 5021 N RIDGE RD EAST, ASHTABULA, OH, 44004         US Mail (1st Class)     <	24846	LESLIE J. LAYTON, 192 SOLOMON ROAD, ALICEVILLE, AL, 35442	US Mail (1st Class)
24846         LIBERTY SOLVENTS AND, CHEMICALS, 9429 RAVENNA ROAD, TWINSBURG, OH, 44097         US Mail (1st Class)           24846         LIBERTY SOLVENTS AND, CHEMICALS, 9429 RAVENNA ROAD, TWINSBURG, OH, 44097         US Mail (1st Class)           24846         LIFESARE SERVICES CLEVELAND, 97 POWERS AVE, SUITE 8, JACKSONVILLE, FL, 32217         US Mail (1st Class)           24846         LIFESERVERS OF NORTHEAST OHIO, 2826 PICKWICK DR, COLUMBUS, OH, 43221-2926         US Mail (1st Class)           24846         LINDER, 1550 AIRPORT RD, GALLATIN, TN, 37066         US Mail (1st Class)           24846         LINDER, 285 GOODRICH ROAD, FOSTORIA, MI, 48435         US Mail (1st Class)           24846         LINDER, 1550 AIRPORT ROAD, FOSTORIA, MI, 48435         US Mail (1st Class)           24846         LINDER, 1550 CARPORT, 781 HOWARD AVENUE, MCOREGOR, ON, NOR 1JOCANADA         US Mail (1st Class)           24846         LINDER, METRIALS, 3180 PARK SQUARE, BIRMINGHAM BUSINESS PARK, BIRMINGHAM, B37         US Mail (1st Class)           24846         LINDER, DES, 6003 KALE, ADAM ROAD, LEAVITTSBURG, OH, 44430         US Mail (1st Class)           24846         LISA BOOZE, 8003 KALE, ADAM ROAD, LEAVITTSBURG, OH, 44004         US Mail (1st Class)           24846         LISA SHALW, WORTH, 3731 COLUMBUS RD NE, CANTON, OH, 44007         US Mail (1st Class)           24846         LISA SHALW, SCARTH, 3131 COLUMBUS RD NE, CANTON, OH, 45070         US Mail (1st C	24846	LESSONS IN LEADERSHIP, SECTION 210, LOUISVILLE, KY, 40289	US Mail (1st Class)
24846   LIBERTY SOLVENTS AND, CHEMICALS, 9429 RAVENNA ROAD, TWINSBURG, OH, 44087   US Mail (1st Class)   24846   LIFESARE SERVICES-CLEVELAND, 5971 POWERS AVE, SUITE 8, JACKSONVILLE, FL, 32217   US Mail (1st Class)   24846   LIRESERVERS OF NORTHEAST OHD, 2926 PICKWICK DR, COLUMBUS, OH, 43221-2926   US Mail (1st Class)   24846   LING SYSTEMS, 16540 SOUTHPARK DR, WESTRIELD, IN, 46074   US Mail (1st Class)   24846   LING SYSTEMS, 16540 SOUTHPARK DR, WESTRIELD, IN, 46074   US Mail (1st Class)   24846   LIND A HAWE, 268 GOODRICH ROAD, POSTORIA, MI, 48435   US Mail (1st Class)   24846   LINDERME TUBE COMPANY, 1500 E 219TH, CLEVELAND, OH, 44117   US Mail (1st Class)   24846   LINDERME TUBE COMPANY, 1500 E 219TH, CLEVELAND, OH, 44117   US Mail (1st Class)   24846   LINDERME TUBE COMPANY, 1500 E 219TH, CLEVELAND, OH, 44117   US Mail (1st Class)   24846   LINDERME TUBE COMPANY, 1500 E 219TH, CLEVELAND, OH, 44117   US Mail (1st Class)   24846   LINDERME TUBE COMPANY, 1500 E 219TH, CLEVELAND, OH, 44419 - 2296   US Mail (1st Class)   24846   LINDERME AND MATERIALS, 3180 PARK SQUARE, BIRMINGHAM BUSINESS PARK, BIRMINGHAM, B37   TYNUNITED KINGDOM   US Mail (1st Class)   24846   LIDAG ENFORCEMENT, 470 PORTAGE LAKES DR # 204, AKRON, OH, 44319 - 2296   US Mail (1st Class)   24846   LISA SHANK, 5021 N RINGE RD EAST, ASHTABULA, OH, 44004   US Mail (1st Class)   24846   LISA SHANK, 5021 N RINGE RD EAST, ASHTABULA, OH, 44004   US Mail (1st Class)   24846   LIVINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA   US Mail (1st Class)   24846   LIVINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA   US Mail (1st Class)   24846   LIVINGSTON INTERNATIONAL, 405 THE WEST MAIN STREET, MADISON, OH, 44057   US Mail (1st Class)   24846   LOCKLE SERVICES, INC., 744 WEST MAIN STREET, MADISON, OH, 44057   US Mail (1st Class)   24846   LOCKLE SERVICES, INC., 744 WEST MAIN STREET, MADISON, OH, 44057   US Mail (1st Class)   24846   LOCKLE CORPORATE ECH., 1009 RAPIDS DR., PO BOX 0032, RACINE, WI, \$404-0032   US Mail	24846	LEWELLYN TECHNOLOGY, INC., 1375 W ROAD, LINTON, IN, 47441	US Mail (1st Class)
24846         LIFESAFE SERVICES-CLEVELAND, 5971 POWERS AVE, SUITE 8, JACKSONVILLE, FL., 32217         US Mail (fat Class)           24846         LIFESERVERS OF NORTHEAST OHIO, 2926 PICKWICK DR, COLUMBUS, OH, 43221-2926         US Mail (fat Class)           24846         LINDER, INC., 1550 AIRPORT RD, GALATIN, TN., 37066         US Mail (fat Class)           24846         LIND SYSTEMS, 16540 SOUTHPARK DR, WESTFIELD, IN, 46074         US Mail (fat Class)           24846         LINDER HE TUBE COMPANY, 1500 E 219TH, CLEVELAND, OH, 44117         US Mail (fat Class)           24846         LINDER ME TUBE COMPANY, 1500 E 219TH, CLEVELAND, OH, 44117         US Mail (fat Class)           24846         LINPAC MATERIALS, 3180 PARK SQUARE, BIRMINGHAM BUSINESS PARK, BIRMINGHAM, B37         US Mail (fat Class)           24846         LINPAC MATERIALS, 3180 PARK SQUARE, BIRMINGHAM BUSINESS PARK, BIRMINGHAM, B37         US Mail (fat Class)           24846         LISA BOOZE, 6403 KALE ADAM ROAD, LEAVITTSBURG, OH, 44430         US Mail (fat Class)           24846         LISA SHANK, 5021 R. NIGGE RE DAST, ASHTABULA, OH, 44004         US Mail (fat Class)           24846         LISA STALLWORTH, 8-7-7 SAMUEL BARTON DR, BELLEVILLE, MI, 48111         US Mail (fat Class)           24846         LISA STALLWORTH, 18-7-7 SAMUEL BARTON, OH, 44705         US Mail (fat Class)           24846         LISA STALLWORTH, INC., 465         US Mail (fat Class)	24846	LIBERTY COMPANIES, 1534 RACE ST, CINCINNATI, OH, 45202-7012	US Mail (1st Class)
24846         LIFESERVERS OF NORTHEAST OHIO, 2926 PICKWICK DR, COLUMBUS, OH, 43221-2926         US Mail (1st Class)           24846         LINATEX, INC., 1550 AIRPORT RD, GALLATIN, TN, 37066         US Mail (1st Class)           24846         LINDA HAWE, 266 GOODRICH ROAD, FOSTORIA, MI, 48435         US Mail (1st Class)           24846         LINDERME TUBE COMPANY, 1500 E 219TH, CLEVELAND, OH, 44117         US Mail (1st Class)           24846         LINDERME TUBE COMPANY, 1500 E 219TH, CLEVELAND, OH, 44117         US Mail (1st Class)           24846         LINDAC MATERIALS, 3180 PARK SQUARE, BIRMINGHAM BUSINESS PARK, BIRMINGHAM, B37         US Mail (1st Class)           24846         LINDAC MATERIALS, 3180 PARK SQUARE, BIRMINGHAM BUSINESS PARK, BIRMINGHAM, B37         US Mail (1st Class)           24846         LISA STALLWORTH, 470 PORTAGE LAKES DR # 204, AKRON, OH, 44319-2296         US Mail (1st Class)           24846         LISA SHANK, 5021 N RIDGE RD EAST, ASHTABULA, OH, 44004         US Mail (1st Class)           24846         LISA SHALLWORTH, 677-SAMUEL BARTON DR, BELLEVILLE, MI, 48111         US Mail (1st Class)           24846         LIVINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LIVINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LIVINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA<	24846	LIBERTY SOLVENTS AND, CHEMICALS, 9429 RAVENNA ROAD, TWINSBURG, OH, 44087	US Mail (1st Class)
24846         LINATEX, INC., 1550 AIRPORT RD, GALLATIN, TN, 37066         US Mail (1st Class)           24846         LINCS YSTEMS, 16540 SOUTHPARK DR, WESTFIELD, IN, 46074         US Mail (1st Class)           24846         LINDERMIE TUBE COMPANY, 1500 E 219TH, CLEVELAND, OH, 44117         US Mail (1st Class)           24846         LINDERMIE TUBE COMPANY, 1500 E 219TH, CLEVELAND, OH, 44117         US Mail (1st Class)           24846         LINEAPROM TETRIALS, 3180 PARK SQUARE, BIRMINGHAM BUSINESS PARK, BIRMINGHAM, B37         US Mail (1st Class)           24846         LINDAPROM TETRIALS, 3180 PARK SQUARE, BIRMINGHAM BUSINESS PARK, BIRMINGHAM, B37         US Mail (1st Class)           24846         LISA BOOZE, 6403 ALE, ADAM ROAD, LEAVITTSBURG, OH, 44430         US Mail (1st Class)           24846         LISA SHANK, 5021 N RIDGE RD EAST, ASHTABULA, OH, 4404         US Mail (1st Class)           24846         LISA SHANK, 5021 N RIDGE RD EAST, ASHTABULA, OH, 44705         US Mail (1st Class)           24846         LISA WADSWORTH, 3131 COLUMBUS RD NE, CANTON, OH, 44705         US Mail (1st Class)           24846         LINGSTON INTERNATL INC., 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LIVINGSTON INTERNATURAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LIVINGSTON INTERNATURAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st	24846	LIFESAFE SERVICES-CLEVELAND, 5971 POWERS AVE, SUITE 8, JACKSONVILLE, FL, 32217	US Mail (1st Class)
24846         LINC SYSTEMS, 16540 SOUTHPARK DR, WESTFIELD, IN, 46074         US Mail (1st Class)           24846         LINDA HAWE, 266 GOODRICH ROAD, FOSTORIA, MI, 48435         US Mail (1st Class)           24846         LINDERME TUBE COMPANY, 1500 E 219TH, CLEVELAND, OH, 44117         US Mail (1st Class)           24846         LINE-DRIVE TRANSPORT, 7781 HOWARD AVENUE, MCGREGOR, ON, NOR 1J0CANADA         US Mail (1st Class)           24846         LINPAC MATERIALS, 3180 PARK SQUARE, BIRMINGHAM BUSINESS PARK, BIRMINGHAM, B37         US Mail (1st Class)           24846         LIGUOR ENFORCEMENT, 470 PORTAGE LAKES DR # 204, AKRON, OH, 44319-2296         US Mail (1st Class)           24846         LISA SANAK, 5021 N RIDGE RD EAST, ASHTABULA, OH, 44004         US Mail (1st Class)           24846         LISA STALLWORTH, 87-7 SAMUEL BARTON DR, BELLEVILLE, MI, 48111         US Mail (1st Class)           24846         LISA STALLWORTH, 87-7 SAMUEL BARTON DR, BELLEVILLE, MI, 48111         US Mail (1st Class)           24846         LISA SHALLWORTH, 87-7 SAMUEL BARTON DR, BELLEVILLE, MI, 48111         US Mail (1st Class)           24846         LIVINGSTON INTERNATI, INC., 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LIVINGSTON INTERNATI, MIC., 405 THE WEST MAIL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LOYD LOLL, 985 FALLS CREEK DR, VANDALIA, OH, 45377         US Mail (	24846	LIFESERVERS OF NORTHEAST OHIO, 2926 PICKWICK DR, COLUMBUS, OH, 43221-2926	US Mail (1st Class)
24846         LINDA HAWE, 266 GOODRICH ROAD, FOSTORIA, MI, 48435         US Mail (1st Class)           24846         LINDERMET TUBE COMPANY, 1506 E 219TH, CLEVELAND, OH, 44117         US Mail (1st Class)           24846         LINPEDRIVE TRANSPORT, 7781 HOWARD AVENUE, MCGREGOR, ON, NOR 1JOCANADA         US Mail (1st Class)           24846         LINPAC MATERIALS, 3180 PARK SQUARE, BIRMINGHAM BUSINESS PARK, BIRMINGHAM, B37         US Mail (1st Class)           24846         LIQOR ENFORCEMENT, 470 PORTAGE LAKES DR # 204, AKRON, OH, 44319-2296         US Mail (1st Class)           24846         LISA BOOZE, 6403 KALE ADAM ROAD, LEAVITTSBURG, OH, 44430         US Mail (1st Class)           24846         LISA SHANK, 5021 N RIDGE RD EAST, ASHTABULA, OH, 44004         US Mail (1st Class)           24846         LISA STALLWORTH, 3731 COLUMBUS RD NE, CANTON, OH, 44705         US Mail (1st Class)           24846         LIVINGSTON INTERNATI INC., 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LIVINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LIVINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LICOYD LOLL, 985 FALLS CREEK DR., VANDALLA, OH, 45370         US Mail (1st Class)           24846         LOCAL PALLET & WAREHOUSE INC. 734 WEST MAIN STREET, MADISON, OH, 44067	24846	LINATEX, INC., 1550 AIRPORT RD, GALLATIN, TN, 37066	US Mail (1st Class)
24846         LINDERME TUBE COMPANY, 1500 E 219TH, CLEVELAND, OH, 44117         US Mail (1st Class)           24846         LINE-DRIVE TRANSPORT, 7781 HOWARD AVENUE, MCGREGOR, ON, NOR 1JOCANADA         US Mail (1st Class)           24846         LINPAC MATERIALS, 3180 PARK SQUARE, BIRMINGHAM BUSINESS PARK, BIRMINGHAM, B37         US Mail (1st Class)           24846         LIQUOR ENFORCEMENT, 470 PORTAGE LAKES DR # 204, AKRON, OH, 44319-2296         US Mail (1st Class)           24846         LISA BOOZE, 6403 KALE ADAM ROAD, LEAVITTSBURG, OH, 44430         US Mail (1st Class)           24846         LISA SHANK, 5021 N RIDGE RD EAST, ASHTABULA, OH, 44004         US Mail (1st Class)           24846         LISA SHANK, 5021 N RIDGE RD EAST, ASHTABULA, OH, 44705         US Mail (1st Class)           24846         LISA SHANK, 5021 N RIDGE RD EAST, ASHTABULA, OH, 44705         US Mail (1st Class)           24846         LISA SHANK, 5021 N RIDGE RD EAST, ASHTABULA, OH, 44705         US Mail (1st Class)           24846         LISA STALLWORTH, 877 SAMUEL BARTON DR, BELLEVILLE, MI, 48111         US Mail (1st Class)           24846         LIVINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C SK7CANADA         US Mail (1st Class)           24846         LOYD LOLL, 985 FALLS CREEK DR, VANDALIA, OH, 45377         US Mail (1st Class)           24846         LOCAL, SERVICES, 10C., 1734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)	24846	LINC SYSTEMS, 16540 SOUTHPARK DR, WESTFIELD, IN, 46074	US Mail (1st Class)
24846         LINE-DRIVE TRANSPORT, 7781 HOWARD AVENUE, MCGREGOR, ON, NOR 1J0CANADA         US Mail (1st Class)           24846         LINPAC MATERIALS, 3180 PARK SQUARE, BIRMINGHAM BUSINESS PARK, BIRMINGHAM, B37         US Mail (1st Class)           24846         LIQUOR ENFORCEMENT, 470 PORTAGE LAKES DR # 204, AKRON, OH, 44319-2296         US Mail (1st Class)           24846         LISA BOOZE, 6403 KALE ADAM ROAD, LEAVITTSBURG, OH, 44430         US Mail (1st Class)           24846         LISA STALLWORTH, 87-75 SAMUEL BARTON DR, BELLEVILLE, MI, 48111         US Mail (1st Class)           24846         LISA STALLWORTH, 87-75 SAMUEL BARTON DR, BELLEVILLE, MI, 48111         US Mail (1st Class)           24846         LIVINGSTON INTERNAT LINC., 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LIVINGSTON INTERNAT LINC., 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LIVINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LOYD LOLL, 985 FALLS CREEK DR., VANDALIA, OH, 45320         US Mail (1st Class)           24846         LOYD LOLL, 985 FALLS CREEK DR., VANDALIA, OH, 45320         US Mail (1st Class)           24846         LOCAL PALLET & WAREHOUSE INC, 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCKING CYLINDER TECH., 1509 RAPIDS DR., PO BOX 0032,	24846	LINDA HAWE, 266 GOODRICH ROAD, FOSTORIA, MI, 48435	US Mail (1st Class)
24846         LINPAC MATERIALS, 3180 PARK SQUARE, BIRMINGHAM BUSINESS PARK, BIRMINGHAM, B37 7YNUNITED KINGDOM         US Mail (1st Class) 7YNUNITED KINGDOM         US Mail (1st Class) 24846         LISA DOOZE, 6403 KALE ADAM ROAD, LEAVITTSBURG, OH, 444319-2296         US Mail (1st Class) 24846         LISA SHANK, 5021 N RIDGE RD EAST, ASHTABULA, OH, 44004         US Mail (1st Class) 24846         LISA SHANK, 5021 N RIDGE RD EAST, ASHTABULA, OH, 44004         US Mail (1st Class) 24846         LISA SHANK, 5021 N RIDGE RD EAST, ASHTABULA, OH, 44004         US Mail (1st Class) 24846         LISA SHANK, 5021 N RIDGE RD EAST, ASHTABULA, OH, 44004         US Mail (1st Class) 24846         LISA STALLWORTH, 87-7 SAMUEL BARTON DR, BELLEVILLE, MI, 48111         US Mail (1st Class) 24846         LISA WADSWORTH, 3131 COLUMBUS RD NE, CANTON, OH, 44705         US Mail (1st Class) 24846         LIVINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class) 24846         LIVINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class) 24846         LLOYD LOLL, 985 FALLS CREEK DR., VANDALIA, OH, 45377         US Mail (1st Class) 24846         LLOYD LOLL, 985 FALLS CREEK DR., VANDALIA, OH, 45307         US Mail (1st Class) 24846         LOCAL PALLET & WAREHOUSE INC, 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class) 24846         LOCKING CYLINDER TECH., 1509 RAPIDS DR., PO BOX 0032, RACINE, WI, 5401-0032         US Mail (1st Class) 24846         LOCTITE CORP., 1001 TROUT BROOK XING, ROCKY HILL, CT, 06067-3562         US Mail (1st Class) 24846         LOCTITE CORP., 1001 TROUT BROOK XING, ROCKY HILL, CT, 06067-3562         US Ma	24846	LINDERME TUBE COMPANY, 1500 E 219TH, CLEVELAND, OH, 44117	US Mail (1st Class)
TYNUNITED KINGDOM         US Mail (1st Class)           24846         LIQUOR ENFORCEMENT, 470 PORTAGE LAKES DR # 204, AKRON, OH, 44319-2296         US Mail (1st Class)           24846         LISA BOOZE, 6403 KALE ADAM ROAD, LEAVITTSBURG, OH, 44430         US Mail (1st Class)           24846         LISA SHANK, 5021 N RIDGE RD EAST, ASHTABULA, OH, 44004         US Mail (1st Class)           24846         LISA STALLWORTH, 87-7 SAMUEL BARTON DR, BELLEVILLE, MI, 48111         US Mail (1st Class)           24846         LIVINGSTON INTERNAT'L INC., 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LIVINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LIVINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LLOYD LOLL, 985 FALLS CREEK DR., VANDALIA, OH, 45377         US Mail (1st Class)           24846         LOCAL PALLET & WAREHOUSE INC., 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCAL SERVICES, INC., 34 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCKING CYLINDER TECH., 1509 RAPIDS DR., PO BOX 0032, RACINE, WI, 53401-0032         US Mail (1st Class)           24846         LOCKING CYLINDER TECH., 1509 RAPIDS DR., PO BOX 0032, RACINE, WI, 53401-0032         US Mail (1st Class) <td>24846</td> <td>LINE-DRIVE TRANSPORT, 7781 HOWARD AVENUE, MCGREGOR, ON, NOR 1J0CANADA</td> <td>US Mail (1st Class)</td>	24846	LINE-DRIVE TRANSPORT, 7781 HOWARD AVENUE, MCGREGOR, ON, NOR 1J0CANADA	US Mail (1st Class)
24846         LISA BOOZE, 6403 KALE ADAM ROAD, LEAVITTSBURG, OH, 44400         US Mail (1st Class)           24846         LISA SHANK, 5021 N RIDGE RD EAST, SATTABULA, OH, 44004         US Mail (1st Class)           24846         LISA STALLWORTH, 87-7 SAMUEL BARTON DR, BELLEVILLE, MI, 48111         US Mail (1st Class)           24846         LISA WADSWORTH, 3131 COLUMBUS RD NE, CANTON, OH, 44705         US Mail (1st Class)           24846         LIVINGSTON INTERNAT LINC., 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LIVYINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LLOYD LOLL, 985 FALLS CREEK DR., VANDALIA, OH, 45377         US Mail (1st Class)           24846         LOYD LOLL, 985 FALLS CREEK DR., VANDALIA, OH, 45377         US Mail (1st Class)           24846         LOCAL PALLET & WARZEHOUSE INC., 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCAL SERVICES, INC., 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCKING CYLINDER TECH., 1509 RAPIDS DR., PO BOX 0032, RACINE, WI, 53401-0032         US Mail (1st Class)           24846         LOCTIVARE, INC., 166 CORPORATE DR., PORTSMOUTH, NH, 03801-8815         US Mail (1st Class)           24846         LOGISTIC SERVICES INC., LOGISTICS INSIGHT CORP., 12225 STEPHENS RD., WARREN, MI, 48089	24846		US Mail (1st Class)
24846         LISA SHANK, 5021 N RIDGE RD EAST, ASHTABULA, OH, 44004         US Mail (1st Class)           24846         LISA STALLWORTH, 87-7 SAMUEL BARTON DR, BELLEVILLE, MI, 48111         US Mail (1st Class)           24846         LISA WADSWORTH, 3131 COLUMBUS RD NE, CANTON, OH, 44705         US Mail (1st Class)           24846         LIVINGSTON INTERNAT'LINC., 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LIVINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LICYD LOLL, 985 FALLS CREEK DR., VANDALIA, OH, 45377         US Mail (1st Class)           24846         LOCAL PALLET & WAREHOUSE INC, 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCKING CYLINDER TECH., 1509 RAPIDS DR., PO BOX 0032, RACINE, WI, 53401-0032         US Mail (1st Class)           24846         LOCTITE CORP., 1001 TROUT BROOK XING, ROCKY HILL, CT, 06067-3582         US Mail (1st Class)           24846         LOGITE CORP., 1001 TROUT BROOK XING, ROCKY HILL, CT, 06067-3582         US Mail (1st Class)           24846         LOGITE CERVICES INC., 166 CORPORATE DR, PORTSMOUTH, NH, 03801-8815         US Mail (1st Class)           24846         LOGITE SERVICES INC., 168 CORPORATE DR, PORTSMOUTH, NH, 03801-8815         US Mail (1st Class)           24846         LOGITE SERVICES INC., 168 CORPORATE DR, PORTSMOUTH, WI, 04404	24846	LIQUOR ENFORCEMENT, 470 PORTAGE LAKES DR # 204, AKRON, OH, 44319-2296	US Mail (1st Class)
24846         LISA STALLWORTH, 87-7 SAMUEL BARTON DR, BELLEVILLE, MI, 48111         US Mail (1st Class)           24846         LISA WADSWORTH, 3131 COLUMBUS RD NE, CANTON, OH, 44705         US Mail (1st Class)           24846         LIVINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LIVINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LLOYD LOLL, 985 FALLS CREEK DR., VANDALIA, OH, 45377         US Mail (1st Class)           24846         LLOYD LOLL, 985 FALLS CREEK DR., VANDALIA, OH, 45370         US Mail (1st Class)           24846         LOCAL PALLET & WAREHOUSE INC, 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCAL SERVICES, INC., 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCKING CYLINDER TECH., 1509 RAPIDS DR., PO BOX 0032, RACINE, WI, 53401-0032         US Mail (1st Class)           24846         LOCTITE CORP., 1001 TROUT BROOK XING, ROCKY HILL, CT, 06067-3582         US Mail (1st Class)           24846         LOGISTIC SERVICES INC., LOGISTICS INSIGHT CORP., 12225 STEPHENS RD., WARREN, MI, 48089         US Mail (1st Class)           24846         LOGISTIC SERVICES INC., SERVICES INC., THE TRAINING NETWORK, INC., 106 CAPITOLA DRIVE, DURHAM, NC, 27713         US Mail (1st Class)           24846         LONE STAR QUA	24846	LISA BOOZE, 6403 KALE ADAM ROAD, LEAVITTSBURG, OH, 44430	US Mail (1st Class)
24846         LISA WADSWORTH, 3131 COLUMBUS RD NE, CANTON, OH, 44705         US Mail (1st Class)           24846         LIVINGSTON INTERNAT'L INC., 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LIVINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LLOYD LOLL, 985 FALLS CREEK DR., VANDALIA, OH, 45377         US Mail (1st Class)           24846         LOCAL PALLET & WAREHOUSE INC, 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCAL SERVICES, INC., 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCALISERVICES, INC., 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCALISERVICES, INC., 1509 RAPIDS DR., PO BOX 0032, RACINE, WI, 53401-0032         US Mail (1st Class)           24846         LOCTITE CORP., 1001 TROUT BROOK XING, ROCKY HILL, CT, 06067-3582         US Mail (1st Class)           24846         LOGISTIC SERVICES INC., LOGISTICS INSIGHT CORP., 12225 STEPHENS RD., WARREN, MI, 48089         US Mail (1st Class)           24846         LOGISTIC SERVICES INC., LOGISTICS INSIGHT CORP., 12225 STEPHENS RD., WARREN, MI, 48089         US Mail (1st Class)           24846         LOGISTIAND PRODUCTIONS, INC., THE TRAINING NETWORK, INC., 106 CAPITOLA DRIVE, DUS Mail (1st Class)         US Mail (1st Class)           24846 </td <td>24846</td> <td>LISA SHANK, 5021 N RIDGE RD EAST, ASHTABULA, OH, 44004</td> <td>US Mail (1st Class)</td>	24846	LISA SHANK, 5021 N RIDGE RD EAST, ASHTABULA, OH, 44004	US Mail (1st Class)
24846         LIVINGSTON INTERNAT'L INC., 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LIVINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LLOYD LOLL, 985 FALLS CREEK DR., VANDALIA, OH, 45377         US Mail (1st Class)           24846         LM PRODUCTS, 201 LEXINGTON RD., EATON, OH, 45320         US Mail (1st Class)           24846         LOCAL PALLET & WAREHOUSE INC., 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCKING CYLINDER TECH., 1509 RAPIDS DR., PO BOX 0032, RACINE, WI, 53401-0032         US Mail (1st Class)           24846         LOCTITE CORP., 1001 TROUT BROOK XING, ROCKY HILL, CT, 06067-3582         US Mail (1st Class)           24846         LOCTITE CORP., 1001 TROUT BROOK XING, ROCKY HILL, CT, 06067-3582         US Mail (1st Class)           24846         LOGISTIC SERVICES INC., LOGISTICS INSIGHT CORP., 12225 STEPHENS RD., WARREN, MI, 48089         US Mail (1st Class)           24846         LOGOS @ WORK, 2874 S. DIXIE HIGHWAY, DAYTON, OH, 45409         US Mail (1st Class)           24846         LONG SILAND PRODUCTIONS, INC., 1117 EAGLE RIDGE DR., EL PASO, TX, 79912         US Mail (1st Class)           24846         LORENSTON MFG. CO. INC., 1111 RANK PKWY, KOKOMO, IN, 46901-3124         US Mail (1st Class)           24846         LORENSZON SALES & MARKETING, 79 WHITE ST.,	24846	LISA STALLWORTH, 87-7 SAMUEL BARTON DR, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846         LIVINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA         US Mail (1st Class)           24846         LLOYD LOLL, 985 FALLS CREEK DR., VANDALIA, OH, 45377         US Mail (1st Class)           24846         LM PRODUCTS, 201 LEXINGTON RD., EATON, OH, 45320         US Mail (1st Class)           24846         LOCAL PALLET & WAREHOUSE INC, 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCAL SERVICES, INC., 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCKING CYLINDER TECH., 1509 RAPIDS DR., PO BOX 0032, RACINE, WI, 53401-0032         US Mail (1st Class)           24846         LOCTITE CORP., 1001 TROUT BROOK XING, ROCKY HILL, CT, 06067-3582         US Mail (1st Class)           24846         LOCTITE CORP., 1001 TROUT BROOK XING, ROCKY HILL, CT, 06067-3582         US Mail (1st Class)           24846         LOFTWARE, INC., 166 CORPORATE DR, PORTSMOUTH, NH, 03801-6815         US Mail (1st Class)           24846         LOGISTIC SERVICES INC., LOGISTICS INSIGHT CORP., 12225 STEPHENS RD., WARREN, MI, 48089         US Mail (1st Class)           24846         LOGOS @ WORK, 2874 S. DIXIE HIGHWAY, DAYTON, OH, 45409         US Mail (1st Class)           24846         LOGNE STAR QUALITY SVS, INC, 1117 EAGLE RIDGE DR., EL PASO, TX, 79912         US Mail (1st Class)           24846         LORENSTON MFG, CO. INC., 1111 RANK PKWY, KOKOMO, IN, 46901-3	24846	LISA WADSWORTH, 3131 COLUMBUS RD NE, CANTON, OH, 44705	US Mail (1st Class)
24846         LLOYD LOLL, 985 FALLS CREEK DR., VANDALIA, OH, 45377         US Mail (1st Class)           24846         LM PRODUCTS, 201 LEXINGTON RD., EATON, OH, 45320         US Mail (1st Class)           24846         LOCAL PALLET & WAREHOUSE INC, 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCAL SERVICES, INC., 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCKING CYLINDER TECH., 1509 RAPIDS DR., PO BOX 0032, RACINE, WI, 53401-0032         US Mail (1st Class)           24846         LOCTITE CORP., 1001 TROUT BROOK XING, ROCKY HILL, CT, 06067-3582         US Mail (1st Class)           24846         LOCTITE CORP., 1001 TROUT BROOK XING, ROCKY HILL, CT, 06067-3582         US Mail (1st Class)           24846         LOCTITE CORP., 1001 TROUT BROOK XING, ROCKY HILL, CT, 06067-3582         US Mail (1st Class)           24846         LOCTITE CORP., 1001 TROUT BROOK XING, ROCKY HILL, CT, 06067-3582         US Mail (1st Class)           24846         LOGISTIC SERVICES INC., LOGISTICS INSIGHT CORP., 12225 STEPHENS RD., WARREN, MI, 48089         US Mail (1st Class)           24846         LOGOS @ WORK, 2874 S. DIXIE HIGHWAY, DAYTON, OH, 45409         US Mail (1st Class)           24846         LONG SLAND PRODUCTIONS, INC., THE TRAINING NETWORK, INC., 106 CAPITOLA DRIVE, DUS Mail (1st Class)         US Mail (1st Class)           24846         LORENSTON MFG. CO. INC., 1111 RANK	24846	LIVINGSTON INTERNAT`L INC., 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA	US Mail (1st Class)
24846         LM PRODUCTS, 201 LEXINGTON RD., EATON, OH, 45320         US Mail (1st Class)           24846         LOCAL PALLET & WAREHOUSE INC, 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCAL SERVICES, INC., 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCKING CYLINDER TECH., 1509 RAPIDS DR., PO BOX 0032, RACINE, WI, 53401-0032         US Mail (1st Class)           24846         LOCKING CYLINDER TECH., 1509 RAPIDS DR., PO BOX 0032, RACINE, WI, 53401-0032         US Mail (1st Class)           24846         LOCKING CYLINDER TECH., 1509 RAPIDS DR., PO BOX 0032, RACINE, WI, 53401-0032         US Mail (1st Class)           24846         LOCTITE CORP., 1001 TROUT BROOK XING, ROCKY HILL, CT, 06067-3582         US Mail (1st Class)           24846         LOFTWARE, INC., 166 CORPORATE DR, PORTSMOUTH, NH, 03801-6815         US Mail (1st Class)           24846         LOGISTIC SERVICES INC., LOGISTICS INSIGHT CORP., 12225 STEPHENS RD., WARREN, MI, 48089         US Mail (1st Class)           24846         LOGOS @ WORK, 2874 S. DIXIE HIGHWAY, DAYTON, OH, 45409         US Mail (1st Class)           24846         LONG STAR QUALITY SVS, INC, 1117 EAGLE RIDGE DR., EL PASO, TX, 79912         US Mail (1st Class)           24846         LONE STAR QUALITY SVS, INC, 1117 EAGLE RIDGE DR., EL PASO, TX, 79912         US Mail (1st Class)           24846         LORENSTON MFG. CO. INC., 111	24846	LIVINGSTON INTERNATIONAL, 405 THE WEST MALL, TORONTO, ON, M9C 5K7CANADA	US Mail (1st Class)
24846         LOCAL PALLET & WAREHOUSE INC, 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCAL SERVICES, INC., 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCKING CYLINDER TECH., 1509 RAPIDS DR., PO BOX 0032, RACINE, WI, 53401-0032         US Mail (1st Class)           24846         LOCTITE CORP., 1001 TROUT BROOK XING, ROCKY HILL, CT, 06067-3582         US Mail (1st Class)           24846         LOFTWARE, INC., 166 CORPORATE DR, PORTSMOUTH, NH, 03801-6815         US Mail (1st Class)           24846         LOGISTIC SERVICES INC., LOGISTICS INSIGHT CORP., 12225 STEPHENS RD., WARREN, MI, 48089         US Mail (1st Class)           24846         LOGOS @ WORK, 2874 S. DIXIE HIGHWAY, DAYTON, OH, 45409         US Mail (1st Class)           24846         LONG ISLAND PRODUCTIONS, INC., 1117 EAGLE RIDGE DR., EL PASO, TX, 79912         US Mail (1st Class)           24846         LORENSTON MFG. CO. INC., 1111 RANK PKWY, KOKOMO, IN, 46901-3124         US Mail (1st Class)           24846         LORENZ & SON SALES & MARKETING, 79 WHITE ST., COBOURG, ON, K9A 4W4CANADA         US Mail (1st Class)           24846         LORMAN EDUCATION SERVICES, 2510 ALPINE RD, EAU CLAIRE, WI, 54702         US Mail (1st Class)           24846         LORNANIE L. CERIMELE, 368 BROADWAY AVE, WARREN, OH, 44484         US Mail (1st Class)           24846         LOUB BOWN, 119 EAST DAYTON ST., WEST	24846	LLOYD LOLL, 985 FALLS CREEK DR., VANDALIA, OH, 45377	US Mail (1st Class)
24846         LOCAL SERVICES, INC., 734 WEST MAIN STREET, MADISON, OH, 44057         US Mail (1st Class)           24846         LOCKING CYLINDER TECH., 1509 RAPIDS DR., PO BOX 0032, RACINE, WI, 53401-0032         US Mail (1st Class)           24846         LOCTITE CORP., 1001 TROUT BROOK XING, ROCKY HILL, CT, 06067-3582         US Mail (1st Class)           24846         LOFTWARE, INC., 166 CORPORATE DR, PORTSMOUTH, NH, 03801-6815         US Mail (1st Class)           24846         LOGISTIC SERVICES INC., LOGISTICS INSIGHT CORP., 12225 STEPHENS RD., WARREN, MI, 48089         US Mail (1st Class)           24846         LOGOS @ WORK, 2874 S. DIXIE HIGHWAY, DAYTON, OH, 45409         US Mail (1st Class)           24846         LONG STAR QUALITY SVS, INC, 1117 EAGLE RIDGE DR., EL PASO, TX, 79912         US Mail (1st Class)           24846         LONG ISLAND PRODUCTIONS, INC., THE TRAINING NETWORK, INC., 106 CAPITOLA DRIVE, DURHAM, NC, 27713         US Mail (1st Class)           24846         LORENSTON MFG. CO. INC., 1111 RANK PKWY, KOKOMO, IN, 46901-3124         US Mail (1st Class)           24846         LORENZ & SON SALES & MARKETING, 79 WHITE ST., COBOURG, ON, K9A 4W4CANADA         US Mail (1st Class)           24846         LORRAINE L. CERIMELE, 368 BROADWAY AVE, WARREN, OH, 44484         US Mail (1st Class)           24846         LOTUSEA, 150 VIEW BEND ST., JOHNSON CITY, TN, 37601-1338         US Mail (1st Class)           24846         LOUBON STEEL INC., 820	24846	LM PRODUCTS, 201 LEXINGTON RD., EATON, OH, 45320	US Mail (1st Class)
24846         LOCKING CYLINDER TECH., 1509 RAPIDS DR., PO BOX 0032, RACINE, WI, 53401-0032         US Mail (1st Class)           24846         LOCTITE CORP., 1001 TROUT BROOK XING, ROCKY HILL, CT, 06067-3582         US Mail (1st Class)           24846         LOFTWARE, INC., 166 CORPORATE DR, PORTSMOUTH, NH, 03801-6815         US Mail (1st Class)           24846         LOGISTIC SERVICES INC., LOGISTICS INSIGHT CORP., 12225 STEPHENS RD., WARREN, MI, 48089         US Mail (1st Class)           24846         LOGOS @ WORK, 2874 S. DIXIE HIGHWAY, DAYTON, OH, 45409         US Mail (1st Class)           24846         LONG ISLAND PRODUCTIONS, INC., 1117 EAGLE RIDGE DR., EL PASO, TX, 79912         US Mail (1st Class)           24846         LONG ISLAND PRODUCTIONS, INC., THE TRAINING NETWORK, INC., 106 CAPITOLA DRIVE, DURHAM, NC, 27713         US Mail (1st Class)           24846         LORENSTON MFG. CO. INC., 1111 RANK PKWY, KOKOMO, IN, 46901-3124         US Mail (1st Class)           24846         LORENZ & SON SALES & MARKETING, 79 WHITE ST., COBOURG, ON, K9A 4W4CANADA         US Mail (1st Class)           24846         LORRAINE L. CERIMELE, 368 BROADWAY AVE, WARREN, OH, 44484         US Mail (1st Class)           24846         LOTUSEA, 150 VIEW BEND ST, JOHNSON CITY, TN, 37601-1338         US Mail (1st Class)           24846         LOUBON STEEL INC., 8208 ELLIS RD., MILLINGTON, MI, 48746         US Mail (1st Class)           24846         LOUDON STEEL INC., 8208 EL	24846	LOCAL PALLET & WAREHOUSE INC, 734 WEST MAIN STREET, MADISON, OH, 44057	US Mail (1st Class)
24846       LOCTITE CORP., 1001 TROUT BROOK XING, ROCKY HILL, CT, 06067-3582       US Mail (1st Class)         24846       LOFTWARE, INC., 166 CORPORATE DR, PORTSMOUTH, NH, 03801-6815       US Mail (1st Class)         24846       LOGISTIC SERVICES INC., LOGISTICS INSIGHT CORP., 12225 STEPHENS RD., WARREN, MI, 48089       US Mail (1st Class)         24846       LOGOS @ WORK, 2874 S. DIXIE HIGHWAY, DAYTON, OH, 45409       US Mail (1st Class)         24846       LONE STAR QUALITY SVS, INC., 1117 EAGLE RIDGE DR., EL PASO, TX, 79912       US Mail (1st Class)         24846       LONG ISLAND PRODUCTIONS, INC., THE TRAINING NETWORK, INC., 106 CAPITOLA DRIVE, DURHAM, NC, 27713       US Mail (1st Class)         24846       LORENSTON MFG. CO. INC., 1111 RANK PKWY, KOKOMO, IN, 46901-3124       US Mail (1st Class)         24846       LORENZ & SON SALES & MARKETING, 79 WHITE ST., COBOURG, ON, K9A 4W4CANADA       US Mail (1st Class)         24846       LORMAN EDUCATION SERVICES, 2510 ALPINE RD, EAU CLAIRE, WI, 54702       US Mail (1st Class)         24846       LORRAINE L. CERIMELE, 368 BROADWAY AVE, WARREN, OH, 44484       US Mail (1st Class)         24846       LOTUSEA, 150 VIEW BEND ST, JOHNSON CITY, TN, 37601-1338       US Mail (1st Class)         24846       LOUBON STEEL INC., 8208 ELLIS RD., MILLINGTON, MI, 48746       US Mail (1st Class)         24846       LOVE BOX COMPANY INC., 700 EAST 37TH ST NORTH, WICHITA, KS, 67219       US Mail (1st Class)	24846	LOCAL SERVICES, INC., 734 WEST MAIN STREET, MADISON, OH, 44057	US Mail (1st Class)
24846       LOFTWARE, INC., 166 CORPORATE DR, PORTSMOUTH, NH, 03801-6815       US Mail (1st Class)         24846       LOGISTIC SERVICES INC., LOGISTICS INSIGHT CORP., 12225 STEPHENS RD., WARREN, MI, 48089       US Mail (1st Class)         24846       LOGOS @ WORK, 2874 S. DIXIE HIGHWAY, DAYTON, OH, 45409       US Mail (1st Class)         24846       LONE STAR QUALITY SVS, INC, 1117 EAGLE RIDGE DR., EL PASO, TX, 79912       US Mail (1st Class)         24846       LONG ISLAND PRODUCTIONS, INC., THE TRAINING NETWORK, INC., 106 CAPITOLA DRIVE, DURHAM, NC, 27713       US Mail (1st Class)         24846       LORENSTON MFG. CO. INC., 1111 RANK PKWY, KOKOMO, IN, 46901-3124       US Mail (1st Class)         24846       LORENZ & SON SALES & MARKETING, 79 WHITE ST., COBOURG, ON, K9A 4W4CANADA       US Mail (1st Class)         24846       LORMAN EDUCATION SERVICES, 2510 ALPINE RD, EAU CLAIRE, WI, 54702       US Mail (1st Class)         24846       LORRAINE L. CERIMELE, 368 BROADWAY AVE, WARREN, OH, 44484       US Mail (1st Class)         24846       LOTUSEA, 150 VIEW BEND ST, JOHNSON CITY, TN, 37601-1338       US Mail (1st Class)         24846       LOU BROWN, 119 EAST DAYTON ST., WEST ALEXANDRIA, OH, 45381       US Mail (1st Class)         24846       LOUDON STEEL INC., 8208 ELLIS RD., MILLINGTON, MI, 48746       US Mail (1st Class)         24846       LOWE'S, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456       US Mail (1st Class) <tr< td=""><td>24846</td><td>LOCKING CYLINDER TECH., 1509 RAPIDS DR., PO BOX 0032, RACINE, WI, 53401-0032</td><td>US Mail (1st Class)</td></tr<>	24846	LOCKING CYLINDER TECH., 1509 RAPIDS DR., PO BOX 0032, RACINE, WI, 53401-0032	US Mail (1st Class)
24846       LOGISTIC SERVICES INC., LOGISTICS INSIGHT CORP., 12225 STEPHENS RD., WARREN, MI, 48089       US Mail (1st Class)         24846       LOGOS @ WORK, 2874 S. DIXIE HIGHWAY, DAYTON, OH, 45409       US Mail (1st Class)         24846       LONE STAR QUALITY SVS, INC, 1117 EAGLE RIDGE DR., EL PASO, TX, 79912       US Mail (1st Class)         24846       LONG ISLAND PRODUCTIONS, INC., THE TRAINING NETWORK, INC., 106 CAPITOLA DRIVE, DURHAM, NC, 27713       US Mail (1st Class)         24846       LORENSTON MFG. CO. INC., 1111 RANK PKWY, KOKOMO, IN, 46901-3124       US Mail (1st Class)         24846       LORENZ & SON SALES & MARKETING, 79 WHITE ST., COBOURG, ON, K9A 4W4CANADA       US Mail (1st Class)         24846       LORMAN EDUCATION SERVICES, 2510 ALPINE RD, EAU CLAIRE, WI, 54702       US Mail (1st Class)         24846       LORRAINE L. CERIMELE, 368 BROADWAY AVE, WARREN, OH, 44484       US Mail (1st Class)         24846       LOTUSEA, 150 VIEW BEND ST, JOHNSON CITY, TN, 37601-1338       US Mail (1st Class)         24846       LOU BROWN, 119 EAST DAYTON ST., WEST ALEXANDRIA, OH, 45381       US Mail (1st Class)         24846       LOUDON STEEL INC., 8208 ELLIS RD., MILLINGTON, MI, 48746       US Mail (1st Class)         24846       LOVE BOX COMPANY INC., 700 EAST 37TH ST NORTH, WICHITA, KS, 67219       US Mail (1st Class)         24846       LOWE'S, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456       US Mail (1st Class)	24846	LOCTITE CORP., 1001 TROUT BROOK XING, ROCKY HILL, CT, 06067-3582	US Mail (1st Class)
24846       LOGOS @ WORK, 2874 S. DIXIE HIGHWAY, DAYTON, OH, 45409       US Mail (1st Class)         24846       LONE STAR QUALITY SVS, INC, 1117 EAGLE RIDGE DR., EL PASO, TX, 79912       US Mail (1st Class)         24846       LONG ISLAND PRODUCTIONS, INC., THE TRAINING NETWORK, INC., 106 CAPITOLA DRIVE, DURHAM, NC, 27713       US Mail (1st Class)         24846       LORENSTON MFG. CO. INC., 1111 RANK PKWY, KOKOMO, IN, 46901-3124       US Mail (1st Class)         24846       LORENZ & SON SALES & MARKETING, 79 WHITE ST., COBOURG, ON, K9A 4W4CANADA       US Mail (1st Class)         24846       LORMAN EDUCATION SERVICES, 2510 ALPINE RD, EAU CLAIRE, WI, 54702       US Mail (1st Class)         24846       LORRAINE L. CERIMELE, 368 BROADWAY AVE, WARREN, OH, 44484       US Mail (1st Class)         24846       LOTUSEA, 150 VIEW BEND ST, JOHNSON CITY, TN, 37601-1338       US Mail (1st Class)         24846       LOUBON STEEL INC., 8208 ELLIS RD., MILLINGTON, MI, 48746       US Mail (1st Class)         24846       LOVE BOX COMPANY INC., 700 EAST 37TH ST NORTH, WICHITA, KS, 67219       US Mail (1st Class)         24846       LOWE'S, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456       US Mail (1st Class)         24846       LOWE'S OF ONTARIO, 940 N LEXINGTON-SPRINGMILL RD, MANSFIELD, OH, 44906       US Mail (1st Class)         24846       LOWELL AMICK, 858 TAYLORTOWN RD., MANSFIELD, OH, 44903       US Mail (1st Class)         248	24846	LOFTWARE, INC., 166 CORPORATE DR, PORTSMOUTH, NH, 03801-6815	US Mail (1st Class)
24846         LONE STAR QUALITY SVS, INC, 1117 EAGLE RIDGE DR., EL PASO, TX, 79912         US Mail (1st Class)           24846         LONG ISLAND PRODUCTIONS, INC., THE TRAINING NETWORK, INC., 106 CAPITOLA DRIVE, DURHAM, NC, 27713         US Mail (1st Class)           24846         LORENSTON MFG. CO. INC., 1111 RANK PKWY, KOKOMO, IN, 46901-3124         US Mail (1st Class)           24846         LORENZ & SON SALES & MARKETING, 79 WHITE ST., COBOURG, ON, K9A 4W4CANADA         US Mail (1st Class)           24846         LORMAN EDUCATION SERVICES, 2510 ALPINE RD, EAU CLAIRE, WI, 54702         US Mail (1st Class)           24846         LORRAINE L. CERIMELE, 368 BROADWAY AVE, WARREN, OH, 44484         US Mail (1st Class)           24846         LOTUSEA, 150 VIEW BEND ST, JOHNSON CITY, TN, 37601-1338         US Mail (1st Class)           24846         LOU BROWN, 119 EAST DAYTON ST., WEST ALEXANDRIA, OH, 45381         US Mail (1st Class)           24846         LOUDON STEEL INC., 8208 ELLIS RD., MILLINGTON, MI, 48746         US Mail (1st Class)           24846         LOVE BOX COMPANY INC., 700 EAST 37TH ST NORTH, WICHITA, KS, 67219         US Mail (1st Class)           24846         LOWE'S, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456         US Mail (1st Class)           24846         LOWE'S OF ONTARIO, 940 N LEXINGTON-SPRINGMILL RD, MANSFIELD, OH, 44906         US Mail (1st Class)           24846         LOWELL AMICK, 858 TAYLORTOWN RD., MANSFIELD, OH, 44903 <td>24846</td> <td>LOGISTIC SERVICES INC., LOGISTICS INSIGHT CORP., 12225 STEPHENS RD., WARREN, MI, 48089</td> <td>US Mail (1st Class)</td>	24846	LOGISTIC SERVICES INC., LOGISTICS INSIGHT CORP., 12225 STEPHENS RD., WARREN, MI, 48089	US Mail (1st Class)
24846         LONG ISLAND PRODUCTIONS, INC., THE TRAINING NETWORK, INC., 106 CAPITOLA DRIVE, DURHAM, NC, 27713         US Mail (1st Class)           24846         LORENSTON MFG. CO. INC., 1111 RANK PKWY, KOKOMO, IN, 46901-3124         US Mail (1st Class)           24846         LORENZ & SON SALES & MARKETING, 79 WHITE ST., COBOURG, ON, K9A 4W4CANADA         US Mail (1st Class)           24846         LORMAN EDUCATION SERVICES, 2510 ALPINE RD, EAU CLAIRE, WI, 54702         US Mail (1st Class)           24846         LORRAINE L. CERIMELE, 368 BROADWAY AVE, WARREN, OH, 44484         US Mail (1st Class)           24846         LOTUSEA, 150 VIEW BEND ST, JOHNSON CITY, TN, 37601-1338         US Mail (1st Class)           24846         LOU BROWN, 119 EAST DAYTON ST., WEST ALEXANDRIA, OH, 45381         US Mail (1st Class)           24846         LOUDON STEEL INC., 8208 ELLIS RD., MILLINGTON, MI, 48746         US Mail (1st Class)           24846         LOVE BOX COMPANY INC., 700 EAST 37TH ST NORTH, WICHITA, KS, 67219         US Mail (1st Class)           24846         LOWE'S, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456         US Mail (1st Class)           24846         LOWE'S OF ONTARIO, 940 N LEXINGTON-SPRINGMILL RD, MANSFIELD, OH, 44906         US Mail (1st Class)           24846         LOWES BUSINESS ACCOUNT, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456         US Mail (1st Class)           24846         LOWEY COMPUTER PRODUCTS, 9420 MALTBY RD, BRIGHTON, MI,	24846	LOGOS @ WORK, 2874 S. DIXIE HIGHWAY, DAYTON, OH, 45409	US Mail (1st Class)
DURHAM, NC, 27713         24846       LORENSTON MFG. CO. INC., 1111 RANK PKWY, KOKOMO, IN, 46901-3124       US Mail (1st Class)         24846       LORENZ & SON SALES & MARKETING, 79 WHITE ST., COBOURG, ON, K9A 4W4CANADA       US Mail (1st Class)         24846       LORMAN EDUCATION SERVICES, 2510 ALPINE RD, EAU CLAIRE, WI, 54702       US Mail (1st Class)         24846       LORRAINE L. CERIMELE, 368 BROADWAY AVE, WARREN, OH, 44484       US Mail (1st Class)         24846       LOTUSEA, 150 VIEW BEND ST, JOHNSON CITY, TN, 37601-1338       US Mail (1st Class)         24846       LOU BROWN, 119 EAST DAYTON ST., WEST ALEXANDRIA, OH, 45381       US Mail (1st Class)         24846       LOUDON STEEL INC., 8208 ELLIS RD., MILLINGTON, MI, 48746       US Mail (1st Class)         24846       LOVE BOX COMPANY INC., 700 EAST 37TH ST NORTH, WICHITA, KS, 67219       US Mail (1st Class)         24846       LOWE'S, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456       US Mail (1st Class)         24846       LOWE'S OF ONTARIO, 940 N LEXINGTON-SPRINGMILL RD, MANSFIELD, OH, 44906       US Mail (1st Class)         24846       LOWES BUSINESS ACCOUNT, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456       US Mail (1st Class)         24846       LOWRY COMPUTER PRODUCTS, 9420 MALTBY RD, BRIGHTON, MI, 48116       US Mail (1st Class)	24846	LONE STAR QUALITY SVS, INC, 1117 EAGLE RIDGE DR., EL PASO, TX, 79912	US Mail (1st Class)
LORENZ & SON SALES & MARKETING, 79 WHITE ST., COBOURG, ON, K9A 4W4CANADA  24846 LORMAN EDUCATION SERVICES, 2510 ALPINE RD, EAU CLAIRE, WI, 54702  24846 LORRAINE L. CERIMELE, 368 BROADWAY AVE, WARREN, OH, 44484  LOTUSEA, 150 VIEW BEND ST, JOHNSON CITY, TN, 37601-1338  24846 LOTUSEA, 150 VIEW BEND ST, JOHNSON CITY, TN, 37601-1338  24846 LOU BROWN, 119 EAST DAYTON ST., WEST ALEXANDRIA, OH, 45381  24846 LOUDON STEEL INC., 8208 ELLIS RD., MILLINGTON, MI, 48746  LOVE BOX COMPANY INC., 700 EAST 37TH ST NORTH, WICHITA, KS, 67219  24846 LOWE'S, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456  LOWE'S OF ONTARIO, 940 N LEXINGTON-SPRINGMILL RD, MANSFIELD, OH, 44906  LOWELL AMICK, 858 TAYLORTOWN RD., MANSFIELD, OH, 44903  24846 LOWES BUSINESS ACCOUNT, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456  LOWRY COMPUTER PRODUCTS, 9420 MALTBY RD, BRIGHTON, MI, 48116  US Mail (1st Class)  US Mail (1st Class)  US Mail (1st Class)	24846		US Mail (1st Class)
24846       LORMAN EDUCATION SERVICES, 2510 ALPINE RD, EAU CLAIRE, WI, 54702       US Mail (1st Class)         24846       LORRAINE L. CERIMELE, 368 BROADWAY AVE, WARREN, OH, 44484       US Mail (1st Class)         24846       LOTUSEA, 150 VIEW BEND ST, JOHNSON CITY, TN, 37601-1338       US Mail (1st Class)         24846       LOU BROWN, 119 EAST DAYTON ST., WEST ALEXANDRIA, OH, 45381       US Mail (1st Class)         24846       LOUDON STEEL INC., 8208 ELLIS RD., MILLINGTON, MI, 48746       US Mail (1st Class)         24846       LOVE BOX COMPANY INC., 700 EAST 37TH ST NORTH, WICHITA, KS, 67219       US Mail (1st Class)         24846       LOWE'S, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456       US Mail (1st Class)         24846       LOWE'S OF ONTARIO, 940 N LEXINGTON-SPRINGMILL RD, MANSFIELD, OH, 44906       US Mail (1st Class)         24846       LOWELL AMICK, 858 TAYLORTOWN RD., MANSFIELD, OH, 44903       US Mail (1st Class)         24846       LOWES BUSINESS ACCOUNT, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456       US Mail (1st Class)         24846       LOWRY COMPUTER PRODUCTS, 9420 MALTBY RD, BRIGHTON, MI, 48116       US Mail (1st Class)	24846	LORENSTON MFG. CO. INC., 1111 RANK PKWY, KOKOMO, IN, 46901-3124	US Mail (1st Class)
24846       LORRAINE L. CERIMELE, 368 BROADWAY AVE, WARREN, OH, 44484       US Mail (1st Class)         24846       LOTUSEA, 150 VIEW BEND ST, JOHNSON CITY, TN, 37601-1338       US Mail (1st Class)         24846       LOU BROWN, 119 EAST DAYTON ST., WEST ALEXANDRIA, OH, 45381       US Mail (1st Class)         24846       LOUDON STEEL INC., 8208 ELLIS RD., MILLINGTON, MI, 48746       US Mail (1st Class)         24846       LOVE BOX COMPANY INC., 700 EAST 37TH ST NORTH, WICHITA, KS, 67219       US Mail (1st Class)         24846       LOWE'S, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456       US Mail (1st Class)         24846       LOWE'S OF ONTARIO, 940 N LEXINGTON-SPRINGMILL RD, MANSFIELD, OH, 44906       US Mail (1st Class)         24846       LOWELL AMICK, 858 TAYLORTOWN RD., MANSFIELD, OH, 44903       US Mail (1st Class)         24846       LOWES BUSINESS ACCOUNT, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456       US Mail (1st Class)         24846       LOWRY COMPUTER PRODUCTS, 9420 MALTBY RD, BRIGHTON, MI, 48116       US Mail (1st Class)	24846	LORENZ & SON SALES & MARKETING, 79 WHITE ST., COBOURG, ON, K9A 4W4CANADA	US Mail (1st Class)
24846       LOTUSEA, 150 VIEW BEND ST, JOHNSON CITY, TN, 37601-1338       US Mail (1st Class)         24846       LOU BROWN, 119 EAST DAYTON ST., WEST ALEXANDRIA, OH, 45381       US Mail (1st Class)         24846       LOUDON STEEL INC., 8208 ELLIS RD., MILLINGTON, MI, 48746       US Mail (1st Class)         24846       LOVE BOX COMPANY INC., 700 EAST 37TH ST NORTH, WICHITA, KS, 67219       US Mail (1st Class)         24846       LOWE'S, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456       US Mail (1st Class)         24846       LOWE'S OF ONTARIO, 940 N LEXINGTON-SPRINGMILL RD, MANSFIELD, OH, 44906       US Mail (1st Class)         24846       LOWELL AMICK, 858 TAYLORTOWN RD., MANSFIELD, OH, 44903       US Mail (1st Class)         24846       LOWES BUSINESS ACCOUNT, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456       US Mail (1st Class)         24846       LOWRY COMPUTER PRODUCTS, 9420 MALTBY RD, BRIGHTON, MI, 48116       US Mail (1st Class)	24846	LORMAN EDUCATION SERVICES, 2510 ALPINE RD, EAU CLAIRE, WI, 54702	US Mail (1st Class)
LOU BROWN, 119 EAST DAYTON ST., WEST ALEXANDRIA, OH, 45381  US Mail (1st Class)  LOUDON STEEL INC., 8208 ELLIS RD., MILLINGTON, MI, 48746  US Mail (1st Class)  LOVE BOX COMPANY INC., 700 EAST 37TH ST NORTH, WICHITA, KS, 67219  US Mail (1st Class)  LOWE'S, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456  US Mail (1st Class)  LOWE'S OF ONTARIO, 940 N LEXINGTON-SPRINGMILL RD, MANSFIELD, OH, 44906  US Mail (1st Class)  LOWELL AMICK, 858 TAYLORTOWN RD., MANSFIELD, OH, 44903  US Mail (1st Class)  LOWES BUSINESS ACCOUNT, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456  LOWRY COMPUTER PRODUCTS, 9420 MALTBY RD, BRIGHTON, MI, 48116  US Mail (1st Class)	24846	LORRAINE L. CERIMELE, 368 BROADWAY AVE, WARREN, OH, 44484	US Mail (1st Class)
24846       LOUDON STEEL INC., 8208 ELLIS RD., MILLINGTON, MI, 48746       US Mail (1st Class)         24846       LOVE BOX COMPANY INC., 700 EAST 37TH ST NORTH, WICHITA, KS, 67219       US Mail (1st Class)         24846       LOWE'S, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456       US Mail (1st Class)         24846       LOWE'S OF ONTARIO, 940 N LEXINGTON-SPRINGMILL RD, MANSFIELD, OH, 44906       US Mail (1st Class)         24846       LOWELL AMICK, 858 TAYLORTOWN RD., MANSFIELD, OH, 44903       US Mail (1st Class)         24846       LOWES BUSINESS ACCOUNT, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456       US Mail (1st Class)         24846       LOWRY COMPUTER PRODUCTS, 9420 MALTBY RD, BRIGHTON, MI, 48116       US Mail (1st Class)	24846	LOTUSEA, 150 VIEW BEND ST, JOHNSON CITY, TN, 37601-1338	US Mail (1st Class)
24846 LOVE BOX COMPANY INC., 700 EAST 37TH ST NORTH, WICHITA, KS, 67219  24846 LOWE'S, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456  24846 LOWE'S OF ONTARIO, 940 N LEXINGTON-SPRINGMILL RD, MANSFIELD, OH, 44906  24846 LOWELL AMICK, 858 TAYLORTOWN RD., MANSFIELD, OH, 44903  24846 LOWES BUSINESS ACCOUNT, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456  24846 LOWRY COMPUTER PRODUCTS, 9420 MALTBY RD, BRIGHTON, MI, 48116  US Mail (1st Class)  US Mail (1st Class)	24846	LOU BROWN, 119 EAST DAYTON ST., WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846 LOWE'S, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456  24846 LOWE'S OF ONTARIO, 940 N LEXINGTON-SPRINGMILL RD, MANSFIELD, OH, 44906  24846 LOWELL AMICK, 858 TAYLORTOWN RD., MANSFIELD, OH, 44903  24846 LOWES BUSINESS ACCOUNT, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456  24846 LOWRY COMPUTER PRODUCTS, 9420 MALTBY RD, BRIGHTON, MI, 48116  US Mail (1st Class)  US Mail (1st Class)	24846	LOUDON STEEL INC., 8208 ELLIS RD., MILLINGTON, MI, 48746	US Mail (1st Class)
24846 LOWE'S OF ONTARIO, 940 N LEXINGTON-SPRINGMILL RD, MANSFIELD, OH, 44906 US Mail (1st Class) 24846 LOWELL AMICK, 858 TAYLORTOWN RD., MANSFIELD, OH, 44903 US Mail (1st Class) 24846 LOWES BUSINESS ACCOUNT, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456 US Mail (1st Class) 24846 LOWRY COMPUTER PRODUCTS, 9420 MALTBY RD, BRIGHTON, MI, 48116 US Mail (1st Class)	24846	LOVE BOX COMPANY INC., 700 EAST 37TH ST NORTH, WICHITA, KS, 67219	US Mail (1st Class)
24846 LOWELL AMICK, 858 TAYLORTOWN RD., MANSFIELD, OH, 44903  24846 LOWES BUSINESS ACCOUNT, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456  24846 LOWRY COMPUTER PRODUCTS, 9420 MALTBY RD, BRIGHTON, MI, 48116  US Mail (1st Class)  US Mail (1st Class)	24846	LOWE'S, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456	US Mail (1st Class)
24846 LOWES BUSINESS ACCOUNT, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456 US Mail (1st Class) 24846 LOWRY COMPUTER PRODUCTS, 9420 MALTBY RD, BRIGHTON, MI, 48116 US Mail (1st Class)	24846	LOWE'S OF ONTARIO, 940 N LEXINGTON-SPRINGMILL RD, MANSFIELD, OH, 44906	US Mail (1st Class)
24846 LOWRY COMPUTER PRODUCTS, 9420 MALTBY RD, BRIGHTON, MI, 48116 US Mail (1st Class)	24846	LOWELL AMICK, 858 TAYLORTOWN RD., MANSFIELD, OH, 44903	US Mail (1st Class)
	24846	LOWES BUSINESS ACCOUNT, 401 ELKIN HWY, NORTH WILKESBORO, NC, 28659-3456	US Mail (1st Class)
24846 LTL SYSTEMS INC., 1431 KINGSLAND AVE, SAINT LOUIS, MO, 63133-1701 US Mail (1st Class)	24846	LOWRY COMPUTER PRODUCTS, 9420 MALTBY RD, BRIGHTON, MI, 48116	US Mail (1st Class)
	24846	LTL SYSTEMS INC., 1431 KINGSLAND AVE, SAINT LOUIS, MO, 63133-1701	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	LUBRICATION ENG. INC., PO BOX 16025, ATTN: SAMANTHA/ CREDIT DEPARTMENT, WICHITA, KS, 67216-6025	US Mail (1st Class)
24846	LUCAS MOLD DESIGN, 7849 DELISLE FOURMAN, ARCANUM, OH, 45304	US Mail (1st Class)
24846	LUIS ALEJANDRO ELIZONDO MARIN, LEMAC QUALITY SUPPORT, SAN FELIPE #3805, MONTERREY N.L., 64310MEXICO	US Mail (1st Class)
24846	LUIS CARLOS HERNANDEZ, 10047 KEYSTONE DRIVE, LAKE CITY, PA, 16423	US Mail (1st Class)
24846	LUMBEE ENTERPRISES, INC., 1576 FENCORP DR., FENTON, MO, 63026	US Mail (1st Class)
24846	LYDEY AIR CONTROLS, INC., 425 TOMAHAWK DR. UNIT T, MAUMEE, OH, 43537	US Mail (1st Class)
24846	LYNETTE GERMAINE, 9344 BRISTOL, DAVISON, MI, 48423	US Mail (1st Class)
24846	LYNN BALAHON, 4239 MAPLE STREET, BROWN CITY, MI, 48416	US Mail (1st Class)
24846	LYNN MORSE, 31572 INDUSTRIAL ROAD, SUITE 300, LIVONIA, MI, 48150	US Mail (1st Class)
24846	M & I FIRST NAT'L LEASING, 250 E WISCONSIN AVE STE 1400, MILWAUKEE, WI, 53202-4219	US Mail (1st Class)
24846	M & L SUPPLY COMPANY, 987 E TALLMADGE AVE, AKRON, OH, 44310-3547	US Mail (1st Class)
24846	M F CACHET COMPANY, 14600 DETROIT AVE, LAKEWOOD, OH, 44107-4207	US Mail (1st Class)
24846	M FASCO, 16199 COMMON RD, ROSEVILLE, MI, 48066-1824	US Mail (1st Class)
24846	M G AUTOMATION & CONTROLS, 1412 EDWARDS AVE, HARAHAN, LA, 70123	US Mail (1st Class)
24846	M V SPORT, 88 SPENCE STREET, BAYSHORE, NY, 11706	US Mail (1st Class)
24846	M&B WELDING, INC., 6411 EUCLID STREET, PO BOX 162, MARLETTE, MI, 48453	US Mail (1st Class)
24846	M&E PUMP & EQUIPMENT CO., 564 OLD ST. RT. 74, CINCINNATI, OH, 45244-1512	US Mail (1st Class)
24846	M&R ELECTRIC MOTOR SERVICE, 1516 EAST FIFTH STREET, DAYTON, OH, 45403	US Mail (1st Class)
24846	M&R SERVICE, EXPEDITING & DELIVERY SERVICE, 175 BARKER RD., WHITMORE LAKE, MI, 48189	US Mail (1st Class)
24846	M&W CONTROLS, INC., 3 RIVERCHASE RDG OFC PLAZ, HOOVER, AL, 35244-2893	US Mail (1st Class)
24846	M. CONLEY CO., 1312 FOURTH ST. SE, PO BOX 21270, CANTON, OH, 44701	US Mail (1st Class)
24846	M.C. MOLDS, INC., 125 INDUSTRIAL DR., WILLIAMSTON, MI, 48895	US Mail (1st Class)
24846	M.D. MACHINING SPECIALISTS, 13435 N STATE RD., OTISVILLE, MI, 48463	US Mail (1st Class)
24846	M.J.A. MACHINE, 2268 EAST AVE., AKRON, OH, 44314	US Mail (1st Class)
24846	M.R.M. TRUCKING, 1325 LONDON RD, DULUTH, MN, 55805-2424	US Mail (1st Class)
24846	MAC EQUIPMENT INC., 7901 NW 107TH TER, KANSAS CITY, MO, 64153-1910	US Mail (1st Class)
24846	MACH 1, W BROADWAY, TEMPE, AZ, 85282	US Mail (1st Class)
24846	MACH MOLD & DIE CO., 360 URBANDALE, BENTON HARBOR, MI, 49022	US Mail (1st Class)
24846	MACH MOVERS & INSTALLATIONS, 100 ELM HILL ROAD, PO BOX 128, CAREY, OH, 43316	US Mail (1st Class)
24846	MACKINTOSH TOOL CO., 10542 SUCCESS LANE, CENTERVILLE, OH, 45458	US Mail (1st Class)
24846	MADSEN MACHINE COMPANY,INC, 500 39TH ST E, TUSCALOOSA, AL, 35405	US Mail (1st Class)
24846	MAGNATAG, 2031 O'NEILL ROAD, MACEDON, NY, 14502-8953	US Mail (1st Class)
24846	MAGNETIC CONCEPTS, 611 THIRD AVENUE, CARMEL, IN, 46032	US Mail (1st Class)
24846	MAGNUM SOLVENTS, 375 COLUMBUS DR NE, PO BOX 1041, KALKASKA, MI, 49646	US Mail (1st Class)
24846	MAHEK TROPHIES & AWARDS, 17 NORTH LONG STREET, SHELBY, OH, 44875	US Mail (1st Class)
24846	MAHONING COUNTY COURT #4, 6000 MAHONING AVENUE, #254, YOUNGSTOWN, OH, 44515	US Mail (1st Class)
24846	MAHONING VALLEY ELECTRIC SRVC., 4326 LAKE PARK RD, YOUNGSTOWN, OH, 44512	US Mail (1st Class)
24846	MAINTENANCE ENGINEERING, 3001 S UNIVERSITY DR., PO BOX 2123, FARGO, ND, 58107-2123	US Mail (1st Class)
24846	MAJOR OIL CO INC., PO BOX 11, MONTGOMERY, AL, 36101	US Mail (1st Class)
24846	MALLAK, JOSEPH, 16420 MAJESTIC OAKS, CHAGRIN FALLS, OH, 44023	US Mail (1st Class)
24846	MANAGEMENT RECRUITERS, 603-A EAST GOVERNMENT ST., PENSACOLA, FL, 32501	US Mail (1st Class)
24846	MANAGEMENT RECRUITERS/NW OHIO, 3450 W CENTRAL AVE., SUITE 360, TOLEDO, OH, 43606	US Mail (1st Class)
24846	MANAGING CHANGE INC., 2655 OAKLEY PARK RD., SUITE 101, WALLED LAKE, MI, 48390	US Mail (1st Class)
24846	MANCOMM, 315 W. 4TH STREET, DAVENPORT, IA, 52801	US Mail (1st Class)
24846	MANN FLUID POWER, 2249 N CLEVELAND MASSILLON RD, AKRON, OH, 44333-1254	US Mail (1st Class)
24846	MANPOWER, 21271 NETWORK PLACE, CHICAGO, IL, 60673-1212	US Mail (1st Class)
24846	MANPOWER INC., BOX 68-6003, MILWAUKEE, WI, 53267-6003	US Mail (1st Class)
24846	MANSFIELD FAIRFIELD INN, 1065 N. LEX. SPRINGMILL RD., MANSFIELD, OH, 44906	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	MANSFIELD FIRE CHIEF, FIRE SERVICES INC., 268 SIXTH AVE., MANSFIELD, OH, 44905	US Mail (1st Class)
24846	MANSFIELD MAINT. SERVICE, 1225 N HOME ROAD, MANSFIELD, OH, 44906	US Mail (1st Class)
24846	MANSFIELD MAINTENANCE SUPPLY, 1225 N. HOME RD., PO BOX 2633, MANSFIELD, OH, 44906	US Mail (1st Class)
24846	MANSFIELD MUNICIPAL COURT, 30 N DIAMOND ST, MANSFIELD, OH, 44902	US Mail (1st Class)
24846	MANSFIELD RUBBER STAMP, 174 SOUTH MULBERRY STREET, MANSFIELD, OH, 44902	US Mail (1st Class)
24846	MANSFIELD STRUCTRAL ERECTING, 429 PARK AVE E, MANSFIELD, OH, 44905	US Mail (1st Class)
24846	MANSFIELD WAREHOUSING & DISTR., 245 E FOURTH STREET, PO BOX 938, MANSFIELD, OH, 44901	US Mail (1st Class)
24846	MANUFACTURER'S NEWS, 1633 CENTRAL ST., EVANSTON, IL, 60201-1569	US Mail (1st Class)
24846	MANUFACTURERS GROUP, INC., 1573 WELLESLEY DR, LEXINGTON, KY, 40513	US Mail (1st Class)
24846	MANUFACTURERS' NEWS, INC., 1633 CENTRAL ST., EVANSTON, IL, 60201-7200	US Mail (1st Class)
24846	MANUFACTURING SEARCH SYSTEMS, 205 NORTH STREET, SUITE 200, CHARDON, OH, 44024	US Mail (1st Class)
24846	MANUFACTURING TECHNOLOGIES, 1525 CORPORATE WOODS PARKWAY, SUITE 500, UNIONTOWN, OH, 44685	US Mail (1st Class)
24846	MAPICS, 1000 WINDWARD CONCOURSE PARKWY, SUITE 100, ALPHARETTA, GA, 30005	US Mail (1st Class)
24846	MAPICS INC, 1000 WINDWARD CONCOURSE, STE 100, ALPHARETTA, GA, 30005	US Mail (1st Class)
24846	MAPLE SYSTEMS, 808 134TH ST SW STE 120, EVERETT, WA, 98204	US Mail (1st Class)
24846	MARATHON INDUSTRIAL, 27581 SCHOOLCRAFT, LIVONIA, MI, 48150	US Mail (1st Class)
24846	MARATHON INDUSTRIAL, 13160 WAYNE RD, LIVONIA, MI, 48150	US Mail (1st Class)
24846	MARCH OF DIMES, 1500 W THIRD AVE-STE 303, COLUMBUS, OH, 43212	US Mail (1st Class)
24846	MARCH OF DIMES, NORTH ALABAMA CHAPTER, BIRMINGHAM, AL, 35209	US Mail (1st Class)
24846	MARCONI DATA SYSTEMS, INC., ATTN: ACCOUNTS RECEIVABLE, PO BOX 93170, CHICAGO, IL, 60673-3170	US Mail (1st Class)
24846	MARGERY KIRBY, 900 S. WILEY STREET, PO BOX 320, CRESTLINE, OH, 44827-0320	US Mail (1st Class)
24846	MARIA DEL CARMEN, VARGAS FLORES, JULIA GARZA ALMAGUER, NO. 119/ COL. MUJERES ILUSTRES, SAN NICOLAS DE LOS GARZA, 66635MEXICO	US Mail (1st Class)
24846	MARIA DIAZ-MARSH, 41104 CANTON CT, BLDG. 12, CANTON, MI, 48188	US Mail (1st Class)
24846	MARIA MCKINNEY, 8707 SAMUEL BARTON DRIVE, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	MARIAH INDUSTRIES, 13125 EAST 8 MILE ROAD, WARREN, MI, 48089	US Mail (1st Class)
24846	MARIO MAZZOLA, 31572 INDUSTRIAL ROAD, SUITE 300, LIVONIA, MI, 48150	US Mail (1st Class)
24846	MARK CHANDLER, 985 FALLS CREEK DR., VANDALIA, OH, 45377-9686	US Mail (1st Class)
24846	MARK DILLEN, ADDRESS UNAVAILABLE, CHARDON, OH, 44024	US Mail (1st Class)
24846	MARK FOUR CAM INC., 22929 INDUSTRIAL DRIVE, ST. CLAIR SHORES, MI, 48080	US Mail (1st Class)
24846	MARK HERWICK, 3131 COLUMBUS RD. NE, CANTON, OH, 44705	US Mail (1st Class)
24846	MARK LIEBER, 290 MCCORMICK, LAPEER, MI, 48446	US Mail (1st Class)
24846	MARKER'S WALLY WORLD, 1212 SOUTH STATE RD 101, LIBERTY, IN, 47353	US Mail (1st Class)
24846	MARKSON LAB SALES, INC., 336 E MONTGOMERY ST, HENDERSON, NC, 27536-4618	US Mail (1st Class)
24846	MARLETTE ROOFING & SHEET METAL, 2650 NORTH VAN DYKE, MARLETTE, MI, 48453	US Mail (1st Class)
24846	MAROKO AND LANDAU, P.C., 31731 NORTHWESTERN HWY., FARMINGTON HILLS, MI, 48334-1662	US Mail (1st Class)
24846	MARON RAYMON, 8707 SAMUEL BARTON DRIVE, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	MAROON INCORPORATED, 1390 JAYCOX ROAD, AVON, OH, 44011	US Mail (1st Class)
24846	MARSHALL CAMPBELL, 2975 LAPEER RD., PO BOX 610947, PORT HURON, MI, 48061-0947	US Mail (1st Class)
24846	MARTHA S. JOINER, 15856 KENWOOD DRIVE, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	MARTY RADOCK, 610 STUART LANE APT. #3, MANSFIELD, OH, 44907	US Mail (1st Class)
24846	MARTY'S TOOL & DIE, 14824 TELEGRAPH RD., DETROIT, MI, 48239	US Mail (1st Class)
24846	MARUBENI SPECIALTY CHEMICALS, 3341 COLLECTION CENTER DR., CHICAGO, IL, 60693-3341	US Mail (1st Class)
24846	MARY BEATTY, 3560 WEST MARKET STREET, AKRON, OH, 44333	US Mail (1st Class)
24846	MARY C. MARTIN, 913 EAST 39TH STREET, ERIE, PA, 16504	US Mail (1st Class)
24846	MARY C. MARTIN, 913 EAST 30TH STREET, ERIE, PA, 16504	US Mail (1st Class)
24846	MARY DAVIS, 3560 W. MARKET STREET,	US Mail (1st Class)
24846	MARY WOISSOL, 2601 SOUTH RIDGE RD. EAST, ASHTABULA, OH, 44004	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	MASSILLON EQUIPMENT RENTAL, 2727 LINCOLN WAY EAST, MASSILLON, OH, 44646	US Mail (1st Class)
24846	MASTER MANUFACTURING CO., INC., 4703 O'HARA DRIVE, EVANSVILLE, IN, 47711	US Mail (1st Class)
24846	MASTER TOOL & MOLD, INC., 5730 WEBSTER ST., DAYTON, OH, 45414	US Mail (1st Class)
24846	MASTER UNIT DIE PRODUCTS, 853 FAIRPLAINE STREET, PO BOX 520, GREENVILLE, MI, 48838	US Mail (1st Class)
24846	MATERIALS TECH. INC., 213 LYON LN, BIRMINGHAM, AL, 35211	US Mail (1st Class)
24846	MATTHEW J. LEBEAU, 1431 MITSON BLVD, FLINT, MI, 48504	US Mail (1st Class)
24846	MATTHEW M. FRANCKIEWICZ, ARBITRATOR, 902 LINCOLN HWY., NORTH VERSAILLES, PA, 15137	US Mail (1st Class)
24846	MATTHEW MILLER, 8660 ST RTE 534, MESOPOTOMIA, OH, 44439	US Mail (1st Class)
24846	MATTHEW W LOGAN, 317 NEEPER, CAPAC, MI, 48014	US Mail (1st Class)
24846	MAVRICK METAL STAMPING INC., 309 S DANFORTH ST, MANCELONA, MI, 49659	US Mail (1st Class)
24846	MAXWELL SIGNS, 7701 OLD BIRMINGHAM HWY LOT 12, COTTONDALE, AL, 35453-0348	US Mail (1st Class)
24846	MAYER ELECTRIC, 3405 4TH AVE S, BIRMINGHAM, AL, 35222-2305	US Mail (1st Class)
24846	MAZZELLA LIFTING TECHNOLOGIES, 21000 AEROSPACE PARKWAY, CLEVELAND, OH, 44142	US Mail (1st Class)
24846	MBS CAR ERIM, 3600 GREEN CT STE 550, ANN ARBOR, MI, 48105-1570	US Mail (1st Class)
24846	MC LAREN OCCUPATIONAL, 1254 N MAIN ST, LAPEER, MI, 48446-1343	US Mail (1st Class)
24846	MC LAREN REGIONAL CENTER, 1375 N. MAIN ST, LAPEER, MI, 48446	US Mail (1st Class)
24846	M-CARE, 2301 COMMONWEALTH BLVD, ANN ARBOR, MI, 48105	US Mail (1st Class)
24846	MCCARL'S SERVICES, INC., 1253 FREEDOM RD, CRANBERRY TWP, PA, 16066-4953	US Mail (1st Class)
24846	MCCARTHY GROUP, 30 CORPORATE DRIVE, AUBURN HILLS, MI, 48326	US Mail (1st Class)
24846	MCCOLLISTER'S, 1800 RT. NORTH 130, BURLINGTON, NJ, 08016	US Mail (1st Class)
24846	MCCORMICK COFFEE COMPANY, 1541 LIBERTY ST, ERIE, PA, 16502-1645	US Mail (1st Class)
24846	MCCORMICK EQUIPMENT CO., INC., 112 NORTHEAST DR, LOVELAND, OH, 45140-7144	US Mail (1st Class)
24846	MCDJFS, 1111 S EDWIN C MOSES, PO BOX 972, DAYTON, OH, 45422	US Mail (1st Class)
24846	MCGIFFERT AND ASSOCIATES LLC, 2814 STILLMAN BLVD, TUSCALOOSA, AL, 35401-2208	US Mail (1st Class)
24846	MCGRANAHAN ASSOCIATES, 27016 KNICKERBOCKER RD., BAY VILLAGE, OH, 44140	US Mail (1st Class)
24846	MCGRAW-HILL, INC., 1221 AVENUE OF THE AMERICAS, NEW YORK, NY, 10020-1001	US Mail (1st Class)
24846	MCGUFFIN MECHANICAL, 5527 S SAGINAW, FLINT, MI, 48507	US Mail (1st Class)
24846	MCGUIREWOODS, LLP, DOMINION TOWER, 23RD FLOOR, 625 LIBERTY AVE, PITTSBURGH, PA, 15222	US Mail (1st Class)
24846	MCINTOSH & MCINTOSH, P.L.L.C., 15 E EIGHTH STREET, SUITE 30C W, CINCINNATI, OH, 45202	US Mail (1st Class)
24846	MCKEE MARKETING, 180A COMMERCE DR, PELHAM, AL, 35124-1838	US Mail (1st Class)
24846	MCKELVEY MECHANICAL, INC., 5240 KAULOOSA AVE, TUSCALOOSA, AL, 35405	US Mail (1st Class)
24846	MCLEAN MIDWEST CORP., DEPT. CH 14197, PALATINE, IL, 60055-4197	US Mail (1st Class)
24846	MCMASTER-CARR SUPPLY CO., 6100 FULTON INDUSTRIAL BLVD, ATLANTA, GA, 30336	US Mail (1st Class)
24846	MCNAUGHTON-MCKAY ELECTRIC, DEPT 70901, PO BOX 67000, DETROIT, MI, 48267-0709	US Mail (1st Class)
24846	MCNAUGHTON-MCKAY ELECTRIC CO., DEPT. 14801, PO BOX 67000, DETROIT, MI, 48267-0148	US Mail (1st Class)
24846	MCNEES WALLACE & NURICK LLC, 100 PINE STREET, HARRISBURG, PA, 17108	US Mail (1st Class)
24846	MCNICHOLS COMPANY, BOX 101211, ATLANTA, GA, 30392-1211	US Mail (1st Class)
24846	MDR, 70 ROSS RD., ASHTABULA, OH, 44004	US Mail (1st Class)
24846	MED GROUP, 8555 SWEET VALLEY DR, SUITE A, VALLEY VIEW, OH, 44125	US Mail (1st Class)
24846	MEDCENTRAL HEALTH SYS/WORKABLE, 335 GLESSNER AVE, MANSFIELD, OH, 44903-2269	US Mail (1st Class)
24846	MEDCENTRAL HEALTH SYSTEM, ATTENTION: CASHIERS, 335 GLESSNER AVENUE, MANSFIELD, OH, 44903	US Mail (1st Class)
24846	MEDCENTRAL HEALTH-CRESTLINE, 291 HEISER COURT, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	MEDIA CAMPAIGN SERVICES, PO BOX 46065, CHICAGO, IL, 60646	US Mail (1st Class)
24846	MEDICAL MUTUAL, FIFTH THIRD BANK, PO BOX 632204, CINCINNATI, OH, 45263-2204	US Mail (1st Class)
24846	MEDICAL MUTUAL, 2060 E 9TH ST, CLEVELAND, OH, 44115	US Mail (1st Class)
24846	MEDICAL MUTUAL OF OHIO, 2060 E 9TH ST, CLEVELAND, OH, 44115-1313	US Mail (1st Class)
24846	MEECH STATIC ELIMINATORS USA, 2915 NEWPARK DRIVE, NORTON, OH, 44203	US Mail (1st Class)
24846	MEGA FLUIDLINE PRODUCTS, INC., 425 KENNEDY RD., AKRON, OH, 44305	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	MEGA VENT QUEBEC INC., 114, RUE POINTE-LANGLOIS, LAVAL, QC, H7L 3M5CANADA	US Mail (1st Class)
24846	MEIJER, ATTN: GIFT CARDS - 982, PO BOX 2102, GRAND RAPIDS, MI, 49501-2102	US Mail (1st Class)
24846	MEIJER, INC., 155 DEMILLE ROAD, LAPEER, MI, 48446	US Mail (1st Class)
24846	MEIJER, INC., 1355 N LEXINGTON-SPRINGMILL, MANSFIELD, OH, 44906	US Mail (1st Class)
24846	MEL WACKER SIGNS, INC., 13076 BARRS SW, MASSILLON, OH, 44647	US Mail (1st Class)
24846	MELCO ENGRAVING, INC., 1809 ROCHESTER INDUSTRIAL DR., ROCHESTER HILLS, MI, 48309	US Mail (1st Class)
24846	MELISSA G. YOUNG, 21 THOMAS DR., MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	MELISSA IRWIN, 3131 COLUMBUS RD NE, CANTON, OH, 44705	US Mail (1st Class)
24846	MELMOR ASSOCIATES, INC., 840 ANN STREET, PO BOX 511, NILES, OH, 44446	US Mail (1st Class)
24846	MEMINDEX, 149 CARTER STREET, ROCHESTER, NY, 14602	US Mail (1st Class)
24846	MENS INDUSTRIAL GOLF LEAGUE, 13404 OLD STATE ROAD, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	MERCHANTS & MEDICAL CREDIT CORP., 6324 TAYLOR DRIVE, FLINT, MI, 48507-4685	US Mail (1st Class)
24846	MERCURY EXPRESS, INC., 14801 AUBURN RD, NEWBURY, OH, 44065-9757	US Mail (1st Class)
24846	MERCY BUSINESS HEALTH, 2540 16TH STREET, PORT HURON, MI, 48060	US Mail (1st Class)
24846	MEREDITH-SPRINGFIELD, 321 MOODY STREET, LUDLOW, MA, 01056	US Mail (1st Class)
24846	MES CO., PO BOX 387, FLINT, MI, 48501-0387	US Mail (1st Class)
24846	METAL & WIRE PRODUCTS CO., 1065 SALEM PKWY., SALEM, OH, 44460	US Mail (1st Class)
24846	METAL CLADDING INC., 230 S NIAGARA ST, LOCKPORT, NY, 14094-1927	US Mail (1st Class)
24846	METRICS UNLIMITED, INC, 6709 CHICAGO ROAD, WARREN, MI, 48092	US Mail (1st Class)
24846	METRO BUILDING SERVICES, 31106 STEPHEN AVE, WESTLAND, MI, 48185-1637	US Mail (1st Class)
24846	METRO MECHANICAL SERVICES, 851 N PINE HILL RD., BIRMINGHAM, AL, 35217	US Mail (1st Class)
24846	METROLOGY SUPPORT GROUP, 9210 WYOMING AVE. N, SUITE 215, BROOKLYN PARK, MN, 55445	US Mail (1st Class)
24846	METROPOLITAN HEATING & COOLING, INC, 29010 WEST 8 MILE RD., FARMINGTON HILLS, MI, 48336	US Mail (1st Class)
24846	METTLER-TOLEDO, INC., 1900 POLARIS PKWY, COLUMBUS, OH, 43240-4035	US Mail (1st Class)
24846	MEXICAN CONSULATE, ADDRESS UNAVAILABLE,	US Mail (1st Class)
24846	MFS PLUS, INC., 4560 INDIANWOOD, CLARKSTON, MI, 48348	US Mail (1st Class)
24846	MGA RESEARCH CORP, BOX 68-9916, MILWAUKEE, WI, 53268-9916	US Mail (1st Class)
24846	MH CONTAINERS & EQUIP, POBOX 808, WILLOUGHBY, OH, 44096	US Mail (1st Class)
24846	MH IND. AND MONETREX, DEPARTMENT 77090, PO BOX 77000, DETROIT, MI, 48277-0090	US Mail (1st Class)
24846	MHI INJECTION MOLDING MACHINERY INC, 520 THOMAS DR, BENSENVILLE, IL, 60106-1621	US Mail (1st Class)
24846	MIAMI INDUSTRIAL TRUCKS, INC., 2830 E RIVER RD, DAYTON, OH, 45439	US Mail (1st Class)
24846	MICH.COM, INC., 10 W. HURON, PONTIAC, MI, 48342	US Mail (1st Class)
24846	MICHAEL A. DIANTONIO, 3131 COLUMBUS RD. NE, CANTON, OH, 44705	US Mail (1st Class)
24846	MICHAEL A. SELBY, 61 LAKEWGREN DR, EATON, OH, 45320	US Mail (1st Class)
24846	MICHAEL BENZA ASSOCIATES, CONSULTING ENGR & SURVEYOR, 6860 WEST SNOWVILLE RD., BRECKSVILLE, OH, 44141	US Mail (1st Class)
24846	MICHAEL BOARDWINE, 3560 WEST MARKET STREET, SUITE 340, AKRON, OH, 44333	US Mail (1st Class)
24846	MICHAEL CUMMINGS, 4602 AMHERST, ERIE, PA, 16506	US Mail (1st Class)
24846	MICHAEL D. NUNLEE, 34314 MCBRIDE, ROMULUS, MI, 48174	US Mail (1st Class)
24846	MICHAEL DAY ENTERPRISES INC., 960 SEVILLE RD, WADSWORTH, OH, 44281	US Mail (1st Class)
24846	MICHAEL DIMEGLIO, 17187 N LAUREL PK DR, SUITE #208, LIVONIA, MI, 48152	US Mail (1st Class)
24846	MICHAEL DOSIAK, 1028 WOODVILLE RD, MANSFIELD, OH, 44907	US Mail (1st Class)
24846	MICHAEL DOSIAK/, AMERICAN EXPRESS, 1028 WOODVILLE RD, MANSFIELD, OH, 44907	US Mail (1st Class)
24846	MICHAEL GENSBURG, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	MICHAEL J. WHITE, 11540 JOE NAMETH ROAD, NORTHPORT, AL, 35475	US Mail (1st Class)
24846	MICHAEL L HARVARD, 18490 ASBURY PARK, DETROIT, MI, 48235	US Mail (1st Class)
24846	MICHAEL LIBERATORE, 8707 SAMUEL BARTON DR., BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	MICHAEL LORENZ, 8707 SAMUEL BARTON DR., BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	MICHAEL P. HART, 59 RT. 193, JEFFERSON, OH, 44047	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	MICHAEL RENCH, 985 FALLS CREEK DR., VANDALIA, OH, 45377	US Mail (1st Class)
24846	MICHAEL ROSE, 567 STATE RTE. 314 N., MANSFIELD, OH, 44903	US Mail (1st Class)
24846	MICHELLE AYRIS, 1142 WILTSHIRE DRIVE, LAPEER, MI, 48446	US Mail (1st Class)
24846	MICHELLE PISCOPO, 9719 CHARDON RD, CHARDON, OH, 44024	US Mail (1st Class)
24846	MICHIGAN DEQ, STORAGE TANK DIVISION, PO BOX 30157, LANSING, MI, 48909-7657	US Mail (1st Class)
24846	MICHIGAN DOOR SYSTEMS, 34355 GROESBECK HWY, CLINTON TWP, MI, 48035-3359	US Mail (1st Class)
24846	MICHIGAN MECHANICAL SERVICE, 25445 BREST ROAD, TAYLOR, MI, 48180	US Mail (1st Class)
24846	MICHIGAN METAL BLDG. ERECTORS, 6630 ROOSEVELT, ALLEN PARK, MI, 48101	US Mail (1st Class)
24846	MICHIGAN MINORITY BUSINESS, DEVELOPMENT COUNCIL, 3011 W GRAND BLVD., DETROIT, MI, 48202-3011	US Mail (1st Class)
24846	MICHIGAN PALLET INC., 1225 N SAGINAW ST, SAINT CHARLES, MI, 48655	US Mail (1st Class)
24846	MICHIGAN POLYMER RECLAIM, 107 E. WALKER RD., ST. JOHNS, MI, 48879	US Mail (1st Class)
24846	MICHIGAN RUBBER PRODUCTS, DEPT. CH 17167, PALATINE, IL, 60055-7167	US Mail (1st Class)
24846	MICHIGAN RUBBER PRODUCTS, DEPT. CH-17167, PALATINE, IL, 60055-7167	US Mail (1st Class)
24846	MICHIGAN TESTING INSTITUTE INC, 42818 MOUND ROAD, STERLING HGTS, MI, 48314	US Mail (1st Class)
24846	MICHIGAN TESTING INSTITUTE INC, 42818 MOUND RD, STERLING HIGHTS, MI, 48314	US Mail (1st Class)
24846	MICRO CENTER COMPUTER EDU, 1349 SOM CENTER ROAD, MAYFIELD HTS, OH, 44124	US Mail (1st Class)
24846	MICRO GROUP, 7 INDUSTRIAL PARK RD, MEDWAY, MA, 02053-1750	US Mail (1st Class)
24846	MICRO LABORATORIES, 7158 INDUSTRIAL PARK BLVD, MENTOR, OH, 44060	US Mail (1st Class)
24846	MICRO MOULES, 5195 RUE ALBERT MILLICHAMP, ST. HUBERT, QC, J3Y 8Z8CANADA	US Mail (1st Class)
24846	MICRO PLASTICS INC., 11 INDUSTRY LANE, FLIPPIN, AR, 72634	US Mail (1st Class)
24846	MICRO WAREHOUSE, 7077 COLLECTION CENTER DRIVE, CHICAGO, IL, 60693-0072	US Mail (1st Class)
24846	MICRODRY, THE NEMETH GROUP, 5901 WEST HIGHWAY 22, CRESTWOOD, KY, 40014	US Mail (1st Class)
24846	MICROWAVE ENGINEERING SERVICES CORP, 52 CHEESE SPRING RD., WILTON, CT, 06897	US Mail (1st Class)
24846	MID STATE, PO BOX 931900, CLEVELAND, OH, 44193	US Mail (1st Class)
24846	MID STATES BOLT & SCREW, 4126 SOMERS DR, BURTON, MI, 48529-2279	US Mail (1st Class)
24846	MIDAS MUFFLER & BRAKE SHOP, 216 CHERRY AVE., CHARDON, OH, 44024	US Mail (1st Class)
24846	MIDDLEFIELD CHAMBER/COMM, 14979 S. STATE AVE., PO BOX 801, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	MIDDLEFIELD SIGN COMPANY, 14895 N STATE AVE, MIDDLEFIELD, OH, 44062-9747	US Mail (1st Class)
24846	MIDDLEFIELD WATER DEPT., ACCT: 9010021300-103 9010021800-103, 14860 N STATE AVE, MIDDLEFIELD, OH, 44062-9747	US Mail (1st Class)
24846	MIDLAND CONSULTANTS, INC., 4401 ROCKSIDE RD., STE 214, INDEPENDENCE, OH, 44131	US Mail (1st Class)
24846	MID-OHIO GRAVELY, INC., 1139 LUCAS ROAD, MANSFIELD, OH, 44905	US Mail (1st Class)
24846	MIDRANGE SOFTWARE, 12716 RIVERSIDE DR., STUDIO CITY, CA, 91607	US Mail (1st Class)
24846	MID-STATES RUBBER PROD., 1230 RACE ST., PO BOX 370, PRINCETON, IN, 47670	US Mail (1st Class)
24846	MIDWEST KNIFE GRINDING, 492 ELM RIDGE AVE, UNIT 4, CANAL FULTON, OH, 44614	US Mail (1st Class)
24846	MIDWEST RETAIL SERVICES INC., 8077 FAIRWAY DRIVE, COLUMBUS, OH, 43235	US Mail (1st Class)
24846	MIDWEST SOCKET SCREW, 12627 FARLEY, DETROIT, MI, 48239	US Mail (1st Class)
24846	MIG PLASTICS INC., DEPARTMENT #200401, PO BOX 67000, DETROIT, MI, 48267-2004	US Mail (1st Class)
24846	MIKA TRADE & MARKETING, 4450 CRYSTAL PKWY, KENT, OH, 44240-8006	US Mail (1st Class)
24846	MIKE CINDIA SALES, INC., 7244 HARTMAN RD., WADSWORTH, OH, 44281	US Mail (1st Class)
24846	MIKE CUMMINGS, 4602 AMHERST RD, ERIE, PA, 16506	US Mail (1st Class)
24846	MIKE DADA, 189 W NEWARK, LAPEER, MI, 48446	US Mail (1st Class)
24846	MIKE HOSTETTLER, 3131 COLUMBUS RD., N.E., CANTON, OH, 44705	US Mail (1st Class)
24846	MIKE KEMP, TUSCALOOSA PLANT, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	MIKE SEIBEL, 1350 WALNUT ST APT 6, BARABOO, WI, 53913-2975	US Mail (1st Class)
24846	MILACRON MARKETING CO., LOCKBOX 1497, SOLUTIONS CENTER, CHICAGO, IL, 60677-1004	US Mail (1st Class)
24846	MILACRON MARKETING COMPANY, PLASTIC TECHNOLOGY GROUP, PO BOX 740440, ATLANTA, GA, 30374-0440	US Mail (1st Class)
24846	MILACRON MARKETING COMPANY, 4701 MARBURG AVE, CINCINNATI, OH, 45209-1025	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	MILACRON MARKETING CO-SERVTEK, 2090 FLORENCE AVE, CINCINNATI, OH, 45206-2484	US Mail (1st Class)
24846	MILDRED FREEMAN, 713 GIBBS AVE. NE, CANTON, OH, 44704	US Mail (1st Class)
24846	MILL VILLAGE VOLUNTEER FIRE DEPT., PO BOX 6L, MILL VILLAGE, PA, 16427	US Mail (1st Class)
24846	MILLCREEK TOWNSHIP, WATER AUTHORITY, 3608 WEST 26TH STREET, ERIE, PA, 16506-2037	US Mail (1st Class)
24846	MILLCREEK TOWNSHIP SEWER, 3608 W 26TH ST, ERIE, PA, 16506-2059	US Mail (1st Class)
24846	MILLENNIUM EXPEDITED, 501 N DIXIE DRIVE, VANDALIA, OH, 45377	US Mail (1st Class)
24846	MILLENNIUM TECHNICAL SERVICE, 4046 ONWAY RD, INDIAN RIVER, MI, 49749	US Mail (1st Class)
24846	MILLER ENGINEERING GROUP INC., 4738 GATEWAY CIR # A100, KETTERING, OH, 45440-1716	US Mail (1st Class)
24846	MILLER MEATS, 1335 S. TRIMBLE ROAD, MANSFIELD, OH, 44907	US Mail (1st Class)
24846	MILLER TRUCK & STORAGE CO, 420 INGHAM ST., JACKSON, MI, 49201	US Mail (1st Class)
24846	MILWAUKEE ELECTRIC TOOL CORP., 4844 COLLECTION CENTER, CHICAGO, IL, 60693	US Mail (1st Class)
24846	MINARIK CORPORATION, CREDIT DEPARTMENT, PO BOX 25033, GLENDALE, CA, 91221	US Mail (1st Class)
24846	MINITAB, INC, 1829 PINE HALL ROAD, STATE COLLEGE, PA, 16801-3008	US Mail (1st Class)
24846	MINNESOTA RUBBER, SDS 10-0130, PO BOX 86, MINNEAPOLIS, MN, 55486-0130	US Mail (1st Class)
24846	MINNIE ZORNES, 81 FOX RUN ROAD, HILLSBORO, KY, 41049	US Mail (1st Class)
24846	MINUTE MAN UPHOLSTERY SUPPLY, 1905 SOUTH ELM STREET, HIGH POINT, NC, 27260	US Mail (1st Class)
24846	MIRA TIDWELL, 1401 INDUSTRIAL PARK DRIVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	MIRAGLIA`S FINE FOOD &, CATERING, 13488 RAVENNA ROAD, CHARDON, OH, 44024	US Mail (1st Class)
24846	MIRSAD ALIC, 1808A BUFFALO ROAD A, APT #3, ERIE, PA, 16510	US Mail (1st Class)
24846	MISC. NO-COST, ADDRESS UNAVAILABLE,	US Mail (1st Class)
24846	MISTY D. BUCKINGHAM, 537 WASHINGTON TWIN, EATON, OH, 45320	US Mail (1st Class)
24846	MITCH BROWN, 4853 S. BALDWIN ROAD, LANGSBURG, MI, 48848	US Mail (1st Class)
24846	MITUTOYO AMERICA CORPORATION, DEPT. CH 17053, PALATINE, IL, 60055-7053	US Mail (1st Class)
24846	MITUTOYO/MTI METROLOGY, CENTER, 45001 5 MILE RD., PLYMOUTH, MI, 48170-2810	US Mail (1st Class)
24846	MM&J ACCOUNTING, 9400 W FOSTER AVE STE 204, CHICAGO, IL, 60656-1571	US Mail (1st Class)
24846	MMBDC, 3011 W GRAND BLVD STE 230, DETROIT, MI, 48202	US Mail (1st Class)
24846	MMTC, 47911 HALYARD, PLYMOUTH, MI, 48170	US Mail (1st Class)
24846	MOBILE COMMUNICATIONS, SALES INC., 15075 WEST 11 MILE RD., ROSEVILLE, MI, 48066	US Mail (1st Class)
24846	MOBILE HANDLING SYSTEMS,, 260 W. PROSPECT AVE., PAINESVILLE, OH, 44077	US Mail (1st Class)
24846	MOBILE OCCUPATIONAL HEALTH, 189 WEST NEWARK, LAPEER, MI, 48446	US Mail (1st Class)
24846	MODERN ENGINEERING, PNC BANK, NA, HIGHPOINT BUSINESS CTR - BOX 772577, 160 HANSON COURT, WOOD DALE, IL, 60191	US Mail (1st Class)
24846	MODERN INDUSTRIES, 613 WEST 11TH STREET, PO BOX 399, ERIE, PA, 16512	US Mail (1st Class)
24846	MODERN METAL PRODUCTS CO., 35053 EAGLE WAY, CHICAGO, IL, 60678-1350	US Mail (1st Class)
24846	MODERN PLASTICS CORP., 489 NORTH SHORE DRIVE, PO BOX 1367, BENTON HARBOR, MI, 49023-1367	US Mail (1st Class)
24846	MOHR PARTNERS, INC, 15305 DALLAS PARTWAY, SUITE 710, ADDISON, TX, 75001	US Mail (1st Class)
24846	MOLD & DIE TECHNOLOGY, 820 DUELL ROAD, TRAVERSE CITY, MI, 49686	US Mail (1st Class)
24846	MOLD & MACHINE SERVICES, 2120 WARFEL AVENUE, ERIE, PA, 16503	US Mail (1st Class)
24846	MOLD MASTERS SYS. INC., 233 ARMSTRONG AVE, GEORGETOWN, ON, L7G 4X5CANADA	US Mail (1st Class)
24846	MOLD SPECIALTIES, INC., 51232 ORO DRIVE, SHELBY TOWNSHIP, MI, 48315	US Mail (1st Class)
24846	MOLD TECH, INC., 1900 COMMERCE PARKWAY, LANCASTER, NY, 14086-1735	US Mail (1st Class)
24846	MOLDERS CHOICE, INC., 27000 RICHMOND RD, UNIT #9, SOLON, OH, 44139	US Mail (1st Class)
24846	MOLDERS SERVICES, 646 AJAX DRIVE, MADISON HEIGHTS, MI, 48071	US Mail (1st Class)
24846	MOLDERS WORLD INC., 6101 CHERRY LANE FARM, WEST CHESTER, OH, 45069	US Mail (1st Class)
24846	MOLDPRO, 4042 MAINWAY DRIVE, BURLINGTON, ON, L7M 4B9CANADA	US Mail (1st Class)
24846	MOLDSHOP TOOLS.COM, 820 DUELL ROAD, TRAVERSE CITY, MI, 49686	US Mail (1st Class)
24846	MOLD-TECH, BOX 101045, ATLANTA, GA, 30392-1045	US Mail (1st Class)
24846	MOLD-TECH (MI), 34497 KELLY RD, FRASER, MI, 48026-3404	US Mail (1st Class)
24846	MOLDTECH, INC., 1900 COMMERCE PKWY., LANCASTER, NY, 14086-1735	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	MONITOR TECHNOLOGIES LLC, 44W320 KESLINGER RD, PO BOX 8048, ELBURN, IL, 60119-8048	US Mail (1st Class)
24846	MONROE AUTO EQUIPMENT, 1 INTERNATIONAL DR, MONROE, MI, 48161-9345	US Mail (1st Class)
24846	MONROE VINYL GRAPHICS INC., 2784 N MONROE ST., MONROE, MI, 48162	US Mail (1st Class)
24846	MONSTER, INC., PO BOX 90364, CHICAGO, IL, 60696-0364	US Mail (1st Class)
24846	MONTGOMERY CTY COURT AREA ONE, CIVIL DEPARTMENT, 3100 SHILOH SPRINGS ROAD, TROTWOOD, OH, 45426	US Mail (1st Class)
24846	MONTREX, 53 ELLIOTT POWER DRIVE, LEXINGTON, TN, 38351	US Mail (1st Class)
24846	MOOG, INC., 300 JAMISON RD, EAST AURORA, NY, 14052	US Mail (1st Class)
24846	MOORE MEDICAL CORP., 389 JOHN DOWNEY DR, NEW BRITAIN, CT, 06051-2924	US Mail (1st Class)
24846	MOORE'S BRANCH #267, 2903 PARK AVE., WEST, MANSFIELD, OH, 44906	US Mail (1st Class)
24846	MORELAND INDUSTRIES, 5135 RICHMOND RD, CLEVELAND, OH, 44146-1330	US Mail (1st Class)
24846	MORHEAT, 63 GALAXY BLVD #11, TORONTO, ON, M9W 1P7CANADA	US Mail (1st Class)
24846	MORLOCK MATERIAL HANDLING, 617 S MICHIGAN AVE, EDGERTON, OH, 43517-9367	US Mail (1st Class)
24846	MORRELL INCORPORATED, DEPT. 20301, PO BOX 67000, DETROIT, MI, 48267-0203	US Mail (1st Class)
24846	MORRIS COUPLING COMPANY, 2240 W 15TH ST, ERIE, PA, 16505-4510	US Mail (1st Class)
24846	MORROW CTY. SHERIFF'S SPEC. UNIT, 101 HOME RD, MOUNT GILEAD, OH, 43338	US Mail (1st Class)
24846	MOSIER FLUID POWER OF OH, 2475 TECHNICAL DRIVE, MIAMISBURG, OH, 45342	US Mail (1st Class)
24846	MOTION INDUSTRIES, 3345 E. WASHINGTON, SAGINAW, MI, 48601	US Mail (1st Class)
24846	MOTION INDUSTRIES, 1605 ALTON RD, BIRMINGHAM, AL, 35210-3770	US Mail (1st Class)
24846	MOTION VIDEO PRODUCTIONS, 904 CANDLEWOOD DR, LEAGUE CITY, TX, 77573-6105	US Mail (1st Class)
24846	MOTOMAN TECHNICAL EDU, 805 LIBERTY LANE, WEST CARROLLTON, OH, 45449	US Mail (1st Class)
24846	MOTOMAN, INC., 805 LIBERTY LN, WEST CARROLLTON, OH, 45449-2158	US Mail (1st Class)
24846	MOTOR CITY FASTENER, INC., 1600 EAST TEN MILE RD., PO BOX 219, HAZEL PARK, MI, 48030-0219	US Mail (1st Class)
24846	MOTOR COACH IND-ROSWELL, 72 EARL CUMMINGS BLVD W, ROSWELL, NM, 88203-6150	US Mail (1st Class)
24846	MOTOR SYSTEMS INC., 501 TECHNICAL CENTER, MILFORD, OH, 45150	US Mail (1st Class)
24846	MOULD-TEK INDUSTRIES, INC, 77 NANTUCKET BLVD, SCARBOROUGH, ON, M1P 2N5CANADA	US Mail (1st Class)
24846	MOVING COST, INC., 1700 NW 64TH STREET, SUITE 400, FT. LAUDERDALE, FL, 33309	US Mail (1st Class)
24846	MPC INC., 835 CANTERBURY ROAD, WESTLAKE, OH, 44143	US Mail (1st Class)
24846	MPI LABEL SYSTEMS, 916 AVENUE N, GRAND PRAIRIE, TX, 75050	US Mail (1st Class)
24846	MQR, 101 E FLAGET ST, BARDSTOWN, KY, 40004-1517	US Mail (1st Class)
24846	MRA INDUSTRIAL MAINTENANCE, 4333 NYE RD., WEST SPRINGFIELD, PA, 16443	US Mail (1st Class)
24846	MRM INCORP, 22660 HESLIP DRIVE, NOVI, MI, 48375	US Mail (1st Class)
24846	MSA SYSTEMS, 2002 FILMORE AVE, ERIE, PA, 16506	US Mail (1st Class)
24846	MSC INDUSTRIAL SUPPLY, DEPT C H 0075, PALATINE, IL, 60055-0075	US Mail (1st Class)
24846	MSC INDUSTRIAL SUPPLY CO, DEPT. CH 0075, PALATINE, IL, 60055-0075	US Mail (1st Class)
24846	MSC INDUSTRIAL SUPPLY CO. INC., DEPT. CH 0075, PALATINE, IL, 60055-0075	US Mail (1st Class)
24846	MSI VIKING GAGE, LLC., BRANCH 10 SERVICES, PO BOX 537, DUNCAN, SC, 29334	US Mail (1st Class)
24846	MT. BUSINESS TECHNOLOGIES, 1150 NATIONAL PKWY, MANSFIELD, OH, 44906-1911	US Mail (1st Class)
24846	MTM DE MEXICO, ATTN: ALEX MUNOZ/MTM, STATE NATIONAL BANK, 2110 YARBROUGH, EL PASO, TX, 79925	US Mail (1st Class)
24846	MULTI SERVICES INPEX INC., 18617 RUE VENNE, PIERREFONDS,, MONTREAL, QC, H9K 1K7CANADA	US Mail (1st Class)
24846	MULTI-PLASTICS, INC., SOUTH STREET, ROUTE 198, SAEGERTOWN, PA, 16433	US Mail (1st Class)
24846	MURRU, ANTHONY, 17584 LAKESEDGE TRL, CHAGRIN FALLS, OH, 44023-2116	US Mail (1st Class)
24846	MUSCULAR DYSTROPHY ASSOC., 7530 LUCERNE DRIVE. # 210, CLEVELAND, OH, 44130	US Mail (1st Class)
24846	MUSCULAR DYSTROPHY ASSOC., 700 CENTURY PARK SOUTH, SUITE 129, BIRMINGHAM, AL, 35226	US Mail (1st Class)
24846	MUSCULAR DYSTROPHY ASSOC., C/O FIFTH THIRD BANK, PO BOX 716, MAUMEE, OH, 43537	US Mail (1st Class)
24846	MUSCULAR DYSTROPHY ASSOCIATION, C/O FIFTH THIRD BANK, G-5030 S. SAGINAW, FLINT, MI, 48507	US Mail (1st Class)
24846	MVHRA, THE MANDALAY CENTER, 140 E MONUMENT AVENUE, DAYTON, OH, 45402	US Mail (1st Class)
24846	MYRNA RUSSELL, 100 SEVENTH AVE, SUITE 100, CHARDON, OH, 44024	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	MYRON CORP., 205 MAYWOOD AVE, MAYWOOD, NJ, 07607-1007	US Mail (1st Class)
24846	MYRON MANUFACTURING CORP., 205 MAYWOOD AVE, MAYWOOD, NJ, 07607-1007	US Mail (1st Class)
24846	MYSQL AB, 20450 STEVENS CREEK BLVD STE 350, CUPERTINO, CA, 95014-6813	US Mail (1st Class)
24846	N.L. CONSTRUCTION CORP., 230 15TH STREET S.E., CANTON, OH, 44707	US Mail (1st Class)
24846	N.O.R.G., C/O H MUEHLSTEIN & CO., ATTN: DON BREECE, AKRON, OH, 44333	US Mail (1st Class)
24846	N.W.P. MANUFACTURING, 23546 US ST RT 224, ALVADA, OH, 44802	US Mail (1st Class)
24846	NAMISCO INC., 1220 BARRANCA DR., SUITE 5AA, EL PASO, TX, 79935	US Mail (1st Class)
24846	NANCY A ADAMS, 9101 STATE ROUTE 303, WINDHAM, OH, 44288	US Mail (1st Class)
24846	NANCY COSTELLO, 3304 32ND AVE E, TUSCALOOSA, AL, 35405	US Mail (1st Class)
24846	NANCY HUNT, 12631 PRINCETON RD, HUNTSBURG, OH, 44046-9794	US Mail (1st Class)
24846	NANCY J. KRESS, 3131 COLUMBUS ROAD NE, CANTON, OH, 44705	US Mail (1st Class)
24846	NANCY MACBRIEN, 483 PLEASANT RIDGE DR., CANTON, MI, 48188	US Mail (1st Class)
24846	NAPA AUTO PARTS, 2999 CIR 75 PKWY, ATLANTA, GA, 30339-3050	US Mail (1st Class)
24846	NAPM OF NORTH CENTRAL OHIO, INTERNATIONAL SYSTEMS MARKETING INC, 611 ROCKVILLE PIKE STE 120, ROCKVILLE, MD, 20852-1178	US Mail (1st Class)
24846	NAS RECRUITMENT COMMUNICATIONS, 1 INFINITY CORPORATE CENTRE DR, CLEVELAND, OH, 44125-5369	US Mail (1st Class)
24846	NAS RECRUITMENT COMMUNICATIONS, INC, 1 INFINITY CORPORATE CENTRE DR, CLEVELAND, OH, 44125-5369	US Mail (1st Class)
24846	NASP PROFESSIONALS, 106-242 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2143	US Mail (1st Class)
24846	NAT'L EXTRUDED SHEET CLAMPING, 1391 WEST M61, GLADWIN, MI, 48624	US Mail (1st Class)
24846	NAT'L PLUS PLAN-PENSION, 6 BLACKSTONE VALLEY PLACE, SUITE 302, LINCOLN, RI, 02865	US Mail (1st Class)
24846	NATHAN STONEBURGH, 290 MCCORMICK RD., LAPEER, MI, 48446	US Mail (1st Class)
24846	NATION GRINDING, INC., 2061 WEBSTER STREET, DAYTON, OH, 45404	US Mail (1st Class)
24846	NATIONAL ASSOC OF PURCHASING, INTERNATIONAL SYSTEMS MARKETING INC, 611 ROCKVILLE PIKE STE 120, ROCKVILLE, MD, 20852-1178	US Mail (1st Class)
24846	NATIONAL BLACK REVIEW, 7215 LINDLEY AVE., RESEDA, CA, 91335	US Mail (1st Class)
24846	NATIONAL CALIBRATION &, TESTING LABORATORIES, 6960 MADISON AVE WEST,STE 11, GOLDEN VALLEY, MN, 55427	US Mail (1st Class)
24846	NATIONAL COMPUTER ENTERPRISES, 439 SOUTH MAIN STREET, SUITE 203, ROCHESTER, MI, 48307	US Mail (1st Class)
24846	NATIONAL FUEL, 6363 MAIN ST, WILLIAMSVILLE, NY, 14221-5855	US Mail (1st Class)
24846	NATIONAL HEAT EXCHANGE CLEAN., 156 NULF DRIVE, COLUMBIANA, OH, 44408	US Mail (1st Class)
24846	NATIONAL INSTITUTE OF, BUSINESS MANAGEMENT, PO BOX 9206, MCLEAN, VA, 22102-0206	US Mail (1st Class)
24846	NATIONAL PAPER, PO BOX 643784, PITTSBURGH, PA, 15264-3784	US Mail (1st Class)
24846	NATIONAL SEMINARS GROUP, 1016 AUTUMN OAKS DR, IMPERIAL, MO, 63052-3452	US Mail (1st Class)
24846	NATIONAL STARCH & CHEMICAL, 10 FINDERNE AVE, BRIDGEWATER, NJ, 08807-3355	US Mail (1st Class)
24846	NATIONAL TECHNOLOGY, TRANSFER, INC., PO BOX 4558, ENGLEWOOD, CO, 80155-4558	US Mail (1st Class)
24846	NATIONAL TOOL & MFG., 100-124 NORTH 12 STREET, KENILWORTH, NJ, 07033	US Mail (1st Class)
24846	NATIONAL UNION/AIGRM BUSINESS, 175 WATER ST 11TH FL, NEW YORK, NY, 10038-4918	US Mail (1st Class)
24846	NATIONAL WELDERS, SUPPLY CO., INC., 1486 BRIDGE STREET, ASHTABULA, OH, 44004	US Mail (1st Class)
24846	NATIONAL WILDLIFE FEDERATION, 11100 WILDLIFE CENTER DR, RESTON, VA, 20190-5361	US Mail (1st Class)
24846	NAVARRE TRAILER SALES INC., 4633 ERIE AVENUE, SW, NAVARRE, OH, 44660	US Mail (1st Class)
24846	NAVIGATING BUSINESS SPACE, 2595 BELLINGHAM, TROY, MI, 48283	US Mail (1st Class)
24846	NBS TRUCKING LTD, SUSQUEHANNA RD, CUSTER CITY, PA, 16725	US Mail (1st Class)
24846	NBW, INC., 4556 INDUSTRIAL PKWY, CLEVELAND, OH, 44135-4542	US Mail (1st Class)
24846	NC STATE COLLEGE FOUNDATION, 2441 KENWOOD CIR, MANSFIELD, OH, 44906-1546	US Mail (1st Class)
24846	NCC ELECTRONICS, 6471 COMMERCE DRIVE, WESTLAND, MI, 48185	US Mail (1st Class)
24846	NCS PEARSON, INC, 4484 COLLECTIONS CENTER DR., CHICAGO, IL, 60693	US Mail (1st Class)
24846	NCS PEARSON, INC., 4484 COLLECTIONS CENTER DRIVE, CHICAGO, IL, 60693	US Mail (1st Class)
24846	NEFF ENGINEERING COMPANY INC., 7114 INNOVATION BLVD, FORT WAYNE, IN, 46818-1373	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	NEIGHBOR HELPING NEIGHBOR INC., GOVERNOR`S TORNADO RELIEF FUND, THE TUSCALOOSA NEWS, TUSCALOOSA, AL, 35402	US Mail (1st Class)
24846	NELMOR CO INC, 100 RODDY AVE., SOUTH ATTLEBORO, MA, 02703	US Mail (1st Class)
24846	NEO-WOOD MFG. CO., 23546 US RT 224, ALVADA, OH, 44802	US Mail (1st Class)
24846	NEW CASTLE ENGINEERING, CORPORATE PROCESSING DEPT, PO BOX 8055, FORT WAYNE, IN, 46898	US Mail (1st Class)
24846	NEW CASTLE INDUSTRIES INC., 36674 TREASURY CENTER, CHICAGO, IL, 60694-6600	US Mail (1st Class)
24846	NEW ENGLAND WHOLESALE, PO BOX 713, ESSEX JUNCTION, VT, 05453	US Mail (1st Class)
24846	NEW HORIZONS, 1890 COMMERCE CENTER, FAIRBORN, OH, 45324-6337	US Mail (1st Class)
24846	NEW HORIZONS, 14115 FARMINGTON RD., LIVONIA, MI, 48154	US Mail (1st Class)
24846	NEW HORIZONS COMPUTER, LEARNING CENTER, 6000 ROCKSIDE WOODS BLVD, INDEPENDENCE, OH, 44131	US Mail (1st Class)
24846	NEW LIFE INDUSTRIES, 140 CHAPPELLS DAIRY RD, SOMERSET, KY, 42503-4589	US Mail (1st Class)
24846	NEW TECHNOLOGIES INC., 4380 E BALDWIN RD BOX 97, GRAND BLANC, MI, 48439	US Mail (1st Class)
24846	NEW YORK DELI & GRILLE, 425 WATER STREET, CHARDON, OH, 44024	US Mail (1st Class)
24846	NEWARK ELECTRONICS, PO BOX 94151, PALATINE, IL, 60094-4151	US Mail (1st Class)
24846	NEWBURY ELECTRIC SUPPLY, CED, PO BOX 307260, GAHANNA, OH, 43230	US Mail (1st Class)
24846	NEWCO, INC., DEPT. #117301, PO BOX 67000, DETROIT, MI, 48267	US Mail (1st Class)
24846	NEWELL EQUIPMENT, INC., 1301 FREESE WORKS PL, GALION, OH, 44833-9368	US Mail (1st Class)
24846	NEWPORT SCIENTIFIC, INC., 8246-E SANDY COURT, JESSUP, MD, 20794-9632	US Mail (1st Class)
24846	NEWSPAPER HOLDINGS INC., CUSTOMER PAYMENT CENTER, PO BOX 60, NEW CASTLE, PA, 16103-0060	US Mail (1st Class)
24846	NEWSWEEK, 251 W 57TH ST, NEW YORK, NY, 10019-1802	US Mail (1st Class)
24846	NFM WELDING ENGINEERS, 577 OBERLIN AVE SW, MASSILLON, OH, 44647-7820	US Mail (1st Class)
24846	NICKLE PARTS INC., 836 HALL AVENUE, DAYTON, OH, 45404	US Mail (1st Class)
24846	NICOLA R. VERZINO, 2636 SHADY BANK LANE, NORTHPORT, AL, 35473	US Mail (1st Class)
24846	NISSAN TRADING CORP, USA, 34405 W 12 MILE RD, SUITE 225, FARMINGTON HILLS, MI, 48334	US Mail (1st Class)
24846	NMHG FINANCIAL SERVICES, INC., 44 OLD RIDGEBURY RD, DANBURY, CT, 06810-5107	US Mail (1st Class)
24846	NOLAR INDUSTRIES, 90 SNOW BLVD., CONCORD, ON, L4K 4A2CANADA	US Mail (1st Class)
24846	NORAL, 23456 MERCANTILE RD., CLEVELAND, OH, 44122	US Mail (1st Class)
24846	NOREN PRODUCTS INC., 1010 O'BRIEN DRIVE, MENLO PARK, CA, 94025	US Mail (1st Class)
24846	NORRELL SERVICES, INC., 925 N POINT PKWY, ALPHARETTA, GA, 30005-5210	US Mail (1st Class)
24846	NORRIS SALES ASSOCIATES, 19111 DETROIT AVE., SUITE 202, ROCKY RIVER, OH, 44116	US Mail (1st Class)
24846	NORTECH VACUUM PRODUCTS, DIV OF GUARDAIR CORP., 54 SECOND AVENUE, CHICOPEE, MA, 01020	US Mail (1st Class)
24846	NORTH AMERICAN MFG. CO., 4455 E 71ST STREET, CLEVELAND, OH, 44105	US Mail (1st Class)
24846	NORTH AMERICAN SECURITY SOLUTIONS, 2844 E RIVER RD., DAYTON, OH, 45439	US Mail (1st Class)
24846	NORTH AMERICAN STEEL CO., 18300 MILES RD, CLEVELAND, OH, 44128-3441	US Mail (1st Class)
24846	NORTH AMERICAN VAN LINES, 33901 TREASURY CENTER, CHICAGO, IL, 60694-3900	US Mail (1st Class)
24846	NORTH CANTON REPAIR SHOP, 1555 N. MAIN ST., NORTH CANTON, OH, 44720	US Mail (1st Class)
24846	NORTH CENTRAL PERIPHERALS, 10640 LYNDALE AVE SOUTH, BLOOMINGTON, MN, 55420	US Mail (1st Class)
24846	NORTH CENTRAL STATE COLLEGE, 2441 KENWOOD CIRCLE, PO BOX 698, MANSFIELD, OH, 44901-0698	US Mail (1st Class)
24846	NORTHBROOK APARTMENT, 800 ENERGY CENTER ROAD, NORTHPORT, AL, 35476	US Mail (1st Class)
24846	NORTHCOAST BUS. SYSTEMS, 7850 HUB PKWY, CLEVELAND, OH, 44125-5711	US Mail (1st Class)
24846	NORTHEAST POLY BAG CO., INC., 2 NORTHEAST BLVD., PO BOX 1460, STERLING, MA, 01564	US Mail (1st Class)
24846	NORTHERN OHIO PIZZA CO.INC., 8100 E 22ND STREET N, BUILDING 200, WICHITA, KS, 67226	US Mail (1st Class)
24846	NORTHERN TOOL & EQUIPMENT CO., PO BOX 1219, 2800 SOUTHCROSS DRIVE, BURNSVILLE, MN, 55337-0129	US Mail (1st Class)
24846	NORTHPORT HOSPITAL-DCH, 2700 HOSPITAL DR, NORTHPORT, AL, 35476-3360	US Mail (1st Class)
24846	NORTHWEST CONTROLS, 1936 SPRUCE ST, DEFIANCE, OH, 43512-2529	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	NORTHWEST SUPPLY, 4006 MCFARLAND BLVD, NORTHPORT, AL, 35476-2820	US Mail (1st Class)
24846	NORTHWESTERN OHIO SECURITY SYSTEMS, 121 E HIGH ST, LIMA, OH, 45801-4417	US Mail (1st Class)
24846	NORTHWOODS DELIVERY, PO BOX 1519, RHINELANDER, WI, 54501-1519	US Mail (1st Class)
24846	NO-SAG SPRING COMPANY LTD, 459 INDUSTRIAL ROAD, LONDON, ON, N5V 3E5CANADA	US Mail (1st Class)
24846	NOT JUST GRAPHICS, 32640 MICHIGAN AVENUE, WAYNE, MI, 48184-1762	US Mail (1st Class)
24846	NOTARY PUBLIC UNDERWRITER, 127 LEE ST, MONTGOMERY, AL, 36104-3516	US Mail (1st Class)
24846	NOVA CHEMICALS, CITICORP SERVICES, 8430 W BRYN MAWR AVE DEPT 2399, CHICAGO, IL, 60631-3473	US Mail (1st Class)
24846	NOVA CHEMICALS, CITICORP SERVICES, 8430 W BRYN MAWR AVE, DEPT 2399, CHICAGO, IL, 60631-3473	US Mail (1st Class)
24846	NOVACHEM, 955 CONNECTICUT AVE, BRIDGEPORT, CT, 06607-1200	US Mail (1st Class)
24846	NOVATEC, 222 THOMAS AVE, BALTIMORE, MD, 21225-3327	US Mail (1st Class)
24846	NOW ONLINE, INC., 781 BETA DRIVE, SUITE C, CLEVELAND, OH, 44143	US Mail (1st Class)
24846	NSA MICHIGAN, INC., PO BOX 4400543, NASHVILLE, TN, 37244-0543	US Mail (1st Class)
24846	NSK INDUSTRIES, INC., 150 ASCOT PARKWAY, P.O. BOX 1089, CUYAHOGA FALLS, OH, 44223	US Mail (1st Class)
24846	NSU CORP., 9385 SONORA RD., SONORA, KY, 42776	US Mail (1st Class)
24846	NTS, TECH PLAZA, UNIT 14, 12601 SOUTHFIELD RD., DETROIT, MI, 48223	US Mail (1st Class)
24846	NU STAR INC., 1425 STAGECOACH RD., SHAKOPEE, MN, 55379	US Mail (1st Class)
24846	NUMATICS INC., 2212 PAYSHERE CIRCLE, CHICAGO, IL, 60674	US Mail (1st Class)
24846	NU-TECH POLMERS, 4670 DUES DRIVE, CINCINNATI, OH, 45246	US Mail (1st Class)
24846	NUTONE PRODUCTS DISTRIBUTORS, 14670 EAST FIRESTON BLVD., SUITE 410, LA MIRADO, CA, 90638	US Mail (1st Class)
24846	O & P OIL & GAS, INC., 220 CENTER ST, ANDOVER, OH, 44003	US Mail (1st Class)
24846	O&D USA LLC, 13536 COLLECTIONS CENTER DR., CHICAGO, IL, 60693	US Mail (1st Class)
24846	O&S TRUCKING, INC., 3769 E EVERGREEN ST, SPRINGFIELD, MO, 65803-5720	US Mail (1st Class)
24846	O.C. TANNER SALES CO., 1930 SOUTH STATE STREET, SALT LAKE CITY, UT, 84115	US Mail (1st Class)
24846	O.E. MEYER COMPANY, 304 6TH AVE, GALION, OH, 44833	US Mail (1st Class)
24846	O'NEAL STEEL, 744 41ST ST N, BIRMINGHAM, AL, 35202	US Mail (1st Class)
24846	O'SULLIVAN MOVING & STORAGE CO., 380 FAIR STREET, FERNDALE, MI, 48220	US Mail (1st Class)
24846	OAKITE PRODUCTS INC., 50 VALLEY RD, BERKELEY HEIGHTS, NJ, 07922-2712	US Mail (1st Class)
24846	OESA, ACCOUNTING DEPT., 10 LABORATORY DRIVE, PO BOX 13966, RESEARCH TRIANGLE PARK, NC, 27709	US Mail (1st Class)
24846	OETIKER LMT., C/O P.O. BOX 337, MARLETTE, MI, 48453-0337	US Mail (1st Class)
24846	OFFICE MAX #296, 150 E PIERCE RD, ITASCA, IL, 60143-1290	US Mail (1st Class)
24846	OFFICE MAX CONTRACT INC, 150 E PIERCE RD, ITASCA, IL, 60143-1290	US Mail (1st Class)
24846	OFFICETEAM, 12400 COLLECTIONS, CENTER DRIVE, CHICAGO, IL, 60693	US Mail (1st Class)
24846	OHIO BANDITS, 76 CR 1975, JEROMESVILLE, OH, 44840	US Mail (1st Class)
24846	OHIO BUREAU OF MOTOR VEHICLES, 15065 KINSMAN RD, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	OHIO BUREAU OF MOTOR VEHICLES, 1970 W BROAD ST, COLUMBUS, OH, 43085	US Mail (1st Class)
24846	OHIO CARBON BLANK, 38403 PELTON RD, WILLOUGHBY, OH, 44094-7721	US Mail (1st Class)
24846	OHIO CARBON BLANK INC., 38403 PELTON RD, WILLOUGHBY, OH, 44094-7721	US Mail (1st Class)
24846	OHIO CHILD SUPPORT, PAYMENT CENTRAL, PO BOX 182394, COLUMBUS, OH, 43218-2394	US Mail (1st Class)
24846	OHIO COUNTING SCALE, O.C.S. TECHNOLOGIES, INC., 1300 EAST GRAINGER RD., BROOKLYN HEIGHTS, OH, 44131	US Mail (1st Class)
24846	OHIO CSPC, 30 E BROAD ST 32ND FL, COLUMBUS, OH, 43215-3414	US Mail (1st Class)
24846	OHIO DEPARTMENT OF TAXATION, PO BOX 16561, COLUMBUS, OH, 43216-6561	US Mail (1st Class)
24846	OHIO DEPT OF JOB & FAMILY, BUREAU OF UC TAX, PO BOX 182413, COLUMBUS, OH, 43218-2413	US Mail (1st Class)
24846	OHIO DEPT OF JOB AND FAMILY SERV, PO BOX 182413, COLUMBUS, OH, 43218	US Mail (1st Class)
24846	OHIO DEPT OF TAXATION, TAX COMMIS WILLIAM W WILKINS, 30 E BROAD ST 22ND FL, COLUMBUS, OH, 43215	US Mail (1st Class)
24846	OHIO DEPT. OF DEVELOPMENT, ADDRESS UNAVAILABLE, OH,	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	OHIO DEPT. OF TAXATION, POBOX 347, COLLUMBUS, OH, 43216-0347	US Mail (1st Class)
24846	OHIO DESK, 3009 SMITH ROAD, SUITE 500, FAIRLAWN, OH, 44333	US Mail (1st Class)
24846	OHIO DRILLING CO., 2405 BOSTIC BLVD SW, MASSILLON, OH, 44647-7686	US Mail (1st Class)
24846	OHIO E-CHECK, 122 S FRONT ST, COLUMBUS, OH, 43215-3425	US Mail (1st Class)
24846	OHIO EDISON, ACCT: 410-13-000016-0-01-5, FIRSTENERGY CORP, 76 S MAIN ST, AKRON, OH, 44308-1812	US Mail (1st Class)
24846	OHIO EDISON, ACCT: 1100 46 4741 09, FIRSTENERGY CORP 76 S MAIN ST, AKRON, OH, 44308-1812	US Mail (1st Class)
24846	OHIO EDISON COMPANY, 76 S MAIN ST, AKRON, OH, 44308-1812	US Mail (1st Class)
24846	OHIO FABRICATORS CO., 111 N. 14TH ST., PO BOX 218, COSHOCTON, OH, 43812-0218	US Mail (1st Class)
24846	OHIO INDUSTRIAL LUBRICANT, PO BOX 94225, CLEVELAND, OH, 44101	US Mail (1st Class)
24846	OHIO INDUSTRIAL SUPPLY INC., 915 LAKE ROAD, MEDINA, OH, 44256	US Mail (1st Class)
24846	OHIO KNIFE, 1280 SOM CENTER RD. #259, MAYFIELD HTS., OH, 44124	US Mail (1st Class)
24846	OHIO MATERIALS HANDLING, 7100 KRICK RD, BEDFORD, OH, 44146	US Mail (1st Class)
24846	OHIO PIPE, VALVES, & FITTINGS, INC., PO BOX 5897-C, CLEVELAND, OH, 44191	US Mail (1st Class)
24846	OHIO SECRETARY OF STATE, J KENNETH BLACKWELL, 180 E BROAD ST, COLUMBUS, OH, 43215	US Mail (1st Class)
24846	OHIO TRANSMISSION & PUMP, 1900 JETWAY BLVD, COLUMBUS, OH, 43219-1681	US Mail (1st Class)
24846	OIL SKIMMERS, INC, 12800 YORK RD STE G, CLEVELAND, OH, 44133-3683	US Mail (1st Class)
24846	OL COLONY, 401 OLD COLONY RD, TUSCALOOSA, AL, 35406	US Mail (1st Class)
24846	OLD DOMINION FREIGHT LINE, INC., ATTN: PAYMENT APPLICATION, 3000 CORPORATE EXCHANGE DR., 5TH FLOOR, COLUMBUS, OH, 43231	US Mail (1st Class)
24846	OLSTEN, PO BOX 371084, PITTSBURGH, PA, 15250-7084	US Mail (1st Class)
24846	OLSTEN STAFFING SERVICES, 312 PLUM STREET, SUITE 950, CINCINNATI, OH, 45202	US Mail (1st Class)
24846	OMEGA ENGINEERING, 1 OMEGA DR, STAMFORD, CT, 06907-2390	US Mail (1st Class)
24846	OMEGA ENGINEERING, INC., 1 OMEGA DR, STAMFORD, CT, 06907-2390	US Mail (1st Class)
24846	OMEGA PLASTICS, 2001 WEST 12TH STREET, ERIE, PA, 16505	US Mail (1st Class)
24846	OMEGADYNE, INC., 149 STELZER COURT, SUNBURY, OH, 43074	US Mail (1st Class)
24846	OMGA, INC., 3705 WILLIAM RICHARDSON DR., SOUTH BEND, IN, 46628	US Mail (1st Class)
24846	OMNI FASTENERS, INC., 909 TOWPATH ROAD, BROADVIEW HTS., OH, 44147	US Mail (1st Class)
24846	OMNI ONE, 72437 EAGLE WAY, CHICAGO, IL, 60678-7254	US Mail (1st Class)
24846	OMNI TECH CORPORATION, 100 SOUTH ALLOY DR., FENTON, MI, 48430	US Mail (1st Class)
24846	OMNIMARK INSTRUMENT, 1320 S PRIEST DR., SUITE 104, TEMPE, AZ, 85281	US Mail (1st Class)
24846	OMRON IDM CONTROLS, 9510 N HOUSTON ROSSLYN RD, HOUSTON, TX, 77088-3904	US Mail (1st Class)
24846	ONSTAR SUBSCRIPTION CENTER, 400 RENAISSANCE CENTER, DETROIT, MI, 48265	US Mail (1st Class)
24846	ONTARIO FAIRFIELD INN, 1065 N LEX. SPRINGMILL RD, MANSFIELD, OH, 44906	US Mail (1st Class)
24846	ONTARIO GLOVE, 500 DOTZERT COURT, WATERLOO, ON, N2L 6A7CANADA	US Mail (1st Class)
24846	ONTARIO/MANSFIELD, FAIRFIELD INN, 1065 N. LEX-SPRINGMILL RD., MANSFIELD, OH, 44906	US Mail (1st Class)
24846	OPEN, 1650 LAKE SHORE DR STE 350, COLUMBUS, OH, 43204-4978	US Mail (1st Class)
24846	OPENONLINE, 1650 LAKE SHORE DR STE 350, COLUMBUS, OH, 43204-4978	US Mail (1st Class)
24846	OPMA, 679 NORBURY DRIVE, HUDSON, OH, 44236-4685	US Mail (1st Class)
24846	OPTICARE VISION CENTERS, 700 SOUTH MAIN ST., SUITE 105, LAPEER, MI, 48446	US Mail (1st Class)
24846	OPTIVA SYSTEMS INC, 23399 COMMERCE DR, STE B-5A, FARMINGTON HILLS, MI, 48335	US Mail (1st Class)
24846	ORBIS, 1055 CORPORATE CENTER DR, PO BOX 389, OCONOMOWOC, WI, 53066-0389	US Mail (1st Class)
24846	ORBIS, DRAWER 346, MILWAUKEE, WI, 53278-0346	US Mail (1st Class)
24846	ORBITFORM, INC., 1600 EXECUTIVE DR, JACKSON, MI, 49203-3469	US Mail (1st Class)
24846	ORLANDO'S SUPERMARKET, 6277 S MAIN STREET, PO BOX 271, NORTH KINGSVILLE, OH, 44068	US Mail (1st Class)
24846	ORYCON CONTROL TECHNO, 3407 ROSE AVENUE, OCEAN, NJ, 07712-3923	US Mail (1st Class)
24846	OSCO INC., 2937 WATERVIEW DR, ROCHESTER HILLS, MI, 48309	US Mail (1st Class)
24846	OTIS ELEVATOR COMPANY, 1 FARM SPRINGS RD, FARMINGTON, CT, 06032-2572	US Mail (1st Class)
24846	OUTBACK STEAKHOUSE, INC., 820 N. LEXINGTON-SPRINGMILL RD, ONTARIO, OH, 44906	US Mail (1st Class)
24846	OUTSOURCE PARTNERS INC., PO BOX 932101, ATLANTA, GA, 31193-2101	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	OVERNITE TRANSPORTION CO., PO BOX 79755, BALTIMORE, MD, 21279-0755	US Mail (1st Class)
24846	OWENS ROAD SERVICE, LLC, 6011 DEERFIELD ST., DAYTON, OH, 45414	US Mail (1st Class)
24846	OXFORD OVERHEAD DOOR, 2118 METAMORA, OXFORD, MI, 48371	US Mail (1st Class)
24846	OYL-AIR SPECIALTY CO., 28196 SCIPPO CREEK RD., CIRCLEVILLE, OH, 43113-9796	US Mail (1st Class)
24846	P.E.T.S., 4141 LUELLA LANE, AUBURN HILLS, MI, 48236	US Mail (1st Class)
24846	PA SCDU, PO BOX 69112, HARRISBURG, PA, 17106-9112	US Mail (1st Class)
24846	PAARLO PLASTICS INC., 7720 TIM AVENUE NW, PO BOX 2556, NORTH CANTON, OH, 44720-0556	US Mail (1st Class)
24846	PAASCHE AIRBRUSH CO, 4311 N NORMANDY AVE, CHICAGO, IL, 60634-1395	US Mail (1st Class)
24846	PACE MOLD & MACHINE INC., 8225 NAVARRE RD SW, MASSILLON, OH, 44646-8813	US Mail (1st Class)
24846	PACESETTER PERSONNEL SERVICE, 4615 POST OAK PLACE DR, HOUSTON, TX, 77027-9731	US Mail (1st Class)
24846	PACKAGING CORP OF AMERICA, 36596 TREASURY CENTER, CHICAGO, IL, 60694-6500	US Mail (1st Class)
24846	PACKAGING CORP OF AMERICA, PACKAGING CREDIT COMPANY LLC, 36596 TREASURY CENTER, CHICAGO, IL, 60694-6500	US Mail (1st Class)
24846	PACKING MATERIAL CO. INC., 27280 HAGGERTY RD., STE C-16, FARMINGTON HILLS, MI, 48331	US Mail (1st Class)
24846	PACKLINE CO., 1675 FRITZ DR., TRENTON, MI, 48183	US Mail (1st Class)
24846	PACUR, 3555 MOSER STREET, OSHKOSH, WI, 54901	US Mail (1st Class)
24846	PADGETT-THOMPSON, 11221 ROE AVE, SHAWNEE MISSION, KS, 66211-1748	US Mail (1st Class)
24846	PAE, 900 C RIVER STREET, WINDSOR, CT, 06095	US Mail (1st Class)
24846	PALAMATIC HANDLING, 901 S BOLMAR ST STE C, WEST CHESTER, PA, 19382-4550	US Mail (1st Class)
24846	PALERMO REALTY, 3435 WEST LAKE RD, PO BOX 8026, ERIE, PA, 16505	US Mail (1st Class)
24846	PALLADIUM ITEM, PO BOX 308, 1175 NORTH A STREET, RICHMOND, IN, 47375-0308	US Mail (1st Class)
24846	PAM DEDICATED SERVICES INC., 297 W HENRI DETONTI BLVD, TONTITOWN, AR, 72770	US Mail (1st Class)
24846	PAM MAXFIELD, 13 GUNRIDGE LANE, LAPEER, MI, 48446	US Mail (1st Class)
24846	PANATEK, 2358 HASSELL RD, SUITE E, HOFFMAN ESTATES, IL, 60195	US Mail (1st Class)
24846	PANDUIT CORP., 16530 163RD STREET, LOCKPORT, IL, 60441	US Mail (1st Class)
24846	PANEL COMPONENTS & SYSTEMS, 149 MAIN STREET, STANHOPE, NJ, 07874	US Mail (1st Class)
24846	PANELOC CORP., 142 BRICKYARD RD., FARMINGTON, CT, 06032	US Mail (1st Class)
24846	PANTHER II TRANSPORTATION, 1114 N COURT ST., POB 713, MEDINA, OH, 44256	US Mail (1st Class)
24846	PAR FOAM PRODUCTS INC., 239 VAN RENSSELAER STREET, BUFFALO, NY, 14210	US Mail (1st Class)
24846	PAR INDUSTRIES, LLC, DEPT. 148401 DRIVE, PO BOX 67000, DETROIT, MI, 48267-1484	US Mail (1st Class)
24846	PAR INDUSTRIES, LLC, DEPT.# 148401, PO BOX 67000, DETROIT, MI, 48267-1484	US Mail (1st Class)
24846	PARAGON PRINT SYSTEMS, INC., 141 E 26TH STREET, ERIE, PA, 16504-1048	US Mail (1st Class)
24846	PARAMETRIC TECHNOLOGY CORP., 140 KENDRICK ST, NEEDHAM, MA, 02494-2739	US Mail (1st Class)
24846	PARAMOUNT STAMPING, 1200 W 58TH STREET, CLEVELAND, OH, 44102	US Mail (1st Class)
24846	PARAMOUNT STAMPING, 1200 W 58TH ST, CLEVELAND, OH, 44102	US Mail (1st Class)
24846	PARATHERM CORP., 1050 COLWELL ROAD, CONSHOHOCKEN, PA, 19428	US Mail (1st Class)
24846	PARK AVENUE CATERING, 15 PARK AVENUE WEST, MANSFIELD, OH, 44902	US Mail (1st Class)
24846	PARK SUPPLY, 2918 10TH AVE., TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	PARK TRANSPORT & STORAGE, 1500 30TH STREET, PO BOX 2508, TUSCALOOSA, AL, 35403	US Mail (1st Class)
24846	PARKER HANNIFIN, 7917 COLLECTION CENTER, CHICAGO, IL, 60693	US Mail (1st Class)
24846	PARKER HANNIFIN CORPORATION, 7911 COLLECTION CENTER, CHICAGO, IL, 60693	US Mail (1st Class)
24846	PARKER SEAL CO., 7917 COLLECTION CENTER DR., CHICAGO, IL, 60693	US Mail (1st Class)
24846	PARKER SEALS, 7917 COLLECTION CENTER DRIVE, CHICAGO, IL, 60693	US Mail (1st Class)
24846	PARKER STEEL COMPANY, 4329 MONROE ST, TOLEDO, OH, 43606-1956	US Mail (1st Class)
24846	PARQUE INDUSTRIAL INT, CALLE SEGUNDA #4, COL. CENTRO, CHIHUAHUA, 31000MEXICO	US Mail (1st Class)
24846	PARTNERS ENVIRONMENTAL, 31100 SOLON ROAD, SUITE G, SOLON, OH, 44139	US Mail (1st Class)
24846	PATCO ELECTRICAL CONTRACT, 3411 MCGEE RD, COTTONDALE, AL, 35453-1723	US Mail (1st Class)
24846	PATE'S WESTERN IMAGE, 918 L. WALLACE BLVD.SOUTH, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	PATH SILICONES, POBOX 430, ELMWOOD PARK, NJ, 07407	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	PATH TECHNOLOGIES, 7505 TYLER BLVD., UNIT 2, MENTOR, OH, 44060-5416	US Mail (1st Class)
24846	PATON PRESS LLC, 60 INDEPENDENCE CIR, CHICO, CA, 95973-4921	US Mail (1st Class)
24846	PATRICIA A MARKER, 14705-2 LAKEVIEW DR., MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	PATRICIA BUTLER, 4345 CO RD 31, GALION, OH, 44833	US Mail (1st Class)
24846	PATRICIA L. WESSLING, 32313 KALAMAZOO, WESTLAND, MI, 48186	US Mail (1st Class)
24846	PATRICIA MARKER, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	PATRICK CLEAVER, 900 S. WILEY STREET, PO BOX 320, CRESTLINE, OH, 44827-0320	US Mail (1st Class)
24846	PATTERSON FAN CO., INC., 1120 NORTHPOINT BLVD., BLYTHEWOOD, SC, 29018	US Mail (1st Class)
24846	PAUL AUBERT, 5805 FARMBROOK, DETROIT, MI, 48224	US Mail (1st Class)
24846	PAUL C. HANAHAN, M.D., 150 MENTOR AVE. SUITE 5, PAINESVILLE, OH, 44077	US Mail (1st Class)
24846	PAUL N. GARDNER CO.INC., 316 NE 1ST ST, POMPANO BEACH, FL, 33060-6608	US Mail (1st Class)
24846	PAUL'S LUMBER DO-IT, CENTER, 8018 FRENCH ST., GARRETTSVILLE, OH, 44231	US Mail (1st Class)
24846	PAXAR, 105 CORPORATE PARK DR, WHITE PLAINS, NY, 10604-3814	US Mail (1st Class)
24846	PAXTON PAPER CO., INC., 6047 WOODLAND FORREST DR, TUSCALOOSA, AL, 35405-5651	US Mail (1st Class)
24846	PBB USA INC., 670 YOUNG ST, TONAWANDA, NY, 14150-4103	US Mail (1st Class)
24846	PCRS TECHNOLOGIES, 11097 BARDELL RD NW, BOLIVAR, OH, 44612-9605	US Mail (1st Class)
24846	PEIDMONT NATIONAL, 1570 JEAN STREET, MONTGOMERY, AL, 36107	US Mail (1st Class)
24846	PENCO TOOL, 2621 WEST AVE, ASHTABULA, OH, 44004-3115	US Mail (1st Class)
24846	PENDELTON TOOL CO, 3214 WEST 22ND STREET, ERIE, PA, 16506	US Mail (1st Class)
24846	PENDLETON TOOL COMPANY INC., 3214 WEST 22ND STREET, ERIE, PA, 16508	US Mail (1st Class)
24846	PENELEC, 395 GHENT RD, FAIRLAWN, OH, 44333-2678	US Mail (1st Class)
24846	PENN COLOR, 400 OLD DUBLIN PIKE, DOYLESTOWN, PA, 18901-2356	US Mail (1st Class)
24846	PENN TECH MACHINE, 992 PERRY HWY., PITTSBURGH, PA, 15237	US Mail (1st Class)
24846	PENN TOOL CO., 1776 SPRINGFIELD AVE., MAPLEWOOD, NJ, 07040-2931	US Mail (1st Class)
24846	PENN-OHIO PALLET SERVICE, 2975 N. SALEM-WARREN RD., NORTH JACKSON, OH, 44451	US Mail (1st Class)
24846	PENSION BENEFIT GUARANTY, DEPT 77430, PO BOX 77000, DETROIT, MI, 48277-0430	US Mail (1st Class)
24846	PENSKE TRUCK LEASING, 2675 MORGANTOWN RD, READING, PA, 19607-9676	US Mail (1st Class)
24846	PERFECT WATER CO., 1720A CRETE ST, MOBERLY, MO, 65270	US Mail (1st Class)
24846	PERFORMANCE ALLOY & SERVICES, N 116 W 18515 MORSE DR., GERMANTOWN, WI, 53022	US Mail (1st Class)
24846	PERFORMANCE ENTERPRISES, INC., 20771 RYAN ROAD, WARREN, MI, 48091	US Mail (1st Class)
24846	PERFORMANCE PLASTICS, INC, C/O AMER COMM, 6500 BROOKTREE RD., WEXFORD, PA, 15090- 9272	US Mail (1st Class)
24846	PERFORMANCE POLYMERS, 13009 COLLECTIONS, CHICAGO, IL, 60693	US Mail (1st Class)
24846	PERFORMANCE POLYMERS INC., 803R LANCASTER ST, LEOMINSTER, MA, 01453-4503	US Mail (1st Class)
24846	PERG, PO BOX 1003, UNIONTOWN, OH, 44685	US Mail (1st Class)
24846	PERMACEL, PO BOX 92263, CHICAGO, IL, 60675-2263	US Mail (1st Class)
24846	PERMACEL AUTOMOTIVE, 8485 PROSPECT AVE, KANSAS CITY, MO, 64132-2376	US Mail (1st Class)
24846	PERSONNEL MANAGEMENT, INC., 1499 WINDHORST WAY STE 220, GREENWOOD, IN, 46143-8800	US Mail (1st Class)
24846	PETAL PUSHER, BOX 813, 14962 S. STATE AVE., MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	PETE'S AUTO DETAILING, 13950 CLARIDON-TROY, BURTON, OH, 44021	US Mail (1st Class)
24846	PETER WANGUNYU, 31572 INDUSTRIAL RD., SUITE #300, LIVONIA, MI, 48150	US Mail (1st Class)
24846	PETERS ENGINEERING LLC, 2193 ASSOCIATION DR., SUITE 300, OKEMOS, MI, 48864	US Mail (1st Class)
24846	PETERSON & COMPANY, PO BOX 408, 27040 PRINCETON AVE., INKSTER, MI, 48141	US Mail (1st Class)
24846	PETERSON'S CLEANING SERVICE, EBCO PARK, 2001 PENINSULA DRIVE, SUITE #7, ERIE, PA, 16506	US Mail (1st Class)
24846	PETTY CASH, LISA SHANK OR TANYA KAPEN, 2601 A SOUTH RIDGE ROAD EAST, ASHTABULA, OH, 44004	US Mail (1st Class)
24846	PETTY CASH, MARK DILLEN / GINNIE HICKS, 119 EAST DAYTON STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	PETTY CASH, BEVERLY SCHULTZ/JERRY SCHARDT, 5633 52ND ST. S.E., GRAND RAPIDS, MI, 49512	US Mail (1st Class)
24846	PETTY CASH, WILLIAM WAHL OR, MICHAEL LORENZ, BELLEVILLE, MI, 48111	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	PETTY CASH, BARBARA ROHARIK, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	PETTY CASH, BEVERLY J. SCHULTZ, 5633 52ND STREET SE, GRAND RAPIDS, MI, 49512	US Mail (1st Class)
24846	PETTY CASH, 1401 INDUSTRIAL DRIVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	PETTY CASH - VANDALIA, 985 FALLS CREEK DR., VANDALIA, OH, 45377	US Mail (1st Class)
24846	PETTY CASH CANTON, 3131 COLUMBUS RD. NE, CANTON, OH, 44705	US Mail (1st Class)
24846	PETTY CASH/CHARDON, BARB WHALEN, 100 7TH AVE./SUITE 100, CHARDON, OH, 44024	US Mail (1st Class)
24846	PETTY CASH/LAPEER, THERESA GILBERT, LAPEER, MI, 48446	US Mail (1st Class)
24846	PETTY CASH-BISHOPVILLE, 15 MYRTLE DR, BISHOPVILLE, SC, 29010	US Mail (1st Class)
24846	PEYTON C. COCHRANE, RM 124, TUSC COUNTY COURTHOUSE, 714 GREENSBORO AVENUE, TUSCALOOSA, AL, 35401-1891	US Mail (1st Class)
24846	PFA, INC., N118 W18251 BUNSEN DRIVE, GERMANTOWN, WI, 53022	US Mail (1st Class)
24846	PH, INC., 4429 HAMANN PKWY., WILLOUGHBY, OH, 44094	US Mail (1st Class)
24846	PHIL HARRELL, 290 MCCORMICK, LAPEER, MI, 48446	US Mail (1st Class)
24846	PHIL HART MANAGEMENT SERVICES, INC, 3985 SILVER MEADOW LANE, GRAND PRAIRIE, TX, 75052	US Mail (1st Class)
24846	PHIL MAUER & ASSOC. INC., 56954 EDEN LINE, EDEN, ON, NOJ 1H0CANADA	US Mail (1st Class)
24846	PHILIP CALAWAY, 222 LOCUST DRIVE, CONNEAUT, OH, 44030	US Mail (1st Class)
24846	PHILIPS ELLIOTT, 1401 INDUSTRIAL PARK DR., TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	PHILLIP KRAMER, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	PHILLIP MILLER, 15450 BURTON WINDSOR RD, MIDDLEFIELD, OH, 44062-9786	US Mail (1st Class)
24846	PHILLIPS CHEMICAL CO., PHILLIPS PETROLEUM CO., P. O. BOX 96001, CHICAGO, IL, 60693	US Mail (1st Class)
24846	PHOENIX ASSOCIATES, 16760 PARK CIRCLE DR, CHAGRIN FALLS, OH, 44023-4563	US Mail (1st Class)
24846	PHOENIX MACHINERY MOVERS, 50555 CORPORATE DR., SHELBY TWP, MI, 48315-3103	US Mail (1st Class)
24846	PHOENIX ROPE & CORDAGE CO. INC, PO BOX 520, 220 ALESSIO DR., JOLIET, IL, 60433	US Mail (1st Class)
24846	PHYGEN, 1400 MARSHALL ST NE, MINNEAPOLIS, MN, 55413	US Mail (1st Class)
24846	PHYSICIAN SALES & SERVICE INC., 2003 CASE PARKWAY SOUTH, TWINSBURG, OH, 44087	US Mail (1st Class)
24846	PIAB USA, INC., 65 SHARP STREET, HINGHAM, MA, 02043	US Mail (1st Class)
24846	PIC GROUP, PC PRODUCTIVITY IMPROVEMENT, CENTER INC., 36510 TREASURY CENTER, CHICAGO, IL, 60694-6500	US Mail (1st Class)
24846	PIC GROUP, DURHAM COLLEGE, 199 WENTWORTH ST. E, OSHAWA, ON, L1H 3V6CANADA	US Mail (1st Class)
24846	PIE POWELL INDUSTRIAL ENGRG., 5455 LENN RD., NEWBURGH, IN, 47630	US Mail (1st Class)
24846	PILOT AIR FREIGHT, 135 S LASALLE, DEPT. 1573, CHICAGO, IL, 60674	US Mail (1st Class)
24846	PINNACLE TECHNOLOGY, 5559 EATON LEWISBURG RD., LEWISBURG, OH, 45338	US Mail (1st Class)
24846	PIONEER CAREER & TECH.CTR, PO BOX 309, 27 RYAN ROAD, SHELBY, OH, 44875	US Mail (1st Class)
24846	PIONEER RANDUSTRIAL, 4529 INDUSTRIAL PWKY, CLEVELAND, OH, 44135	US Mail (1st Class)
24846	PIONEER RESEARCH, 3443 N CENTRAL AVE., SUITE 1200, PHOENIX, AZ, 85012	US Mail (1st Class)
24846	PIPES BY PAPPY'S, 4469 FUNK RD, SHELBY, OH, 44875-9701	US Mail (1st Class)
24846	PIQUA TECHNOLOGIES, 1620 S MAIN ST, PIQUA, OH, 45356-8320	US Mail (1st Class)
24846	PIQUA TECHNOLOGIES LLC, 1620 S MAIN ST, PIQUA, OH, 45356-8320	US Mail (1st Class)
24846	PITNEY BOWES PURCHASE POWER, FIRST EXPRESS REMITTANCE PROCESSING, PURCHASE POWER, 5101 INTERCHANGE WAY, LOUISVILLE, KY, 40285-6042	US Mail (1st Class)
24846	PITNEY BOWES PURCHASE POWER, 1 ELMCROFT ROAD, STAMFORD, CT, 06926-0700	US Mail (1st Class)
24846	PITTOHIO EXPRESS, 15 27TH ST, PITTSBURGH, PA, 15222-4729	US Mail (1st Class)
24846	PITTS FIRE EXTINGUISHER, 26 6TH ST. SOUTHWEST, MASSILLON, OH, 44647	US Mail (1st Class)
24846	PJ'S FABRICATING, 1511 LINWOOD AVENUE SW, CANTON, OH, 44710	US Mail (1st Class)
24846	PLANTERRA TROPICAL GREENHOUSES, 7315 DRAKE RD., W. BLOOMFIELD, MI, 48322	US Mail (1st Class)
24846	PLANTSCAPING, INC., 1865 EAST 40TH STREET, CLEVELAND, OH, 44103	US Mail (1st Class)
24846	PLASTEC U.S.A. INC., 7752 N.W. 74 AVE., MIAMI, FL, 33166	US Mail (1st Class)
24846	PLASTEK INDUSTRIES INC., 2425 W 23RD ST, ERIE, PA, 16506-2920	US Mail (1st Class)
24846	PLASTIC COMPOUNDERS, 300 TAYLOR AVE, CAMBRIDGE, OH, 43725-1348	US Mail (1st Class)
24846	PLASTIC MACHINERY, 500 INDUSTRIAL PKWY, PO BOX 1027, RICHMOND, IN, 47374	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	PLASTIC MATERIALS INC., 868 E TALLMADGE AVE., AKRON, OH, 44310	US Mail (1st Class)
24846	PLASTIC MLDG DEVELOPMENT, 42400 YEARGO DR, STERLING HEIGHTS, MI, 48314	US Mail (1st Class)
24846	PLASTIC MOLD TECHNOLOGY, 4201 BROADMOOR S.E., KENTWOOD, MI, 49512	US Mail (1st Class)
24846	PLASTIC PROCESS EQUIP INC, 8303 CORPORATE PARK DR, CLEVELAND, OH, 44102	US Mail (1st Class)
24846	PLASTIC PROCESS EQUIPMENT, 8303 CORPORATE PARK DR, CLEVELAND, OH, 44102	US Mail (1st Class)
24846	PLASTIC SERVICES, 15809 CLAIRE COURT, MACOMB TOWNSHIP, MI, 48042	US Mail (1st Class)
24846	PLASTICS MOLDING CO., 4211 NORTH BROADWAY, ST. LOUIS, MO, 63147	US Mail (1st Class)
24846	PLASTICS NEWS, 1725 MERRIMAN RD, AKRON, OH, 44313-9004	US Mail (1st Class)
24846	PLASTOHM SA, RUE GRANGE MORIN, Z.I. ARNAS BP 178, VILLEFRANCHE S/SAONE, 69656FRANCE	US Mail (1st Class)
24846	PLASTOMER CORPORATION, COMERICA BK DEPT #15601, PO BOX 67000, DETROIT, MI, 48267	US Mail (1st Class)
24846	PLYMOUTH HIGH SCHOOL, 400 TRUX STREET, PLYMOUTH, OH, 44865	US Mail (1st Class)
24846	PME COMPANIES, INC., 13870 E ELEVEN MILE RD., WARREN, MI, 48089	US Mail (1st Class)
24846	PNR AMERICA, LLC, ONE CIVIC CENTER PLAZA, SUITE 508, POUGHKEEPSIE, NY, 12547	US Mail (1st Class)
24846	POLYMER CLEANING TECHNOLOGY, 9 ILENE COURT, UNIT 16, BELLE MEAD, NJ, 08502	US Mail (1st Class)
24846	POLYMER COLOR SERVICES, 153 HITT ROAD, EASLEY, SC, 29640	US Mail (1st Class)
24846	POLYMER MACHINERY CO., INC., 466 COMMERCE ST, TALLMADGE, OH, 44278-2132	US Mail (1st Class)
24846	POLYMER TESTING INSTR., INC., PO BOX 840045, ST. AUGUSTINE, FL, 32084	US Mail (1st Class)
24846	POLYMERLAND, 7663 INDUSTRIAL BLVD, ALLENTOWN, PA, 18106-9265	US Mail (1st Class)
24846	POLYMEROHIO, INC., C/O WAYNE EARLY, PO BOX 2098, WESTERVILLE, OH, 43086	US Mail (1st Class)
24846	POLYONE CORP./HARWICK CHEMICAL, DEPT. CH 10489, PALATINE, IL, 60055-0489	US Mail (1st Class)
24846	POLYONE DISTRIBUTION-LOCKBOX, DEPT. CH 14046, PALATINE, IL, 60055-4046	US Mail (1st Class)
24846	POLYONE DISTRIBUTION-LOCKBOX, DEPT CH 10489, PALANTINE, IL, 60055-0489	US Mail (1st Class)
24846	POLYSHOT CORP., 75 LUCIUS GORDON DRIVE, WEST HENRIETTA, NY, 14586-9682	US Mail (1st Class)
24846	PONTIAC ELECTRIC MOTORS, 224 WEST SHEFFIELD, PONTIAC, MI, 48340	US Mail (1st Class)
24846	PONZELLA LUNCEFORD, 1401 INDUSTRIAL PK. DR., TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	PORT HURON TIMES HERALD, ATTN: CLASSIFIED DEPT., PO BOX 5009, PORT HURON, MI, 48061	US Mail (1st Class)
24846	PORT-A-COOK, INC., 309 W MAIN ST, LUCAS, OH, 44843-9568	US Mail (1st Class)
24846	PORTAGE PRECISION POLYMERS, 215 W LAKE ST, RAVENNA, OH, 44266	US Mail (1st Class)
24846	PORTER & HECKMAN, 3056 DAVISON RD., LAPEER, MI, 48446	US Mail (1st Class)
24846	PORTER CABLE CORPORATION, 4825 US HIGHWAY 45 N, JACKSON, TN, 38305-7900	US Mail (1st Class)
24846	POSITIVE PROMOTIONS INC., 15 GILPIN AVE., PO BOX 18021, HAUPPAUGE, NY, 11788-8821	US Mail (1st Class)
24846	POSITRON CORPORATION, ATTN: MATT OXENDER, 4614 WYLAND DRIVE, ELKHART, IN, 46516	US Mail (1st Class)
24846	POSTMASTER, 15993 E HIGH ST, MIDDLEFIELD, OH, 44062-9800	US Mail (1st Class)
24846	POSTMASTER, U S POSTAL SERVICE, 567 W NEPESSING STREET, LAPEER, MI, 48446-9998	US Mail (1st Class)
24846	POTOMAC DISPLAY & MANUFACTURING, 737 MULLIGAN LN, WESTMINSTER, MD, 21158-9435	US Mail (1st Class)
24846	POWER & SIGNAL GROUP, 4670 RICHMOND RD, SUITE 120, WARRENVILLE HEIGHTS, OH, 44128	US Mail (1st Class)
24846	POWER DRIVES INC., 133 HOPKINS ST, BUFFALO, NY, 14220-2101	US Mail (1st Class)
24846	POWERMOTION, 90 ROBERT JEMISON ROAD, BIRMINGHAM, AL, 35209	US Mail (1st Class)
24846	POWERVOLT, INC., 300 W FACTORY ROAD, PO BOX 383, ADDISON, IL, 60101-5004	US Mail (1st Class)
24846	POWERWARE, PO BOX 93810, CHICAGO, IL, 60673-3810	US Mail (1st Class)
24846	POWERWAY, INC., L-2522, COLUMBUS, OH, 43260	US Mail (1st Class)
24846	PRAIRIE BUSINESS CREDIT, INC., ACCOUNT OF SKILLWOOD PRODUCTS, PO BOX 802269, CHICAGO, IL, 60680-2269	US Mail (1st Class)
24846	PREACHER KAIN, 15689 MAYFIELD RD., HUNSTBURG, OH, 44061	US Mail (1st Class)
24846	PREBLE COUNTY CLERK OF COURTS, 985 FALLS CREEK DR., VANDALIA, OH, 45377	US Mail (1st Class)
24846	PREBLE COUNTY TREASURER, BRENDA WHITE, PO BOX 341, EATON, OH, 45320	US Mail (1st Class)
24846	PREBLE COUNTY TREASURER, BRENDA K WHITE, 101 EAST MAIN ST, EATON, OH, 45320	US Mail (1st Class)
24846	PREBLE COUNTY TREASURER, BRENDA K. WHITE, 101 EAST MAIN ST, EATON, OH, 45320	US Mail (1st Class)
24846	PREBLE LUMBER CO., 645 CENTRAL AVE., CARLISLE, OH, 45005	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	PRECEPT SOFTWARE, LLC., 217 FIRST STREET, HURON, OH, 44839	US Mail (1st Class)
24846	PRECISE INC., 5223 NORTH RIDGE WEST, SUITE 5, ASHTABULA, OH, 44004	US Mail (1st Class)
24846	PRECISION AUTO PLASTICS, 1617 TURNER RD, WINDSOR, ON, N8W 3J8CANADA	US Mail (1st Class)
24846	PRECISION BOLT, INC., 1509 VESHECCO DRIVE, ERIE, PA, 16501	US Mail (1st Class)
24846	PRECISION CERTIFIED WELDING, 203 N. JERSEY, DAYTON, OH, 45403	US Mail (1st Class)
24846	PRECISION MACHINE CO., PO BOX 634325, CINCINNATI, OH, 45263-4325	US Mail (1st Class)
24846	PRECISION PLATING, INC., 980 N. PINE HILL RD., BIRMINGHAM, AL, 35217	US Mail (1st Class)
24846	PRECISION TECHNOLOGIES, 1740 STATE RTE. 61 SOUTH, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	PREFERRED INC., 4871 NEO PARKWAY, CLEVELAND, OH, 44128	US Mail (1st Class)
24846	PREMIER PLASTICS GROUP, 4400 RAY BOLL BLVD., COLUMBUS, IN, 47203	US Mail (1st Class)
24846	PREMIER PRINT & SERVICE, 120 SOUTH RIVERSIDE PLAZA, CHICAGO, IL, 60606-3938	US Mail (1st Class)
24846	PRENTICE HALL, PAYMENT PROCESSING CENTER, PO BOX 10806, DES MOINES, IA, 50336-0806	US Mail (1st Class)
24846	PRESIDENT'S CLUB SERVICE CENTER, RCHPM, PO BOX 4555, HOUSTON, TX, 77210-4555	US Mail (1st Class)
24846	PRESS AUTOMATION/FLUIDLINE, 2008 HOLLAND-SYLVANIA ROAD, TOLEDO, OH, 43615	US Mail (1st Class)
24846	PREST SALES COMPANY, 14963 PARK VIEW CT, STERLING HEIGHTS, MI, 48313-5771	US Mail (1st Class)
24846	PRESTON, 13600 WEST CENTER STREET, BURTON, OH, 44021	US Mail (1st Class)
24846	PRIDE SEALS, INC., 10859 SHARONDALE RD., CINCINNATI, OH, 45241-2835	US Mail (1st Class)
24846	PRIMA TECH, INC., 24837 SHERWOOD AVE., CENTERLINE, MI, 48015	US Mail (1st Class)
24846	PRIME EQUIPMENT, PO BOX 840514, DALLAS, TX, 75284-0514	US Mail (1st Class)
24846	PRIME TECHNOLOGY GROUP LLC, 14500 LOCHRIDGE BLVD, SUITE D, COVINGTON, GA, 30014	US Mail (1st Class)
24846	PRINTING INC., 344 N ST. FRANCIS, WICHITA, KS, 67202	US Mail (1st Class)
24846	PRIORITY SERVICE, BY PORTER & HECKMAN, 3056 DAVISON RD., LAPEER, MI, 48446	US Mail (1st Class)
24846	PRIORTY PRINTER SALE, 11660 ALPHARETTA HWY, SUITE 110, ROSWELL, GA, 30076	US Mail (1st Class)
24846	PRIORTY SERVICE, PORTER & HECKMAN, 3056 DAVISON RD., LAPEER, MI, 48446	US Mail (1st Class)
24846	PRISM TECHNOLOGY & ASSEMBLIES, 13690 S. MOISERTOWN RD., MEADVILLE, PA, 16335	US Mail (1st Class)
24846	PRO CRETE INDUSTRIES, 803 WAREHOUSE RD., TOLEDO, OH, 43615	US Mail (1st Class)
24846	PROACTIVE QUALITY, 77 COX PLAZA, GILBERTSVILLE, KY, 42044	US Mail (1st Class)
24846	PROACTIVE TECHNOLOGIES, 7849 COLLIN CT., YPSILANTI, MI, 48197	US Mail (1st Class)
24846	PROAUTO, 6152 RIVER RD., EAST CHINA, MI, 48054	US Mail (1st Class)
24846	PROCESS CONTROL GROUP INC., 8101 TOWER POINT DR, CHARLOTTE, NC, 28227	US Mail (1st Class)
24846	PROCESS DEVELOPMENT CORP., 33027 SCHOOLCRAFT RD., LIVONIA, MI, 48150	US Mail (1st Class)
24846	PRODUCT ACTION INTERNATIONAL,INC., 2506 RELIABLE PARKWAY, CHICAGO, IL, 60686-0025	US Mail (1st Class)
24846	PRODUCTION TOOL SUPPLY CO, 2100 MAC ARTHUR BLVD, WARREN, MI, 48089	US Mail (1st Class)
24846	PRODUCTION TOOL SUPPLY CO., DEPT. #587, PO BOX 67000, DETROIT, MI, 48267-0587	US Mail (1st Class)
24846	PRODUCTIVITY IMPROVEMENT CENTER INC, 36510 TREASURY CENTER, CHICAGO, IL, 60694-6500	US Mail (1st Class)
24846	PRODUCTIVITY IMPROVEMENT CTR, DURHAM COLLEGE, 1610 CHAMPLAIN AVE., WHITBY, ON, L1N 6A7CANADA	US Mail (1st Class)
24846	PRODUCTIVITY QUALITY, 15200 25TH AVE N, PLYMOUTH, MN, 55447-2190	US Mail (1st Class)
24846	PROFESSIONAL ASSEMBLY CORP., 704 S HAMILTON, SAGINAW, MI, 48602	US Mail (1st Class)
24846	PROFESSIONAL CONSULT, 428 EAST STREET, WETUMPKA, AL, 36092	US Mail (1st Class)
24846	PROFESSIONAL CONSULTANTS, 300 PATTON HILL ROAD, TOWNLEY, AL, 35587-3024	US Mail (1st Class)
24846	PROGRESS ENERGY CAROLINAS, INC., ACCT: 125-744-8389, 410 WEST WILMINGTON S, RALEIGH, NC, 27601	US Mail (1st Class)
24846	PROGRESS FOR INDUSTRY, PO BOX 192, 201 GRANT STREET, SAEGERTOWN, PA, 16433	US Mail (1st Class)
24846	PROGRESS MOTOR & CONTROL, 463 2ND ST NW, NEW PHILA, OH, 44663-1849	US Mail (1st Class)
24846	PROGRESSIVE BUSINESS PUB., 370 TECHNOLOGY DR., PO BOX 3019, MALVERN, PA, 19355	US Mail (1st Class)
24846	PROGRESSIVE CRANE, INC., 13721 BENNINGTON AVE., CLEVELAND, OH, 44135-4951	US Mail (1st Class)
24846	PROGRESSIVE SYSTEMS, INC., PO BOX 49399, GREENWOOD, SC, 29649-8513	US Mail (1st Class)
24846	PROGRESSIVE TOOL & IND. CO., DBA COMAU PICO, PO BOX 78037, DETROIT, MI, 48278	US Mail (1st Class)
24846	PROHEAT, INC., 117 E ADAMS ST, LAGRANGE, KY, 40031-1229	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	PRO-MOLD LLC, 350 BUELL ROAD, ROCHESTER, NY, 14692	US Mail (1st Class)
24846	PROPANE SERVICES, BOX 35, TAYLOR, MI, 48180	US Mail (1st Class)
24846	PROTOTYPE SOLUTIONS, INC, 1801 E. 79TH STREET, SUITE 9, BLOOMINGTON, MN, 55425	US Mail (1st Class)
24846	PROVANTAGE, 7249 WHIPPLE AVE NW, NORTH CANTON, OH, 44720	US Mail (1st Class)
24846	PRUDENT PUBLISHING, 65 CHALLENGE RD., RIDGEFIELD PARK, NJ, 07660	US Mail (1st Class)
24846	PRYOR SEMINAR, 8700 NW RIVER PARK DR, PARKVILLE, MO, 64152-4358	US Mail (1st Class)
24846	PSM FASTENER CORPORATION, 12223 C R KOON HWY, NEWBERRY, SC, 29108-9354	US Mail (1st Class)
24846	PTX INC., 9841 YORK ALPHA DR., NORTH ROYALTON, OH, 44133	US Mail (1st Class)
24846	PUNDERSON MANOR HOUSE, 11755 KINSMAN RD, NEWBURY, OH, 44065-9691	US Mail (1st Class)
24846	Q TECH, LLC, 5068 FOUR SEASONS RD., SMITHVILLE, TN, 37166	US Mail (1st Class)
24846	QC INSPECTION SERVICES, 11975 PORTLAND AVE SOUTH, SUITE NO. 102, BURNSVILLE, MN, 55337	US Mail (1st Class)
24846	QCR TECH, LLC., 1605 EAST AVIS DR., MADISON HEIGHTS, MI, 48071	US Mail (1st Class)
24846	QIC, INC., PO BOX 66, WRIGHT BROTHERS BRANCH, DAYTON, OH, 45409	US Mail (1st Class)
24846	QIS, INC., 27481 BEVERLY RD, ROMULUS, MI, 48174-2049	US Mail (1st Class)
24846	QLS-NORTH AMERICA LLC, 1790 STONEBRIDGE DR. N, ANN ARBOR, MI, 48108	US Mail (1st Class)
24846	QMI, PO BOX 98184, CHICAGO, IL, 60693	US Mail (1st Class)
24846	QPMR INC., 1027 INDUSTRIAL PARKWAY, MEDINA, OH, 44256	US Mail (1st Class)
24846	QS-9000 OFICIAL MERCHANDISE, 30707 GREENFIELD ROAD, SOUTHFIELD, MI, 48076	US Mail (1st Class)
24846	Q-SHACK, 2604 E LEXINGTON RD., EATON, OH, 45320	US Mail (1st Class)
24846	QUALCHOICE HEALTH PLAN, INC., 6000 PARKLAND BLVD, MAYFIELD HEIGHTS, OH, 44124-6120	US Mail (1st Class)
24846	QUALITY & DURABLE INDUSTRIAL FLOORS, PO BOX 3413, FARMINGTON HILLS, MI, 48333	US Mail (1st Class)
24846	QUALITY ASSISTANCE, LLC, 7480 BROCK RD., PLAIN CITY, OH, 43064	US Mail (1st Class)
24846	QUALITY CALIBRATION SERVICE, 10205 W GREENFIELD AVE., WEST ALLIS, WI, 53214	US Mail (1st Class)
24846	QUALITY CONCEPTS, LLC, 2605 CROOKS ROAD, TROY, MI, 48084	US Mail (1st Class)
24846	QUALITY CONTAINMENT SERVICES, 1547 LYELL AVENUE, ROCHESTER, NY, 14606	US Mail (1st Class)
24846	QUALITY COUNCIL OF INDIANA, 602 WEST PARIS AVENUE, WEST TERRE HAUTE, IN, 47885	US Mail (1st Class)
24846	QUALITY FABRICATED METALS INC., 14000 W MIDDLETOWN RD, SALEM, OH, 44460-9184	US Mail (1st Class)
24846	QUALITY FIRST METROLOGY, SOLUTIONS, 1320 SPRINGBORROW DRIVE, FLINT, MI, 48532-2143	US Mail (1st Class)
24846	QUALITY FIRST SYSTEMS,INC, 14901 32 MILE ROAD, ROMEO, MI, 48065-9419	US Mail (1st Class)
24846	QUALITY LUBRICATION, 806 W US HWY 90, CRESTVIEW, FL, 32536	US Mail (1st Class)
24846	QUALITY MACHINE & AUTOMATION, 184 MANUFACTURERS` DRIVE, HOLLAND, MI, 49424	US Mail (1st Class)
24846	QUALITY MEASURING TOOL REPAIR, 103 W CHURCH STREET, PO BOX 326, CAPAC, MI, 48014	US Mail (1st Class)
24846	QUALITY METAL IMAGES, INC., 1226 RANKIN STREET, TROY, MI, 48083	US Mail (1st Class)
24846	QUALITY PLASTIC, MACHINE REPAIR INC., 1027 INDUSTRIAL PARKWAY, MEDINA, OH, 44256	US Mail (1st Class)
24846	QUALITY PURSUIT, 625 PANORAMA TRAIL, BLDG. 2, SUITE 200, ROCHESTER, NY, 14625	US Mail (1st Class)
24846	QUALITY SPECIALIST GROUP, 3505 BOCA CHICA BLVD., SUITE 178, BROWNSVILLE, TX, 78521	US Mail (1st Class)
24846	QUALITY TECH, INC., 39337 MOUND RD, STERLING HEIGHTS, MI, 48310	US Mail (1st Class)
24846	QUAL-TECH FINISHING, INC., 12201 ARCHBOLD-WHITEHOUSE RD., WHITEHOUSE, OH, 43571	US Mail (1st Class)
24846	QUAL-TECH PRINTING, 8911 DUTCH RD., WATERVILLE, OH, 43566	US Mail (1st Class)
24846	QUEST QUALITY SERVICES, 1545 HOLLAND ROAD, STE. E, MAUMEE, OH, 43537	US Mail (1st Class)
24846	QUICK MOLD & REPAIR, 935 SOUTHEAST AVE., PO BOX 138, TALLMADGE, OH, 44278	US Mail (1st Class)
24846	QUINLAN PUBLISHING CO., 23 DRYDOCK AVENUE, BOSTON, MA, 02210	US Mail (1st Class)
24846	QUINN, BUSECK, LEEMHUIS,, TOOHEY AND KROTO, 2222 WEST GRANDVIEW BLVD., ERIE, PA, 16506-4508	US Mail (1st Class)
24846	R & L ENTERPRISES, 9919 E. WILLER ROAD, DURAND, MI, 48429	US Mail (1st Class)
24846	R & R INC., 1810 INDUSTRIAL PARK DR STE F, GRAND HAVEN, MI, 49417-8930	US Mail (1st Class)
24846	R D MFG. CORPORATION, 4558 50TH STREET SE, KENTWOOD, MI, 49512	US Mail (1st Class)
24846	R&L CARRIERS INC., 375 2ND ST, ELGIN, IL, 60123-7032	US Mail (1st Class)
24846	R. LENNY EWERS, 79 SHERMAN AVE., MANSFIELD, OH, 44906	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	R.A. RODRIGUEZ, 1326 HENRY BRENNAN, EL PASO, TX, 79936	US Mail (1st Class)
24846	R.E. CARROLL INC., 1570 NORTH OLDEN AVE, PO BOX 5806, TRENTON, NJ, 08638	US Mail (1st Class)
24846	R.G. DINE REFUSE, INC., 1921 CLARENDON AVENUE NW, CANTON, OH, 44708	US Mail (1st Class)
24846	R.G. KENRICK, G3530 FLUSHING ROAD, FLINT, MI, 48504	US Mail (1st Class)
24846	R.J.R. & SONS, INC., PARKING LOT MAINT., 12711 FARMINGTON RD., LIVONIA, MI, 48150	US Mail (1st Class)
24846	R.M.A., PO BOX 3147, MEDINA, OH, 44258	US Mail (1st Class)
24846	R.T. VANDERBILT COMPANY, DEPT. 2133, NORTH SUBURBAN, IL, 60132	US Mail (1st Class)
24846	R/D TECH INSTRUMENTS, 48 WOERD AVE, WALTHAM, MA, 02453-3824	US Mail (1st Class)
24846	R/D TECHNICAL INSTRUMENTS, 48 WOERD AVE, WALTHAM, MA, 02453-3824	US Mail (1st Class)
24846	RACHELL DICROSS, 3515 MASSILON RD., SUITE 320, UNIONTOWN, OH, 44685	US Mail (1st Class)
24846	RACING COMMISSION, 37650 PROFESSIONAL CENTER DR, LIVONIA, MI, 48154-1197	US Mail (1st Class)
24846	RACK N` ROLL MATERIAL, HANDLING SOLUTIONS, INC., OR THERESA RESH, 218 CLAIRBORNE CIRCLE, WHITE LAKE, MI, 48383	US Mail (1st Class)
24846	RACKLEY OIL INC., PO DRAWER 826, STARKVILLE, MS, 39760	US Mail (1st Class)
24846	RADIANCE MOLD & ENGR., INC., 28195 KEHRIG DRIVE, CHESTERFIELD, MI, 48047	US Mail (1st Class)
24846	RADIO STATION, 9215 E ASTER DR, SCOTTSDALE, AZ, 85260-4572	US Mail (1st Class)
24846	RADIOLOGY ASSOC. OF WARREN INC, PO BOX 75259, CLEVELAND, OH, 44101-2199	US Mail (1st Class)
24846	RADIRECT, INC., 900 CORPORATE DRIVE, MAHWAH, NJ, 07430	US Mail (1st Class)
24846	RADIX CONTROLS INC., 2105 FASAN DR., OLDCASTLE, ON, NOR 1L0CANADA	US Mail (1st Class)
24846	RAEDER CONSTRUCTION INC., 601 W 15TH STREET, DOVER, OH, 44622	US Mail (1st Class)
24846	RAGAN COMMUNICATIONS,INC., 316 N MICHIGAN AVE., CHICAGO, IL, 60601-9756	US Mail (1st Class)
24846	RAGON ELECTRIC SERVICE, SUPPLY, 1240 DUEBER AVE. SW, CANTON, OH, 44706	US Mail (1st Class)
24846	RALPH A. HILLER CO., 6005 ENTERPRISE DRIVE, EXPORT, PA, 15632	US Mail (1st Class)
24846	RAMADA INN, 6101 WATTSBURG RD., ERIE, PA, 16509	US Mail (1st Class)
24846	RAMSAY CORPORATION, 1050 BOYCE ROAD, PITTSBURGH, PA, 15241-3907	US Mail (1st Class)
24846	RAND NORTH AMERICA, 6320 CANOGA AVENUE, TRILLIUM EAST TOWER, WOODLAND HILLS, CA, 91367-2526	US Mail (1st Class)
24846	RANDSTAD, 2015 S PARK PL SE, ATLANTA, GA, 30339-2058	US Mail (1st Class)
24846	RANDY COULTER, 290 MCCORMICK, LAPEER, MI, 48446	US Mail (1st Class)
24846	RANDY K. AMAKER, 3782 FOX STREET, INKSTER, MI, 48141	US Mail (1st Class)
24846	RANDY LACHOWSKI, 20 NIAGARA PIER, ERIE, PA, 16507	US Mail (1st Class)
24846	RANSOM AND BENJAMIN, 21 E MAIN ST, MYSTIC, CT, 06355-2647	US Mail (1st Class)
24846	RAPID GRANULATOR, INC., PO BOX 5887, ROCKFORD, IL, 61125-0887	US Mail (1st Class)
24846	RAPID MOLD SOLUTIONS, 4820 PACIFIC AVE., ERIE, PA, 16506	US Mail (1st Class)
24846	RAPIDFORMS INC., 301 GROVE ROAD, THOROFARE, NJ, 08086	US Mail (1st Class)
24846	RATLIFF & TAYLOR, INC., 31105 BAINBRIDGE RD., STE. 4, SOLON, OH, 44139	US Mail (1st Class)
24846	RAVAGO-HINDS, 25 FOUNDATION PLACE, BARBERTON, OH, 44203	US Mail (1st Class)
24846	RAY GEKLER, 900 S. WILEY, PO BOX 320, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	RAYL INDUSTRIAL SUPPLY CO., 1625 EAST AVIS DRIVE, MADISON HEIGHTS, MI, 48071	US Mail (1st Class)
24846	RAYTECH CORP., 12833 COLLECTIONS CENTER DR., CHICAGO, IL, 60693	US Mail (1st Class)
24846	RC SIMPSON, INC., 5950 FAIRVIEW RD, SUITE 604, CHARLOTTE, NC, 28210	US Mail (1st Class)
24846	READY TECHNOLOGY, 333 PROGRESS RD, DAYTON, OH, 45449-2321	US Mail (1st Class)
24846	REBA MARKHAM, 428 LUM-FIFE ROAD, GORDO, AL, 35466	US Mail (1st Class)
24846	RECORD PUBLISHING CO., 206 W. ERIE ST., PO BOX 70, KENT, OH, 44240	US Mail (1st Class)
24846	RECYCLING CENTER, PO BOX 640749, CINCINNATI, OH, 45264-0749	US Mail (1st Class)
24846	RED LOBSTER, 2322 W 4TH STREET, ONTARIO, OH, 44862	US Mail (1st Class)
24846	RED MAPLE INN, 14707 S. CHESHIRE ST., BURTON, OH, 44021	US Mail (1st Class)
24846	REECE FLUID POWER COMPANY, 5245 SECOR RD., UNIT 1, PO BOX 8751, TOLEDO, OH, 43623	US Mail (1st Class)
24846	REEK CORP, 1100 PRODUCE ST, PO BOX 1648, HIDALGO, TX, 78557	US Mail (1st Class)
24846	REEL-NEAT SYSTEMS, INC., 6408 S EASTERN AVE, OKLAHOMA CITY, OK, 73149-5134	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	REG WILKINS, 44 NIAGARA DR., OSHAWA, ON, L1G 8A5CANADA	US Mail (1st Class)
24846	REGAL INDUSTRIES, INC., 9564 EAST 600 SOUTH, CROTHERSVILLE, IN, 47229	US Mail (1st Class)
24846	REGAL TOURS, MARTHA DIETRICH, PO BOX 245, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	REGENSTRIEF BOYS AND GIRLS CLUB, ATTN: GOLF, 825 CENTRAL AVENUE, CONNERSVILLE, IN, 47331	US Mail (1st Class)
24846	REGIS INDUSTRIAL SUPPLY, PO BOX 39038, N RIDGEVILLE, OH, 44039-0038	US Mail (1st Class)
24846	REGISTER NOW, 9625 W 76TH ST, EDEN PRAIRIE, MN, 55344-3765	US Mail (1st Class)
24846	REGISTRAR ACCREDITATION BOARD, 600 N PLANKINTON AVE, MILWAUKEE, WI, 53203-2914	US Mail (1st Class)
24846	REH ASSOCIATES, 21637 MELROSE AVE, SOUTHFIELD, MI, 48075	US Mail (1st Class)
24846	REHAU INC., 1501 EDWARDS FERRY RD NE, LEESBURG, VA, 20176-6680	US Mail (1st Class)
24846	REID TOOL SUPPLY CO., 2265 BLACK CREEK RD, MUSKEGON, MI, 49444-2673	US Mail (1st Class)
24846	REINSHAW,INC, DEPT. 77-6758, CHICAGO, IL, 60678-6758	US Mail (1st Class)
24846	REKO TOOL & MOULD, 5390 BRENDAN LANE, OLDCASTLE, ON, NOR ILOCANADA	US Mail (1st Class)
24846	REKO TOOL & MOULD INC., 5390 BRENDAN LANE RR 1, OLDCASTLE, ON, NOR 1L0CANADA	US Mail (1st Class)
24846	RELIABLE ANALYSIS, 379 INDUSCO COURT, TROY, MI, 48083	US Mail (1st Class)
24846	RELIABLE OFFICE SUPPLY, DEPT. 8001, 135 LASALLE ST., CHICAGO, IL, 60674-8001	US Mail (1st Class)
24846	RELIZON COMPANY, WORKFLOWONE, 220 E MONUMENT AVE, DAYTON, OH, 45402-1223	US Mail (1st Class)
24846	REM ELECTRONICS SUPPLY CO., 525 S. PARK, BOX 831, WARREN, OH, 44483	US Mail (1st Class)
24846	REM SALES, 34 BRADLEY PARK RD, EAST GRANBY, CT, 06026-9789	US Mail (1st Class)
24846	RENAISSANCE THEATER, 138 PARK AVENUE WEST, MANSFIELD, OH, 44902	US Mail (1st Class)
24846	RENCO MOLD, 2801 OME AVENUE, DAYTON, OH, 45414	US Mail (1st Class)
24846	RENEE L REYNIER, 4783 APACHE TRAIL, COLUMBIAVILLE, MI, 48421	US Mail (1st Class)
24846	RENEE L. VOGEL, 10273 ROSE STREET, LAKE CITY, PA, 16423	US Mail (1st Class)
24846	RENISHAW, INC., 6758 EAGLE WAY, CHICAGO, IL, 60678-1067	US Mail (1st Class)
24846	RENNCO AUTOMATION SYSTEMS, 971 HAMILTON DR, HOLLAND, OH, 43528-8211	US Mail (1st Class)
24846	RENNER PLUMBING & SUPPLY, 1100 NAVARRE ROAD S.W., CANTON, OH, 44706	US Mail (1st Class)
24846	RENTAL CORRAL, 840 MAHONING AVE., WARREN, OH, 44483	US Mail (1st Class)
24846	RENTAL SERVICE CORP., 6929 E GREENWAY PKWY STE 200, SCOTTSDALE, AZ, 85254-2171	US Mail (1st Class)
24846	REP CORPORATION, 8N470 TAMELING COURT, PO BOX 8146, BARTLETT, IL, 60103-8146	US Mail (1st Class)
24846	REPROGRAPHICS ONE INC., 36060 INDUSTRIAL ROAD, LIVONIA, MI, 48150	US Mail (1st Class)
24846	REPROS, INC., 7219 WHIPPLE AVENUE, NW, NORTH CANTON, OH, 44720	US Mail (1st Class)
24846	REQUIPCO, INC., 9700 CLEVELAND AVE. SOUTH, EAST SPARTA, OH, 44626	US Mail (1st Class)
24846	RESERVE ACCOUNT, ACCT # 18723056, PO BOX 856056, LOUISVILLE, KY, 40285-6056	US Mail (1st Class)
24846	RESIN EXPRESS, UNIT 11, THE VALLEY CENTRE, GORDON ROAD, HIGH WYCOMBE, BUCKINGHAMSHIRE, HP13 6EQUNITED KINGDOM	US Mail (1st Class)
24846	RESPOND FIRST AID, 8320 PEBBLE CREEK CT, CHAGRIN FALLS, OH, 44023-4865	US Mail (1st Class)
24846	RESPONSE MECHANICAL, 4981 TRINITY CHURCH ROAD, DAYTON, OH, 45415	US Mail (1st Class)
24846	RETCO INDUSTRIES CORP., 981 HOME AVE., AKRON, OH, 44310	US Mail (1st Class)
24846	RETURNABLE PKG RESOURCE, P. O. BOX 381322, BIRMINGHAM, AL, 35238	US Mail (1st Class)
24846	REVENUE DEPT, 518 19TH AVE, TUSCALOOSA, AL, 35401-1510	US Mail (1st Class)
24846	REVERE PRODUCTS, PO BOX 35311, CLEVELAND, OH, 44135-0311	US Mail (1st Class)
24846	REX PIPE & SUPPLY CO. MANSFIEL, 294 E 3RD ST, MANSFIELD, OH, 44902-1535	US Mail (1st Class)
24846	REXEL ELECTRICAL, 6700 LBJ FWY STE 3200, DALLAS, TX, 75240-6526	US Mail (1st Class)
24846	REXEL SOUTHERN, 3106 GREENSBORO AVE., TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	REYNOLDS MACHINERY, 760 LIBERTY LANE, DAYTON, OH, 45449	US Mail (1st Class)
24846	RHB SORTING SERVICES, 5940 COMMERCE, W. BLOOMFIELD, MI, 48324	US Mail (1st Class)
24846	RHEIN CHEMIE CORP., 111 RIDC PARK WEST DR, PITTSBURGH, PA, 15275-1112	US Mail (1st Class)
24846	RHEIN CHEMIE CORPORATION, 111 RIDC PARK WEST DR, PITTSBURGH, PA, 15275-1112	US Mail (1st Class)
24846	RHETECH, DEPT. 77214, PO BOX 77000, DETROIT, MI, 48277-0214	US Mail (1st Class)
24846	RHM FLUID POWER, 375 MANUFACTURERS DR, WESTLAND, MI, 48186-4038	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	RICH NEWSOME, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	RICHARD BOWER, 1401 INDUSTRIAL PARK, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	RICHARD DAVIS, 359 HICKORY LANE, MOUNDVILLE, AL, 35474	US Mail (1st Class)
24846	RICHARD EQUIPMENT CO., WOODLAND TECH CENTER, 1008 SEABROOK WAY, CINCINNATI, OH, 45245-1963	US Mail (1st Class)
24846	RICHARD FRANCIS, 119 E DAYTON STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	RICHARD HETHERINGTON, 1609 PLEASANT POINTE CIR, BRYANT, AR, 72022-3948	US Mail (1st Class)
24846	RICHARD J. HOOVER, 1245 DETWILER DRIVE, YORK, PA, 17404	US Mail (1st Class)
24846	RICHARD J. PRANGER, 31572 INDUSTRIAL RD., SUITE #300, LIVONIA, MI, 48150	US Mail (1st Class)
24846	RICHARD MILLER, 10400 DAVISON RD, DAVISON, MI, 48423	US Mail (1st Class)
24846	RICHARD PAGE, 2628 COVENTRY BLVD NE, CANTON, OH, 44705	US Mail (1st Class)
24846	RICHARDS LANDSCAPING, 5451 CRAWFORD SENECA LINE RD., PO BOX 400, NEW WASHINGTON, OH, 44854	US Mail (1st Class)
24846	RICHARDSON`S TRANSPORT, 112 WYNONA DR., EATON, OH, 45320	US Mail (1st Class)
24846	RICHLAND NEWHOPE, 150 EAST FOURTH STREET, PO BOX 916, MANSFIELD, OH, 44902	US Mail (1st Class)
24846	RICHLAND SEALCOATING CO., 4156 PARK AVE. WEST, MANSFIELD, OH, 44903	US Mail (1st Class)
24846	RICK CHAMBERS, 1100 N OPDYKE RD, STE. 200, AUBURN HILLS, MI, 48326	US Mail (1st Class)
24846	RICK FREDERICK, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	RICK LACHAT, 119 E DAYTON STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	RICK SZOSZOREK, C/O BUNDY, 1901 WAGER RD, ERIE, PA, 16509	US Mail (1st Class)
24846	RICK WARGAL, CEP PRODUCTS LLC, 31557 SCHOOLCRAFT RD STE 100, LIVONIA, MI, 48150-1847	US Mail (1st Class)
24846	RIDGE-U-RAK, INC., 12340 GAY RD, NORTH EAST, PA, 16428-3576	US Mail (1st Class)
24846	RIDGEVIEW FARM, C/O STEPHEN D. GROVER, 5488 KINSMAN ROAD, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	RIFKIN STEEL & ALUMINUM, 1443 N NIAGRA STREET, SAGINAW, MI, 48602-4782	US Mail (1st Class)
24846	RIGGER LIFT MFG., INC., 1255 BALMORAL RD., CAMBRIDGE, ON, N1T 1C4CANADA	US Mail (1st Class)
24846	RITA HARKEY, 832 2ND AVE SOUTHWEST, FAYETTE, AL, 35555	US Mail (1st Class)
24846	RITE AID PHARMACIES, 15596 W HIGH STREET, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	RITTER ENGINEERING COMPANY, 23717 RESEARCH DRIVE, FARMINGTON HILLS, MI, 48335-2625	US Mail (1st Class)
24846	RITTER TECHNOLOGIES, 1952 PAINTERS RUN RD, PITTSBURGH, PA, 15241-3014	US Mail (1st Class)
24846	RIXAN ASSOCIATES, INC., 7560 PARAGON RD., DAYTON, OH, 45459	US Mail (1st Class)
24846	RJG TECHNOLOGIES, 3111 PARK DRIVE, TRAVERSE CITY, MI, 49686	US Mail (1st Class)
24846		US Mail (1st Class)
24846	RNR LAND INC., PO BOX 768, ASHTABULA, OH, 44005	US Mail (1st Class)
24846	ROADWAY EXPRESS, 1077 GORGE BLVD, AKRON, OH, 44310-2408	US Mail (1st Class)
24846	ROADWAY EXPRESS INC., 1330 HENRY BRENNAN, EL PASO, TX, 79936	US Mail (1st Class)
24846	ROB WHITNEY, PO BOX 320, 900 S. WILEY STREET, CRESTLINE, OH, 44827-0302	US Mail (1st Class)
24846	ROB'S WELDING, 2920 PRODUCTION COURT, DAYTON, OH, 45414	US Mail (1st Class)
24846	ROBERT CRAMER, 716 N MAIN, PERRYVILLE, MO, 63775	US Mail (1st Class)
24846	ROBERT E. MILLS, C/O CEP PRODUCTS, LLC, PO BOX 320, CRESTLINE, OH, 44827-0320	US Mail (1st Class)
24846	ROBERT L. COTTON, 900 S. WILEY ST., PO BOX 320, CRESTLINE, OH, 44820	US Mail (1st Class)
24846	ROBERT L. HARVEY, ASHTABULA COUNTY TREASURER, 25 WEST JEFFERSON STREET, JEFFERSON, OH, 44047	US Mail (1st Class)
24846	ROBERT L. WILLIAMS, 42850 MOUND RD., STERLING HEIGHTS, MI, 48314	US Mail (1st Class)
24846	ROBERT LANGFORD, 1401 INDUSTRIAL PARK DRIVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	ROBERT LEE, 3131 COLUMBUS RD., CANTON, OH, 44705	US Mail (1st Class)
24846	ROBERT MANUFACTURING, DEPT. #2668, LOS ANGELES, CA, 90084-2668	US Mail (1st Class)
24846	ROBERT PFROGNER, 136 RIVER RD, DELLE VERNON, PA, 15012	US Mail (1st Class)
24846	ROBERT POYNTER, 230 TOWNSHIP ROAD, JEROMEVILLE, OH, 44840	US Mail (1st Class)
24846	ROBERT ROBINSON, 1249 44TH STREET SE, GRAND RAPIDS, MI, 49508	US Mail (1st Class)
24846	ROBERT ROHARIK, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	ROBERT TURNER, 5127 COLDWATER RD, LAPEER, MI, 48446	US Mail (1st Class)
24846	ROBERTA SUTTON, 111 THIRD AVE., GALION, OH, 44833	US Mail (1st Class)
24846	ROBERTO ARMANDO SANTANA FLORES, CHUCHO MARTINEZ GIL, COL. CHURUBUSCO CP., 31120MEXICO	US Mail (1st Class)
24846	ROBINSON INDUSTRIES, INC., 3051 CURTICE ROAD, COLEMAN, MI, 48618	US Mail (1st Class)
24846	ROBOHAND, 482 PEPPER STREET, MONROE, CT, 06468-2653	US Mail (1st Class)
24846	ROBOTIC ACCESSORIES, 6555 S STATE ST., ROUTE 202, TIPP CITY, OH, 45371	US Mail (1st Class)
24846	ROBOTIC PROD. TECH, 1255 HARMON ROAD, AUBURN HILLS, MI, 48326	US Mail (1st Class)
24846	ROBOTICS INC., 2421 ROUTE 9, BALLSTON SPA, NY, 12020-4407	US Mail (1st Class)
24846	ROBS CATERING, 705 ARLINGTON ROAD, BROOKVILLE, OH, 45309	US Mail (1st Class)
24846	ROCKBROOK PRODUCTS INC., 28 SOUTH MAIN STREET, #123, RANDOLPH, MA, 02368	US Mail (1st Class)
24846	ROCKET ENTERPRISES, INC., 30660 RYAN ROAD, WARREN, MI, 48092	US Mail (1st Class)
24846	ROCKHURST COLLEGE, CONTINUING EDUCATION CTR., INC, PO BOX 419107, KANSAS CITY, MO, 64141-6107	US Mail (1st Class)
24846	ROCKHURST UNIVERSITY, 1100 ROCKHURST RD, KANSAS CITY, MO, 64110-2508	US Mail (1st Class)
24846	ROCKWELL AUTOMATION, 1201 S 2ND ST, MILWAUKEE, WI, 53204-2410	US Mail (1st Class)
24846	ROCKWOOD AUTOMATIC MACHINE, 55 FESSENDEN ST., ROCHESTER, NY, 14611	US Mail (1st Class)
24846	ROCKY'S CRANE SERVICE, 15400 OLD STATE RD, MIDDLEFIELD, OH, 44062-9260	US Mail (1st Class)
24846	ROD R. SELOGY, 6340 BRINKER ST. SW, NAVARRE, OH, 44662	US Mail (1st Class)
24846	RODEN ELECTRIC-TUSCALOOSA, 170 MABRY HOOD RD, KNOXVILLE, TN, 37922-2211	US Mail (1st Class)
24846	RODERICK MULLINS, 232 2ND STREET, SANTEE, SC, 29142	US Mail (1st Class)
24846	RODNEY SHAFFER, 5276 OAKVALE ST. SW, CANTON, OH, 44706	US Mail (1st Class)
24846	ROG`S, INC., PO BOX 1026, 327 EAST 18TH STREET, ERIE, PA, 16512-1026	US Mail (1st Class)
24846	ROGER A HEDDERICK, PO BOX 1026, ERIE, PA, 16512	US Mail (1st Class)
24846	ROGERS COMPANY, 7550 TYLER BLVD, MENTOR, OH, 44060-4868	US Mail (1st Class)
24846	ROGUE MANUFACTURING, 304 STOTLER ROAD, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	ROHM & HAAS COMPANY, 100 INDEPENDENCE MALL WEST, PHILADELPHIA, PA, 19106-2399	US Mail (1st Class)
24846	ROILGARD, 5600 THIRTEENTH ST, MENOMINEE, MI, 49858	US Mail (1st Class)
24846	ROLL IN SAW, 15401 COMMERCE PARK DR., BROOKPARK, OH, 44142	US Mail (1st Class)
24846	ROLLER FABRICS, 6600 W CALUMET RD, MILWAUKEE, WI, 53223-4185	US Mail (1st Class)
24846	ROLLING HILLS, 3274 DAVISON ROAD, LAPEER, MI, 48446	US Mail (1st Class)
24846	ROLL-KRAFT, 8901 TYLER BLVD, MENTOR, OH, 44060-2184	US Mail (1st Class)
24846	ROMAIN MASONRY, INC., 400 KIRKWOOD DRIVE, NORTHPORT, AL, 35473	US Mail (1st Class)
24846	RON FRYE, 900 S WILEY STREET, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	RON NEWNHAM, 1925 38TH ST NE, CANTON, OH, 44705	US Mail (1st Class)
24846	RON WEISMILLER, 4395 MAGNOLIA CT., ERIE, PA, 16510	US Mail (1st Class)
24846	RONALD L. LINSENBIGLER, 1053 HOLLAND AVE NE, EAST CANTON, OH, 44730	US Mail (1st Class)
24846	RONDY & COMPANY, INC., 255 WOOSTER RD. NORTH, BARBERTON, OH, 44203	US Mail (1st Class)
24846	RONICK'S SPORTS & APPAREL, 1261 HARMAN RD, ASHTABULA, OH, 44004	US Mail (1st Class)
24846	ROONEY OPTICAL, INC., 5440 W. 164TH ST., CLEVELAND, OH, 44142	US Mail (1st Class)
24846	ROSA'S OFFICE PLUS, 20 SOUTH 11 STREET, PO BOX 369, RICHMOND, IN, 47375-0369	US Mail (1st Class)
24846	ROSCOM, 2925 STATE ROAD, CROYDON, PA, 19021	US Mail (1st Class)
24846	ROSCOM, INC. W510104, 2925 STATE RD, CROYDON, PA, 19021-6960	US Mail (1st Class)
24846	ROSE FANNING, 23420 BELMONT DR, CLEVELAND, OH, 44145-2709	US Mail (1st Class)
24846	ROSEMARY TOROK, 13277 BASS LAKE RD., CHARDON, OH, 44024	US Mail (1st Class)
24846	ROSS ASPHALT PAVING SYSTEMS, 2150 N DORT HWY., FLINT, MI, 48506	US Mail (1st Class)
24846	ROTO ROOTER, 1200 12TH ST. NW, CANTON, OH, 44703	US Mail (1st Class)
24846	ROTOGRAN INTERNATIONAL INC., 3 BRADWICK DRIVE, CONCORD, ON, L4K 2T4CANADA	US Mail (1st Class)
24846	ROW ELECTRICAL EQUIPMENT, INC., 3300 BISHOP ST, TOLEDO, OH, 43606-4219	US Mail (1st Class)
24846	ROXANNE SPRINGSTEEN, 182 JAMESTOWNE LANE, DAVISON, MI, 48423	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	ROY HARRISON, 119 EAST DAYTON STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	ROY ROBINSON, 1292 1/2 BROADWAY, FLINT, MI, 48506	US Mail (1st Class)
24846	ROYAL SOUND INC., 1455 WINONA RD., STOW, OH, 44224	US Mail (1st Class)
24846	ROYALITE, 101 BURTON ST., FLINT, MI, 48502	US Mail (1st Class)
24846	RTI LABORATORIES, INC., 31628 GLENDALE, LIVONIA, MI, 48150	US Mail (1st Class)
24846	RTI TECHNOLOGIES INC, PO BOX 3099, YORK, PA, 17402-0099	US Mail (1st Class)
24846	RU VAN, POBOX 930855, ATLANTA, GA, 31193-0855	US Mail (1st Class)
24846	RUA REPORTING SERVICE, SUITE 1009, 55 PUBLIC SQUARE, CLEVELAND, OH, 44113-1967	US Mail (1st Class)
24846	RUBBER & PLASTICS NEWS, SUBSCRIPTION DEPARTMENT, PO BOX 07940, DETROIT, MI, 48207-9905	US Mail (1st Class)
24846	RUBBER DIVISION-ACS, AMERICAN CHEMICAL SOCIETY, 1155 16TH ST NW, WASHINGTON, DC, 20036-4892	US Mail (1st Class)
24846	RUBBER MOLDING TECHNOLOGY, 36036 LAKELAND BLVD., EASTLAKE, OH, 44095	US Mail (1st Class)
24846	RUBBERMAID INC., RT 1 INDUSTRIAL PK, CENTERVILLE, IA, 52544	US Mail (1st Class)
24846	RUBIN BROWNLEE, 8836 NOTTINGHAM DR, SUPERIOR TOWNSHIP, MI, 48198-3226	US Mail (1st Class)
24846	RUDY MIES, 1165 BECK RD., GALION, OH, 44833	US Mail (1st Class)
24846	RUMPKE, PO BOX 538708, CINCINNATI, OH, 45381	US Mail (1st Class)
24846	RUPP VALLEY FARM, 6700 LINCOLN HWY., CRESTLINE, OH, 44827	US Mail (1st Class)
24846	RURAL/METRO AMBULANCE, 4349 MONTICELLO BLVD, SOUTH EUCLID, OH, 44121	US Mail (1st Class)
24846	RUSH TRANSPORTATION, 2388 ARBOR BLVD, DAYTON, OH, 45439-1745	US Mail (1st Class)
24846	RUSS GIBBS, 900 S. WILEY STREET, PO BOX 320, CRESTLINE, OH, 44827-0320	US Mail (1st Class)
24846	RUSS RUTHERFORD, 4783 APACHE TRAIL, COLUMBIAVILLE, MI, 48421	US Mail (1st Class)
24846	RUSSEL W. STROUP, 1452 VALLEY DRIVE, LAPEER, MI, 48446	US Mail (1st Class)
24846	RUSSELL T. GILMAN, INC., BIN #87, MILWAUKEE, WI, 53288	US Mail (1st Class)
24846	RUSSELL W. KLING, 3911 YORKLAND DRIVE, APT 11, COMSTOCK PARK, MI, 49321	US Mail (1st Class)
24846	RUSSELL WRIGHT, 2501 LIPPINCOTT BLVD, LOT 107, FLINT, MI, 48507	US Mail (1st Class)
24846	RUSTY SCHWENDEMAN, 16865 HUNTLEY RD, HUNTSBURG, OH, 44046-9767	US Mail (1st Class)
24846	RUTH KARSGAARD, 2565 SOUTH WAYNE RD., WESTLAND, MI, 48186	US Mail (1st Class)
24846	RYAN ALEXANDER, 119 EAST DAYTON ST., WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	RYAN ALTERNATIVE STAFFING, 4489 MARKET ST, YOUNGSTOWN, OH, 44512-1513	US Mail (1st Class)
24846	RYAN ENTERPRISE, 1465 LINNET RD., PO BOX 3100, WRIGHTWOOD, CA, 92397	US Mail (1st Class)
24846	RYAN T. ALEXANDER, 1051 COURTNEY DRIVE, APT D, RIVERSIDE, OH, 45431	US Mail (1st Class)
24846	RYDER TRANSPORTATION SERVICES, 11690 NW 105TH ST, MEDLEY, FL, 33178-1103	US Mail (1st Class)
24846	RYDER TRUCK RENTAL, INC., 11690 NW 105TH ST, MEDLEY, FL, 33178-1103	US Mail (1st Class)
24846	RYTAM TECHNOLOGY, 1749 W HAMLIN, ROCHESTER HILLS, MI, 48309	US Mail (1st Class)
24846	S & R EQUIPMENT CO., 5517 DISTRIBUTION DR, FORT WAYNE, IN, 46825-5115	US Mail (1st Class)
24846	S & S INC., 21300 ST. CLAIR AVE, CLEVELAND, OH, 44117	US Mail (1st Class)
24846	S & S PACKAGING PRODUCTS, INC., 10549 CROSBY CIR, CRANESVILLE, PA, 16410-9337	US Mail (1st Class)
24846	S & Z TOOL & DIE, 3180 BEREA RD, CLEVELAND, OH, 44111-1501	US Mail (1st Class)
24846	S AND N PALLETS, INC., 5610 LINDENTREE RD., MINERAL CITY, OH, 44656	US Mail (1st Class)
24846	S M T INDUSTRIES, INC., 2540 ROSS STREET, SIDNEY, OH, 45365-0766	US Mail (1st Class)
24846	S S LANDSCAPE, 17640 NASH ROAD, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	S&K ASPHALT & CONCRETE, INC., 2275 MANCHESTER RD, AKRON, OH, 44314-3745	US Mail (1st Class)
24846	S&S COMPANY OF GEORGIA, 827 PINE AVE, ALBANY, GA, 31701-2460	US Mail (1st Class)
24846	S.A. COMMUNALE CO., INC., 2900 NEWPARK DR., PO BOX 150, BARBERTON, OH, 44203	US Mail (1st Class)
24846	S.D. MYERS, INC., 180 SOUTH AVE, TALLMADGE, OH, 44278-2813	US Mail (1st Class)
24846	S.G. MORRIS CO., 27439 HOLIDAY LANE, PERRYSBURG, OH, 43551	US Mail (1st Class)
24846	S.G. MORRIS COMPANY, 699 MINER RD, CLEVELAND, OH, 44143-2115	US Mail (1st Class)
24846	SABO FASTENERS & SUPPLY, UNIT D, 64 EAST MAIN STREET, LEXINGTON, OH, 44905	US Mail (1st Class)
24846	SACS CONSULTING &, INVESTIGATIVE SERVICES, 400 N CLEVELAND AVE., MOGADORE, OH, 44260	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	SADDLE ISLAND INSTITUTE, 126 STATE STREET, BOSTON, MA, 02109	US Mail (1st Class)
24846	SAE INTERNATIONAL, 400 COMMONWEALTH DRIVE, WARRENDALE, PA, 15096	US Mail (1st Class)
24846	SAFECO, 3700 NORTHBROOK DR. SUITE 11, NORTHPORT, AL, 35473	US Mail (1st Class)
24846	SAFETY KLEEN SYSTEMS, 5400 LEGACY DR, BUILDING # 3, PLANO, TX, 75024	US Mail (1st Class)
24846	SAFETY SIGN COMPANY, 13700 PROSPECT RD, STRONGSVILLE, OH, 44149-3862	US Mail (1st Class)
24846	SAFETY SOLUTIONS, INC., 6161 SHAMROCK CT, DUBLIN, OH, 43016-1275	US Mail (1st Class)
24846	SAFETY TODAY, INC., 2425 SPIEGEL DR STE A, GROVEPORT, OH, 43125-9112	US Mail (1st Class)
24846	SAFETY-KLEEN CORP., ATTN: KATHY, 1002 HOOKE AVE, DOLOMITE, AL, 35061	US Mail (1st Class)
24846	SAIA MOTOR FREIGHT LINE INC., 11465 JOHNS CREEK PKWY STE 400, DULUTH, GA, 30097-1572	US Mail (1st Class)
24846	SAILOR AUTOMATION INC., 981 VIA RODEO, PLACENTIA, CA, 92870	US Mail (1st Class)
24846	SAINT-GOBAIN/NORTON PERF. PLAS, 1 SEALANTS PARK, GRANVILLE, NY, 12832-1652	US Mail (1st Class)
24846	SALCO MACHINE, INC., 3822 VICTORY ROAD N.E., LOUISVILLE, OH, 44641	US Mail (1st Class)
24846	SALESMASTER CORPORATION, 197 CANAL ROAD WEST, FAIRLESS HILLS, PA, 19030	US Mail (1st Class)
24846	SALLY SILBAUGH, 6900 THOMAS RD., MCKLEAN, PA, 16426	US Mail (1st Class)
24846	SALVATION ARMY, 270 FRANKLIN STREET, WARREN, OH, 44483	US Mail (1st Class)
24846	SALVATORE ZEOLLA, 4569 W 38TH STREET, ERIE, PA, 16506	US Mail (1st Class)
24846	SAM BERNIK, 1901 WAGER RD, ERIE, PA, 16509	US Mail (1st Class)
24846	SAM`S CLUB, 4790 PORTAGE STREET, NORTH CANTON, OH, 44720	US Mail (1st Class)
24846	SAM`S CLUB, 27853 CHARDON RD., WICKLIFFE, OH, 44092	US Mail (1st Class)
24846	SAM'S CLUB, 608 SW 8TH ST, BENTONVILLE, AR, 72712-6207	US Mail (1st Class)
24846	SAMIR A. AL-MANSORY, 254 EAST 3RD STREET, ERIE, PA, 16507	US Mail (1st Class)
24846	SAMUEL CHIPNER III, 207 N HENRY, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	SANDRA BRUNSKOLE, 8707 SAMUEL BARTON DR, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	SANDRA HENNING, 9363 FOSTER RD., FOSTORIA, MI, 48435	US Mail (1st Class)
24846	SANDRA K. TERRASI, 4411 OAK ARBOR COURT NE, GRAND RAPIDS, MI, 49525	US Mail (1st Class)
24846	SANDRA LENK, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	SANDUSKY ELECTRIC, INC., 1516 MILAN RD, SANDUSKY, OH, 44870-4135	US Mail (1st Class)
24846	SANDUSKY MACHINE & TOOL, 2223 TIFFEN AVE, SANDUSKY, OH, 44670	US Mail (1st Class)
24846	SANTEK ENVIRONMENTAL, 650 25TH STREET NW, SUITE 100, CLEVELAND, TN, 37311	US Mail (1st Class)
24846	SANYO PLASTICS COMPOUND AMERICA,INC, 765 THIRD AVE, SUITE 300-14, CHULA VISTA, CA, 91910	US Mail (1st Class)
24846	SARAH E. GARNER, 235 EAST AVENUE, EATON, OH, 45320	US Mail (1st Class)
24846	SARNAMOTIVE BLUE WATER INC., DEPT. 257001, PO BOX 67000, DETROIT, MI, 48267-2570	US Mail (1st Class)
24846	SARTAIN LUMBER, 14684 HIGHWAY 82 W, COKER, AL, 35452-3503	US Mail (1st Class)
24846	SARTOMER COMPANY, 502 THOMAS JONES WAY, EXTON, PA, 19341-2530	US Mail (1st Class)
24846	SAUDER SEAMLESS SPOUTING, 2620 CRIDER ROAD, MANSFIELD, OH, 44903	US Mail (1st Class)
24846	SAV ENTERPRISES, INC., 8786 35W WEST SERVICE DRIVE NE, BLAINE, MN, 55449	US Mail (1st Class)
24846	SAVAGE & SAVAGE, INC., PO BOX 121, MONROEVILLE, OH, 44847	US Mail (1st Class)
24846	SBC, BILL PAYMENT CENTER, SAGINAW, MI, 48663-0003	US Mail (1st Class)
24846	SBC CAPITAL SERVICES, PO BOX 71614, CHICAGO, IL, 60694	US Mail (1st Class)
24846	SBC LONG DISTANCE, PO BOX 660688, DALLAS, TX, 75266-0688	US Mail (1st Class)
24846	SC EMPLOYMENT SECURITY COMMISSION, PO BOX 7103, COLUMBIA, SC, 29202	US Mail (1st Class)
24846	SC EMPLOYMENT SECURITY COMMISSION, 1550 GADSDEN ST, COLUMBIA, SC, 29202	US Mail (1st Class)
24846	SCAGLIA AMERICA INC., 3630 GREEN PARK CIRCLE, CHARLOTTE, NC, 28217	US Mail (1st Class)
24846	SCHMID MECHANICAL, INC., 1801 MARTIN LUTHER KING JR DR, CLEVELAND, OH, 44106-1757	US Mail (1st Class)
24846	SCHMID PLUMBING & HEATING, 207 N HILLCREST DR, WOOSTER, OH, 44691-3720	US Mail (1st Class)
24846	SCHWAAB, INC., 11415 W BURLEIGH ST, MILWAUKEE, WI, 53222-3217	US Mail (1st Class)
24846	SCLMC, 6200 FRANK AVE. NW, CANTON, OH, 44720	US Mail (1st Class)
24846	SCORI MOLD & ENG, INC, 688 E. MAIN ST, CENTREVILLE, MI, 49032	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	SCOT PUMP CO., 77 SW 20TH STREET, FORT LAUDERDALE, FL, 33315	US Mail (1st Class)
24846	SCOTT KINGSLEY, ADDRESS UNAVAILABLE, CHARDON, OH, 44024	US Mail (1st Class)
24846	SCOTT LEVASSEUR, 290 MCCORMICK, LAPEER, MI, 48446	US Mail (1st Class)
24846	SCOTT PAULOT, 31572 INDUSTRIAL RD., SUITE #300, LIVONIA, MI, 48150	US Mail (1st Class)
24846	SCOTT STUMBO, 1729 SOUTH FORMAN RD, JEFFERSON, OH, 44047	US Mail (1st Class)
24846	SCOTT V. PERKINS, 15307 RENSHAW DRIVE, CLINTON TWP, MI, 48038	US Mail (1st Class)
24846	SCR-EXECUTIVE SEARCH, RECRUITING CO., 6750 PONTEBERRY ST. NW, CANTON, OH, 44718	US Mail (1st Class)
24846	SCR-EXECUTIVE SEARCH &, RECRUITING COMPANY, 6750 PONTEBERRY ST. NW, SUITE 101, CANTON, OH, 44718	US Mail (1st Class)
24846	SCUDDER INVESTMENT SERVICE CO., ATTN: SCUDDER RETIREMENT SVC., PO BOX 219197, KANSAS CITY, MO, 64121-9197	US Mail (1st Class)
24846	SD MYERS, 180 SOUTH AVE, TALLMADGE, OH, 44278-2813	US Mail (1st Class)
24846	SE BE PALLET, 148 CATHCART ST., PO BOX 1183, BLENHEIM, ON, NOP 1A0CANADA	US Mail (1st Class)
24846	SEAL SOURCE INC., 31408 HARPER AVE., SUITE 172, SAINT CLAIR SHORES, MI, 48082	US Mail (1st Class)
24846	SEAN ENSINGER, 1149 CHATWELL DR., DAVISON, MI, 48423	US Mail (1st Class)
24846	SEARCH AMERICA, INC., 459 NORTH GILBERT RD, SUITE B-140, GILBERT, AZ, 85234	US Mail (1st Class)
24846	SEARCH MASTERS, INC., 1148 EUCLID AVE STE 509, CLEVELAND, OH, 44115-1604	US Mail (1st Class)
24846	SEARS, ADDRESS UNAVAILABLE, OH,	US Mail (1st Class)
24846	SEARS, CREDIT SALES DEPT., PO BOX 740020, ATLANTA, GA, 30374	US Mail (1st Class)
24846	SEARS COMMERCIAL ONE, 75 REMITTANCE DRIVE, SUITE 1674, CHICAGO, IL, 60675-1674	US Mail (1st Class)
24846	SEBRENA SINGLETON, 1222 8TH ST NE, CANTON, OH, 44704	US Mail (1st Class)
24846	SECRETARY OF STATE, TERRI LYNN LAND, PO BOX 30045, LANSING, MI, 48918	US Mail (1st Class)
24846	SECRETARY OF STATE, HON MARK HAMMOND, 1205 PENDLETON STREET SUITE 525, COLUMBIA, SC, 29201	US Mail (1st Class)
24846	SECRETARY OF STATE, HON NANCY WORLEY, PO BOX 5616, MONTGOMERY, AL, 36103-5616	US Mail (1st Class)
24846	SECURITY ACCESS & PARKING SYSTEMS, 7991 PENNSYLVANIA AVE., MAPLE HTS., OH, 44137	US Mail (1st Class)
24846	SECURITY FIRE & EQUIPMENT CO., 3700 NORTHBROOK DRIVE, SUITE 11, NORTHPORT, AL, 35473	US Mail (1st Class)
24846	SECURITY STORAGE & VAN CO., 12671 MCMANUS BLVD, NEWPORT NEWS, VA, 23602-4434	US Mail (1st Class)
24846	SEEGOTT, INC., PO 75591, CLEVELAND, OH, 44101-4755	US Mail (1st Class)
24846	SEIPLE LITHOGRAPH CO., 4390 MT. PLEASANT RD. NW, NORTH CANTON, OH, 44720-8098	US Mail (1st Class)
24846	SELECT AUTOMOTIVE SERVICES, 872 MYERS STREET, OSHAWA, ON, L1H 5N4CANADA	US Mail (1st Class)
24846	SELECT TOOL INC., 4125 DEL DUCA DR, OLD CASTLE, ON, N0R 1L0CANADA	US Mail (1st Class)
24846	SELECT TOOL, INC., 4125 DELDUCA DRIVE, OLDCASTLE, ON, NOR 1L0CANADA	US Mail (1st Class)
24846	SELECTCARE, DEPARTMENT 771008, DETROIT, MI, 48277-0008	US Mail (1st Class)
24846	SELECT-STAFFING, 277 W CAPAC RD, IMLAY CITY, MI, 48444-1071	US Mail (1st Class)
24846	SEMBLEX CORPORATION, 199 W DIVERSEY AVE, ELMHURST, IL, 60126-1103	US Mail (1st Class)
24846	SEMCO ENERGY, 1411 3RD ST STE A, PORT HURON, MI, 48060-5467	US Mail (1st Class)
24846	SENECA COUNTY TREASURER, MARGUERITE O. BERNARD, 109 S. WASHINGTON STREET, TIFFIN, OH, 44883	US Mail (1st Class)
24846	SENSOR PRODUCTS, 188 ROUTE 10, DEPT. 307, EAST HANOVER, NJ, 07936-2108	US Mail (1st Class)
24846	SERIGRAPH, BOX 68-9574, MILWAUKEE, WI, 53268-9574	US Mail (1st Class)
24846	SERV TEK, 4165 HALF ACRE RD., BATAVIA, OH, 45103	US Mail (1st Class)
24846	SERVEX ELECTRONIC DISTR., 2050 MARION-MT. GILEAD RD., MARION, OH, 43302	US Mail (1st Class)
24846	SERVICE ELECTRIC SUPPLY, INC., DRAWER 1535, PO BOX 79001, DETROIT, MI, 48279-1535	US Mail (1st Class)
24846	SERVICE STAR, 381 US HWY 11, ARGO, AL, 35173	US Mail (1st Class)
24846	SERVICEMASTER PREMIER, OF RICHLAND COUNTY, INC., PO BOX 216, SHELBY, OH, 44875	US Mail (1st Class)
24846	SET ENTERPRISES, INC., DEPT. 225901, PO BOX 67000, DETROIT, MI, 48267-2259	US Mail (1st Class)
24846	SETON, PO BOX 819, 20 THOMPSON ROAD, BRANFORD, CT, 06405-2842	US Mail (1st Class)
24846	SETON NAME PLATE CO., PO BOX 819, 20 THOMPSON ROAD, BRANFORD, CT, 06405-2842	US Mail (1st Class)
24846	SHADEL COMPANY INC., 233 LA GRANGE AVE., ROCHESTER, NY, 14613-1562	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	SHAFER INDUSTRIAL SERVICE, 4565 N. LEAVITT RD., WARREN, OH, 44485	US Mail (1st Class)
24846	SHAFER SANITARY SERVICE, INC., 1240 N RIVER RD., NE, WARREN, OH, 44483	US Mail (1st Class)
24846	SHALTZ FLUID POWER, 5163 COMMERCE ROAD, FLINT, MI, 48507	US Mail (1st Class)
24846	SHAMROCK HOSE & FITTINGS, 1771 IVANHOE RD, CLEVELAND, OH, 44112	US Mail (1st Class)
24846	SHANE DALE, PETERSON & IBOLD, 401 SOUTH STREET, CHARDON, OH, 44024	US Mail (1st Class)
24846	SHANKS CONVERTING EQUIPMENT, 39 AVENUE A, NEWARK, NJ, 07114	US Mail (1st Class)
24846	SHANN HUMPHRES, 184 RAILROAD STREET, MILLPORT, AL, 35576	US Mail (1st Class)
24846	SHANNON BOGUTH, 30821 UNDERWOOD DR., ROSEVILLE, MI, 48066	US Mail (1st Class)
24846	SHANNON RYAN, 442 AUGUST DR., FOSTORIA, MI, 48435	US Mail (1st Class)
24846	SHARI SHAFER, 6114 N. GALE RD, DAVISON, MI, 48425	US Mail (1st Class)
24846	SHARRON PESHA, 290 MC CORMICK ST, LAPEER, MI, 48446	US Mail (1st Class)
24846	SHAY KIMPLE CONSULTING GROUP, 49 WELSHS LN, SOMERSET, NJ, 08873-7114	US Mail (1st Class)
24846	SHEFFCO INC., N27 W 23310 ROUNDY DR., PEWAUKEE, WI, 53072	US Mail (1st Class)
24846	SHEILA COMER, 15972 NORTHWIND CR, NORTHPORT, AL, 35475	US Mail (1st Class)
24846	SHEILA CRAFT, 1374 W DOWNEY AVE., FLINT, MI, 48505	US Mail (1st Class)
24846	SHEILA LORENZEN, 2788 S DECKER RD., MARLETTE, MI, 48453-8993	US Mail (1st Class)
24846	SHELBY WELDING & RACK CO., 75 WALL STREET, SHELBY, OH, 44875	US Mail (1st Class)
24846	SHELL, PROCESSING CENTER, DES MOINES, IA, 50367-0400	US Mail (1st Class)
24846	SHELL FLEET, SHELL OIL COMPANY, 910 LOUISIANA ST, HOUSTON, TX, 77002-4916	US Mail (1st Class)
24846	SHELLEY TURNER, 5127 COLDWATER ROAD, LAPEER, MI, 48446	US Mail (1st Class)
24846	SHELLY SMITH, 2278 CLARK RD, LAPEER, MI, 48446	US Mail (1st Class)
24846	SHELTON STATE COMMUNITY, COLLEGE, 9500 OLD GREENSBORO ROAD, TUSCALOOSA, AL, 35405	US Mail (1st Class)
24846	SHERATON AIRPORT HOTEL CLEVE., 5300 RIVERSIDE DRIVE, CLEVELAND, OH, 44135	US Mail (1st Class)
24846	SHEREE FOX, 238 DELAWARE CROSSING, EATON, OH, 45320	US Mail (1st Class)
24846	SHERMAN INDUSTRIES, 37701 STEVENS BLVD., WILLOUGHBY, OH, 44094	US Mail (1st Class)
24846	SHERWIN WILLIAMS, 324 S. SANDUSKY, BUCYRUS, OH, 44820	US Mail (1st Class)
24846	SHERWIN-WILLIAMS, 66 W NEPESSING, LAPEER, MI, 48446-2144	US Mail (1st Class)
24846	SHERWIN-WILLIAMS, 2305 COLUMBUS ROAD NE, CANTON, OH, 44705	US Mail (1st Class)
24846	SHILOH INDUSTRIES, INC., WELLINGTON STAMPING DIV., DEPT. 6262, CHICAGO, IL, 60674-6262	US Mail (1st Class)
24846	SHIN HWA T&T CO., 142-3 GAMSAM-DONG DALSEO-GU, TAEGU, SOUTH KOREA	US Mail (1st Class)
24846	SHINTECH, INC., 3 E GREENWAY PLZ STE 1150, HOUSTON, TX, 77046-0320	US Mail (1st Class)
24846	SHIP-PAQ, INC, 3845 PORT UNION ROAD, FAIRFIELD, OH, 45014	US Mail (1st Class)
24846	SHIRLEY PAVING, PO BOX 469, NORTHPORT, AL, 35476-0469	US Mail (1st Class)
24846	SHRED-IT, 9860 WINDISCH RD., WEST CHESTER, OH, 45069	US Mail (1st Class)
24846	SHRED-X-OF ERIE, 7320 STERRETTANIA RD, FAIRVIEW, PA, 16415-2811	US Mail (1st Class)
24846	SHUMAN PLASTICS, INC., 35 NEOGA ST, DEPEW, NY, 14043-3315	US Mail (1st Class)
24846	SIDNEY TRUCK & STORAGE, 777 WEST RUSSELL RD., PO BOX 748, SIDNEY, OH, 45365	US Mail (1st Class)
24846	SIEGLING AMERICA LLC, 918 CENTRAL AVENUE, WOOD DALE, IL, 60191-1216	US Mail (1st Class)
24846	SIEI AMERICA, INC., 14201 S LAKES DR STE D, CHARLOTTE, NC, 28273-7708	US Mail (1st Class)
24846	SIEMENS ENERGY & AUTOMATION, 3333 OLD MILTON PKWY, ALPHARETTA, GA, 30005-4437	US Mail (1st Class)
24846	SIEMENS-WESTINGHOUSE, 21741 MELROSE, SOUTHFIELD, MI, 48075	US Mail (1st Class)
24846	SIESEL DISTRIBUTING, 6831 STATE ROUTE 602, PO BOX 273, NEW WASHINGTON, OH, 44854	US Mail (1st Class)
24846	SIGN PRO OF CANTON, INC., 2700 FULTON DR. N.W., CANTON, OH, 44718-2220	US Mail (1st Class)
24846	SILCO FIRE PROTECTION CO., 10765 MEDALLION DRIVE, CINCINNATI, OH, 45241	US Mail (1st Class)
24846	SILENT SENTRY ALARM SVC. LLC, 3401 CORUNNA RD, FLINT, MI, 48503-3265	US Mail (1st Class)
24846	SILVIA MARTINEZ, 826 BLACKBURN ST. SW, WYOMING, MI, 49509	US Mail (1st Class)
24846	SIMPSON ELECTRIC, 853 DUNDEE AVE., ELGIN, IL, 60120	US Mail (1st Class)
24846	SINNETT-ELPACO COATINGS CORP., 1378 KINGSLAND AVE, SAINT LOUIS, MO, 63133-1614	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	SIPCO MOLDING TECHNOLOGIES, PO BOX 725, 1099 MORGAN VILLAGE RD., MEADVILLE, PA, 16335-6725	US Mail (1st Class)
24846	SISTEMAS HORMIGA SA DE CV, AV. 20 DE NOVIEMBRE #15, COL. STA. MARIA AZTAHUACAN, MEXICO, DF, 095000MEXICO	US Mail (1st Class)
24846	SIVON MANUFACTURING CO INC., 3131 PERRY PARK RD., PERRY, OH, 44081	US Mail (1st Class)
24846	SIX FLAGS OHIO, 1060 N. AURORA RD., AURORA, OH, 44202-8710	US Mail (1st Class)
24846	SJF MATERIAL HANDLING, LOCK BOX 32, PO BOX 1575, MINNEAPOLIS, MN, 55480-1575	US Mail (1st Class)
24846	SKD, L.P., 1450 W LONG LAKE RD STE 210, TROY, MI, 48098-6330	US Mail (1st Class)
24846	SKILL TOOL & DIE CORP., 16151 PURITAS AVE, CLEVELAND, OH, 44135-2617	US Mail (1st Class)
24846	SKILLPATH SEMINARS, 6900 SQUIBB RD STE 300, SHAWNEE MISSION, KS, 66202-3256	US Mail (1st Class)
24846	SKIMMER, INC, 6969 METROPLEX DRIVE, ROMULUS, MI, 48174	US Mail (1st Class)
24846	SKYLINE CENTER, INC., 2600 N 4TH STREET, PO BOX 3064, CLINTON, IA, 52732-3064	US Mail (1st Class)
24846	SLESNICK STRUCTURAL STEEL, 927 WARNER ROAD, S.E., CANTON, OH, 44707	US Mail (1st Class)
24846	SMALL PARTS, INC., 600 HUMPHREY STREET, PO BOX 23, LOGANSPORT, IN, 46947	US Mail (1st Class)
24846	SMC (SOUTHERN MACHINERY), PO BOX 3407, OXFORD, AL, 36203	US Mail (1st Class)
24846	SMC PNEUMATICS, 3011 N FRANKLIN RD, INDIANAPOLIS, IN, 46226-6308	US Mail (1st Class)
24846	SMITH WELDING SUPPLY, AND EQUIPMENT CO., INC., 1200 FARROW, FERNDALE, MI, 48220	US Mail (1st Class)
24846	SMITH'S MACHINES, 14120 HWY 11 NORTH, COTTONDALE, AL, 35453	US Mail (1st Class)
24846	SMITHERS SCIENTIFIC, 425 W MARKET STREET, AKRON, OH, 44303	US Mail (1st Class)
24846	SMYER CONSTRUCTION, 5239 NEW PARIS PIKE, RICHMOND, IN, 47374	US Mail (1st Class)
24846	SNAP-ON TOOLS, 21755 NETWORK PLACE, CHICAGO, IL, 60673-1217	US Mail (1st Class)
24846	SNAP-ON TOOLS, SNAP-ON TOOLS BIRMINGHAM, PO BOX 6900, NORCROSS, GA, 30091-6900	US Mail (1st Class)
24846	SNELLING PERSONNEL SERVICES, 12801 N CENTRAL EXPY STE 600, DALLAS, TX, 75243-1725	US Mail (1st Class)
24846	SNS-IEC, PO BOX 951164, CLEVELAND, OH, 44193	US Mail (1st Class)
24846	SOCIETY FOR HUMAN RESOURCE, 1800 DUKE ST, ALEXANDRIA, VA, 22314-3494	US Mail (1st Class)
24846	SOCIETY FOR HUMAN RESOURCE, HUMAN RESOURCE CERTIFICATION, PO BOX 79905, BALTIMORE, MD, 21279-0905	US Mail (1st Class)
24846	SOCIETY OF MANUFACTURING, ENGINEERS, ONE SME DRIVE, DEARBORN, MI, 48121	US Mail (1st Class)
24846	SOCIETY OF MFG ENGINEERS, 1 SME DR, DEARBORN, MI, 48128-2408	US Mail (1st Class)
24846	SOCIETY OF PLASTIC ENGINEER, 14 FAIRFIELD DR, BROOKFIELD, CT, 06804-3911	US Mail (1st Class)
24846	SOCIETY OF PLASTICS, ENGINEERING, 1195 CENTRE RD., AUBURN HILLS, MI, 48236	US Mail (1st Class)
24846	SOFCO, INC., 702 POTENTIAL PKWY, SCOTIA, NY, 12302-1011	US Mail (1st Class)
24846	SOFTWARE-WHOLESALE COM, 13312 CARINITO CIERA, SUITE 194, SAN DIEGO, CA, 92129	US Mail (1st Class)
24846	SOJOURNER CAFE`, 267 W. MAIN STREET, LOUDONVILLE, OH, 44842	US Mail (1st Class)
24846	SOLARTEC, INC., 250 PENNSYLVANIA AVE., SALEM, OH, 44460	US Mail (1st Class)
24846	SOLID CONCEPTS, INC., 28309 AVENUE CROCKER, VALENCIA, CA, 91355-1251	US Mail (1st Class)
24846	SOLID CONTROLS, INC., 2155 NIAGARA LANE N, SUITE 110, PLYMOUTH, MN, 55447	US Mail (1st Class)
24846	SOLID GOLD SOUND, 40951 IRWIN, HARRISON TWP., MI, 48045	US Mail (1st Class)
24846	SOLVAY ENGINEERED POLYMERS, 88186 EXPEDITE WAY, CHICAGO, IL, 60695-0001	US Mail (1st Class)
24846	SOMMER ELECTRIC CORPORATION, 818 3RD ST NE, CANTON, OH, 44704-1139	US Mail (1st Class)
24846	SONYA MIKESELL, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	SOTA TECHNOLOGY, INC., 2470 X-CELSIOR DR., PO BOX 563, OXFORD, MI, 48371-0563	US Mail (1st Class)
24846	SOUTH CAROLINA EMPLOYMENT SECURITY, ATTN: CONTRIBUTION SECTION, PO BOX 7103, COLUMBIA, SC, 29202	US Mail (1st Class)
24846	SOUTH EASTERN CONTROLS, PO BOX 660864, VESTAVIA HILLS, AL, 35266	US Mail (1st Class)
24846	SOUTH LYON HOME OF, COMPASSIONATE CARE, 700 REYNOLDS SUITE PARKWAY, SOUTH LYON, MI, 48178	US Mail (1st Class)
24846	SOUTHEASTERN FREIGHT LINES, 420 DAVEGA DR, LEXINGTON, SC, 29073-7485	US Mail (1st Class)
24846	SOUTHEASTERN POLYMER, 2515 IVY ST W, CUMMING, GA, 30041	US Mail (1st Class)
24846	SOUTHERN CONTROLS, 3511 WETUMPKA HWY, MONTGOMERY, AL, 36110-2751	US Mail (1st Class)
24846	SOUTHERN ELECTRIC SUPPLY CO., PO BOX 1690, DAYTON, OH, 45401-1690	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	SOUTHERN HEAT EXCHANGE, 6100 OLD MONTGOMERY HIGHWAY, TUSCALOOSA, AL, 35404	US Mail (1st Class)
24846	SOUTHERN RESEARCH, PO BOX 0375, BIRMINGHAM, AL, 35246-0375	US Mail (1st Class)
24846	SOUTHLAND EQUIPMENT SERVICE, INC., 109 N. SHORECREST RD., COLUMBIA, SC, 29209	US Mail (1st Class)
24846	SOUTHLAND PALLET CO, RT. 3, BOX 347, EUTAW, AL, 35462	US Mail (1st Class)
24846	SOUTHWEST AIR PRODUCTS, 13108 EULESS ST, EULESS, TX, 76040-7266	US Mail (1st Class)
24846	SOUTHWEST ENGINEERS, INC., 39478 HIGHWAY 190 E, SLIDELL, LA, 70461-2357	US Mail (1st Class)
24846	SOUTHWEST HEATER & CONTROLS, 10610 CONTROL PLACE, DALLAS, TX, 75238	US Mail (1st Class)
24846	SOUTHWEST TRANE, 4401 ELLISON NE, ALBURQUERQUE, NM, 87109	US Mail (1st Class)
24846	SOUTHWESTERN INDUSTRIAL, CONTRACTORS AND RIGGERS, INC., 7155 INDUSTRIAL AVE., EL PASO, TX, 79915	US Mail (1st Class)
24846	SOVEREIGN CHEMICAL CO., 1225 W MARKET ST, AKRON, OH, 44313-7107	US Mail (1st Class)
24846	SPALLINGER AUTOCLAVE SYS. INC., 1155 E HAWTHORN RD., LIMA, OH, 45804	US Mail (1st Class)
24846	SPAN MANUFACTURING, 125 GIBSON DR., MARKHAM, ON, L3R 3K7CANADA	US Mail (1st Class)
24846	SPARKLE MARKET, 169 SOUTH HIGH ST., CORTLAND, OH, 44410	US Mail (1st Class)
24846	SPARKS BELTING CO., 3800 STAHL DR SE, GRAND RAPIDS, MI, 49546	US Mail (1st Class)
24846	SPARKS BELTING COMPANY, 4653 SPRING ROAD, CLEVELAND, OH, 44131-1021	US Mail (1st Class)
24846	SPARTAN STORES, INC., PO BOX 79001 DRAWER 1733, DETROIT, MI, 48279-1733	US Mail (1st Class)
24846	SPARTECH COLOR, 577 ERIE STREET, PO BOX 601, STRATFORD, ON, N5A 6V6CANADA	US Mail (1st Class)
24846	SPARTECH POLYCOM, 7174 COLLECTION CENTER DRIVE, CHICAGO, IL, 60693-0071	US Mail (1st Class)
24846	SPE AUTOMOTIVE DIVISION, 343 E MAPLE, TROY, MI, 48083	US Mail (1st Class)
24846	SPE BLOW MOLDING DIVISION, 14 FAIRFIELD DR, BROOKFIELD, CT, 06804-3911	US Mail (1st Class)
24846	SPE DETROIT SECTION, 1800 CROOKS RD., SUITE A, TROY, MI, 48084	US Mail (1st Class)
24846	SPE DETROIT SECTION SPRING, GOLF CLASSIC, L. LEWALLAN CO., 22900 INTERSTATE DR., CLINTON TWP., MI, 48035	US Mail (1st Class)
24846	SPEC CHECK INC., 6200 26 MILE ROAD, SHELBY TOWNSHIP, MI, 48316	US Mail (1st Class)
24846	SPECIAL TAX BOARD, 714 GREENSBORO AVE, TUSCALOOSA, AL, 35401-1859	US Mail (1st Class)
24846	SPECIAL TOOL & ENGINEERING, 33910 JAMES J POMPO, FRASER, MI, 48026	US Mail (1st Class)
24846	SPECIAL WASTE SYSTEM, PO BOX 350, TIPP CITY, OH, 45371-0350	US Mail (1st Class)
24846	SPECIALIZED GLOBAL LOGISTICS, HAGGERTY CENTER, 801 WILLOW RUN AIRPORT HNGR 1, YPSILANTI, MI, 48198-0899	US Mail (1st Class)
24846	SPECIALIZED INSPECTION GROUP, L-902 ST. ROUTE 108, NAPOLEON, OH, 43545	US Mail (1st Class)
24846	SPECIALTY BLADES, INC., 9 TECHNOLOGY DR, STAUNTON, VA, 24401	US Mail (1st Class)
24846	SPECIALTY CHEMICAL SALES, INC., 4561 WEST 160TH STREET, CLEVELAND, OH, 44135-2647	US Mail (1st Class)
24846	SPECIALTY FITTERS INC., 563 TERNES LN, ELYRIA, OH, 44035-6251	US Mail (1st Class)
24846	SPECIALTY MINERALS INC., THE CHRYSLER BUILDING, 405 LEXINGTON AVE, NEW YORK, NY, 10174-0002	US Mail (1st Class)
24846	SPECIALTY NAMEPLATE CORP., 4670 GROVES ROAD, COLUMBUS, OH, 43232	US Mail (1st Class)
24846	SPECIALTY STAMPINGS, LLC ADEL, 423 MILL ST, ADEL, GA, 31620-2308	US Mail (1st Class)
24846	SPECIALTY TECH PUBLISHER, UNIT 10, 1225 KEITH RD., NORTH VANCOUVER, BC, V7J 1J3CANADA	US Mail (1st Class)
24846	SPECTRUM HEALTH, OCCUPATIONAL BILLING, PO BOX 2048, GRAND RAPIDS, MI, 49501	US Mail (1st Class)
24846	SPEDE TECHNOLOGIES, ATTN: JACQUIE ALAR, 24864 DETROIT RD., WESTLAKE, OH, 44145	US Mail (1st Class)
24846	SPEEDWRENCH, 3364 QUINCY STREET, HUDSONVILLE, MI, 49426	US Mail (1st Class)
24846	SPENCER PRODUCTS CO., 1859 SUMMIT COMMERCE PARK, TWINSBURG, OH, 44087	US Mail (1st Class)
24846	SPERRY & RICE MFG CO LLC, 9146 US 52, BROOKVILLE, IN, 47012-9657	US Mail (1st Class)
24846	SPHERION CORPORATION, 4259 COLLECTIONS CENTER DR., CHICAGO, IL, 60693	US Mail (1st Class)
24846	SPHERION OF MID-OHIO, INC., L-2427, COLUMBUS, OH, 43260	US Mail (1st Class)
24846	SPIRIT SERVICES CO., 2001 ARLINGATE LANE, COLUMBUS, OH, 43228	US Mail (1st Class)
24846	SPIROL INTERNATIONAL CORP, DEPT CH 14018, PALATINE, IL, 60055-4018	US Mail (1st Class)
24846	SPITFIRE GRINDING, PO BOX 971243, YPSILANT, MI, 48197	US Mail (1st Class)
24846	SPITFIRE GRINDING, PO BOX 97-1243, YPSILANT, MI, 48197	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	SPITFIRE GRINDING, PO BOX 97-1243, YPSILANTI, MI, 48197	US Mail (1st Class)
24846	SPITFIRE GRINDING INC., 1575 N PROSPECT RD, YPSILANTI, MI, 48198-9403	US Mail (1st Class)
24846	SPORTS BARN, 14960 S STATE RD., MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	SPORTS UNLIMITED, INC., 9961 ANDOVERS DRIVE, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	SPRAYING SYSTEMS CO., NORTH AVE. AT SCHMALE RD, WHEATON, IL, 60189-7900	US Mail (1st Class)
24846	SPRING TEAM, 2851 INDUSTRIAL PARK DR, AUSTINBURG, OH, 44010-9764	US Mail (1st Class)
24846	SPRINGFIELD METAL WORKS, 9676 NORTHWEST COURT, CLARKSTON, MI, 48346	US Mail (1st Class)
24846	SPRINGFIELD MOLD & DIE, INC., 2420 W FOURTH ST., MANSFIELD, OH, 44906	US Mail (1st Class)
24846	SPS TECHNOLOGIES WATERFORD CO., 1956 SOLUTIONS CENTER, LOCKBOX 771956, CHICAGO, IL, 60677-1009	US Mail (1st Class)
24846	SPSS INC., 1213 PAYSPHERE CIRCLE, CHICAGO, IL, 60674	US Mail (1st Class)
24846	SQ INCORPORATED, 1456 BOWERS ROAD, LAPEER, MI, 48446	US Mail (1st Class)
24846	SQS, 14403 EXPORT ROAD, ITC INDUSTRIAL PARK, LAREDO, TX, 78045	US Mail (1st Class)
24846	SQUARE D GROUP - SNIEDER, 1960 RESEARCH DRIVE, SUITE 100, TROY, MI, 48083	US Mail (1st Class)
24846	SRLMC, 6200 FRANK AVENUE NW, CANTON, OH, 44720	US Mail (1st Class)
24846	SSA GLOBAL TECHNOLOGIES, 36549 EAGLE WAY, CHICAGO, IL, 60678-1365	US Mail (1st Class)
24846	ST. CLAIR COMMUNITY COLLEGE, 323 ERIE STREET, PO BOX 5015, PORT HURON, MI, 48061	US Mail (1st Class)
24846	ST. CLAIR COMMUNITY COLLEGE, 444 WEST THIRD STREET, DAYTON, OH, 45402	US Mail (1st Class)
24846	ST. CLAIR SUPPLY COMPANY, PO BOX 387, 231 EAST MAIN STREET, EATON, OH, 45320	US Mail (1st Class)
24846	ST. JOHNS GOLF &, CONFERENCE CENTER, 44045 FIVE MILE RD., PLYMOUTH, MI, 48170	US Mail (1st Class)
24846	ST. JOSEPH CATHOLIC CHURCH, 119 N. LIBERTY, GALION, OH, 44833	US Mail (1st Class)
24846	ST. JOSEPH MERCY HOSPITAL, C/O THOMAS NEFF, 371 N. MAIN ST., MILFORD, MI, 48381	US Mail (1st Class)
24846	ST. VINCENT REHAB SOLUTIONS, 3530 PEACH STREET, LL SUITE 1, ERIE, PA, 16508	US Mail (1st Class)
24846	STACK PAC COR., 5633 52ND STREET SE, GRAND RAPIDS, MI, 49512	US Mail (1st Class)
24846	STACY KING, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	STAFAST PRODUCTS INC., 505 LAKESHORE BLVD, PAINESVILLE, OH, 44077-1121	US Mail (1st Class)
24846	STAFFINGMASTER, LLC., 1188 BELL ROAD, SUITE 206, CLEVELAND, OH, 44022	US Mail (1st Class)
24846	STAFFMARK, INC., 111 CENTER ST, LITTLE ROCK, AR, 72201-4401	US Mail (1st Class)
24846	STAMM CONTRACTING CO INC., PO BOX 450, 4566 ORCHARD ST., MANTUA, OH, 44255	US Mail (1st Class)
24846	STAMPINGS INC., 34152 DOREKA DR., FRASER, MI, 48026	US Mail (1st Class)
24846	STAN SMITH CONSTRUCTION, 16199 WESLEY CHAPEL ROAD, RALPH, AL, 35480	US Mail (1st Class)
24846	STANDARD INSURANCE, UNIT 16, PO BOX 5000, PORTLAND, OR, 97208-5000	US Mail (1st Class)
24846	STANDARD PLUMBING & HTG., 435 WALNUT AVENUE, S.E., PO BOX 20650, CANTON, OH, 44701	US Mail (1st Class)
24846	STANDARD PLUMBING & HTG., 435 WALNUT AVE SE, CANTON, OH, 44702-1348	US Mail (1st Class)
24846	STANDARD ROOFING, OF BIRMINGHAM, PO BOX 1309, MONTGOMERY, AL, 36102-1309	US Mail (1st Class)
24846	STANLEY BOSTITCH, S COUNTY BRIG RR 2, EAST GREENWICH, RI, 02818	US Mail (1st Class)
24846	STANLEY INDUSTRIES INC., 641 N. ROCHESTER ROAD, CLAWSON, MI, 48017-1729	US Mail (1st Class)
24846	STANLEY M. PROCTOR CO., 2016 MIDWAY DR, TWINSBURG, OH, 44087-1934	US Mail (1st Class)
24846	STANLEY R BARTLING, 258 FESCUE DR SE, KENTWOOD, MI, 49548-5953	US Mail (1st Class)
24846	STANWADE METAL PRODUCTS, 6868 STATE ROUTE 305, FOWLER, OH, 44418-9780	US Mail (1st Class)
24846	STAPLES-BUSINESS ADVANTAGE, DETROIT DEPT. DET, PO BOX A3689, CHICAGO, IL, 60690-3689	US Mail (1st Class)
24846	STARK COUNTY AUDITOR, 200 TUSCARAWAS ST W # 1, CANTON, OH, 44702-2202	US Mail (1st Class)
24846	STARK COUNTY CLERK OF COURTS, PHIL G GIAVASIS, 115 CENTRAL PLAZA NORTH, CANTON, OH, 44701-1160	US Mail (1st Class)
24846	STARK COUNTY ESTATE TAX, 110 CENTRAL PLZ S # 220, CANTON, OH, 44702-1410	US Mail (1st Class)
24846	STARK COUNTY HEALTH DEPARTMENT, 3951 CONVENIENCE NW, CANTON, OH, 44718-2660	US Mail (1st Class)
24846	STARK COUNTY TREASURER, 200 TUSCARAWAS ST W # 2, CANTON, OH, 44702-2202	US Mail (1st Class)
24846	STARK INDUSTRIAL, INC., 5103 STONEHAM RD, CANTON, OH, 44720-1540	US Mail (1st Class)
24846	STARLINGER NORTH AMERICA, 900 MUIRFIELD DRIVE, HANOVER PARK, IL, 60133	US Mail (1st Class)
24846	STATE AUDITOR, 111 2ND ST NW FL 4, CANTON, OH, 44702-1509	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	STATE OF ALABAMA DEPARTMENT OF REVENUE, PO BOX 327790, MONTGOMERY, AL, 36132-7790	US Mail (1st Class)
24846	STATE OF MICHIGAN, DEPARTMENT OF THE TREASURY, LANSING, MI, 48922	US Mail (1st Class)
24846	STATE OF MICHIGAN, 7064 CROWNER DR, LANSING, MI, 48980-0001	US Mail (1st Class)
24846	STATE OF MICHIGAN, DEQ WATER BUREAU, CASHIERS OFFICE, WB-SINV/PO BOX 30657, LANSING, MI, 48909-8157	US Mail (1st Class)
24846	STATE OF MICHIGAN, MICHIGAN DEPT OF TREASURY, DEPT 77003, DETROIT, MI, 48277-0003	US Mail (1st Class)
24846	STATE OF MICHIGAN, MICHIGAN DEPT OF STATE, LANSING, MI, 48980-0001	US Mail (1st Class)
24846	STATE OF MICHIGAN DEPT OF LABOR & GROWTH, UNEMPLOYMENT INSURANCE AGENCY, TAX OFFICE - SUITE 11-500, 3024 W GRAND BLVD, DETROIT, MI, 48202	US Mail (1st Class)
24846	STATE SCREW PRODUCTS, 44605 MACOMB INDUST.DRIVE, CLINTON TOWNSHIP, MI, 48036	US Mail (1st Class)
24846	STAUB LASER CUTTING, 3525 STOP EIGHT RD., DAYTON, OH, 45414	US Mail (1st Class)
24846	STAUBLI, 201 PARKWAY WESTY, PO BOX 189, DUNCAN, SC, 29334	US Mail (1st Class)
24846	STAVELEY SERVICES, 192 INTERNATIONALE BLVD, GLENDALE HEIGHTS, IL, 60139-2094	US Mail (1st Class)
24846	STEAM ECONOMIES COMPANY, 9325 SUTTON PL, HAMILTON, OH, 45011-9705	US Mail (1st Class)
24846	STEMACO USA INC., 25 MILFORD DRIVE, SUITE 1, HUDSON, OH, 44236	US Mail (1st Class)
24846	STEPHEN J. TOMASKO, 1176 MEADOW SPUR, AKRON, OH, 44333	US Mail (1st Class)
24846	STEPHEN MALGAY, 38490 SUPERIOR, ROMULUS, MI, 48174	US Mail (1st Class)
24846	STEPHEN T. CAMPBELL, CARTER GIBSON MARTIN ETAL, 18748 JAMESTOWN CIR, NORTHVILLE, MI, $48168\hbox{-}3530$	US Mail (1st Class)
24846	STERICYCLE INC., 28161 N KEITH DR, LAKE FOREST, IL, 60045-4528	US Mail (1st Class)
24846	STERLING BROKERAGE DIVISION, PO BOX 426, 1000 ISLAND AVENUE, MCKEES ROCKS, PA, 15136	US Mail (1st Class)
24846	STERLING CONTRACTORS, 50413 CENTRAL INDUSTRIAL DRIVE, SHELBY TOWNSHIP, MI, 48315	US Mail (1st Class)
24846	STERLING, INC., 2900 S 1060TH ST, NEW BERLIN, WI, 53151	US Mail (1st Class)
24846	STERLING, INC., DEPT. 59902, MILWAUKEE, WI, 53259-0902	US Mail (1st Class)
24846	STEVE HALLOWELL, 314 CRESTHAVEN PL, SIMPSONVILLE, SC, 29681-4854	US Mail (1st Class)
24846	STEVE MARTIN, 1401 INDUSTRIAL PARK, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	STEVE NEWBAUER, 985 FALLS CREEK DR., VANDALIA, OH, 45377-9686	US Mail (1st Class)
24846	STEVE PALMER, 15 MYRTLE DR., BISHOPVILLE, SC, 29010	US Mail (1st Class)
24846	STEVE RINGE, 31572 INDUSTRIAL RD., SUITE #300, LIVONIA, MI, 48150	US Mail (1st Class)
24846	STEVEN BOYD, 7707 MT. PLEASANT RD., NORTH CANTON, OH, 44720	US Mail (1st Class)
24846	STEVEN RUSSELL, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	STEWART EFI LLC, 45 OLD WATERBURY RD, THOMASTON, CT, 06787	US Mail (1st Class)
24846	STIK II PRODUCTS, 41 ONEILL ST, EASTHAMPTON, MA, 01027-1103	US Mail (1st Class)
24846	STIRNEMANN, INC., 772 E MAPLE, BIRMINGHAM, MI, 48009	US Mail (1st Class)
24846	STOERMER-ANDERSON, INC., 3818 RED BANK ROAD, CINCINNATI, OH, 45227	US Mail (1st Class)
24846	STONE CONTAINER CORP., 21514 NETWORK PLACE, CHICAGO, IL, 60673-1215	US Mail (1st Class)
24846	STONER INC, 1070 ROBERT FULTON HWY, QUARYVILLE, PA, 17566	US Mail (1st Class)
24846	STONER INC., 1070 ROBERT FULTON HWY, PO BOX 65, QUARRYVILLE, PA, 17566	US Mail (1st Class)
24846	STONER INC., 1070 ROBERT FULTON HWY, QUARYVILLE, PA, 17566	US Mail (1st Class)
24846	STONEY CREST REGRIND SERVICE, 6243 DIXIE HIGHWAY, BRIDGEPORT, MI, 48722	US Mail (1st Class)
24846	STOUT RISIUS ROSS, INC., 32255 NORTHWESTERN HWY, SUITE 201, FARMINGTON HILLS, MI, 48334-1574	US Mail (1st Class)
24846	STOW & GO SELF STORAGE, 41999 ANN ARBOR RD., PLYMOUTH, MI, 48170	US Mail (1st Class)
24846	STRAIGHT AWAY FABRICATION, 471 B US HWY 250 EAST, ASHLAND, OH, 44805	US Mail (1st Class)
24846	STRATO LIFTS, PARTS FOR LIFTS, INC., 320 ELM AVE., NORTH WALES, PA, 19454	US Mail (1st Class)
24846	STRAUB DESIGN COMPANY, 2238 FLORIDA AVE SOUTH, MINNEAPOLIS, MN, 55426	US Mail (1st Class)
24846	STRAUCH & OBERMILLER, 209 NORTH SELTZER ST, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	STREAMLINE PACKAGING, INC, 5241 SOUTHWAY STREET SW, CANTON, OH, 44706	US Mail (1st Class)
24846	STRETCH ASSOCIATES, 2510 B W WHITNER ST, ANDERSON, SC, 29625	US Mail (1st Class)
24846	STROUP & SON, LTD., 2164 STATE RT 43, MOGADORE, OH, 44260	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	STRUKTOL COMPANY, 201 E STEELS CORNERS RD, STOW, OH, 44224-4921	US Mail (1st Class)
24846	STRYKER SALES CORP., 2725 FAIRFIELD RD, PORTAGE, MI, 49002-1747	US Mail (1st Class)
24846	STS MANUFACTURING, 1325 EL JARDIN HEIGHTS RD, BROWNSVILLE, TX, 78526-9707	US Mail (1st Class)
24846	STYMO CONSTRUCTION, 2875 SPRINGBORO WEST, DAYTON, OH, 45439-1673	US Mail (1st Class)
24846	SUBEK BUCLDAY, 558 E 70TH STREET, ERIE, PA, 16503	US Mail (1st Class)
24846	SUBURBAN LANES, 975 HOPLEY AVE., BUCYRUS, OH, 44820	US Mail (1st Class)
24846	SUBURBAN TOOL & DIE CO., 4940 PACIFIC AVE., ERIE, PA, 16506-4978	US Mail (1st Class)
24846	SUCCESSORIES, 2520 DIEHL RD, AURORA, IL, 60502-9497	US Mail (1st Class)
24846	SUE EAGLETON, 3131 COLUMBUS ROAD NE, CANTON, OH, 44705	US Mail (1st Class)
24846	SUMMERVILLES, INC., 2375 ROMIG RD, AKRON, OH, 44320-3824	US Mail (1st Class)
24846	SUMMIT COUNTY AUDITOR, 1030 E TALLMADGE AVE, AKRON, OH, 44310-3516	US Mail (1st Class)
24846	SUMMIT COUNTY TREASURER, JOHN A DONOFRIO, FISCAL OFFICER-COUNTY OF SUMMIT, 175 S MAIN STREET - ROOM 211, AKRON, OH, 44308	US Mail (1st Class)
24846	SUMMIT CTY CLERK OF COURTS, CLERK OF COURTS, AKRON, OH, 44308	US Mail (1st Class)
24846	SUMMIT ELECTRIC, INC., 2544 ACACIA, TROY, MI, 48083	US Mail (1st Class)
24846	SUMMIT MFG. & ASSY.,INC., 4655 PATTERSON AVE. SE, SUITE C, GRAND RAPIDS, MI, 49512	US Mail (1st Class)
24846	SUMMIT ON THE PARK, CONTINENTAL SERVICES, 46000 SUMMIT PARKWAY, CANTON, MI, 48188	US Mail (1st Class)
24846	SUMMIT PRECISION TOOL, 34112 DOREKA DRIVE, FRASER, MI, 48026	US Mail (1st Class)
24846	SUMTER PACKAGING CORP., 2341 CORPORATE HGWY., SUMTER, SC, 29154	US Mail (1st Class)
24846	SUMTER PRECISION MACHINE & FAB, 1320 SOUTH GUIGNARD DRIVE, SUMTER, SC, 29150	US Mail (1st Class)
24846	SUN PLASTECH, DEPT. CH-17059, PALATINE, IL, 60055	US Mail (1st Class)
24846	SUNOCO, INC., 1735 MARKET ST STE LL, PHILADELPHIA, PA, 19103-7528	US Mail (1st Class)
24846	SUNSOURCE (FAUVER), 1450 EAST AVIS DRIVE, MADISON HEIGHTS, MI, 48071	US Mail (1st Class)
24846	SUPER BOWL, 45100 FORD ROAD, CANTON, MI, 48187	US Mail (1st Class)
24846	SUPER TRANSPORTATION, INC., PO BOX 489, ASHTABULA, OH, 44005	US Mail (1st Class)
24846	SUPERIOR BUILDING SERVICES, 2917 PHILADELPHIA DR, DAYTON, OH, 45405-1913	US Mail (1st Class)
24846	SUPERIOR DIE SET CORP., 900 W DREXEL AVE, OAK CREEK, WI, 53154-1924	US Mail (1st Class)
24846	SUPERIOR EQUIPMENT SOLUTIONS, 219 S PIONEER BLVD., SUITE C, SPRINGBORO, OH, 45066	US Mail (1st Class)
24846	SUPERIOR GAGE, 1200 DEGURSE, MARINE CITY, MI, 48039	US Mail (1st Class)
24846	SUPERIOR LAMP INC., 8003 LINCOLN DR W STE F, MARLTON, NJ, 08053-3210	US Mail (1st Class)
24846	SUPERIOR MEASURING, LLC, 1605 EAST AVIS DRIVE, MADISON HEIGHTS, MI, 48071	US Mail (1st Class)
24846	SUPERIOR PRODUCTS, 510 COUNTY ROAD D W, SAINT PAUL, MN, 55112-3520	US Mail (1st Class)
24846	SUPERIOR TANK COMPANY, INC., 1955 US HIGHWAY 62 W, PRINCETON, KY, 42445-6104	US Mail (1st Class)
24846	SUPPLIER DEVELOPMENT INC., 802 N 5TH AVE, PHOENIX, AZ, 85003-1316	US Mail (1st Class)
24846	SUPPLIER INSPECTION, 2941 S GETTYSBURG, DAYTON, OH, 45418	US Mail (1st Class)
24846	SUPPLIERLINK SERVICES, INC., PO BOX 19519, RENO, NV, 89511-9519	US Mail (1st Class)
24846	SUP-R-DIE, INC., 10003 MEMPHIS AVE., CLEVELAND, OH, 44144	US Mail (1st Class)
24846	SURFACE WATER QUALITY DIV., STORM WATER, MICHIGAN DEPT OF ENVIRONMENTAL, PO BOX 30657, LANSING, MI, 48909-8157	US Mail (1st Class)
24846	SURFCONTROL, INC., DEPT. CH 10987, PALATINE, IL, 60055-0987	US Mail (1st Class)
24846	SUSAN COOK, 3560 W. MARKET STREET, AKRON, OH, 44333	US Mail (1st Class)
24846	SUSAN RAINEY, 1401 INDUSTRIAL DR, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	SUSAN ZEUCH, 443 LOHR RD, MANSFIELD, OH, 44903	US Mail (1st Class)
24846	SUSIE HALLETT, 3560 W. MARKET, SUITE 340, AKRON, OH, 44333	US Mail (1st Class)
24846	SW CONTROLS, 45345 FIVE MILE ROAD, PLYMOUTH, MI, 48170	US Mail (1st Class)
24846	SWAN CLEANERS, LAUNDRY & LINEN, 165 PARK AVENUE WEST, MANSFIELD, OH, 44902	US Mail (1st Class)
24846	SWAN'S LOCKSMITH SERVICE, 354 MAIN STREET, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	SWECO, DIV. OF EMERSON, 1115 HOME PL, MATTHEWS, NC, 28105-6892	US Mail (1st Class)
24846	SWIGART ELECTRIC MOTORS,INC., 301 WEST WENGER RD., ENGLEWOOD, OH, 45322	US Mail (1st Class)
24846	SYMBOL TECHNOLOGIES, INC., DEPT. AT 40229, ATLANTA, GA, 31192-0229	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	SYMMCO INC., 40 S PARK ST, SYKESVILLE, PA, 15865	US Mail (1st Class)
24846	SYNCHRONOUS INDUSTRIAL SERVICES, 10400 TECHNOLOGY DR, COTTONDALE, AL, 35453	US Mail (1st Class)
24846	SYNVENTIVE MOLDING SOLUTIONS, DEPT 294401, 39200 W SIX MILE RD, LIVONIA, MI, 48152	US Mail (1st Class)
24846	SYSTEM INTEGRATION SUPPORT, 449 LAURIAN WAY, KENNESAW, GA, 30144	US Mail (1st Class)
24846	SYSTEMS PACK, INC., 649 EAST HIGHLAND ROAD, MACEDONIA, OH, 44056	US Mail (1st Class)
24846	SYSTRAN FINANCIAL, SERVICES CORPORATION, PO BOX 640296, PITTSBURGH, PA, 15264-0296	US Mail (1st Class)
24846	T & G PROFESSIONAL COMPUTING, 1000 QUEEN CITY AVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	T C M PRODUCTS, 4500 LEE RD, SUITE 218, CLEVELAND, OH, 44128	US Mail (1st Class)
24846	T L SERVICE CENTER, 4626 STATE ROUTE 82, MANTUA, OH, 44255	US Mail (1st Class)
24846	T MOBILE, 12920 SE 38TH ST, BELLEVUE, WA, 98006-1350	US Mail (1st Class)
24846	T R GOLDSMITH & SON, 16 PEUQUET PARKWAY, TONAWANDA, NY, 14150-2413	US Mail (1st Class)
24846	T&H MACHINE TOOLS, 4345 MORRIS PARK DRIVE, CHARLOTTE, NC, 28227	US Mail (1st Class)
24846	T&J LUMBER SVC LLC., HIGHWAY 11, EUTAW, AL, 35462	US Mail (1st Class)
24846	T.H.E.M. OF OHIO INC., 16645 GRANITE RD, CLEVELAND, OH, 44137-4301	US Mail (1st Class)
24846	T.P. COMPOSITES INC., 8 CROZERVILLE ROAD, ASTON, PA, 19014	US Mail (1st Class)
24846	T.S. SERVICES, PO BOX 1206 RT. 608, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	T.S. TRANSPORT LTD, PO BOX 1206, 15668 OLD STATE RD., MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	TA SYSTEMS, INC., 1873 ROCHESTER INDUST. CT, ROCHESTER HILLS, MI, 48309	US Mail (1st Class)
24846	TACONIC PLASTICS, 136 COON BROOK RD, PETERSBURGH, NY, 12138-4303	US Mail (1st Class)
24846	TAMARA L. MEYER, 4632 N SULPHUR-SPRINGS, BROOKVILLE, OH, 45309	US Mail (1st Class)
24846	TAMEKA THOMPSON, 1802 CONCORD, FLINT, MI, 48503	US Mail (1st Class)
24846	TAMMY A NALLEY, PO BOX 152, CHATFIELD, OH, 44825-0152	US Mail (1st Class)
24846	TAMMY L. MELLING, 18 W. SOUTH STREET, WEST ALEXANDER, OH, 45381	US Mail (1st Class)
24846	TANNER AND GUIN, L.L.C., 2711 UNIVERSITY BLVD, TUSCALOOSA, AL, 35403	US Mail (1st Class)
24846	TAPESWITCH CORPORATION, 100 SCHMITT BLVD, FARMINGDALE, NY, 11735-1424	US Mail (1st Class)
24846	TARA SKELTON, 13131 NORTHSIDE ROAD, BERRY, AL, 35546	US Mail (1st Class)
24846	TARGET LOGISTIC SERVICES, DEPT 0942, LOS ANGELES, CA, 90084-0942	US Mail (1st Class)
24846	TARIMAS DEL NORTE, CALLE 36 # 7430, CHIHUAHUA, MEXICO	US Mail (1st Class)
24846	TAXATION DEPT-INCOME TAX DIV, 161 S HIGH ST # 501, AKRON, OH, 44308-1666	US Mail (1st Class)
24846	TAYLOR AUTOMOTIVE, 3294 ST RTE 35 EAST, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	TAYLOR ELECTRICAL CONTRACTORS INC., 5479 SKYLAND BLVD E, COTTONDALE, AL, 35453-1963	US Mail (1st Class)
24846	TAYLOR IND SERVICES (HPM DIV), 820 MARION RD, MT GILEAD, OH, 43338	US Mail (1st Class)
24846	TAYLOR IND. SERVICES (HPM DIV), 820 MARION ROAD, MT. GILEAD, OH, 43338	US Mail (1st Class)
24846	TAYLOR'S SUPPORT SERVICES, 820 MARIO RD., MT. GILEAD, OH, 43338	US Mail (1st Class)
24846	TAYLORS INDUSTRIAL SERVICES, HPM DIVISION, 820 MARION ROAD, MOUNT GILEAD, OH, 43338	US Mail (1st Class)
24846	TB&P EXPRESS, PO BOX 2022, CINCINNATI, OH, 45263-2022	US Mail (1st Class)
24846	TBDN TENNESSEE COMPANY, 1410 US HIGHWAY 70 BYPASS, JACKSON, TN, 38301	US Mail (1st Class)
24846	TDL TOOL, 1296 SOUTH PATTON STREET, XENIA, OH, 45385	US Mail (1st Class)
24846	TDS AUTOMOTIVE U.S. INC., 850 CHAMPLAIN AVE., OSHAWA, ON, L1J 8C3CANADA	US Mail (1st Class)
24846	TEAM RESOURCES INC., 1091 CENTRE RD/SUITE 130, AUBURN HILLS, MI, 48326	US Mail (1st Class)
24846	TEC ENGINEERING CORP, 31 TOWN FORREST RD, OXFORD, MA, 01540	US Mail (1st Class)
24846	TECH DEPOT, BANK OF AMERICA, PO BOX 33074, HARTFORD, CT, 06150-3074	US Mail (1st Class)
24846	TECH ENGINEERING, INC., 1050 W. HURON STREET, WATERFORD, MI, 48328	US Mail (1st Class)
24846	TEC-HACKETT, INC., 3418 CAVALIER DR, FORT WAYNE, IN, 46808-4428	US Mail (1st Class)
24846	TECHNICAL EQUIPMENT SALES CO., NW 7968-06, PO BOX 1450, MINNEAPOLIS, MN, 55485-7968	US Mail (1st Class)
24846	TECHNICAL LOADARM, INC, 2620 GRISWOLD STREET, PORT HURON, MI, 48060	US Mail (1st Class)
24846	TECHNICO, INC., 766 NORTH RIVER ROAD NW, WARREN, OH, 44483	US Mail (1st Class)
24846	TECHNICOMP, 1111 CHESTER AVENUE, CLEVELAND, OH, 44114	US Mail (1st Class)
24846	TECHNIFOR INC., 9800-J SOUTHERN PINE BLVD., CHARLOTTE, NC, 28273	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	TECHNIMARK, INC., MEXICAN OPERATIONS, POST OFFICE BOX 2068, ASHEBORO, NC, 27204-2068	US Mail (1st Class)
24846	TECHNOLOGY REVIEW, 1 MAIN ST 7TH FL, CAMBRIDGE, MA, 02142-1531	US Mail (1st Class)
24846	TEC-MAR DISTRIBUTION SERVICES INC., 16150 GROVE ROAD, LANSING, MI, 48906	US Mail (1st Class)
24846	TED TSIKHLAKIS, 1401 INDUSTRIAL DRIVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	TELAR CORPORATION, 11470 HAWKE RD UNIT 17, COLUMBIA STATION, OH, 44028-9818	US Mail (1st Class)
24846	TELCO FLUID POWER, INC., 7169 E PLEASANT VALLEY RD, CLEVELAND, OH, 44131-5541	US Mail (1st Class)
24846	TELEVAN SALES, 5451 SYLVIA AVE., DEARBORN HGTS, MI, 48125	US Mail (1st Class)
24846	TEL-X CORPORATION, 32400 FORD ROAD, GARDEN CITY, MI, 48135	US Mail (1st Class)
24846	TEMPCO LECTRIC HEATER CORP, DEPT. 77-9194, CHICAGO, IL, 60678-9194	US Mail (1st Class)
24846	TEMPTEK INC., 525 E STOP 18 RD., GREENWOOD, IN, 46143	US Mail (1st Class)
24846	TEMTRON INC., 7530 MARSHALL AVE., ST. LOUIS, MO, 63119	US Mail (1st Class)
24846	TENIBAC-GRAPHION INC., 35155 AUTOMATION DRIVE, CLINTON TWP., MI, 48035	US Mail (1st Class)
24846	TENNESSEAN / NASHVILLE BANNER, 226 W STEVENS ST, COOKEVILLE, TN, 38501-3296	US Mail (1st Class)
24846	TENNESSEE STAMPING LLC, 119 KIRBY DR, PORTLAND, TN, 37148-2004	US Mail (1st Class)
24846	TERESA HAMBLIN, 1210 CLARK RD, LAPEER, MI, 48446	US Mail (1st Class)
24846	TERPCO INDUSTRIAL PDTS. CO., PO BOX 7394, AKRON, OH, 44306-0394	US Mail (1st Class)
24846	TERRY'S SPECIALTIES, 5639 MAJOR LANE, PLATTEVILLE, WI, 53818	US Mail (1st Class)
24846	TEXAS MACHINE & TOOL CO., 4428 MONTANA AVE., EL PASO, TX, 79903	US Mail (1st Class)
24846	TEXTAPE INC., 915 PENDALE ROAD, EL PASO, TX, 79907	US Mail (1st Class)
24846	TEXTILE WORKERS-PENSION, FUND, 6 BLACKSTONE VALLEY PLACE, LINCOLN, RI, 02865	US Mail (1st Class)
24846	TEXTRON FASTENING SYSTEMS, P.O. BOX 77082, DETROIT, MI, 48277	US Mail (1st Class)
24846	TF WAREHOUSING INC., 1036 GREEN VALLEY ROAD, LONDON, ON, N6N 1E3CANADA	US Mail (1st Class)
24846	THE ANVIL PRODUCTS CO., 4535 EAST 71ST STREET, CLEVELAND, OH, 44105	US Mail (1st Class)
24846	THE BALANCING COMPANY, 898 CENTER DR, VANDALIA, OH, 45377-3130	US Mail (1st Class)
24846	THE BANK OF NEW YORK, FINANCIAL CONTROL, BILLING DEPARTMENT, PO BOX 19445, NEWARK, NJ, 07195-0445	US Mail (1st Class)
24846	THE BIRMINGHAM NEWS, 2200 4TH AVE N, BIRMINGHAM, AL, 35203-3802	US Mail (1st Class)
24846	THE BRENNAN INDUSTRIAL, TRUCK COMPANY, PO BOX 939, HOLLAND, OH, 43528	US Mail (1st Class)
24846	THE CARMON GROUP, SUMMIT OFFICE PARK, THREE SUMMIT PARK DR., SUITE 300, INDEPENDENCE, OH, 44131	US Mail (1st Class)
24846	THE CARPET COMPANY, 8089 MAYFIELD RD, CHESTERLAND, OH, 44026	US Mail (1st Class)
24846	THE CHALLENGERS, 2425 ROSCOE COURT, DUBLIN, OH, 43016	US Mail (1st Class)
24846	THE CHARLES RITTER CO., 35 WEST SIXTH STREET, PO BOX 215, MANSFIELD, OH, 44901	US Mail (1st Class)
24846	THE CITY OF TUSCALOOSA, TUSCALOOSA CTY PORT AUTHORITY, 2008 12TH STREET, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	THE CONNECTION CO., PO BOX 28429, COLUMBUS, OH, 43228-0429	US Mail (1st Class)
24846	THE COUNTY PRESS, 1521 IMLAY CITY RD, LAPEER, MI, 48446-3175	US Mail (1st Class)
24846	THE DAILY CHIEF-UNION, 111 W WYANDOT AVE., PO BOX 180, UPPER SANDUSKY, OH, 43351	US Mail (1st Class)
24846	THE DRAWING BOARD, 1418 STIRLING ST, COATESVILLE, PA, 19320-2535	US Mail (1st Class)
24846	THE ECONOMIC CLUB OF DETROIT, 333 WEST FORT STREET, SUITE 100, DETROIT, MI, 48226-3134	US Mail (1st Class)
24846	THE ECONOMICS PRESS INC., PO BOX 1249, WEST CALDWELL, NJ, 07007-1249	US Mail (1st Class)
24846	THE EQUIPMENT LEASING CO., DIV. OF SANDY SPRING, NATIONAL BANK OF MARYLAND, 53 LOVETON CIRCLE, STE 100, SPARKS, MD, 21152	US Mail (1st Class)
24846	THE ESTATE OF JEAN BRIDGHAM, 9636 DONLIN ROAD, ERIE, PA, 16510	US Mail (1st Class)
24846	THE ESTATE OF LOUIS S. EIMER, C/O ATTORNEY TERRY L. GERNERT, PO BOX 191, BUCYRUS, OH, 44820	US Mail (1st Class)
24846	THE ESTATE OF RONALD LEE GUY, C/O BRYDON LAW OFFICE, ATTORNEY JOHN C. BRYDON, 78 E. MAIN STREET, NORTH EAST, PA, 16428	US Mail (1st Class)
24846	THE GALLERY COLLECTION, 65 CHALLENGER ROAD, RIDGEFIELD PARK, NJ, 07660	US Mail (1st Class)
24846	THE GYM, INC., 15980 VALPLAST ST, MIDDLEFIELD, OH, 44062-9399	US Mail (1st Class)
24846	THE HOLLAND GROUP, DEPT. #94761, LOUISVILLE, KY, 40294-4761	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	THE HUNTSMAN HUNT CLUB, 3166 HAVENS ROAD, DRYDEN, MI, 48428	US Mail (1st Class)
24846	THE ILLUMINATING COMPANY, ATTN: TREASURY, FIRSTENERGY, 76 SOUTH MAIN STREET, AKRON, OH, 44308-1890	US Mail (1st Class)
24846	THE INFORMED OUTLOOK, 15913 EDGEWOOD DRIVE, MONTCLAIR, VA, 22026	US Mail (1st Class)
24846	THE JOSEPH JEFFRIES CO., 5211 LOUISVILLE ROAD NE, LOUISVILLE, OH, 44641	US Mail (1st Class)
24846	THE KIPLINGER LETTER, 1729 H ST NW, WASHINGTON, DC, 20006-3901	US Mail (1st Class)
24846	THE LANCE ARMSTRONG FOUNDATION, 1221 S MO PAC EXPY STE 320, AUSTIN, TX, 78746-7666	US Mail (1st Class)
24846	THE MACOMB GROUP, DEPT 166401, PO BOX 67000, DETROIT, MI, 48267	US Mail (1st Class)
24846	THE MAHER GROUP, L.L.C., 575 BYRNE INDUSTRIAL DRIVE, ROCKFORD, MI, 49341	US Mail (1st Class)
24846	THE MINDS PROGRAM, 30233 SOUTHFIELD ROAD, SUITE 113, SOUTHFIELD, MI, 48076	US Mail (1st Class)
24846	THE NEWS-HERALD, 7085 MENTOR AVE., WILLOUGHBY, OH, 44094	US Mail (1st Class)
24846	THE NORTHVILLE LOCKSMITH, 22857 FREDERICK ST, FARMINGTON, MI, 48336-3928	US Mail (1st Class)
24846	THE OAKLAND PRESS, 48 W HURON ST, PONTIAC, MI, 48342-2101	US Mail (1st Class)
24846	THE OAKWOOD GROUP, COMERICA, DRAWER 99901, PO BOX 67000, DETROIT, MI, 48267-0999	US Mail (1st Class)
24846	THE PAQUIN COMPANY, 26981 LAKELAND BLVD., CLEVELAND, OH, 44132	US Mail (1st Class)
24846	THE PLAIN DEALER, 1801 SUPERIOR AVE E, CLEVELAND, OH, 44114-2107	US Mail (1st Class)
24846	THE PLYMOUTH MANOR, 345 N MAIN STREET, PLYMOUTH, MI, 48170	US Mail (1st Class)
24846	THE QUAIL HOLLOW RESORT, 11080 CONCORD-HAMBDEN, PAINESVILLE, OH, 44077	US Mail (1st Class)
24846	THE REGISTER HERALD, 1332 NORTH BARRON STREET, EATON, OH, 45320	US Mail (1st Class)
24846	THE REHAB CTR. (PROGRESS IND.), 270 STERKEL BLVD., MANSFIELD, OH, 44907	US Mail (1st Class)
24846	THE RESIN ENTERPRISE, 307 AUBURN DR., DOWNINGTOWN, PA, 19335	US Mail (1st Class)
24846	THE ROBOT COMPANY, 300 RYAN ROAD, SEVILLE, OH, 44273	US Mail (1st Class)
24846	THE SCHOLL-CHOFFIN CO., 4150 SIMON ROAD, YOUNGSTOWN, OH, 44512	US Mail (1st Class)
24846	THE SELINICK COMPANY, 15879 MADISON RD., MIDDLEFIED, OH, 44062	US Mail (1st Class)
24846	THE SHERWIN WILLIAMS COMPANY, 15421 WEST HIGH STREET, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	THE SKY WAY EAST, 2461 EMMA LANE, MANSFIELD, OH, 44903	US Mail (1st Class)
24846	THE STATIONERY SHOP, 30 NORTH SUMMIT ST., AKRON, OH, 44308	US Mail (1st Class)
24846	THE SWEEP MASTER INC., 2248 STAR COURT, ROCHESTER HILLS, MI, 48309	US Mail (1st Class)
24846	THE TOWN LOCKSMITH, INC., 1270 SOUTH MAIN, PLYMOUTH, MI, 48170	US Mail (1st Class)
24846	THE TRAINING NETWORK, 106 CAPITOLA DRIVE, DURHAM, NC, 27713	US Mail (1st Class)
24846	THE TUSCALOOSA NEWS, 315 28TH AVE, TUSCALOOSA, AL, 35401-1022	US Mail (1st Class)
24846	THE UNIVERSITY OF ALABAMA, 4200 MCFARLAND BLVD E, TUSCALOOSA, AL, 35405-3853	US Mail (1st Class)
24846	THE URSEM COMPANY, 1548 WEST 117TH STREET, CLEVELAND, OH, 44107	US Mail (1st Class)
24846	THE VINDICATOR PRINTING CO., 107 VINDICATOR SQ, YOUNGSTOWN, OH, 44503-1136	US Mail (1st Class)
24846	THE WALL STREET JOURNAL, 200 LIBERTY ST, NEW YORK, NY, 10281-1003	US Mail (1st Class)
24846	THEODORE A. STRIMBACK, 4878 BUEHLER ROAD, HASTINGS, MI, 49058	US Mail (1st Class)
24846	THERESA M. GILBERT, 290 MCCORMICK, LAPEER, MI, 48446	US Mail (1st Class)
24846	THERMAL CORP., 1264 SLAUGHTER RD, MADISON, AL, 35758	US Mail (1st Class)
24846	THERMAL LABEL WAREHOUSE, 111 CENTER PARK DR STE 100, KNOXVILLE, TN, 37922-2121	US Mail (1st Class)
24846	THERMASYS HEAT TRANSFER, 2776 GUNTER PARK DR E STE R, S, MONTGOMERY, AL, 36109-1402	US Mail (1st Class)
24846	THERMOPLASTIC IND.REPLACE, 668 LEESBURG STATION ROAD, VOLANT, PA, 16156	US Mail (1st Class)
24846	THERMOPLASTICS ACQUISITION, LLC, 3560 WEST MARKET ST., SUITE 340, AKRON, OH, 44333	US Mail (1st Class)
24846	THERMWOOD CORPORATION, PO BOX 436, ATTN: A/R, 904 BUFFALOVILLE ROAD WEST, DALE, IN, 47523	US Mail (1st Class)
24846	THIEL TOOL & ENGINEERING, 4622 BULWER AVE, SAINT LOUIS, MO, 63147-3204	US Mail (1st Class)
24846	THOMAS D. JABLONSKI, 5020 HOLLYWOOD CT., BRIGHTON, MI, 48116	US Mail (1st Class)
24846	THOMAS G. FROMKNECHT, 1901 WAGER RD., ERIE, PA, 16509	US Mail (1st Class)
24846	THOMAS R. SKULINA, 24803 DETROIT ROAD, WEST LAKE, OH, 44145	US Mail (1st Class)
24846	THOMAS REGIONAL DIRECTORY, FIVE PENN PLAZA, NEW YORK, NY, 10001	US Mail (1st Class)
24846	THOMAS ROOFING, 6351 WEST SIXTH ST, P O BOX 129, CASS CITY, MI, 48726	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	THOMAS STREASICK, 477 CASTLE RD, FOSTORIA, MI, 48435	US Mail (1st Class)
24846	THOMPSON TRACTOR, 2401 PENSON HWY, BIRMINGHAM, AL, 35202	US Mail (1st Class)
24846	THOMPSON-GUNNER ENTERPRISES, 5514 GRUBB ROAD, ERIE, PA, 16506	US Mail (1st Class)
24846	THOMSON PENN-OHIO, CUSTOMER PAYMENT CENTER, PO BOX 711822, COLUMBUS, OH, 43271-1822	US Mail (1st Class)
24846	THRASHER, DINSMORE & DOLAN, 100 7TH AVENUE, SUITE 150, CHARDON, OH, 44024	US Mail (1st Class)
24846	THREE-D SALES INC., 604 SMITH ST, ALGONAC, MI, 48001-1441	US Mail (1st Class)
24846	THRUWAY FASTENERS, INC, 2910 NIAGARA FALLS BLVD, NORTH TONAWANDA, NY, 14120-1140	US Mail (1st Class)
24846	THYSSEN SPECIALTY STEELS, 44500 REYNOLDS DR, CLINTON TWP, MI, 48036-1246	US Mail (1st Class)
24846	TICONA (HOECHST GROUP-MEMBER), POB 96205, CHICAGO, IL, 60693	US Mail (1st Class)
24846	TIER ONE RECRUITING CONSULTANTS, 1776 MENTOR AVE., STE 75-408, CINCINNATI, OH, 45212	US Mail (1st Class)
24846	TIGER DIRECT.COM, 7795 WEST FLAGLER ST., SUITE 35, MIAMI, FL, 33144	US Mail (1st Class)
24846	TIGER RIDGE MANUFACTURING, INC., 12656 STATE HIGHWAY 11, WHITEWRIGHT, TX, 75491-5285	US Mail (1st Class)
24846	TIM BRUMBAUGH, 3844 CINWOOD NW, MASSILLON, OH, 44646	US Mail (1st Class)
24846	TIM DICKINSON, DICKINSON, TIMOTHY A, 6870 WASHBURN RD, VASSAR, MI, 48768-9532	US Mail (1st Class)
24846	TIM LANGLEY, 11378 OAK ROAD, OTISVILLE, MI, 48463	US Mail (1st Class)
24846	TIM SNOW, 3131 COLUMBUS ROAD, CANTON, OH, 44705	US Mail (1st Class)
24846	TIM SUMMERS, 315 BUTLER STREET, VASSAR, MI, 48768	US Mail (1st Class)
24846	TIM`S AUTO SERVICE, 24 EAST DAYTON STREET, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	TIMOTHY CAHILL, 2601 S RIDGE RD, ASHTABULA, OH, 44004	US Mail (1st Class)
24846	TINIUS OLSEN TESTING, MACHINE CO., INC., PO BOX 7780-1204, PHILADELPHIA, PA, 19182	US Mail (1st Class)
24846	TINNERMAN PALNUT, PRECISION STAMPING DIV., PO BOX 74891, CHICAGO, IL, 60694-4891	US Mail (1st Class)
24846	TINNERMAN PALNUT ENGINEERED PROD., NATIONAL CITY BANK, ENGINEERED PRODUCTS, CANADA, PO BOX 92374, CLEVELAND, OH, 44193	US Mail (1st Class)
24846	TITAN FIRE & INDUSTRIAL CONTROL, 7184 CRESTRIDGE ROAD, MEMPHIS, TN, 38119	US Mail (1st Class)
24846	TMG LLC, 575 BYRNE INDUSTRIAL DR., ROCKFORD, MI, 49341	US Mail (1st Class)
24846	TODD DREISBACH, 17187 LAUREL PARK PLACE, SUITE # 208, LIVONIA, MI, 48152	US Mail (1st Class)
24846	TODD HAIBACH, 14166 MIDDLETON ROAD, WATERFORD, PA, 16441	US Mail (1st Class)
24846	TODD HAMRICK, 40 DOVE DRIVE, MORGANTOWN, WV, 26508	US Mail (1st Class)
24846	TODD TINGLE, 1401 INDUSTRIAL PARK DR., TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	TOLEDO ABRASIVE SUPPLY, (AKA) RICK TURNER & ASSOCIATES INC, 1735 EILEEN RD, TOLEDO, OH, 43615-3962	US Mail (1st Class)
24846	TOM ATTON, 8707 SAMUEL BARTON DRIVE, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	TOM D'ARCA, 31572 INDUSTRIAL RD., SUITE #300, LIVONIA, MI, 48150	US Mail (1st Class)
24846	TOM DELANO, 31572 INDUSTRIAL ROAD, LIVONIA, MI, 48150	US Mail (1st Class)
24846	TOM PETERSON, 1401 INDUSTRIAL PK DR, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	TOM RANDALL, 1401 INDUSTRIAL DRIVE, TUSCALOOSA, AL, 35461	US Mail (1st Class)
24846	TOM TEETRICK, 900 S. WILEY STREET, CRESTLINE, OH, 44827-0320	US Mail (1st Class)
24846	TONI ANN BERTOLO, 2009 WAGNER AVE, ERIE, PA, 16510	US Mail (1st Class)
24846	TONY SHIRLEY, 15 MYRTLE DRIVE, BISHOPVILLE, SC, 29010	US Mail (1st Class)
24846	TONY W GARNER, PO BOX 795, MOUNDVILLE, AL, 35474	US Mail (1st Class)
24846	TOOLS UNLIMITED, 1941 W ALEXIS RD, TOLEDO, OH, 43613	US Mail (1st Class)
24846	TOOL-SMITH, DEPT. 3422, P.O. BOX 2153, BIRMINGHAM, AL, 35287-3422	US Mail (1st Class)
24846	TOOLTEX, INC., 6160 SEEDS ROAD, GROVE CITY, OH, 43123-8603	US Mail (1st Class)
24846	TOPOG-E GASKET COMPANY, 1224 NORTH UTICA AVE., TULSA, OK, 74110	US Mail (1st Class)
24846	TORAY INTERNATIONAL DE MEXICO, SA DE CV, CALZADA SAN PEDRO #250, MONTERREY NL, 11570MEXICO	US Mail (1st Class)
24846	TORAY RESIN CO., 888 WEST BIG BEAVER RD., SUITE 290, TROY, MI, 48084	US Mail (1st Class)
24846	TOSHIBA AMERICA INFORMATION, 1251 AVENUE OF THE AMERICAS STE 4110, NEW YORK, NY, 10020-4110	US Mail (1st Class)
24846	TOTAL AUTOMATION, 837 DEGURSE AVE, MARINE CITY, MI, 48039	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	TOTAL AUTOMATION, 837 DEGURSE AVENUE, MARINE CITY, MI, 48039	US Mail (1st Class)
24846	TOTAL FIRST AID, 3551 JOHNSTOWN RD NE, DOVER, OH, 44622-7565	US Mail (1st Class)
24846	TOTAL HANDLING EQUIP. CO. INC, 1660 VANDERBILT RD, BIRMINGHAM, AL, 35234-1414	US Mail (1st Class)
24846	TOWER GROUP INTERNATIONAL, 128 DEARBORN ST, BUFFALO, NY, 14207-3122	US Mail (1st Class)
24846	TOWLES TRANSPORT, 54600 PONTIAC TRL, MILFORD, MI, 48381-4346	US Mail (1st Class)
24846	TOWN PLACE SUITES BY MARRIOTT, 9369 WATERSTONE BLVD., CINCINNATI, OH, 45249	US Mail (1st Class)
24846	TOWNE AIR FREIGHT, 36506 TREASURY CENTER, CHICAGO, IL. 60694-6500	US Mail (1st Class)
24846	TOYO LEX INDUSTRY CORP, 2284 DANFORTH DR STE 600, LEXINGTON, KY, 40511-2505	US Mail (1st Class)
24846	TOYOTA MATERIALS HANDLING, 2816 5TH AVE S, IRONDALE, AL, 35210-1210	US Mail (1st Class)
24846	TOYOTA MOTOR CREDIT CORP., COMMERCIAL FINANCE, PO BOX 2431, CAROL STREAM, IL, 60132-	US Mail (1st Class)
	2431	
24846	TPC TRAINING SYSTEMS, 750 LAKE-COOK ROAD, SUITE 250, BUFFALO GROVE, IL, 60089	US Mail (1st Class)
24846	TPC WIRE & CABLE, 4570 PAYSHPERE CIRCLE, CHICAGO, IL, 60674	US Mail (1st Class)
24846	TPG GLASS, INC., 2008 8TH STREET, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	TPI SOLUTIONS INC., 2012 W 16TH STREET, ERIE, PA, 16505	US Mail (1st Class)
24846	TRACTOR SUPPLY CREDIT PLAN, DEPT 30-1100243482, PO BOX 689020, DES MOINES, IA, 50368-9020	US Mail (1st Class)
24846	TRANE SERVICE AGENCY, 1900 28TH AVE SOUTH, SUITE 200, HOMEWOOD, AL, 35209	US Mail (1st Class)
24846	TRANS AMERICA GLOBAL, INTERESTATE CAPITAL CORPORATION, PO BOX 1229, SANTA TERESA, NM, 88008	US Mail (1st Class)
24846	TRANS AMERICA GLOBAL, INTERSTATE CAPITAL CORP, PO BOX 1229, SAN TERESA, NM, 88008	US Mail (1st Class)
24846	TRANSCAT CALIBRATION & REPAIR, 7640 MCEWEN RD, CENTERVILLE FINANCE, OH, 45459-3908	US Mail (1st Class)
24846	TRANSFORMER INSPECTION, RETROFILL, 2704 NORMANDY RD., ROYAL OAK, MI, 48073	US Mail (1st Class)
24846	TRANSGROUP, 18850 8TH AVE S, SEATTLE, WA, 98148-1959	US Mail (1st Class)
24846	TRANSLIFT INC., 1960 CHANDALAR DR, STE C, PELHAM, AL, 35124	US Mail (1st Class)
24846	TRANSMATIC, 300 EAST 48TH STREET, HOLLAND, MI, 49423	US Mail (1st Class)
24846	TRANSPORT CLEARINGS EAST, INC., 210 E WOODLAWN RD # 210, CHARLOTTE, NC, 28217-2202	US Mail (1st Class)
24846	TRANS-TEK, RT. 83, PO BOX 338, ELLINGTON, CT, 06029	US Mail (1st Class)
24846	TRAVIS L. SMITH, PO BOX 127, WEST SPRINGFIELD, PA, 16443	US Mail (1st Class)
24846	TRC/SILMIX, INC, 1777 COMMERCE DR., STOW, OH, 44224	US Mail (1st Class)
24846	TREASE L. ABBOTT, 524 WEBB AVE SW, MASSILLON, OH, 44647	US Mail (1st Class)
24846	TREASUER, CITY OF FLINT, PO BOX 367, FLINT, MI, 48501	US Mail (1st Class)
24846	TREASURER OF CRAWFORD COUNTY, BOX 565, BUCYRUS, OH, 44820-0565	US Mail (1st Class)
24846	TREASURER OF STATE OHIO, 30 EAST BROAD STREET, 9TH FLOOR, COLUMBUS, OH, 43266-0001	US Mail (1st Class)
24846	TREASURER, CITY OF CANTON (400), PO BOX 8876, CANTON, OH, 44711	US Mail (1st Class)
24846	TREASURER, CITY OF LAPEER, 576 LIBERTY PK, LAPEER, MI, 48446	US Mail (1st Class)
24846	TREASURER, STATE OF OHIO, OHIO EPA, DEPT L-2711, COLUMBUS, OH, 43260-2711	US Mail (1st Class)
24846	TREASURER, STATE OF OHIO, OFFICE OF FISCAL ADMIN, PO BOX 1049, COLUMBUS, OH, 43216	US Mail (1st Class)
24846	TREASURER, STATE OF OHIO, ADDRESS UNAVAILABLE, COLUMBUS, OH,	US Mail (1st Class)
24846	TREASURER, STATE OF OHIO, OEPA NORTHWEST DISTRICT OFFICE, 347 N DUNBRIDGE RD, BOWLING GREEN, OH, 43402	US Mail (1st Class)
24846	TREE SAVER PALLETS, 1784 MINI RANCH TRAIL, GENEVA, OH, 44041	US Mail (1st Class)
24846	TREXEL, INC., 45 SIXTH ROAD, WOBURN, MA, 01801	US Mail (1st Class)
24846	TRI CORP DESIGN, PO BOX 70055, ROCHESTER HILLS, MI, 48307-0001	US Mail (1st Class)
24846	TRI CORP TECHNOLOGIES, TRI CORP DESIGN, PO BOX 70055, ROCHESTER HILLS, MI, 48307-0001	US Mail (1st Class)
24846	TRI COUNTY EXPRESS DELIVERY, 8099 GLEN PARK RD, KIRTLAND, OH, 44094	US Mail (1st Class)
24846	TRI COUNTY FARMS, 8424 N. STATE ROAD, OTISVILLE, MI, 48463	US Mail (1st Class)
24846	TRI ELECTRIC SUPPLY CO., 8141 TYLER BLVD, MENTOR, OH, 44060-4855	US Mail (1st Class)
24846	TRIAD INC, TU, LLC, PO BOX 691602, CINCINNATI, OH, 45269-1602	US Mail (1st Class)
24846	TRIAD TECHNOLOGIES, LLC, 100 ROCKRIDGE RD, ENGLEWOOD, OH, 45322-2737	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	TRIANGLE SALES CO. INC., 1128 SOUTH STREET, NOBLESVILLE, IN, 46060-3815	US Mail (1st Class)
24846	TRI-CITY TIMES, 594 NORTH ALMONT AVE, IMLAY CITY, MI, 48444	US Mail (1st Class)
24846	TRICO PAINTING, INC., 41 AIRPORT LOOP RD, PHILLIPSBURG, OH, 45354	US Mail (1st Class)
24846	TRICO PRODUCTS CORP., 405 N FRENCH RD, AMHERST, NY, 14228-2010	US Mail (1st Class)
24846	TRICO, INC., 655 BALLARD ROAD, JACKSON, MI, 49201	US Mail (1st Class)
24846	TRICOR INDUSTRIAL, INC., PO BOX 752, 823 W BOWMAN STREET, WOOSTER, OH, 44691-0752	US Mail (1st Class)
24846	TRIDENT LIGHTING LLC, 2929 32ND ST SE, KENTWOOD, MI, 49512-1754	US Mail (1st Class)
24846	TRIDENT TOOLING PTY. LTD., 8 TRANSPORT AVENUE, NETLEY, SA, 5037AUSTRALIA	US Mail (1st Class)
24846	TRIDON, 8100 TRIDON DR, SMYRNA, TN, 37167-6603	US Mail (1st Class)
24846	TRIDON, INC, ATTN DAVID ENGLISH, 3200 PARKER DR, ST AUGUSTINE, FL, 32084	US Mail (1st Class)
24846	TRIDON, INC, (RE: TRIDON), ATTN DAVID ENGLISH, 3200 PARKER DR, ST AUGUSTINE, FL, 32084	US Mail (1st Class)
24846	TRI-MI MOULD(1990) INC., 1740 MORO DRIVE R.R.#1, WINDSOR, ON, N9A 6J3CANADA	US Mail (1st Class)
24846	TRINETICS GROUP INC., 530 S DOCK STREET, SHARON, PA, 16146	US Mail (1st Class)
24846	TRINITY TOOL COMPANY, PO BOX 98, FRASER, MI, 48026-0098	US Mail (1st Class)
24846	TRIOLOGY INC., 1070 E AVON RD, ROCHESTER HILLS, MI, 48307-2424	US Mail (1st Class)
24846	TRI-PALM INTERNATIONAL, 265 N. HAMILTON RD., COLUMBUS, OH, 43213-0150	US Mail (1st Class)
24846	TRIPLE R AMERICA, 225 MACPHERSON AVENUE, UNIT C, TORONTO, ON, M4V 1A1CANADA	US Mail (1st Class)
24846	TRIPLE S CONSTRUCTION, 8563 SILVERWOOD RD, SILVERWOOD, MI, 48760	US Mail (1st Class)
24846	TRI-POWER MPT, INC., 1447 SOUTH MAIN ST, AKRON, OH, 44301	US Mail (1st Class)
24846	TRIPPLEHORNS TRAVELIN TRACTOR, 2896 KLAM RD., LAPEER, MI, 48446	US Mail (1st Class)
24846	TRISM SPECIALIZED CARRIER, 8141 E 7TH ST, JOPLIN, MO, 64801-9283	US Mail (1st Class)
24846	TRI-STATE EXPEDITED, 27681 CUMMINGS RD, MILLBURY, OH, 43447-9762	US Mail (1st Class)
24846	TROY TOOLING, INC., 1890 ROCHESTER INDUSTRIAL DR, ROCHESTER HILLS, MI, 48309-3341	US Mail (1st Class)
24846	TRUGREEN-CHEMLAWN (4740), 110 INDUSTRIAL DRIVE, PO BOX 3088, MANSFIELD, OH, 44904-0088	US Mail (1st Class)
24846	TRUMBULL COUNTY COURT, EASTERN DISTRICT, 7130 BROOKWOOD DR., BROOKFIELD, OH, 44403	US Mail (1st Class)
24846	TRUMBULL INDUSTRIES, 400 DIETZ RD NE, WARREN, OH, 44483-2749	US Mail (1st Class)
24846	TRUMBULL INDUSTRIES, INC., ATTN: CREDIT DEPT, 400 DIETZ RD NE, WARREN, OH, 44483	US Mail (1st Class)
24846	TRUMBULL STEEL, PHILIP METALS INC., PO BOX 931939-N, CLEVELAND, OH, 44193-0246	US Mail (1st Class)
24846	TSE INDUSTRIES, INC., 4370 112TH TER N, CLEARWATER, FL, 33762-4902	US Mail (1st Class)
24846	TTL, INC., PO DRAWER 1128, TUSCALOOSA, AL, 35403	US Mail (1st Class)
24846	TUBERIA DE ACERO, AVE. INDEPENDENCIA NO. 4604, CHIHUAHUA, CHIH., MEXICO	US Mail (1st Class)
24846	TUCKER MASSEY MKTG, 2201 11TH ST, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	TURBEN TRUCKING, RD 1 BOX 40D, BEAR LAKE, PA, 16402	US Mail (1st Class)
24846	TURBO BRAZE CORP., 687 LEHIGH AVE., PO BOX 897, UNION, NJ, 07083-0897	US Mail (1st Class)
24846	TURBOSPRAY, 1172 FENDT DRIVE, BUILDING 300, HOWELL, MI, 48843	US Mail (1st Class)
24846	TURNKEY MANUFACTURING SEMINARS, PO BOX 803370, KANSAS CITY, MO, 64180-3370	US Mail (1st Class)
24846	TUSCALOOSA BLUEPRINTING, 2009 UNIVERSITY BLVD., PO BOX 21304, TUSCALOOSA, AL, 35402	US Mail (1st Class)
24846	TUSCALOOSA CHEVROLET, 6500 INTERSTATE PKWY, COTTONDALE, AL, 35453-4327	US Mail (1st Class)
24846	TUSCALOOSA CITY REVENUE, PO BOX 2089, TUSCALOOSA, AL, 35403-2089	US Mail (1st Class)
24846	TUSCALOOSA CITY REVENUE DEPT, 2201 UNIVERSITY BLVD, TUSCALOOSA, AL, 35401-1541	US Mail (1st Class)
24846	TUSCALOOSA CITY SALES TAX, 2201 UNIVERSITY BLVD, TUSCALOOSA, AL, 35401-1541	US Mail (1st Class)
24846	TUSCALOOSA COUNTY, USE TAX RETURN, PO BOX 20738, TUSCALOOSA, AL, 35402-0738	US Mail (1st Class)
24846	TUSCALOOSA COUNTY INDUSTRIAL, DEVELOPMENT AUTHORITY, PO BOX 2667, TUSCALOOSA, AL, $35403$	US Mail (1st Class)
24846	TUSCALOOSA HUMAN RESOURCE PROF, 2201 UNIVERSITY BLVD, TUSCALOOSA, AL, 35401-1541	US Mail (1st Class)
24846	TUSCALOOSA IRON AND METAL, 2701 31ST ST, TUSCALOOSA, AL, 35401-7084	US Mail (1st Class)
24846	TUSCALOOSA MACHINE & WELDING, 1715 A HOLT ROAD, TUSCALOOSA, AL, 35404	US Mail (1st Class)
24846	TUSCALOOSA MACHINE & WELDING, 1715 A HOLT RD, TUSCALOOSA, AL, 35404	US Mail (1st Class)
24846	TUSCALOOSA OPTICAL LAB, 2324 6TH STREET, TUSCALOOOSA, AL, 35401	US Mail (1st Class)

1956AL   1	Svc Lst	Name and Address of Served Party	Mode of Service
24846   TUSCARORA INC., 2902 EXPANSION BLVD., STORM LAKE, IA. 50588   US Mail (1st Class)   US Mail (1st Cla	24846	TUSCALOOSA SCALES, 4050 INDUSTRIAL DRIVE, COTTONDALE, AL, 35453	US Mail (1st Class)
24816         TUTHILL PNEUMATICS & VACUUM, 8500 S MADISON ST, BURR RIDGE, IL, 60527-6284         US Malf (1sc Class)           24846         TVASA, PO BOX 125, WEST ALEXANDRIA, OH, 45831         US Malf (1sc Class)           24846         TWO MEN 8 A TRUCK, 1420 LLOYD ROAD, WICKLIFFE, OH, 44092         US Malf (1sc Class)           24846         TYCO ELECTRONICS, 449 EISENHOWER BLVD, HARRISBURG, PA, 17111-2302         US Malf (1sc Class)           24846         TYCO ELECTRONICS, 449 EISENHOWER BLVD, HARRISBURG, PA, 17111-2302         US Malf (1sc Class)           24846         US BUREAU OF CITIZENSHIP, AND IMMIGRATION SERVICES, CIC ELI MAROKO, 31731         US Malf (1sc Class)           24846         US MOLDING MACHINERY CO., 38294 PELTON RD, WILLOUGHBY, OH, 44094         US Malf (1sc Class)           24846         US SOFTAL SERVICE, CIMERS-PR, DO BOX 0566, CAROL STERAM IL, 60132-0566         US Malf (1sc Class)           24846         US, DEPT. OF EDUCATION, 400 MARYLAND AVE SW, WASHINGTON, DC, 20202-0001         US Malf (1sc Class)           24846         US, DOSTMA, SERVICE, CMRS-POC, PO BOX 0575, ACCT #88109132, CAROL STREAM, IL, 60132-0557         US Malf (1sc Class)           24846         US, DOSTMA, SERVICE, CMRS-POC, PO BOX 0576, ACCT #88109132, CAROL STREAM, IL, 60132-0557         US Malf (1sc Class)           24846         US, DOSTMA, STER, MAIN POST OFFICE, 2600 CLEVELAND AVE WIN, CANTON, OH, 44711         US Malf (1sc Class)           24846	24846	TUSCARORA CT., 29 PARK ST., PUTNAM, CT, 06260	US Mail (1st Class)
1758A, PO BOX 125, WEST ALEXANDRIA, OH, 45381   US Mail (tst Class)   US Mail (tst Cla	24846	TUSCARORA INC., 2902 EXPANSION BLVD., STORM LAKE, IA, 50588	US Mail (1st Class)
TWIN TOOL LLC, 4848 GATEWAY CIRCLE, KETTERING, OH, 45440   US Mail (1st Class)   US Ma	24846	TUTHILL PNEUMATICS & VACUUM, 8500 S MADISON ST, BURR RIDGE, IL, 60527-6284	US Mail (1st Class)
TWO MEN & A TRUCK, 1420 LLOYD ROAD, WICKLIFFE, OH, 44092	24846	TVSAA, PO BOX 125, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846   UDO IT, 13315 15 MILE RD., STERLING HEIGHTS, MI. 48312   US Mail (1st Class)	24846	TWIN TOOL LLC, 4648 GATEWAY CIRCLE, KETTERING, OH, 45440	US Mail (1st Class)
24846   U.S. BUREAU OF CITIZENSHIP, AND IMMIGRATION SERVICES, CIO ELI MAROKO, 31731   US Mail (1st Class) NORTHWESTERN HUY, SUITE 165, FARMINGTON HILLS, MI, 48334-1663   US MAICHWESTERN HUY, SUITE 165, FARMINGTON HILLS, MI, 48334-1663   US MAICHWESTERN HUY, SUITE 165, FARMINGTON HILLS, MI, 48334-1663   US MAICHWESTERN HUY, SUITE 165, FARMINGTON HULLS, MI, 48334-1663   US MAICHWESTER, 4420 JEFFERIES HWY, WALTERBORO, SC, 29488-6488   US Mail (1st Class) 124846   US WASTE INDUSTRIES, 4420 JEFFERIES HWY, WALTERBORO, SC, 29488-6488   US Mail (1st Class) 124846   US. POSTAL SERVICE, CMRS-PB, PO BOX 0566, CAROL STREAM, II., 60132-0566   US Mail (1st Class) 124846   US. PETROLON IND., INC., 4900 RENTYORTH DRIVE, LINCOLIN, NE, 68516   US MAIL (1st Class) 124846   US. POSTAL SERVICE, CMRS-POC, PO BOX 0575, ACCT #88109132, CAROL STREAM, III., 60132-0575   US Mail (1st Class) 124846   US. POSTAL SERVICE, CMRS-POC, PO BOX 0575, ACCT #88109132, CAROL STREAM, III., 60132-0575   US Mail (1st Class) 124846   US. DE, CAROL STREAM, III., 60132-0575   US Mail (1st Class) 124846   US. DE, CAROL STREAM, III., 60132-0575   US Mail (1st Class) 124846   US. DE, CAROL STREAM, III., 60132-0575   US Mail (1st Class) 124846   US. DE, CAROL STREAM, III., 60132-0575   US Mail (1st Class) 124846   US. DE, ALDET 262, 4260 WISCONSIN AVE NW. WASHINGTON, DC, 2016-2157   US Mail (1st Class) 124846   US. DE, ALDET 262, 4260 WISCONSIN AVE NW. WASHINGTON, DC, 2016-2157   US Mail (1st Class) 124846   US. DE, ALDET 262, 4260 WISCONSIN AVE NW. WASHINGTON, DC, 2016-2157   US Mail (1st Class) 124846   US. DE, ALDET 262, 4260 WISCONSIN AVE NW. WASHINGTON, DC, 2016-2157   US Mail (1st Class) 124846   US. MILL (1st Class) 124846   U	24846	TWO MEN & A TRUCK, 1420 LLOYD ROAD, WICKLIFFE, OH, 44092	US Mail (1st Class)
24846         U S BUREAU OF CITIZENSHIP, AND IMMICRATION SERVICES, C/O ELI MAROKO, 31731         US Mail (1st Class)           24846         U S MOLDING MACHINERY CO., 32924 PELTON RD., WILLOUGHBY, OH, 44094         US Mail (1st Class)           24846         U S MOSTE INDUSTRIES, 4420 JEFFERIES HWY, WALTERBORO, SC. 29488-6488         US Mail (1st Class)           24846         U S. POSTAL SERVICE. CMRS-PB, PO BOX 0566, CAROL STREAM, IL., 60132-0566         US Mail (1st Class)           24846         U.S. DEPT. OF EDUCATION, 400 MARYLAND AVE SW, WASHINGTON, DC, 20202-0001         US Mail (1st Class)           24846         U.S. POSTAL SERVICE. CMRS-POC, PO BOX 0575, ACCT #88109132, CAROL STREAM, IL., 60132-0575         US Mail (1st Class)           24846         U.S. POSTAL SERVICE, CMRS-POC, PO BOX 0575, ACCT #88109132, CAROL STREAM, IL., 60132-0575         US Mail (1st Class)           24846         U.S. POSTAL SERVICE, CMRS-POC, PO BOX 0575, ACCT #88109132, CAROL STREAM, IL., 60132-0575         US Mail (1st Class)           24846         U.S. POSTMASTER, MAIN POST OFFICE, 2650 CLEVELAND AVE NW, CANTON, OH, 44711         US Mail (1st Class)           24846         U.S. POSTMASTER, MAIN POST OFFICE, 2650 CLEVELAND AVE NW, CANTON, OH, 44711         US Mail (1st Class)           24846         US S. D.S. A., DEPT 282, 4200 WISCONSIN AVE NW, WASHINGTON, OC., 2016-2157         US Mail (1st Class)           24846         US S. D.S. A., DEPT 382, 4200 WISCONSIN AVE NW, WASHINGTON, OC., 2016-2157 <t< th=""><th>24846</th><td>TYCO ELECTRONICS, 449 EISENHOWER BLVD, HARRISBURG, PA, 17111-2302</td><td>US Mail (1st Class)</td></t<>	24846	TYCO ELECTRONICS, 449 EISENHOWER BLVD, HARRISBURG, PA, 17111-2302	US Mail (1st Class)
NORTHWESTERN HWY, SUITE 165, FARMINGTON HILLS, MI, 48334-1663  24846 U S MOLDING MACHINERY CO., 38294 PELTON RD., WILLOUGHBY, OH, 240994  24846 U S MOLDING MACHINERY CO., 38294 PELTON RD., WILLOUGHBY, OH, 240964  24846 U.S. POSTAL SERVICE, CMRS-PB, PO BOX 0566, CAROL STREAM, IL, 60132-0566  24846 U.N.I.T.E. LOCAL #1902. C/O CHUCK HOMA, PO BOX 127, MIDDLEFIELD, OH, 44062  24846 U.S. PETROLON IND., INC., 4900 RENTWORTH DRIVE, LINCOLN, NE, 68516  24846 U.S. PETROLON IND., INC., 4900 RENTWORTH DRIVE, LINCOLN, NE, 68516  24846 U.S. POSTAL SERVICE, CMRS-PO, PO BOX 0575, ACCT #86109132, CAROL STREAM, IL, 60132-0575  24846 U.S. POSTAL SERVICE, CMRS-PO, PO BOX 0575, ACCT #86109132, CAROL STREAM, IL, 60132-0575  24846 U.S. POSTAMSTER MAIN POST OFFICE, 2680 CLEVELAND AVE NW, CANTON, OH, 44711  24846 U.S. DOSTAMSTER MAIN POST OFFICE, 2680 CLEVELAND AVE NW, CANTON, OH, 44711  24846 U.S. D.S. A., DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157  24846 U.HAUL, 2727 N CENTRAL AVE, PHOENIX, AZ, 85004-1155  24846 U.HAUL, 2727 N CENTRAL AVE, PHOENIX, AZ, 85004-1155  24846 U.HINE, 2200 S LAKESIDE DRIVE, WAUKEGAN, IL, 60085  24846 ULTIMA ED OFFICE, INC., 317 FAIRFIELD RD, FREEHOLD, NJ, 07728-7829  24846 ULTRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429  24846 ULTRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429  24846 ULTRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429  24846 ULTRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429  24846 ULTRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429  24846 ULTRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429  24846 ULTRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429  24846 ULTRA CHURCH STAMBER RD, SOUTHFIELD, MI, 48903  24846 ULTRA CHURCH STAMBER, PO PO BOX 8211, ASHTABULA, OH, 44005-0321  24846 ULTRA CHURCH STAMBER, PO PO BOX 8211, ASHTABULA, OH, 44005-0321  24846 UNISCEN THE COUNTER, INC., 5023 STATE ROAD	24846	U DO IT, 13315 15 MILE RD., STERLING HEIGHTS, MI, 48312	US Mail (1st Class)
24846         U.S. WASTE INDUSTRIES, 4420 JEFFERIES HWY, WALTERBORO, SC, 29488-6488         U.S. Mail (1st Class)           24846         U.S. POSTAL SERVICE, CMRS-PB, PO BOX 0566, CAROL STREAM, II., 60132-0566         US Mail (1st Class)           24846         U.S. DEPT. OF EDUCATION, 400 MARYLAND AVE SW, WASHINGTON, DC, 20202-0001         US Mail (1st Class)           24846         U.S. DEPT. OF EDUCATION, 400 MARYLAND AVE SW, WASHINGTON, DC, 20202-0001         US Mail (1st Class)           24846         U.S. DEPT. OF EDUCATION, 400 MARYLAND AVE SW, WASHINGTON, DC, 20202-0007         US Mail (1st Class)           24846         U.S. DEPT. SERVICE, CMRS-POC, PO BOX 0575, ACCT #88109132, CAROL STREAM, II., 60132-0575         US Mail (1st Class)           24846         U.S. D.S.A., DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         U.S. D.S.A., DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         U.S. D.S.A., DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         U.S. DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         U-HAUL, 2727 N CENTRAL AVE, PHOENIX, AZ, 85004-1155         US Mail (1st Class)           24846         U-HAUL, 2727 N CENTRAL AVE, PHOENIX, AZ, 85004-1155         US Mail (1st Class)           24846         ULITRA,	24846		US Mail (1st Class)
24846         U.S. POSTAL SERVICE, CMRS-PB, PO BOX 0566, CAROL STREAM, IL, 60132-0566         US Mail (1st Class)           24846         U.S. DEPT. OF EDUCATION, 400 MARYLAND AVE SW, WASHINGTON, DC, 20202-0001         US Mail (1st Class)           24846         U.S. PETROLON IND., INC., 4900 RENTWORTH DRIVE, LINCOLN, NE, 68516         US Mail (1st Class)           24846         U.S. POSTAL SERVICE, CMRS-POC, PO BOX 0575, ACCT 488109132, CAROL STREAM, IL, 60132-0575         US Mail (1st Class)           24846         U.S. DOSTAL SERVICE, CMRS-POC, PO BOX 0575, ACCT 488109132, CAROL STREAM, IL, 60132-0575         US Mail (1st Class)           24846         U.S. D. SO STMASTER, MAIN POST OFFICE, 2650 CLEVELAND AVE NW, CANTON, OH, 44711         US Mail (1st Class)           24846         U.S. D. S. D. DEPT 222 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         U.S. D. S. A., DEPT 224 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         UHAUL, 2727 N CENTRAL AVE, PHOENIX, AZ, 85004-1155         US Mail (1st Class)           24846         UHINE, 200 S LAKESIDE DRIVE, WAUKEGAN, IL, 60085         US Mail (1st Class)           24846         ULTIMATE OFFICE, INC., 317 FAIRFIELD RD, FREEHOLD, NJ, 07728-7829         US Mail (1st Class)           24846         ULTRA FAB, INC., CHURCH STREET STATION, PO BOX 4229, NEW YORK, NY, 10249-6429         US Mail (1st Class)           24846	24846	U S MOLDING MACHINERY CO., 38294 PELTON RD., WILLOUGHBY, OH, 44094	US Mail (1st Class)
24846         U.N.I.T.E. LOCAL #1902, C/O CHUCK HOMA, PO BOX 127, MIDDLEFIELD, OH, 44062         U.S. DEPT. OF EDUCATION, 400 MARYLAND AVE SW, WASHINGTON, DC, 2020-20001         US Mail (1st Class)           24846         U.S. DEPT. OF EDUCATION, 400 MARYLAND AVE SW, WASHINGTON, DC, 2020-20001         US Mail (1st Class)           24846         U.S. POSTAL SERVICE, CMRS-POC, PO BOX 0575, ACCT #88109132, CAROL STREAM, IL, 60132-0575         US Mail (1st Class)           24846         U.S. POSTMASTER, MAIN POST OFFICE, 2650 CLEVELAND AVE NW, CANTON, OH, 44711         US Mail (1st Class)           24846         U.S. D.S.A., DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         U.S. CORP, 13690 RIVERPORT DR, MARYLAND HEIGHTS, MO, 63043-4801         US Mail (1st Class)           24846         U.S. POSTMAS REGIONAL, 13207 RAVENNA RD, CHARDON, OH, 44024-7032         US Mail (1st Class)           24846         U.HINE, 2200 S LAKESIGE DRIVE, WAUKEGAN, IL, 60085         US Mail (1st Class)           24846         U.LITRA, FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429         US Mail (1st Class)           24846         U.LTRA, FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429         US Mail (1st Class)           24846         U.LTRA, MAINTENANCE CO., PO BOX 3311, ERIE, PA, 16508         US Mail (1st Class)           24846         U.LTRA, MAINTENANCE CO., PO BOX 3311, ERIE, PA, 16508         US	24846	U S WASTE INDUSTRIES, 4420 JEFFERIES HWY, WALTERBORO, SC, 29488-6488	US Mail (1st Class)
24846         U.S. DEPT. OF EDUCATION, 400 MARYLAND AVE SW, WASHINGTON, DC, 20202-0001         US Mail (1st Class)           24846         U.S. PETROLON IND., INC., 4900 RENTWORTH DRIVE, LINCOLN, NE, 68516         US MAIL (1st Class)           24846         U.S. POSTAL SERVICE, CMRS-POC, PO BOX 0575, ACCT #88109132, CAROL STREAM, IL, 60132-0575         US Mail (1st Class)           24846         U.S. POSTALSTER, MAIN POST OFFICE, 2550 CLEVELAND AVE NW, CANTON, OH, 44711         US Mail (1st Class)           24846         U.S. D.S.A., DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         U.S. D.S.A., DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         U.S. D.S.A., DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         U.S. D.S.A. DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         U.S. D.S.A. DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         U.H. D.S. D.S. S.	24846	U. S. POSTAL SERVICE, CMRS-PB, PO BOX 0566, CAROL STREAM, IL, 60132-0566	US Mail (1st Class)
24846         U.S. PETROLON IND., INC., 4900 RENTWORTH DRIVE, LINCOLN, NE, 68516         US Mail (1st Class)           24846         U.S. POSTAL SERVICE, CMRS-POC, PO BOX 0575, ACCT #88109132, CAROL STREAM, IL, 60132-0575         US Mail (1st Class)           24846         U.S. D.SAL, DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         U.S. D.S.A., DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         U.S. D.S.A., DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         U.S. D.S.A., DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         U.HAUL, 2727 N CENTRAL AVE, PHOENIX, AZ, 85004-1155         US Mail (1st Class)           24846         U.HINS, GEAUGA REGIONAL, 13207 RAVENNA RD, CHARDON, OH, 44024-7032         US Mail (1st Class)           24846         U.HINE, 2200 S LAKESIDE DRIVE, WAUKEGAN, IL, 60085         US Mail (1st Class)           24846         U.LITRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429         US Mail (1st Class)           24846         U.LTRA MAINTENANCE CO., PO BOX 3311, ERIE, PA, 16508         US Mail (1st Class)           24846         U.LTRA, MAINTENANCE CO., PO BOX 3311, ERIE, PA, 16508         US Mail (1st Class)           24846         U.LTRA, MAINTENANCE CO., PO BOX 3311, ERIE,	24846	U.N.I.T.E. LOCAL #1902, C/O CHUCK HOMA, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846         U.S. POSTAL SERVICE, CMRS-POC, PO BOX 0575, ACCT #88109132, CAROL STREAM, IL, 60132-0575         US Mail (1st Class)           24846         U.S. POSTMASTER, MAIN POST OFFICE, 2650 CLEVELAND AVE NW, CANTON, OH, 44711         US Mail (1st Class)           24846         U.S. D.S.A., DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         U.S. D.S.A., DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         U.S. D.S.A., DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         U.HAUL, 2727 N CENTRAL AVE, PHOENIX, AZ, 85004-1155         US Mail (1st Class)           24846         U.HIR, 2200 S LAKESIDE DRIVE, WAUKEGAN, IL, 60085         US Mail (1st Class)           24846         U.LTRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429         US Mail (1st Class)           24846         ULTRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429         US Mail (1st Class)           24846         ULTRA FAB, INC., CS860 LAHSER RD, SOUTHFIELD, MI, 48034         US Mail (1st Class)           24846         ULTRA MAINTENANCE CO., PO BOX 3311, ERIE, PA, 16508         US Mail (1st Class)           24846         ULTRA-TECH AUTOMATION, INC, 33671 DOREKA DRIVE, FRASER, MI, 48026         US Mail (1st Class)           24846         UNBER THE COUNTER, INC, 5023 ST	24846	U.S. DEPT. OF EDUCATION, 400 MARYLAND AVE SW, WASHINGTON, DC, 20202-0001	US Mail (1st Class)
24846         U.S. POSTMASTER, MAIN POST OFFICE, 2650 CLEVELAND AVE NW, CANTON, OH, 44711         US Mail (1st Class)           24846         U.S. D.S.A., DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         U.S. D.S.A., DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         U.HAUL, 2727 N CENTRAL AVE, PHOENIX, AZ, 85004-1155         US Mail (1st Class)           24846         U.HIS GEAUGA REGIONAL, 13207 RAVENNA RD, CHARDON, OH, 44024-7032         US Mail (1st Class)           24846         U.LINE, 2200 S LAKESIDE DRIVE, WAUKEGAN, IL, 60085         US Mail (1st Class)           24846         U.LITRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429         US Mail (1st Class)           24846         ULTRA MAINTENANCE CO., PO BOX 3311, ERIE, PA, 16508         US Mail (1st Class)           24846         ULTRA MAINTENANCE CO., PO BOX 3311, ERIE, PA, 16508         US Mail (1st Class)           24846         ULTRACOM, INC., 25860 LAHSER RD, SOUTHFIELD, MI, 48034         US Mail (1st Class)           24846         ULTRACOM, INC., DEPAWER 20428, WACO, TX, 76702-0428         US Mail (1st Class)           24846         ULTRA-TECH AUTOMATION, INC, 27560 HAGGERTY RD., SUITE A-11, FARMINGTON HILLS, MI, 48331         US Mail (1st Class)           24846         UNDER THE COUNTER, INC., 5023 STATE ROAD, PO BOX 321, ASHTABULA, OH, 44005-0321	24846	U.S. PETROLON IND., INC., 4900 RENTWORTH DRIVE, LINCOLN, NE, 68516	US Mail (1st Class)
24846         U.S.D.S.A., DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157         US Mail (1st Class)           24846         UGS CORP, 13690 RIVERPORT DR, MARYLAND HEIGHTS, MO, 63043-4801         US Mail (1st Class)           24846         U-HAUL, 2727 N CENTRAL AVE, PHOENIX, AZ, 85004-1155         US Mail (1st Class)           24846         U-HINE, 2200 S LAKESIDE DRIVE, WAUKEGAN, IL, 60085         US Mail (1st Class)           24846         ULTIMATE OFFICE, INC., 317 FAIRFIELD RD, FREEHOLD, NJ, 07728-7829         US Mail (1st Class)           24846         ULTRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429         US Mail (1st Class)           24846         ULTRA MAINTENANCE CO., PO BOX 3311, ERIE, PA, 16508         US Mail (1st Class)           24846         ULTRACOM, INC., 25860 LAHSER RD, SOUTHFIELD, MI, 48034         US Mail (1st Class)           24846         ULTRALAMATION INC., PO DRAWER 20428, WACO, TX, 76702-0428         US Mail (1st Class)           24846         ULTRA-TECH AUTOMATION, INC, 33671 DOREKA DRIVE, FRASER, MI, 48026         US Mail (1st Class)           24846         UNBERTLA DIGITAL SOLUTIONS INC, 27260 HAGGERTY RD, SUITE A-11, FARMINGTON HILLS, MI, 48331         US Mail (1st Class)           24846         UNDER WILLE DIGITAL SOLUTIONS INC, MILACRON, PO BOX 321, ASHTABULA, OH, 44005-0321         US Mail (1st Class)           24846         UNILOY MILACRON, INC., MILACRON MARKETING, PO BOX 77199, DETR	24846	U.S. POSTAL SERVICE, CMRS-POC, PO BOX 0575, ACCT #88109132, CAROL STREAM, IL, 60132-0575	US Mail (1st Class)
24846         UGS CORP, 13690 RIVERPORT DR, MARYLAND HEIGHTS, MO, 63043-4801         US Mail (1st Class)           24846         U-HAUL, 2727 N CENTRAL AVE, PHOENIX, AZ, 85004-1155         US Mail (1st Class)           24846         UHHS GEAUGA REGIONAL, 13207 RAVENNA RD, CHARDON, OH, 44024-7032         US Mail (1st Class)           24846         U-LINE, 2200 S LAKESIDE DRIVE, WAUKEGAN, IL, 60085         US Mail (1st Class)           24846         ULTRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429         US Mail (1st Class)           24846         ULTRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429         US Mail (1st Class)           24846         ULTRA MAINTENANCE CO., PO BOX 3311, ERIE, PA, 16508         US Mail (1st Class)           24846         ULTRACOM, INC., 25860 LAHSER RD, SOUTHFIELD, MI, 48034         US Mail (1st Class)           24846         ULTRACOM, INC., 25860 LAHSER RD, SOUTHFIELD, MI, 48034         US Mail (1st Class)           24846         ULTRACTECH AUTOMATION, INC., 23671 DOREKA DRIVE, FRASER, MI, 48026         US Mail (1st Class)           24846         UNDER THE COUNTER, INC., 5023 STATE ROAD, PO BOX 321, ASHTABULA, OH, 44005-0321         US Mail (1st Class)           24846         UNILOY MILACRON INC., MILACRON MARKETING, PO BOX 77199, DETROIT, MI, 48277-0199         US Mail (1st Class)           24846         UNILOY MILACRON, INC., MILACRON MARKETING, PO BOX 77199, DETROIT, MI, 482	24846	U.S. POSTMASTER, MAIN POST OFFICE, 2650 CLEVELAND AVE NW, CANTON, OH, 44711	US Mail (1st Class)
24846         U-HAUL, 2727 N CENTRAL AVE, PHOENIX, AZ, 85004-1155         US Mail (1st Class)           24846         UHHS GEAUGA REGIONAL, 13207 RAVENNA RD, CHARDON, OH, 44024-7032         US Mail (1st Class)           24846         ULINE, 2200 S LAKESIDE DRIVE, WAUKEGAN, IL, 60085         US Mail (1st Class)           24846         ULTRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429         US Mail (1st Class)           24846         ULTRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429         US Mail (1st Class)           24846         ULTRA MAINTENANCE CO., PO BOX 3311, ERIE, PA, 16508         US Mail (1st Class)           24846         ULTRACOM, INC., 25860 LAHSER RD, SOUTHFIELD, MI, 48034         US Mail (1st Class)           24846         ULTRAMATION INC., PO DRAWER 20428, WACO, TX, 76702-0428         US Mail (1st Class)           24846         ULTRAMATION INC., PO DRAWER 20428, WACO, TX, 76702-0428         US Mail (1st Class)           24846         UNIDER FILE COUNTER, INC., 33671 DOREKA DRIVE, FRASER, MI, 48026         US Mail (1st Class)           24846         UNDER FILE COUNTER, INC., 5023 STATE ROAD, PO BOX 321, ASHTABULA, OH, 44005-0321         US Mail (1st Class)           24846         UNILOY MILACRON INC., 5550 OCCIDENTAL HWY, SUITE B, TECUMSEH, MI, 49286         US Mail (1st Class)           24846         UNILOY MILACRON, INC., MILACRON MARKETING, PO BOX 77199, DETROIT, MI, 48277-0199	24846	U.S.D.S.A., DEPT 262, 4200 WISCONSIN AVE NW, WASHINGTON, DC, 20016-2157	US Mail (1st Class)
24846         UHHS GEAUGA REGIONAL, 13207 RAVENNA RD, CHARDON, OH, 44024-7032         US Mail (1st Class)           24846         U-LINE, 2200 S LAKESIDE DRIVE, WAUKEGAN, IL, 60085         US Mail (1st Class)           24846         ULTIMATE OFFICE, INC., 317 FAIRFIELD RD, FREEHOLD, NJ, 07728-7829         US Mail (1st Class)           24846         ULTRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429         US Mail (1st Class)           24846         ULTRA MAINTENANCE CO., PO BOX 3311, ERIE, PA, 16508         US Mail (1st Class)           24846         ULTRACOM, INC., 25860 LAHSER RD, SOUTHFIELD, MI, 48034         US Mail (1st Class)           24846         ULTRAMATION INC., PO DRAWER 20428, WACO, TX, 76702-0428         US Mail (1st Class)           24846         ULTRA-TECH AUTOMATION,INC, 33671 DOREKA DRIVE, FRASER, MI, 48026         US Mail (1st Class)           24846         UNDER THE COUNTER, INC., 5023 STATE ROAD, PO BOX 321, ASHTABULA, OH, 44005-0321         US Mail (1st Class)           24846         UNIDOY MILACRON INC., 5550 OCCIDENTAL HWY, SUITE B, TECUMSEH, MI, 49286         US Mail (1st Class)           24846         UNILOY MILACRON, 1165 HALFACRE RD., BATAVIA, OH, 45103         US Mail (1st Class)           24846         UNILOY MILACRON, 1165 HALFACRE RD., BATAVIA, OH, 45103         US Mail (1st Class)           24846         UNINON COUNTY TREASURER, 233 W 6TH ST., MARYSVILLE, OH, 43040-5513         US Mail (1st Cl	24846	UGS CORP, 13690 RIVERPORT DR, MARYLAND HEIGHTS, MO, 63043-4801	US Mail (1st Class)
24846         U-LINE, 2200 S LAKESIDE DRIVE, WAUKEGAN, IL, 60085         US Mail (1st Class)           24846         ULTIMATE OFFICE, INC., 317 FAIRFIELD RD, FREEHOLD, NJ, 07728-7829         US Mail (1st Class)           24846         ULTRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429         US Mail (1st Class)           24846         ULTRA AMAINTENANCE CO., PO BOX 3311, ERIE, PA, 16508         US Mail (1st Class)           24846         ULTRACOM, INC., 25860 LAHSER RD, SOUTHFIELD, MI, 48034         US Mail (1st Class)           24846         ULTRAMATION INC., PO DRAWER 20428, WACO, TX, 76702-0428         US Mail (1st Class)           24846         ULTRA-TECH AUTOMATION,INC, 33671 DOREKA DRIVE, FRASER, MI, 48026         US Mail (1st Class)           24846         UNBRELLA DIGITAL SOLUTIONS INC, 27260 HAGGERTY RD., SUITE A-11, FARMINGTON HILLS, MI, 48331         US Mail (1st Class)           24846         UNDER THE COUNTER, INC., 5023 STATE ROAD, PO BOX 321, ASHTABULA, OH, 44005-0321         US Mail (1st Class)           24846         UNILOY MILACRON INC., 5550 OCCIDENTAL HWY, SUITE B, TECUMSEH, MI, 49286         US Mail (1st Class)           24846         UNILOY MILACRON, INC., MILACRON MARKETING, PO BOX 77199, DETROIT, MI, 48277-0199         US Mail (1st Class)           24846         UNILOY MILICRON, 4165 HALFACRE RD, BATAVIA, OH, 45103         US Mail (1st Class)           24846         UNILOY MILICRON, 4165 HALFACRE RD, BATAVIA, O	24846	U-HAUL, 2727 N CENTRAL AVE, PHOENIX, AZ, 85004-1155	US Mail (1st Class)
24846         ULTIMATE OFFICE, INC., 317 FAIRFIELD RD, FREEHOLD, NJ, 07728-7829         US Mail (1st Class)           24846         ULTRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429         US Mail (1st Class)           24846         ULTRA MAINTENANCE CO., PO BOX 3311, ERIE, PA, 16508         US Mail (1st Class)           24846         ULTRACOM, INC., 25860 LAHSER RD, SOUTHFIELD, MI, 48034         US Mail (1st Class)           24846         ULTRAAMATION INC., PO DRAWER 20428, WACO, TX, 76702-0428         US Mail (1st Class)           24846         ULTRA-TECH AUTOMATION,INC, 33671 DOREKA DRIVE, FRASER, MI, 48026         US Mail (1st Class)           24846         UMBRELLA DIGITAL SOLUTIONS INC, 27260 HAGGERTY RD., SUITE A-11, FARMINGTON HILLS, MI, 48331         US Mail (1st Class)           24846         UNDER THE COUNTER, INC., 5023 STATE ROAD, PO BOX 321, ASHTABULA, OH, 44005-0321         US Mail (1st Class)           24846         UNILOY MILACRON INC., S550 OCCIDENTAL HWY, SUITE B, TECUMSEH, MI, 49286         US Mail (1st Class)           24846         UNILOY MILACRON, INC., MILACRON MARKETING, PO BOX 77199, DETROIT, MI, 48277-0199         US Mail (1st Class)           24846         UNILOY MILICRON, 4165 HALFACRE RD., BATAVIA, OH, 45103         US Mail (1st Class)           24846         UNION COUNTY TREASURER, 233 w 6TH ST, MARYSVILLE, OH, 43040-5513         US Mail (1st Class)           24846         UNION COUNTY TREASURER, 233 w	24846	UHHS GEAUGA REGIONAL, 13207 RAVENNA RD, CHARDON, OH, 44024-7032	US Mail (1st Class)
24846         ULTRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429         US Mail (1st Class)           24846         ULTRA MAINTENANCE CO., PO BOX 3311, ERIE, PA, 16508         US Mail (1st Class)           24846         ULTRACOM, INC., 25860 LAHSER RD, SOUTHFIELD, MI, 48034         US Mail (1st Class)           24846         ULTRAMATION INC., PO DRAWER 20428, WACO, TX, 76702-0428         US Mail (1st Class)           24846         ULTRA-TECH AUTOMATION,INC, 33671 DOREKA DRIVE, FRASER, MI, 48026         US Mail (1st Class)           24846         UMBRELLA DIGITAL SOLUTIONS INC, 27260 HAGGERTY RD., SUITE A-11, FARMINGTON HILLS, MI, 48331         US Mail (1st Class)           24846         UNDER THE COUNTER, INC., 5023 STATE ROAD, PO BOX 321, ASHTABULA, OH, 44005-0321         US Mail (1st Class)           24846         UNDERWRITER'S LAB., INC., 333 PFINGSTEN RD, NORTHBROOK, IL, 60062-2002         US Mail (1st Class)           24846         UNILOY MILACRON INC., MILACRON MARKETING, PO BOX 77199, DETROIT, MI, 48277-0199         US Mail (1st Class)           24846         UNILOY MILICRON, 4165 HALFACRE RD., BATAVIA, OH, 45103         US Mail (1st Class)           24846         UNION COUNTY TREASURER, 233 W 6TH ST, MARYSVILLE, OH, 43040-5513         US Mail (1st Class)           24846         UNION COUNTY TREASURER, TAMARA LOWE, 233 W SIXTH ST, MARYSVILLE, OH, 43040-0420         US Mail (1st Class)           24846         UNIQUE FABRICATING,	24846	U-LINE, 2200 S LAKESIDE DRIVE, WAUKEGAN, IL, 60085	US Mail (1st Class)
24846         ULTRA MAINTENANCE CO., PO BOX 3311, ERIE, PA, 16508         US Mail (1st Class)           24846         ULTRACOM, INC., 25860 LAHSER RD, SOUTHFIELD, MI, 48034         US Mail (1st Class)           24846         ULTRAMATION INC., PO DRAWER 20428, WACO, TX, 76702-0428         US Mail (1st Class)           24846         ULTRA-TECH AUTOMATION,INC, 33671 DOREKA DRIVE, FRASER, MI, 48026         US Mail (1st Class)           24846         UMDRELLA DIGITAL SOLUTIONS INC, 27260 HAGGERTY RD., SUITE A-11, FARMINGTON HILLS, MI, 48331         US Mail (1st Class)           24846         UNDER THE COUNTER, INC., 5023 STATE ROAD, PO BOX 321, ASHTABULA, OH, 44005-0321         US Mail (1st Class)           24846         UNDERWRITER'S LAB., INC., 333 PFINGSTEN RD, NORTHBROOK, IL., 60062-2002         US Mail (1st Class)           24846         UNILOY MILACRON INC., 5550 OCCIDENTAL HWY, SUITE B, TECUMSEH, MI, 49286         US Mail (1st Class)           24846         UNILOY MILACRON, INC., MILACRON MARKETING, PO BOX 77199, DETROIT, MI, 48277-0199         US Mail (1st Class)           24846         UNILOY MILICRON, 4165 HALFACRE RD., BATAVIA, OH, 45103         US Mail (1st Class)           24846         UNION COUNTY TREASURER, 233 W 6TH ST, MARYSVILLE, OH, 43040-5513         US Mail (1st Class)           24846         UNION COUNTY TREASURER, TAMARA LOWE, 233 W SIXTH ST, MARYSVILLE, OH, 43040-0420         US Mail (1st Class)           24846         UNIQUE FABRICATING, INC	24846	ULTIMATE OFFICE, INC., 317 FAIRFIELD RD, FREEHOLD, NJ, 07728-7829	US Mail (1st Class)
24846         ULTRACOM, INC., 25860 LAHSER RD, SOUTHFIELD, MI, 48034         US Mail (1st Class)           24846         ULTRAMATION INC., PO DRAWER 20428, WACO, TX, 76702-0428         US Mail (1st Class)           24846         ULTRA-TECH AUTOMATION,INC, 33671 DOREKA DRIVE, FRASER, MI, 48026         US Mail (1st Class)           24846         UMBRELLA DIGITAL SOLUTIONS INC, 27260 HAGGERTY RD., SUITE A-11, FARMINGTON HILLS, MI, 48331         US Mail (1st Class)           24846         UNDER THE COUNTER, INC., 5023 STATE ROAD, PO BOX 321, ASHTABULA, OH, 44005-0321         US Mail (1st Class)           24846         UNIDERWRITER'S LAB., INC., 333 PFINGSTEN RD, NORTHBROOK, IL, 60062-2002         US Mail (1st Class)           24846         UNILOY MILACRON INC., 5550 OCCIDENTAL HWY, SUITE B, TECUMSEH, MI, 49286         US Mail (1st Class)           24846         UNILOY MILACRON, INC., MILACRON MARKETING, PO BOX 77199, DETROIT, MI, 48277-0199         US Mail (1st Class)           24846         UNILOY MILICRON, 4165 HALFACRE RD., BATAVIA, OH, 45103         US Mail (1st Class)           24846         UNION COUNTY TREASURER, 233 W 6TH ST, MARYSVILLE, OH, 43040-5513         US Mail (1st Class)           24846         UNIPAK LOGISTICS, LLC, ATTN: CANDACE MILLER, 3980 STATE RD. 38 EAST, LAFAYETTE, IN, 47905         US Mail (1st Class)           24846         UNIQUE FABRICATING, INC., DRAWER #641431, PO BOX 64000, DETROIT, MI, 48264-1431         US Mail (1st Class)           24846 <th>24846</th> <th>ULTRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429</th> <th>US Mail (1st Class)</th>	24846	ULTRA FAB, INC., CHURCH STREET STATION, PO BOX 6429, NEW YORK, NY, 10249-6429	US Mail (1st Class)
24846         ULTRAMATION INC., PO DRAWER 20428, WACO, TX, 76702-0428         US Mail (1st Class)           24846         ULTRA-TECH AUTOMATION,INC, 33671 DOREKA DRIVE, FRASER, MI, 48026         US Mail (1st Class)           24846         UMBRELLA DIGITAL SOLUTIONS INC, 27260 HAGGERTY RD., SUITE A-11, FARMINGTON HILLS, MI, 48331         US Mail (1st Class)           24846         UNDER THE COUNTER, INC., 5023 STATE ROAD, PO BOX 321, ASHTABULA, OH, 44005-0321         US Mail (1st Class)           24846         UNDERWRITER'S LAB., INC., 333 PFINGSTEN RD, NORTHBROOK, IL, 60062-2002         US Mail (1st Class)           24846         UNILOY MILACRON INC., 5550 OCCIDENTAL HWY, SUITE B, TECUMSEH, MI, 49286         US Mail (1st Class)           24846         UNILOY MILACRON, INC., MILACRON MARKETING, PO BOX 77199, DETROIT, MI, 48277-0199         US Mail (1st Class)           24846         UNILOY MILICRON, 4165 HALFACRE RD., BATAVIA, OH, 45103         US Mail (1st Class)           24846         UNION COUNTY TREASURER, 233 W 6TH ST, MARYSVILLE, OH, 43040-5513         US Mail (1st Class)           24846         UNIPAK LOGISTICS, LLC, ATTN: CANDACE MILLER, 3980 STATE RD. 38 EAST, LAFAYETTE, IN, 47905         US Mail (1st Class)           24846         UNIQUE FABRICATING, INC, DRAWER #641431, PO BOX 64000, DETROIT, MI, 48264-1431         US Mail (1st Class)           24846         UNIGUE TOOL & GAUGE INC., 1505 MORO DRIVE, RR 1, WINDSOR, ON, N9A 6J3CANADA         US Mail (1st Class)	24846	ULTRA MAINTENANCE CO., PO BOX 3311, ERIE, PA, 16508	US Mail (1st Class)
24846         ULTRA-TECH AUTOMATION,INC, 33671 DOREKA DRIVE, FRASER, MI, 48026         US Mail (1st Class)           24846         UMBRELLA DIGITAL SOLUTIONS INC, 27260 HAGGERTY RD., SUITE A-11, FARMINGTON HILLS, MI, 48331         US Mail (1st Class)           24846         UNDER THE COUNTER, INC., 5023 STATE ROAD, PO BOX 321, ASHTABULA, OH, 44005-0321         US Mail (1st Class)           24846         UNDERWRITER'S LAB., INC., 333 PFINGSTEN RD, NORTHBROOK, IL, 60062-2002         US Mail (1st Class)           24846         UNILOY MILACRON INC., 5550 OCCIDENTAL HWY, SUITE B, TECUMSEH, MI, 49286         US Mail (1st Class)           24846         UNILOY MILACRON, INC., MILACRON MARKETING, PO BOX 77199, DETROIT, MI, 48277-0199         US Mail (1st Class)           24846         UNILOY MILICRON, 4165 HALFACRE RD., BATAVIA, OH, 45103         US Mail (1st Class)           24846         UNION COUNTY TREASURER, 233 W 6TH ST, MARRYSVILLE, OH, 43040-5513         US Mail (1st Class)           24846         UNION COUNTY TREASURER, TAMARA LOWE, 233 W SIXTH ST, MARRYSVILLE, OH, 43040-0420         US Mail (1st Class)           24846         UNIQUE FABRICATING, INC, DRAWER #641431, PO BOX 64000, DETROIT, MI, 48264-1431         US Mail (1st Class)           24846         UNIQUE TOOL & GAUGE INC., 1505 MORO DRIVE, RR 1, WINDSOR, ON, N9A 6J3CANADA         US Mail (1st Class)           24846         UNISCAL, 1NC., 1014 UHLHORN ST, EVANSVILLE, IN, 47710-2734         US Mail (1st Class)	24846	ULTRACOM, INC., 25860 LAHSER RD, SOUTHFIELD, MI, 48034	US Mail (1st Class)
24846         UMBRELLA DIGITAL SOLUTIONS INC, 27260 HAGGERTY RD., SUITE A-11, FARMINGTON HILLS, MI, 48331         US Mail (1st Class)           24846         UNDER THE COUNTER, INC., 5023 STATE ROAD, PO BOX 321, ASHTABULA, OH, 44005-0321         US Mail (1st Class)           24846         UNDERWRITER'S LAB., INC., 333 PFINGSTEN RD, NORTHBROOK, IL, 60062-2002         US Mail (1st Class)           24846         UNILOY MILACRON INC., 5550 OCCIDENTAL HWY, SUITE B, TECUMSEH, MI, 49286         US Mail (1st Class)           24846         UNILOY MILACRON, INC., MILACRON MARKETING, PO BOX 77199, DETROIT, MI, 48277-0199         US Mail (1st Class)           24846         UNILOY MILICRON, 4165 HALFACRE RD., BATAVIA, OH, 45103         US Mail (1st Class)           24846         UNION COUNTY TREASURER, 233 W 6TH ST, MARYSVILLE, OH, 43040-5513         US Mail (1st Class)           24846         UNION COUNTY TREASURER, TAMARA LOWE, 233 W SIXTH ST, MARYSVILLE, OH, 43040-0420         US Mail (1st Class)           24846         UNIQUE FABRICATING, INC, DRAWER #641431, PO BOX 64000, DETROIT, MI, 48264-1431         US Mail (1st Class)           24846         UNIQUE TOOL & GAUGE INC., 1505 MORO DRIVE, RR 1, WINDSOR, ON, N9A 6J3CANADA         US Mail (1st Class)           24846         UNISCAL, INC., 1014 UHLHORN ST, EVANSVILLE, IN, 47710-2734         US Mail (1st Class)           24846         UNISCAL, INC., 1014 UHLHORN ST, EVANSVILLE, IN, 47710-2734         US Mail (1st Class)           24	24846	ULTRAMATION INC., PO DRAWER 20428, WACO, TX, 76702-0428	US Mail (1st Class)
48331  24846 UNDER THE COUNTER, INC., 5023 STATE ROAD, PO BOX 321, ASHTABULA, OH, 44005-0321  US Mail (1st Class)  24846 UNDERWRITER'S LAB., INC., 333 PFINGSTEN RD, NORTHBROOK, IL, 60062-2002  US Mail (1st Class)  24846 UNILOY MILACRON INC., 5550 OCCIDENTAL HWY, SUITE B, TECUMSEH, MI, 49286  US Mail (1st Class)  24846 UNILOY MILACRON, INC., MILACRON MARKETING, PO BOX 77199, DETROIT, MI, 48277-0199  US Mail (1st Class)  24846 UNILOY MILICRON, 4165 HALFACRE RD., BATAVIA, OH, 45103  US Mail (1st Class)  24846 UNION COUNTY TREASURER, 233 W 6TH ST, MARYSVILLE, OH, 43040-5513  US Mail (1st Class)  24846 UNIPAK LOGISTICS, LLC, ATTN: CANDACE MILLER, 3980 STATE RD. 38 EAST, LAFAYETTE, IN, 47905  24846 UNIQUE FABRICATING, INC, DRAWER #641431, PO BOX 64000, DETROIT, MI, 48264-1431  US Mail (1st Class)  24846 UNIQUE TOOL & GAUGE INC., 1505 MORO DRIVE, RR 1, WINDSOR, ON, N9A 6J3CANADA  US Mail (1st Class)  24846 UNISEAL, INC., 1014 UHLHORN ST, EVANSVILLE, IN, 47710-2734  UNISOURCE, 7472 COLLECTIONS CENTER DRIVE, CHICAGO, IL, 60693  UNITEC INDUSTRIES, INC, 20100 CORNILLIE RD, ROSEVILLE, MI, 48066  US Mail (1st Class)	24846	ULTRA-TECH AUTOMATION,INC, 33671 DOREKA DRIVE, FRASER, MI, 48026	US Mail (1st Class)
24846         UNDERWRITER'S LAB., INC., 333 PFINGSTEN RD, NORTHBROOK, IL, 60062-2002         US Mail (1st Class)           24846         UNILOY MILACRON INC., 5550 OCCIDENTAL HWY, SUITE B, TECUMSEH, MI, 49286         US Mail (1st Class)           24846         UNILOY MILACRON, INC., MILACRON MARKETING, PO BOX 77199, DETROIT, MI, 48277-0199         US Mail (1st Class)           24846         UNILOY MILICRON, 4165 HALFACRE RD., BATAVIA, OH, 45103         US Mail (1st Class)           24846         UNION COUNTY TREASURER, 233 W 6TH ST, MARYSVILLE, OH, 43040-5513         US Mail (1st Class)           24846         UNIPAK LOGISTICS, LLC, ATTN: CANDACE MILLER, 3980 STATE RD. 38 EAST, LAFAYETTE, IN, 47905         US Mail (1st Class)           24846         UNIQUE FABRICATING, INC, DRAWER #641431, PO BOX 64000, DETROIT, MI, 48264-1431         US Mail (1st Class)           24846         UNIQUE TOOL & GAUGE INC., 1505 MORO DRIVE, RR 1, WINDSOR, ON, N9A 6J3CANADA         US Mail (1st Class)           24846         UNIROYAL CHEMICAL COMPANY, POBOX 2049, CAROL STREAM, IL, 60132         US Mail (1st Class)           24846         UNISCAL, INC., 1014 UHLHORN ST, EVANSVILLE, IN, 47710-2734         US Mail (1st Class)           24846         UNITED INDUSTRIES, INC, 20100 CORNILLIE RD, ROSEVILLE, MI, 48066         US Mail (1st Class)           24846         UNITED BANK, PUBLIC SQUARE, C/O JEREMIAH&EMILY WOODMANSEE, GALION, OH, 44833         US Mail (1st Class)           24846	24846		US Mail (1st Class)
24846         UNILOY MILACRON INC., 5550 OCCIDENTAL HWY, SUITE B, TECUMSEH, MI, 49286         US Mail (1st Class)           24846         UNILOY MILACRON, INC., MILACRON MARKETING, PO BOX 77199, DETROIT, MI, 48277-0199         US Mail (1st Class)           24846         UNILOY MILICRON, 4165 HALFACRE RD., BATAVIA, OH, 45103         US Mail (1st Class)           24846         UNION COUNTY TREASURER, 233 W 6TH ST, MARYSVILLE, OH, 43040-5513         US Mail (1st Class)           24846         UNION COUNTY TREASURER, TAMARA LOWE, 233 W SIXTH ST, MARYSVILLE, OH, 43040-0420         US Mail (1st Class)           24846         UNIPAK LOGISTICS, LLC, ATTN: CANDACE MILLER, 3980 STATE RD. 38 EAST, LAFAYETTE, IN, 47905         US Mail (1st Class)           24846         UNIQUE FABRICATING, INC, DRAWER #641431, PO BOX 64000, DETROIT, MI, 48264-1431         US Mail (1st Class)           24846         UNIQUE TOOL & GAUGE INC., 1505 MORO DRIVE, RR 1, WINDSOR, ON, N9A 6J3CANADA         US Mail (1st Class)           24846         UNIROYAL CHEMICAL COMPANY, POBOX 2049, CAROL STREAM, IL, 60132         US Mail (1st Class)           24846         UNISEAL, INC., 1014 UHLHORN ST, EVANSVILLE, IN, 47710-2734         US Mail (1st Class)           24846         UNITEC INDUSTRIES, INC, 20100 CORNILLIE RD, ROSEVILLE, MI, 48066         US Mail (1st Class)           24846         UNITED BANK, PUBLIC SQUARE, C/O JEREMIAH&EMILY WOODMANSEE, GALION, OH, 44833         US Mail (1st Class)           24846 <th>24846</th> <th>UNDER THE COUNTER, INC., 5023 STATE ROAD, PO BOX 321, ASHTABULA, OH, 44005-0321</th> <th>US Mail (1st Class)</th>	24846	UNDER THE COUNTER, INC., 5023 STATE ROAD, PO BOX 321, ASHTABULA, OH, 44005-0321	US Mail (1st Class)
UNILOY MILACRON, INC., MILACRON MARKETING, PO BOX 77199, DETROIT, MI, 48277-0199  US Mail (1st Class)	24846	UNDERWRITER`S LAB., INC., 333 PFINGSTEN RD, NORTHBROOK, IL, 60062-2002	US Mail (1st Class)
UNILOY MILICRON, 4165 HALFACRE RD., BATAVIA, OH, 45103  US Mail (1st Class)  UNIQUE FABRICATING, INC, DRAWER #641431, PO BOX 64000, DETROIT, MI, 48264-1431  UNIQUE TOOL & GAUGE INC., 1505 MORO DRIVE, RR 1, WINDSOR, ON, N9A 6J3CANADA  US Mail (1st Class)	24846	UNILOY MILACRON INC., 5550 OCCIDENTAL HWY, SUITE B, TECUMSEH, MI, 49286	US Mail (1st Class)
UNION COUNTY TREASURER, 233 W 6TH ST, MARYSVILLE, OH, 43040-5513  US Mail (1st Class)  UNION COUNTY TREASURER, TAMARA LOWE, 233 W SIXTH ST, MARYSVILLE, OH, 43040-0420  US Mail (1st Class)  UNIPAK LOGISTICS, LLC, ATTN: CANDACE MILLER, 3980 STATE RD. 38 EAST, LAFAYETTE, IN, 47905  UNIQUE FABRICATING, INC, DRAWER #641431, PO BOX 64000, DETROIT, MI, 48264-1431  UNIQUE TOOL & GAUGE INC., 1505 MORO DRIVE, RR 1, WINDSOR, ON, N9A 6J3CANADA  US Mail (1st Class)	24846	UNILOY MILACRON, INC., MILACRON MARKETING, PO BOX 77199, DETROIT, MI, 48277-0199	US Mail (1st Class)
UNION COUNTY TREASURER, TAMARA LOWE, 233 W SIXTH ST, MARYSVILLE, OH, 43040-0420 US Mail (1st Class) UNIPAK LOGISTICS, LLC, ATTN: CANDACE MILLER, 3980 STATE RD. 38 EAST, LAFAYETTE, IN, 47905 US Mail (1st Class) UNIQUE FABRICATING, INC, DRAWER #641431, PO BOX 64000, DETROIT, MI, 48264-1431 US Mail (1st Class) UNIQUE TOOL & GAUGE INC., 1505 MORO DRIVE, RR 1, WINDSOR, ON, N9A 6J3CANADA US Mail (1st Class) UNIROYAL CHEMICAL COMPANY, POBOX 2049, CAROL STREAM, IL, 60132 US Mail (1st Class) UNISEAL, INC., 1014 UHLHORN ST, EVANSVILLE, IN, 47710-2734 UNISOURCE, 7472 COLLECTIONS CENTER DRIVE, CHICAGO, IL, 60693 UNITEC INDUSTRIES, INC, 20100 CORNILLIE RD, ROSEVILLE, MI, 48066 UNITED BANK, PUBLIC SQUARE, C/O JEREMIAH&EMILY WOODMANSEE, GALION, OH, 44833 US Mail (1st Class)	24846	UNILOY MILICRON, 4165 HALFACRE RD., BATAVIA, OH, 45103	US Mail (1st Class)
UNIPAK LOGISTICS, LLC, ATTN: CANDACE MILLER, 3980 STATE RD. 38 EAST, LAFAYETTE, IN, 47905  US Mail (1st Class)  UNIQUE FABRICATING, INC, DRAWER #641431, PO BOX 64000, DETROIT, MI, 48264-1431  US Mail (1st Class)	24846	UNION COUNTY TREASURER, 233 W 6TH ST, MARYSVILLE, OH, 43040-5513	US Mail (1st Class)
UNIQUE FABRICATING, INC, DRAWER #641431, PO BOX 64000, DETROIT, MI, 48264-1431  US Mail (1st Class)  UNIQUE TOOL & GAUGE INC., 1505 MORO DRIVE, RR 1, WINDSOR, ON, N9A 6J3CANADA  US Mail (1st Class)  UNIROYAL CHEMICAL COMPANY, POBOX 2049, CAROL STREAM, IL, 60132  US Mail (1st Class)  UNISEAL, INC., 1014 UHLHORN ST, EVANSVILLE, IN, 47710-2734  UNISOURCE, 7472 COLLECTIONS CENTER DRIVE, CHICAGO, IL, 60693  UNITEC INDUSTRIES, INC, 20100 CORNILLIE RD, ROSEVILLE, MI, 48066  UNITED BANK, PUBLIC SQUARE, C/O JEREMIAH&EMILY WOODMANSEE, GALION, OH, 44833  UNITED COMMUNICATIONS GROUP, 11300 ROCKVILLE PIKE STE 1100, ROCKVILLE, MD, 20897-5210  US Mail (1st Class)	24846	UNION COUNTY TREASURER, TAMARA LOWE, 233 W SIXTH ST, MARYSVILLE, OH, 43040-0420	US Mail (1st Class)
UNIQUE TOOL & GAUGE INC., 1505 MORO DRIVE, RR 1, WINDSOR, ON, N9A 6J3CANADA  US Mail (1st Class)  UNIROYAL CHEMICAL COMPANY, POBOX 2049, CAROL STREAM, IL, 60132  US Mail (1st Class)  UNISEAL, INC., 1014 UHLHORN ST, EVANSVILLE, IN, 47710-2734  US Mail (1st Class)	24846	UNIPAK LOGISTICS, LLC, ATTN: CANDACE MILLER, 3980 STATE RD. 38 EAST, LAFAYETTE, IN, 47905	US Mail (1st Class)
UNIROYAL CHEMICAL COMPANY, POBOX 2049, CAROL STREAM, IL, 60132  US Mail (1st Class)  UNISEAL, INC., 1014 UHLHORN ST, EVANSVILLE, IN, 47710-2734  US Mail (1st Class)	24846	UNIQUE FABRICATING, INC, DRAWER #641431, PO BOX 64000, DETROIT, MI, 48264-1431	US Mail (1st Class)
UNISEAL, INC., 1014 UHLHORN ST, EVANSVILLE, IN, 47710-2734 US Mail (1st Class)	24846	UNIQUE TOOL & GAUGE INC., 1505 MORO DRIVE, RR 1, WINDSOR, ON, N9A 6J3CANADA	US Mail (1st Class)
UNISOURCE, 7472 COLLECTIONS CENTER DRIVE, CHICAGO, IL, 60693  US Mail (1st Class)  UNITEC INDUSTRIES, INC, 20100 CORNILLIE RD, ROSEVILLE, MI, 48066  US Mail (1st Class)  UNITED BANK, PUBLIC SQUARE, C/O JEREMIAH&EMILY WOODMANSEE, GALION, OH, 44833  US Mail (1st Class)  UNITED COMMUNICATIONS GROUP, 11300 ROCKVILLE PIKE STE 1100, ROCKVILLE, MD, 20897-5210  UNITED FEED SCREWS LTD, 487 WELLINGTON AVE, PO BOX 9433, AKRON, OH, 44305  US Mail (1st Class)  US Mail (1st Class)	24846	UNIROYAL CHEMICAL COMPANY, POBOX 2049, CAROL STREAM, IL, 60132	US Mail (1st Class)
24846 UNITEC INDUSTRIES, INC, 20100 CORNILLIE RD, ROSEVILLE, MI, 48066 US Mail (1st Class) 24846 UNITED BANK, PUBLIC SQUARE, C/O JEREMIAH&EMILY WOODMANSEE, GALION, OH, 44833 US Mail (1st Class) 24846 UNITED COMMUNICATIONS GROUP, 11300 ROCKVILLE PIKE STE 1100, ROCKVILLE, MD, 20897-5210 US Mail (1st Class) 24846 UNITED FEED SCREWS LTD, 487 WELLINGTON AVE, PO BOX 9433, AKRON, OH, 44305 US Mail (1st Class)	24846	UNISEAL, INC., 1014 UHLHORN ST, EVANSVILLE, IN, 47710-2734	US Mail (1st Class)
24846 UNITED BANK, PUBLIC SQUARE, C/O JEREMIAH&EMILY WOODMANSEE, GALION, OH, 44833 US Mail (1st Class) 24846 UNITED COMMUNICATIONS GROUP, 11300 ROCKVILLE PIKE STE 1100, ROCKVILLE, MD, 20897-5210 US Mail (1st Class) 24846 UNITED FEED SCREWS LTD, 487 WELLINGTON AVE, PO BOX 9433, AKRON, OH, 44305 US Mail (1st Class)	24846	UNISOURCE, 7472 COLLECTIONS CENTER DRIVE, CHICAGO, IL, 60693	US Mail (1st Class)
24846 UNITED COMMUNICATIONS GROUP, 11300 ROCKVILLE PIKE STE 1100, ROCKVILLE, MD, 20897-5210 US Mail (1st Class) 24846 UNITED FEED SCREWS LTD, 487 WELLINGTON AVE, PO BOX 9433, AKRON, OH, 44305 US Mail (1st Class)	24846	UNITEC INDUSTRIES, INC, 20100 CORNILLIE RD, ROSEVILLE, MI, 48066	US Mail (1st Class)
24846 UNITED FEED SCREWS LTD, 487 WELLINGTON AVE, PO BOX 9433, AKRON, OH, 44305 US Mail (1st Class)	24846	UNITED BANK, PUBLIC SQUARE, C/O JEREMIAH&EMILY WOODMANSEE, GALION, OH, 44833	US Mail (1st Class)
	24846	UNITED COMMUNICATIONS GROUP, 11300 ROCKVILLE PIKE STE 1100, ROCKVILLE, MD, 20897-5210	US Mail (1st Class)
24846 UNITED GLOVE, INC., 2017 N STEWART AVE, NEWTON, NC, 28658 US Mail (1st Class)	24846	UNITED FEED SCREWS LTD, 487 WELLINGTON AVE, PO BOX 9433, AKRON, OH, 44305	US Mail (1st Class)
	24846	UNITED GLOVE, INC., 2017 N STEWART AVE, NEWTON, NC, 28658	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	UNITED HEALTHCARE OF OHIO,INC., DEPT. 658, COLUMBUS, OH, 43265-0658	US Mail (1st Class)
24846	UNITED PLASTICS, INC., 1227 GARFIELD AVE, FLINT, MI, 48505-3962	US Mail (1st Class)
24846	UNITED RENTALS, 8401 EAST FWY, HOUSTON, TX, 77029-1613	US Mail (1st Class)
24846	UNITED RENTALS, 6298 PARK SOUTH DR, BESSEMER, AL, 35022-5655	US Mail (1st Class)
24846	UNITED RENTALS, 13461 HURON RIVER DRIVE, ROMULUS, MI, 48174	US Mail (1st Class)
24846	UNITED SILICONES CORP., 4471 WALDEN AVE, LANCASTER, NY, 14086-9754	US Mail (1st Class)
24846	UNITED STATES CRAYON, 4608 W. 20TH STREET, CICERO, IL, 60804	US Mail (1st Class)
24846	UNITED STATES POSTAL SERV, NORTH SELTZER ST, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	UNITED STATES POSTAL SERVICE, 2670 MARKET ST LOBBY A, PHILADELPHIA, PA, 19104-9775	US Mail (1st Class)
24846	UNITED STATES POSTAL SERVICE, CMRS-TMS, PO BOX 0527, CAROL STREAM, IL, 60132-0527	US Mail (1st Class)
24846	UNITED STATES TREASURY, INTERNAL REVENUE SERVICE, CINCINNATI, OH, 45999-0039	US Mail (1st Class)
24846	UNITED STEELWORKER'S OF AMERICA, 5 GATEWAY CTR, 60 BOULEVARD OF THE ALLIES, PITTSBURGH, PA, 15222	US Mail (1st Class)
24846	UNITED VAN LINES LLC, 22304 NETWORK PLACE, CHICAGO, IL, 60673-1223	US Mail (1st Class)
24846	UNITED VAN LINES, LLC, ONE UNITED DRIVE, FENTON, MO, 63026-1350	US Mail (1st Class)
24846	UNITED WAY, 1212 GRISWOLD ST, DETROIT, MI, 48226-1848	US Mail (1st Class)
24846	UNITED WAY, 701 N FAIRFAX ST, ALEXANDRIA, VA, 22314-2058	US Mail (1st Class)
24846	UNITED WAY, 35 PARK ST. N, SUITE 179, MANSFIELD, OH, 44902	US Mail (1st Class)
24846	UNITED WAY OF CENTRAL, STARK COUNTY, PO BOX 9161, CANTON, OH, 44711-9161	US Mail (1st Class)
24846	UNITED WAY OF THE GREATER DAYTON, 184 SALEM AVE, DAYTON, OH, 45406-5804	US Mail (1st Class)
24846	UNITED WAY OF TUSCALOOSA, 2720 6TH ST, TUSCALOOSA, AL, 35401-1731	US Mail (1st Class)
24846	UNITED WAY SERVICES, 1331 EUCLID AVE, CLEVELAND, OH, 44115-1819	US Mail (1st Class)
24846	UNIV ER SPEC. OF BEDFORD-GEA, PO BOX 631009, CINCINNATI, OH, 45263-1009	US Mail (1st Class)
24846	UNIVERSAL DATA, INC., 4150 GRANGE HALL RD., HOLLY, MI, 48442	US Mail (1st Class)
24846	UNIVERSAL DATA, INCORP., 4150 GRANGE HALL RD., HOLLY, MI, 48442	US Mail (1st Class)
24846	UNIVERSAL DYNAMICS INC., 13600 DABNEY RD., PO DRAWER X, WOODBRIDGE, VA, 22194-0396	US Mail (1st Class)
24846	UNIVERSAL ELECTRIC PROD., 840 MOTOR DR., HOWELL, MI, 48843	US Mail (1st Class)
24846	UNIVERSAL METAL PRODUCTS, BOX #73757-N, CLEVELAND, OH, 44193-1058	US Mail (1st Class)
24846	UNIVERSITY BUSINESS INSTALLATIONS, 23231 B INDUSTRIAL DR., FARMINGTON HILLS, MI, 48335	US Mail (1st Class)
24846	UNIVERSITY FAMILY PRACTICE, 904 ANNA AVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	UNIVERSITY MEDNET, 22750 ROCKSIDE RD, CLEVELAND, OH, 44146-1574	US Mail (1st Class)
24846	UNIVERSITY OF ALABAMA, 4200 MCFARLAND BLVD E, TUSCALOOSA, AL, 35405-3853	US Mail (1st Class)
24846	UNIVERSITY OF DAYTON, BURSAR, DAYTON, OH, 45469-1600	US Mail (1st Class)
24846	UNIVERSITY OUTREACH, UW-MILWAUKEE, DRAWER NO. 491, MILWAUKEE, WI, 53293-0491	US Mail (1st Class)
24846	UNLIMITED SERVICES, INC., 35960 31 MILE RD, LENOX, MI, 48050-1406	US Mail (1st Class)
24846	UNUM LIFE INSURANCE COMPANY, OF AMERICA, PO BOX 406955, ATLANTA, GA, 30384-6955	US Mail (1st Class)
24846	UPBEAT INC., 4350 DUNCAN AVE, SAINT LOUIS, MO, 63110-1110	US Mail (1st Class)
24846	UPLAND HILLS FARM, 481 LAKE GEORGE ROAD, OXFORD, MI, 48370	US Mail (1st Class)
24846	URBANIAK & ASSOCIATES, 35109 DEARING DRIVE, STERLING HEIGHTS, MI, 48312	US Mail (1st Class)
24846	US COMEX CORPORATION, 13853 PERKINS RD, BATON ROUGE, LA, 70810-3436	US Mail (1st Class)
24846	US CUSTOMS SVC, 3575 CONCORDE DR, VANDALIA, OH, 45377-3308	US Mail (1st Class)
24846	US CUSTOMS SVC, 5590 LAUBY RD, CANTON, OH, 44720-1500	US Mail (1st Class)
24846	US DEPT OF LABOR OSHA, 36 TRIANGLE PARK DR, CINCINNATI, OH, 45246	US Mail (1st Class)
24846	US FLAGPOLE, 23920 NAPIER ROAD, NOVI, MI, 48374	US Mail (1st Class)
24846	US LABOR DEPT WAGE & HOUR DIV, 2 S MAIN ST, AKRON, OH, 44308-1813	US Mail (1st Class)
24846	US OFFICE PRODUCTS, EAST, 1025 THOMAS JEFFERSON ST NW STE 600, WASHINGTON, DC, 20007-5201	US Mail (1st Class)
24846	${\tt US\ OFFICE\ PRODUCTS\ SOUTHFIELD,\ DEPARTMENT\ 77636,\ PO\ BOX\ 77000,\ DETROIT,\ MI,\ 48277-0636}$	US Mail (1st Class)
24846	US SAFETYGEAR/OHIO GLOVE, 4196 W MARKET, LEAVITTSBURG, OH, 44430	US Mail (1st Class)
24846	US TREASURY DEPT, 441 WOLF LEDGES PKWY # 403, AKRON, OH, 44311-1039	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	USA PAINTING SERVICE, 5719 WILSON, TRENTON, MI, 48183	US Mail (1st Class)
24846	USCARORA INC., 1323 NORTH MAIN STREET, MT. PLEASANT, TN, 38474	US Mail (1st Class)
24846	USF BESTWAY, 17200 N PERIMETER DR., SCOTTSDALE, AZ, 85255-5400	US Mail (1st Class)
24846	USF HOLLAND INC., 750 E 4TH ST, HOLLAND, MI, 49423	US Mail (1st Class)
24846	USF HOLLAND MOTOR EXPRESS, 750 EAST 40TH STREET, HOLLAND, MI, 49423	US Mail (1st Class)
24846	USFLOW CORP., 3300 SPRING GROVE AVE, CINCINNATI, OH, 45225-1327	US Mail (1st Class)
24846	UTILITIES PLUS ENERGY, 16864 HIGHWAY 10 NW, ELK RIVER, MN, 55330	US Mail (1st Class)
24846	UTILITY RESEARCH INC., 3363 CEDAR ROAD, FOWLERVILLE, MI, 48836	US Mail (1st Class)
24846	UW-MILWAUKEE, SCHOOL OF CONT. ED, DRAWER 491, MILWAUKEE, WI, 53293-0491	US Mail (1st Class)
24846	V.W. EIMICKE ASSOCIATES INC., 35 E GRASSY SPRAIN RD, YONKERS, NY, 10710-4620	US Mail (1st Class)
24846	VACUUM TUBES, INC., 10995-2A STATE RTE 128, HARRISON, OH, 45030	US Mail (1st Class)
24846	VALENTINE & KEBARTAS INC., 15 UNION ST, LAWRENCE, MA, 01840-1866	US Mail (1st Class)
24846	VALEO ELECTRICAL SYSTEMS INC., 3000 UNIVERSITY DR, AUBURN HILLS, MI, 48326-2356	US Mail (1st Class)
24846	VALEO SYLVANIA, LLC., 1231 A AVE N, SEYMOUR, IN, 47274	US Mail (1st Class)
24846	VALEO WIPER SYSTEMS, 3000 UNIVERSITY DR, AUBURN HILLS, MI, 48326-2356	US Mail (1st Class)
24846	VALLEY NATIONAL GASES, INC., 200 W BEAU ST, WASHINGTON, PA, 15301-4404	US Mail (1st Class)
24846	VALLEY PRODUCTS, INC., 1025 BROOKS ST SE, DECATUR, AL, 35601	US Mail (1st Class)
24846	VALLEY RUBBER MIXING, 520 S MAIN STREET, SUITE 2445, AKRON, OH, 44311-1010	US Mail (1st Class)
24846	VALLEY TOOL INC., 111 INDUSTRIAL PARK RD, WATER VALLEY, MS, 38965-4681	US Mail (1st Class)
24846	VALLEY VIEW PALLETS, 22414 HOSTETLER ROAD, DANVILLE, OH, 43014	US Mail (1st Class)
24846	VALMEC, INC., 1274 SOUTH HOLLY ROAD, FENTON, MI, 48430	US Mail (1st Class)
24846	VAN BUREN TOWNSHIP, ACCT: 004823-000, 46425 TYLER ROAD, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	VAN DORN DEMAG, DEMAG PLASTICS GROUP, 88049 EXPEDITE WAY, CHICAGO, IL, 60695-8049	US Mail (1st Class)
24846	VAN TIEM, JAMES, 3560 WEST MARKET STREET, SUITE 300, AKRON, OH, 44333	US Mail (1st Class)
24846	VANCO CORP., 1030 N MERRIFIELD AVE, MISHAWAKA, IN, 46545-6713	US Mail (1st Class)
24846	VANDALIA RENTAL, 950 ENGLE RD, VANDALIA, OH, 45377	US Mail (1st Class)
24846	VANGUARD LEASING SOLUTIONS LLC, 21719 NETWORK PLACE, CHICAGO, IL, 60673-1217	US Mail (1st Class)
24846	VANGUARD SECURITY SYSTEMS, 995 JOHN A PAPALAS, LINCOLN PARK, MI, 48146	US Mail (1st Class)
24846	VARI-FORM, INC., 233 LOTHIAN AVENUE, STRATHROY, ON, N7G 3J3CANADA	US Mail (1st Class)
24846	VAUGHAN BUILDING & GROUNDS, PO BOX 771, LAPEER, MI, 48446	US Mail (1st Class)
24846	VEHICLE RESEARCH & DEVELOPMENT, 3863 VAN DYKE ROAD, ALMONT, MI, 48003	US Mail (1st Class)
24846	VEKTEK, INC., 1334 E 6TH AVE, EMPORIA, KS, 66801-3306	US Mail (1st Class)
24846	VELCRO AUTOMOTIVE DIVISION, 406 BROWN AVE, MANCHESTER, NH, 03103-7202	US Mail (1st Class)
24846	VENANGO MACHINE COMPANY, BOX 239, WATTSBURG, PA, 16442-0239	US Mail (1st Class)
24846	VENTURE MEASUREMENT CO., LLC, 13163 COLLECTIONS CENTER DR., CHICAGO, IL, 60693	US Mail (1st Class)
24846	VENTURE TOOL & METAL, 42 EAST MAIN STREET, SEBAWING, MI, 48759	US Mail (1st Class)
24846	VERIAD, UAL 3102 PAYSPHERE CR, CHICAGO, IL, 60674	US Mail (1st Class)
24846	VERIZON DIRECTORY STORE, 140 WEST ST, NEW YORK, NY, 10007-2109	US Mail (1st Class)
24846	VESTIL MANUFACTURING CO., 2999 NORTH WAYNE ST., ANGOLA, IN, 46703	US Mail (1st Class)
24846	VIBROMATIC COMPANY, INC., 1301 S 6TH ST, NOBLESVILLE, IN, 46060-3712	US Mail (1st Class)
24846	VIC BOND SALES, INC., 1400 IMLAY CITY RD., LAPEER, MI, 48446	US Mail (1st Class)
24846	VI-CHEM CORPORATION, 55 COTTAGE GROVE SW, GRAND RAPIDS, MI, 49507	US Mail (1st Class)
24846	VICKERY TAPE & LABEL CO., PO BOX 8, 20 WEST CANAL STREET, PERU, IN, 46970	US Mail (1st Class)
24846	VICKI ANN COON, 5765 HEMINGWAY LAKE RD, OTTER LAKE, MI, 48464	US Mail (1st Class)
24846	VICKI COON, 5765 HEMINGWAY LK. RD., OTTER LAKE, MI, 48464	US Mail (1st Class)
24846	VICKIE L WILLARD, 526 SUPERIOR AVE., SUITE 935, CLEVELAND, OH, 44114	US Mail (1st Class)
24846	VICTORY TUBE COMPANY, 4521 WILLOW PKWY, CLEVELAND, OH, 44125	US Mail (1st Class)
24846	VIDON PLASTICS INC., 3171 JOHN CONLEY DR., PO BOX 56, LAPEER, MI, 48446	US Mail (1st Class)
24846	VIKAS SAINI, 3515 MASSILON RD., SUITE 320, UNIONTOWN, OH, 44685	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	VIKING OFFICE PRODUCTS, OFFICE DEPOT CORPORATE, 2200 OLD GERMANTOWN RD, DELRAY BEACH, FL, 33445-8299	US Mail (1st Class)
24846	VIKING TOOL & ENGINEERING, 2780 E. COLBY STREET, WHITEHALL, MI, 49461	US Mail (1st Class)
24846	VILLAGE PRINTING, 349 MCCORMICK DRIVE, LAPEER, MI, 48446	US Mail (1st Class)
24846	VINCENT COLBY, 1401 INDUSTRIAL PARK DRIVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	VIRTUAL SERVICES, INC., 17333 FEDERAL DR., SUITE 220, ALLEN PARK, MI, 48101	US Mail (1st Class)
24846	VISION GRAPHICS & SIGN, 560 OAK STREET, LAPEER, MI, 48446	US Mail (1st Class)
24846	VISIONSOFT LTD, 115 ROUTE 46 WEST, SUITE F1000, MOUNTAIN LAKES, NJ, 07046	US Mail (1st Class)
24846	VISTEON CORP INDIANAPOLIS, INDIANAPOLIS PLANT, 6900 E ENGLISH AVE., INDIANAPOLIS, IN, 46219	US Mail (1st Class)
24846	VISTEON CORPORATION, ATTN: VISTEON TECH CENTER-AR DEPT., 16630 SOUTHFIELD ROAD, SUITE 22B-CUBE 30, ALLEN PARK, MI, 48101	US Mail (1st Class)
24846	VISTEON CURE FOR DIABETES, ADDRESS UNAVAILABLE,	US Mail (1st Class)
24846	VISTEON/FORD/CNC LOGISTICS, 735 W 21ST STREET, CONNERSVILLE, IN, 47331	US Mail (1st Class)
24846	VISUAL PRECISION INC., 1300 BATAVIA, ROYAL OAK, MI, 48067	US Mail (1st Class)
24846	VITRAN EXPRESS, 5770 ESTE AVE, CINCINNATI, OH, 45232-1435	US Mail (1st Class)
24846	VOELKER CONTROLS, 3000 COMMERCE CENTER DR, FRANKLIN, OH, 45005-1819	US Mail (1st Class)
24846	VOIGT-ENGLAND CO. INC., 5018 1ST AVE N, BIRMINGHAM, AL, 35212-3102	US Mail (1st Class)
24846	VQ SUPPORT, 3300 HIAWATHA, EL PASO, TX, 79936	US Mail (1st Class)
24846	VWR SCIENTIFIC PRODUCTS, DIR. OF VWR SCIENTIFIC PROD, PO BOX 640169, PITTSBURGH, PA, 15264-0169	US Mail (1st Class)
24846	W H OSWALT COMPANY, 557 N HOME RD., PO BOX 1241, MANSFIELD, OH, 44901-1241	US Mail (1st Class)
24846	W. ALEXANDRIA FOOD & CLOTHING BANK, 20 E SOUTH ST, WEST ALEXANDRIA, OH, 45381-1257	US Mail (1st Class)
24846	W. K. INDUSTRIES INC., 6120 MILLETT, STERLING HEIGHTS, MI, 48312	US Mail (1st Class)
24846	W.A.HAMMOND DRIERITE CO., 138 DAYTON AVE, XENIA, OH, 45385-2830	US Mail (1st Class)
24846	W.D. SUPPLY, 836 DEPOT STREET, CINCINNATI, OH, 45204	US Mail (1st Class)
24846	W.E. BASSETT COMPANY, 100 TRAP FALLS ROAD EXT, SHELTON, CT, 06484-4655	US Mail (1st Class)
24846	W.M. WILSON CORP., 19571 PROGRESS DRIVE, CLEVELAND, OH, 44136	US Mail (1st Class)
24846	W.N. SHAW CO. INC., 853 PARK LAMAR DR, VILLA RIDGE, MO, 63089-2547	US Mail (1st Class)
24846	W.S. SERVICES, 10179 COLONIAL INDUSTRIAL, SOUTH LYON, MI, 48178	US Mail (1st Class)
24846	W.T. LIVINGSTON TIRE COMPANY, 15150 KINSMAN ROAD, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
	W.W. GRAINGER, 100 GRAINGER PKWY., LAKE FOREST, IL, 60045	US Mail (1st Class)
24846	W.W. GRAINGER, INC., DEPT 825120991, PALATINE, IL, 60038-0001	US Mail (1st Class)
24846	W.W. SCHAUB ELECTRIC CO., 501 APPLEGROVE ROAD NW, NORTH CANTON, OH, 44720	US Mail (1st Class)
24846	WA BASEBALL ORGANIZATION, PO BOX 196, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24846	WACHOVIA - NJ/NY/PA, COMMERCIAL BILLING DEPARTMENT, PO BOX 601013, CHARLOTTE, NC, 28260	US Mail (1st Class)
24846	WACHOVIA CAPITAL FINANCE CORPORATION (CENTRAL), 150 SOUTH WACKER DRIVE, SUITE 2200, CHICAGO, IL, 60606	US Mail (1st Class)
24846	WACHS ELECTRIC SERVICE INC., PO BOX 685, 15550 MAIN MARKET RD., PARKMAN, OH, 44080	US Mail (1st Class)
24846	WACKER SILICONES CORP., 3301 SUTTON RD, ADRIAN, MI, 49221	US Mail (1st Class)
24846	WAGONER STORES INC., 324 UNION BLVD., ENGLEWOOD, OH, 45322	US Mail (1st Class)
24846	WAINWRIGHT INDUSTRIES, 17 CERMAK BLVD, SAINT PETERS, MO, 63376-1019	US Mail (1st Class)
24846	WALBRO, INC., 2988 ANGLE ROAD, ORCHARD PARK, NY, 14127	US Mail (1st Class)
24846	WALCO CORPORATION, 1651 EAST SUTTER RD, PO BOX 9, GLENSHAW, PA, 15116	US Mail (1st Class)
24846	WALDAMEER PARK, 220 PENINSULA DR, ERIE, PA, 16505-2036	US Mail (1st Class)
24846	WALKER PATTERN & MOLD INC., 14485 HWY 43 NORTH, NORTHPORT, AL, 35475	US Mail (1st Class)
24846	WALL STREET JOURNAL, 200 LIBERTY ST, NEW YORK, NY, 10281-1003	US Mail (1st Class)
24846	WALLACE, PO BOX 905046, CHARLOTTE, NC, 28290-5046	US Mail (1st Class)
24846	WALLSTABE & SCHNEIDER GMBH, DICHTUNGSTECHNIK, POSTFACH 1163, BOGEN, 94322GERMANY	US Mail (1st Class)
24846	WALMART, 359 LEXINGTON SPRINGMILL, MANSFIELD, OH, 44906	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	WALMART, 555 E GENESEE STREET, LAPEER, MI, 48446	US Mail (1st Class)
24846	WAL-MART STORES, INC., ADDRESS UNAVAILABLE,	US Mail (1st Class)
24846	WALTON/STOUT INC., 6863 CHAPMAN ROAD, LITHONIA, GA, 30058	US Mail (1st Class)
24846	WARD DISTRIBUTION, LLC, PO BOX 1764, ASHTABULA, OH, 44005	US Mail (1st Class)
24846	WARD TRUCKING, 2ND AVE. AND 7TH ST., GREENWOOD RD., ALTOONA, PA, 16602	US Mail (1st Class)
24846	WARD'S COMMUNICATIONS, 3000 TOWN CENTER, SUITE 2750, SOUTHFIELD, MI, 48075	US Mail (1st Class)
24846	WAREHOUSE EQUIPMENT AND, 116 WEST PARK DRIVE, BIRMINGHAM, AL, 35211-4469	US Mail (1st Class)
24846	WAREHOUSE RACK & SHELF, 1 BURGUNDY DRIVE, LAKE ST. LOUIS, MO, 63367	US Mail (1st Class)
24846	WARREN KNIPPLE, 3560 W. MARKET ST., STE 340, AKRON, OH, 44333	US Mail (1st Class)
24846	WARREN MILL SUPPLY, 2 GEORGE ST, NEW CASTLE, PA, 16101-3952	US Mail (1st Class)
24846	WARREN MUNICIPAL COURT, 141 SOUTH ST SE, WARREN, OH, 44483-5747	US Mail (1st Class)
24846	WARREN PIPE & SUPPLY CO., 18660 15 MILE ROAD, FRASER, MI, 48026	US Mail (1st Class)
24846	WARREN THOMAS, 3217 GREEN GROVE DR NE, TUSCALOOSA, AL, 35404	US Mail (1st Class)
24846	WARRIOR PRECISION, INC., 4514 REICHOLD RD, TUSCALOOSA, AL, 35404	US Mail (1st Class)
24846	WASHINGTON PENN PLASTIC CO., 2080 N MAIN ST, WASHINGTON, PA, 15301	US Mail (1st Class)
24846	WASHINGTON PENN PLASTICS, 2080 N MAIN ST, WASHINGTON, PA, 15301-6146	US Mail (1st Class)
24846	WASHINGTON'S CARPET ONE, 1790 ASHLAND RD., PO BOX 2253, MANSFIELD, OH, 44905	US Mail (1st Class)
24846	WASTE MANAGEMENT, WASTE MANAGEMENT RMC, 2421 W PEORIA AVE, PHOENIX, AZ, 85029	US Mail (1st Class)
24846	WATERLOO TRANSMISSION, 3603 CLEVELAND AVE., CANTON, OH, 44709	US Mail (1st Class)
24846	WATSON, DEGRAFFENRIED, HARDIN & TYRA, LLP, 1651 MEFARLAND BLVD. N., TUSCALOOSA, AL, 35406-2212	US Mail (1st Class)
24846	WAYNE DALTON CORPORATION, 1 DOOR DRIVE, MT HOPE, OH, 44660	US Mail (1st Class)
24846	WAYNE WALKER, 5963 N FOX STREET, MAYVILLE, MI, 48744	US Mail (1st Class)
24846	WCAWA, C/O BRENDA MILES, UOC 305 BRYANT DR. E, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	WEAR TECHNOLOGY, 1672 SOLUTIONS CENTER, CHICAGO, IL, 60677-1006	US Mail (1st Class)
24846	WEBB/MASON, 10830 GILROY ROAD, HUNT VALLEY, MD, 21031-1301	US Mail (1st Class)
24846	WEBB/MASON, 10830 GILROY RD., HUNT VALLEY, MD, 21031-1301	US Mail (1st Class)
24846	WEBER MARKING SYSTEMS, 711 W ALGONQUIN RD, ARLINGTON HEIGHTS, IL, 60005	US Mail (1st Class)
24846	WEIDIGS FLORAL, 200 CENTER ST., CHARDON, OH, 44024	US Mail (1st Class)
24846	WELCH PACKAGING, 2555 BREST ROAD, TAYLOR, MI, 48180	US Mail (1st Class)
24846	WELCH PACKAGING DETROIT, 25555 BREST, TAYLOR, MI, 48180	US Mail (1st Class)
24846	WELLMAN INC., DEPT. 952138, ATLANTA, GA, 31192-2138	US Mail (1st Class)
24846	WELLNESS COUNCIL, STEPHEN MUSGRAVE, 433 S STATE STREET, PAINESVILLE, OH, 44077	US Mail (1st Class)
24846	WELSH MACHINERY SALES, INC., 7471-P TYLER BLVD., MENTOR, OH, 44060-5413	US Mail (1st Class)
24846	WENDY A. PAYNE, PO BOX 2, OAKMAN, AL, 35579	US Mail (1st Class)
24846	WESBELL DEDICATED, ASSEMBLY LTD, PO BOX 12833, RESEARCH TRIANGLE, NC, 27709	US Mail (1st Class)
24846	WESLEY INTERNATIONAL, 3680 CHESTNUT ST, SCOTTDALE, GA, 30079-1206	US Mail (1st Class)
24846	WEST ALABAMA, CHAMBER OF COMMERCE, TUSCALOOSA, AL, 35402	US Mail (1st Class)
24846	WEST ALABAMA CHAMBER OF COMMERCE, 2200 UNIVERSITY BLVD, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24846	WEST GATE FORD, 3680 LEHARPS DR, YOUNGSTOWN, OH, 44515-1436	US Mail (1st Class)
24846	WEST GROUP, PO BOX 64779, ST. PAUL, MN, 55164-0779	US Mail (1st Class)
24846	WEST OIL COMPANY, 435 BANKHEAD HWY, WINFIELD, AL, 35594-5417	US Mail (1st Class)
24846	WEST SHORE SIGNS, 12455 HALE STRET, RIVERVIEW, MI, 48192	US Mail (1st Class)
24846	WESTERN CONSOLIDATED, TECHNOLOGIES, INC., 135 S LASALLE STREET, CHICAGO, IL, 60674-2957	US Mail (1st Class)
24846	WESTERN CONSOLIDATED TECH., 135 S LASALLE STREET, DEPT. 2957, CHICAGO, IL, 60674-2957	US Mail (1st Class)
24846	WESTERN PATHOLOGY CONSULTANTS, 1321 BROADWAY, PO BOX 1936, SCOTTSBLUFF, NE, 69363-1936	US Mail (1st Class)
24846	WESTERN RESERVE FARM, COOPERATIVE, PO BOX 365, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	WESTERN RESERVE FARM COOPERATIVE, PO BOX 365, MIDDLEFIELD, OH, 44062	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	WESTERN RESERVE OFFICE SUPPLY, 11993 RAVENNA RD., STE 2, CHARDON, OH, 44024	US Mail (1st Class)
24846	WESTERN RUBBER COMPANY, 135 SOUTH LASALLE, DEPT. 1568, CHICAGO, IL, 60603	US Mail (1st Class)
24846	WESTERN SKY INDUSTRIES, 21300 CLOUD WAY, HAYWARD, CA, 94545	US Mail (1st Class)
24846	WESTHOFF, 9462 WATSON INDUSTRIAL PARK, ST. LOUIS, MO, 63126	US Mail (1st Class)
24846	WESTLAKE TOOL, DIE & MFG., 1280 MOORE RD., AVON, OH, 44011	US Mail (1st Class)
24846	WESTLAND FIRE EXTINGUISHER INC, 28754 PENNSYLVANIA, ROMULUS, MI, 48174	US Mail (1st Class)
24846	WESTON HURD, LLP, 1301 EAST 9TH ST, SUITE 1900, CLEVELAND, OH, 44114-1862	US Mail (1st Class)
24846	WETZEL INC., 5001 ENTERPRISE DR. NW, WARREN, OH, 44481	US Mail (1st Class)
24846	WHELCO INDUSTRIAL, LTD., 1605 INDIANWOOD CIRCLE, SUITE 200, MAUMEE, OH, 43537	US Mail (1st Class)
24846	WHIPPLE ENTERPRISES INC., 21150 SHAKE RIDGE RD., VOLCANO, CA, 95689	US Mail (1st Class)
24846	WHITESELL CANADA, 590 BASALTIC RD, VAUGHAN, ON, L4K 5A2CANADA	US Mail (1st Class)
24846	WHITESELL OF MICHIGAN INC., WACHOVIA BANK/WHITESELL CORPORATION, PO BOX 934536, ATLANTA, GA, 31193-4536	US Mail (1st Class)
24846	WHITMAN MOLD, INC., 4336 BAMBURG RD, COPE, SC, 29038	US Mail (1st Class)
24846	WI SCTF, BOX 74400, MILWAUKEE, WI, 53274-0400	US Mail (1st Class)
24846	WICKED WELDING, INC., 5665 WEBSTER ST., DAYTON, OH, 45414-3515	US Mail (1st Class)
24846	WICKES LUMBER, 2275 STUMBO ROAD, MANSFIELD, OH, 44906	US Mail (1st Class)
24846	WIDE AREA MEDIA, LLC, 19915 LAKE ROAD, SUITE C, ROCKY RIVER, OH, 44116	US Mail (1st Class)
24846	WIEMANN, GERARD, 109 COUNTRYSIDE DRIVE, CHAGRIN FALLS, OH, 44022	US Mail (1st Class)
24846	WILCOX SALES & SERVICE CO, 77 SOUTH ILLINOIS, MANSFIELD, OH, 44905	US Mail (1st Class)
24846	WILLARD HIGH SCHOOL, 123 DR. L.H. WHISLER AVENUE, PO BOX 410, WILLARD, OH, 44890	US Mail (1st Class)
24846	WILLIAM ALMASHY, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	WILLIAM COOLEY, 900 S. WILEY ST, PO BOX 320, CRESTLINE, OH, 44827	US Mail (1st Class)
24846	WILLIAM DILWORTH, 549 ST. RT. 60, GREENWICH, OH, 44837	US Mail (1st Class)
24846	WILLIAM G. HOYT, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	WILLIAM J. SNYDER, 10047 KEYSTONE DRIVE, LAKE CITY, PA, 16423	US Mail (1st Class)
24846	WILLIAM WAHL, 8707 SAMUEL BARTON DR, BELLEVILLE, MI, 48111	US Mail (1st Class)
24846	WILLIAM WRIGHT, 17187 N LAUREL PK DR, SUITE #208, LIVONIA, MI, 48152	US Mail (1st Class)
24846	WILLIAMS TOYOTA LIFT, 9462 MAIN ST SE, PO BOX 359, EAST SPARTA, OH, 44626	US Mail (1st Class)
24846	WILMA POST, PO BOX 127, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	WILMA POST, 15332 OLD STATE RD., MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24846	WILSON EQUIPMENT CO., 29561 PURITAN, LIVONIA, MI, 48154	US Mail (1st Class)
24846	WILSON'S WELDING &, 5381 SMITH-KRAMER ROAD NE, HARTVILLE, OH, 44632	US Mail (1st Class)
24846	WINANS, INC., 1460 E. HEMPHILL, BURTON, MI, 48529	US Mail (1st Class)
24846	WINDBIGLER PHOTOGRAPHY, 12300 KINSMAN RD, NEWBURY, OH, 44065	US Mail (1st Class)
24846	WINDOW PLACE, 1334 PARK AVENUE WEST, MANSFIELD, OH, 44906	US Mail (1st Class)
24846	WINDSOR MOLD, 1628 DURHAM PLACE, WINDSOR, ON, N8W 2Z8CANADA	US Mail (1st Class)
24846	WINDSOR MOLD, 4035 MALDEN ROAD, WINDSOR, ON, N9C 2G4CANADA	US Mail (1st Class)
24846	WINDWARD PETROLEUM, 1064 GOFFS FALLS RD, MANCHESTER, NH, 03103-6128	US Mail (1st Class)
24846	WINN DIXIE, 10 MCFARLAND BLVD., NORTHPORT, AL, 35476	US Mail (1st Class)
24846	WINSTON HEAT TREATING INC, 711 E 2ND ST, DAYTON, OH, 45402-1319	US Mail (1st Class)
24846	WINTERHALTER FLUID POWER INC., 6571 RIDGE RD, WADSWORTH, OH, 44281-9704	US Mail (1st Class)
24846	WINTHROP & WEINSTINE, P.A., ATTN: JUSTICE ERICSON LINDELL, 3200 MINNESOTA WORLD TRADE CENTER, 30 EAST SEVENTH STREET, ST. PAUL, MN, 55101	US Mail (1st Class)
24846	WISCO SUPPLY INC., 815 S ST VRAIN ST, PO BOX 214, EL PASO, TX, 79942-0214	US Mail (1st Class)
24846	WITT ENTERPRISES INC., POB 54, ASHTABULA, OH, 44005-0054	US Mail (1st Class)
24846	WITT MECHANICAL, INC., 28722 WALL STREET, WIXOM, MI, 48393	US Mail (1st Class)
24846	WIX FILTRATION CORP, C/O BANK ONE, DEPT 73071, CHICAGO, IL, 60673-7071	US Mail (1st Class)
24846	WMT MACHINE TOOL COMPANY, 600A HOLLISTER ROAD, TETERBORO, NJ, 07608	US Mail (1st Class)
24846	WOHRLE METALLWAREN FABRIK GMBH, POSTFACH 61, WILDBERG, D-72214GERMANY	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24846	WOLFE MARKETING CORP., PO BOX 38307, OLMSTED FALLS, OH, 44138	US Mail (1st Class)
24846	WONDERLIC, INC., 1795 N BUTTERFIELD RD., LIBERTYVILLE, IL, 60048-1238	US Mail (1st Class)
24846	WOOD LEASING INC., 3716 SOUTH ELYRIA RD., SHREVE, OH, 44676	US Mail (1st Class)
24846	WORKER'S COMPENSATION ASSOC., OF WEST ALABAMA, PO BOX 1248, TUSCALOOSA, AL, 35403	US Mail (1st Class)
24846	WORK-LOSS DATA INSTITUTE, 500 N SHORELINE BLVD, SUITE 1101 N, CORPUS CHRISTI, TX, 78471	US Mail (1st Class)
24846	WORKMAN HOME IMPROVEMENTS, 466 CENTRAL AVE., MANSFIELD, OH, 44902	US Mail (1st Class)
24846	WORLD CLASS SAFETY, 3590 CASSANDRA DR., TIPP CITY, OH, 45371	US Mail (1st Class)
24846	WORLD GLOBAL, 733 EL PARQUEETER, EL PASO, TX, 79912	US Mail (1st Class)
24846	WORLD WASTE SERVICES, 3860 NORTH VANDYLEE, ALMONT, MI, 48003	US Mail (1st Class)
24846	WORLDWIDE INDUSTRIAL CONTRACTING, 13227 DARRYL, WARREN, MI, 48088	US Mail (1st Class)
24846	WORN FURNACE CO., 7541 MINES RD SE., WARREN, OH, 44484	US Mail (1st Class)
24846	WORXZ, 3540 NORTH HURDS CORNER, SUITE L, CARO, MI, 48723-9335	US Mail (1st Class)
24846	WRIGHT POLYMER COMPANY, 12740 STONECREEK DRIVE, PICKERINGTON, OH, 43147	US Mail (1st Class)
24846	WRIGHT STATE UNIVERSITY, CENTER FOR PROFESSIONAL DEV., 140 EAST MONUMENT AVE, DAYTON, OH, 45402-1267	US Mail (1st Class)
24846	WURZBURG INC., 710 S FOURTH STREET, MEMPHIS, TN, 38101	US Mail (1st Class)
24846	WYANDOTTE ALARM SYSTEM, 1409 OAK STREET, WYANDOTTE, MI, 48192	US Mail (1st Class)
24846	WYATT SALES CORPORATION, 19645 PROGRESS DRIVE, PO BOX 360350, STRONGSVILLE, OH, 44149	US Mail (1st Class)
24846	XALOY INC., 102 XALOY WAY, PULASKI, VA, 24301	US Mail (1st Class)
24846	XEROX CORPORATION, 800 LONG RIDGE RD, STAMFORD, CT, 06902-1227	US Mail (1st Class)
24846	XEROX CORPORATION, 350 S NORTHWEST HWY, PARK RIDGE, IL, 60068	US Mail (1st Class)
24846	XEROX-OMNIFAX, 800 LONG RIDGE RD, STAMFORD, CT, 06902-1227	US Mail (1st Class)
24846	XPEDX, 6285 TRI RIDGE BLVD, LOVELAND, OH, 45140-8318	US Mail (1st Class)
24846	X-RITE, INCORPORATED, 3100 44TH ST SW, GRANDVILLE, MI, 49418-2567	US Mail (1st Class)
24846	YASUFUKU U.S.A. INC., 3201 NW 12TH ST, LINCOLN, NE, 68521-3612	US Mail (1st Class)
24846	YATES INDUSTRIES, 23050 INDUSTRIAL DR. EAST, ST. CLAIR SHORES, MI, 48080	US Mail (1st Class)
24846	YELLOW CAB LEASING, INC., 4137 MARTINDALE RD NE, CANTON, OH, 44705	US Mail (1st Class)
24846	YELLOW FREIGHT SYSTEM, WORLD HEADQUARTERS, 10990 ROE AVE, OVERLAND PARK, KS, 66211-1213	US Mail (1st Class)
24846	YELLOW TRANSPORTATION, INC (YFSY), WORLD HEADQUARTERS, 10990 ROE AVE, OVERLAND PARK, KS, 66211-1213	US Mail (1st Class)
24846	YKK (USA) INC., 2700 PATRIOT BLVD., SUITE, GLENVIEW, IL, 60025	US Mail (1st Class)
24846	YORK BUILDING MAINTENANCE, 4748 BROADVIEW BLVD., CLEVELAND, OH, 44109	US Mail (1st Class)
24846	YORK SAW & KNIFE CO, INC, 295 EMIG ROAD, YORK, PA, 17405	US Mail (1st Class)
24846	YOU AND THE LAW, PO BOX 9206, MCLEAN, VA, 22102-0206	US Mail (1st Class)
24846	YOUNGSTOWN HARD CHROME PLATING &, GRINDING, INC., 8451 SOUTHERN BLVD., PO BOX 3508, YOUNGSTOWN, OH, 44513	US Mail (1st Class)
24846	YOUNGSTOWN HEARING & SPEECH CN, 6614 SOUTHERN BLVD., BOARDMAN, OH, 44512-3455	US Mail (1st Class)
24846	YOUR FLOWER SHOP, 200 EAST MAIN STREET, EATON, OH, 45320	US Mail (1st Class)
24846	YUDO, INC., 3275 GRANDE VISTA DR., UNIT 9, NEWBURRY PARK, CA, 91320-1193	US Mail (1st Class)
24846	ZARCO ELECTRONIC SUPPLY INC., 6831 COMMERCE AVE., EL PASO, TX, 79915	US Mail (1st Class)
24846	ZATKOFF, 23230 INDUSTRIAL PARK DR, FARMINGTON HILLS, MI, 48335-2850	US Mail (1st Class)
24846	ZATKOFF SEALS & PACKINGS, 23231 INDUSTRIAL PARK DR, FARMINGTON HILLS, MI, 48335-2851	US Mail (1st Class)
24846	ZEANAH, HUST, SUMMERFORD, DAVIS & WILLIAMSON LLC, PO BOX 1310, TUSCALOOSA, AL, 35403	US Mail (1st Class)
24846	ZEE MEDICAL INC., 22 CORPORATE PARK, IRVINE, CA, 92606-3112	US Mail (1st Class)
24846	ZENITH CUTTER CO., DEPT. 77-2761, CHICAGO, IL, 60678-2761	US Mail (1st Class)
24846	ZENITH CUTTER CO., 2761 EAGLE WAY, CHICAGO, IL, 60678-1027	US Mail (1st Class)
24846	ZEP MANUFACTURING COMPANY, 13237 COLLECTIONS CTR DRIVE, CHICAGO, IL, 60693-0132	US Mail (1st Class)
24846	ZEP MFG, 13237 COLLECTIONS CTR DRIVE, CHICAGO, IL, 60693-0132	US Mail (1st Class)
24846	ZEPPE`S PIZZERIA, 15065 KINSMAN RD., MIDDLEFIELD, OH, 44062	US Mail (1st Class)

## **Exhibit B - Solicitation**

Svc Lst	Name and Address of Served Party	Mode of Service
24846	ZESCO INC., 6500 MILLER ROAD, CLEVELAND, OH, 44141	US Mail (1st Class)
24846	ZIEGLER BOLT & NUT HOUSE, 4848 CORPORATE ST SW, CANTON, OH, 44706	US Mail (1st Class)
24846	ZIPPS, LLC, 1207 EAST MAIN STREET, LURAY, VA, 22835	US Mail (1st Class)
24846	ZM ENTERPRISES, ASPHALT PAVING, 2761 KLOTZ AVE NE, CANTON, OH, 44730-1549	US Mail (1st Class)
24846	ZUMHAGEN COMPANY, LLC, 2875 GLADSTONE AVENUE, ANN ARBOR, MI, 48104-6432	US Mail (1st Class)
24846	ZUMHAGEN, CONRAD, 2875 GLADSTONE AVENUE, ANN ARBOR, MI, 48104	US Mail (1st Class)

Subtotal for this group: 4436

## **EXHIBIT C**

Svc Lst	Name and Address of Served Party	Mode of Service
24800	A & M PALLET CO, 15340 GEORGIA RD, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24800	A A JANSSON, INC., 2070 AIRPORT ROAD, WATERFORD, MI, 48327-1204	US Mail (1st Class)
24800	A SCHULMAN INC, ATTN MELISSA WOLFF, 3550 W MARKET ST, AKRON, OH, 44333	US Mail (1st Class)
24800	A T & T, ONE AT&T WAY, BEDMINSTER, NJ, 07921-2693	US Mail (1st Class)
24800	A T & T, ACCT: GEGA, ONE AT&T WAY, BEDMINSTER, NJ, 07921-2693	US Mail (1st Class)
24800	A. LOUIS SUPPLY, 5610 MAIN AVENUE, ASHTABULA, OH, 44004	US Mail (1st Class)
24800	A.P. PLASMAN INC., 24 INDUSTRIAL PARK RD., PO BOX 538, TILBURY, ON, N0P 2L0CANADA	US Mail (1st Class)
24800	A.P. PLASMAN TECUMSEH PLANT, 418 SILVER CREEK INDUSTRIAL DR, RR#1, TECUMSEH, ON, N8N 4Y3CANADA	US Mail (1st Class)
24800	ABA WATER SYSTEMS, INC., 226 WEST BROADWAY, PLAINVIEW, MN, 55964	US Mail (1st Class)
24800	ABC FIRE EQUIPMENT CO., C/O BANK OF TUSCALOOSA, PO BOX 3086, TUSCALOOSA, AL, 35403	US Mail (1st Class)
24800	ABRASIVE PRODUCTS LLC, 6330 W LOOPS, STE 900, BELAIR, TX, 77401	US Mail (1st Class)
24800	ABZ RENT ALL, 1331 MCFARLAND BLVD N.E., TUSCALOOSA, AL, 35406	US Mail (1st Class)
24800	ACCOUNTEMPS, ATTN KAREN LIMA, DIV OF ROBERT HALF INTERNATIONAL, 5720 STONERIDGE DR, STE 3, PLEASANTON, CA, 94588	US Mail (1st Class)
24800	ACCU GRIND, 4430 CRYSTAL PARKWAY, KENT, OH, 44240	US Mail (1st Class)
24800	ACCUDYN PRODUCTS, INC., 2400 YODER DR, ERIE, PA, 16506-2364	US Mail (1st Class)
24800	ACCULUBE, 403 HOMESTEAD AVE, DAYTON, OH, 45408	US Mail (1st Class)
24800	ACCU-TECH, INC., 43718 UTICA RD, STERLING HGTS., MI, 48314	US Mail (1st Class)
24800	ACE, 6599 GRANGER RD, INDEPENDENCE, OH, 44131	US Mail (1st Class)
24800	ACE CONTROLS, INC., 6745 RELIABLE PARKWAY, CHICAGO, IL, 60686-0067	US Mail (1st Class)
24800	ACE PACKAGING SYSTEMS, 3746 COLLECTION CENTER DR., CHICAGO, IL, 60693	US Mail (1st Class)
24800	ACT LABORATORIES, INC., 273 INDUSTRIAL DR, HILLSDALE, MI, 49242	US Mail (1st Class)
24800	ACTIVE PLASTIC INDUSTRIES PTY. LTD., 104-118 LINK DRIVE, CAMPBELLFIELD VIC, 3061AUSTRALIA	US Mail (1st Class)
24800	ADDRESSER SALES, 13864 BENNETT ROAD, CLEVELAND, OH, 44133	US Mail (1st Class)
24800	ADECCO EMPLOYMENT SERVICES, 10A OAKBRANCH DR, GREENSBORO, NC, 27407	US Mail (1st Class)
24800	ADP, INC., 1 ADP BLVD, ROSELAND, NJ, 07068-1728	US Mail (1st Class)
24800	ADT SECURITY SERVICES, 1 TOWN CENTER RD, BOCA RATON, FL, 33486-1002	US Mail (1st Class)
24800	ADT SECURITY SYSTEMS, 1 TOWN CENTER RD, BOCA RATON, FL, 33486-1002	US Mail (1st Class)
24800	ADVANCED COMPOSITES INC, ROBERT G HANSEMAN ESQ, SEBALY SHILLITO & DYER, 1900 KETTERING TOWER, 40 N MAIN ST, DAYTON, OH, 45423	US Mail (1st Class)
24800	ADVANCED ELASTOMER SYSTEMS, 388 S MAIN ST, AKRON, OH, 44311-1064	US Mail (1st Class)
24800	ADVANCED POLYMERS ALLOYS, DIV. OF FERRO CORP., PO BOX 5831, CLEVELAND, OH, 44193	US Mail (1st Class)
24800	ADVANCED TECHNOLOGIES, SERVICE BUREAU, 44978 FORD ROAD - SUITE D, CANTON, MI, 48187	US Mail (1st Class)
24800	ADVANTAGE ELECTRONICS, PO BOX 407, GREENWOOD, IN, 46143	US Mail (1st Class)
24800	ADVANTAGE PRECISION, 11283 BACO RD, MEADVILLE, PA, 16335	US Mail (1st Class)
24800	ADVANTAGE SERVICES, INC., 409 EAST COOK RD., FORT WAYNE, IN, 46825	US Mail (1st Class)
24800	ADY, PAULETTE K, 112 SHADY LANE CIR NE, WARREN, OH, 44483	US Mail (1st Class)
24800	AEC AUTOMATION, DEPT. 59916, MILWAUKEE, WI, 53259-0916	US Mail (1st Class)
24800	AEC INC., 801 AEC DR, WOOD DALE, IL, 60191-1123	US Mail (1st Class)
24800	AEC/NELMOR, 108 RODDY AVE, ATTLEBORO, MA, 02703-7900	US Mail (1st Class)
24800	AEROTEK, 7301 PARKWAY DR, HANOVER, MD, 21076	US Mail (1st Class)
24800	AGILITY INC, 7000 FITZWATER RD, STE 4, BRECKSVILLE, OH, 44141	US Mail (1st Class)
24800	AICCO, INC., ATT: BARBARA BLAHOUS, 300 S. RIVERSIDE, STE. 200, CHICAGO, IL, 60606	US Mail (1st Class)
24800	AIM SUPPLY/ACE TOOL, 33827 TREASURY CENTER, CHICAGO, IL, 60694-3800	US Mail (1st Class)
24800	AIR HANDLING EQUIPMENT INC, 1389 RIVERSIDE DR., SIDNEY, OH, 45365	US Mail (1st Class)
24800	AIR POWER OF OHIO, 1999 LONGWOOD AVE, COLUMBUS, OH, 43123	US Mail (1st Class)
24800	AIRGAS GREAT LAKES, 311 COLUMBUS AVE, BAY CITY, MI, 48708	US Mail (1st Class)
24800	AKRON BEACON JOURNAL, PO BOX 3661, 44 EAST EXCHANGE ST, AKRON, OH, 44309-3661	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	AKRON BELTING & SUPPLY CO, 145 OSCEOLA AVE., TALLMADGE, OH, 44278	US Mail (1st Class)
24800	AKZO COATINGS, INC., 21442 NETWORK PLACE, CHICAGO, IL, 60673-1214	US Mail (1st Class)
24800	ALABAMA BOLT & SUPPLY, 630 AIR BASE BLVD, MONTGOMERY, AL, 36108	US Mail (1st Class)
24800	ALABAMA GAS CORP., ACCT: 7001831213001, 605 RICHARD ARRINGTON JR BLVD N, BIRMINGHAM, AL, 35203-2707	US Mail (1st Class)
24800	ALABAMA POWER COMPANY, 600 N 18TH ST, BIRMINGHAM, AL, 35291-0001	US Mail (1st Class)
24800	ALABAMA POWER COMPANY, ACCT: 47123-93009 / 32310-08006, 600 NORTH 18TH STREET, BIRMINGHAM, AL, 35291-0001	US Mail (1st Class)
24800	ALABAMA PROTECTIVE SERV., 1019 19TH AVENUE EAST, TUSCALOOSA, AL, 35404	US Mail (1st Class)
24800	ALAMO IRON WORKS INC, 943 AT&T CENTER PKWY, SAN ANTONIO, TX, 78219-3107	US Mail (1st Class)
24800	ALBA ENTERPRISES, INC., 10260 INDIANA COURT, RANCHO CUCAMONGA, CA, 91730	US Mail (1st Class)
24800	ALBECO ALUMINUM BEARING CO., 5091 WEST 164TH ST., BROOK PARK, OH, 44142	US Mail (1st Class)
24800	ALI MED INC, ACCOUNTS RECEIVABLE, PO BOX 9135, DEDHAM, MA, 02027	US Mail (1st Class)
24800	ALL ABOUT PLASTICS, 5339 ST RT 571 E, GREENVILLE, OH, 45331	US Mail (1st Class)
24800	ALL PRO INC., 6770 FIFTH ST, NORTHPORT, AL, 35476	US Mail (1st Class)
24800	ALL RITE INDUSTRIES, 470 OAKWOOD, LAKE ZURICH, IL, 60047	US Mail (1st Class)
24800	ALLEGHENY-CLEARFIELD INC, DBA HAWK PRECISION COMPONENT, 596 W OAK ST, CAMPBELLSBURG, IN, 47108	US Mail (1st Class)
24800	ALLIANCE STAFFING DIV., 25811 ST. RT. 62, BELOIT, OH, 44609	US Mail (1st Class)
24800	ALLIED ELECTRONICS INC, RECEIVABLE MANAGEMENT SERVICES, PO BOX 5126, TIMONIUM, MD, 21094	US Mail (1st Class)
24800	ALLIED ELECTRONICS, INC, ACCTS. RECEIVABLE DEPT., PO BOX 2325, FT. WORTH, TX, 76113-2325	US Mail (1st Class)
24800	ALLIED WASTE SERVICES #223, 15880 N GREENWAY HAYDEN LOOP, SCOTTSDALE, AZ, 85260-1648	US Mail (1st Class)
24800	ALLTEL, 1 ALLIED DR, LITTLE ROCK, AR, 72202-2013	US Mail (1st Class)
24800	ALLTEL CORPORATION, 1 ALLIED DR, LITTLE ROCK, AR, 72202-2013	US Mail (1st Class)
24800	ALPHA TECHNOLOGIES SERV., 2689 WINGATE AVE, AKRON, OH, 44314-1301	US Mail (1st Class)
24800	ALRO STEEL CORP., 821 SPRINGFIELD ST, DAYTON, OH, 45403	US Mail (1st Class)
24800	AMERICAN ELECTRIC POWER, ACCT: 072-571-100-0-2, 1 RIVERSIDE PLZ, COLUMBUS, OH, 43215-2355	US Mail (1st Class)
24800	AMERICAN EXPRESS TRAVEL RELATED, C/O BECKET & LEE LLP, PO BOX 3001, MALVERN, PA, 19355-0701	US Mail (1st Class)
24800	AMERICAN INDUSTRIAL CORP., 1400 AMERICAN WAY, GREENWOOD, IN, 46143	US Mail (1st Class)
24800	AMERICAN OSMENT, 2923 5TH AVE S, BIRMINGHAM, AL, 35233-1503	US Mail (1st Class)
24800	AMERICAN ROTARY TOOLS INC, 1825 D SOUTH PECK RD, MONROVIA, CA, 91016-4778	US Mail (1st Class)
24800	AMERICAN SHOE SERVICE, 230 S. SANDUSKY AVE., BUCYRUS, OH, 44820	US Mail (1st Class)
24800	AMERICAN SYNERGISTICS INC, 35355 SCHOOLCRAFT RD, LIVONIA, MI, 48150	US Mail (1st Class)
24800	AMERICAN ULTRA SPECIALTIES,INC, HINES HILL ROAD, 6855 INDUSTRIAL PARKWAY, HUDSON, OH, 44236-1158	US Mail (1st Class)
24800	AMERIGAS, PO BOX 965, VALLEY FORGE, PA, 19482	US Mail (1st Class)
24800	AMERISOURCE INDUSTRIAL SUPPLY, 9145 VINCENT, HAMTRAMCK, MI, 48211	US Mail (1st Class)
24800	ANDERSON PALLET SERVICE INC., 210 WESTERN AVENUE, PO BOX 669, LEWISBURG, OH, 45338	US Mail (1st Class)
24800	ANDREWS MOVING & STORAGE, ONE ANDREWS CIRCLE, CLEVELAND, OH, 44141	US Mail (1st Class)
24800	ANDROID INDUSTRIES INC, 3800 EAST AVENUE E, ARLINGTON, TX, 76011	US Mail (1st Class)
24800	ANDROID INDUSTRIES, INC., 3800 EAST AVENUE E, ARLINGTON, TX, 76011	US Mail (1st Class)
24800	ANGLE CALIBRATION, 40 SOUTH LANE, TROY, OH, 45373	US Mail (1st Class)
24800	ANTHEM BCBS OH GROUP, 120 MONUMENT CIR, INDIANAPOLIS, IN, 46204-4906	US Mail (1st Class)
24800	APPLIED IND TECHNOLOGIES 2014 NETWORK BL. CHICAGO II. 20072 1005	US Mail (1st Class)
24800	APPLIED IND TECHNOLOGIES, 22510 NETWORK PL., CHICAGO, IL, 60673-1225	US Mail (1st Class)
24800	APPLIED INDUSTRIAL TECH., 22510 NETWORK PLACE, CHICAGO, IL, 60673-1225	US Mail (1st Class)
24800	AQUAPERFECT INC, 5684 W 74TH ST, INDIANAPOLIS, IN, 46278-1752	US Mail (1st Class)
24800	ARAMARK UNIFORM SERVICES, 5120 ADVANTAGE DR, TOLEDO, OH, 43612	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	ARBILL GLOVE & SAFETY PRODUCTS, 10450 DRUMMOND RD, PHILADELPHIA, PA, 19154	US Mail (1st Class)
24800	ARD TRUCKING CO INC, PO BOX 504, DARLINGTON, SC, 29540	US Mail (1st Class)
24800	ARGOSYS AUTOMATION, 840 WOODED HILLS CT, FREMONT, CA, 94539-7246	US Mail (1st Class)
24800	ARIZONA INSTRUMENT CORP., 1912 W 4TH STREET, TEMPE, AZ, 85281	US Mail (1st Class)
24800	ARJ MANUFACTURING LLC, 3300 RIDGECREST ROAD EXT, JACKSON, TN, 38305-7500	US Mail (1st Class)
24800	ARMADA RUBBER, 24586 ARMADA RIDGE RD, ARMADA, MI, 48005	US Mail (1st Class)
24800	ARMSTRONG, TED M, 6581 KALE ADAMS RD, LEAVITTSBURG, OH, 44430	US Mail (1st Class)
24800	ARROW UNIFORM RENTAL, DEPT. 039101, PO BOX 67000, DETROIT, MI, 48267-0391	US Mail (1st Class)
24800	ARROWHEAD SALES CORP, PO BOX 92199, CLEVELAND, OH, 44191	US Mail (1st Class)
24800	ARTX, INC., DBA XTRA STORAGE RENTAL CO., PO BOX 1909, DAYTON, OH, 45401-1909	US Mail (1st Class)
24800	ASHEVILLE HOSE & EQUIPMENT, 192 SWANNANOA RIVER RD, ASHEVILLE, NC, 28813	US Mail (1st Class)
24800	ASHLAND DISTRIBUTION CO., 5200 BLAZER PKWY, DUBLIN, OH, 43017	US Mail (1st Class)
24800	ASHLAND INC., 5200 BLAZER PKWY, DUBLIN, OH, 43017	US Mail (1st Class)
24800	ASHLAND SCALE CO., INC., 2210 ROCKY LANE, ASHLAND, OH, 44805	US Mail (1st Class)
24800	ASSURANT EMPLOYEE BENEFITS, 2323 GRAND BLVD, KANSAS CITY, MO, 64108-2670	US Mail (1st Class)
24800	A-STAMP INDUSTRIES, 633 COMMERCE DR, BRYAN, OH, 43506	US Mail (1st Class)
24800	ASTM, 100 BARR HARBOR DR., WEST CONSHOHOCKEN, PA, 19428-2959	US Mail (1st Class)
24800	AT&T, ACCT: 937R0600053315, N17W24300 RIVERWOOD, WAUKESHA, WI, 53187	US Mail (1st Class)
24800	AT&T UNIVERSAL BILLER, ONE AT&T WAY, BEDMINSTER, NJ, 07921-2693	US Mail (1st Class)
24800	ATCO PRODUCTS, INC., 1002 PERFORMANCE RD., FOREST, VA, 24551	US Mail (1st Class)
24800	ATLAS WELDING SUPPLY, 3530 GREENSBORO AVE, TUSCALOOSA, AL, 35401-7002	US Mail (1st Class)
24800	ATS AUTOMATION TECHNOLOGY SCHWOPE, ATTN KEVIN N SUMMERS, PEAR SPERLING EGGAN & DANIELS PC, 1349 S HURON ST, STE 1, YPSILANTI, MI, 48197	US Mail (1st Class)
24800	ATTICA HYDRAULIC EXHANGE CORP, 48175 GRATIOT AVE., CHESTERFIELD, MI, 48051	US Mail (1st Class)
24800	AUBURN ENGINEERING, 2961 BOND, ROCHESTER, MI, 48309	US Mail (1st Class)
24800	AURORA PLASTICS INC., 9280 JEFFERSON DR, STREETSBORO, OH, 44241	US Mail (1st Class)
24800	AUTOMATED FASTENING TECH. INC., 3637 CORPORATE DRIVE, COLUMBUS, OH, 43231	US Mail (1st Class)
24800	AUTOMATIONDIRECT.COM, INC., 3505 HUTCHINSON RD, CUMMING, GA, 30040	US Mail (1st Class)
24800	AUTOMATRICS, 4400 DONKER CT. SE, GRAND RAPIDS, MI, 49512	US Mail (1st Class)
24800	AUTOMOTIVE INDUSTRIAL SUPPLY, 241 QUADRAL DR, WADSWORTH, OH, 44281	US Mail (1st Class)
24800	AUTOMOTIVE RENTALS INC, ATTN R MOYER, 9000 MIDLANTIC DR, MT LAUREL, NJ, 08054	US Mail (1st Class)
24800	AVI FOODSYSTEMS INC, 2590 ELM RD NE, WARREN, OH, 44483	US Mail (1st Class)
24800	AVIS RENT-A-CAR, VEHICLE DAMAGE CLAIMS, PO BOX 409309, ATLANTA, GA, 30384-9309	US Mail (1st Class)
24800	BACKTRACK EMPLOYMENT, 8850 TYLER BLVD, MENTOR, OH, 44060	US Mail (1st Class)
24800	BALZER, CYNTHIA S, 5085 PRICE WEST, NEWTON FALLS, OH, 44444	US Mail (1st Class)
24800	BANK OF AMERICA, ATTN: AT&T, PO BOX 277019, ATLANTA, GA, 30384-7019	US Mail (1st Class)
24800	BARBER, RONALD, 6159 ST RTE 88, KINSMAN, OH, 44428	US Mail (1st Class)
24800	BARCODE ID SYSTEMS, 1100 JOHNSON FERRY ROAD, SUITE 575, ATLANTA, GA, 30342	US Mail (1st Class)
24800	BARNETT, TAMMY L, PO BOX 80, BRISTOLVILLE, OH, 44402	US Mail (1st Class)
24800	BASELL USA INC., 912 APPLETON RD, ELKTON, MD, 21921-3920	US Mail (1st Class)
24800	BASF CORPORATION, 100 CAMPUS DR, FLORHAM PARK, NJ, 07932	US Mail (1st Class)
24800	BASF-MELLON BANK, 1609 BIDDLE AVE, WYANDOTTE, MI, 48192	US Mail (1st Class)
24800	BASIC ALARM MONITORING, 15163 BURTON-WINDSOR ROAD, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24800	BATTENFELD OF AMERICA, 1620 SHANAHAN DR, SOUTH ELGIN, IL, 60177	US Mail (1st Class)
24800	BC MACHINING INC, PO BOX 365, CONNEAUT, OH, 44030	US Mail (1st Class)
24800	BEACH MANUFACTURING, 118 NORTHAMPTON RD, DONNELSVILLE, OH, 45319	US Mail (1st Class)
24800	BEALS, KATHERINE A, 9460 ROUTE 193, FARMDALE, OH, 44417	US Mail (1st Class)
24800	BEAR FLUID POWER, 34612 CENTAUR, CLINTON TWP., MI, 48035	US Mail (1st Class)
24800	BEARING DISTRIBUTORS INC BDI, ATTN ANDREA MOELLER, PO BOX 936, WATERLOO, IA, 50704	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	BEARING DISTRIBUTORS, INC, 8000 HUB PARKWAY, CLEVELAND, OH, 44125	US Mail (1st Class)
24800	BEARING HDQTRS COMP., 13603 ASHURST ST, LIVONA, MI, 48150	US Mail (1st Class)
24800	BEHCO INC, 32613 FOLSOM RD, FARMINGTON HILLS, MI, 48336	US Mail (1st Class)
24800	BELL FORK LIFT INC, 34660 CENTAUR DR, CLINTON TOWNSHIP, MI, 48035	US Mail (1st Class)
24800	BELLSOUTH, 1155 PEACHTREE ST NE, ATLANTA, GA, 30309-7629	US Mail (1st Class)
24800	BELLSOUTH TELECOMMUNICATIONS INC, 220 SOUTH ST, RM 317, GASTONIA, NC, 28052	US Mail (1st Class)
24800	BEST MOLD & MFG INC, 1546 EAST TURKEYFOOT LAKE RD, AKRON, OH, 44312	US Mail (1st Class)
24800	BETTY SHAFER, 9333 E. LIPPINCOTT BLVD., DAVISON, MI, 48423	US Mail (1st Class)
24800	BIG CHIEF SUPPLY, INC., 5150 BIG CHIEF DR, CINCINNATI, OH, 45227	US Mail (1st Class)
24800	BIRMINGHAM-TOLEDO INC., 3620 VANN RD., BIRMINGHAM, AL, 35235	US Mail (1st Class)
24800	BLACK BOX NETWORK SERVICES, SDS 12-0976, PO BOX 86, MINNEAPOLIS, MN, 55486-0976	US Mail (1st Class)
24800	BLUE WATER PRINTING CO INC, PO BOX 241, 645 MCCORMICK DR, LAPEER, MI, 48446	US Mail (1st Class)
24800	BODYCOTE ACT LABORATORIES, 5996 COLLECTION CENTER DR., CHICAGO, IL, 60693	US Mail (1st Class)
24800	BOSTIK, 22838 NETWORK PLACE, CHICAGO, IL, 60673-1227	US Mail (1st Class)
24800	BOSTWICK BRAUN CO, 5000 GROVE AVE, LORAIN, OH, 44055	US Mail (1st Class)
24800	BOTTING-THOMPSON REALTY, 8569 GERMANTOWN ROAD, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24800	BOWLES FLUIDICS CORP., 6625 DOBBIN RD, COLUMBIA, MD, 21045-4707	US Mail (1st Class)
24800	BP PRODUCTS NORTH AMERICA,INC., 1837 CARILLON PARK DR, OVIEDO, FL, 32765-5148	US Mail (1st Class)
24800	BRAKEY CONSULTING, INC., 3309 GLENCAIRN ROAD, SHAKER HEIGHTS, OH, 44122	US Mail (1st Class)
24800	BRANSON ULTRASONICS, 1585 BARCLAY BLVD, BUFFALO GROVE, IL, 60089	US Mail (1st Class)
24800	BRECKENRIDGE PAPER & PACKAGING, 2425 W MONROE ST, SANDUSKY, OH, 44870	US Mail (1st Class)
24800	BRENDA K. WHITE, PREBLE COUNTY TREASURER, EATON, OH, 45320	US Mail (1st Class)
24800	BRENNAN INDUSTRIAL TRUCK CO., 6940 HALL ST, HOLLAND, OH, 43528	US Mail (1st Class)
24800	BRINKMAN, 325 KISER STREET, DAYTON, OH, 45404-1621	US Mail (1st Class)
24800	BRONER GLOVE & SUPPLY, 1750 HARMON RD, AUBURN HILLS, MI, 48326	US Mail (1st Class)
24800	BROWN CORP OF GREENVILLE, 1927 N THEOBALD STREET EXT, GREENVILLE, MS, 38703-2207	US Mail (1st Class)
24800	BRY-AIR INC, 10793 ST RT 37 WEST, SUNBURY, OH, 43074	US Mail (1st Class)
24800	BRYLER, DANIEL J, 16986 NAUVOO RD, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24800	BSI AMERICA, INC., 13910 COLLECTIONS CTR. DR., CHICAGO, IL, 60693	US Mail (1st Class)
24800	BUCKEYE BUSINESS PRODUCTS INC, 3830 KELLEY AVE, CLEVELAND, OH, 44114	US Mail (1st Class)
24800	BUCKHORN, 55 W TECHNE CENTER DRIVE, MILFORD, OH, 45150	US Mail (1st Class)
24800	BUDGET MOLDERS SUPPLY CO, 8303 CORPORATE PARK DR, MACEDONIA, OH, 44056	US Mail (1st Class)
24800	BUFORD'S LOCKSMITH, 1802 LURLEEN WALLACE BLVD., NORTHPORT, AL, 35476	US Mail (1st Class)
24800	BURNEX CORPORATION, 703 W. ALGONQUIN RD., PO BOX 7357, ALGONQUIN, IL, 60102	US Mail (1st Class)
24800	BUSH, CHERYL L, 7714 STATE ROUTE 7, KINGMAN, OH, 44428	US Mail (1st Class)
24800	BUSINESS & QUALITY INTERGRATIONS, 2159 STATE RTE 121 N, NEW MADISON, OH, 45346	US Mail (1st Class)
24800	BUTLER MACDONALD, INC., 5955 W. 80TH STREET, INDIANAPOLIS, IN, 46278	US Mail (1st Class)
24800	CA JANITORIAL SERVICES INC., 6845 GREENGFIELD, SUITE 203, DETROIT, MI, 48228	US Mail (1st Class)
24800	CAIN STEEL & SUPPLY INC, 2650 20TH ST, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24800	CALIFORNIA INDUSTRIAL PRODUCTS, 33355 TREASURY CENTER, CHICAGO, IL, 60694-3300	US Mail (1st Class)
24800	CALLOS GROUP II, 5083 MARKET STREET, YOUNGSTOWN, OH, 44512-2190	US Mail (1st Class)
24800	CALVIN SPIESS TREASURER, PO BOX 546, NAPOLEAN, OH, 43545-0546	US Mail (1st Class)
24800	CAMCAR DIVISION OF TEXTRON, TEXTRON FASTENING SYSTEMS, 500 18TH AVE, ROCKFORD, IL, 61104	US Mail (1st Class)
24800	CAMDEN FIRE EXTINGUISHER, LLC, 510 S BOUNDARY RD, CAMDEN, SC, 29020	US Mail (1st Class)
24800	CAMMEL SAW CO., 4898 HILLS & DALES RD., CANTON, OH, 44708	US Mail (1st Class)
24800	CANTON CITY BLUE PRINT, 1334 OXFORD AVE NW, CANTON, OH, 44703	US Mail (1st Class)
24800	CANTON CITY UTILITIES, ACCT: 49099508200, 218 CLEVELAND AVE SW, CANTON, OH, 44702-1906	US Mail (1st Class)
24800	CANTON STERILIZED WIPING CLOTH, PO BOX 7227, CANTON, OH, 44705	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	CAPP, INC., 201 MARPLE AVE, CLIFTON HEIGHTS, PA, 19018	US Mail (1st Class)
24800	CAPRO INC A TELEFLEX CO, 640 N LEWIS RD, LIMERICK, PA, 19468-5048	US Mail (1st Class)
24800	CARBIDE SPECIALTIES INC., 3941 EASTERN AVE. SE, GRAND RAPIDS, MI, 49508	US Mail (1st Class)
24800	CARL ZEISS IMT CORPORATION, 6250 SYCAMORE LANE N., MINNEAPOLIS, MN, 55369	US Mail (1st Class)
24800	CARL ZEISS IMT CORPORATION, NW 7241, PO BOX 1450, MINNEAPOLIS, MN, 55485-7241	US Mail (1st Class)
24800	CARLISLE COMPANIES, INC., 13925 BALLANTYNE CORP. PLACE, CHARLOTTE, NC, 28277	US Mail (1st Class)
24800	CARLISLE ENGINEERED PRODUCTS INC, C/O CARRIE M BROSIUS ESQ, VORYS SATER SEYMOUR & PEASE LLP, 2100 ONE CLEVELAND CENTER, 1375 EAST NINTH ST, CLEVELAND, OH, 44114	US Mail (1st Class)
24800	CARLISLE ENGINEERED PRODUCTS INC, CARLISLE COMPANIES INC, MICHAEL L ROBERSON / STEVEN FORD, SENIOR CORPORATE COUNSEL, 250 SOUTH CLINTON STREET, SYRACUSE, NY, 13202-1258	US Mail (1st Class)
24800	CARNEGIE TEXTILE, 1734 IVANHOE RD, CLEVELAND, OH, 44112	US Mail (1st Class)
24800	CAROLINA COMMERCIAL HEAT TREAT, 100 SOUTH MAIN ST, FOUNTAIN INN, SC, 29644	US Mail (1st Class)
24800	CARRIER CORPORATION, P.O. BOX 905303, CHARLOTTE, NC, 28290-5303	US Mail (1st Class)
24800	CARSON FISCHER, P.L.C., 4111 ANDOVER W, 2ND FLOOR, BLOOMFIELD HILLS, MI, 48302-1924	US Mail (1st Class)
24800	CATALINA TOOL & MOLD, 6230 S COUNTRY CLUB ROAD, TUCSON, AZ, 85706	US Mail (1st Class)
24800	CATHERINE M. COX, 4553 RIDGETON RD., BUCYRUS, OH, 44820	US Mail (1st Class)
24800	CD&R COMPONENTS INC, 3247 W STATE RD 229, BATESVILLE, IN, 47006	US Mail (1st Class)
24800	CENTERPOINT INDUSTRIES, 1449 BOWERS RD, LAPEER, MI, 48446	US Mail (1st Class)
24800	CENTURY SPRING CORP., 231274 MOMENTUM PLACE, CHICAGO, IL, 60689-5311	US Mail (1st Class)
24800	CH RACHES INC, DARRELL A CLAY ESQ, WALTER & HAVERFIELD LLP, 1301 EAST NINTH ST, STE 3500, CLEVELAND, OH, 44114-1821	US Mail (1st Class)
24800	CH RACHES INC, C/O DARRELL A CLAY ESQ, WALTER & HAVERFIELD LLP, 1301 E 9TH ST, STE 3500, CLEVELAND, OH, 44114-1821	US Mail (1st Class)
24800	CH ROBINSON WORLDWIDE, INC., 9201 W BROADWAY AVE, MINNEAPOLIS, MN, 55455	US Mail (1st Class)
24800	CHANNEL PRIME ALLIANCE, 1803 HULL AVE, DES MOINES, IA, 50313	US Mail (1st Class)
24800	CHARDON LABORATORIES INC, C/O C L FULLER CFO, 7300 TUSSING RD, REYNOLDSBURG, OH, 43068-4111	US Mail (1st Class)
24800	CHARLES RITTER CO THE, PO BOX 215, 35 WEST SIXTH ST, MANSFIELD, OH, 44901	US Mail (1st Class)
24800	CHASE EQUIPMENT LEASING INC., CORP. PROCESSING DEPT. 0993, COLUMBUS, OH, 43271-0993	US Mail (1st Class)
24800	CHASE PLASTIC SERVICES, DEPARTMENT 231101, PO BOX 67000, DETROIT, MI, 48267	US Mail (1st Class)
24800	CHB TECHNOLOGIES, 4924 CONTEC DRIVE, LANSING, MI, 48910	US Mail (1st Class)
24800	CHEMBAR INC, 1089 CLAYCRAFT RD, GAHANNA, OH, 43230	US Mail (1st Class)
24800	CHEVRON PHILLIPS CHEMICAL CO LP, PO BOX 4910, THE WOODLANDS, TX, 77380	US Mail (1st Class)
24800	CHISSO AMERICA INC, 1800 CENTURY BLVD, STE 1250, ATLANTA, GA, 30345	US Mail (1st Class)
24800	CINCINNATI BELL WIRELESS, 221 E 4TH ST STE 221, CINCINNATI, OH, 45202-4133	US Mail (1st Class)
24800	CINGULAR WIRELESS, GLENRIDGE HIGHLANDS TWO, 5565 GLENRIDGE CONNECTOR NE, ATLANTA, GA, 30342-4756	US Mail (1st Class)
24800	CINTAS CORPORATION, C/O JASON V STITT, KEATING MUETHING & KLEKAMP, ONE EAST FOURTH ST, STE 1400, CINCINNATI, OH, 45202	US Mail (1st Class)
24800	CINTAS FIRST AID & SAFETY, 8400 SWEETVALLEY DRIVE #408, CLEVELAND, OH, 44125	US Mail (1st Class)
24800	CITIBANK (SOUTH DAKOTA) NA, DBA SHELL, 4740 121ST ST, URBANDALE, IA, 50323	US Mail (1st Class)
24800	CITIBANK (SOUTH DAKOTA) NA, DBA OFFICE DEPOT, PO BOX 9025, DES MOINES, IA, 50368	US Mail (1st Class)
24800	CITICORP LEASING, INC., 1615 BRETT RD, NEW CASTLE, DE, 19720	US Mail (1st Class)
24800	CITY BARREL & DRUM, 2881 EAST 91ST STREET, CLEVELAND, OH, 44104	US Mail (1st Class)
24800	CITY OF BISHOPVILLE, ACCT: 103064, 135 E CHURCH ST, BISHOPVILLE, SC, 29010	US Mail (1st Class)
24800	CITY OF TUSCALOOSA, ACCT: 24143, 2201 UNIVERSITY BLVD, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24800	CITY OF VANDALIA WATER, 333 JAMES E BOHANAN MEMORIAL DR, VANDALIA, OH, 45377-2394	US Mail (1st Class)
24800	CLARIANT, MASTERBATCH DIVISION, 3631 COLLECTION CENTER DRIVE, CHICAGO, IL, 60693	US Mail (1st Class)
24800	CLARIANT CORPORATION, 3618 COLLECTION CENTER DR., CHICAGO, IL, 60693	US Mail (1st Class)
24800	CLARK FIRE & SAFETY INC, 619 SOUTH WASHINGTON ST, OWOSSO, MI, 48867	US Mail (1st Class)
24800	CLARK, MICHELE L, 1506 CLEMMENS AVE NW, WARREN, OH, 44485	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	CLEAN NET, 9861 BROKEN LAND PARKWAY, SUITE 208, COLUMBIA, MD, 21046	US Mail (1st Class)
24800	CLEARY MACHINE, 304 STOTLER ROAD, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24800	CLEVELAND ELECTRIC LABS, 1776 ENTERPRISE PKWY, TWINSBURG, OH, 44087	US Mail (1st Class)
24800	CLEVELAND PLUMBING SUPPLY CO., 143 E WASHINGTON STREET, CHAGRIN FALLS, OH, 44022	US Mail (1st Class)
24800	CLEVELAND TOOL & CUTTER, 14181 FOLTZ INDUSTRIAL PARKWAY, STRONGSVILLE, OH, 44136	US Mail (1st Class)
24800	CMD, 3121 INTERSTATE PARKWAY, PO BOX 845, BRUNSWICK, OH, 44212	US Mail (1st Class)
24800	CMM TECHNOLOGY INC., 1230 PUERTO DEL SOL, SAN CLEMENT, CA, 92673	US Mail (1st Class)
24800	CMS SOLUTIONS, 3041 DOUGALL AVE, SUITE 215, WINDSOR, ON, N9E 1S3CANADA	US Mail (1st Class)
24800	CND INDUSTRIES, 4520 SOUTHWAY SW, CANTON, OH, 44706	US Mail (1st Class)
24800	COASTAL WIPERS, 6803 PARKE EAST BLVD, TAMPA, FL, 33610	US Mail (1st Class)
24800	COASTAL WIPERS INC, 6803 PARKE E BLVD, TAMPA, FL, 33610	US Mail (1st Class)
24800	COBER ELECTRONICS, 151 WOODWARD AVE., S. NORWALK, CT, 06854	US Mail (1st Class)
24800	COLONIAL ICE, 2323 GREENVILLE RD., CORTLAND, OH, 44410	US Mail (1st Class)
24800	COMDOC, 3458 MASSILLON RD, UNIONTOWN, OH, 44685	US Mail (1st Class)
24800	COMET AUTOMATION SYSTEMS, 2220 W. DOROTHY LANE, DAYTON, OH, 45439	US Mail (1st Class)
24800	COMMERCIAL TRAFFIC COMPANY, 12487 PLAZA DR, PARMA, OH, 44130	US Mail (1st Class)
24800	COMMUNITY SOLUTIONS ASSOC., 320 HIGH ST. NE, WARREN, OH, 44481	US Mail (1st Class)
24800	COMPONENT MANUFACTURING &, DESIGN INC., PO BOX 845, 3121 INTERESTATE PKWY, BRUNSWICK, OH, 44212	US Mail (1st Class)
24800	COMPRESSOR ENGR CORP, 13330 FOLEY, DETROIT, MI, 48227	US Mail (1st Class)
24800	CONAIR, DEPARTMENT 1119, PO BOX 40000, HARTFORD, CT, 06151-1119	US Mail (1st Class)
24800	CONAIR, INC., DEPARTMENT 1119, PO BOX 40000, HARTFORD, CT, 06151-1119	US Mail (1st Class)
24800	CONCENTRA MEDICAL CENTERS, 26185 GREENFIELD RD, SOUTHFIELD, MI, 48076-4709	US Mail (1st Class)
24800	CONCOURS MOLD AL INC., 651 24TH STREET SW, CULLMAN, AL, 35055	US Mail (1st Class)
24800	CONCOURS MOLD INC., 3400 ST. ETIENNE BLVD., WINDSOR, ON, N8W 5E1CANADA	US Mail (1st Class)
24800	CONFORMANCE MGMT ASSOC LLC, 12460 WOODGATE DR, PLYMOUTH, MI, 48170	US Mail (1st Class)
24800	CONSOLIDATED MACHINERY MOVERS, 15 HEID AVE, DAYTON, OH, 45404	US Mail (1st Class)
24800	CONSOLIDATED PLASTICS, 8181 DARROW RD, TWINSBURG, OH, 44087	US Mail (1st Class)
24800	CONSUMERS ENERGY, ACCT: 04 16 15 7007 2 5, CONSUMERS ENERGY, LANSING, MI, 48937-0001	US Mail (1st Class)
24800	CONSUMERS ENERGY, ACCT: 0526362990008, 1 ENERGY PLAZA DR, JACKSON, MI, 49201	US Mail (1st Class)
24800	CONTINENTAL RIGGERS, INC., 1422 SOUTHFIELD RD, LINCOLN PARK, MI, 48146	US Mail (1st Class)
24800	CONTRACT PROFESSIONALS OF OHIO, DEPARTMENT #165201, PO BOX 67000, DETROIT, MI, 48267-1652	US Mail (1st Class)
24800	CONTRACTOR'S STEEL COMPANY, 1648 RELIABLE PARKWAY, CHICAGO, IL, 60686	US Mail (1st Class)
24800	CONTROL EQUIPMENT, INC., 35401 EUCLID AVE, STE 217, WILLOUGHBY, OH, 44094	US Mail (1st Class)
24800	CONTROL SYSTEM LABORATORIES, 1501 KENSINGTON AVE, BUFFALO, NY, 14215	US Mail (1st Class)
24800	COOPER-STANDARD AUTOMOTIVE FHS, DRAWER #67-399, DETROIT, MI, 48267	US Mail (1st Class)
24800	COPPER & BRASS SALES, 6555 E DAVIDSON ST, DETROIT, MI, 48212	US Mail (1st Class)
24800	CORBIN CORPORATION, RICHLAND DR-SOUTHLAND PK, PO BOX 647, TIFFIN, OH, 44883	US Mail (1st Class)
24800	CORPAD COMPANY INC, PO BOX 1492, 555 PARK AVE EAST, MANSFIELD, OH, 44901-1492	US Mail (1st Class)
24800	CORPORATE EXPRESS, 33722 TREASURY CENTER, CHICAGO, IL, 60694-3722	US Mail (1st Class)
24800	CORRUGATED CONTAINERS INC, 2125 LANDMARK DR, CAYCE, SC, 29033	US Mail (1st Class)
24800	COSAR MOLD INC, 395 PORTAGE BLVD., BRIMFIELD, OH, 44240	US Mail (1st Class)
24800	COSAR MOLDS, 395 PORTAGE BLVD., KENT, OH, 44240	US Mail (1st Class)
24800	COUNTY SEWER DEPT, 1701 MAHONING RD NE, CANTON, OH, 44705	US Mail (1st Class)
24800	COVERT MANUFACTURING, INC, 328 S. EAST STREET, PO BOX 608, GALION, OH, 44833	US Mail (1st Class)
24800	COXCO INC, 3603 PINE LN SE, BESSEMER, AL, 35022	US Mail (1st Class)
24800	CRAIG SMITH AUTO GROUP, 7459 STATE ROUTE 309, PO BOX 850, GALION, OH, 44833	US Mail (1st Class)
24800	CRANE TECHNOLOGIES, 1954 ROCHESTER INDUSTRIAL DR, ROCHESTER HILLS, MI, 48309	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	CRAWFORD CITY BOARD OF COMMISSIONERS, C/O CITY OF CRESTLINE, 100 N SELTZER ST, CRESTLINE, OH, 44827	US Mail (1st Class)
24800	CRAWFORD COUNTY, GARY COLE TREASURER, PO BOX 565, BUCYRUS, OH, 44820-0565	US Mail (1st Class)
24800	CRAWFORD PRODUCTS, 3637 CORPORATE DR, COLUMBUS, OH, 43231	US Mail (1st Class)
24800	CREATIVE PROMOTIONS OF JACKSON, 1106 W. MICHIGAN AVE, JACKSON, MI, 49202	US Mail (1st Class)
24800	CRESTLINE, CITY OF, 100 N SELTZER ST, CRESTLINE, OH, 44827	US Mail (1st Class)
24800	CRIZAF AUTOMATION SYSTEMS, 427 WEST PIKE, PO BOX 590, JACKSON CENTER, OH, 45334-0590	US Mail (1st Class)
24800	CROAL-PALUMBO, ELENA, 4701 STATE ROUTE 534, WEST FARMINGTON, OH, 44491	US Mail (1st Class)
24800	CROSS, ROBERT, 157 WASHINGTON ST NE, WARREN, OH, 44483	US Mail (1st Class)
24800	CROUCH FIRE & SAFETY PRODUCTS INC, 8210 EXPANSION WAY, DAYTON, OH, 45424	US Mail (1st Class)
24800	CRUZ, ENRIQUE, 527 LANSDOWNE BLVD, YOUNGSTOWN, OH, 44505	US Mail (1st Class)
24800	CRUZ, MARIA C, 527 LANSDOWNE BLVD, YOUNGSTOWN, OH, 44505	US Mail (1st Class)
24800	CTR SURPLUS, 202 W. LIVINGSTON AVENUE, CRESTLINE, OH, 44827	US Mail (1st Class)
24800	CTX ACCRA LABORATORIES LLC, PO BOX 472, 1393 E HIGHLAND RD, TWINSBURG, OH, 44087	US Mail (1st Class)
24800	CULLIGAN WATER CONDITIONING, 4404 EAST MAIN, RICHMOND, IN, 47374	US Mail (1st Class)
24800	CURBELL PLASTICS INC, 28455 SCHOOLCRAFT RD, #5, LIVONIA, MI, 48150	US Mail (1st Class)
24800	CURRENCE, ROGER L, 1525 MAIN AVE SW, WARREN, OH, 44483	US Mail (1st Class)
24800	CUSTOM ENGINEERED MACHINE, 2888 SWALLEN AVE, LOUISVILLE, OH, 44641	US Mail (1st Class)
24800	CUT-N-EDGE LAWN CARE, 12811 LAKE LURLEEN RD, COKER, AL, 35452	US Mail (1st Class)
24800	CYBERBOND, LLC, 39105 TREASURY CENTER, CHICAGO, IL, 60694-9100	US Mail (1st Class)
24800	D.R.Y.E. CUSTOM PALLETS, 19400 ALLEN RD., MELVINDALE, MI, 48122	US Mail (1st Class)
24800	D.W. SALES, INC., 1953 BRINSTON, TROY, MI, 48083	US Mail (1st Class)
24800	DALE, SHANE B, 14289 MAIN MARKET, HIRAM, OH, 44234	US Mail (1st Class)
24800	DALIN MOLD & MACHINE CO, PO BOX 220, 2915 INDUSTRIAL PARK DR, AUSTINBURG, OH, 44010	US Mail (1st Class)
24800	DANBOISE MECHANICAL INC, 31625 GRAND RIVER AVE, FARMINGTON, MI, 48336	US Mail (1st Class)
24800	DANKA OFFICE IMAGING, ATTN LEE ACEVEDO, 11101 ROOSEVELT BLVD, ST PETERSBURG, FL, 33716	US Mail (1st Class)
24800	DANLY IEM, SUITE A-F, 6779 ENGLE RD, CLEVELAND, OH, 44130-7952	US Mail (1st Class)
24800	DANYELL WILSON, 8707 SAMUEL BARTON DRIVE, BELLEVILLE, MI, 48111	US Mail (1st Class)
24800	DAR-TECH INC, 16485 ROCKSIDE RD, CLEVELAND, OH, 44137	US Mail (1st Class)
24800	DATA RANGER COMPUTER PDTS., ATTN: ACCOUNTS RECEIVABLE, PO BOX 835, MANCHESTER, MI, 48158-0835	US Mail (1st Class)
24800	DAVID PRICE METAL SERVICES INC, 360 EASTPARK DR, NORWALK, OH, 44857	US Mail (1st Class)
24800	DAVIS STANDARD LLC, 3262 CROWN POINTE DR, STOW, OH, 44224	US Mail (1st Class)
24800	DAY-PAK, 2208 SANDRIDGE DR, DAYTON, OH, 45439	US Mail (1st Class)
24800	DAYTON CAP SCREW, 5747 WEBSTER ST., DAYTON, OH, 45414	US Mail (1st Class)
24800	DAYTON DOOR SALES INC, 1112 SPRINGFIELD ST, DAYTON, OH, 45403	US Mail (1st Class)
24800	DAYTON FREIGHT LINES INC, 6450 POE AVE, STE 311, DAYTON, OH, 45414-2647	US Mail (1st Class)
24800	DAYTON POWER & LIGHT, ACCT: 2835789713, 1065 WOODMAN DR, DAYTON, OH, 45432-1423	US Mail (1st Class)
24800	DAYTON POWER & LIGHT CO., ACCT: 5260978744, 1065 WOODMAN DR, DAYTON, OH, 45432-1423	US Mail (1st Class)
24800	DC-SCIENTIFIC PEST CONTROL, 1814 GREENSBORO AVE., TUSCALOOSA, AL, 35401	US Mail (1st Class)
24800	DE LAGE LANDEN FINANCIAL SERVICES INC, DBA AMERICAN FINANCIAL RESOURCES, 1111 OLD EAGLE SCHOOL RD, WAYNE, PA, 19087	US Mail (1st Class)
24800	DECK VEND SERVICES INC, 1031 CHAMBERLAIN BLVD, CONNEAUT, OH, 44030	US Mail (1st Class)
24800	DELL INC, ATTN MICHAEL KELLER, ONE DELL WAY, BLDG 1 MS 8052, ROUND ROCK, TX, 78682	US Mail (1st Class)
24800	DELPHI AUTOMOTIVE SYSTEMS, 5725 DELPHI DR, TROY, MI, 48098	US Mail (1st Class)
24800	DELPHI PACKARD ELECTRIC SYS, 408 DANA ST NE, WARREN, OH, 44483-3852	US Mail (1st Class)
24800	DELTA DENTAL, 550 POLARIS PARKWAY, SUITE 550, WESTERVILLE, OH, 43082	US Mail (1st Class)
24800	DELTA DIST. & WAREHOUSE INC., 1600 DELTA DR., EL PASO, TX, 79901	US Mail (1st Class)
24800	DELTA TECH INC., 9437 MERCANTILE DRIVE, MENTOR, OH, 44060	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	DEMAG PLASTICS GROUP, 11792 ALAMEDA DR, STRONGSVILLE, OH, 44149	US Mail (1st Class)
24800	DESALVO, LINDA M, PO BOX 256, WINDHAM, OH, 44288	US Mail (1st Class)
24800	DESHAZO CRANE COMPANY, 190 AIRPARK INDUSTRIAL R, ALABASTER, AL, 35007	US Mail (1st Class)
24800	DESIGNETICS, 1624 SOUTH EBER RD., HOLLAND, OH, 43528	US Mail (1st Class)
24800	DETAIL BORING, 190 COY CT, OXFORD, MI, 48371	US Mail (1st Class)
24800	DETROIT BAND SAW WORKS,, P.O. BOX 530566, LIVONIA, MI, 48153	US Mail (1st Class)
24800	DETROIT DOOR & HARDWARE, 111 EAST 12 MILE RD, MADISON HEIGHTS, MI, 48071	US Mail (1st Class)
24800	DETROIT RIGGERS INC, 3519 LAPEER RD, METAMORA, MI, 48455	US Mail (1st Class)
24800	DETROIT TUBULAR RIVET INC, 1213 GROVE ST, WYANDOTTE, MI, 48192	US Mail (1st Class)
24800	DEVICH, JOHN, 823 HOUSEL CRAFT RD, CORTLAND, OH, 44410	US Mail (1st Class)
24800	DEXTER STAMPING COMPANY, 69 ENTERPRISE DR., ANN ARBOR, MI, 48103	US Mail (1st Class)
24800	DIAGRAPH MARKING & STENCILING PDTS, AN ITW COMPANY, C/O THE NORTHERN TRUST COMPANY, 75 REMITTANCE DRIVE, SUITE # 3271, CHICAGO, IL, 60675-3271	US Mail (1st Class)
24800	DIE TECH TOOL, 385 INDUSTRIAL PKWY., IMLAY CITY, MI, 48444	US Mail (1st Class)
24800	DIGITAL TECHNOLOGIES, 135 DIXIE HWY, ROSSFORD, OH, 43460	US Mail (1st Class)
24800	DIRECT LINE PRODUCTS, 427 W PIKE ST, JACKSON CENTER, OH, 45334	US Mail (1st Class)
24800	DISTILLATA COMPANY, 1608 E 24TH ST, CLEVELAND, OH, 44114	US Mail (1st Class)
24800	D-M-E COMPANY, DEPT. LOCKBOX 78242, PO BOX 78000, DETROIT, MI, 48278-0242	US Mail (1st Class)
24800	D-M-E COMPANY, DEPT LOCKBOX 78242, PO BOX 78000, DETROIT, MI, 48278-0424	US Mail (1st Class)
24800	DOCKERY, DONNA J, 1457 AUTUMN DR NW, WARREN, OH, 44485	US Mail (1st Class)
24800	DOCUMENT CONCEPTS INC, 1835 DUEBER AVE SW, CANTON, OH, 44706	US Mail (1st Class)
24800	DOMINION EAST OHIO, ACCT: 50657, ATTN: CREDIT DEPT, 701 E CARY, 18TH FLOOR, RICHMOND, VA, 23219	US Mail (1st Class)
24800	DONNA LEYMAN, 18755 SUMPTER, BELLEVILLE, MI, 48111-9140	US Mail (1st Class)
24800	DORRIE PROCESS COMPANY, DIV. OF HICKS & OTIS PRIN, PO BOX 188, NORWALK, CT, 06852	US Mail (1st Class)
24800	DOUG PAYNE, 900 S. WILEY STREET, PO BOX 320, CRESTLINE, OH, 44827-0320	US Mail (1st Class)
24800	DOUGHERTY'S FLOWERS INC., 3717 TULANE AVE NE, LOUISVILLE, OH, 44641	US Mail (1st Class)
24800	DOVER HYDRAULICS INC, PO BOX 2239, 2996 PROGRESS ST, DOVER, OH, 44622	US Mail (1st Class)
24800	DOW CORNING CORPORATION, ATTN ANGELA COLE, 2200 W SALZBURG RD, MIDLAND, MI, 48686	US Mail (1st Class)
24800	DOW CORNING STI, ATTN ANGELA COLE, 2200 W SALZBURG RD, MIDLAND, MI, 48686	US Mail (1st Class)
24800	DOWDLE GAS COMPANY, 22561 HIGHWAY 82, GORDO, AL, 35466-5373	US Mail (1st Class)
24800	DRUID CITY ELECTRIC, 916 UNIVERSITY LN, STE A, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24800	DRUKENBROD, LANDSCAPING & NURSERY, 8795 CLEVELAND AVE. NW, NORTH CANTON, OH, 44720	US Mail (1st Class)
24800	DRUMMOND AMERICAN, 2721 PAYSPHERE CIRCLE, CHICAGO, IL, 60674	US Mail (1st Class)
24800	DS QUALITY SYSTEMS, 3308 DOTY LANE, ARLINGTON, TX, 76001	US Mail (1st Class)
24800	DSCL ENTERPRISES, INC, C/O DWIGHT CAMERON & ASSOC., 24 KENNY CRES., BARRIE, ON, L4N 6C9CANADA	US Mail (1st Class)
24800	DSM ENGINEERING PLASTICS, 4094 PAYSPHERE CIRCLE, CHICAGO, IL, 60674-0040	US Mail (1st Class)
24800	DSM THERMOPLASTIC ELASTOMERS, 31 FULLER ST, LEOMINSTER, MA, 01453-4451	US Mail (1st Class)
24800	DT FOWLER MFG CO INC, PO BOX 70, 101 N MAPLELEAF RD, LAPEER, MI, 48446	US Mail (1st Class)
24800	DTE ENERGY, ACCT: 0000-3706-9, 2000 2ND AVE, DETROIT, MI, 48226-1203	US Mail (1st Class)
24800	DTE ENERGY, ACCT: 000071530, 2000 2ND AVE, DETROIT, MI, 48226-1203	US Mail (1st Class)
24800	DTE ENERGY, ACCT: 00003706-9, 2000 2ND AVE, DETROIT, MI, 48226-1203	US Mail (1st Class)
24800	DTE ENERGY, ACCT: 5666 678 000 1 5, 2000 2ND AVE, DETROIT, MI, 48226-1203	US Mail (1st Class)
24800	DU PONT MEXICO SA DE CV, HOMERO NO 206, COL CHAPULTEPEC MORALES, MEXICO DF 11570, MEXICO	US Mail (1st Class)
24800	DUBOIS CHEMICALS, JOHNSON DIVERSEY, 1589 PAYSPHERE CIRCLE, CHICAGO, IL, 60674	US Mail (1st Class)
24800	DUKANE CORP/ULTRASONICS, 2900 DUKANE DR, ST CHARLES, IL, 60174-3348	US Mail (1st Class)
24800	DUN & BRADSTREET, ATTN: CLAUDIA CARROS, 340 INTERSTATE NORTH PARKWAY, ATLANTA, GA, 30339	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	DUNN BLUE, ROCHESTER HILLS, 2950 TECHNOLOGY DR., ROCHESTER HILLS, MI, 48309	US Mail (1st Class)
24800	DURABOX DE CHIHUAHUA SA DE CV, 6248 EDGEMERE, STE 757, EL PASO, TX, 79925	US Mail (1st Class)
24800	DYCE, MYRTLE, 9581 GREEN MEADOW RD, APT A, WINDHAM, OH, 44288	US Mail (1st Class)
24800	E I DUPONT DE NEMOURS CO., 6324 FAIRVIEW RD, CHARLOTTE, NC, 28210-3271	US Mail (1st Class)
24800	E.I. DUPONT, CORPORATE HEADQUARTERS 1007 MARKET STREET, WILMINGTON, DE, 19898-0001	US Mail (1st Class)
24800	EAGLE MOLD COMPANY, 315 INDUSTRY DRIVE, CARLISLE, OH, 45005	US Mail (1st Class)
24800	EAR TECH INDUSTRIAL HEARING, CONSERVATION SERVICES, 18212 RUSTIC HOLLOW, STRONGSVILLE, OH, 44136	US Mail (1st Class)
24800	EASTERN SINTERED ALLOYS INC, PO BOX 708, 126 ACCESS RD, ST MARYS, PA, 15857	US Mail (1st Class)
24800	EATON AUTOMOTIVE-FLUID CONNECTORS, PO BOX 12 THORNS RD., BRIERLY HILL, WEST MIDLANDS, DY5 2LBUNITED KINGDOM	US Mail (1st Class)
24800	EATON SUPPLY CO., ROUTE 35 WEST, EATON, OH, 45320	US Mail (1st Class)
24800	EDICT SYSTEMS INC, 2680 INDIAN RIPPLE RD, DAYTON, OH, 45440-3605	US Mail (1st Class)
24800	EDMAC COMPRESSOR CO., 701 E SALEM AVE, WINSTON SALEM, NC, 27101	US Mail (1st Class)
24800	EDS CANADA INC., 1615 DUNDAS ST E, WHITBY, ON, L1N 7S6CANADA	US Mail (1st Class)
24800	EDWARD BONDERENKA, 8707 SAMUEL BARTON DRIVE, BELLEVILLE, MI, 48111	US Mail (1st Class)
24800	EDWARD SEGAL, INC., 360 REYNOLDS BRIDGE RD, THOMASTON, CT, 06787	US Mail (1st Class)
24800	EDWARDS III, JOHNNIE, 2767 LEXINGTON AVE, WARREN, OH, 44425	US Mail (1st Class)
24800	EFC INTERNATIONAL, 1940 CRAIGSHIRE RD, SAINT LOUIS, MO, 63146	US Mail (1st Class)
24800	EFD INC., 977 WATERMAN AVE, EAST PROVIDENCE, RI, 02914	US Mail (1st Class)
24800	EI DUPONT DE NEMOURS, ATTN SUSAN HERR, DUPONT LEGAL D-4026, 1007 MARKET ST, WILMINGTON, DE, 19898-0001	US Mail (1st Class)
24800	EIS FIBER COATING INC, 616 E MAIN ST, LOGANSPORT, IN, 46947	US Mail (1st Class)
24800	EISC INC, 5555 AIRPORT HWY, STE 100, TOLEDO, OH, 43615-7320	US Mail (1st Class)
24800	ELAINE CHANEY, 290 MCCORMICK, LAPEER, MI, 48446	US Mail (1st Class)
24800	ELCO TEXTRON INC., 1111 SAMUELSON RD, ROCKFORD, IL, 61109	US Mail (1st Class)
24800	ELECTRO SUPPLY INC., 2144 ARDMORE BLVD., FOREST HILLS, PA, 15221	US Mail (1st Class)
24800	ELECTRONIC DEV. LABS INC., 244 OAKLAND DR., DANVILLE, VA, 24540	US Mail (1st Class)
24800	EMEDCO, INC., 39209 TREASURY CENTER, CHICAGO, IL, 60694-9200	US Mail (1st Class)
24800	EMHART TEKNOLOGIES, INC., 12337 COLLECTIONS CENTER DRIVE, CHICAGO, IL, 60693	US Mail (1st Class)
24800	EMI PLASTICS EQUIPMENT, PO BOX 590, 427 WEST PIKE STREET, JACKSON CENTER, OH, 45334-0590	US Mail (1st Class)
24800	ENERSYS, INC., 2809 E MILLBROOK RD, RALEIGH, NC, 27604	US Mail (1st Class)
24800	ENGEL MACHINERY INC, 3740 BOARD RD, #5, YORK, PA, 17402	US Mail (1st Class)
24800	ENGINEERING & INDUSTRIAL, SERVICES, INC., 2095 S CO RD 150 E, LOGANSPORT, IN, 46947	US Mail (1st Class)
24800	ENVIRONMENT CONTROL, SHILOH 676, 7939 SOUTH COUNTY RD 25 A, STE A, TIPP CITY, OH, 45371	US Mail (1st Class)
24800	ENVIROSERVE, 5502 SCHAAF RD., INDEPENDENCE, OH, 44131	US Mail (1st Class)
24800	EPCO PRODUCTS, INC., 3736 VANGUARD DR, FORT WAYNE, IN, 46809	US Mail (1st Class)
24800	ERIE CONCRETE STEEL, 1301 CRANBERRY ST, ERIE, PA, 16501	US Mail (1st Class)
24800	ERVIN LEASING CO, PO BOX 1689, ANN ARBOR, MI, 48106	US Mail (1st Class)
24800	ESTATE OF IDA RUTH BURKHART, C/O INA GAIL PRITCHARD, PO BOX 2354, MORRISTOWN, TN, 37816	US Mail (1st Class)
24800	EUCLID BLUE PRINT, 908 E. 222 ST., CLEVELAND, OH, 44123	US Mail (1st Class)
24800	EUROTHERM INC., 741-F MILLER DRIVE, LEESBURG, VA, 20175	US Mail (1st Class)
24800	EVERHARD PRODUCTS INC., 1016 NINTH STREET S.W., CANTON, OH, 44707	US Mail (1st Class)
24800	EXCEL PERSONNEL, INC., 3737, RUE NOTRE-DAME QUEST, MONTREAL, QC, H4C 1P8CANADA	US Mail (1st Class)
24800	EXCEL POLYMERS LLC, COFACE NORTH AMERICA INC, ATTN DAVID MILLER, PO BOX 2102, CRANBURY, NJ, 08512	US Mail (1st Class)
24800	EXELON ENERGY, ACCT: 6420500078440, 21425 NETWORK PLACE, CHICAGO, IL, 60673-1214	US Mail (1st Class)
24800	EXONIC SYSTEMS, LLC, 149 DELTA DR, PITTSBURGH, PA, 15238	US Mail (1st Class)
24800	EXPORT DEVELOPMENT CANADA, 151 O'CONNOR ST, OTTAWA, ON, K1A 1K3CANADA	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	EXPORT DEVELOPMENT CANADA (EDC), 151 O'CONNOR ST, 4TH FL, OTTAWA, ON, K1A 1K3CANADA	US Mail (1st Class)
24800	F A INTEGRATION INC DBA LAKEWOOD AUTOMATION, 27911 CLEMENS RD, WESTLAKE, OH, 44145	US Mail (1st Class)
24800	FABNET ASSOCIATES INC, C/O DARRELL A CLAY ESQ, WALTER & HAVERFIELD LLP, 1301 E 9TH ST, STE 3500, CLEVELAND, OH, 44114-1821	US Mail (1st Class)
24800	FABNET ASSOCIATES INC, DARRELL A CLAY ESQ, WALTER & HAVERFIELD LLP, 1301 EAST NINTH ST, STE 3500, CLEVELAND, OH, 44114-1821	US Mail (1st Class)
24800	FACIL LLC, 1972 BROWN ROAD, AUBURN HILLS, MI, 48326	US Mail (1st Class)
24800	FAIRBANKS SCALES, 821 LOCUST ST, KANSAS CITY, MO, 64106	US Mail (1st Class)
24800	FAIRCHILD INDUSTRIES, 1200 ENSELL RD, LAKE ZURICH, IL, 60047	US Mail (1st Class)
24800	FAIRFIELD INN OF LAPEER, 927 DEMILLE RD., LAPEER, MI, 48446	US Mail (1st Class)
24800	FAIRLAWN, CITY OF, PO BOX 5433, FAIRLAWN, OH, 44334	US Mail (1st Class)
24800	FALCON TOOL COMPANY INC, 7500 HUB PKWY, CLEVELAND, OH, 44125	US Mail (1st Class)
24800	FAMILY ENTERPRISES, INC., PO 88937, MILWAUKEE, WI, 53288-0937	US Mail (1st Class)
24800	FANUC ROBOTICS, ATTN LEGAL, 3900 W HAMLIN RD, ROCHESTER HILLS, MI, 48309-3253	US Mail (1st Class)
24800	FANUC ROBOTICS, ATTN LEGAL, PO BOX 79001, DRAWER #5739, DETROIT, MI, 48279-5739	US Mail (1st Class)
24800	FARMERS WELDING SUPPLY, 14943 MADISON RD, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24800	FASTENAL COMPANY, 4730 SERVICE DR, WINONA, MN, 55987	US Mail (1st Class)
24800	FAY INDUSTRIES INC, 17200 FOLTZ PKWY, STRONGSVILLE, OH, 44149	US Mail (1st Class)
24800	FB SERVICE CORPORATION, 1139 TANGLEWOOD DR., COOKEVILLE, TN, 38501	US Mail (1st Class)
24800	FEDEX CUSTOMER INFORMATION SERVICE, ATTN REVENUE RECOVERY/BANKRUPTCY, 2005 CORPORATE AVE, 2ND FL, MEMPHIS, TN, 38132	US Mail (1st Class)
24800	FEI BEHAVIORAL HEALTH, BOX 88937, MILWAUKEE, WI, 53288-0937	US Mail (1st Class)
24800	FELDER, KATHLEEN, 135 UPLAND, YOUNGSTOWN, OH, 44505	US Mail (1st Class)
24800	FELLENGER JR, LAWRENCE M, 2816 OAKWOOD AVE, SOUTHINGTON, OH, 44470	US Mail (1st Class)
24800	FIES SCALES & SYSTEMS INC, 570 LEO ST, DAYTON, OH, 45404-1506	US Mail (1st Class)
24800	FILING SCALE COMPANY, 1500 ENTERPRISE PKWY, TWINSBURG, OH, 44087	US Mail (1st Class)
24800	FIRST COMMUNICATIONS, ACCT: 3306642940, 3340 W MARKET ST, AKRON, OH, 44333	US Mail (1st Class)
24800	FIRST COMMUNICATIONS, ACCT: 6642940; 4196831500; 3304562481; 4406321671, 7343671403; 4402867111; 8106647212; 2053304926;, 7343975000, 3340 W MARKET ST, AKRON, OH, 44333	US Mail (1st Class)
24800	FIRST COMMUNICATIONS, ACCT: 9374153773; 8034286125, 3340 W MARKET ST, AKRON, OH, 44333	US Mail (1st Class)
24800	FIRTH JOHNSTON MARTINEZ, (RE: DURABOX DE CHIHUAHUA SA DE CV), ATTN VICTOR FIRTH, 415 NORTH MESA, THIRD FLOOR, EL PASO, TX, 79901	US Mail (1st Class)
24800	FIVE STAR LIGHTING, 9001 DUTTON DR, TWINSBURG, OH, 44087	US Mail (1st Class)
24800	FLEX TECHNOLOGIES INC, PO BOX 400, MIDVALE, OH, 44653	US Mail (1st Class)
24800	FLINT WELDING SUPPLY CO., 2201 BRANCH RD., FLINT, MI, 48506	US Mail (1st Class)
24800	FLOW SOLUTIONS, INC., 1725 RUTAN DR, LIVERMORE, CA, 94551-7638	US Mail (1st Class)
24800	FLUIDTROLS CORP, 909 CANTERBURY RD STE G, WESTLAKE, OH, 44145-7212	US Mail (1st Class)
24800	FOAMEX INTERNATIONAL, ATTN DONNA GRASSA, 1000 COLUMBIA AVE, LINWOOD, PA, 19061	US Mail (1st Class)
24800	FOSTER BLUE WATER OIL LLC, ATTN BARBARA SCALPATO CREDIT, PO BOX 4430, RICHMOND, MI, 48062	US Mail (1st Class)
24800	FOX JR, MADISON M, 5986 SODOM HUTCHINGS RD, FARMDALE, OH, 44417	US Mail (1st Class)
24800	FOX, ROSEANN I, 6330 BAZETTA RD, CORTLAND, OH, 44410	US Mail (1st Class)
24800	FRANKLIN COVEY, 2200 PARKWAY BLVD, SALT LAKE CITY, UT, 84119	US Mail (1st Class)
24800	FRASER PRECISION TUBULAR, 11915 53RD ST NE, ALBERTVILLE, MN, 55301	US Mail (1st Class)
24800	FREA, LLC, 4000 EMBASSY PKWY., STE 400, AKRON, OH, 44333	US Mail (1st Class)
24800	FREDERICK, RICK, 1715 ESTABROOK AVE NW, WARREN, OH, 44485	US Mail (1st Class)
24800	FREEMAN, MILDRED, 713 GIBBS AVE NE, CANTON, OH, 44704	US Mail (1st Class)
24800	FREUDENBERG-NOK GENERAL PARTNERSHIP, ATTN KARL BJORKLUND, PO BOX 2001, BRISTOL, NH, 03222	US Mail (1st Class)
24800	FT.WAYNE RUBBER & PLASTICS GROUP, 51 E. NELSON DRIVE, MILFORD, IN, 46542	US Mail (1st Class)
24800	FTC, ACCT: 85195000, 1101 E MAIN ST, KINGSTREE, SC, 29556	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	FUJI COMPONENT PARTS USA, 4115 W 54TH STREET, INDIANAPOLIS, IN, 46254	US Mail (1st Class)
24800	FULLER, NANCY, 12335 HANEY LOCUST LN, LOT 34, GARRETSVILLE, OH, 44231	US Mail (1st Class)
24800	FURBAY/MANSFIELD ELECT. SUPPLY, 208 SCHROYER AVE SW, CANTON, OH, 44702	US Mail (1st Class)
24800	G & K SERVICES, 12875 HURON RIVER DRIVE, ROMULUS, MI, 48174	US Mail (1st Class)
24800	G E POLYMERLAND SERVICE CENTER, 1 PLASTICS AVE, PITTSFIELD, MA, 01201-3662	US Mail (1st Class)
24800	G&C MOLD CO., 11430 CEDAR OAK DRIVE, EL PASO, TX, 79936	US Mail (1st Class)
24800	G. NEIL DIRECT MAIL INC., 720 INTERNATIONAL PARKWAY, FORT LAUDERDALE, FL, 33325	US Mail (1st Class)
24800	GALE, JOHN, 1190 ANDREW AVE, SALEM, OH, 44460	US Mail (1st Class)
24800	GALION COMMUNITY HOSPITAL, ACCOUNTS RECEIVABLE, 269 PORTLAND WAY SOUTH, GALION, OH, 44833	US Mail (1st Class)
24800	GALION TIRE SHOP, 200 SOUTH STREET, GALION, OH, 44833	US Mail (1st Class)
24800	GALION-MANSFIELD PETROLEUM, 1075 KNORR RD., PO BOX 492, GALION, OH, 44833	US Mail (1st Class)
24800	GAMAS XCEL FORWARDING, 911 NORTH INDUSTRIAL PARK, NOGALES, AZ, 85621	US Mail (1st Class)
24800	GANDERT DOOR CO, PO BOX 143, 3077 PARK AVE, ONTARIO, OH, 44862	US Mail (1st Class)
24800	GARCOR SUPPLY CO., INC., 1736 S MILLER AVE., PO BOX 1377, MARION, IN, 46952-1377	US Mail (1st Class)
24800	GARDINER TRANE, 31200 BAINBRIDGE RD, SOLON, OH, 44139	US Mail (1st Class)
24800	GB SALES & SERVICE INC, 39550 SCHOOLCRAFT RD, PLYMOUTH, MI, 48170	US Mail (1st Class)
24800	GBA SYSTEMS, 3200 ATLANTIC AVENUE, SUITE 200, RALEIGH, NC, 27604	US Mail (1st Class)
24800	GBIE INC, 4461 4TH CONCESSION RD NORTH, RR #4, AMHERSTBURG, ON, N9V 2Y9CANADA	US Mail (1st Class)
24800	GBS, 7233 FREEDOM AVE NW, NORTH CANTON, OH, 44720	US Mail (1st Class)
24800	GE BETZ, 7796 COLLECTION CENTER DRIVE, CHICAGO, IL, 60693-0077	US Mail (1st Class)
24800	GE CAPITAL, 260 LONG RIDGE RD, STAMFORD, CT, 06927-1600	US Mail (1st Class)
24800	GE PLASTICS, ATTN VAL VENABLE, 9930 KINCEY AVE, HUNTERSVILLE, NC, 28078	US Mail (1st Class)
24800	GE POLYMERSHAPES, BUSINESS UNIT 0049, 4168 COLLECTIONS CENTER DR., CHICAGO, IL, 60693	US Mail (1st Class)
24800	GE SUPPLY, 684 ROBBINS DR, TROY, MI, 48083	US Mail (1st Class)
24800	GEAUGA DOOR SERVICE, 14249 KINSMAN RD., BURTON, OH, 44021	US Mail (1st Class)
24800	GEAUGA MECHANICAL, 12585 CHARDON-WINDSOR, CHARDON, OH, 44024-8968	US Mail (1st Class)
24800	GEM GRAVURE CO., 112 SCHOOL STREET, WEST HANOVER, MA, 02339	US Mail (1st Class)
24800	GENERAL MACHINERY, 921 FIRST AVE NORTH, BIRMINGHAM, AL, 35203	US Mail (1st Class)
24800	GENERAL PLASTEX INC., 35 STUVER PLACE, BARBERTON, OH, 44203	US Mail (1st Class)
24800	GENERAL TAPE & SUPPLY INC., 21500 WEST EIGHT MILE ROAD, SOUTHFIELD, MI, 48075	US Mail (1st Class)
24800	GENTRAN, 42025 OSGOOD ROAD, FREMONT, CA, 94539	US Mail (1st Class)
24800	GEORGE CONDY, 144 CAUTHEN CIRCLE, ROANOKE, AL, 36274	US Mail (1st Class)
24800	GEORGE PUCCI, 3560 W. MARKET ST., STE. 340, AKRON, OH, 44333	US Mail (1st Class)
24800	GEORGIA, BEVERLY, 269 13TH ST, NILES, OH, 44446	US Mail (1st Class)
24800	GERALD L BAKER, (RE: CND INDUSTRIES), ATTORNEY AT LAW, 3711 WHIPPLE AVENUE NW, GANTON, OH, 44718-2933	US Mail (1st Class)
24800	GERLACH INC., 881 RAVENNA ROAD, KENT, OH, 44240	US Mail (1st Class)
24800	GF GOODMAN & SON INC, 2 IVYBROOK BLVD, WARMINSTER, PA, 18974	US Mail (1st Class)
24800	GIANT EAGLE MIDDLEFIELD, 907 W MAPLE ST, HARTVILLE, OH, 44632	US Mail (1st Class)
24800	GLA WATER INC, 17802 RAPIDS RD, HIRAM, OH, 44234	US Mail (1st Class)
24800	GLA WATER MANAGEMENT, 325 SUPERIOR STREET, ROSSFORD, OH, 43460	US Mail (1st Class)
24800	GLEDHILL LUMBER, 235 BLOOMER ST., GALION, OH, 44833	US Mail (1st Class)
24800	GLO MOLD, INC., 520 S. MAIN STREET, SUITE 2445, AKRON, OH, 44311	US Mail (1st Class)
24800	GLOBAL COMPUTER SUPPLIES, 7795 W FLAGLER, SUITE 35, MIAMI, FL, 33144	US Mail (1st Class)
24800	GLOBAL ENVIRONMENTAL, P.O. BOX 352, ELK RAPIDS, MI, 49629	US Mail (1st Class)
24800	GLOBAL EQUIPMENT, 2505 MILL CENTER PKWY, SUITE 100, BUFFORD, GA, 30518	US Mail (1st Class)
24800	GLOBALSTAR, 461 S MILPITAS BLVD, STE 5, MILPITAS, CA, 95035	US Mail (1st Class)
24800	GOLD KEY PROCESSING LTD, ATTN DAVID MILLER, PO BOX 2102, COFACE NORTH AMERICA INC, CRANBURY, NJ, 08512	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	GOLUK, PAULA J, 4997 ROUTE 322 WEST, ORWELL, OH, 44076	US Mail (1st Class)
24800	GOODWIN, RAYMOND, 397 PARK AVE, CHARDON, OH, 44024	US Mail (1st Class)
24800	GORDINIER ELECTRONICS, 29975 PARKWAY, ROSEVILLE, MI, 48066	US Mail (1st Class)
24800	GOTHAM STAPLE CO., 26101 MILES ROAD, CLEVELAND, OH, 44128	US Mail (1st Class)
24800	GRAND RIVER RUBBER, 2029 AETNA RD, ASHTABULA, OH, 44004	US Mail (1st Class)
24800	GRANT, CALVIN J, 10195 SILICA SAND RD, GARRESTTVILLE, OH, 44231	US Mail (1st Class)
24800	GRAYBAR ELECTRIC, 12431 COLLECTIONS CENTER DRIVE, CHICAGO, IL, 60693	US Mail (1st Class)
24800	GRAYBAR ELECTRIC CO., 900 RIDGE AVE., PITTSBURGH, PA, 15212	US Mail (1st Class)
24800	GREAT LAKES TAPE CORPORATION, 42345 YEAREGO DR, STERLING HEIGHTS, MI, 48314	US Mail (1st Class)
24800	GREAT LAKES WASTE SERVICE, 1633 HIGHWOOD W, PONTIAC, MI, 48340-1244	US Mail (1st Class)
24800	GREATAMERICA LEASING CORPORATION, PO BOX 609, CEDAR RAPIDS, IA, 52406	US Mail (1st Class)
24800	GREGORY, WILLIAM, 8418 DENNISON ASHTABOLA RD, N BLOOMFIELD, OH, 44450	US Mail (1st Class)
24800	GRETAG MACBETH, 617 LITTLE BRITAIN RD, NEW WINDSOR, NY, 12553-6150	US Mail (1st Class)
24800	GRIM, NANCY L, 5212 WARREN RD, CORTLAND, OH, 44410	US Mail (1st Class)
24800	GROFFRE INVESTMENTS, 5211 LOUISVILLE STREET NE, LOUISVILLE, OH, 44641	US Mail (1st Class)
24800	GRUBB JR, RONALD R, 653 BRIGGS RD, LEAVITTSBURG, OH, 44430	US Mail (1st Class)
24800	GSC ENTERPRISES, INC., 720 W LONGVIEW AVE, MANSFIELD, OH, 44906	US Mail (1st Class)
24800	GXS - GLOBAL EXCHANGE SERVICES, 100 EDISON PARK DR MS52A2, GAITHERSBURG, MD, 20878	US Mail (1st Class)
24800	H S SPRING GROUP, 3805 BUSINESS PARK DRIVE, LOUISVILLE, KY, 40213	US Mail (1st Class)
24800	H&H MACHINED PRODUCTS, 2540 MANCHESTER RD, ERIE, PA, 16506	US Mail (1st Class)
24800	H&H TUBE & MFG CO, PO BOX 455, VANDERBILT, MI, 49795	US Mail (1st Class)
24800	H.H. BARNUM COMPANY, 7915 LOCHLIN DR, BRIGHTON, MI, 48116	US Mail (1st Class)
24800	H.R. HOLP & SONS, INC., 530 HAY AVE, BROOKVILLE, OH, 45309	US Mail (1st Class)
24800	HALL, TIMOTHY A, 3256 DUNSTAN DR NW, WARREN, OH, 44485	US Mail (1st Class)
24800	HALLSTAR SPORTS, 105 S. PROSPECT AVE., HARTVILLE, OH, 44630	US Mail (1st Class)
24800	HAND SALES LLC, 1953 BRINSTON, TROY, MI, 48083	US Mail (1st Class)
24800	HANDEL COMPANY, 23980 CHAGRIN BLVD, STE 202, BEACHWOOD, OH, 44122	US Mail (1st Class)
24800	HANSALOY CORPORATION, 820 W 35TH ST, DAVENPORT, IA, 52806	US Mail (1st Class)
24800	HAPCO INC, 390 PORTAGE BLVD, KENT, OH, 44240	US Mail (1st Class)
24800	HARRIS PEST CONTROL, INC., 4928 S IRBY ST, EFFINGHAM, SC, 29541	US Mail (1st Class)
24800	HARWICK STANDARD DISTRIBUTION, 60 S SEIBERLING ST, AKRON, OH, 44305	US Mail (1st Class)
24800	HASTINGS MANUFACTURING, 12951 GRAVOIS RD., SUITE 110, ST. LOUIS, MO, 63127-1714	US Mail (1st Class)
24800	HAWLEY & HAWLEY PLUMBING, 200 W. MONFRONT STREET, EATON, OH, 45320	US Mail (1st Class)
24800	HAYLO MANUFACTURING CO, 5100 RICHMOND ROAD, BEDFORD HEIGHTS, OH, 44146	US Mail (1st Class)
24800	HAYS, DENNIS, 7 ORANGE DR SW, WARREN, OH, 44485	US Mail (1st Class)
24800	HEARN LOGISTICS INC., 2480 SEMINOLE ST., WINDSOR, ON, N8Y 1X3CANADA	US Mail (1st Class)
24800	HELPING HANDS & MORE, PO BOX 271, BROOKVILLE, OH, 45309-0271	US Mail (1st Class)
24800	HENRY M WOOD CO., ATTN: ACCOUNTS RECEIVABLE, 9774 WINDISCH ROAD, WEST CHESTER, OH, 45069	US Mail (1st Class)
24800	HERALD OFFICE SYSTEMS, 110 E ROOSEVELT ST, DILLON, SC, 29536	US Mail (1st Class)
24800	HERCULES FASTENERS INC., 5310 W 161ST, BROOKPARK, OH, 44142-1610	US Mail (1st Class)
24800	HERITAGE CRYSTAL CLEAN, 2175 POINT BLVD, STE 375, ELGIN, IL, 60123	US Mail (1st Class)
24800	HESS INDUSTRIES LTD., 201 EAST FIFTH STREET, SUITE 121, MANSFIELD, OH, 44902	US Mail (1st Class)
24800	HI TECH MOLD & ENGINEERING SE, 466 BAXTER LN, WINCHESTER, TN, 37398	US Mail (1st Class)
24800	HINES INDUSTRIES, INC., 739 AIRPORT BLVD, ANN ARBOR, MI, 48108	US Mail (1st Class)
24800	HJS INC., 2965 E 13TH ST # C, BROWNSVILLE, TX, 78521-3344	US Mail (1st Class)
24800	HOLLINGSWORTH LOGISTIC MGMT., 14225 WEST WARREN AVE., DEARBORN, MI, 48126	US Mail (1st Class)
24800	HOMA II, CHARLES R, 1303 WARREN RD, APT B, NEWTON FALLS, OH, 44444	US Mail (1st Class)
24800	HOMA, ANITA D, 1303 WARREN RD, APT B, NEWTON FALLS, OH, 44444	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	HOME CITY ICE, 20282 HANNAN PKWY, WALTON HILLS, OH, 44146	US Mail (1st Class)
24800	HOME DEPOT, 2455 PACES FERRY RD SE, ATLANTA, GA, 30339-1834	US Mail (1st Class)
24800	HOME DEPOT CREDIT SERVICES, DEPT. 32-2501172740, PO BOX 9055, DES MOINES, IA, 50368-9055	US Mail (1st Class)
24800	HONDA OF AMERICA MFG INC, JOSEPH F LAFLEUR, CORPORATE COUNSEL, 24000 HONDA PARKWAY, MARYSVILLE, OH, 43040-9251	US Mail (1st Class)
24800	HONDA OF AMERICA MFG INC, ROBERT SIDMAN & BRENDA BROWES, PO BOX 1008, VORYS SATER SEYMOUR & PEASE LLP, 52 EAST GAY ST, COLUMBUS, OH, 43216-1008	US Mail (1st Class)
24800	HONDA TRADING AMERICA, 19900 STATE RT 739, MARYSVILLE, OH, 43040	US Mail (1st Class)
24800	HONDA TRADING AMERICA, 19900 STATE ROUTE 739, MARYSVILLE, OH, 43040-9256	US Mail (1st Class)
24800	HONE ENGINEERING INC, 30409 FOX CLUB, FARMINGTON HILLS, MI, 48331	US Mail (1st Class)
24800	HOOVER INSTRUMENT SERVICE, 401 NORTH HOME RD, MANSFIELD, OH, 44906	US Mail (1st Class)
24800	HOPKINS ASSEMBLY & PACKAGING, 3440 KAULOOSA AVE., TUSCALOOSA, AL, 35401	US Mail (1st Class)
24800	HORAUF & KOHLER GMBH, PETER-DORFLER-STIABE 27, D-86195 AUGSBURG, GERMANY	US Mail (1st Class)
24800	HORRELL, JESSICA, 683 B CENTER ST, WARREN, OH, 44481	US Mail (1st Class)
24800	HOSTETLER, PETE P, 15500 SHEDD RD, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24800	HOWELL, JANE S, PO BOX 426, 10001 NORTH BRADFIELD, GARRETTSVILLE, OH, 44231	US Mail (1st Class)
24800	HP PRODUCTS INC, 512 W GORGAS ST, LOUISVILLE, OH, 44641-3912	US Mail (1st Class)
24800	HR HOLP & SON, INC., 530 HAY AVE, BROOKEVILLE, OH, 45309	US Mail (1st Class)
24800	HTC/TOPS ALL CLAMPS, INC., 26627 W. 8 MILE RD, REDFORD, MI, 48240	US Mail (1st Class)
24800	HTG RIMES LOGISTIC SERVICES INCO, PO BOX 943, CHARDON, OH, 44024-0943	US Mail (1st Class)
24800	HUBBARD INDUSTRIAL SUPPLY, 901 W 2ND ST, FLINT, MI, 48503-2680	US Mail (1st Class)
24800	HUDSON HIGHLAND GROUP INC., 6980 NORTH PORT WASHINGTON RD., SUITE 203, GLENDALE, WI, 53217	US Mail (1st Class)
24800	HUNGRY HOWIES #73, PO BOX 1035, 10784 BELLEVILLE RD, BELLEVILLE, MI, 48112	US Mail (1st Class)
24800	HURLEY DESIGN, 5051 RIMERS DR., JACKSON, MI, 49201	US Mail (1st Class)
24800	HURST, JOYCE L, 5472 RISHER RD, LEAVITTSBURG, OH, 44430	US Mail (1st Class)
24800	HUSKY INJECTION MOLDING SYS, 55 AMHERST VILLA RD, CHEEKTOWAGA, NY, 14225	US Mail (1st Class)
24800	HUSMAN, CAROL L, PO BOX 500, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24800	HUSMAN, KEITH E, PO BOX 500, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24800	HYDRAULIC CONNECTIONS, 2848 INTERSTATE PKWY, BRUNSWICK, OH, 44212-5626	US Mail (1st Class)
24800	HYDROFORM SOLUTIONS, 1995 WILLIAMS PARKWAY, BRAMPTON, ON, L6S 6E5CANADA	US Mail (1st Class)
24800	HYDROTECH, INC., 10052 COMMERCE PARK DRIVE, CINCINNATI, OH, 45246	US Mail (1st Class)
24800	HYNST, DALE W, 4445 SOUTH PARK RD, W FARMINGTON, OH, 44491	US Mail (1st Class)
24800	HYNST, DARRYLL J, 635 BRISTOL-CHAMPION TWLN RD, BRISTOLVILLE, OH, 44402	US Mail (1st Class)
24800	I.M.S. COMPANY, 10373 STAFFORD RD, CHAGRIN FALLS, OH, 44023	US Mail (1st Class)
24800	IBM CREDIT LLC, BH SHIDELER, 2 LINCOLN CENTRE, OAKBROOK TERRACE, IL, 60181	US Mail (1st Class)
24800	IER INDUSTRIES INC., 8271 BAVARIA DR, MACEDONIA, OH, 44056	US Mail (1st Class)
24800	IET LABS, INC., 534 MAIN ST., WESTBURY, NY, 11590	US Mail (1st Class)
24800	IKON FINANCIAL SERCICES, 70 VALLEY STREAM PKWY, MELBOURNE, PA, 19355	US Mail (1st Class)
24800	IKON FINANCIAL SERVICES, BANKRUPTCY ADMINISTRATION, PO BOX 13708, MACON, GA, 31208-3708	US Mail (1st Class)
24800	IKON FINANCIAL SERVICES, BANKRUPTCY ADMINISTRATION, PO BOX 13708, MACON, GA, 31208	US Mail (1st Class)
24800	IKON OFFICE SOLUTIONS, 70 VALLEY STREAM PKWY, MELBOURNE, PA, 19355	US Mail (1st Class)
24800	IMP, 32 LOUCK, OXFORD, MI, 48371	US Mail (1st Class)
24800	IMS COMPANY, 10373 STAFFORD RD, CHAGRIN FALLS, OH, 44023	US Mail (1st Class)
24800	INCAT, 41370 BRIDGE ST., NOVI, MI, 48375	US Mail (1st Class)
24800	INCAT SOLUTIONS, INC., DEPT. # 123301, PO BOX 67000, DETROIT, MI, 48267-1233	US Mail (1st Class)
24800	INCOE CORPORATION, 1740 E MAPLE RD, TROY, MI, 48083	US Mail (1st Class)
24800	INCONTROL INC, 117 HILLWOOD CIR, NEWNAN, GA, 30263	US Mail (1st Class)
24800	INDENCO, 1515 WHIPPLE AVE SW, CANTON, OH, 44710-1323	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	INDEPENDENCE BUSINESS, 4550 HINCKLEY PKWY, CLEVELAND, OH, 44109-6010	US Mail (1st Class)
24800	INDIANS HILLS COUNTRY CLUB, 1650 MCFARLAND BLVD NORTH, TUSCALOOSA, AL, 35406	US Mail (1st Class)
24800	INDUSTRIAL MAINTENANCE EQUIP CENTER, 20638 KRICK RD, WALTON HILLS, OH, 44146	US Mail (1st Class)
24800	INDUSTRIAL MOLD & MACHINE, 2057 E AURORA RD, UNIT H, TWINSBURG, OH, 44087	US Mail (1st Class)
24800	INDUSTRIAL TAPES, INC., 10357 KINSMAN RD, NEWBURY, OH, 44065	US Mail (1st Class)
24800	INDUSTRIAL WAREHOUSE SERVICE, TUSCALOOSA WAREHOUSE, PO BOX 2177, TUSCALOOSA, AL, 35403	US Mail (1st Class)
24800	INDUSTRY PRODUCTS COMPANY, 500 STATLER RD, PIQUA, OH, 45356	US Mail (1st Class)
24800	INEOS USA LLC, DBA INEOS OLEFINS & POLYMERS USA, 2600 SOUTH SHORE BLVD, LEAGUE CITY, TX, 77573	US Mail (1st Class)
24800	INFOR GLOBAL SOLUTIONS, 13560 MORRIS RD, SUITE 4100, ALPHARETTA, GA, 30004	US Mail (1st Class)
24800	INPRAX, ATTN JENELLE MADONNA, 7301 PARKWAY DR, HANOVER, MD, 21076	US Mail (1st Class)
24800	INSIGHT DIRECT, INC., 6820 S HARL AVE, TEMPE, AZ, 85283	US Mail (1st Class)
24800	INSTRUMENT SERV & EQUIP, 10100 ROYALTON RD, CLEVELAND, OH, 44133	US Mail (1st Class)
24800	INTEGRATED LOGISTICS, 23000 EUCLID AVE, CLEVELAND, OH, 44117	US Mail (1st Class)
24800	INTERLAND, INC., 303 PEACHTREE CENTER AVE NE #500, ATLANTA, GA, 30303	US Mail (1st Class)
24800	INTERNATIONAL FIRE PROTECTION, 243 ROYAL DRIVE, MADISON, AL, 35758	US Mail (1st Class)
24800	INTERNATIONAL OUTSOURCING SERVICES LLC, 1600 W BLOOMFIELD RD, STE A, BLOOMINGTON, IN, 47403	US Mail (1st Class)
24800	INTERSTATE BATTERIES, 320 NORTH MULBERRY ST., MANSFIELD, OH, 44902	US Mail (1st Class)
24800	INTERSTATE CAPITAL CORPORATION, C/O TRANS AMERICA CARRIERS, INC., PO BOX 915183, DALLAS, TX, 75391-5183	US Mail (1st Class)
24800	INTERSTATE OPTICAL CO, PO BOX 308, 680 LINDAIRE LN, MANSFIELD, OH, 44901	US Mail (1st Class)
24800	IOS CAPITAL, 1738 BASS RD, MACON, GA, 31210-1043	US Mail (1st Class)
24800	IRON AGE CORP., 13898 COLLECTIONS CENTER DR., CHICAGO, IL, 60693	US Mail (1st Class)
24800	ITWCIP/ANCHOR STAMPINGS, 33355 TREASURY CENTER, CHICAGO, IL, 60694-3300	US Mail (1st Class)
24800	IWATA BOLT USA INC, 7131 ORANGEWOOD AVE, GARDEN GROVE, CA, 92841-1409	US Mail (1st Class)
24800	J & L INDUSTRIAL SUPPLY, 31800 INDUSTRIAL ROAD, LIVONIA, MI, 48150	US Mail (1st Class)
24800	J&H MACHINE TOOLS INC, 4345 MORRIS PARK DR, CHARLOTTE, NC, 28227	US Mail (1st Class)
24800	J&L INDUSTRIAL SUPPLY, 20921 LAHSER RD, SOUTHFIELD, MI, 48033-4432	US Mail (1st Class)
24800	J.A. SHOMER CO., INC., 1951 WEST 112TH STREET, CLEVELAND, OH, 44102	US Mail (1st Class)
24800	J.B. QUALITY SERVICES, 449 SUNSET CRESCENT, ROSENEATH LANDING, ON, K0K 2X0CANADA	US Mail (1st Class)
24800	J.H. BENNETT & COMPANY, 22975 VENTURE DR, NOVI, MI, 48376	US Mail (1st Class)
24800	J.J. KELLER & ASSOCIATES INC., 3003 W. BREEZEWOOD LANE, PO BOX 548, NEENAH, WI, 54957-0548	US Mail (1st Class)
24800	J.K. JIM`S ELECTRIC INC., 6585 WINDFALL ROAD, GALION, OH, 44833	US Mail (1st Class)
24800	JACKSON, GAIL, 5942 BEACH SMITH RD, KINSMAN, OH, 44428	US Mail (1st Class)
24800	JACO PRODUCTS, 15060 MADISON RD, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24800	JAMES LOMAS, 8707 SAMUEL BARTON DRIVE, BELLEVILLE, MI, 48111	US Mail (1st Class)
24800	JAMESTOWN CONTAINER, 85 GRAND ST, LOCKPORT, NY, 14094	US Mail (1st Class)
24800	JAN-PRO, OF NORTHEAST OHIO, 2858 SOUTH ARLINGTON STE. 100, AKRON, OH, 44312	US Mail (1st Class)
24800	JASPER RUBBER, 1010 1ST AVE W, JASPER, IN, 47546	US Mail (1st Class)
24800	JAY INDUSTRIAL SALES CO., 30559 PINETREE RD., STE. 216, CLEVELAND, OH, 44124	US Mail (1st Class)
24800	J-COM - E.D.I. SERVICES, 1670 N KOLB RD, TUSCON, AZ, 85715	US Mail (1st Class)
24800	JEFCO MACHINE SERVICE INC, 460 VANESSA DR, W ALEXANDRIA, OH, 45381	US Mail (1st Class)
24800	JEFF NORRIS, 3709 GREENBROOK DRIVE, NORTHPORT, AL, 35475	US Mail (1st Class)
24800	JELLIFF CORPORATION, 1351 NE 51ST STREET, POMPANO BEACH, FL, 33064	US Mail (1st Class)
24800	JERGENS, INC., 15700 S WATERLOO RD, CLEVELAND, OH, 44110	US Mail (1st Class)
24800	JET ELECTRIC CO., 21333 HILLTOP, SOUTHFIELD, MI, 48034	US Mail (1st Class)
24800	JF GOOD COMPANY, COMMERCIAL COLLECTION CORP, PO BOX 288, TONAWANDA, NY, 14150	US Mail (1st Class)
24800	JM MOLD, 1707 COMMERCE DRIVE, PO BOX 730, PIQUA, OH, 45356	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	JO KING, 8041-B ARROW RIDGE RD., CHARLOTTE, NC, 28273	US Mail (1st Class)
24800	JOHN L AND ASSOC., 140 BUSINESS CENTER DR, BIRMINGHAM, AL, 35244	US Mail (1st Class)
24800	JOHNSON CONTROLS INC, PO BOX 242, MILWAUKEE, WI, 53278	US Mail (1st Class)
24800	JONES, BRENDA J, 2789 RED FOX RUN DR NW, WARREN, OH, 44485	US Mail (1st Class)
24800	JORDAN, KAREN L, 300 MAPLEWOOD DR, APT 5, CORTLAND, OH, 44410	US Mail (1st Class)
24800	JSW PLASTICS MACHINERY, 555 PROMENADE AVE., CORONA, CA, 92879	US Mail (1st Class)
24800	KABRE CONVEYOR, 3864 CENTER ROAD, SUITE A-9, BRUNSWICK, OH, 44212	US Mail (1st Class)
24800	KANO LABORATORIES, 1000 EAST THOMPSON LANE, NASHVILLE, TN, 37211	US Mail (1st Class)
24800	KARDEX SYSTEMS, INC., 1189 PAXSPHERE CIRCLE, MARIETTA, OH, 45750	US Mail (1st Class)
24800	KAULOOSA SOLUTIONS, 6072 LOBLOLLY LANE, TUSCALOOSA, AL, 35405	US Mail (1st Class)
24800	KAUTEX INC, ATTN DIANE STRICKLAND, 11182 GA HWY 17 SOUTH, LEVONIA, GA, 30553	US Mail (1st Class)
24800	KELLER AUTO PARTS-GALION, 121 HARDING WAY WEST, GALION, OH, 44833	US Mail (1st Class)
24800	KELLY SERVICES INC., 999 W BIG BEAVER RD, TROY, MI, 48084	US Mail (1st Class)
24800	KEMP, WILLIAM M, 373 ELDON DR, WARREN, OH, 44483	US Mail (1st Class)
24800	KENT MANUFACTURING CO, 1840 OAK INDUSTRIAL DR NE, GRAND RAPIDS, MI, 49505	US Mail (1st Class)
24800	KESSLER SIGN COMPANY, 2669 NATIONAL RD, ZANESVILLE, OH, 43701	US Mail (1st Class)
24800	KEVIN DENLAR, 8707 SAMUEL BARTON DRIVE, BELLEVILLE, MI, 48111	US Mail (1st Class)
24800	KEYENCE CORP. OF AMERICA, DEPT. CH 17128, PALATINE, IL, 60055-7128	US Mail (1st Class)
24800	KEYSTONE COMPONENTS, INC., 1960 CASE PARKWAY SOUTH, TWINSBURG, OH, 44087	US Mail (1st Class)
24800	KINCSES TOOL & MOLDING, KEARNEY INDUSTRIAL PARK, PO BOX 69, FLORA, MS, 39071	US Mail (1st Class)
24800	KIRTLAND LANDSCAPE INC., 4791 MOUNT MORRIS RD, COLUMBIAVILLE, MI, 48421-8945	US Mail (1st Class)
24800	KLINEDINST, SCOTT H, 3245 TOD AVE NW, WARREN, OH, 44483	US Mail (1st Class)
24800	KOLAT, SEAN E, 5541 STATE ROUTE 45, BRISTOLVILLE, OH, 44402	US Mail (1st Class)
24800	KONGSKILDE INDUSTRIES INC., 2439 S. MAIN STREET, SUITE 2, BLOOMINGTON, IL, 61704-7362	US Mail (1st Class)
24800	KONICA MINOLTA-ALBIN, 46921 ENTERPRISE CT., WIXOM, MI, 48393	US Mail (1st Class)
24800	KORTHAS, SHARON, 26 EAST WALNUT ST, JEFFERSON, OH, 44047	US Mail (1st Class)
24800	KOTZ SANGSTER WYSOCKI BERG PC, (RE: QIS INC AKA QUALITY INDUSTRIAL SRVCS INC), ATTN REBECCA M DECOSTER, 400 RENAISSANCE CENTER, SUITE 2555, DETROIT, MI, 48243-1675	US Mail (1st Class)
24800	KOZAK, CHARLES V, PO BOX 426, LEAVITTSBURG, OH, 44430	US Mail (1st Class)
24800	KOZAK, RUSSELL, 645 PARKMAN RD NW, WARREN, OH, 44485	US Mail (1st Class)
24800	KRAFT FLUID SYSTEMS INC, 14300 FOLTZ PKWY, STRONGSVILLE, OH, 44149	US Mail (1st Class)
24800	KROY LLC, 3830 KELLEY AVE, CLEVELAND, OH, 44114	US Mail (1st Class)
24800	KUNTZMAN TRUCKING, 1805 W STATE ST, ALLIANCE, OH, 44601	US Mail (1st Class)
24800	KW PLASTICS SALES & MKTG, 1 SANDERS RD, TROY, AL, 36079-2504	US Mail (1st Class)
24800	KYLE OFFICE SUPPLY, 1020 21ST AVE, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24800	L & L PRODUCTS, 160 MCLEAN, ROMEO, MI, 48065	US Mail (1st Class)
24800	LAGLENDELL MULLINS, 3820 1ST AVE., APT. 210, TUSCALOOSA, AL, 35405	US Mail (1st Class)
24800	LAKE ERIE WAREHOUSE &, DISTRIBUTION CENTER, 5650 WATTSBURG ROAD, ERIE, PA, 16509	US Mail (1st Class)
24800	LAKE HOSPITAL SYSTEM, P.O. BOX 714031, COLUMBUS, OH, 43271	US Mail (1st Class)
24800	LAMKO TOOL & MOLD INC, 105 TOWERLINE PL, LONDON, ON, N6E 2T3CANADA	US Mail (1st Class)
24800	LANSMONT CORPORATION, 6539 WESTLAND WAY, SUITE 24, LANSING, MI, 48917	US Mail (1st Class)
24800	LANXESS CORP, ATTN DEBBIE KRAUSE, 111 RIDC PARK WEST DR, PITTSBURGH, PA, 15275-1112	US Mail (1st Class)
24800	LARCO INDUSTRIES INC, 9833 5 MILE RD, NORTHVILLE, MI, 48168-9403	US Mail (1st Class)
24800	LASER 3 LC, 12794 CURRIE CT., LIVONIA, MI, 48150	US Mail (1st Class)
24800	LAVERGNE GROUP INC., MONTCAP FINANCIAL CORPORATION, 3500 DEMAISONNEUVE BLVD. WEST., SUITE 1510, MONTREAL, QC, H3Z 3C1CANADA	US Mail (1st Class)
24800	LAW THE FLORIST, 121 S. WASHINGTON AVE., PO BOX 411, CRESTLINE, OH, 44827	US Mail (1st Class)
24800	LAYNE, HARRY J, PO BOX 634, 2850 HAGUE RD, ORWELL, OH, 44076	US Mail (1st Class)
24800	LEGACY MOLD AND TOOL, INC., 4525 SPRINGFIELD RD., BARDSTOWN, KY, 40004	US Mail (1st Class)
24800	LEGGETT & PLATT, 901 W MAPCE, CLAWSON, MI, 48017	US Mail (1st Class)

24800 LEHMANN, WILLIAM T. 3308 WARREN BURTON RD, SOUTHINGTON, OH, 44470         US Mail (1st Class)           24800 LEMAC PACKAGING, 2121 MCKINLEY AVE, ERIE, PA, 16514-0788         US Mail (1st Class)           24800 LEMACD PACKAGING, 2121 MCKINLEY AVE, ERIE, PA, 165140         US Mail (1st Class)           24800 LEO'S PHELMATICS, 1821 FRANKLIN AVE, ERIE, PA, 16510         US Mail (1st Class)           24800 LEONAD, ROSEANNA, 2929 RED FOX RUN, WARREN, OH, 44485         US Mail (1st Class)           24800 LICHTY, JENNIFER, 2822 LCIBY OSBORNE RD, SOUTHINGTON, OH, 444707         US Mail (1st Class)           24800 LICHTY, JENNIFER, 2822 LCIBY OSBORNE RD, SOUTHINGTON, OH, 444707         US Mail (1st Class)           24800 LINDE GAS LLC, 6858 ROCKSIDE WOODS BLVD, INDEPENDENCE, OH, 44191-2301         US Mail (1st Class)           24800 LINDERA MOLD & ENGINEERING, 34435 GLENDALE, LIVONIA, MI, 49150         US Mail (1st Class)           24800 LINDERA ROLD & ENGINEERING, 34435 GLENDALE, LIVONIA, MI, 49150         US Mail (1st Class)           24800 LORES TER, GRACE, E, 5107 COLD STATE RO, WEST EARMINGTON, OH, 444891         US Mail (1st Class)           24800 LORES TAR QUALITY SVS INC, 1117 EAGLE RIDGE DR, EL PASO, TX, 79912         US Mail (1st Class)           24800 LOPEZ, JAMIES D, 501 PARKMAN RD NW, WARREN, OH, 44483         US Mail (1st Class)           24800 LOUGH, DAYTON, 461 BELMONT NE, WARREN, OH, 44483         US Mail (1st Class)           24800 LOUGH, DAYTON, 461 DEL MEN CONTROL ON WARREN, OH, 44483	Svc Lst	Name and Address of Served Party	Mode of Service
24800         LENARD TOOL & MACHINE, 3125 E. ALLERTON AVE., ST FRANCIS, WI. 52355         US Mail (1st Class)           24800         LEO'S PNEUMATICS, 1821 FRANKLIN AVE, ERIE, PA, 16510         US Mail (1st Class)           24800         LEONARD, ROSEANMA, 2928 RED FOR KIN, WARREN, OH, 44465         US Mail (1st Class)           24800         LIBERTY SOLVENTS & CHEMICALS CO INC, 9429 RAVENNA RD, TWINSBURG, OH, 44087         US Mail (1st Class)           24800         LIDERTY, JENNIFER, 2822 LCIBY O'SBORNE RD, SOUTHINGTON, OH, 44470         US Mail (1st Class)           24800         LINEAR MOLD & ENGINEERING, 34435 GLENDALE, LIVONIA, MI, 48150         US Mail (1st Class)           24800         LINEAR MOLD & ENGINEERING, 34435 GLENDALE, LIVONIA, MI, 48150         US Mail (1st Class)           24800         LINEAR MOLD & ENGINEERING, 34435 GLENDALE, LIVONIA, MI, 4491         US Mail (1st Class)           24800         LOPEZ, JAMES D, 501 PARKAMA RD NW, WARREN, OH, 44485         US Mail (1st Class)           24800         LOPEZ, JAMES D, 501 PARKAMA RD NW, WARREN, OH, 44483         US Mail (1st Class)           24800         LORD CORPORATION, 2000 W GRANDVEW BLUD, ERIE, PA, 16509         US Mail (1st Class)           24800         LORD CORPORATION, 2000 W GRANDVEW BLUD, ERIE, PA, 16509         US Mail (1st Class)           24800         LORD CORPORATION, 2000 W GRANDVEW BLUD, ERIE, PA, 16509         US Mail (1st Class) <th< td=""><td>24800</td><td>LEHMANN, WILLIAM T, 3308 WARREN BURTON RD, SOUTHINGTON, OH, 44470</td><td>US Mail (1st Class)</td></th<>	24800	LEHMANN, WILLIAM T, 3308 WARREN BURTON RD, SOUTHINGTON, OH, 44470	US Mail (1st Class)
24800         LEO'S PNEUMATICS, 1821 FRANKLIN AVE, ERIE, PA, 16510         US Mail (1st Class)           24800         LEONARD, ROSEANNA, 2929 RED FOX RUN, WARREN, OH, 44485         US Mail (1st Class)           24800         LIENTY SOLVENTS & CHEMICALS CO INC, 4928 RAVENNA RD, TWINSBURG, OH, 44087         US Mail (1st Class)           24800         LICHTY, JENNIFER, 2822 LCIBY OSBORNE RD, SOUTHINGTON, OH, 44470         US Mail (1st Class)           24800         LINDE GAS LLC, 6055 ROCKSIDE WOODS BLVD, INDEPENDENCE, OH, 44131-2301         US Mail (1st Class)           24800         LINIFORM SERVICE, 1050 NORTHYIEW AVE., BARBERTON, OH, 44203-7197         US Mail (1st Class)           24800         LINFORM SERVICE, 1050 NORTHYIEW AVE., BARBERTON, OH, 444991         US Mail (1st Class)           24800         LIPPERT, CRACE E, 5017 OLD STATE RD, WEST FARMINGTON, OH, 444991         US Mail (1st Class)           24800         LORD STAR OUALITY SSY INC. 1117 EAGLE RIDGE OR E, ELPASO, TX. 79912         US Mail (1st Class)           24800         LORD CORPORATION, 2000 W GRANDYIEW BLVD, ERIE, PA, 16509         US Mail (1st Class)           24800         LORD CORPORATION, 2000 W GRANDYIEW BLVD, ERIE, PA, 16509         US Mail (1st Class)           24800         LUSKI, MICHAEL T, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LUSKI, MICHAEL T, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)	24800	LEMAC PACKAGING, 2121 MCKINLEY AVE., ERIE, PA, 16514-0788	US Mail (1st Class)
24800         LEONARD, ROSEANNA, 2929 RED FOX RUN, WARREN, OH, 44485         US Mail (tst class)           24800         LIBERTY SOLVENTS & CHEMICALS CO INC, 9429 RAVENNA RD, TWINSBURG, OH, 44087         US Mail (tst class)           24800         LINTY, JENNIFER, 222 LCIBY OSBORNE RD, SOLVIHINGTON, OH, 44470         US Mail (tst class)           24800         LINDE GAS LLC, 6055 ROCKSIDE WOODS BLVD, INDEPENDENCE, OH, 44131-2301         US Mail (tst class)           24800         LINDERO AS ENCISCE, 1605 NORTHIVEW AVE, BARBERTON, OH, 44203-7197         US Mail (tst class)           24800         LIPPERT, GRACE E, 5017 OLD STATE RD, WEST FARMINGTON, OH, 44491         US Mail (tst class)           24800         LORD STAR QUALITY SVS INC, 1117 EAGLE RIDGE DR. EL PASO, TX, 79912         US Mail (tst class)           24800         LORD CORPORATION, 2000 W GRANDVIEW BLVD, ERIE, PA, 16509         US Mail (tst class)           24800         LORD CORPORATION, 2000 W GRANDVIEW BLVD, ERIE, PA, 16509         US Mail (tst class)           24800         LORD CORPORATION, 2000 W GRANDVIEW BLVD, ERIE, PA, 16509         US Mail (tst class)           24800         LUCIK, JANET E, 457 ELDON DR, WARREN, OH, 44483         US Mail (tst class)           24800         LUCIK, MICHAEL T, 457 ELDON DR, WARREN, OH, 44483         US Mail (tst class)           24800         MAC BANCHINE, 14725 SISSON ROAD, CHARDON, OH, 44024         US Mail (tst class)	24800	LENARD TOOL & MACHINE, 3125 E. ALLERTON AVE., ST FRANCIS, WI, 53235	US Mail (1st Class)
24800         LIBERTY SOLVENTS & CHEMICALS CO INC, 9429 RAVENNA RD, TWINSBURG, OH, 44087         US Mail (tst class)           24800         LICHTY, JENNIFER, 2822 LICIBY OSBORNE RD, SOUTHINGTON, OH, 44437         US Mail (tst class)           24800         LINGE GAS LLG, 6055 ROCKSIDE WOODS BLVD, INDEPENDENCE, OH, 44131-2301         US Mail (tst class)           24800         LINIFORM SERVICE, 1050 NORTHYIEW AVE, BARBERTON, OH, 44203-7197         US Mail (tst class)           24800         LINIFORM SERVICE, 1050 NORTHYIEW AVE, BARBERTON, OH, 44491         US Mail (tst class)           24800         LIONE STAR QUALITY SVS INC, 1117 EAGLE RIDGE DR, EL PASO, TX, 79912         US Mail (tst class)           24800         LOPEZ, JAMES D, 501 PARKMAN RD NW, WARREN, OH, 44485         US Mail (tst class)           24800         LOPG CORPORATION, 2000 W GRANDVIEW BLVD, ERIE, PA, 16509         US Mail (tst class)           24800         LOSH, DAYTON, 461 BELMONT NE, WARREN, OH, 44483         US Mail (tst class)           24800         LOSH, MICHAEL E, 457 ELDON DR, WARREN, OH, 44483         US Mail (tst class)           24800         LICKIK, JANET E, 457 ELDON DR, WARREN, OH, 44483         US Mail (tst class)           24800         LYDEN OIL COMPANY, 2649 TRACX ROAD, NORTHWOOD, OH, 43619         US Mail (tst class)           24800         MA S B MACHINE, 14725 SISSON ROAD, CHARDON, OH, 44024         US Mail (tst class)           24800 </td <td>24800</td> <td>LEO'S PNEUMATICS, 1821 FRANKLIN AVE, ERIE, PA, 16510</td> <td>US Mail (1st Class)</td>	24800	LEO'S PNEUMATICS, 1821 FRANKLIN AVE, ERIE, PA, 16510	US Mail (1st Class)
24800         LICHTY, JENNIFER, 2822 LCIBY OSBORNE RD, SOUTHINGTON, OH, 44470         US Mail (tst Class)           24800         LINDE GAS LLC, 6055 ROCKSIDE WOODS BLVD, INDEPENDENCE, OH, 44131-2301         US Mail (tst Class)           24800         LINFAR MOLD & ENDIKERING, 34455 GLENDALE, LIVONIA, MI, 48150         US Mail (tst Class)           24800         LINFORM SERVICE, 1050 NORTHVIEW AVE, BARBERTON, OH, 44203-7197         US Mail (tst Class)           24800         LIPPERT, GRACE E, 5017 OLD STATE RD, WEST FARMINGTON, OH, 44491         US Mail (tst Class)           24800         LOPEZ, JAMES D, 501 PARKMAN RD NW, WARREN, OH, 44495         US Mail (tst Class)           24800         LODEZ, JAMES D, 501 PARKMAN RD NW, WARREN, OH, 44483         US Mail (tst Class)           24800         LODG, DORPORATION, 2000 W. GRANDVIEW BLVD, ERIE, PA, 16509         US Mail (tst Class)           24800         LOUGH, DAYTON, 461 BELMONT NE, WARREN, OH, 44483         US Mail (tst Class)           24800         LUCIK, JANET E, 457 ELDON DR, WARREN, OH, 44483         US Mail (tst Class)           24800         LUCIK, MICHAEL T, 457 ELDON DR, WARREN, OH, 44483         US Mail (tst Class)           24800         LVDEN OIL COMPANY, 2649 TRACX ROAD, NORTHWOOD, OH, 43619         US Mail (tst Class)           24800         MAS B MACHINE, 44725 SISSON ROAD, CHARDON, OH, 44024         US Mail (tst Class)           24800         MAC SUTO	24800	LEONARD, ROSEANNA, 2929 RED FOX RUN, WARREN, OH, 44485	US Mail (1st Class)
24800         LINDE GAS LLC, 8055 ROCKSIDE WOODS BLVD, INDEPENDENCE, OH, 44131-2301         US Mail (1st Class)           24800         LINIFORM SERVICE, 1050 NORTHYIEW AVE., BARBERTON, OH, 44203-7197         US Mail (1st Class)           24800         LINIFORM SERVICE, 1050 NORTHYIEW AVE., BARBERTON, OH, 44203-7197         US Mail (1st Class)           24800         LOPEZ, GRACE E, 5017 OLD STATE RD, WEST FARMINGTON, OH, 44491         US Mail (1st Class)           24800         LOPEZ, JAMES D, 501 PARKMAN RD NW, WARREN, OH, 44485         US Mail (1st Class)           24800         LOPEZ, JAMES D, 501 PARKMAN RD NW, WARREN, OH, 44483         US Mail (1st Class)           24800         LORD CORPORATION, 2000 W GRANDVIEW BLVD, ERIE, PA, 16509         US Mail (1st Class)           24800         LORD CORPORATION, 461 BELMONT NE, WARREN, OH, 44483         US Mail (1st Class)           24800         LORD KIN, JANET E, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LUCIK, MICHAEL T, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LVEDEN OIL COMPANY, 2649 TRACX ROAD, NORTHYMOD, OH, 43619         US Mail (1st Class)           24800         MAC SMITH, 5201 VIRGINIA WAY, BRENTWOOD, TN, 37024-5094         US Mail (1st Class)           24800         MAC SMITH, 5201 VIRGINIA WAY, BRENTWOOD, TN, 37024-5094         US Mail (1st Class)           24800         MACH	24800	LIBERTY SOLVENTS & CHEMICALS CO INC, 9429 RAVENNA RD, TWINSBURG, OH, 44087	US Mail (1st Class)
24800         LINEAR MOLD & ENGINEERING, 34435 GLENDALE, LIVONIA, MI, 48150         US Mail (1st Class)           24800         LINIFORM SERVICE, 1050 NORTH/WEW AVE., BARBERTON, OH, 44203-7197         US Mail (1st Class)           24800         LIPPERT, GRACE E, 5017 OLD STATE RD, WEST FARMINGTON, OH, 44491         US Mail (1st Class)           24800         LONE STAR QUALITY SVS INC., 1117 EAGLE RIDGE DR. EL PASO, TX, 79912         US Mail (1st Class)           24800         LORD CORPORATION, 2000 W GRANDVIEW BLVD, ERIE, PA, 16509         US Mail (1st Class)           24800         LORD CORPORATION, 2000 W GRANDVIEW BLVD, ERIE, PA, 16509         US Mail (1st Class)           24800         LORD, DATTON, 461 BELMONT NE, WARREN, OH, 44483         US Mail (1st Class)           24800         LUCIK, JANET E, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LUCIK, MICHAEL T, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LYDEN OIL COMPANY, 2649 TRACX ROAD, NORTH-WOOD, OH, 43619         US Mail (1st Class)           24800         M. EE SMITH, 5201 VIRIGINIA WAY, BRENTWOOD, TN, 37024-5094         US Mail (1st Class)           24800         MAC AUTOMATION CONCEPTS, INC., 1760 KILKENNY COURT, WOODSTOCK, IL, 60098         US Mail (1st Class)           24800         MACEY, RUSSELL D, 6425 NORTH RIDGE EAST, GENEVA, OH, 44041         US Mail (1st Class)           24800 </td <td>24800</td> <td>LICHTY, JENNIFER, 2822 LCIBY OSBORNE RD, SOUTHINGTON, OH, 44470</td> <td>US Mail (1st Class)</td>	24800	LICHTY, JENNIFER, 2822 LCIBY OSBORNE RD, SOUTHINGTON, OH, 44470	US Mail (1st Class)
24800         LINFORM SERVICE, 1050 NORTHYJEW AVE., BARBERTON, OH, 44491         US Mail (1st Class)           24800         LIPPERT, GRACE E, 5017 OLD STATE RD, WEST FARMINGTON, OH, 44491         US Mail (1st Class)           24800         LOPEZ, JAMIES D, 501 PARKMAN RD NW, WARREN, OH, 44485         US Mail (1st Class)           24800         LOPEZ, JAMIES D, 501 PARKMAN RD NW, WARREN, OH, 44483         US Mail (1st Class)           24800         LODEN CORPORATION, 2000 W GRANDVIEW BLVD, ERIE, PA, 16509         US Mail (1st Class)           24800         LOUGH, DAYTON, 461 BELMONT NE, WARREN, OH, 44483         US Mail (1st Class)           24800         LUCIK, JANET E, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LUCIK, JANET E, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LYDEN OIL COMPANY, 2649 TRACX ROAD, NORTHWOOD, OH, 43619         US Mail (1st Class)           24800         M. EE SMITH, 5201 VIRGINIA WAY, BRENTWOOD, TN, 37024-5094         US Mail (1st Class)           24800         M. EE SMITH, 5201 VIRGINIA WAY, BRENTWOOD, TN, 37024-5094         US Mail (1st Class)           24800         MACE WILLIAM SAMATION CONCEPTS, INC., 1750 KILKENNY COURT, WOODSTOCK, IL, 60098         US Mail (1st Class)           24800         MACH MOLD INC, 360 URBANDALE AVE, BENTON HARBOR, MI, 49022         US Mail (1st Class)           24800         MACH M	24800	LINDE GAS LLC, 6055 ROCKSIDE WOODS BLVD, INDEPENDENCE, OH, 44131-2301	US Mail (1st Class)
24800         LIPPERT, GRACE E, 5017 OLD STATE RD, WEST FARMINGTON, OH, 44491         US Mail (1st Class)           24800         LONE STAR QUALITY SVS INC, 1117 EAGLE RIDGE DR, EL PASO, TX, 79912         US Mail (1st Class)           24800         LOPEZ, JAMES D, 501 PARKMAN RD NW, WARREN, OH, 44485         US Mail (1st Class)           24800         LORD CORPORATION, 2000 W GRANDVIEW BLVD, ERIE, PA, 16509         US Mail (1st Class)           24800         LOUGH, DAYTON, 461 BELMONT NE, WARREN, OH, 44483         US Mail (1st Class)           24800         LUCIK, JANET E, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LUCIK, MICHAEL T, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LYDEN OIL COMPANY, 2649 TRACX ROAD, NORTHWOOD, OH, 43619         US Mail (1st Class)           24800         M. EE SMITH, 5201 VIRGINIA WAY, BRENTWOOD, TN, 37024-5094         US Mail (1st Class)           24800         MACE AUTOMATION CONCEPTS, INC., 1760 KILKENNY COURT, WOODSTOCK, IL, 60098         US Mail (1st Class)           24800         MACEY, RUSSELL D, 6425 NORTH RIDGE EAST, GENEVA, OH, 44041         US Mail (1st Class)           24800         MACH MOLD INC, 360 URBANDALE AVE, BENTON HARDER, MI, 49022         US Mail (1st Class)           24800         MAG, INC., DEPT, #216401, PO BOX 67000, DETROIT, MI, 48267-2164         US Mail (1st Class)           24800	24800	LINEAR MOLD & ENGINEERING, 34435 GLENDALE, LIVONIA, MI, 48150	US Mail (1st Class)
24800         LONE STAR QUALITY SVS INC, 1117 EAGLE RIDGE DR, EL PASO, TX, 79912         US Mail (1st Class)           24800         LOPEZ, JAMES D, 501 PARKMAN RO NW, WARREN, OH, 44485         US Mail (1st Class)           24800         LORD CORPORATION, 2000 W GRANDVIEW BLVD, ERIE, PA, 16509         US Mail (1st Class)           24800         LOUGH, DAYTON, 461 BELMONT NE, WARREN, OH, 44483         US Mail (1st Class)           24800         LUCIK, JANET E, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LUCIK, MICHAEL T, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LUCIK, MICHAEL T, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LUCIK, MICHAEL T, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LUCIK, MICHAEL T, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         M. LEE SMITH, 5201 VIRGINIA WAY, BRENTWOOD, TN, 37024-5094         US Mail (1st Class)           24800         M. LEE SMITH, 5201 VIRGINIA WAY, BRENTWOOD, TN, 37024-5094         US Mail (1st Class)           24800         MACCH MOLD INC, 360 URBANDALE AVE, BENTON HARBOR, MI, 49022         US Mail (1st Class)           24800         MACH MOLD INC, 360 URBANDALE AVE, BENTON HARBOR, MI, 49262         US Mail (1st Class)           24800         MACH MOLD INC, 360 URBANDALE AVE, BENTON	24800	LINIFORM SERVICE, 1050 NORTHVIEW AVE., BARBERTON, OH, 44203-7197	US Mail (1st Class)
24800         LOPEZ, JAMES D, 501 PARKMAN RD NW, WARREN, OH, 44485         US Mail (1st Class)           24800         LOUGH, DAYTON, 461 BELMONT NE, WARREN, OH, 44483         US Mail (1st Class)           24800         LOUGH, DAYTON, 461 BELMONT NE, WARREN, OH, 44483         US Mail (1st Class)           24800         LUCIK, JANET E, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LUCIK, JANET E, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LUCICK, MICHAEL T, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LYDEN OIL COMPANY, 2649 TRACX ROAD, NORTHWOOD, OH, 43619         US Mail (1st Class)           24800         M. S. MACHINE, 14725 SISSON ROAD, CHARDON, OH, 44024         US Mail (1st Class)           24800         MA.C. BMITH, 5201 VIRGINIA WAY, BRENTWOOD, TN, 37024-5094         US Mail (1st Class)           24800         MACCEY, RUSSELL D, 6425 NORTH RIDGE EAST, GENEVA, OH, 44011         US Mail (1st Class)           24800         MACH MOLD INC, 380 URBANDALE AVE, BENTON HARBOR, MI, 49022         US Mail (1st Class)           24800         MAGNIER PRODUCTS, INC., 11 CROZERVILLE RD., ASTON, PA, 19014-1431         US Mail (1st Class)           24800         MAGNETIC SCREW MACHINE, 23241 ST. RT. 37, MARYSVILLE, OH, 43904         US Mail (1st Class)           24800         MARLE FILTER SYSTEMS, 966 BUTLER DR.,	24800	LIPPERT, GRACE E, 5017 OLD STATE RD, WEST FARMINGTON, OH, 44491	US Mail (1st Class)
24800         LORD CORPORATION, 2000 W GRANDVIEW BLVD, ERIE, PA, 16509         US Mail (1st Class)           24800         LOUGH, DAYTON, 461 BELMONT NE, WARREN, OH, 44483         US Mail (1st Class)           24800         LRS INC COMPANY, 41190 JOY RD, PLYMOUTH, MI, 48170-4634         US Mail (1st Class)           24800         LUCIK, JANET E, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LUCIK, MICHAEL T, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LYDEN OIL COMPANY, 2649 TRACX ROAD, NORTHWOOD, OH, 43619         US Mail (1st Class)           24800         M. S MACHINE, 14725 SISSON ROAD, CHARDON, OH, 44024         US Mail (1st Class)           24800         M. LEE SMITH, 5201 VIRGINIA WAY, BRENTWOOD, TN, 37024-5094         US Mail (1st Class)           24800         MAC AUTOMATION CONCEPTS, INC., 1760 KILKENNY COURT, WOODSTOCK, IL, 60098         US Mail (1st Class)           24800         MACCH, RUSSELL D, 6425 NORTH RIDGE EAST, GENEVA, OH, 44041         US Mail (1st Class)           24800         MAC, DEPT, #216401, PO BOX 67000, DETROIT, MI, 48267-2164         US Mail (1st Class)           24800         MAGNETIC SCREW MACHINE, 23241 ST. RT. 37, MARYSVILLE, OH, 43040         US Mail (1st Class)           24800         MALLE FILTER SYSTEMS, 906 BUTLER DR., MURFREESBORD, TN., 37127         US Mail (1st Class)           24800         MALLE F	24800	LONE STAR QUALITY SVS INC, 1117 EAGLE RIDGE DR, EL PASO, TX, 79912	US Mail (1st Class)
24800         LOUGH, DAYTON, 461 BELMONT NE, WARREN, OH, 44483         US Mail (1st Class)           24800         LRS INC COMPANY, 41190 JOY RD, PLYMOUTH, MI, 48170-4634         US Mail (1st Class)           24800         LUCIK, JANAET E, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LUCIK, MICHAEL T, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LYDEN OIL COMPANY, 2649 TRACX ROAD, NORTHWOOD, OH, 43619         US Mail (1st Class)           24800         M & S MACHINE, 14725 SISSON ROAD, CHARDON, OH, 44024         US Mail (1st Class)           24800         MA. LEE SMITH, 5201 VIRGINIA WAY, BRENTWOOD, TN, 37024-5094         US Mail (1st Class)           24800         MACCAUTOMATION CONCEPTS, INC., 1760 KILKENNY COURT, WOODSTOCK, IL, 60098         US Mail (1st Class)           24800         MACCEY, RUSSELL D, 6425 NORTH RIDGE EAST, GENEVA, OH, 44041         US Mail (1st Class)           24800         MACH MOLD INC, 360 URBANDALE AVE, BENTON HARBOR, MI, 49022         US Mail (1st Class)           24800         MAG, INC., DEPT. #216401, PO BOX 67000, DETROIT, MI, 48267-2164         US Mail (1st Class)           24800         MAGNETIC SCREW MACHINE, 23241 ST, RT. 37, MARYSVILLE, OH, 43040         US Mail (1st Class)           24800         MACHE FILLER SYSTEMS, 906 BUTLER DR., MURFREESBORO, TN, 37127         US Mail (1st Class)           24800	24800	LOPEZ, JAMES D, 501 PARKMAN RD NW, WARREN, OH, 44485	US Mail (1st Class)
24800         LRS INC COMPANY, 41190 JOY RD, PLYMOUTH, MI, 48170-4634         US Mail (1st Class)           24800         LUCIK, JANET E, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LUCIK, JANET E, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LYDEN OIL COMPANY, 2649 TRACX ROAD, NORTHWOOD, OH, 43619         US Mail (1st Class)           24800         M & S MACHINE, 14725 SISSON ROAD, CHARDON, OH, 44024         US Mail (1st Class)           24800         M. LEE SMITH, 5201 VIRGINIA WAY, BRENTWOOD, TN, 37024-5094         US Mail (1st Class)           24800         MACA AUTOMATION CONCEPTS, INC., 1760 KILKENNY COURT, WOODSTOCK, IL, 60098         US Mail (1st Class)           24800         MACEY, RUSSELL D, 6425 NORTH RIDGE EAST, GENEVA, OH, 44041         US Mail (1st Class)           24800         MACEY, RUSSELL D, 6425 NORTH RIDGE EAST, GENEVA, OH, 44041         US Mail (1st Class)           24800         MAG, INC., DEPT. #216401, PO BOX 67000, DETROIT, MI, 48267-2164         US Mail (1st Class)           24800         MAG, INC., DEPT. #216401, PO BOX 67000, DETROIT, MI, 48267-2164         US Mail (1st Class)           24800         MAGURE PRODUCTS, INC., 11 CROZERVILLE RD., ASTON, PA, 19014-1431         US Mail (1st Class)           24800         MAGURE PRODUCTS, INC., 23710 SAMOSET TRAIL, SOUTHFIELD, MI, 48034         US Mail (1st Class)           248	24800	LORD CORPORATION, 2000 W GRANDVIEW BLVD, ERIE, PA, 16509	US Mail (1st Class)
24800         LUCIK, JANET E, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LUCIK, MICHAEL T, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LYDEN OIL COMPANY, 2649 TRACX ROAD, NORTHWOOD, OH, 43619         US Mail (1st Class)           24800         M & S MACHINE, 14725 SISSON ROAD, CHARDON, OH, 44024         US Mail (1st Class)           24800         M. LEE SMITH, 5201 VIRGINIA WAY, BRENTWOOD, TN, 37024-5094         US Mail (1st Class)           24800         MAC AUTOMATION CONCEPTS, INC., 1760 KILKENNY COURT, WOODSTOCK, IL, 60098         US Mail (1st Class)           24800         MACEY, RUSSELL D, 6425 NORTH RIDGE EAST, GENEVA, OH, 44041         US Mail (1st Class)           24800         MACH MOLD INC., 360 URBANDALE AVE, BENTON HARBOR, MI, 49022         US Mail (1st Class)           24800         MAG, INC., DEPT. #216401, PO BOX 67000, DETROIT, MI, 48267-2164         US Mail (1st Class)           24800         MAGURE PRODUCTS, INC., 11 CROZERVILLE RD., ASTON, PA, 19014-1431         US Mail (1st Class)           24800         MAGURE PRODUCTS, INC., 23710 SAMOSET TRAIL, SOUTHFIELD, MI, 48034         US Mail (1st Class)           24800         MALONE, PATRICIA A, 258 FEDERAL ST NW, WARREN, OH, 44483         US Mail (1st Class)           24800         MAROSEMENT RECRUITERS TALLAHASSEE, 743 EAST TENNESSEE ST., TALLAHASSEE, FL, 32308         US Mail (1st Class)	24800	LOUGH, DAYTON, 461 BELMONT NE, WARREN, OH, 44483	US Mail (1st Class)
24800         LUCIK, MICHAEL T, 457 ELDON DR, WARREN, OH, 44483         US Mail (1st Class)           24800         LYDEN OIL COMPANY, 2649 TRACX ROAD, NORTHWOOD, OH, 43619         US Mail (1st Class)           24800         M & S MACHINE, 14725 SISSON ROAD, CHARDON, OH, 44024         US Mail (1st Class)           24800         M. LEE SMITH, 5201 VIRGINIA WAY, BRENTWOOD, TN, 37024-5094         US Mail (1st Class)           24800         MAC AUTOMATION CONCEPTS, INC., 1760 KILKENNY COURT, WOODSTOCK, IL, 60098         US Mail (1st Class)           24800         MACEY, RUSSELL D, 6425 NORTH RIDGE EAST, GENEVA, OH, 44041         US Mail (1st Class)           24800         MACH MOLD INC, 360 URBANDALE AVE, BENTON HARBOR, MI, 49022         US Mail (1st Class)           24800         MAG, INC., DEPT, #216401, PO BOX 67000, DETROIT, MI, 48267-2164         US Mail (1st Class)           24800         MAGUIRE PRODUCTS, INC., 11 CROZERVILLE RD., ASTON, PA, 19014-1431         US Mail (1st Class)           24800         MAGUIRE PRODUCTS, INC., 23710 SAMOSET TRAIL, SOUTHFIELD, MI, 48034         US Mail (1st Class)           24800         MALONE, PATRICIA A, 258 FEDERAL ST NW, WARREN, OH, 44483         US Mail (1st Class)           24800         MANAGEMENT RECRUITERS TALLAHASSEE, 743 EAST TENNESSEE ST., TALLAHASSEE, FL, 32308         US Mail (1st Class)           24800         MARCHIN, MINDY, 2715 RUSSELL RD, ASHTABULA, OH, 444004         US Mail (1st Class)     <	24800	LRS INC COMPANY, 41190 JOY RD, PLYMOUTH, MI, 48170-4634	US Mail (1st Class)
24800         LYDEN OIL COMPANY, 2649 TRACX ROAD, NORTHWOOD, OH, 43619         US Mail (1st Class)           24800         M & S MACHINE, 14725 SISSON ROAD, CHARDON, OH, 44024         US Mail (1st Class)           24800         M. LEE SMITH, 5201 VIRGINIA WAY, BRENTWOOD, TN, 37024-5094         US Mail (1st Class)           24800         MAC AUTOMATION CONCEPTS, INC., 1760 KILKENNY COURT, WOODSTOCK, IL, 60098         US Mail (1st Class)           24800         MACEY, RUSSELL D, 6425 NORTH RIDGE EAST, GENEVA, OH, 44041         US Mail (1st Class)           24800         MACH MOLD INC, 360 URBANDALE AVE, BENTON HARBOR, MI, 49022         US Mail (1st Class)           24800         MAG, INC., DEPT. #216401, PO BOX 67000, DETROIT, MI, 48267-2164         US Mail (1st Class)           24800         MAGNETIC SCREW MACHINE, 23241 ST. RT. 37, MARYSVILLE, OH, 43040         US Mail (1st Class)           24800         MAGUIRE PRODUCTS, INC., 11 CROZERVILLE RD., ASTON, PA, 19014-1431         US Mail (1st Class)           24800         MALORE PATRICIA A, 258 FEDERAL ST NW, WARREN, OH, 44483         US Mail (1st Class)           24800         MALONE, PATRICIA A, 258 FEDERAL ST NW, WARREN, OH, 44485         US Mail (1st Class)           24800         MARNES BUPERMARKETS #74, 223 AUKERMAN STREET, EATON, OH, 45320         US Mail (1st Class)           24800         MARSH SUPERMARKETS #74, 223 AUKERMAN STREET, EATON, OH, 45345         US Mail (1st Class)	24800	LUCIK, JANET E, 457 ELDON DR, WARREN, OH, 44483	US Mail (1st Class)
24800         M. & S. MACHINE, 14725 SISSON ROAD, CHARDON, OH, 44024         US Mail (1st Class)           24800         M. LEE SMITH, 5201 VIRGINIA WAY, BRENTWOOD, TN, 37024-5094         US Mail (1st Class)           24800         MAC AUTOMATION CONCEPTS, INC., 1760 KILKENNY COURT, WOODSTOCK, IL, 60098         US Mail (1st Class)           24800         MACEY, RUSSELL D, 6425 NORTH RIDGE EAST, GENEVA, OH, 44041         US Mail (1st Class)           24800         MACH MOLD INC, 360 URBANDALE AVE, BENTON HARBOR, MI, 49022         US Mail (1st Class)           24800         MAG, INC., DEPT. #216401, PO BOX 67000, DETROIT, MI, 48267-2164         US Mail (1st Class)           24800         MAGUIRE PRODUCTS, INC., 11 CROZERVILLE RD., ASTON, PA, 19014-1431         US Mail (1st Class)           24800         MAGUIRE PRODUCTS, INC., 11 CROZERVILLE RD., ASTON, PA, 19014-1431         US Mail (1st Class)           24800         MALONE, PATRICIA A, 258 FEDERAL ST NW, WARREN, OH, 44483         US Mail (1st Class)           24800         MALONE, PATRICIA A, 258 FEDERAL ST NW, WARREN, OH, 44487         US Mail (1st Class)           24800         MAPLE CITY PRODUCTS, 170 BENEDICT AVE, NORWALK, OH, 44857         US Mail (1st Class)           24800         MARSHO, MINDY, 2715 RUSSELL RD, ASHTABULA, OH, 44004         US Mail (1st Class)           24800         MARSH SUPERMARKETS #74, 223 AUKERMAN STREET, EATON, OH, 45320         US Mail (1st Class)	24800	LUCIK, MICHAEL T, 457 ELDON DR, WARREN, OH, 44483	US Mail (1st Class)
24800         M. LEE SMITH, 5201 VIRGINIA WAY, BRENTWOOD, TN, 37024-5094         US Mail (1st Class)           24800         MAC AUTOMATION CONCEPTS, INC., 1760 KILKENNY COURT, WOODSTOCK, IL, 60098         US Mail (1st Class)           24800         MACEY, RUSSELL D, 6425 NORTH RIDGE EAST, GENEVA, OH, 44041         US Mail (1st Class)           24800         MACH MOLD INC, 360 URBANDALE AVE, BENTON HARBOR, MI, 49022         US Mail (1st Class)           24800         MAG, INC., DEPT. #216401, PO BOX 67000, DETROIT, MI, 48267-2164         US Mail (1st Class)           24800         MAGNETIC SCREW MACHINE, 23241 ST. RT. 37, MARYSVILLE, OH, 43040         US Mail (1st Class)           24800         MAGUIRE PRODUCTS, INC., 11 CROZERVILLE RD., ASTON, PA, 19014-1431         US Mail (1st Class)           24800         MALORE PRODUCTS, INC., 23710 SAMOSET TRAIL, SOUTHFIELD, MI, 48034         US Mail (1st Class)           24800         MAJOR MICRO SYSTEMS, INC., 23710 SAMOSET TRAIL, SOUTHFIELD, MI, 48034         US Mail (1st Class)           24800         MANAGEMENT RECRUITERS TALLAHASSEE, 743 EAST TENNESSEE ST., TALLAHASSEE, FL, 32308         US Mail (1st Class)           24800         MARINO, MINDY, 2715 RUSSELL EN, ASHTABULA, OH, 444064         US Mail (1st Class)           24800         MARSH SUPERMARKETS #74, 223 AUKERMAN STREET, EATON, OH, 45320         US Mail (1st Class)           24800         MARTIN FLUID POWER, 84 MINNESOTA DR, TROY, MI, 48093	24800	LYDEN OIL COMPANY, 2649 TRACX ROAD, NORTHWOOD, OH, 43619	US Mail (1st Class)
24800         MAC AUTOMATION CONCEPTS, INC., 1760 KILKENNY COURT, WOODSTOCK, IL, 60098         US Mail (1st Class)           24800         MACEY, RUSSELL D, 6425 NORTH RIDGE EAST, GENEVA, OH, 44041         US Mail (1st Class)           24800         MACH MOLD INC, 360 URBANDALE AVE, BENTON HARBOR, MI, 49022         US Mail (1st Class)           24800         MAG, INC., DEPT. #216401, PO BOX 67000, DETROIT, MI, 48267-2164         US Mail (1st Class)           24800         MAGNETIC SCREW MACHINE, 23241 ST. RT. 37, MARYSVILLE, OH, 43040         US Mail (1st Class)           24800         MAGUIRE PRODUCTS, INC., 11 CROZERVILLE RD., ASTON, PA, 19014-1431         US Mail (1st Class)           24800         MAHLE FILTER SYSTEMS, 906 BUTLER DR., MURFREESBORO, TN, 37127         US Mail (1st Class)           24800         MALONE, PATRICIA A, 258 FEDERAL ST NW, WARREN, OH, 44483         US Mail (1st Class)           24800         MALONE, PATRICIA A, 258 FEDERAL ST NW, WARREN, OH, 44483         US Mail (1st Class)           24800         MARJE CITY PRODUCTS, 170 BENEDICT AVE, NORWALK, OH, 44857         US Mail (1st Class)           24800         MARSH SUPERMARKETS #74, 223 AUKERMAN STREET, EATON, OH, 45320         US Mail (1st Class)           24800         MARTIN FLUID POWER, 84 MINNESOTA DR, TROY, MI, 48083         US Mail (1st Class)           24800         MARTIN SUPPLY, 200 E APPLETON AVE, SHEFFIELD, AL, 35660         US Mail (1st Class) <t< td=""><td>24800</td><td>M &amp; S MACHINE, 14725 SISSON ROAD, CHARDON, OH, 44024</td><td>US Mail (1st Class)</td></t<>	24800	M & S MACHINE, 14725 SISSON ROAD, CHARDON, OH, 44024	US Mail (1st Class)
24800       MACEY, RUSSELL D, 6425 NORTH RIDGE EAST, GENEVA, OH, 44041       US Mail (1st Class)         24800       MACH MOLD INC, 360 URBANDALE AVE, BENTON HARBOR, MI, 49022       US Mail (1st Class)         24800       MAG, INC., DEPT. #216401, PO BOX 67000, DETROIT, MI, 48267-2164       US Mail (1st Class)         24800       MAGNETIC SCREW MACHINE, 23241 ST. RT. 37, MARYSVILLE, OH, 43040       US Mail (1st Class)         24800       MAGUIRE PRODUCTS, INC., 11 CROZERVILLE RD., ASTON, PA, 19014-1431       US Mail (1st Class)         24800       MAHLE FILTER SYSTEMS, 906 BUTLER DR., MURFREESBORO, TN, 37127       US Mail (1st Class)         24800       MAJOR MICRO SYSTEMS, INC., 23710 SAMOSET TRAIL, SOUTHFIELD, MI, 48034       US Mail (1st Class)         24800       MALONE, PATRICIA A, 258 FEDERAL ST NW, WARREN, OH, 44483       US Mail (1st Class)         24800       MANCEMENT RECRUITERS TALLAHASSEE, 743 EAST TENNESSEE ST., TALLAHASSEE, FL, 32308       US Mail (1st Class)         24800       MAPLE CITY PRODUCTS, 170 BENEDICT AVE, NORWALK, OH, 444857       US Mail (1st Class)         24800       MARSH SUPERMARKETS #74, 223 AUKERMAN STREET, EATON, OH, 45320       US Mail (1st Class)         24800       MARTIN STREASICK, 290 MCCORMICK, LAPEER, MI, 48446       US Mail (1st Class)         24800       MARTIN SUPPLY, 200 E APPLETON AVE, SHEFFIELD, AL, 35660       US Mail (1st Class)         24800       MARTIN BEVERLY	24800	M. LEE SMITH, 5201 VIRGINIA WAY, BRENTWOOD, TN, 37024-5094	US Mail (1st Class)
24800         MACH MOLD INC, 360 URBANDALE AVE, BENTON HARBOR, MI, 49022         US Mail (1st Class)           24800         MAG, INC., DEPT. #216401, PO BOX 67000, DETROIT, MI, 48267-2164         US Mail (1st Class)           24800         MAGNETIC SCREW MACHINE, 23241 ST. RT. 37, MARYSVILLE, OH, 43040         US Mail (1st Class)           24800         MAGUIRE PRODUCTS, INC., 11 CROZERVILLE RD., ASTON, PA, 19014-1431         US Mail (1st Class)           24800         MAHLE FILTER SYSTEMS, 906 BUTLER DR., MURFREESBORO, TN, 37127         US Mail (1st Class)           24800         MAJOR MICRO SYSTEMS, INC., 23710 SAMOSET TRAIL, SOUTHFIELD, MI, 48034         US Mail (1st Class)           24800         MALONE, PATRICIA A, 258 FEDERAL ST NW, WARREN, OH, 44483         US Mail (1st Class)           24800         MANAGEMENT RECRUITERS TALLAHASSEE, 743 EAST TENNESSEE ST., TALLAHASSEE, FL, 32308         US Mail (1st Class)           24800         MAPLE CITY PRODUCTS, 170 BENEDICT AVE, NORWALK, OH, 44857         US Mail (1st Class)           24800         MARSINO, MINDY, 2715 RUSSELL RD, ASHTABULA, OH, 44004         US Mail (1st Class)           24800         MARSH SUPERMARKETS #74, 223 AUKERMAN STREET, EATON, OH, 45320         US Mail (1st Class)           24800         MARTIN FLUID POWER, 84 MINNESOTA DR, TROY, MI, 48083         US Mail (1st Class)           24800         MARTIN STREASICK, 290 MCCORMICK, LAPEER, MI, 48046         US Mail (1st Class)	24800	MAC AUTOMATION CONCEPTS, INC., 1760 KILKENNY COURT, WOODSTOCK, IL, 60098	US Mail (1st Class)
24800       MAG, INC., DEPT. #216401, PO BOX 67000, DETROIT, MI, 48267-2164       US Mail (1st Class)         24800       MAGNETIC SCREW MACHINE, 23241 ST. RT. 37, MARYSVILLE, OH, 43040       US Mail (1st Class)         24800       MAGUIRE PRODUCTS, INC., 11 CROZERVILLE RD., ASTON, PA, 19014-1431       US Mail (1st Class)         24800       MAHLE FILTER SYSTEMS, 906 BUTLER DR., MURFREESBORO, TN, 37127       US Mail (1st Class)         24800       MAJOR MICRO SYSTEMS, INC., 23710 SAMOSET TRAIL, SOUTHFIELD, MI, 48034       US Mail (1st Class)         24800       MALONE, PATRICIA A, 258 FEDERAL ST NW, WARREN, OH, 44483       US Mail (1st Class)         24800       MANGEMENT RECRUITERS TALLAHASSEE, 743 EAST TENNESSEE ST., TALLAHASSEE, FL, 32308       US Mail (1st Class)         24800       MAPLE CITY PRODUCTS, 170 BENEDICT AVE, NORWALK, OH, 44857       US Mail (1st Class)         24800       MARRINO, MINDY, 2715 RUSSELL RD, ASHTABULA, OH, 44004       US Mail (1st Class)         24800       MARSH SUPERMARKETS #74, 223 AUKERMAN STREET, EATON, OH, 45320       US Mail (1st Class)         24800       MARTIN FLUID POWER, 84 MINNESOTA DR, TROY, MI, 48046       US Mail (1st Class)         24800       MARTIN STREASICK, 290 MCCORMICK, LAPEER, MI, 48446       US Mail (1st Class)         24800       MARTIN SUPPLY, 200 E APPLETON AVE, SHEFFIELD, AL, 35660       US Mail (1st Class)         24800       MARTIN, BEVERLY J, 639 ROU	24800	MACEY, RUSSELL D, 6425 NORTH RIDGE EAST, GENEVA, OH, 44041	US Mail (1st Class)
24800       MAGNETIC SCREW MACHINE, 23241 ST. RT. 37, MARYSVILLE, OH, 43040       US Mail (1st Class)         24800       MAGUIRE PRODUCTS, INC., 11 CROZERVILLE RD., ASTON, PA, 19014-1431       US Mail (1st Class)         24800       MAHLE FILTER SYSTEMS, 906 BUTLER DR., MURFREESBORO, TN, 37127       US Mail (1st Class)         24800       MAJOR MICRO SYSTEMS, INC., 23710 SAMOSET TRAIL, SOUTHFIELD, MI, 48034       US Mail (1st Class)         24800       MALONE, PATRICIA A, 258 FEDERAL ST NW, WARREN, OH, 44483       US Mail (1st Class)         24800       MARAGEMENT RECRUITERS TALLAHASSEE, 743 EAST TENNESSEE ST., TALLAHASSEE, FL, 32308       US Mail (1st Class)         24800       MAPLE CITY PRODUCTS, 170 BENEDICT AVE, NORWALK, OH, 44857       US Mail (1st Class)         24800       MARINO, MINDY, 2715 RUSSELL RD, ASHTABULA, OH, 44004       US Mail (1st Class)         24800       MARSH SUPERMARKETS #74, 223 AUKERMAN STREET, EATON, OH, 45320       US Mail (1st Class)         24800       MARTIN FLUID POWER, 84 MINNESOTA DR, TROY, MI, 48083       US Mail (1st Class)         24800       MARTIN STREASICK, 290 MCCORMICK, LAPEER, MI, 48446       US Mail (1st Class)         24800       MARTIN WELDING, 1472 W MAIN STREET, NEW LEBANON, OH, 45345       US Mail (1st Class)         24800       MARTIN, BEVERLY J, 639 ROUTE 534 NW, NEWTON FALLS, OH, 44444       US Mail (1st Class)         24800       MARUBENI SPECIALTY CHEMICA	24800	MACH MOLD INC, 360 URBANDALE AVE, BENTON HARBOR, MI, 49022	US Mail (1st Class)
24800       MAGUIRE PRODUCTS, INC., 11 CROZERVILLE RD., ASTON, PA, 19014-1431       US Mail (1st Class)         24800       MAHLE FILTER SYSTEMS, 906 BUTLER DR., MURFREESBORO, TN, 37127       US Mail (1st Class)         24800       MAJOR MICRO SYSTEMS, INC., 23710 SAMOSET TRAIL, SOUTHFIELD, MI, 48034       US Mail (1st Class)         24800       MALONE, PATRICIA A, 258 FEDERAL ST NW, WARREN, OH, 44483       US Mail (1st Class)         24800       MANAGEMENT RECRUITERS TALLAHASSEE, 743 EAST TENNESSEE ST., TALLAHASSEE, FL, 32308       US Mail (1st Class)         24800       MAPLE CITY PRODUCTS, 170 BENEDICT AVE, NORWALK, OH, 44857       US Mail (1st Class)         24800       MARINO, MINDY, 2715 RUSSELL RD, ASHTABULA, OH, 44004       US Mail (1st Class)         24800       MARSH SUPERMARKETS #74, 223 AUKERMAN STREET, EATON, OH, 45320       US Mail (1st Class)         24800       MARTIN FLUID POWER, 84 MINNESOTA DR, TROY, MI, 48083       US Mail (1st Class)         24800       MARTIN STREASICK, 290 MCCORMICK, LAPEER, MI, 48446       US Mail (1st Class)         24800       MARTIN WELDING, 1472 W MAIN STREET, NEW LEBANON, OH, 45345       US Mail (1st Class)         24800       MARTIN, BEVERLY J, 639 ROUTE 534 NW, NEWTON FALLS, OH, 44444       US Mail (1st Class)         24800       MARVIN, MARTHA B, PO BOX 453, LEAVITTSBURG, OH, 44430       US Mail (1st Class)         24800       MASTERMAN'S, 11 C ST., BUILDING 10,	24800	MAG, INC., DEPT. #216401, PO BOX 67000, DETROIT, MI, 48267-2164	US Mail (1st Class)
24800       MAHLE FILTER SYSTEMS, 906 BUTLER DR., MURRREESBORO, TN, 37127       US Mail (1st Class)         24800       MAJOR MICRO SYSTEMS, INC., 23710 SAMOSET TRAIL, SOUTHFIELD, MI, 48034       US Mail (1st Class)         24800       MALONE, PATRICIA A, 258 FEDERAL ST NW, WARREN, OH, 44483       US Mail (1st Class)         24800       MANAGEMENT RECRUITERS TALLAHASSEE, 743 EAST TENNESSEE ST., TALLAHASSEE, FL, 32308       US Mail (1st Class)         24800       MAPLE CITY PRODUCTS, 170 BENEDICT AVE, NORWALK, OH, 44857       US Mail (1st Class)         24800       MARINO, MINDY, 2715 RUSSELL RD, ASHTABULA, OH, 44004       US Mail (1st Class)         24800       MARSH SUPERMARKETS #74, 223 AUKERMAN STREET, EATON, OH, 45320       US Mail (1st Class)         24800       MARTIN FLUID POWER, 84 MINNESOTA DR, TROY, MI, 48083       US Mail (1st Class)         24800       MARTIN STREASICK, 290 MCCORMICK, LAPEER, MI, 48446       US Mail (1st Class)         24800       MARTIN SUPPLY, 200 E APPLETON AVE, SHEFFIELD, AL, 35660       US Mail (1st Class)         24800       MARTIN WELDING, 1472 W MAIN STREET, NEW LEBANON, OH, 45345       US Mail (1st Class)         24800       MARTIN, BEVERLY J, 639 ROUTE 534 NW, NEWTON FALLS, OH, 44444       US Mail (1st Class)         24800       MARUBENI SPECIALTY CHEMICALS INC, 10 BANK ST, STE 740, WHITE PLAINS, NY, 10606       US Mail (1st Class)         24800       MASTERMAN'S, 11 C ST.	24800	MAGNETIC SCREW MACHINE, 23241 ST. RT. 37, MARYSVILLE, OH, 43040	US Mail (1st Class)
24800       MAJOR MICRO SYSTEMS, INC., 23710 SAMOSET TRAIL, SOUTHFIELD, MI, 48034       US Mail (1st Class)         24800       MALONE, PATRICIA A, 258 FEDERAL ST NW, WARREN, OH, 44483       US Mail (1st Class)         24800       MANAGEMENT RECRUITERS TALLAHASSEE, 743 EAST TENNESSEE ST., TALLAHASSEE, FL, 32308       US Mail (1st Class)         24800       MAPLE CITY PRODUCTS, 170 BENEDICT AVE, NORWALK, OH, 44857       US Mail (1st Class)         24800       MARINO, MINDY, 2715 RUSSELL RD, ASHTABULA, OH, 44004       US Mail (1st Class)         24800       MARSH SUPERMARKETS #74, 223 AUKERMAN STREET, EATON, OH, 45320       US Mail (1st Class)         24800       MARTIN FLUID POWER, 84 MINNESOTA DR, TROY, MI, 48083       US Mail (1st Class)         24800       MARTIN STREASICK, 290 MCCORMICK, LAPEER, MI, 48446       US Mail (1st Class)         24800       MARTIN SUPPLY, 200 E APPLETON AVE, SHEFFIELD, AL, 35660       US Mail (1st Class)         24800       MARTIN WELDING, 1472 W MAIN STREET, NEW LEBANON, OH, 45345       US Mail (1st Class)         24800       MARTIN, BEVERLY J, 639 ROUTE 534 NW, NEWTON FALLS, OH, 444444       US Mail (1st Class)         24800       MARUBENI SPECIALTY CHEMICALS INC, 10 BANK ST, STE 740, WHITE PLAINS, NY, 10606       US Mail (1st Class)         24800       MARVIN, MARTHA B, PO BOX 453, LEAVITTSBURG, OH, 44430       US Mail (1st Class)         24800       MASTERMAN'S, 11 C ST., BUILD	24800	MAGUIRE PRODUCTS, INC., 11 CROZERVILLE RD., ASTON, PA, 19014-1431	US Mail (1st Class)
24800       MALONE, PATRICIA A, 258 FEDERAL ST NW, WARREN, OH, 44483       US Mail (1st Class)         24800       MANAGEMENT RECRUITERS TALLAHASSEE, 743 EAST TENNESSEE ST., TALLAHASSEE, FL, 32308       US Mail (1st Class)         24800       MAPLE CITY PRODUCTS, 170 BENEDICT AVE, NORWALK, OH, 44857       US Mail (1st Class)         24800       MARINO, MINDY, 2715 RUSSELL RD, ASHTABULA, OH, 44004       US Mail (1st Class)         24800       MARSH SUPERMARKETS #74, 223 AUKERMAN STREET, EATON, OH, 45320       US Mail (1st Class)         24800       MARTIN FLUID POWER, 84 MINNESOTA DR, TROY, MI, 48083       US Mail (1st Class)         24800       MARTIN STREASICK, 290 MCCORMICK, LAPEER, MI, 48446       US Mail (1st Class)         24800       MARTIN SUPPLY, 200 E APPLETON AVE, SHEFFIELD, AL, 35660       US Mail (1st Class)         24800       MARTIN WELDING, 1472 W MAIN STREET, NEW LEBANON, OH, 45345       US Mail (1st Class)         24800       MARTIN, BEVERLY J, 639 ROUTE 534 NW, NEWTON FALLS, OH, 44444       US Mail (1st Class)         24800       MARUBENI SPECIALTY CHEMICALS INC, 10 BANK ST, STE 740, WHITE PLAINS, NY, 10606       US Mail (1st Class)         24800       MARVIN, MARTHA B, PO BOX 453, LEAVITTSBURG, OH, 44430       US Mail (1st Class)         24800       MASTERMAN'S, 11 C ST., BUILDING 10, AUBURN INDUSTRIAL PARK, AUBURN, MA, 01501-0411       US Mail (1st Class)         24800       MATZKA, 25255 MO	24800	MAHLE FILTER SYSTEMS, 906 BUTLER DR., MURFREESBORO, TN, 37127	US Mail (1st Class)
24800       MANAGEMENT RECRUITERS TALLAHASSEE, 743 EAST TENNESSEE ST., TALLAHASSEE, FL, 32308       US Mail (1st Class)         24800       MAPLE CITY PRODUCTS, 170 BENEDICT AVE, NORWALK, OH, 44857       US Mail (1st Class)         24800       MARINO, MINDY, 2715 RUSSELL RD, ASHTABULA, OH, 44004       US Mail (1st Class)         24800       MARSH SUPERMARKETS #74, 223 AUKERMAN STREET, EATON, OH, 45320       US Mail (1st Class)         24800       MARTIN FLUID POWER, 84 MINNESOTA DR, TROY, MI, 48083       US Mail (1st Class)         24800       MARTIN STREASICK, 290 MCCORMICK, LAPEER, MI, 48446       US Mail (1st Class)         24800       MARTIN SUPPLY, 200 E APPLETON AVE, SHEFFIELD, AL, 35660       US Mail (1st Class)         24800       MARTIN WELDING, 1472 W MAIN STREET, NEW LEBANON, OH, 45345       US Mail (1st Class)         24800       MARTIN, BEVERLY J, 639 ROUTE 534 NW, NEWTON FALLS, OH, 44444       US Mail (1st Class)         24800       MARUBENI SPECIALTY CHEMICALS INC, 10 BANK ST, STE 740, WHITE PLAINS, NY, 10606       US Mail (1st Class)         24800       MARVIN, MARTHA B, PO BOX 453, LEAVITTSBURG, OH, 44430       US Mail (1st Class)         24800       MASTERMAN'S, 11 C ST., BUILDING 10, AUBURN INDUSTRIAL PARK, AUBURN, MA, 01501-0411       US Mail (1st Class)         24800       MATZKA, 25255 MOUND RD, WARREN, MI, 48091       US Mail (1st Class)         24800       MAVRICK METAL STAMPING INC, PO	24800	MAJOR MICRO SYSTEMS, INC., 23710 SAMOSET TRAIL, SOUTHFIELD, MI, 48034	US Mail (1st Class)
24800       MAPLE CITY PRODUCTS, 170 BENEDICT AVE, NORWALK, OH, 44857       US Mail (1st Class)         24800       MARINO, MINDY, 2715 RUSSELL RD, ASHTABULA, OH, 44004       US Mail (1st Class)         24800       MARSH SUPERMARKETS #74, 223 AUKERMAN STREET, EATON, OH, 45320       US Mail (1st Class)         24800       MARTIN FLUID POWER, 84 MINNESOTA DR, TROY, MI, 48083       US Mail (1st Class)         24800       MARTIN STREASICK, 290 MCCORMICK, LAPEER, MI, 48446       US Mail (1st Class)         24800       MARTIN SUPPLY, 200 E APPLETON AVE, SHEFFIELD, AL, 35660       US Mail (1st Class)         24800       MARTIN WELDING, 1472 W MAIN STREET, NEW LEBANON, OH, 45345       US Mail (1st Class)         24800       MARTIN, BEVERLY J, 639 ROUTE 534 NW, NEWTON FALLS, OH, 44444       US Mail (1st Class)         24800       MARUBENI SPECIALTY CHEMICALS INC, 10 BANK ST, STE 740, WHITE PLAINS, NY, 10606       US Mail (1st Class)         24800       MARVIN, MARTHA B, PO BOX 453, LEAVITTSBURG, OH, 44430       US Mail (1st Class)         24800       MASTERMAN'S, 11 C ST., BUILDING 10, AUBURN INDUSTRIAL PARK, AUBURN, MA, 01501-0411       US Mail (1st Class)         24800       MATZKA, 25255 MOUND RD, WARREN, MI, 48091       US Mail (1st Class)         24800       MAVRICK METAL STAMPING INC, PO BOX 247, 309 S DANFORTH ST, MANCELONA, MI, 49659       US Mail (1st Class)	24800	MALONE, PATRICIA A, 258 FEDERAL ST NW, WARREN, OH, 44483	US Mail (1st Class)
24800       MARINO, MINDY, 2715 RUSSELL RD, ASHTABULA, OH, 44004       US Mail (1st Class)         24800       MARSH SUPERMARKETS #74, 223 AUKERMAN STREET, EATON, OH, 45320       US Mail (1st Class)         24800       MARTIN FLUID POWER, 84 MINNESOTA DR, TROY, MI, 48083       US Mail (1st Class)         24800       MARTIN STREASICK, 290 MCCORMICK, LAPEER, MI, 48446       US Mail (1st Class)         24800       MARTIN SUPPLY, 200 E APPLETON AVE, SHEFFIELD, AL, 35660       US Mail (1st Class)         24800       MARTIN WELDING, 1472 W MAIN STREET, NEW LEBANON, OH, 45345       US Mail (1st Class)         24800       MARTIN, BEVERLY J, 639 ROUTE 534 NW, NEWTON FALLS, OH, 44444       US Mail (1st Class)         24800       MARUBENI SPECIALTY CHEMICALS INC, 10 BANK ST, STE 740, WHITE PLAINS, NY, 10606       US Mail (1st Class)         24800       MARVIN, MARTHA B, PO BOX 453, LEAVITTSBURG, OH, 44430       US Mail (1st Class)         24800       MASTERMAN'S, 11 C ST., BUILDING 10, AUBURN INDUSTRIAL PARK, AUBURN, MA, 01501-0411       US Mail (1st Class)         24800       MAZIKA, 25255 MOUND RD, WARREN, MI, 48091       US Mail (1st Class)         24800       MAU SHERWOOD SUPPLY, 33220 LAKELAND BLVD, EASTLAKE, OH, 44095       US Mail (1st Class)         24800       MAVRICK METAL STAMPING INC, PO BOX 247, 309 S DANFORTH ST, MANCELONA, MI, 49659       US Mail (1st Class)	24800	MANAGEMENT RECRUITERS TALLAHASSEE, 743 EAST TENNESSEE ST., TALLAHASSEE, FL, 32308	US Mail (1st Class)
24800       MARSH SUPERMARKETS #74, 223 AUKERMAN STREET, EATON, OH, 45320       US Mail (1st Class)         24800       MARTIN FLUID POWER, 84 MINNESOTA DR, TROY, MI, 48083       US Mail (1st Class)         24800       MARTIN STREASICK, 290 MCCORMICK, LAPEER, MI, 48446       US Mail (1st Class)         24800       MARTIN SUPPLY, 200 E APPLETON AVE, SHEFFIELD, AL, 35660       US Mail (1st Class)         24800       MARTIN WELDING, 1472 W MAIN STREET, NEW LEBANON, OH, 45345       US Mail (1st Class)         24800       MARTIN, BEVERLY J, 639 ROUTE 534 NW, NEWTON FALLS, OH, 44444       US Mail (1st Class)         24800       MARUBENI SPECIALTY CHEMICALS INC, 10 BANK ST, STE 740, WHITE PLAINS, NY, 10606       US Mail (1st Class)         24800       MARVIN, MARTHA B, PO BOX 453, LEAVITTSBURG, OH, 44430       US Mail (1st Class)         24800       MASTERMAN'S, 11 C ST., BUILDING 10, AUBURN INDUSTRIAL PARK, AUBURN, MA, 01501-0411       US Mail (1st Class)         24800       MATZKA, 25255 MOUND RD, WARREN, MI, 48091       US Mail (1st Class)         24800       MAU SHERWOOD SUPPLY, 33220 LAKELAND BLVD, EASTLAKE, OH, 44095       US Mail (1st Class)         24800       MAVRICK METAL STAMPING INC, PO BOX 247, 309 S DANFORTH ST, MANCELONA, MI, 49659       US Mail (1st Class)	24800	MAPLE CITY PRODUCTS, 170 BENEDICT AVE, NORWALK, OH, 44857	US Mail (1st Class)
24800       MARTIN FLUID POWER, 84 MINNESOTA DR, TROY, MI, 48083       US Mail (1st Class)         24800       MARTIN STREASICK, 290 MCCORMICK, LAPEER, MI, 48446       US Mail (1st Class)         24800       MARTIN SUPPLY, 200 E APPLETON AVE, SHEFFIELD, AL, 35660       US Mail (1st Class)         24800       MARTIN WELDING, 1472 W MAIN STREET, NEW LEBANON, OH, 45345       US Mail (1st Class)         24800       MARTIN, BEVERLY J, 639 ROUTE 534 NW, NEWTON FALLS, OH, 44444       US Mail (1st Class)         24800       MARUBENI SPECIALTY CHEMICALS INC, 10 BANK ST, STE 740, WHITE PLAINS, NY, 10606       US Mail (1st Class)         24800       MARVIN, MARTHA B, PO BOX 453, LEAVITTSBURG, OH, 44430       US Mail (1st Class)         24800       MASTERMAN'S, 11 C ST., BUILDING 10, AUBURN INDUSTRIAL PARK, AUBURN, MA, 01501-0411       US Mail (1st Class)         24800       MATZKA, 25255 MOUND RD, WARREN, MI, 48091       US Mail (1st Class)         24800       MAU SHERWOOD SUPPLY, 33220 LAKELAND BLVD, EASTLAKE, OH, 44095       US Mail (1st Class)         24800       MAVRICK METAL STAMPING INC, PO BOX 247, 309 S DANFORTH ST, MANCELONA, MI, 49659       US Mail (1st Class)	24800	MARINO, MINDY, 2715 RUSSELL RD, ASHTABULA, OH, 44004	US Mail (1st Class)
24800       MARTIN STREASICK, 290 MCCORMICK, LAPEER, MI, 48446       US Mail (1st Class)         24800       MARTIN SUPPLY, 200 E APPLETON AVE, SHEFFIELD, AL, 35660       US Mail (1st Class)         24800       MARTIN WELDING, 1472 W MAIN STREET, NEW LEBANON, OH, 45345       US Mail (1st Class)         24800       MARTIN, BEVERLY J, 639 ROUTE 534 NW, NEWTON FALLS, OH, 44444       US Mail (1st Class)         24800       MARUBENI SPECIALTY CHEMICALS INC, 10 BANK ST, STE 740, WHITE PLAINS, NY, 10606       US Mail (1st Class)         24800       MARVIN, MARTHA B, PO BOX 453, LEAVITTSBURG, OH, 44430       US Mail (1st Class)         24800       MASTERMAN'S, 11 C ST., BUILDING 10, AUBURN INDUSTRIAL PARK, AUBURN, MA, 01501-0411       US Mail (1st Class)         24800       MATZKA, 25255 MOUND RD, WARREN, MI, 48091       US Mail (1st Class)         24800       MAU SHERWOOD SUPPLY, 33220 LAKELAND BLVD, EASTLAKE, OH, 44095       US Mail (1st Class)         24800       MAVRICK METAL STAMPING INC, PO BOX 247, 309 S DANFORTH ST, MANCELONA, MI, 49659       US Mail (1st Class)	24800	MARSH SUPERMARKETS #74, 223 AUKERMAN STREET, EATON, OH, 45320	US Mail (1st Class)
24800       MARTIN SUPPLY, 200 E APPLETON AVE, SHEFFIELD, AL, 35660       US Mail (1st Class)         24800       MARTIN WELDING, 1472 W MAIN STREET, NEW LEBANON, OH, 45345       US Mail (1st Class)         24800       MARTIN, BEVERLY J, 639 ROUTE 534 NW, NEWTON FALLS, OH, 44444       US Mail (1st Class)         24800       MARUBENI SPECIALTY CHEMICALS INC, 10 BANK ST, STE 740, WHITE PLAINS, NY, 10606       US Mail (1st Class)         24800       MARVIN, MARTHA B, PO BOX 453, LEAVITTSBURG, OH, 44430       US Mail (1st Class)         24800       MASTERMAN'S, 11 C ST., BUILDING 10, AUBURN INDUSTRIAL PARK, AUBURN, MA, 01501-0411       US Mail (1st Class)         24800       MATZKA, 25255 MOUND RD, WARREN, MI, 48091       US Mail (1st Class)         24800       MAU SHERWOOD SUPPLY, 33220 LAKELAND BLVD, EASTLAKE, OH, 44095       US Mail (1st Class)         24800       MAVRICK METAL STAMPING INC, PO BOX 247, 309 S DANFORTH ST, MANCELONA, MI, 49659       US Mail (1st Class)	24800	MARTIN FLUID POWER, 84 MINNESOTA DR, TROY, MI, 48083	US Mail (1st Class)
24800       MARTIN WELDING, 1472 W MAIN STREET, NEW LEBANON, OH, 45345       US Mail (1st Class)         24800       MARTIN, BEVERLY J, 639 ROUTE 534 NW, NEWTON FALLS, OH, 44444       US Mail (1st Class)         24800       MARUBENI SPECIALTY CHEMICALS INC, 10 BANK ST, STE 740, WHITE PLAINS, NY, 10606       US Mail (1st Class)         24800       MARVIN, MARTHA B, PO BOX 453, LEAVITTSBURG, OH, 44430       US Mail (1st Class)         24800       MASTERMAN'S, 11 C ST., BUILDING 10, AUBURN INDUSTRIAL PARK, AUBURN, MA, 01501-0411       US Mail (1st Class)         24800       MATZKA, 25255 MOUND RD, WARREN, MI, 48091       US Mail (1st Class)         24800       MAU SHERWOOD SUPPLY, 33220 LAKELAND BLVD, EASTLAKE, OH, 44095       US Mail (1st Class)         24800       MAVRICK METAL STAMPING INC, PO BOX 247, 309 S DANFORTH ST, MANCELONA, MI, 49659       US Mail (1st Class)	24800	MARTIN STREASICK, 290 MCCORMICK, LAPEER, MI, 48446	US Mail (1st Class)
24800       MARTIN, BEVERLY J, 639 ROUTE 534 NW, NEWTON FALLS, OH, 44444       US Mail (1st Class)         24800       MARUBENI SPECIALTY CHEMICALS INC, 10 BANK ST, STE 740, WHITE PLAINS, NY, 10606       US Mail (1st Class)         24800       MARVIN, MARTHA B, PO BOX 453, LEAVITTSBURG, OH, 44430       US Mail (1st Class)         24800       MASTERMAN'S, 11 C ST., BUILDING 10, AUBURN INDUSTRIAL PARK, AUBURN, MA, 01501-0411       US Mail (1st Class)         24800       MATZKA, 25255 MOUND RD, WARREN, MI, 48091       US Mail (1st Class)         24800       MAU SHERWOOD SUPPLY, 33220 LAKELAND BLVD, EASTLAKE, OH, 44095       US Mail (1st Class)         24800       MAVRICK METAL STAMPING INC, PO BOX 247, 309 S DANFORTH ST, MANCELONA, MI, 49659       US Mail (1st Class)	24800	MARTIN SUPPLY, 200 E APPLETON AVE, SHEFFIELD, AL, 35660	US Mail (1st Class)
24800       MARUBENI SPECIALTY CHEMICALS INC, 10 BANK ST, STE 740, WHITE PLAINS, NY, 10606       US Mail (1st Class)         24800       MARVIN, MARTHA B, PO BOX 453, LEAVITTSBURG, OH, 44430       US Mail (1st Class)         24800       MASTERMAN'S, 11 C ST., BUILDING 10, AUBURN INDUSTRIAL PARK, AUBURN, MA, 01501-0411       US Mail (1st Class)         24800       MATZKA, 25255 MOUND RD, WARREN, MI, 48091       US Mail (1st Class)         24800       MAU SHERWOOD SUPPLY, 33220 LAKELAND BLVD, EASTLAKE, OH, 44095       US Mail (1st Class)         24800       MAVRICK METAL STAMPING INC, PO BOX 247, 309 S DANFORTH ST, MANCELONA, MI, 49659       US Mail (1st Class)	24800	MARTIN WELDING, 1472 W MAIN STREET, NEW LEBANON, OH, 45345	US Mail (1st Class)
24800       MARVIN, MARTHA B, PO BOX 453, LEAVITTSBURG, OH, 44430       US Mail (1st Class)         24800       MASTERMAN'S, 11 C ST., BUILDING 10, AUBURN INDUSTRIAL PARK, AUBURN, MA, 01501-0411       US Mail (1st Class)         24800       MATZKA, 25255 MOUND RD, WARREN, MI, 48091       US Mail (1st Class)         24800       MAU SHERWOOD SUPPLY, 33220 LAKELAND BLVD, EASTLAKE, OH, 44095       US Mail (1st Class)         24800       MAVRICK METAL STAMPING INC, PO BOX 247, 309 S DANFORTH ST, MANCELONA, MI, 49659       US Mail (1st Class)	24800	MARTIN, BEVERLY J, 639 ROUTE 534 NW, NEWTON FALLS, OH, 44444	US Mail (1st Class)
24800       MASTERMAN'S, 11 C ST., BUILDING 10, AUBURN INDUSTRIAL PARK, AUBURN, MA, 01501-0411       US Mail (1st Class)         24800       MATZKA, 25255 MOUND RD, WARREN, MI, 48091       US Mail (1st Class)         24800       MAU SHERWOOD SUPPLY, 33220 LAKELAND BLVD, EASTLAKE, OH, 44095       US Mail (1st Class)         24800       MAVRICK METAL STAMPING INC, PO BOX 247, 309 S DANFORTH ST, MANCELONA, MI, 49659       US Mail (1st Class)	24800	MARUBENI SPECIALTY CHEMICALS INC, 10 BANK ST, STE 740, WHITE PLAINS, NY, 10606	US Mail (1st Class)
24800 MATZKA, 25255 MOUND RD, WARREN, MI, 48091 US Mail (1st Class) 24800 MAU SHERWOOD SUPPLY, 33220 LAKELAND BLVD, EASTLAKE, OH, 44095 US Mail (1st Class) 24800 MAVRICK METAL STAMPING INC, PO BOX 247, 309 S DANFORTH ST, MANCELONA, MI, 49659 US Mail (1st Class)	24800	MARVIN, MARTHA B, PO BOX 453, LEAVITTSBURG, OH, 44430	US Mail (1st Class)
24800 MAU SHERWOOD SUPPLY, 33220 LAKELAND BLVD, EASTLAKE, OH, 44095 US Mail (1st Class) 24800 MAVRICK METAL STAMPING INC, PO BOX 247, 309 S DANFORTH ST, MANCELONA, MI, 49659 US Mail (1st Class)	24800	MASTERMAN'S, 11 C ST., BUILDING 10, AUBURN INDUSTRIAL PARK, AUBURN, MA, 01501-0411	US Mail (1st Class)
24800 MAVRICK METAL STAMPING INC, PO BOX 247, 309 S DANFORTH ST, MANCELONA, MI, 49659 US Mail (1st Class)	24800	MATZKA, 25255 MOUND RD, WARREN, MI, 48091	US Mail (1st Class)
	24800	MAU SHERWOOD SUPPLY, 33220 LAKELAND BLVD, EASTLAKE, OH, 44095	US Mail (1st Class)
24800 MAXI-BLAST INC., 3650 NORTH OLIVE ROAD, SOUTH BEND, IN, 46628 US Mail (1st Class)	24800	MAVRICK METAL STAMPING INC, PO BOX 247, 309 S DANFORTH ST, MANCELONA, MI, 49659	US Mail (1st Class)
	24800	MAXI-BLAST INC., 3650 NORTH OLIVE ROAD, SOUTH BEND, IN, 46628	US Mail (1st Class)
24800 MCFEELY, JACKSON J, PO BOX 650, 5305 E MAPLE, GENEVA, OH, 44041 <b>US Mail (1st Class)</b>	24800	MCFEELY, JACKSON J, PO BOX 650, 5305 E MAPLE, GENEVA, OH, 44041	US Mail (1st Class)
24800 MCGEE CHEMICAL, 9 CORZERVILLE RD, ASTON, PA, 19014 US Mail (1st Class)	24800	MCGEE CHEMICAL, 9 CORZERVILLE RD, ASTON, PA, 19014	US Mail (1st Class)
24800 MCLAREN MEDICAL MANAGEMENT, 1254 N MAIN STREET, LAPEER, MI, 48446 US Mail (1st Class)	24800	MCLAREN MEDICAL MANAGEMENT, 1254 N MAIN STREET, LAPEER, MI, 48446	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	MCMASTER-CARR SUPPLY CO., 600 COUNTY LINE RD, ELMHURST, IL, 60126-2034	US Mail (1st Class)
24800	MCMASTER-CARR SUPPLY CO., 600 COUNTY LINE RD, ELMHURST, IL, 60126	US Mail (1st Class)
24800	MCNAUGHTON-MCKAY ELECTRIC, DEPT. # 14801, PO BOX 67000, DETROIT, MI, 48267	US Mail (1st Class)
24800	MCNAUGHTON-MCKAY ELECTRIC CO., 1357 E LINCOLN AVE, MADISON HEIGHTS, MI, 48071-4134	US Mail (1st Class)
24800	MEAD ELECTRONICS, 910 W. MANSFIELD ST, PO BOX 147, BUCYRUS, OH, 44820	US Mail (1st Class)
24800	MEADORS, SALLY, 3770 BULAH RD, JEFFERSON, OH, 44047	US Mail (1st Class)
24800	MEASURE-ALL INC., 447 NILLES RD, FAIRFIELD, OH, 45014-2633	US Mail (1st Class)
24800	MEASURING SOLUTIONS, WESSCOTT PLAZA, SUITE G, 832 SNOW ST, OXFORD, AL, 36203-1217	US Mail (1st Class)
24800	MEDLER ELECTRIC CO., 1313 MICHIGAN AVE, ALMA, MI, 48801	US Mail (1st Class)
24800	MEDWORK OCCUPATIONAL, 725 S LUDLOW ST, DAYTON, OH, 45402-2610	US Mail (1st Class)
24800	MEGA FLUIDLINE PRODUCTS INC., 425 KENNEDY ROAD, AKRON, OH, 44305	US Mail (1st Class)
24800	MEIER TRANSMISSION LTD., 1845 E 40TH ST, CLEVELAND, OH, 44103-3503	US Mail (1st Class)
24800	MEIKLE NEW YORK, INC., 36639 TREASURY CENTER, CHICAGO, IL, 60694-6600	US Mail (1st Class)
24800	MELANIE LIMBERG-WARNER, 8707 SAMUEL BARTON DRIVE, BELLEVILLE, MI, 48111	US Mail (1st Class)
24800	MERCER STAFFING, ATTN DEEGAN FARDON, 1600 MANOR DR STE 110, CHALFORT, PA, 18914	US Mail (1st Class)
24800	MERCY MEDICAL CENTER, 1320 MERCY DRIVE NW, CANTON, OH, 44708	US Mail (1st Class)
24800	METAL CLADDING INC, 230 S NIAGARA ST, LOCKPORT, NY, 14094-1927	US Mail (1st Class)
24800	METALLURGICAL SERVICE INC., 2681 E RIVER RD., DAYTON, OH, 45439	US Mail (1st Class)
24800	METRO MECHANICAL SERVICES INC, ZARZAUR & SCHWARTZ PC, PO BOX 11366, BIRMINGHAM, AL, 35202	US Mail (1st Class)
24800	METRO WELDING SUPPLY CORP., 12620 SOUTHFIELD, DETROIT, MI, 48223	US Mail (1st Class)
24800	METROPOLITAN LIFE INSUR., DEPT. CH10579, PALATINE, IL, 60055-0579	US Mail (1st Class)
24800	METZELER AUTOMOTIVE PROFILE SYSTEMS, 900 E WHITCOMB AVE, MADISON HEIGHTS, MI, 48071-5612	US Mail (1st Class)
24800	MG AUTOMATION & CONTROLS, 1412 EDWARDS AVE, HARAHAN, LA, 70123	US Mail (1st Class)
24800	MIAMI INDUSTRIAL TRUCKS INC, 2830 E RIVER RD, DAYTON, OH, 45439	US Mail (1st Class)
24800	MICHIGAN AIRGAS, G5075 N DORT HWY, FLINT, MI, 48505	US Mail (1st Class)
24800	MICHIGAN FIRST AID, 16199 COMMON RD, ROSEVILLE, MI, 48066	US Mail (1st Class)
24800	MICHIGAN SHIPPERS SUPPLY, 600 MARYLAND AVE NE, GRAND RAPIDS, MI, 49505-6002	US Mail (1st Class)
24800	MICHIGAN, STATE OF, DEPARTMENT OF TREASURY, CADILLAC PLACE, 3030 W GRAND BLVD, DETROIT, MI, 48202	US Mail (1st Class)
24800	MICRO PLASTICS INC, 11 INDUSTRY LN, FLIPPIN, AR, 72634	US Mail (1st Class)
24800	MICRO TOOL INC., PO BOX 227, 433 S DIAMOND MILL RD., NEW LEBANON, OH, 45345	US Mail (1st Class)
24800	MICROFINISH, 865 SCHOLZ DR., VANDALIA, OH, 45377	US Mail (1st Class)
24800	MICROWAVE SPECIALITIES INC, 8671 CHERRY LANE, LAUREL, MD, 20707	US Mail (1st Class)
24800	MID OHIO PACKAGING, 2135 INNOVATION DR, MARION, OH, 43302-8261	US Mail (1st Class)
24800	MID-OHIO WATER GROUP L.L.C., 4116 PARK AVE WEST, PO BOX 150, ONTARIO, OH, 44862-0150	US Mail (1st Class)
24800	MIDWEST DIE SUPPLY CO., 6240 AMERICAN ROAD, PO BOX 6657, TOLEDO, OH, 43612	US Mail (1st Class)
24800	MIDWEST EXPANDED METAL, 301 INDUSTRIAL BLVD., WACONIA, MN, 55387	US Mail (1st Class)
24800	MIDWEST MOLD & TEXTURE, 4270 ARMSTRONG BLVD, BATAVIA, OH, 45103-1670	US Mail (1st Class)
24800	MIDWEST OPTICAL SUPPLY, PO BOX 293930, LEWISVILLE, TX, 75029-3930	US Mail (1st Class)
24800	MIDWEST PAPER SPECIALTIES CO., 322 RYDER RD, TOLEDO, OH, 43607-3104	US Mail (1st Class)
24800	MIDWEST REGIONAL BOARD, ACTWU - SUITE #205, 333 SOUTH ASHLAND AVENUE, CHICAGO, IL, 60607	US Mail (1st Class)
24800	MIFAST, 1145 JANSEN FARM DR, ELGIN, IL, 60123-2596	US Mail (1st Class)
24800	MILACRON MARKETING COMPANY, ATTN TOM AUSTING, 4165 HALF ACRE RD, BATAVIA, OH, 45103	US Mail (1st Class)
24800	MILLENNIUM BUSINESS SYSTEMS, 38281 SCHOOLCRAFT RD, SUITE B, LIVONIA, MI, 48150	US Mail (1st Class)
24800	MILLER, JUDITH A, 15365 NASH RD, BURTON, OH, 44021	US Mail (1st Class)
24800	MILLIRON WASTE MANAGEMENT, ACCOUNT #1987, 2384 S.R. 39 NW, MANSFIELD, OH, 44903	US Mail (1st Class)
24800	MILLS JR, FORREST ROGER, 2908 DODGEVILLE RD, ROME, OH, 44085	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	MILLWOOD INC., 986 TIBBETTS WICK RD, GIRARD, OH, 44420	US Mail (1st Class)
24800	MINCO TOOL & MOLD, INC., 5690 WEBSTER ST., DAYTON, OH, 45414-3530	US Mail (1st Class)
24800	MINETT, GARY L, 5401 COREY HUNT RD, BRISTOLVILLE, OH, 44402	US Mail (1st Class)
24800	MINI U STORAGE, 13450 MERRIMAN RD., LIVONIA, MI, 48150	US Mail (1st Class)
24800	MINNESOTA RUBBER, 1100 XENUIM LN N, MINNEAPOLIS, MN, 55441-7000	US Mail (1st Class)
24800	MIRANDA, RUSSELL E, 10576 STATE ROUTE 303, WINDHAM, OH, 44288	US Mail (1st Class)
24800	MIRSA, 501 N. BRIDGE ST, PMB # 148, HIDALGO, TX, 78557	US Mail (1st Class)
24800	MITCHELL, MARY C, 9973 N CHAPLE, GARRETTSVILLE, OH, 44231	US Mail (1st Class)
24800	MODERN METAL PRODUCTS CO, JESSICA GROSSARTH ESQ, PO BOX 7006, PULLMAN & COMLEY LLC, 850 MAIN ST, BRIDGEPORT, CT, 06601-7006	US Mail (1st Class)
24800	MODULAR AUTOMOTIVE SYSTEMS, 26195 BUNERT ROAD, WARREN, MI, 48089	US Mail (1st Class)
24800	MOLD MASTERS CO., 1455 IMLAY CITY RD., LAPEER, MI, 48446	US Mail (1st Class)
24800	MOLD MASTERS SYS INC, 233 ARMSTRONG AVE, GEORGETOWN, ON, L7G 4X5CANADA	US Mail (1st Class)
24800	MOLD MEDIC, INC., 12395 BAYVIEW DR., WAYLAND, MI, 49348	US Mail (1st Class)
24800	MOLDERS SERVICES, 646 AJAX DR, MADISON HEIGHTS, MI, 48071	US Mail (1st Class)
24800	MOLDFLOW CORP., 492 OLD CONNECTICUT PATH, SUITE 401, FRAMINGHAM, MA, 01701	US Mail (1st Class)
24800	MOLDING AUTOMATION CONCEPT, 1686 MOMENTUM PLACE, CHICAGO, IL, 60689-5316	US Mail (1st Class)
24800	MOLD-MASTERS LIMITED, 233 ARMSTRONG AVE, GEORGETOWN, ON, L7G 4X5CANADA	US Mail (1st Class)
24800	MOLDTECH INC, 1900 COMMERCE PKWY, LANCASTER, NY, 14086-1735	US Mail (1st Class)
24800	MOLD-TECH MICHIGAN, 34497 KELLY RD, FRASER, MI, 48026-3404	US Mail (1st Class)
24800	MONROE, WENDY, 600 PRENTICE RD, WARREN, OH, 44481	US Mail (1st Class)
24800	MONTGOMERY COUNTY TREASURER, HUGH QUILL TREASURER, PO BOX 972, 451 W THIRD ST, DAYTON, OH, 45422	US Mail (1st Class)
24800	MOORE, FRED M, 612 IOWA AVE, NILES, OH, 44446	US Mail (1st Class)
24800	MORGAN, PAUL, 1190 NILES CORTLAND RD SE, WARREN, OH, 44484-2541	US Mail (1st Class)
24800	MORROW CONTROL, 810 CARNAHAN AVE. NE, CANTON, OH, 44705	US Mail (1st Class)
24800	MOTION INDUSTRIES, PO BOX 1477, BIRMINGHAM, AL, 35201-1477	US Mail (1st Class)
24800	MOTION INDUSTRIES, 3440 NEEDMORE RD, DAYTON, OH, 45414	US Mail (1st Class)
24800	MOTION INDUSTRIES, INC., 4103 STRAUSSER ST NW, NORTH CANTON, OH, 44720	US Mail (1st Class)
24800	MSC INDUSTRIAL SUPPLY, 75 MAXESS RD, MELVILLE, NY, 11747	US Mail (1st Class)
24800	MT. CLEMENS CRANE, 42827 IRWIN DRIVE, HARRISON TWP., MI, 48045	US Mail (1st Class)
24800	MULTIBASE, INC., 22556 NETWORK PLACE, CHICAGO, IL, 60673-1225	US Mail (1st Class)
24800	MUTUAL OF OMAHA INSUR. CO., GROUP PREMIUM & ENROLLMENT SERVICES, PO BOX 740705, ATLANTA, GA, 30374-0705	US Mail (1st Class)
24800	MUZAK-DAYTON, MIAMI VALLEY AUDIO, LTD., 175 E. ALEX BELL RD., STE. 228, DAYTON, OH, 45459	US Mail (1st Class)
24800	MUZZALL BUSINESS FORMS, 1640 HASLETT RD., SUITE 7A, HASLETT, MI, 48840	US Mail (1st Class)
24800	MYERS, MONA M, 8907 ROUTE 193, ORWELL, OH, 44076	US Mail (1st Class)
24800	MYTEX POLYMERS, 1403 PORT RD, JEFFERSONVILLE, IN, 47130-8411	US Mail (1st Class)
24800	N.T. RUDDOCK COMPANY, 26123 BROADWAY AVE, CLEVELAND, OH, 44146-6512	US Mail (1st Class)
24800	NANGLE, HEIDI, PO BOX 426, LEAVITTSBURG, OH, 44430	US Mail (1st Class)
24800	NARA MOLD & DIE & LTD, 641-711, 391-2 GA EUM JEONG-DONG, CHANGWON CITY, GYEON NAM, SOUTH KOREA	US Mail (1st Class)
24800	NATIONAL BULK EQUIPMENT,INC., 1645 RELIABLE PARKWAY, CHICAGO, IL, 60686	US Mail (1st Class)
24800	NATIONAL CITY BANK, NATIONAL CITY CENTER, 1900 EAST NINTH STREET, CLEVELAND, OH, 44114-3484	US Mail (1st Class)
24800	NATIONAL CONSULTING & DEVELOPMENT, 33801 CURTIS BLVD., #100, EASTLAKE, OH, 44095	US Mail (1st Class)
24800	NATIONAL PUMP & PROCESS, INC., 180 TREAT RD, AURORA, OH, 44202-8704	US Mail (1st Class)
24800	NATIONS RENT, 450 E LAS OLAS BLVD 14TH FL, FT LAUDERDALE, FL, 33301-2292	US Mail (1st Class)
24800	NAUMANN HOBBS, 4335 E WOOD ST, PHOENIX, AZ, 85040-2045	US Mail (1st Class)
24800	NELMOR CO., INC., 100 FRED RODDY AVE, ATTLEBORO, MA, 02703	US Mail (1st Class)
24800	NELS JORGENSON & COMPANY, 20400 NINE MILE RD, ST CLAIR SHORES, MI, 48080	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	NEOPOST LEASING, INC., 3400 BRIDGE PKWY, STE 201, REDWOOD CITY, CA, 94065	US Mail (1st Class)
24800	NEW HAVEN SUPPLY CO, 2744 BULLHEAD RD., PO BOX 155, NEW HAVEN, OH, 44850	US Mail (1st Class)
24800	NEW PACIFIC MACHINERY, PO BOX 467, 1264 HWY 45 BY-PASS, TRENTON, TN, 38382	US Mail (1st Class)
24800	NEW PIG CORPORATION, ONE PORK AVE, TIPTON, PA, 16684	US Mail (1st Class)
24800	NEWARK INONE, 4801 N REVENSWOOD AVE, CHICAGO, IL, 60640-4457	US Mail (1st Class)
24800	NEWNHAM, RON, 1925 38TH ST NE, CANTON, OH, 44705	US Mail (1st Class)
24800	NEWSPAPER NETWORK OF CENTRAL OHIO, 2 N 1ST ST, NEWARK, OH, 43055-5608	US Mail (1st Class)
24800	NICKERSON MACHINERY, 10 KEITH WAY, HINGHAM, MA, 02043-4209	US Mail (1st Class)
24800	NIFCO AMERICA CORP., 8015 DOVE PARKWAY CENTRAL, WINCHESTER, OH, 43110	US Mail (1st Class)
24800	NISSEI AMERICA, INC., 1480 NORTH HANCOCK STREET, ANAHEIM, CA, 92807	US Mail (1st Class)
24800	NL VENTURES V CARLISLE LP, C/O AIC VENTURES, 8080 NORTH CENTRAL EXPWY, STE 1220, DALLAS, TX, 75206	US Mail (1st Class)
24800	NORDSON CORP, 28601 CLEMENS RD, WESTLAKE, OH, 44145-1148	US Mail (1st Class)
24800	NORRIS SALES ASSOCIATES, C/O DARRELL A CLAY ESQ, WALTER & HAVERFIELD LLP, 1301 E 9TH ST, STE 3500, CLEVELAND, OH, 44114-1821	US Mail (1st Class)
24800	NORRIS SALES ASSOCIATES, DARRELL A CLAY ESQ, WALTER & HAVERFIELD LLP, 1301 EAST NINTH ST, STE 3500, CLEVELAND, OH, 44114-1821	US Mail (1st Class)
24800	NORTH COAST SALES, 16345 KINSMAN RD., MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24800	NORTH PAK PALMER CONTAINER CO., 1842 ENTERPRISE PKWY, TWINSBURG, OH, 44087-2206	US Mail (1st Class)
24800	NORTHEAST POLY BAG CO INC, PO BOX 1460, 2 NORTHEAST BLVD, STERLING, MA, 01564	US Mail (1st Class)
24800	NORTHERN INDUSTRIAL SUPPLY, 2800 E HOLLAND AVE, SAGINAW, MI, 48601	US Mail (1st Class)
24800	NORTHERN PLASTIC MACHINE, REPAIR INC., PO BOX 182106, SHELBY TWP., MI, 48318-2106	US Mail (1st Class)
24800	NORTHERN SAFETY CO, 232 INDUSTRIAL PARK DRIVE, FRANKFORT, NY, 13340	US Mail (1st Class)
24800	NORTHERN SUPPLY, 1453 SOLUTIONS CENTER, CHICAGO, IL, 60677-1004	US Mail (1st Class)
24800	NORTHPORT ELECTRICAL, 3111 MCFARLAND BLVD, NORTHPORT, AL, 35476-3170	US Mail (1st Class)
24800	NORTHSIDE TRUE VALUE HRD, 2912 S. WAYNE RD., WAYNE, MI, 48184	US Mail (1st Class)
24800	NORTHSTAR BUSINESS AVIATION, 5430 LAUBY RD., N. CANTON, OH, 44720	US Mail (1st Class)
24800	NORTHWEST REGIONAL SYSTEMS, ACCT: N/A, 210 W MARKET ST, CELINA, OH, 45822-2151	US Mail (1st Class)
24800	NOVOTECHNIK U.S., INC., 155 NORTHBORO RD, SOUTHBOROUGH, MA, 01772	US Mail (1st Class)
24800	NSF INTERNATIONAL STRATEGIC REG LTD, 6201 RELIABLE PARKWAY, CHICAGO, IL, 60686-0062	US Mail (1st Class)
24800	NSU CORPORATION, 9385 SONORA RD, SONORA, KY, 42776	US Mail (1st Class)
24800	OE MEYER COMPANY, 304 6TH AVE, GALION, OH, 44833	US Mail (1st Class)
24800	OFC CAPITOL CORP., P.O. BOX 863981, ORLANDO, FL, 32886-3981	US Mail (1st Class)
24800	OFFICE DEPOT, 2200 OLD GERMANTOWN RD, DELRAY BEACH, FL, 33445-8223	US Mail (1st Class)
24800	OFFICE DEPOT D564203449177, 2200 OLD GERMANTOWN RD, DELRAY BEACH, FL, 33445-8223	US Mail (1st Class)
24800	OFFNER, DEBORAH L, 23 ROBERT DR, VIENNA, OH, 44473	US Mail (1st Class)
24800	OHIO AIR PROD. OF CANTON, 2350 13TH ST NE, CANTON, OH, 44705	US Mail (1st Class)
24800	OHIO CENTRAL SUPPLY, 239 E HELENA STREET, DAYTON, OH, 45404	US Mail (1st Class)
24800	OHIO DESK COMPANY, 1122 PROSPECT AVE E, CLEVELAND, OH, 44115-1229	US Mail (1st Class)
24800	OHIO EDISON COMPANY, BANKRUPTCY DEPT, 6896 MILLER RD, RM 204, BRECKSVILLE, OH, 44141	US Mail (1st Class)
24800	OHIO FABRICATORS CO, PO BOX 218, 111 N 14TH ST, COSHOCTON, OH, 43812-0218	US Mail (1st Class)
24800	OHIO INDUSTRIAL SUPPLY INC, 915 LAKE RD, MEDINA, OH, 44256	US Mail (1st Class)
24800	OHIO MATERIALS HANDLING, 8155 ROLL & HOLD PKWY, MACEDONIA, OH, 44056	US Mail (1st Class)
24800	OLYMPUS NDT, INC., 1329 E KEMPER RD, BUILDING 400, SUITE 4222, CINCINNATI, OH, 45246	US Mail (1st Class)
24800	ONLINE EMPLOYMENT SERVICE INC., PO BOX 273, HASTINGS, MI, 49058	US Mail (1st Class)
24800	ORBIT MOVERS & ERECTORS INC., 1101 WEGLEY PLACE, DAYTON, OH, 45407	US Mail (1st Class)
24800	ORKIN PEST CONTROL, 3404 SUCCESSFUL WAY, DAYTON, OH, 45414	US Mail (1st Class)
24800	OVERHEAD DOOR CO OF TUSCALOOSA, 1181 WHIGHAM PL, TUSCALOOSA, AL, 35405	US Mail (1st Class)
24800	OVERHEAD DOOR CO OF TUSCALOOSA, 1181 WHIGHAM PLACE, TUSCALOOSA, AL, 35405	US Mail (1st Class)
24800	OVERLY, FLOYD, 10817 SHANKDOWN, GARRETTSVILLE, OH, 44231	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	OVERLY, JANET M, 10817 SHANKDOWNS, GARRETTSVILLE, OH, 44231	US Mail (1st Class)
24800	P & L PROPERTIES, 5480 INDUSTRIAL DRIVE, CONNERSVILLE, IN, 47331	US Mail (1st Class)
24800	P/A INDUSTRIES INC., 522 COTTAGE GROVE ROAD, BLOOMFIELD, CT, 06002	US Mail (1st Class)
24800	PABCO FLUID CO., SUNSOURCE, NW 7809, MINNEAPOLIS, MN, 55485-7809	US Mail (1st Class)
24800	PACE MOLD & MACHINE INC, 8225 NAVARRE RD SW, MASSILLON, OH, 44646	US Mail (1st Class)
24800	PACK, TAMMY, 5914 GRANGER RD, ASHTABULA, OH, 44004	US Mail (1st Class)
24800	PACKAGING CORPORATION OF AMERICA, C/O PACKAGING CREDIT COMPANY LLC, 900 E DIEHL RD, STE 131, NAPERVILLE, IL, 60563	US Mail (1st Class)
24800	PAKRITE INC., 6050 CAREY DR, VALLEY VIEW, OH, 44125	US Mail (1st Class)
24800	PALUMBO, JASON, 4701 STATE ROUTE 534, WEST FARMINGTON, OH, 44491	US Mail (1st Class)
24800	PANTHER II TRANSPORTATION INC, 4940 PANTHER PKWY, SEVILLE, OH, 44273	US Mail (1st Class)
24800	PAQUIN COMPANY, 26981 LAKELAND BLVD, EUCLID, OH, 44132-2656	US Mail (1st Class)
24800	PAR FOAM PRODUCTS INC, 239 VAN RENSSELAER ST, BUFFALO, NY, 14210	US Mail (1st Class)
24800	PARAGON DIE & ENGINEERING, 5225 33RD STREET S.E., GRAND RAPIDS, MI, 49512	US Mail (1st Class)
24800	PARAGON TECHNOLOGIES, INC, 5775 TEN MILE ROAD, WARREN, MI, 48091	US Mail (1st Class)
24800	PARAMOUNT STAMPING & WELDING, 1200 W 58TH, CLEVELAND, OH, 44102	US Mail (1st Class)
24800	PARKER HANNIFIN CORPORATION, JEREMY M CAMPANA ESQ, THOMPSON HINE LLP, 127 PUBLIC SQ, 3900 KEY CENTER, CLEVELAND, OH, 44114	US Mail (1st Class)
24800	PARKER HANNIFIN CORPORATION, PARKER SERVICE CENTER, 6035 PARKLAND BLVD, CLEVELAND, OH, 44124	US Mail (1st Class)
24800	PARKER SEALS/O-RING DIV, 7891 COLLECTION CENTER, CHICAGO, IL, 60693	US Mail (1st Class)
24800	PARKER, LYNDA M, 15 TIOGA ST, NEWTON FALLS, OH, 44494	US Mail (1st Class)
24800	PARKER, PAUL D, 2406 JACKSON ST SW, WARREN, OH, 44485	US Mail (1st Class)
24800	PARKER-HANNIFIN CORP., ATTN: CORP. REAL ESTATE MANAGER, 6035 PARKLAND BLVD., CLEVELAND, OH, 44124-4141	US Mail (1st Class)
24800	PARKWAY PLASTICS, 612 PARKWAY DR, WEST BRANCH, MI, 48661	US Mail (1st Class)
24800	PARTICULAR TECHNOLOGY, 3909 GROVE AVE., GURNEE, IL, 60031	US Mail (1st Class)
24800	PARTNERS EQUITY CAPITAL CO, PO BOX 809314, CHICAGO, IL, 60680-9314	US Mail (1st Class)
24800	PATRICK K SMITH, (RE: CONSOLIDATED MACHINERY MOVERS), ATTORNEY AT LAW, 120 W SECOND STREET, 13TH FLOOR LIBERTY TOWER, DAYTON, OH, 45402	US Mail (1st Class)
24800	PATTON PEST CONTROL, 15526 CHILLICOTHE RD, NOVELTY, OH, 44072	US Mail (1st Class)
24800	PATTON'S INC., 3201 SOUTH BLVD, CHARLOTTE, NC, 28209-1950	US Mail (1st Class)
24800	PAUL'S LUMBER DO-IT CENTER, PO BOX 307, 8018 FRENCH ST, GARRETTSVILLE, OH, 44231	US Mail (1st Class)
24800	PAUST PRINTERS, 14 N 10TH ST, RICHMOND, IN, 47374-3142	US Mail (1st Class)
24800	PAXTON PAPER CO INC, 6047 WOODLAND FORREST DR, TUSCALOOSA, AL, 35405-5651	US Mail (1st Class)
24800	PAYCHEX, INC., 4760 RICHMOND RD. STE 100, WARRENSVILLE HTS., OH, 44128	US Mail (1st Class)
24800	PBCC, PO BOX 85460, LOUISVILLE, KY, 40285-5360	US Mail (1st Class)
24800	PBCC, PO BOX 856460, LOUISVILLE, KY, 40285-6460	US Mail (1st Class)
24800	PCS COMPANY, 34488 DOREKA DR., FRASER, MI, 48026	US Mail (1st Class)
24800	PEACOCK WATER CONDITIONING, 1800 MARION MARYSVILLE RD., MARION, OH, 43302	US Mail (1st Class)
24800	PEGUIGNOT SR, RONALD, 3244 KINSMAN RD, NORTH BLOOMFIELD, OH, 44450	US Mail (1st Class)
24800	PENNINGTON, RAY E, 100 MADISON ST, LEETONIA, OH, 44431	US Mail (1st Class)
24800	PENSKE TRUCK LEASING, PO BOX 563, READING, PA, 19603	US Mail (1st Class)
24800	PERFECT WATER/DOLPHIN CAPITAL CORP, PO BOX 605, MOBERLY, MO, 65270	US Mail (1st Class)
24800	PERKINS, DONNA M, 5100 STATE ROUTE 45, BRISTOLVILLE, OH, 44402	US Mail (1st Class)
24800	PERKINS' FLOWERS INC., 323 JEFFERSON ST., LAPEER, MI, 48446	US Mail (1st Class)
24800	PERMAFIX, PO BOX 642998, PITTSBURGH, PA, 15264-2998	US Mail (1st Class)
24800	PERSONNEL CONCEPTS LIMITED, 2865 METROPOLITAN PL, POMONA, CA, 91767-1853	US Mail (1st Class)
24800	PETERS HEAT TREATING, 215 RACE ST, MEADVILLE, PA, 16335-1829	US Mail (1st Class)
24800	PETERSON, CLARA E, 2457 TOD AVE SW, WARREN, OH, 44485	US Mail (1st Class)
24800	PETERSON, RANDALL, 580 RUTH AVE, LEAVITTSBURG, OH, 44430	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	PFP, 68 SUBLER DR, VERSAILLES, OH, 45380-9782	US Mail (1st Class)
24800	PGI SUPPLY, A DOALLL COMPANY, 4436 PAYSHERE CIRCLE, CHICAGO, IL, 60674	US Mail (1st Class)
24800	PHIL HART MANAGEMENT SERVICES INC, 3985 SILVER MEADOW LN, GRAND PRAIRIE, TX, 75052	US Mail (1st Class)
24800	PHILLIPS PLATING CO., 984 NORTH LAKE AVE., PO BOX 72, PHILLIPS, WI, 54555	US Mail (1st Class)
24800	PICKREL BROS. INC., 901 S PERRY ST., DAYTON, OH, 45402	US Mail (1st Class)
24800	PIC-PRODUCTIVITY IMPROVEMENT CENTER, 700 EAST MANDOLINE DR., MADISON HEIGHTS, MI, 48071	US Mail (1st Class)
24800	PILOT AIR FREIGHT, PO BOX 97, 314 N MIDDLETOWN RD, LIMA, PA, 19037-0097	US Mail (1st Class)
24800	PINNACLE PLASTIC PRODUCTS, PO BOX 286, 513 NAPOLEON RD, BOWLING GREEN, OH, 43402	US Mail (1st Class)
24800	PIOLAX CORPORATION, 139 ETOWAH INDUSTRIAL CT, CANTON, GA, 30114-8017	US Mail (1st Class)
24800	PIRTEK-METROPOLITAN DETROIT, 25363 DEQUINDRE RD., MADISON HEIGHTS, MI, 48071	US Mail (1st Class)
24800	PITNEY BOWES CREDIT CORP, FIRST EXPRESS REMITTANCE PROCESSING, PURCHASE POWER, 5101 INTERCHANGE WAY, LOUISVILLE, KY, 40285-6042	US Mail (1st Class)
24800	PITNEY BOWES CREDIT CORPORATION, ATTN RECOVERY DEPT, 27 WATERVIEW DR, SHELTON, CT, 06484-4361	US Mail (1st Class)
24800	PITNEY BOWES INC, 27 WATERVIEW DR, SHELTON, CT, 06484	US Mail (1st Class)
24800	PITNEY BOWES INC., POBOX 856390, LOUISVILLE, KY, 40285-6390	US Mail (1st Class)
24800	PITTSFIELD OF INDIANA, 5741 JACKSON RD, ANN ARBOR, MI, 48103-9504	US Mail (1st Class)
24800	PLASTECH-H&P DIE, 4650 PIEDMAN, BROOKLYN, OH, 44144	US Mail (1st Class)
24800	PLASTEK GROUP, 2425 W 23RD ST, ERIE, PA, 16506-2920	US Mail (1st Class)
24800	PLASTIC ENGINEERING&TECHNICAL, 4141 LUELLA LANE, AUBURN HILLS, MI, 48326	US Mail (1st Class)
24800	PLASTIC PROCESS EQUIPMENT INC, 8303 CORPORATE PARK DR, MACEDONIA, OH, 44056	US Mail (1st Class)
24800	PLASTICS MACHINERY LLC, 141 CLIFFE RUN, FRANKLIN, TN, 37067	US Mail (1st Class)
24800	PLASTOMER, 37819 SCHOOLCRAFT RD, LIVONIA, MI, 48150-1096	US Mail (1st Class)
24800	PLYMOUTH-WAYNE WELDING, 28240 BECK RD, WIXOM, MI, 48393	US Mail (1st Class)
24800	PME COMPANIES, 13870 E 11 MILE RD, WARREN, MI, 48089-1471	US Mail (1st Class)
24800	PME COMPANIES INC, 13870 E ELEVEN MILE RD, WARREN, MI, 48089	US Mail (1st Class)
24800	PNEUMATIC TECHNOLOGY INC., 26673 LAWRENCE, CENTER LINE, MI, 48015	US Mail (1st Class)
24800	POLTA, JEAN L, 326 N PARK AVE, WARREN, OH, 44481	US Mail (1st Class)
24800	POLYMER PRODUCTS GROUP, 100 STATION AVENUE, STOCKERTOWN, PA, 18083	US Mail (1st Class)
24800	POLYONE COLORS GRP./M.A. HANNA, ATTN: MARYANN WATERS - CREDIT, 990 EAST 107TH ST., LEMONT, IL, 60439	US Mail (1st Class)
24800	POLYONE CORP., DEPT. PH 10489, PALATINE, IL, 60055	US Mail (1st Class)
24800	POLYONE DISTRIBUTION, 2600 107TH ST, LEMONT, IL, 60439	US Mail (1st Class)
24800	PORTER SEAL COMPANY, 1301 SOUTH ARARD, DALLAS, TX, 75215	US Mail (1st Class)
24800	POSTLE INDUSTRIES, INC., 18743 SHELDON RD, CLEVELAND, OH, 44130	US Mail (1st Class)
24800	POWER & RUBBER SUPPLY, 1501 35TH ST, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24800	POWER & SIGNAL GROUP, 4670 RICHMOND RD, STE 120, CLEVELAND, OH, 44128	US Mail (1st Class)
24800	PRECISION AUTOMOTIVE PLASTICS, DIV OF WINDSOR MOLD USA INC, 444 HANNA ST EAST, WINDSOR, ON, N8X 2N4CANADA	US Mail (1st Class)
24800	PRECISION STAMPING CO. INC., 1244 GRAND OAKS DRIVE, HOWELL, MI, 48843	US Mail (1st Class)
24800	PRECISION TOOL, DIE & MACHINE, 6901 PRESTON HWY., LOUISVILLE, KY, 40219	US Mail (1st Class)
24800	PREFERRED SOURCING, INC., 3802 N 600 W, GREENFIELD, IN, 46140-9642	US Mail (1st Class)
24800	PREMIER PRINT & SERVICE GROUP, 120 SOUTH RIVERSIDE PLZ, STE 1650, CHICAGO, IL, 60606	US Mail (1st Class)
24800	PRESCOTECH INDUSTRIES, INC, DEPT. #5027, PO BOX 740041, LOUISVILLE, KY, 40201-7441	US Mail (1st Class)
24800	PRIME OFFICE TECHNOLOGIES, 3081 COMMERCE DR., SUITE 300, FORT GRATIOT, MI, 48059	US Mail (1st Class)
24800	PRO TRANS INTERNATIONAL, 8311 N PERIMETER RD, INDIANAPOLIS, IN, 46241-3628	US Mail (1st Class)
24800	PROCESS DEVELOPMENT CORPORATION, 33027 SCHOOLCRAFT RD, LIVONIA, MI, 48150	US Mail (1st Class)
24800	PRODATA COMPUTER SERVICES, 2809 S 160TH STREET, SUITE 401, OMAHA, NE, 68130	US Mail (1st Class)
24800	PRODUCT ACTION INTERNATIONAL, 7998 CENTERPOINT DR, STE 800, INDIANAPOLIS, IN, 46256-3342	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	PRODUCT SUPPORT SOLUTIONS, 46 DONNA LANE, COLUMBUS, MS, 39702	US Mail (1st Class)
24800	PRODUCTION DESIGN SERVICES, INC, 401 FAME ROAD, DAYTON, OH, 45449	US Mail (1st Class)
24800	PRODUCTION RESOURCES INC., 118 SEABOARD LN. SUITE 106, FRANKLIN, TN, 37067-2819	US Mail (1st Class)
24800	PRODUCTION TOOL SUPPLY CO, PO BOX 67000, DEPT #587, DETROIT, MI, 48267-0587	US Mail (1st Class)
24800	PROFORMA, PO BOX 640814, CINCINNATI, OH, 45264-0814	US Mail (1st Class)
24800	PROGRESSIVE COMPONENTS, 5337 PAYSPHERE CIRCLE, CHICAGO, IL, 60674	US Mail (1st Class)
24800	PROGRESSIVE SOFTWARE CORP., 2440 LANCASTER RD., RICHMOND, KY, 40475	US Mail (1st Class)
24800	PROGRESSIVE STAMPING & FAB, 1139 BRITTAIN RD., AKRON, OH, 44305	US Mail (1st Class)
24800	PROSOURCE, 4720 GLENDALE MILFORD RD., CINCINNATI, OH, 45242-3847	US Mail (1st Class)
24800	PROTECTION ONE, 600 HUNTER DR STE 101, OAK BROOK, IL, 60523-8862	US Mail (1st Class)
24800	PROTEK LIMITED, 12260 NORTH ROCKHILL AVE, ALLIANCE, OH, 44601	US Mail (1st Class)
24800	PROWDLEY, TIMOTHY J, 300 MAPLEWOOD DR, APT 5, CORTLAND, OH, 44410	US Mail (1st Class)
24800	PRUDENTIAL, 200 WOOD AVE S, ISELIN, NJ, 08830-2706	US Mail (1st Class)
24800	PSA QUALITY SYSTEMS, BANK ONE WHOLESALE LOCKBOX, ATTN: PSA HOLDINGS 771337, 9000 HAGGERTY RD., BELLEVILLE, MI, 48111	US Mail (1st Class)
24800	PSM INTERNATIONAL, 2800 N LAKE SHORE DR, CHICAGO, IL, 60657-6232	US Mail (1st Class)
24800	PSYCHEMEDICS CORP., 125 NAGOG PARK, ACTON, MA, 01720-3451	US Mail (1st Class)
24800	PSYCHEMEDICS CORPORATION, 125 NAGOG PARK, ACTON, MA, 01720-3451	US Mail (1st Class)
24800	PTM CORPORATION, 6560 BETHUY, FAIR HAVEN, MI, 48023	US Mail (1st Class)
24800	PURCHASE POWER, 5101 INTERCHANGE WAY, LOUISVILLE, KY, 40229-2161	US Mail (1st Class)
24800	PWC INTERNATIONAL, 100 E BROAD ST STE 2100, COLUMBUS, OH, 43215-3607	US Mail (1st Class)
24800	QIS INC AKA QUALITY INDUSTRIAL SRVCS INC, 27481 BEVERLY RD, ROMULUS, MI, 48174	US Mail (1st Class)
24800	QUAD STATE GAUGING & MEASUREMENT, 622 ENTERPRISE DR, CEDAR FALLS, IA, 50613	US Mail (1st Class)
24800	QUALITY ASSISTANCE, LLC, 5119 RESEVE DRIVE, DUBLIN, OH, 43017	US Mail (1st Class)
24800	QUALITY DIE CASTING, 261 EAST GOLDEN GATE, DETROIT, MI, 48203	US Mail (1st Class)
24800	QUALITY INSPECTION & CONTAINMENT INC, PAIGE LEIGH ELLERMAN ESQ, TAFT STETTINIUS & HOLLISTER LLP, 425 WALNUT ST, CINCINNATI, OH, 45202	US Mail (1st Class)
24800	QUALITY PLUS, INC., 36300 EUREKA RD., ROMULUS, MI, 48174	US Mail (1st Class)
24800	QUALITY STEEL, 2221 ARBOR BLVD, DAYTON, OH, 45439	US Mail (1st Class)
24800	QUALITY TECH LOGISTICS, P.O. BOX 907184, GAINESVILLE, GA, 30501	US Mail (1st Class)
24800	QUALITY TOOL REPAIR, 6180 HIGHWAY 31, CALERA, AL, 35040	US Mail (1st Class)
24800	QUALITY VISION, 7220 INVESTMENT DR., CHARLESTON, SC, 29418	US Mail (1st Class)
24800	QUALITY VISION SERVICE INC, 1175 NORTH STREET, ROCHESTER, NY, 14621	US Mail (1st Class)
24800	QUINN, MICHAEL J, 376 S COLONIAL DR, CORTLAND, OH, 44410	US Mail (1st Class)
24800	QW EXPRESS, 135 E ALGONQUIN RD, ARLINGTON HEIGHTS, IL, 60005-5321	US Mail (1st Class)
24800	R S BUSINESS MACHINES INC., 8742 INNOVATION WAY, CHICAGO, IL, 60682-0087	US Mail (1st Class)
24800	RACKLEY OIL INC, PO BOX 826, STARKVILLE, MS, 39760	US Mail (1st Class)
24800	RAE, THOMAS I, 4921 MYRTLE AVE, WARREN, OH, 44483	US Mail (1st Class)
24800	RAF FLUID POWER, INC., 6750 ARNOLD MILLER PARKWAY, SOLON, OH, 44139	US Mail (1st Class)
24800	RAINBOW TAPE & LABEL, INC, 11600 S WAYNE RD., PO BOX 74453, ROMULUS, MI, 48174-0453	US Mail (1st Class)
24800	RALPH C. WILLIAMS, INC., 429 WAYNESURG ROAD SE, P. O. BOX 20527, CANTON, OH, 44701	US Mail (1st Class)
24800	RAMEY, ALBERTA J, 3000 NORTH PARK AVE, WARREN, OH, 44481	US Mail (1st Class)
24800	RAND IMAGINIT TECHNOLOGIES, 5285 SOLAR DR, MISSISSAUGA, ON, L4W 5B8CANADA	US Mail (1st Class)
24800	RANDOLPH TOOL CO., INC., 750 WALES DRIVE SE, HARTVILLE, OH, 44632	US Mail (1st Class)
24800	RAPID GRANULATOR INC, PO BOX 5887, ROCKFORD, IL, 61125-0887	US Mail (1st Class)
24800	RAYMOND STONE, 2922 HEMLOCK DR., ERIE, PA, 16506	US Mail (1st Class)
24800	RD HOLDER OIL COMPNAY, INC, DIV. OF BP PRODUCTS N. AMERICA, INC, 2219 FOLK REAM ROAD, SPRINGFIELD, OH, 45502	US Mail (1st Class)
24800	READY BORING & TOOLING, 32645 FOLSOM RD, FARMINGTON HILLS, MI, 48336-4424	US Mail (1st Class)
24800	RELEASE COATINGS OF NEW YORK, 125 SOUTH BROOKLYN AVE., WELLSVILLE, NY, 14895	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	RELEASE INDUSTRIES, 517 SALTLICK TRCE, PEACHTREE CITY, GA, 30269-1849	US Mail (1st Class)
24800	RELIABLE MINI-STORAGE, 1800 64TH AVENUE, NORTHPORT, AL, 35476	US Mail (1st Class)
24800	REM ELECTRONICS SUPPLY CO, PO BOX 831, 525 S PARK, WARREN, OH, 44483	US Mail (1st Class)
24800	REMINGTON STEEL INC., 1120 S BURNETT RD, SPRINGFIELD, OH, 45505-3408	US Mail (1st Class)
24800	RENT WEAR, INC., 7944 WHIPPLE AVE., N.W., NORTH CANTON, OH, 44720	US Mail (1st Class)
24800	REP CORPORATION, PO BOX 8146, 8N470 TAMELING CT, BARTLETT, IL, 60103-8146	US Mail (1st Class)
24800	REPACORP LABEL PRODUCTS, 31 INDUSTRY PARK CT, TIPP CITY, OH, 45371	US Mail (1st Class)
24800	REPROS INC, 7219 WHIPPLE AVE NW, NORTH CANTON, OH, 44720	US Mail (1st Class)
24800	RESERVE GROUP, SUITE 300, 3560 WEST MARKET STREET, AKRON, OH, 44333	US Mail (1st Class)
24800	RESERVE GROUP MANAGEMENT COMPANY, C/O MARC B MERKLIN, 388 S MAIN ST, STE 500, AKRON, OH, 44311	US Mail (1st Class)
24800	REULAND ELECTRIC, 17969 RAILROAD ST, CITY OF INDUSTRY, CA, 91748-1113	US Mail (1st Class)
24800	REXEL, PO 223356, PITTSBURGH, PA, 15251-2356	US Mail (1st Class)
24800	REXEL SACKS, 6700 LBJ FWY STE 3200, DALLAS, TX, 75240-6526	US Mail (1st Class)
24800	RHODIA, AV. INSURGENTES SUR NO. 1971, TORRE III PISO 6, COLONIA GUADALUPE INN, DELEG. ALVARO OBREGON, C.P., 01020MEXICO	US Mail (1st Class)
24800	RICOH CORPORATION, AMERICAS REGIONAL HEADQUARTERS, 5 DEDRICK PL, WEST CALDWELL, NJ, 07006-6304	US Mail (1st Class)
24800	RILEY, DIANA, 1440 KINSMAN RD, N BLOOMFIELD, OH, 44450	US Mail (1st Class)
24800	RINZELLA, RONALD, 5891 BURNETT RD, LEAVITTSBURG, OH, 44430	US Mail (1st Class)
24800	RIO GRANDE QUALITY SVS, CALLE VISTA AZUL NO. 13, FRACCIONAMIENTO LAS ALAMEDAS, NUEVO LAREDO, TAM., CP, 88274MEXICO	US Mail (1st Class)
24800	ROADWAY EXPRESS INC, RECEIVABLE MANAGEMENT SERVICES, PO BOX 5126, TIMONIUM, MD, 21094	US Mail (1st Class)
24800	ROBBINS, LLC, 3415 THOMPSON ST, MUSCLE SHOALS, AL, 35661-6502	US Mail (1st Class)
24800	ROBECK FLUID POWER CO., 350 LENA DR, AURORA, OH, 44202-8098	US Mail (1st Class)
24800	ROBERT HALF MANAGEMENT RESOURCES, 12400 COLLECTIONS CENTER DRIVE, CHICAGO, IL, 60693	US Mail (1st Class)
24800	ROBERT MANUFACTURING CO., DEPT. 2668, LOS ANGELES, CA, 90084-2668	US Mail (1st Class)
24800	ROBINSON INDUSTRIES INC, 3051 CURTICE RD, COLEMAN, MI, 48618	US Mail (1st Class)
24800	ROBINSON, BARBARA A, 9572 NEWTON FALLS RD, RAVENNA, OH, 44266	US Mail (1st Class)
24800	RON-AL MOLD & MACHINE, 1057 MASON AVE, KENT, OH, 44240-2718	US Mail (1st Class)
24800	RONK'S AUTO & TRUCK TOWING, 5680 U.S. ROUTE 30, CRESTLINE, OH, 44827	US Mail (1st Class)
24800	ROSE PEST SOLUTIONS, 1130 LIVERNOIS RD, TROY, MI, 48083-2711	US Mail (1st Class)
24800	ROSS AUTOMOTIVE SUPPLY, 433 N. SAGINAW ST., LAPEER, MI, 48446	US Mail (1st Class)
24800	ROUPE, LUEVENIA, 661 STATE ROUTE 534 NW, NEWTON FALLS, OH, 44444	US Mail (1st Class)
24800	ROWLAND, RONALD, 4500 HERNER COUNTYLINE RD, SOUTHINGTON, OH, 44470	US Mail (1st Class)
24800	ROYAL CONTAINER CORP, 21100 HUBBELL, OAK PARK, MI, 48237	US Mail (1st Class)
24800	ROYAL SOUND INC, 1455 WINONA RD, STOW, OH, 44224	US Mail (1st Class)
24800	ROYAL-ACME CORPORATION, 3110 PAYNE AVE, CLEVELAND, OH, 44114	US Mail (1st Class)
24800	RTM SALES COMPANY, INC., PO BOX 31545, CHARLOTTE, NC, 28231	US Mail (1st Class)
24800	RTP COMPANY, 580 EAST FRONT STREET, WINONA, MN, 55987	US Mail (1st Class)
24800	RUNNER TOOL & DIE, 1678 MASSILLON RD., AKRON, OH, 44312	US Mail (1st Class)
24800	RUNNING, DAVID, 15692 WEST HIGH ST REAR, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24800	RWJ CORPORATION, BOX 288, 1945 S. 12TH STREET, SEBRINGND, OH, 44672	US Mail (1st Class)
24800	S&K ASPHALT & CONCRETE INC, 2275 MANCHESTER RD, AKRON, OH, 44314-3746	US Mail (1st Class)
24800	S. SLESNICK COMPANY, 700 THIRD ST., S.E., CANTON, OH, 44707-3223	US Mail (1st Class)
24800	SAE, 755 W BIG BEAVER RD STE 1600, TROY, MI, 48084-4906	US Mail (1st Class)
24800	SAFETY-KLEEN SYSTEMS, 5400 LEGACY DR, BLDG 3, PLANO, TX, 75024	US Mail (1st Class)
24800	SAFETY-KLEEN SYSTEMS INC, 5400 LEGACY DR, CLUSTER II BLDG 3, PLANO, TX, 75024	US Mail (1st Class)
24800	SAFETY-KLEEN SYSTEMS INC, 5400 LEGACY DR, CLUSTER 2 BLDG 3, PLANO, TX, 75024	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	SAFETY-KLEEN SYSTEMS INC, 5400 LEGACY DR, CLUSTER II BLDG 3, PLANO, TX, 75024	US Mail (1st Class)
24800	SAFEWAY PACKAGING, 245 BOULDER LN, MINSTER, OH, 45865-1219	US Mail (1st Class)
24800	SANDUSKY ELECTRIC INC, 1516 MILAN RD, SANDUSKY, OH, 44870-4135	US Mail (1st Class)
24800	SANTAMARY FLORISTS, 15460 MAIN MARKET ROAD, BURTON, OH, 44021	US Mail (1st Class)
24800	SANYO PLASTICS COMPOUND AMERICA INC, 2297 NIELS BOHR CT, STE 111, SAN DIEGO, CA, 92154	US Mail (1st Class)
24800	SBC, ACCT: 734-397-5000-9122, BILL PAYMENT CENTER, SAGINAW, MI, 48663	US Mail (1st Class)
24800	SCHAEFER SYSTEMS, 10021 WESTLAKE DR, CHARLOTTE, NC, 28273	US Mail (1st Class)
24800	SCHLEGEL CORP., 742 J A COCHRAN BYP, CHESTER, SC, 29706-3602	US Mail (1st Class)
24800	SCHMIDT SECURITY PRO, 241 MANSFIELD INDUSTRIAL PKWY, MANSFIELD, OH, 44903	US Mail (1st Class)
24800	SCHRADER, DIANN, 15036 WOODSONG DR, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24800	SCHWENDEMAN, STEPHEN R, JONATHAN P BLAKELY, 23 S MAIN ST, 3RD FL, AKRON, OH, 44308	US Mail (1st Class)
24800	SCOT PUMP CO, 77 SW 20TH ST, FORT LAUDERDALE, FL, 33315	US Mail (1st Class)
24800	SCOTT INDUSTRIAL SYSTEMS, 4433 INTERPOINT BLVD, DAYTON, OH, 45424	US Mail (1st Class)
24800	SCOTTISSUE, 3249 DRYDEN ROAD, DAYTON, OH, 45439	US Mail (1st Class)
24800	SEAGATE CONTROL SYSTEMS, 57 N. WESTWOOD, TOLEDO, OH, 43607	US Mail (1st Class)
24800	SEALANT EQUIPMENT, 45677 HELM ST, PLYMOUTH, MI, 48170-6025	US Mail (1st Class)
24800	SECURITAS SECURITY SERVICES USA,INC, 2 CAMPUS DR, PARSIPPNAY, NJ, 07054	US Mail (1st Class)
24800	SEE-BEE DISTRIBUTING CO., 4500 CRAYTON AVE., CLEVELAND, OH, 44104	US Mail (1st Class)
24800	SEFAR AMERICA INC., CHURCH STREET STATION, PO BOX 6771, NEW YORK, NY, 10249-6771	US Mail (1st Class)
24800	SENSMEIER OIL COMPANY, 375 NORTH MAIN ST, MANSFIELD, OH, 44903	US Mail (1st Class)
24800	SERIGRAPH, 3801 DECORAH RD, WEST BEND, WI, 53095-9597	US Mail (1st Class)
24800	SERVICE ELECTRIC SUPPLY INC, 15424 OAKWOOD DR, ROMULUS, MI, 48174	US Mail (1st Class)
24800	SERVICEMASTER PREMIER OF RICHLAND CO INC, PO BOX 216, 16 SHELBY AVE, SHELBY, OH, 44875	US Mail (1st Class)
24800	SERVISOFT OF MIDDLEFIELD, PO BOX 174, 14299 KINSMAN RD., BURTON, OH, 44021	US Mail (1st Class)
24800	SERVO KINETICS INC., 3716 PLAZA DR, ANN ARBOR, MI, 48108-1654	US Mail (1st Class)
24800	SERVTEK MILICRON, MILACRON MARKETING CO., PO BOX 740440, ATLANTA, GA, 30374-0440	US Mail (1st Class)
24800	SEVEN SEVENTEEN CREDIT UNION, 3181 LARCHMONT AVE NE, WARREN, OH, 44483	US Mail (1st Class)
24800	SHERWIN WILLIAMS, 2702 W. 26TH STREET, ERIE, PA, 16506	US Mail (1st Class)
24800	SHETLER, CHESTER J, 16494 S WINE CREEK RD, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24800	SHIP-PAQ INC, 3845 PORT UNION RD, FAIRFIELD, OH, 45014	US Mail (1st Class)
24800	SHIRKS FORKLIFT, 1170 W LONGVIEW AVE, MANSFIELD, OH, 44906-1906	US Mail (1st Class)
24800	SHRADER TIRE & OIL OF MICH, P.O. BOX 13079, TOLEDO, OH, 43613	US Mail (1st Class)
24800	SHUMAN PLASTICS INC, 35 NEOGA ST, DEPEW, NY, 14043-3315	US Mail (1st Class)
24800	SIKA CORPORATION, 14201 BOTTS RD, GRANDVIEW, MO, 64030	US Mail (1st Class)
24800	SILCO FIRE PROTECTION CO, 10765 MEDALLION DR, CINCINNATI, OH, 45241	US Mail (1st Class)
24800	SILMIX DIV-WACKER SILICONE, 2215 INTERNATIONAL PKWY., NORTH CANTON, OH, 44720	US Mail (1st Class)
24800	SIMMONS, MARK J, 3312 SELKIRK BUSH RD, WARREN, OH, 44481	US Mail (1st Class)
24800	SIMPLEX GRINNELL, DEPT. CH 10320, PALATINE, IL, 60055-0320	US Mail (1st Class)
24800	SINGLETON REELS, 11783 TIMBER POINTE TRAIL, MANTUA, OH, 44255	US Mail (1st Class)
24800	SKIMMER INC, 6969 METROPLEX DR, ROMULUS, MI, 48174	US Mail (1st Class)
24800	SKLAR INSTRUMENTS, 889 S MATLACK STREET, WEST CHESTER, PA, 19382	US Mail (1st Class)
24800	SLEYZAK, NOOLOY, PO BOX 155, SOUTHINGTON, OH, 44470	US Mail (1st Class)
24800	SMARTSET AUTOMATION, 2155 NIAGARA LN N STE 110, PLYMOUTH, MN, 55447-4654	US Mail (1st Class)
24800	SMETZ'S TIRE CENTER, INC., 1220 E MAIN STREET, ASHLAND, OH, 44805	US Mail (1st Class)
24800	SMT INDUSTRIES INC, 2540 ROSS ST, SIDNEY, OH, 45365-0766	US Mail (1st Class)
24800	SNYDER BENTLEY, 1119 MANNING ROAD, PO BOX 1316, YOUNGSTOWN, OH, 44501-1316	US Mail (1st Class)
24800	SOFTWARE TECHNOLOGY, 3310 NORFOLK DRIVE, COOKEVILLE, TN, 38506	US Mail (1st Class)
24800	SOLIDEAL TIRE, INC., 3005 5TH AVE. NORTH, BIRMINGHAM, AL, 35203	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	SOLON INDUSTRIAL GRINDING, 1400 MILLER PKWY, STREETSBORO, OH, 44241	US Mail (1st Class)
24800	SOLVAY ADVANCED POLYMERS, 4500 MCGINNIS FERRY RD, ALPHARETTA, GA, 30005-2203	US Mail (1st Class)
24800	SOMERVILLE, DEBORAH L, 7748 STANHOPE-KELLOGGSVILLE RD, WILLIAMSFIELD, OH, 44093	US Mail (1st Class)
24800	SOUTH EAST EXPRESS, INC., 2222 CALE YARBOROUGH HWY, TIMMONSVILLE, SC, 29161	US Mail (1st Class)
24800	SOUTHLAND EQUIPMENT SERVICE INC, 109 N SHORECREST RD, COLUMBIA, SC, 29209	US Mail (1st Class)
24800	SOUTHWESTERN BELL TELEPHONE LP, JAMES CRUDUS ESQ, AT&T INC, ONE AT&T WAY, RM 3A218, BEDMINSTER, NJ, 07921	US Mail (1st Class)
24800	SPARKS BELTING CO, 3800 STAHL DR SE, GRAND RAPIDS, MI, 49546	US Mail (1st Class)
24800	SPARTANBURG STEEL PRODUCTS, INC., DEPT. 142301, PO BOX 67000, DETROIT, MI, 48267-1423	US Mail (1st Class)
24800	SPECIAL MACHINE CONTROLS, 9149 ARTZ RD., NEW CARLISLE, OH, 45344-2202	US Mail (1st Class)
24800	SPECIAL MACHINES, INC., 60880 VAN DYKE RD., WASHINGTON, MI, 48094	US Mail (1st Class)
24800	SPECIALTY BLADES INC, 9 TECHNOLOGY DR, STAUNTON, VA, 24401	US Mail (1st Class)
24800	SPECIALTY RISK SERVICES, DEPT. #5454, PO BOX 30000, HARTFORD, CT, 06150-5454	US Mail (1st Class)
24800	SPECIALTY RISK SERVICES LLC, BANKRUPTCY UNIT T-1-55, HARTFORD PLAZA - TOWER, HARTFORD, CT, 06115	US Mail (1st Class)
24800	SPENCER ELECTRIC INC., 9856 CUTTS RD, CHARDON, OH, 44024-9181	US Mail (1st Class)
24800	SPENCER PRODUCTS CO, 1859 SUMMIT COMMERCE PARK, TWINSBURG, OH, 44087	US Mail (1st Class)
24800	SPINA, ERIN, 3740 SNYDER RD, LEAVITTSBURG, OH, 44430	US Mail (1st Class)
24800	SPIREX CORP., 8469 SOUTHERN BLVD, YOUNGSTOWN, OH, 44512-6709	US Mail (1st Class)
24800	SPIRIT BUSINESS SYSTEMS INC, 100 EDMONTON RD, BELLEVILLE, OH, 44813	US Mail (1st Class)
24800	SPIRIT BUSINESS SYSTEMS INC., 100 EDMONTON RD, BELLEVILLE, OH, 44813	US Mail (1st Class)
24800	SPIRIT SERVICES CO, 2001 ARLINGATE LN, COLUMBUS, OH, 43228	US Mail (1st Class)
24800	SPRADLIN SERVICES COMPANY, 4838 E 355TH ST STE 12, WILLOUGHBY, OH, 44094-4659	US Mail (1st Class)
24800	SPRINT, 2001 EDMUND HALLEY DR, RESTON, VA, 20191-3436	US Mail (1st Class)
24800	SPRINT, 6200 SPRINT PKWY, SHAWNEE MISSION, KS, 66251-6117	US Mail (1st Class)
24800	SPRINTER MARKING, 1805 CHANDLERSVILLE RD, ZANESVILLE, OH, 43701-4604	US Mail (1st Class)
24800	SQUIRE, SANDERS & DEMPSEY, 312 WALNUT ST, CINCINNATI, OH, 45202-4024	US Mail (1st Class)
24800	ST. JOSEPH HEALTH CENTER, 667 EASTLAND AVE SE, WARREN, OH, 44484	US Mail (1st Class)
24800	ST. JUDE RESEARCH HOSP., TIFFANY KINZER, 1335 DUBLIN RD., STE 200C, COLUMBUS, OH, 43215	US Mail (1st Class)
24800	STANDARD PLUMBING & HEATING CO, ATTN HILDA JUMPS, 435 WALNUT AVE SE, CANTON, OH, 44702	US Mail (1st Class)
24800	STANLEY M PROCTOR CO, 2016 MIDWAY DR, TWINSBURG, OH, 44087-1934	US Mail (1st Class)
24800	STAPLES BUSINESS ADVANTAGE, PO BOX 83689, DEPT 2368, CHICAGO, IL, 60690-3689	US Mail (1st Class)
24800	STAR AUTOMATION, N90W14401 COMMERCE DR, MENOMONEE FALLS, WI, 53051-2340	US Mail (1st Class)
24800	STAR ELECTRIC, INC., 9360 INDUSTRIAL TRACE, ALPHARETTA, GA, 30004	US Mail (1st Class)
24800	STAR LEASING COMPANY, L-2738, COLUMBUS, OH, 43260-2738	US Mail (1st Class)
24800	STARK MOLD & PATTERN, INC., 4111 MAHONING ROAD N.E., CANTON, OH, 44705	US Mail (1st Class)
24800	STARK STATE COLLEGE, 6200 FRANK AVE NW, CANTON, OH, 44720-7299	US Mail (1st Class)
24800	STATCARE/WORK HEALTH & SAFETY, 6200 WHIPPLE AVE. NW, NORTH CANTON, OH, 44720	US Mail (1st Class)
24800	STATE CHEMICAL MFG CO., 3100 HAMILTON AVE, CLEVELAND, OH, 44114-3701	US Mail (1st Class)
24800	STERLING SCALE CO., 20950 BOENING, SOUTHFIELD, MI, 48075-5783	US Mail (1st Class)
24800	STERLING, INC, 2900 S 160TH ST, NEW BERLIN, WI, 53151-3606	US Mail (1st Class)
24800	STEWART EFI CONNECTICUT LLC, EULER HERMES ACI, 800 RED BROOK BLVD, OWINGS MILLS, MD, 21117	US Mail (1st Class)
24800	STONER INC, ATTN JON FARRELL, 1070 ROBERT FULTON HWY, QUARRYVILLE, PA, 17566	US Mail (1st Class)
24800	STONERIDGE, ALPHABET DIVISION, 700 INDUSTRIAL DR, PORTLAND, IN, 47371	US Mail (1st Class)
24800	STRONG TOOL COMPANY, 1251 E 286TH ST, CLEVELAND, OH, 44132-2137	US Mail (1st Class)
24800	SUBURBAN PROPANE, 600 E ST. CLAIR ST., EATON, OH, 45320	US Mail (1st Class)
24800	SULFLO, INC, 1158 ERIE AVENUE, BOX 285, NORTH TONAWANDA, NY, 14120	US Mail (1st Class)
24800	SULLIVAN SUPPLY CO., 1433 E. 10TH ST., PO BOX 467, ERIE, PA, 16512	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	SUMMERS RUBBER COMPANY, 12555 BEREA RD, CLEVELAND, OH, 44111	US Mail (1st Class)
24800	SUMMIT DYNAMICS, 2253 E ENTERPRISE PKWY, TWINSBURG, OH, 44087	US Mail (1st Class)
24800	SUMTER PACKAGING CORPORATION, 2341 CORPORATE WAY, SUMTER, SC, 29154	US Mail (1st Class)
24800	SUN PLASTECH, INC., 1140 PARSIPPANY BLVD, PARSIPPANY, NJ, 07054-1880	US Mail (1st Class)
24800	SUNGARD AVAILABILITY SERVICES LP, ATTN MAUREEN A MCGREEVEY ESQ, 680 EAST SWEDESFORD RD, WAYNE, PA, 19087	US Mail (1st Class)
24800	SUNSOURCE/PABCO, NW 7809 P.O. BOX 1450, MINNEAPOLIS, MN, 55485-7809	US Mail (1st Class)
24800	SUPERIOR MEASURING LLC, 1605 EAST AVIS DR, MADISON HEIGHTS, MI, 48071	US Mail (1st Class)
24800	SUPERIOR MOLD AND DIE CO, 449 N. MAIN ST., MUNROE FALLS, OH, 44262	US Mail (1st Class)
24800	SWEEP MASTER INC THE, 2248 STAR CT, ROCHESTER HILLS, MI, 48309	US Mail (1st Class)
24800	SYNCHRONOUS INDUSTRIAL, 10400 TECHNOLOGY DR, COTTONDALE, AL, 35453	US Mail (1st Class)
24800	SYNVENTIVE MOLDING SOLUTI, DEPT. 294401, P.O. BOX 67000, DETROIT, MI, 48267-2944	US Mail (1st Class)
24800	SYNVENTIVE MOLDING SOLUTIONS, DEPT. 294401, PO BOX 67000, DETROIT, MI, 48267-2944	US Mail (1st Class)
24800	SZEMENYEI, MARY D, 11609 KYLE RD, GARRETTSVILLE, OH, 44231	US Mail (1st Class)
24800	T.G.S., INC., 9980 HOBART RD., KIRTLAND, OH, 44094	US Mail (1st Class)
24800	TALENT TREE INC., 312 WILLOW BEND RD, PEACHTREE CITY, GA, 30269-1600	US Mail (1st Class)
24800	TALX, 135 S LASALLE, DEPT. 3065, CHICAGO, IL, 60674-3065	US Mail (1st Class)
24800	TAPEMAN, 48131 WEST RD, WIXOM, MI, 48393-3674	US Mail (1st Class)
24800	TAWNEY, TERRY A, 347 RIDGEWAY PL, NEWTOWN FALLS, OH, 44444	US Mail (1st Class)
24800	TC CONSULTING, 2324 REFSET DR., JANESVILLE, WI, 53545-1373	US Mail (1st Class)
24800	TEAM QUALITY SERVICES, INC., 1675 MOMENTUM PLACE BOX #231675, CHICAGO, IL, 60689-5316	US Mail (1st Class)
24800	TECH MOLD AND TOOL CO., 4333 LISA DRIVE, TIPP CITY, OH, 45371	US Mail (1st Class)
24800	TECHFORM PRODUCTS LTD., 14 CENTENNIAL DRIVE, PENETANGUISHENE, ON, L9M 1R8CANADA	US Mail (1st Class)
24800	TEKNOR APEX COMPANY, 505 CENTRAL AVE, PAWTUCKET, RI, 02861	US Mail (1st Class)
24800	TEL-COM, 3503 FORTUNA DR, AKRON, OH, 44312	US Mail (1st Class)
24800	TELETRONICS, INC., 1200 VALLEY BELT RD, CLEVELAND, OH, 44131	US Mail (1st Class)
24800	TENNANT CO, ATTN: 7TH FLOOR REMITTANCE PROCESSING, TENNANT LOCKBOX 71414, 311 W MONROE, CHICAGO, IL, 60690	US Mail (1st Class)
24800	TENNESSEE STAMPINGS (PORTLAND) LLC, C/O PHILLIP G YOUNG JR, BASS BERRY & SIMS PLC, 315 DEADERICK ST, STE 2700, NASHVILLE, TN, 37238-3001	US Mail (1st Class)
24800	TERMINEX PROCESSING CENTER, 4180 PIER N BLVD, SUITE A, FLINT, MI, 48504	US Mail (1st Class)
24800	TERMINIX INTERNATIONAL, SUITE A, 2680 ROBERTS AVE. NW, NORTH CANTON, OH, 44709	US Mail (1st Class)
24800	TERRY ZAGAR, P O BOX 654, CHARDON, OH, 44024	US Mail (1st Class)
24800	TFT GLOBAL, INC., 500 HIGHWAY #3, TILLSONBURG, ON, N4G 4H5CANADA	US Mail (1st Class)
24800	THE DAYTON SUPPLY & TOOL CO., 507 E 1ST ST, DAYTON, OH, 45402	US Mail (1st Class)
24800	THE DETROIT LIONS, INC., 2000 BRUSH STREET, DETROIT, MI, 48226	US Mail (1st Class)
24800	THE ILLUMINATING COMPANY, ACCT: 110028426614, 76 S MAIN ST, AKRON, OH, 44308-1812	US Mail (1st Class)
24800	THE PAINT SPOT, 501 15TH STREET, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24800	THE RURAL NATURAL GAS CO., 8671 N GILMORE RD, FAIRFIELD, OH, 45014-2104	US Mail (1st Class)
24800	THE ULTIMATE CONTROL, 12838 STAINLESS DRIVE, HOLLAND, MI, 49424	US Mail (1st Class)
24800	THE WORKSHOPS, INC., 2950 WHIPPLE AVE. NW, CANTON, OH, 44708-1534	US Mail (1st Class)
24800	THEM OF OHIO, 16645 GRANITE RD, MAPLE HEIGHTS, OH, 44137	US Mail (1st Class)
24800	THERMAL CARE/MAYER, C/O BANK OF AMERICA, PO BOX 99990, COLLECTION CENTER DR, CHICAGO, IL, 60693	US Mail (1st Class)
24800	THERMAL CORP, 1264 SLAUGHTER RD, MADISON, AL, 35758	US Mail (1st Class)
24800	THI INC, EULER HERMES ACI, 800 RED BROOK BLVD, OWINGS MILLS, MD, 21117	US Mail (1st Class)
24800	THORESON MCCOSH INC., 1885 THUNDERBIRD ST., TROY, MI, 48084	US Mail (1st Class)
24800	TICONA, 8040 DIXIE HWY, FLORENCE, KY, 41042-2904	US Mail (1st Class)
24800	TIERCON INDUSTRIES, 591 ARVIN AVENUE, STONEY CREEK, ON, L8E 5N7CANADA	US Mail (1st Class)
24800	TIGER RIDGE MANUFACTURING INC, PO BOX 1087, WHITEWRIGHT, TX, 75491-5285	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	TIME SERVICES, INC., 6422 LIMA ROAD, FT. WAYNE, IN, 46818	US Mail (1st Class)
24800	TINNERMAN PALNUT, 1060 WEST 130TH ST, BRUNSWICK, OH, 44212	US Mail (1st Class)
24800	TINNERMAN PALNUT ENGINEERED, PRODUCTS, LLC, POB 92368, CLEVELAND, OH, 44193	US Mail (1st Class)
24800	TITAN TOOL COMPANY INC., 7410 WEST RIDGE ROAD, PO BOX H, FAIRVIEW, PA, 16415	US Mail (1st Class)
24800	TOOL CRIB, 3002 INDUSTRIAL PKWY, KNOXVILLE, TN, 37921	US Mail (1st Class)
24800	TOOLS FOR INDUSTRY, 3015 PRODUCTION CT, DAYTON, OH, 45414-3514	US Mail (1st Class)
24800	TOPY PRECISION MFG., INC., 1375 LUNT AVENUE, ELK GROVE VILLAGE, IL, 60007	US Mail (1st Class)
24800	TORAY MARKETING & SALES, 461 5TH AVENUE 9TH FLOOR, NEW YORK, NY, 10017	US Mail (1st Class)
24800	TOSHIBA MACHINE CO., AMERICA, 755 GREENLEAF AVE, ELK GROVE VILLAGE, IL, 60007	US Mail (1st Class)
24800	TOTAL VENDING & COFFEE SERVICE, 13570 WAYNE ROAD, LIVONIA, MI, 48150	US Mail (1st Class)
24800	TOYOTA MOTOR CREDIT CORP, 19001 S WESTERN AVE, TORRANCE, CA, 90501-1106	US Mail (1st Class)
24800	TOYOTA MOTOR CREDIT CORP., 19001 S WESTERN AVE, TORRANCE, CA, 90501-1106	US Mail (1st Class)
24800	TP COMPOSITES INC, 8 CROZERVILLE RD, ASTON, PA, 19014	US Mail (1st Class)
24800	TREASURER, STATE OF OHIO, DIV OF INDUSTRIAL COMPLIANCE, 6606 TUSSING ROAD, REYNOLDSBURG, OH, 43068-9009	US Mail (1st Class)
24800	TREE CITY MOLD MACH, 6752 STATE ROUTE 43 NORTH, KENT, OH, 44240	US Mail (1st Class)
24800	TREHARN, MICHAEL, PO BOX 161, 15822 HIGH ST, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24800	TRENT INC., 201 LEVERINGTON AVE., PHILADELPHIA, PA, 19127	US Mail (1st Class)
24800	TRIAD TECHNOLOGIES LLC, 100 ROCKRIDGE RD, ENGLEWOOD, OH, 45322-2737	US Mail (1st Class)
24800	TRIDON INC, ATTN DAVID ENGLISH, 3200 PARKER DR, ST AUGUSTINE, FL, 32084	US Mail (1st Class)
24800	TRISTAR FIRE PROTECTION, 47810 GALLEON DR, PLYMOUTH, MI, 48170-2468	US Mail (1st Class)
24800	TROYER, ELWIN D, 15371 GEORGIA RD, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24800	TROYER, MATTIE, 14380 OLD STATE RD, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24800	TRUMBULL INDUSTRIES INC, ATTN CREDIT DEPT, 400 DIETZ RD NE, WARREN, OH, 44483	US Mail (1st Class)
24800	TRW FASTENERS DIV., 12025 TECH CENTER DR, LIVONIA, MI, 48150-2122	US Mail (1st Class)
24800	TRW FASTENING SYSTEMS, 12025 TECH CENTER DR, LIVONIA, MI, 48150-2122	US Mail (1st Class)
24800	TURNER, BETTY, 1250 CEDAR ST SW, NEWTON FALLS, OH, 44444	US Mail (1st Class)
24800	TUSCALOOSA FLOWER SHOP, 2208 UNIVERSITY BLVD., TUSCALOOSA, AL, 35401	US Mail (1st Class)
24800	TUSCALOOSA VENDING CO, 1207 17TH STREET, TUSCALOOSA, AL, 35401	US Mail (1st Class)
24800	TWIN CORPORATION, 10456 N. HOLLY RD, HOLLY, MI, 48442-9319	US Mail (1st Class)
	TYCO ELECTRONICS, 2901 FULLING MILL RD, MIDDLETOWN, PA, 17057	US Mail (1st Class)
24800	UGS CORP, 2000 EASTMAN DR, MILFORD, OH, 45150	US Mail (1st Class)
24800	ULINE CO., ATTN: ACCOUNTS RECEIVABLE, 2200 S LAKESIDE DR., WAUKEGAN, IL, 60085	US Mail (1st Class)
24800	UNDERWRITERS LABORATORIES, 333 PFINGSTEN RD, NORTHBROOK, IL, 60062-2002	US Mail (1st Class)
24800	UNIFIRST CORP., 68 JONSPIN RD, WILMINGTON, MA, 01887-1090	US Mail (1st Class)
24800	UNIFORM COLOR COMPANY, 1562 MOMENTUM PLACE, CHICAGO, IL, 60689-5315	US Mail (1st Class)
24800	UNION COUNTY TREASURER, PO BOX 420, MARYSVILLE, OH, 43040	US Mail (1st Class)
24800	UNIQUE FABRICATING INC, 800 STANDARD PKWY, AUBURN HILLS, MI, 48326	US Mail (1st Class)
24800	UNISOURCE WORLDWIDE, INC., 7472 COLLECTION CENTER DRIVE, CHICAGO, IL, 60693	US Mail (1st Class)
24800	UNITED AUTOMATIC HEATING SUPPLY, 399 PHILLIPS AVE, TOLEDO, OH, 43612-1349	US Mail (1st Class)
24800	UNITED INTERMODE, LLC, ONE UNITED DRIVE, FENTON, MO, 63026	US Mail (1st Class)
24800	UNITED PARCEL SERVICE, LOCKBOX 577, CAROL STREAM, IL, 60132-0577	US Mail (1st Class)
24800	UNITED PARCEL SERVICE, RMS BANKRUPTCY RECOVERY SERVICES, PO BOX 5126, TIMONIUM, MD, 21094	US Mail (1st Class)
24800	UNITED SPRING TECHNOLOGY, 24600 NORTH INDUSTRIAL DR, FARMINGTON HILLS, MI, 48335	US Mail (1st Class)
24800	UNITED STEELWORKERS OF AMERICA, LOCAL UNION # 563, INTERNATIONAL SECTREAS, PO BOX 951667, CLEVELAND, OH, 44193	US Mail (1st Class)
24800	UNITED TESTING SYSTEMS, INC., 5171 EXCHANGE DRIVE, FLINT, MI, 48507-2941	US Mail (1st Class)
24800	UNIVERSAL DATA INC, 4150 GRANGER HALL RD, HOLLY, MI, 48442	US Mail (1st Class)
24800	UPPER VALLEY MEDICAL CTR, 3130 NORTH DIXIE HGWY #3011, TROY, OH, 45373	US Mail (1st Class)

Svc Lst	Name and Address of Served Party	Mode of Service
24800	UPS, 55 GLENLAKE PKWY. NE, ATLANTA, GA, 30328	US Mail (1st Class)
24800	UPS CUSTOMHOUSE BROKERAGE, 55 GLENLAKE PKWY NE, ATLANTA, GA, 30328-3474	US Mail (1st Class)
24800	US SAFETYGEAR INC, PO BOX 309, LEAVITTSBURG, OH, 44430	US Mail (1st Class)
24800	USF HOLLAND, 750 E 40TH ST, HOLLAND, MI, 49423	US Mail (1st Class)
24800	USHER OIL COMPANY, 9000 ROSELAWN AVE., DETROIT, MI, 48204	US Mail (1st Class)
24800	UZ ENGINEERED PROD, 3100 HAMILTON AVE, CLEVELAND, OH, 44114-3701	US Mail (1st Class)
24800	VALLEY NATIONAL GASES INC, PO BOX 6628, WHEELING, WV, 26003	US Mail (1st Class)
24800	VALLEY PRODUCTS INC, 1025 BROOKS ST SE, DECATUR, AL, 35601	US Mail (1st Class)
24800	VECTREN, ACCT: 03401512813222427313, ONE VECTREN SQUARE, EVANSVILLE, IN, 47708	US Mail (1st Class)
24800	VECTREN, ACCT: 034001525792328158, ONE VECTREN SQUARE, EVANSVILLE, IN, 47708	US Mail (1st Class)
24800	VEHICLE RESEARCH & DEVELOPMENT, 3863 VAN DYKE RD, ALMONT, MI, 48003	US Mail (1st Class)
24800	VERIZON, ACCT: 814 838 3427 555 42 Y; 814 838 3464 643 23 Y, 27 5430 2618014970 08, 140 WEST ST, NEW YORK, NY, 10007-2109	US Mail (1st Class)
24800	VERIZON, ACCT: 21 1130 2336246414 06, 140 WEST ST, NEW YORK, NY, 10007-2109	US Mail (1st Class)
24800	VERIZON, ACCT: 27 4116 2619265770 03;, 27 4116 2614502375 02, 140 WEST ST, NEW YORK, NY, 10007-2109	US Mail (1st Class)
24800	VERIZON (FIRST COMMUNICATIONS), ACCT: 603419771; 511060331, 140 WEST ST, NEW YORK, NY, 10007-2109	US Mail (1st Class)
24800	VERIZON NORTH, ACCT: 27 5430 2618014970 08;, 27 5430 2676013239 08, 140 WEST ST, NEW YORK, NY, 10007-2109	US Mail (1st Class)
24800	VERIZON NORTH, ACCT: 27 5220 2640072412 04; 27 5220 2645238475 08, 27 5220 2640065254 09; 27 5220 2632209110 06, 140 WEST ST, NEW YORK, NY, 10007-2109	US Mail (1st Class)
24800	VERIZON WIRELESS, ACCT: 511643175; 603465076; 510162418, 140 WEST ST, NEW YORK, NY, 10007-2109	US Mail (1st Class)
24800	VERIZON WIRELESS, GREAT LAKES, PO BOX 25506, LEHIGH VALLEY, PA, 18002-5506	US Mail (1st Class)
24800	VERIZON WIRELESS, ACCT: 603630392, 140 WEST ST, NEW YORK, NY, 10007-2109	US Mail (1st Class)
24800	VERSA MACHINERY, 4850 GREEN CT, ELKHART, IN, 46516	US Mail (1st Class)
24800	VICAS MANUFACTURING CO., 8407 MONROE AVE, CINCINNATI, OH, 45236	US Mail (1st Class)
24800	VICTORY PACKAGING, 3061 W SANER AVE, DALLAS, TX, 75233	US Mail (1st Class)
24800	VIDEOJET TECHNOLOGIES, INC., ATTN: ACCOUNTS RECEIVABLE, 12113 COLLECTION CENTER DRIVE, CHICAGO, IL, 60693	US Mail (1st Class)
24800	VITRAN EXPRESS, PO BOX 7004, INDIANAPOLIS, IN, 46207	US Mail (1st Class)
24800	VULCAN OIL & CHEMICAL CO., 5353 SPRING GROVE AVE., CINCINNATI, OH, 45217	US Mail (1st Class)
24800	VWR SCIENTIFIC INC., 1310 GOSHEN PARKWAY, WEST CHESTER, PA, 19380	US Mail (1st Class)
24800	W ALEXANDRIA WATER DEPT, ACCT: 00301900, 16 NORTH MAIN ST, W. ALEXANDRIA, OH, 45381	US Mail (1st Class)
24800	W W GRAINGER INC, 7300 N MELVINA AVE M240, NILES, IL, 60714-3998	US Mail (1st Class)
24800	W.W. GRAINGER, INC., DEPT 828031229, PALATINE, IL, 60038-0001	US Mail (1st Class)
24800	WACKER SILICONES CORP, 3301 SUTTON RD, ADRIAN, MI, 49221	US Mail (1st Class)
24800	WAGANEK, BARBARA J, 5689 WARREN MEADVILLE RD, CORTLAND, OH, 44410	US Mail (1st Class)
24800	WALKER, GABRIEL, 261 WEST MAIN ST, ANDOVER, OH, 44003	US Mail (1st Class)
24800	WALMEAD, 7400 ALUMAX DR., TEXARKANA, TX, 75501-8694	US Mail (1st Class)
24800	WARN, HOFFMANN MILLER & LALONE PC, 691 N SQUIRREL ROAD, AUBURN HILLS, MI, 48326	US Mail (1st Class)
24800	WARREN FIRE EQUIPMENT, 6880 TOD AVE, WARREN, OH, 44481-8628	US Mail (1st Class)
24800	WASHINGTON PENN PLASTIC COMPANY INC, ATTN KIMBERLY A COLEMAN, LEECH TISHMAN FUSCALDO & LAMPL, 525 WILLIAM PENN PL, 30TH FL, PITTSBURGH, PA, 15219	US Mail (1st Class)
24800	WASTE MANAGEMENT, 2421 W PEORIA AVE, PHOENIX, AZ, 85029	US Mail (1st Class)
24800	WASTE MANAGEMENT, 1001 FANNIN ST STE 4000, HOUSTON, TX, 77002-6711	US Mail (1st Class)
24800	WASTE MANAGEMENT OF ALA, 1001 FANNIN ST STE 4000, HOUSTON, TX, 77002-6711	US Mail (1st Class)
24800	WASTE MANAGEMENT OF OHIO, DAYTON HAULING, PO BOX 9001054, LOUISVILLE, KY, 40290-1054	US Mail (1st Class)
24800	WATKINS GRINDING, 1245 LEONHARD STREET, DAYTON, OH, 45404	US Mail (1st Class)
24800	WATSON WYATT & COMPANY, 1001 LAKESIDE AVE, SUITE 1900, CLEVELAND, OH, 44114-1172	US Mail (1st Class)
24800	WCT-FREMONT, 2800 E SWAGER DR., FREMONT, IN, 46737	US Mail (1st Class)

## **Exhibit C - Solicitation**

Svc Lst	Name and Address of Served Party	Mode of Service
24800	WEAVER, JAKE P, 15078 GEORGIA RD, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24800	WEBER COMPANY, 8140 MAYFIELD RD, CHESTERLAND, OH, 44026	US Mail (1st Class)
24800	WEILER WELDING CO., 324 EAST 2ND STREET, DAYTON, OH, 45402-1759	US Mail (1st Class)
24800	WELCH PACKAGING DETROIT INC, 1020 HERMAN ST, ELKHART, IN, 46516	US Mail (1st Class)
24800	WELLMAN INC, DEPT 952138, ATLANTA, GA, 31192-2138	US Mail (1st Class)
24800	WELLS FARGO FINANCIAL LEASING INC, 400 LOCUST ST, STE 500, MAC F4045-050, DES MOINES, IA, 50309-2331	US Mail (1st Class)
24800	WENDY L. MOORE, 46000 GEDDES #190, CANTON, MI, 48188	US Mail (1st Class)
24800	WESCHLER INSTRUMENTS, 16900 FOLTZ PKWY, STRONGSVILLE, OH, 44149	US Mail (1st Class)
24800	WEST ALEXANDRIA EMS, 8 MARTY LN, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24800	WEST ALEXANDRIA WATER DEPT., PO BOX 265, WEST ALEXANDRIA, OH, 45381	US Mail (1st Class)
24800	WESTERN RESERVE MINI STORAGE, ATTN: RALPH JANOSKE, PO BOX 753, CHARDON, OH, 44024	US Mail (1st Class)
24800	WESTOVER, MARY E, PO BOX 232, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24800	WHITESELL CORPORATION, PO BOX 2570, MUSCLE SHOALS, AL, 35661	US Mail (1st Class)
24800	WHITLAM LABEL COMPANY, INC., 24800 SHERWOOD AVE., CENTER LINE, MI, 48015-1059	US Mail (1st Class)
24800	WHITMAN MOLD INC, 4336 BAMBURG RD, COPE, SC, 29038	US Mail (1st Class)
24800	WICKED WELDING INC, 5665 WEBSTER ST, DAYTON, OH, 45414-3518	US Mail (1st Class)
24800	WILFONG, KAREN D, 1326 BARCLAY MESSERLY RD, LEAVITTSBURG, OH, 44430	US Mail (1st Class)
24800	WILKES, MOLLY L, 151 ARHAVEN DR, NEWTON FALLS, OH, 44444	US Mail (1st Class)
24800	WILLIAMS, JACQUELINE G, 15547 CLOVERDALE DR, MIDDLEFIELD, OH, 44062	US Mail (1st Class)
24800	WILLIS JR, GEORGE E, 6742 IRELAND RD, WINDSOR, OH, 44099	US Mail (1st Class)
24800	WILSON/SHORE INSTRUMENTS, 825 UNIVERSITY AVE, NORWOOD, MA, 02062-2643	US Mail (1st Class)
24800	WOLFE MARKETING CORP, PO BOX 38307, OLMSTED FALLS, OH, 44138	US Mail (1st Class)
24800	WOLFF BROS. SUPPLY INC., 6078 WOLFF ROAD, PO BOX 508, MEDINA, OH, 44256	US Mail (1st Class)
24800	WOLVERINE PRODUCTS, 35220 GROESBECK HWY, CLINTON TOWNSHIP, MI, 48035	US Mail (1st Class)
24800	WOODLANDS OF VAN BUREN, 39670 ECORSE RD, WAYNE, MI, 48184	US Mail (1st Class)
24800	WORKSRIGHT SOFTWARE INC., 501 COBBLESTONE CT, MADISON, MS, 39110	US Mail (1st Class)
24800	WPS ENERGY SERVICES, INC., ACCT: 14057, 1716 LAWRENCE DR, DE PERE, WI, 54115-9108	US Mail (1st Class)
24800	WRIGHTSMAN, JAMES A, 1000 EVERETT HULL RD, CORTLAND, OH, 44410	US Mail (1st Class)
24800	WS SERVICES, 10179 COLONIAL INDUSTRIAL, SOUTH LYON, MI, 48178	US Mail (1st Class)
24800	WW SCHAUB ELECTRIC CO, 501 APPLEGROVE RD NW, NORTH CANTON, OH, 44720	US Mail (1st Class)
24800	XEROX CORPORATION, ATTN LESLEY BANKS, PO BOX 660506, DALLAS, TX, 75266-9937	US Mail (1st Class)
24800	YASUFUKU USA INC, 3201 NW 12TH, LINCOLN, NE, 68521	US Mail (1st Class)
24800	YOHE SUPPLY COMPANY, 499 CHERRY AVE NE, CANTON, OH, 44702	US Mail (1st Class)
24800	YUDO INC, 8163 BUSINESS WAY, PLAIN CITY, OH, 43064	US Mail (1st Class)
24800	YUSA CORPORATION, 151 JAMISON RD., SW, WASHINGTON C.H., OH, 43160	US Mail (1st Class)
24800	ZANDARSKI, BEN, 17546 BEACH LN, LAKE MILTON, OH, 44429	US Mail (1st Class)
24800	ZATKOFF SEALS & PACKINGS, PO BOX 486, FARMINGTON, MI, 48332	US Mail (1st Class)
24800	ZUZAN, KELLY K, 539 WESTGATE BLVD, AUSTINTOWN, OH, 44515	US Mail (1st Class)

Subtotal for this group: 1332

# **EXHIBIT D**

## **Exhibit D - Solicitation**

Svc Lst	Name and Address of Served Party	Mode of Service
24801	A-1 VENDING, 713 FENTON RD, FLINT, MI, 48503	US Mail (1st Class)
24801	BOTTING-THOMPSON REALTY, ROBERT G HANSEMAN ESQ, SEBALY SHILLITO & DYER, 1900 KETTERING TOWER, 40 N MAIN ST, DAYTON, OH, 45423	US Mail (1st Class)
24801	CARLISLE ENGINEERED PRODUCTS INC, C/O CARRIE M BROSIUS ESQ, VORYS SATER SEYMOUR & PEASE LLP, 2100 ONE CLEVELAND CENTER, 1375 EAST NINTH ST, CLEVELAND, OH, 44114	US Mail (1st Class)
24801	CONTRARIAN FUNDS LLC, ATTN ALISA MUMOLA, 411 WEST PUTNAM AVE, STE 225, GREENWICH, CT, 06830	US Mail (1st Class)
24801	DELPHI CORPORATION, C/O MATTHEW E WILKINS, BUTZEL LONG PC, 150 W JEFFERSON, STE 100, DETROIT, MI, 48226	US Mail (1st Class)
24801	GENERAL MOTORS CORPORATION, SARAH SEEWER, HONIGMAN MILLER SCHWARTZ, 2290 FIRST NATIONAL BLDG, 660 WOODWARD AVE, DETROIT, MI, 48226	US Mail (1st Class)
24801	HARTFORD FIRE INSURANCE COMPANY, BANKRUPCY UNIT T-1-55, HARTFORD PLAZA, HARTFORD, CT, $06115$	US Mail (1st Class)
24801	HI-TECH MANUFACTURING, 8687 HWY 22 S, LEXINGTON, TN, 38351	US Mail (1st Class)
24801	HORRELL, WILLIAM, 683 B CENTER ST, WARREN, OH, 44481	US Mail (1st Class)
24801	IRON MOUNTAIN INFORMATION MANAGEMENT INC, R FREDERICK LINFESTY ESQ, 745 ATLANTIC AVE, 10TH FL, BOSTON, MA, 02111	US Mail (1st Class)
24801	KING, CORA M, 910 BEDFORD AVE SW, CANTON, OH, 44710	US Mail (1st Class)
24801	PAARLO PLASTICS INC, PO BOX 2556, 7720 TIM AVENUE NW, NORTH CANTON, OH, 44720-0556	US Mail (1st Class)
24801	PENSION BENEFIT GUARANTY CORPORATION, ATTN GENNICE D BRICKHOUSE ESQ, OFFICE OF THE CHIEF COUNSEL, 1200 K STREET NW, STE 340, WSHINGTON, DC, 20005	US Mail (1st Class)
24801	PENSION BENEFIT GUARANTY CORPORATION, ATTN GENNICE D BRICKHOUSE ESQ, OFFICE OF THE CHIEF COUNSEL, 1200 K STREET NW, STE 340, WASHINGTON, DC, 20005	US Mail (1st Class)
24801	PENSION BENEFIT GUARANTY CORPORATION, ATTN GENNICE D BRICKHOUSE ESQ, OFFICE OF THE CHIEF COUNSEL, 1200 K STREET NW, STE 340, WASHINGTON, DC, 20005	US Mail (1st Class)
24801	TIM FRANK SEPTIC TANK CLEANING, PO BOX 277, 12761 MADISON RD, HUNTSBURG, OH, 44046	US Mail (1st Class)
24801	UNITED STEELWORKERS, DAVID M FUSCO, 1300 EAST NINTH ST, STE 616, CLEVELAND, OH, 44114-1503	US Mail (1st Class)
24801	VISTEON CORPORATION, MICHAEL C HAMMER, DICKINSON WRIGHT PLLC, 301 E LIBERTY, STE 500, ANN ARBOR, MI, 48104	US Mail (1st Class)
24801	WACHOVIA CAPITAL FINANCE CORP (CENTRAL), JEREMY M DOWNS, GOLDBERG KOHN, 55 EAST MONROE ST, STE 3300, CHICAGO, IL, 60603-5802	US Mail (1st Class)

Subtotal for this group: 19

# **EXHIBIT E**

## **Exhibit E - Solicitation**

Svc Lst	Name and Address of Served Party	Mode of Service
24803	ARGO PARTNERS, (TRANSFEROR: WELCH PACKAGING DETROIT INC), 12 WEST 37TH ST 9TH FL, NEW YORK, NY, 10018	US Mail (1st Class)
24803	ASM CAPITAL LP, (TRANSFEROR: ECS SERVICES), 7600 JERICHO TURNPIKE, STE 302, WOODBURY, NY, 11797	US Mail (1st Class)
24803	ATRADIUS TRADE CREDIT INS, (TRANSFEROR: ENTROPEX), ATTN CLAIMS DEPT, 5026 CAMPBELL BLVD STE C, BALTIMORE, MD, 21236	US Mail (1st Class)
24803	CONTRARIAN FUNDS, LLC, (TRANSFEROR: AT&T CORP), 411 WEST PUTNAM AVE, STE 225, GREENWICH, CT, 06830	US Mail (1st Class)
24803	LIQUIDITY SOLUTIONS / REVENUE MANAGEMENT, (TRANSFEROR: MICHIGAN TESTING INSTITUTE INC), ONE UNIVERSITY PLAZA STE 312, HACKENSACK, NJ, 07601	US Mail (1st Class)

Subtotal for this group: 5